BellSouth Telecommunications, Inc. 333 Commerce Street, Suite 2101 Nashville, TN 37201-3300

guy.hicks@bellsouth.com

July 31, 2001

OFFICE SECRETAR Fax 615 214 7406

VIA HAND DELIVERY

Mr. David Waddell Executive Secretary Tennessee Regulatory Authority 460 James Robertson Parkway Nashville, Tennessee 37243-0505

Re: Approval of the Interconnection Agreement, together with the Amendments, Negotiated by BellSouth Telecommunications, Inc. and NewSouth Communications, Corp. Pursuant to Sections 251 and 252 of the Telecommunications Act of 1996.

Docket No. 01-00338 0/-00680

Dear Mr. Waddell:

Pursuant to Section 252(e) of the Telecommunications Act of 1996, NewSouth Communications, Corp. and BellSouth Telecommunications, Inc. are hereby submitting to the Tennessee Regulatory Authority the original and thirteen copies of the attached Petition for Approval of the Interconnection Agreement, together with the Amendments to the Interconnection Agreement. The Amendments incorporate rates established by the Tennessee Regulatory Authority in Dockets 97-01262 and 00-00544. The rates from Docket 00-00544 are subject to true-up once permanent rates for such products and services have been established by the TRA.

Thank you for your attention to this matter.

Sincerely yours,

Guy M. Hicks

GMH/dt

Enclosure

cc: Senior Vice President of Network Planning & Provisioning, NewSouth Communications Corp.

BEFORE THE TENNESSEE REGULATORY AUTHORITY Nashville, Tennessee

In re:

Approval of the Interconnection Agreement and Amendments Thereto Negotiated by BellSouth Telecommunications, Inc. and NewSouth Communications, Corp. Pursuant to Sections 251 and 252 of the Telecommunications Act of 1996

Docket No. 01-00338 0/-00680

PETITION FOR APPROVAL OF THE INTERCONNECTION AGREEMENT AND AMENDMENTS THERETO NEGOTIATED BETWEEN BELLSOUTH TELECOMMUNICATIONS, INC. AND NEWSOUTH COMMUNICATIONS, CORP. PURSUANT TO THE TELECOMMUNICATIONS ACT OF 1996

COME NOW, NewSouth Communications, Corp. ("NewSouth") and BellSouth Telecommunications, Inc., ("BellSouth"), and file this request for approval of the Interconnection Agreement dated May 18, 2001, together with the Amendments to the Interconnection Agreement dated June 20, 2001 (sometimes collectively referred to as the "Agreement") negotiated between the two companies pursuant to Sections 251 and 252 of the Telecommunications Act of 1996, (the "Act"). In support of their request, NewSouth and BellSouth state the following:

1. New South and BellSouth have successfully negotiated an agreement for interconnection of their networks, the unbundling of specific network elements offered by BellSouth and the resale of BellSouth's telecommunications services to NewSouth. The parties have also recently negotiated amendments to the Interconnection Agreement. The first Amendment incorporates rates established by the Tennessee Regulatory Authority ("TRA") in Docket 97-01262. The second Amendment Incorporates rates established by the TRA in Docket 00-00544. The rates from Docket 00-00544 are subject to true-up once permanent rates for such products and services have been established by the TRA. A copy of the Agreement and the Amendments is attached hereto and incorporated herein by reference.

2. Pursuant to Section 252(e) of the Telecommunications Act of 1996, NewSouth and BellSouth are submitting their Agreement to the TRA for its consideration and approval.

3. In accordance with Section 252(e) of the Act, the TRA is charged with approving or rejecting the negotiated Agreement between BellSouth and NewSouth within 90 days of its submission. The Act provides that the TRA may only reject such an agreement if it finds that the agreement or any portion of the agreement discriminates against a telecommunications carrier not a party to the agreement or the implementation of the agreement or any portion of the agreement is not consistent with the public interest, convenience and necessity.

4. NewSouth and BellSouth aver that the Agreement is consistent with the standards for approval.

5. Pursuant to Section 252(i) of the Act, BellSouth shall make the Agreement available upon the same terms and conditions contained therein.

NewSouth and BellSouth respectfully request that the TRA approve the Agreement, including the Amendments, negotiated between the parties.

This 3/57 day of 5, 2001.

Respectfully submitted,

BELLSOUTH TELECOMMUNICATIONS, INC.

By:

Ony M.) Hicks

333 Commerce Street, Suite 2101 Nashville, Tennessee 37201-3300

(615) 214-6301

Attorney for BellSouth

CERTIFICATE OF SERVICE

I, Guy M. Hicks, hereby certify that I have served a copy of the foregoing Petition for Approval of the Interconnection Agreement on the following via United States Mail on this 3/5 day of 3/2001:

Senior Vice President of Network Planning & Provisioning NewSouth Communications, Corp. NewSouth Center Two N. Main Street Greenville, SC 29601

Guly M. Hicks

AMENDMENT TO INTERCONNECTION AGREEMENT BETWEEN BELLSOUTH TELECOMMUNICATIONS, INC. AND NEWSOUTH COMMUNICATIONS CORP. DATED MAY 18, 2001

This Agreement (the "Amendment") is made and entered into between BellSouth Telecommunications, Inc. ("BellSouth") a Georgia corporation, and NewSouth Communications Corp ("NewSouth") a Delaware corporation.

WHEREAS, The Parties desire to amend that certain Interconnection Agreement between BellSouth and NewSouth dated May 18, 2001 (the "Interconnection Agreement") in order to incorporate rates established by the Tennessee Regulatory Authority ("TRA") in Docket Number 00-00544, on September 26, 2000 and November 7, 2000;

NOW THEREFORE, in consideration of the mutual provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, BellSouth and NewSouth hereby convenant and agree as follows:

- 1. Those interim rates established by the TRA in Docket No. 00-00544 for certain Unbundled Network Elements in Tennessee are as set forth in Exhibit 1-TN-Int attached hereto and incorporated herein by this reference.
- 2. To the extent that any rate element set forth in Exhibit 1-TN-Int corresponds to a rate element set forth in the Interconnection Agreement, such rate element in the Interconnection Agreement is hereby deleted and replaced with the corresponding rate element in Exhibit 1-TN-Int. These rates shall be subject to retroactive true-up once permanent rates for such products and services have been established by the TRA.
- 3. To the extent that the existing Interconnection Agreement does not contain terms and conditions for such products and services, then prior to NewSouth's ordering any such elements pursuant to this Amendment, NewSouth and BellSouth shall amend the existing Interconnection Agreement to incorporate such terms and conditions.
- 4. Any rate element in the Interconnection Agreement that is not expressly replaced by the rates set forth in Exhibit 1-TN-Int as described in paragraphs 2 and 3 above shall remain in full force and effect in accordance with the terms of the Interconnection Agreement.
- 5. The Parties agree that all of the other provisions of the Interconnection Agreement, dated July 14, 2000 shall remain in full force and effect.
- 6. The Parties further agree that either or both of the Parties is authorized to submit this Amendment to the Tennessee Regulatory Authority or other regulatory body having jurisdiction over the subject matter of this Amendment, for approval subject to Section 252(e) of the federal Telecommunications Act of 1996.

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed by their respective duly authorized representatives on the data indicated below.

BellSouth Telecommunications, Inc.	NewSouth Communications Corp
(w) Belto	Sele E. 1 in
Signature	Signature
C.W. BOCTZ	Jake E. Jennings
Name	Name
MANASING DIRECTOR	Vice President Regulatory Affairs Title
6-20-01 Date	June 19, 2001 Date

_	_														
										Svc Order	Svc Order	Charge Charge	Charge.	Charge Charge	Charge
UNBUNDLED NETWORK ELEMENT	Permanent	Zone	BCS	USOC				Nonre	Nonrecurring		Submitted	Manual Svc	Manual Svc	•	Manual Svc
						Nonre	Nonrecurring	Disc	Disconnect	per LSR	╄-	Electronic-1st	Electronic-	Electronic-	Electronic-
LOOP MODIFICATION					700	7	Year 1	7.53	7001	SOME	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Unbundled Loop Modification - Load Coil / Equipment Removal per 2 Wire pair - short	_		UAL, UHL,	ULM2L		\$65.40				\$3.50		\$10.00	¢10 00		
Unbundled Loop Modification - Load Coil / Equipment Removal per 2 Wire pair - long	-		OCT OVT, OHIT	ULM2G		\$710.71	77 70					• 10 00	• 10.00		
Unbundled Loop Modification - Load Coil / Equipment Removal per 4 Wire pair - short	-		UCL UAL, UHL,	ULM4I		\$5.40				2 50		• 10.00			
Unbundled Loop Modification - Load Coil / Equipment Removal per 4 Wire pair - long	I		OCT UAL, UHL,	ULM46		\$710.71	\$24.77			63.60		• 10.00	* 10.00		
Unbundled Loop Modification - Bridged Tap Removal, per pair unloaded	-		UAL, UHL, UCL	ULMBT		\$65.44				\$ 3.50		\$19.99	\$19.99		
>LOOPS															
Sub-Loop Distribution															
Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Ser-Up	-		UEANL	USBSC		\$3 13.01						\$20.00	C 10 C 4	613.33	
Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set Up	- -		UEANL	USBSD		\$0.801	\$108.06							200	
Sub-Loop 2-Wire Intrabuilding Network Cable (INC)	-		UEANL	USBR2	\$1.35	\$94.56	\$29.35					\$20.35	\$10.54	\$13.32	
Sub-Loop 4- Wire intrabuilding Network Cable (INC)	-		UEANL	USBR4	\$2.26	\$116.14	\$37.10					\$20.35	\$10.54	\$13.32	
Unbundled Network Terminating Wire (UNTW)															
Unbunded Network Terminating Wire (UNTW) per Pair	-		UENTW	UENPP	\$0.4555	\$2.48	\$2.48								
I MII - Preordering Without Receivation for working facility		-													
queried (Manual).	-			UMKLW		\$100.00	\$100.00	!							
LMU - Preordering Without Reservation, per spare facility queried (Manual)	<u>- 8.</u>			UMKLW		\$ 100.00	\$100.00								
LMU - Preordering With Reservation, per spare facility queried (Manual)	-		-	UMKLP		\$100.00	\$100.00								
LMU - Preordering Without Reservation, per working facility queried (Mechanized)	_					\$0.6888	\$0.6888								
LMU - Preordering Without Reservation, per spare facility queried (Mechanized)	<u> </u>					\$0.6888	\$0.6888								
LMU - Preordering With Reservation, per spare facility queried (Mechanized)	-					\$0.6888	\$0,6888								
Note: Max number of spare facilities per manual LMUSI is 3. Max number of spare facilities per mechanized LMUSI is 10															
LINE SHARING															
2-Wire Analog VG (SL1) for Line Sharing *	P		UEANL	UEAL2	\$13.19	\$31.99	\$20.02					\$20.35	\$10.54	\$13.32	
Line Sharing Splitter															
Line Sharing Splitter, per System 96 Line Capacity	-			ULSDA	\$100.00	\$150.00	\$ 0.00	\$150.00	\$ 0.08	\$3.50		\$19.99	\$19.99	\$1999	00 013
Line Sharing Splitter, per System 24 Line Capacity	-			ULSDB	\$25.00	\$150.00	\$ 0.00	\$150.00	\$ 6.00	\$3.50		\$19.99	\$19.99	\$19.99	\$19.99
Line Sharing Splitter - per Subsequent Activity per Line	-			JUSTO	30.61	¥40.00	\$21.39			\$3.50		\$19.99	\$19.99	\$19.99	\$19.99
Rearrangement	-			ULSDS		\$30.00	\$ 15.00			\$3.50		\$19.99	\$19.99	\$19.99	\$19.99
Note: Interim rates are subject to respective tours															

AMENDMENT TO INTERCONNECTION AGREEMENT BETWEEN BELLSOUTH TELECOMMUNICATIONS, INC. AND NEWSOUTH COMMUNICATIONS CORP. DATED MAY 18, 2001

This Agreement (the "Amendment") is made and entered into between BellSouth Telecommunications, Inc. ("BellSouth") a Georgia corporation, and NewSouth Communications Corp., a Delaware corporation.

WHEREAS, The Parties desire to amend that certain Interconnection Agreement between BellSouth and NewSouth dated May 18, 2001 (the "Interconnection Agreement") in order to incorporate rates established by the Tennessee Regulatory Authority ("TRA") in Docket Number 97-01262, on December 19, 2000, as amended by BellSouth's corrected submissions of January 31, 2001 and February 12, 2001;

NOW THEREFORE, in consideration of the mutual provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, BellSouth and NewSouth hereby convenant and agree as follows:

- 1. Those permanent rates established by the TRA in Docket No. 97-01262 for certain Unbundled Network Elements and Local Interconnection in Tennessee are as set forth in Exhibit 1-TN attached hereto and incorporated herein by this reference.
- 2. To the extent that any product or service set forth in Exhibit 2-TN corresponds to a product or service set forth in the Interconnection Agreement, all rate elements and rates associated with such product or service in the Interconnection Agreement are hereby deleted and replaced with the corresponding rates and rate elements in Exhibit 1-TN.
- 3. Any rate element and rate for products or services in the Interconnection Agreement that is not expressly replaced by the rates and rate elements set forth in Exhibit 1-TN as described in paragraph 2 above shall remain in full force and effect in accordance with the terms of the Interconnection Agreement.
- 4. To the extent NewSouth and BellSouth have not previously negotiated terms and conditions corresponding to any rate element set forth in Exhibit 1-TN, then any Order for such element shall be provisioned in accordance to the terms and conditions set forth in the Competitive Local Exchange Carrier Tariff for the State of Tennessee, incorporated herein by this reference.
- 5. These rates shall be subject to retroactive true-up in accordance with the Agreement. Such true-up shall be retroactive to December 19, 2000.

- 6. The Parti agree that all of the other provis is of the Interconnection Agreement, dated July 14, 2000, shall remain in full force and effect.
- 7. The Parties further agree that either or both of the Parties is authorized to submit this Amendment to the Tennessee Regulatory Authority or other regulatory body having jurisdiction over the subject matter of this Amendment, for approval subject to Section 252(e) of the federal Telecommunications Act of 1996.
- 8. This Amendment is made effective upon the date that it is signed by both Parties.

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed by their respective duly authorized representatives on the data indicated below.

BellSouth Telecommunications, Inc.	NewSouth Communications Corp
Solto Signature	Signature
C.W. BOUTE	Jake E. Jennings
Name	Name
MANAGING DIRECTOR	Vice President of Regulatory Affair
Title	Title
6-20-01	June 19, 2001
Date	Date

ODUF/EODUF Tennessee

Exhibit 1-TN Atachment 1 Exhibit A

L					-	۲		RATES					SSO	OSS RATES		
			-											Presentantel		Incremental
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETWORK ELEMENT AS	TRA Z	o e		SOC.				Nonrecurring	Sve Order	Svc Order	heremental	Charge - Manuel Svc	Incremental Charge - Manual	Charge -
			97-01262			<u> </u>		Nonecuring	<u></u>	Disconnect	Submitted Elec Ber LSR		0 -		Svc Order vs. Electronic-Disc 1st	_
						1_	Jage C	First	Add7	First Add?	_	SOMAN	SOMAN	SOMAN	SOMAN	BOMAN
					H	-		-	┝	-		L				
00	ODUF/EDOUF/CMDS/CNAM-Resate				_			-								
					-											
	OPTIONAL DAILY USAGE FILE (ODUF)					-			_							
		OSS OLEC Daily Usage File		-		-	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,									
	OCCIT: Recording, per message	Hecorong per message		-	-	Y 2	\$0.000044									
								_	_							
		OSS OLEC Daily Usage File		,							_					
	ODUF: Message Processing, per message	Message Distribution per message				¥	\$0.0027368									
		OSS OLEC Daily Usage File			_	-										
	ODUF: Message Processing, per Magnetic Tape Message Distribution per magnetic	Message Distribution per magnetic		_												
	provisioned	tape provisioned		-		¥	\$52.75									
		OSS OLEC Dally Usage File Data		-		_										
	ODUF: Data Transmission (CONNECT:DIRECT), Transmission (Connect: Direct), per	Transmission (Connect: Direct), per														
	per message	message				ΝA	\$0.0000339									
										_						

02009/01

			-	L	L	_		RATES					OSS	OSS RATES		
							_								Incremental Charge -	Incremental
UNBUNDLED NETWORK ELEMENT	LEWENT	UNBUNDLED NETEORIX ELEMENT AS STATED IN DOCKET 97-01262	Docker 97- 2	Zone BC\$	nsoc				Norrecurring	er F	Svc Order Submitted	Svc Order Submitted	Incremental Charge - Manual	Incremental Charge - Manual	Manual Svc Order vs.	Manual Svc Order vs.
							- Form	Norrecurring	Disconnect	15	Per LSR	Manually per LSR	Svc Order vs. Electronic-1st	Svc Order vs. Electronic-Add1	¥	Electronic-Disc Add'i
			1	4	4	ğ.	Fre	Add1	First	Add1	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOWAN
Geographically Degrapes and UNE Zopes and	Zones and			\mid	-											
Geographically Deaveraged UNE Zones and applicable rates have been established for certain	E Zones and hed for certain									-						
services, as shown in this Agreement. Where Geographically Deaveraged UNE Zones and	Former and										-					
applicable rates are established, Statewide rates are	tewide rates are						• • • •									
obsolete. Further, BellSouth is in the process of	the process of			•												
this Geographically Deaveraged UNE Zone Rate	NE Zone Rate															
Structure. Until these enhancements are	ments are															
all services residing in Zones 1, 2, 3 or 4, i.e., once	or 4, i.e., once															
billing enhancements are complete, all applicable	, all applicable															
UNE Zone rates reflected in this Agreement will be	reement will be															
billed. Hererence internet Website http://www.interconnection.bellsouth.com/become cl	vebsite com/become_cl }															
ec/ docs/interconnection/desvuzns.pdf to view	s.pdf to view															
Geographically Deaveraged UNE Zone Designations	ne Designations													· · · · ·		
by Central Office.	_															
	_															
UNBUNDLED EXCHANGE ACCESS LOOP	90c															
					_											
2-WIRE ANALOG VOICE GRADE	LOOP			-												
2-Wire Analog Voice Grade Loop - Service	op · Service	2-Wire Analog Voice Grade Loop			-											
Level 1- Zone 1		Service Level 1 - Zone 1		1 UEANL	NL UEAL2	2 \$13.19	\$31.99	\$20.02	\$10.65	\$1.41			\$20.35	\$10.54	\$13.32	
2-Wire Analog Voice Grade Loop - Service	oob - Service	2-Wire Analog Voice Grade Loop -		I ICANII					40.00	-			400.05	***	613 33	
D.Wire Analog Voice Grade I con - Service	Service .	2-Wire Apalon Voice Grade Loop :		-	-	67./16	\dagger	350.02	CO.01				\$20.33	\$10.01 \$	913.32	
Level 1- Zone 3	POINTS - NO	Service Level 1 · Zone 3		3 UEANL	NL UEAL2	\$22.53	\$31.99	\$20.02	\$10.65	\$1.41			\$20.35	\$10.54	\$13.32	
Manual Order Coordination for UVL-SL1s	UVL-SL18	2-WAVGL - SL1 - Manual Order				_	-									
(per loop)*		Coordination		NEAN	N CEAMO	Q	\$36.52	\$36.52	\$9.18	\$9.18						
Order Coordination for Specified Conversion		Coordination for Specified		T A		77	4.77 30	¢24 20								
2:Wire Analog Voice Grade Loop - Service	Service	COLNECTION OF THE PROPERTY OF		5	_	1	67:50	87.63								
Level 2 w/Loop or Ground Start Signaling - Zone 1		2-Wire Analog Voice Grade Loop - Service Level 2 - Zone 1		1 UEA	A UEAL2	2 \$16.58	\$75.08	\$48.20	\$28.70	\$17.64			\$20.35	\$10.54	\$13.32	
2:Wire Analog Voice Grade Loop - Service	Service															
Level 2 w/Loop or Ground Start Signaling - Zone 2	- Gullangis n	2-Wire Analog Voice Grade Loop - Service Level 2 - Zone 2	•	2 UEA	A UEAL2	\$21.63	\$75.06	\$48.20	\$28.70	\$17.64			\$20.35	\$10.54	\$13.32	
2-Wire Analog Voice Grade Loop - Service	Т			-	 		-	├								
Level 2 w/Loop or Ground Start Signaling -		2-Wire Analog Volce Grade Loop -	•	1154	A LIFAL 2	428.28	475 OR	CAB 20	£28.70	£17 64			\$20.35	£10 54	\$13.32	
0 9 5 5 7				╀	+	╀	3	2					20.03			
Order Coordination for Specified Conversion						,	_									
2.Wire Analog Voice Grade I con - Service	50 A	Conversion 11me		5	2000	3F	25.0	\$34°C3								
Level 2 w/Reverse Battery Signating - Zone		2-Wire Analog Voice Grade Loop -													;	
1		Service Level 2 - Zone 1		1 UEA	A UEAR2	12 \$18.58	\$75.08	\$48.20	\$28.70	\$17.64			\$20.35	\$10.54	\$13.32	
Z-Wire Analog Votce Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zone		2-Wire Analog Voice Grade Loop -											;			
2		Service Level 2 - Zone 2		S CEA	A UEAR2	12 \$21.63	\$75.06	\$48.20	\$28.70	\$.7.8 8.			\$20.35	\$10.54	\$13.32	
2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zone		2-Wire Analog Voice Grade Loop -			A LIFABO	\$08 08	£75.08	648 20	£28 70	21.78			\$20.35	\$10.54	\$13.32	
2				+		\perp	╁	+-	2	5					10:01	
Order Coordination for Specified Conversion Time (per LSB)		Coordination for Specified Conversion Time		UEA	000		\$34.29	\$34.29			•					
/: :>= ;>d/: 2::::!		2000	1	-	٦.	1.5										

			-	-	H		4	RATES					088	OSS RATES	1	
					<u> </u>										_	Incremental Charge
UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97- 2	Zone		2082				Norrecuring	out.	Svc Order Submitted	Svc Order Submitted	Incremental Cherge - Manuel	Incremental Cherge - Manual		Manual Svc Order vs.
		7875			<u></u>		Morrecurring	<u> </u>	Discomed	Ų	Per LSR	Manually per LSM	Sve Order vs. Electronic-1st	Svc Order vs. Electronic-Add7	Electronic-Disc E	Electronic-Disc Add1
			\dashv	-	1	ž	Fret	L PQQ	First	Add1	SOMEC	SOMAN	SOMAN	BOWAN	SOWAN	BOWAN
			+	+												
4-WRE ANALOG VOICE GRADE LOOP	4-Wire Analog Voice Grade Loop -		$\frac{1}{1}$	+												
4-Wire Analog Voice Grade Loop - Zone 1	Zone 1		-	NEA U	UEAL4	\$24.70	\$122.76	\$85.57	\$78.35	\$39.18			\$20.35	\$10.54	\$13.32	
4-Wire Analog Voice Grade Loop - Zone 2	4-Wire Analog Voice Grade Loop - Zone 2		~	UEA	UEAL4	\$32.25	\$122.78	\$85.57	\$76.35	\$39.16			\$20.35	\$10.54	\$13.32	
4-Wire Analog Voice Grade Loop - Zone 3	4-Wire Analog Voice Grade Loop - Zone 3		9	UEA	UEAL4	\$42.17	\$122.78	\$85.57	\$76.35	\$39.16			\$20.35	\$10.54	\$13.32	
Order Coordination for Specified Conversion Time (per LSR)	n 4-Wire AVGL - Order Coordination for Specified Conversion Time			UEA 00	OCOSI		\$34.29	\$34.29								
2:WIRE ISDN DIGITAL GRADE LOOP			+		+											
2-Wire ISDN Digital Grade Loop - Zone 1	2-Wire ISDN Digital Grade Loop - Zone 1		-	NOD	U112X	\$22.00	\$142.78	\$88.88	\$76.35	\$39.16			\$20.35	\$10.54	\$13.32	
2-Wire ISDN Digital Grade Loop - Zone 2	2-Wire ISDN Digital Grade Loop - Zone 2		2	n QD T	V1L2X	\$29.02	\$142.78	\$88.88	\$78.35	\$39.18			\$20.35	\$10.54	\$13.32	
2-Wire ISDN Digital Grade Loop - Zone 3	2-Wire ISDN Digital Grade Loop - Zone 3		<u>۔</u>	n Ngn	U1L2X	\$37.95	\$142.78	\$88.88	\$78.35	\$39.16			\$20.35	\$10.54	\$13.32	
Order Coordination For Specified Conversion Time (per LSR)	2-Wire ISDN Digital Grade Loop - Order Coordination for Specified Conversion Time		٠,	NQN	COSL		\$34.29	\$34.29								
			+													
2-WARE ASYMME RICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOUP 2-WIRE ADSL Compatible Loop Ingluding	SER LINE (ADSL) COMPATIBLE LO	3			-											
Man Svc Inquiry and racility reservation :	Zone 1		-	UAL	UAL2X	\$13.82	\$270.01	\$234.63	\$74.54	\$39.14			\$20.35	\$10.54	\$13.32	
2-Wire ADSL Compatible Loop Including. Man Svc Inquiry and Facility Reservation - Zone 2	2-Wire ADSL Compatible Loop - Zone 2		~	UAL	UAL2X	\$18.05	\$270.01	\$234.63	\$74.54	\$39.14			\$20.35	\$10.54	\$13.32	
2-Wire ADSL Compatible Loop Including. Man Svc Inquiry and Facility Reservation - Zone 3	2-Wire ADSL Compatible Loop - Zone 3			n de	UAL2X	\$23.60	\$270.01	\$234.63	\$74.54	\$39.14			\$20.35	\$10.54	\$13.32	
Order Coordination for Specified Conversion Time	2-Wire ADSL Digital Grade Loop - order Coordination for Specified Conversion Time			8	OCOSL		\$34.29	\$34.29	-							
2-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP	ER LINE (HDSL) COMPATIBLE LOO	٩														
2-Wire HDSL Compatible Loop Including, manual service Inquiry and facility reservation - Zone 1	2-Wire High Bit Rate DSL Compatible Loop - Zone 1			UHL UHL	CHL2X	\$10.83	\$270.01	\$234.63	\$74.54	\$39.14			\$20.35	\$10.54	\$13.32	
2-Wire HDSL Compatible Loop <u>Including</u> manual service inquiry and facility reservation - Zone 2	2-Wire High Bit Rate DSL Compatible Loop - Zone 2		~	포	UHL2X	\$14.15	\$270.01	\$234.63	\$74.54	\$39.14			\$20.35	\$10.54	\$13.32	
2-Wire HDSL Compatible Loop Including manual service inquiry and facility reservation - Zone 3	2-Wire High Bit Rate DSL Compatible Loop - Zone 3		8	UH	UHL2X	\$18.50	\$270.01	\$234.63	\$74.54	\$39.14			\$20.35	\$10.54	\$13.32	
Order Coordination for Specified Conversion Time				UH.	JSOOO		\$34.29	\$34.29			1	بمردهون				
	O 19 Idit Auto College		+	$\frac{1}{1}$	†											
4-WHE HIGH BIT HAS E DIGITAL SUBSCRIBER LINE (NOS.) COMPATIBLE COOR AND HE HIGH HOS. COMPATIBLE COOR MICHAELTS (NOS.) (A WIRE HOS! Compatible Loop Manaus service inquiry and facility	4-Wire HDSL Compatible Loop															
reservation - Zone 1	Zone 1		-	품	UHL4X	\$13.93	\$279.60	\$244.22	\$74.54	\$39.14			\$20.35	\$10.54	\$13.32	

				\vdash	L	H		2	RATES					058	OSS RATES		
						<u> </u>			-								Incremental Charge -
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORIX ELEMENT AS	Docker 97.	Zone	So	2080				Moreocurine		Submitted		Incremental Charge - Manuel	Incremental Cherge - Manual	Menual Svc Order vs.	Mamail Svc Order vs.
•		SALED BOOKE STORE	252			<u> </u>		Nonrecuring	130	Disconnect			Manuelly per LSR	Svc Order vs. Electronic-1st	Bvc Order vs. Electronic-Add [*]	ŝ	Electronic-Disc Add1
_				\dashv	-	Н	ě	Firet	Pdd:	First	Add1	П	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	4-Wire HDSL Compatible Loop Including manual service inquiry and facility reservation - Zone 2	4-Wire HDSL Compatible Loop - Zone 2		7 C	UHL	UHL4X	\$18.20	\$279.60	\$244.22	\$74.54	\$39.14			\$20.35	\$10.54	\$13.32	
	4-Wire HDSL Compatible Loop Including manual service inquiry and facility reservation - Zone 3	4-Wire HDSL Compatible Loop - Zone 3		9	UHL	UHL4X	\$23.80	\$279.60	\$244.22	\$74.54	\$39.14			\$20.35	\$10.54	\$13.32	
	Order Coordination for Specified Conversion Time	4-Wire HDSL Loop Order Coordination for Specified Conversion Time		ر	ZH.	nsoco		\$34.29	\$34.29								
14	4-WIRE DS1 DIGITAL LOOP			-													
	4-Wire DS1 Digital Loop - Zone 1	4-Wire DS1 Digital Loop - Zone 1		-	usr us	USLXX	\$57.73	\$313.08	\$219.72	\$96.86	\$40.45			\$18.98	\$8.43	\$11.95	\$0.00
	4-Wire DS1 Digital Loop - Zone 2	4-Wire DS1 Digital Loop - Zone 2		ر ~	nsr nsr	NSUXX	\$75.40	\$313.08	\$219.72	\$96.86	\$40.45			\$18.98	\$8.43	\$11.95	\$0.00
	4-Wire DS1 Digital Loop - Zone 3	4-Wire DS1 Digital Loop - Zone 3			USL	nscxx	\$98.59	\$313.08	\$219.72	\$96.86	\$40.45			\$18.98	\$8.43	\$11.95	\$0.00
	Order Coordination for Specified Conversion Time	4-Wire DS1 Loop - Order Coordination for Specified Conversion Time			OSt.	OCCOSIL		\$34.59	\$34.59								
1	WAS ER OF ALKERS DICTAL COADE LOOP			$\frac{1}{1}$	+	+											
	4-Wire 58 Kbps Digital Grade Loop - Zone 1	4-Wire 56 or 64 Kbps Digital Grade Loop - Zone 1		-	ID C	UDL56	\$31.10	\$207.01	\$141.38	\$90.70	\$44.18			\$20.35	\$10.54	\$13.32	
				7	יט תטר	UDLS6	\$40.61	\$207.01	\$141.38	\$90.70	\$44.18			\$20.35	\$10.54	\$13.32	
		4-Wire 56 or 64 Kbps Digital Grade Loop - Zone 3		3 (nDr.	UDLSB	\$53.11	\$207.01	\$141.38	\$90.70	\$44.18			\$20.35	\$10.54	\$13.32	
		4-Wire 56/64 kbps Dig. GL · Order Coordination for Specified Conversion Time			Z Z	OCOSL		\$34.29	\$34.29								
	4-Wire 64 Kbps Digital Grade Loop - Zone 1	4-Wire 56 or 64 Kbps Digital Grade Loop - Zone 1		-	ת תם	UDL64	\$31,10	\$207.01	\$141.38	\$90.70	\$44.18			\$20.35	\$10.54	\$13.32	
				~	n nor	MDL64	\$40.61	\$207.01	\$141.38	\$90.70	\$44.18			\$20.35	\$10.54	\$13.32	
	4-Wire 64 Kbps Digital Grade Loop - Zone 3			<u>د</u>	UDF	UDL64	\$53.11	\$207.01	\$141.38	\$90.70	\$44.18			\$20.35	\$10.54	\$13.32	
	Order Coordination for Specified Conversion Time	4-Wire 56/84 kbps Dig. GL - Order Coordination for Specified Conversion Time		-	S P	OCOSE		\$34.29	\$34.29								
SUB-LOOPS	OOPS																
	in the District Advantage			+	+												
n	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-Up		•	5	UEANL	USBSA		\$517.25	\$517.25					\$20.35	\$10.54	\$13.32	
<u></u>	Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-Up		•	5	UEANL	USBSB		\$42.68	\$42.68					\$20.35	\$10.54	\$13.32	
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statewide	Loop Distribution - per 2-WAVGL		sw U	UEANL	USBN2	\$10.02	\$148.84	\$112.34	\$73.14	\$36.65			\$20.35	\$10.54	\$13.32	
	Sub-Loop Distribution - Order Coordination, per sub-loop pair		•	5	UEANL	USBMC		\$36.52									
	Sub-Loop Distribution - Order Coordination for Specified Time Conversion, per LSR	Sub-Loop Distribution - Order Coordination for Specified Conversion Time		5	UEANL	CCOST		\$34.29	-								
				1	1	1											

03/09/01

Exhibit 1-14 Attachment ? Exhibit G

-				-	-	-		۵	PATES					980	OSS RATES		
						J						-				1	Incremental Charge -
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Docker 97- 2	Zone		S S				Nomecuming	guiring	Svc Order Submitted	Submitted	Incremental Charge - Manuel	Incremental Charge - Manuel	Manual Sve Order vs.	Manuel Svc Order vs.
								Nomecu	guling	Disconnect	rect	Elec Per LSR	Manually per LSR	Svc Order vs. Electronic-1st	Svc Order vs. Electronic-Add71	¥	Addi
				+	+	\dagger	ž	First	Addi	Fret	LPQ41	SOMEC	SONAN	SOMAN	SOMAN	SOMAN	SOMAN
no.	Sub-Loop reduct			-		+											
	DSO Set-up per Cross Box location - CLEC Distribution Facility set-up (for access to Feeder)		•	<u> </u>	UEA. UDN,UC L,UDL US	USBFW		\$517.25	\$517.25					\$20.35	\$10.54	\$13.32	
	DSO Set-up per Cross Box location - per 25 pair pasnel set-up (for access to Feeder)		•	> <u>₽</u> Ξ	UEA, UDN,UC L,UDL	USBFX		\$42.68	\$42.68					\$20.35	\$10.54	\$13.32	
	Sub-Loop Feeder- Per 2-Wire Analog Volce Grade Ground-Start Loop - Statewide	Loop Feeder per 2-WAVGL		⊃ *s	ne.	USBFA	\$12.05	\$122.24	\$85.05	\$76.35	\$39.16			\$20.35	\$10.54	\$13.32	
	Sub-Loop Feeder- Per 2-Wire Analog Voice Grade Loop-Start Loop - Statewide	Loop Feeder per 2-WAVGL		sw U	UEA US	USBFB	\$12.05	\$122.24	\$85.05	\$76.35	\$39.18			\$20.35	\$10.54	\$13.32	
	Sub-Loop Feeder - Per 2-Wire Analog Voice Grade Reverse Battery Loop - Statewide	Loop Feeder per 2-WAVGL		sw U	UEA US	USBFC	\$12.05	\$122.24	\$85.05	\$76.35	\$39.16			\$20.35	\$10.54	\$13.32	
	Sub-Loop Feeder - Order Coordination for Specified Conversion Time, per LSR	Sub-Loop Feeder - Order Coordination for Specified Conversion Time		ם	UEA OC	18000		\$34.29									
	VIII			+	+	+											
2	Network Interface Device (NID)	NID cer 2-Wire Loco		Ü	UENTW	UNDC2	\$1.15	\$0.74				\$3.50		\$19.99	\$19.99	\$19.99	\$19.99
	Network Interface Device Cross Connect	NID per 4-Wire Loop		E E		UNDC4	\$1.27	\$0.74				\$3.50		\$19.99	\$19.99	\$19.99	\$19.99
NO.	Unburbleb Loop Concentration	Loop Channelization system - DLC (Inside CO)		-) 5) 5) 5	SOOTA	\$307.07	\$307.34	\$74.37	\$4.18		\$3.50		\$20.35	\$10.54	\$13.32	
	CO Channel Interface - 2-Wire Voice Grade			-	\vdash	ULCC2	\$1.20	\$9.57	\$9.52	\$8.66	\$8.60	\$3.50		\$20.35	\$10.54	\$13.32	
	Unbundled Loop Concentration - Channel Interface-2 Wire Voice-Loop Start or Ground Start	CO Channel Interface - 2-Wire Voice Grade	***	2	UEA	ULCC2	\$1.20	\$9.57	\$9.52	\$8.88	\$8.60	\$3.50		\$20.35	\$10.54	\$13.32	
UNBUN	UNBUNDLED SUB-LOOP CONCENTRATION (OUTSIDE CO)	DE CO)				\parallel											
	Loop Concentration - Channelization System (CO)	Loop Concentration - Channelization System (Outside CO)		-	78D CK	UCT8A	\$328.28	\$651.09	\$283.42	\$207.92	\$50.94	\$3.50		\$20.35	\$10.54	\$13.32	
	Remote Channel Interface - 2-Wire Voice Grade Loop	Loop Concentration - Remote Channel Interface - 2-WAVGL (Outside CO)			180 12	ULCC2	\$0.88	\$9.43	\$9.40	\$4.71	\$4.70	\$3.50		\$20.35	\$10.54	\$13.32	
	Loop Concentration - Remote terminal Cabinet (Outside CO)	Loop Concentration - Remote Terminal Cabinet (Outside CO)					82					\$3.50					
	Channel Interface - 2 Wire Voice-Loop Start or Ground Start	Loop Concentration - Remote Channel Interface - 2-WAVGL (Outside CO)		-	180 12	ULCC2	\$0.88	\$9.43	\$9.40	\$4.71	\$4.70	\$3.50	-	\$20.35	\$10.54	\$13.32	
	OTO COLUMN STATE S	re)		+		+											
2000	DLEU LOCAL EXCHANGE SWITCHING(POR	/61		+	-	+											
ă	Exchange Ports				$\ \cdot \ $												
	Cuthange Date 2 Wite Angled I he Dod.	Evrhance Dorte , 2.Wire Analon			- ··												
	- 1	Line Port (Res./Bus.)		픠	UEPSR UEPRL	PRL	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40

F							_	RATES					OSSF	OSS RATES		
			ţ,								-				Incremental Charge -	Incremental Charge -
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK BLEMENT AS 8TATED IN DOCKET 97-01262	Docket 97- Zone	Ş	20\$0				Nonrecurring	urring	Svc Order Submitted		Incremental Charge - Manual	Incremental Charge - Manuel		Manual Sve Order vs.
			_				Movement	-	Democrato	į	2 2	Manuelly per	Svc Order vs.	Svc Order vs. Flactmotics Addit	Electronic-Disc	Electronic-Disc
_						ž	First	Add:	First	Lppy	SOMEC		SOMAN	SOMAN	SOMAN	SOMAN
2	Note: Although the Portrate includes all available vertical features in TN, the desired feature(s	s vertical features in TN, the desired	feature(s) will	need to be	e ordered.	(
	Exchange Ports - 2-Wire Analog Line Port with Caller ID - Res.	Exchange Ports - 2-Wire Analog Line Port (Res./Bus.)		UEPSH	UEPAC	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Res.	Exchange Ports - 2-Wire Analog Une Port (Res./Bus.)		UEPSH	UEPRO	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	Exchange Ports - 2-Wire VG unbundled TN extended local dialing parity Port with Caller ID - Res.	Exchange Ports - 2-Wire Analog Une Port (Res./Bus.)		UEPSR	UEPAO	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	Exchange Ports · 2-Wire VG unbundled Tennessee Area Calling port with Caller ID · Res (F2R)	+		UEPSR		\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	Exchange Ports - 2-Wire VG unbundled Tennessee Area Calling port with Caller ID - Res (TACER)	_		UEPSR	UEPAL	\$1.89	\$9.93	\$9.19	\$3.68	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	Exchange Ports - 2-Wire VG unbundled Tennessee Area Calling port with Caller ID - Res (TACSR)	Exchange Ports · 2-Wire Analog Une Port (Res./Bus.)		UEPSR	UEPAM	\$1.89	\$9.93	\$9.19	\$3.68	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	Exchange Ports - 2-Wire VG unbundled Tennessee Area Calling port with Caller ID - Res (1MF2X)	Exchange Ports - 2-Wire Analog Line Port (Res./Bus.)		UEPSR	UEPAN	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	Exchange Ports - 2-Wire VG unbundled Tennessee Area Calling port with Caller ID - Res (2MR)	Exchange Ports - 2-Wire Analog Line Port (Res./Bus.)		UEPSR	UEPAO	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LUM)	Exchange Ports - 2-Wire Analog Une Port (Res./Bus.)		UEPSR	UEPAP	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
H	Subsequent Activity		•	UEPSR	USASC		\$10.00	\$10.00								
	Al Available Vertical Features	[Exchange Ports Includes all Applicable Features.]		UEPSR	UEPVF	\$0.00	\$0.00	\$0.00								
	Exchange Ports - 2-Wire Analog Line Port without Caller ID - Bus	Exchange Ports - 2-Wire Analog Une Port (Res./Bus.)		UEPSB	UEPBL	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	Exchange Ports - 2-Wire VG unbundled Line Port with unbundled port with Caller+E484 ID Exchange Ports - 2-Wire Analog - Bus.	Exchange Ports • 2-Wire Analog Une Port (Res/Bus.)		UEPSB	UEPBC	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Bus.	Exchange Ports - 2-Wire Analog Une Port (Res./Bus.)		UEPSB	UEPBO	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	Wire VG unbundled TN ng parity Port with Caller	Exchange Ports - 2-Wire Analog Line Port (Res./Bus.)		UEPSB	UEPAV	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
 	Exhange Ports - 2-Wire VG unbundled incoming only port with Caller ID - Bus	Exchange Ports - 2-Wire Analog Une Port (Res./Bus.)		UEPSB	UEPB1	\$1.89	89.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	Exchange Ports - 2-Wire VG unbundled TN Bus 2-Way Area Calling Port Economy Option - Bus (TACC1)	Exchange Ports - 2-Wire Analog Line Port (Res./Bus.)		UEPSB	UEPAC	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	Exchange Ports - 2-Wire VG unbundled TN Bus 2-Way Area Calling Port Standard Option - Bus (TACC2)	Exchange Ports - 2-Wire Analog Une Port (Res/Bus.)		UEPSB	UEPAD	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	Exchange Ports-2-Wire VG unbundled TN Bus 2-Way Collierville & Memphis Local Calling Port-Bus (82F)	Exchange Ports - 2-Wire Analog Line Port (Res./Bus.)		UEPSB		\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	Subsequent Activity		•	UEPSB	USASC		\$10.00	\$10.00								
	All Available Vertical Features	[Exchange Ports includes all Applicable Features.]		UEPSB	UEPVF	\$0.00	\$0.00	\$0.00								
	Exchange Ports · 4·Wire Analog Voce Grade Port	Exchange Ports - 4-Wire Analog Volce Grade Port			UEP4A	\$8.27	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
-	Subsequent Activity		•		USASC		\$10.00	\$10.00								

Page 5 of 29

							6	4750					OSS BATES	ATES		
			_				1	KAIES		T	-			-	Incremental	Incremental
													;	,	_	Charge.
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS	Not in TRA Docket 97- Zone	S	2080				Manage	1	Submitted	Submitted	Charge - Manual	Incremental Charge - Manuel	Manual Svc Order ve.	Manual Sve Order vs.
			01262				;	-!	NO.				Svc Order vs.	Sve Order vs.	ž	Electronic-Olac
						Pec	First	Add1	First	Add1	SOMEC	SOMAN	17	SOMAN	SOMAN	SOMAN
Alf Avail	Al Available Vertical Features	[Exchange Ports includes all Applicable Features.]			UEPVF	\$0.00	\$0.00	\$0.00								
Exitiang	Port	Exchange Ports - 2-Wire DID Port		UEPEX	UEPP2	\$8.97	\$47.75	\$47.01	\$9.21	\$8.47			\$20.35	\$10.54	\$13.32	\$1.40
Subsec			•	UEPEX	USASC		\$10.00	\$10.00								
N A		Exchange Ports includes all Applicable Features.]		UEPEX	UEPVF	\$0.00	\$0.00	\$0.00								
Erstan Parist	Extrange Ports - DDITS Port - 4-Wire DS1	Exchange Ports - 4-Wire DID Port		UEPDD	UEPDD	\$35.74	\$75.93	\$38.15	\$8.77	\$8.04			\$20.35	\$10.54	\$13.32	\$1.40
Subs and			*	UEPDD	USACP		\$10.00	\$10.00								
Exchinge Por	Wire ISDN Port (See	Exchange Ports - 2-Wire ISDN Port		UIPMA	U1PMA	\$16.26	\$30.23	\$29.49	\$4.10	\$4.10			\$41.43	\$42.17	\$9.80	\$9.80
Suhred	ctivity		•	UEPDD	USASC		\$10.00	\$10.00								
All Avail	All Available Vertical Features	[Exchange Ports includes all Applicable Features.]		UEPDD	UEPVF	\$0.00	\$0.00	\$0.00								
with POTS to drout so data transmi	NOTE: 1 ransmission/usage charges associated with PoTS circuit switched usage will also apply to circuit switched voice and/or circuit switched data transmission by Buhannels associated with chart transmission by Buhannels associated with		-													
NOT(:) Packe: cupi BFR/16, w B	NOTI: Access to Bicharperia Packet: Capabilities will be available only through BFM is a Business Request Process. Rates for the packet capabilities will be determined via the															
Exchan	Exchange Ports - 2-Wire ISDN Port		•		U1UMA	\$0.00	\$0.00	\$0.00								
Fraise	de Ports - 4-Wire ISDN DS1 Port	Exchange Ports - 4-Wire ISDN DS1 Port		UEPEX	_	\$75.04	\$148.68	\$147.18	\$38.46	\$36.98			\$40.69	\$42.17	\$9.07	\$10.54
2. 4.10	2. Vire VG Unbundled 2-Way PBX Trunk -	Exchange Ports - 2-Wire Analog Une Port (PBX)		UEPSE	UEPRD	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2.V'i VGT Trunk - Bus	2-V'i VG Line Side Unbundled 2-Way PBX Trunk - Bus	Exchange Ports Line Port (PBX)		UEPSP	UEPPC	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2 Wire PBX Ta	2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus	Exchange Ports - 2-Wire Analog Une Port (PBX)		UEPSP	UEPPO	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire PBX Tn	2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus	Exchange Ports - 2-Wire Analog Line Port (PBX)		UEPSP	UEPP1	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire Ana Trunk - Bus	2-Wire Analog Long Distance Terminal PBX Trunk - Bus			UEPSP	UEPLD	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire Anal Trunk - Bus	2-Wire Analog TN 2-Way Calling Plan PBX Trunk - Bus	Exchange Ports - 2-Wire Analog Une Port (PBX)		UEPSP	UEPT2	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire Bus	2-Wire TN Outward Calling Plan PBX Trunk - Bus	Exchange Ports - 2-Wire Analog Une Port (PBX)		UEPSP	UEPTO	\$1.79	\$9.93	\$9.19	\$3.86	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire	2-Wire Voice Unbundled PBX LD Terminal Ports	•		UEPSP	UEPLD	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-wire	2-Wire Voice Unbundled 2-Way PBX Tennessee Calling Port	Exchange Ports - 2-Wire Analog Line Port (PBX)		UEPSP	UEPT2	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-V/ire	2-Wire Voice Unbundled 1-Way Outgoing PRY Tennessee Calling Port	Exchange Ports - 2-Wire Analog Line Port (PBX)		UEPSP	UEPTO	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire Port	2-Wire Vice Unbundled 2-Way PBX Usage Port	Exchange Ports - 2-Wire Analog Line Port (PBX)		UEPSP	UEPXA	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire Voic Hotel Ports	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports	Exchange Ports - 2-Wire Analog Une Port (PBX)		UEPSP	UEPXB	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
2-Wire Termin	2-Wire Voice Unbundled PBX LD DDD Terminals Port	Exchange Ports - 2-Wire Analog Line Port (PBX)		UEPSP	UEPSP UEPXC	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40

				L	L			RATES					1880	OSS RATES		
															Incremental	Incremental
	TOTAL SE SECULTAR CE STANDARD	UNBUNDLED NETEORK ELEMENT AS	Not in TRA							T	Svc Order	Svc Order	Incremental	Incremental	Menue	Menual Sys
_	ORGANICED NETWORK ELEMENT	STATED IN DOCKET 87-01282	01262		8				Now	Nonrecurring	Submitted	k	· •	ნ-	Order va.	Order vs. Electronic-Disc
							Monrecurring	urring	Disc	Disconnect	per LSR		Electronic-1st	Electronic-Add1	15	Add1
1	F 0 1 X 0 0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1					Pec	First	LPP¥	Firet	LPPY	SOMEC	ا۔ا	SOMAN	BOMAN	SOMAN	SOMAN
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port	Exchange Ports - 2-Wire Analog Line Port (PBX)		UEPSP	UEPXD	\$1.79	\$9.93	\$9.19	\$3.68	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port	Exchange Ports - 2-Wire Analog Une Port (PBX)		UEPSP	UEPXE	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port	Exchange Ports - 2-Wire Analog Line Port (PBX)		UEPSP	UEPXL	\$1.79	\$9.93	\$9.19	\$3.86	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port	Exchange Ports - 2-Wire Analog Line Port (PBX)		UEPSP	UEPXM	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Economy Admin Calling Port TN Calling Port	Exchange Ports • 2-Wire Analog Line Port (PBX)		UEPSP	UEPXN	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port			UEPSP	UEPXO	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port	Exchange Ports - 2-Wire Analog Une Port (PBX)		UEPSP	UEPXS	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	2-Wire Voice Unbundled PBX Collerville and Memphis Calling Port	Exchange Ports - 2-Wire Analog Une Port (PBX)		UEPSP	UEPXU	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	2-Wire Voice Unbundled 2-Way PBX Tennessee RegionServ Calling Port	Exchange Ports - 2-Wire Analog Line Port (PBX)		UEPSP		\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	Subsequent Activity			UEPSP	USASC		\$10.00	\$10.00								
	All Available Vertical Features	[Exchange Ports includes all Applicable Features.]		UEPSP	UEPVF	\$0.00	\$0.00	\$0.00								
+	Exchange Ports - Coin Port	Exchange Ports - Coin Port				\$2.11	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
UNBON	UNBUNDLED LOCAL SWITCHING, PORT USAGE															
- -																
ត	End Office Switching (Port Usage)	End Office Sudisting Francisco				*********										
	End Office Trunk Port - Shared, Per MOU	End Office Interoffice Trunk Port - Shared, per MOU				90.0000										
=	Tandem Switching (Port Usage) (Local or Access Tandem)	Tondem Suitables Elizables				02200000										
	Tandem Trunk Port - Shared, Per MOU	Tandem Interoffice Trunk Port - Shared, per MOU				8//8000.0¢										
UNBUN	UNBUNDLED TRANSPORT															
8	COMMON TRANSPORT (Shared)															
	Common Transport - Per Mile, Per MOU	Common Transport - per mile, per MOU				\$0.0000084										
	Common Transport - Factities Termination Per MOU	Common Transport - Facilities Termination per MOU				\$0.0003871										
Ξ	INTEROFFICE CHANNEL - DEDICATED TRANSPORT - VOICE GRADE	SPORT - VOICE GRADE														
	Interoffice Channel - Dedicated Transport - 2-Interoffice Transport - Dedicated Wire Voice Grade - per mile per month	Interoffice Transport - Dedicated - Voice Grade - per mile per MQU		XVTIV	1LSXX	\$0.0174				·						
	Interoffice Channel - Dedicated Transport 2- Wire Volce Grade - Facility Termination per month	Interoffice Transport - Dedicated - 2-Wire Voice Grade - Facility Termination		VITVX	U1TV2	\$18.58	\$55.39	\$17.37	\$27.98	\$3.51			\$20.35	\$21.09	080	\$10.54

Page 7 of 29

nbundled Network Elements TENNESSEE
--

					-			RATES					SSO	OSS RATES		
			£												Charge -	Incremental Charge -
UNBUNDLED NETWORK ELEMENT	ORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Docket 97- Zone 01262	808	0080				Norrecurring	guling		Svc Order Submitted	Incremental Charge - Manual	Incremental Charge - Manuel	Menuel Svc Order vs.	Manuel Svc Order vs.
							Nonrecuring	pulsa	Discomed	nect		Manuelly per LSR	Svc Order vs. Electronic-1st	Svc Order vs. Electronic-Add7	Electronic-Disc	Electronic-Disc
				\downarrow	-	Jec.	First	Addi	First	Vqq.1	т	SOMAN	SOWAN	BOMAN	SOMAN	SOMAN
Interoffice Channel - Dedicated Transpor t- 2-Wire Volce Grade Rev Bat Per Mile per month	dicated Transpor t- Bat Per Mile per	Interoffice Transport - Dedicated - Voice Grade - per mile per MOU			- ILSXX	\$0.0174										
Interoffice Channel-Dedicated Transport: 2- Wire Voice Grade Rev Bat. Facility	cated Transport- 2- atFacility	Interoffice Transport - Dedicated - 2-Wire Volce Grade - Facility Termination		7/1	+		00.13	1	2	14 6			90.00	8	6	
		Sagar		5		9	600.00	/c./1	957.30	60.0			\$50.35	E0.126	00.64	1
INTEROFFICE CHANNEL - DEDICATED TRANSPORT- 56/64 KBPS	DEDICATED TRAN	SPORT- 56/64 KBPS														
Interoffice Channel - Dedicated Transport - 58 kbps - per mile per month	licated Transport - onth	Interoffice Transport - Dedicated - DSO - 56/64 kbps - per mile		U1TDX	X 1L5XX	\$0.0174										
Interoffice Channel - Dedicated Transport	licated Transport -	Interoffice Transport - Dedicated - DSO - 56/64 kbps - Facility														
56 kbps - Facility Termination per month	ation per month	Termination		VITDX	X U1TDS	\$17.98	\$55.39	\$17.37	\$27.98	\$3.51			\$20.35	\$21.09	\$9.80	\$10.54
Interoffice Channel - Dedicated Transport - 64 kbps - per mile per month	cated Transport	Interoffice Transport - Dedicated - IDSO - 56/64 kbps - per mile		U1TDX	X 1L5XX	\$0.0174										
Interoffice Channel - Dedicated Transport 64 kbps - Facility Termination per month	licated Transport -	Interoffice Transport - Dedicated - DSO - 56/84 kbps - Facility Termination		XOTIO	NAT NA	\$17.08	\$55.39	\$17.37	\$27.08	63.51			\$20.35	25	0	70.5
					-											
INTEROFFICE CHANNEL - DEDICATED TRANSPORT - DS1	DEDICATED TRANS	SPORT - DS1														
Interoffice Channel - Dedicated Channel - IDS1 - Per Mile per month	icated Channel -	Interoffice Transport - Dedicated - IDS1 - per mile		5	115XX	\$0.3562										
Interoffice Channel - Dedicated Tranport	cated Tranport -	Interoffice Transport - Dedicated -			+	╄										
DS: - Facility Termination per month	n per month	DS1 - Facility Termination		U1TD	UITE	\$77.88	\$112.40	\$76.27	\$19.55	\$14.99			\$20.35	\$21.09	\$9.80	\$10.54
LOCAL CHANNEL - DEDICA	TED TRANSPORT			<u> </u>												
Local Channel - Dedicated - 2-Wire Voice	d - 2-Wire Voice		•	2	- A	5										
Orace Channel - Definition	d. 2.Wire Volce	Local Channel - Dedicated - 2- Wire Volce Grade (shown here		3	$\overline{}$											
Grade per month - Zone 1	1	deaveraged)	_	ULDVX	X ULDV2	\$17.18	\$199.33	\$24.18	\$54.81	\$4.80			\$20.35	\$10.54	\$13.30	\$0.00
Local Channel - Dedicated - 2-Wire Volce Grade per month - Zone 2	id - 2-Wire Volce	Local Channel - Dedicated · 2- Wire Volce Grade [shown here deaveraged]	2	חרמא	X ULDV2	\$22.44	\$199.33	\$24.16	\$54.81	\$4.80			\$20.35	\$10.54	\$13.30	\$0.00
Local Channel - Dedicated - 2-Wire Volce Grade per month - Zone 3	d - 2-Wire Voice	Local Channel - Dedicated - 2- Wire Volce Grade [shown here deaveraged]	8	חראםא	X ULDV2	\$29.34	\$199.33	\$24.18	\$54.81	8.80			\$20.35	\$10.54	\$13.30	\$0.00
Local Channel - Dedicated - 2-Wire Volce Grade Rev. Bat. Per Mile per month	d - 2-Wire Volce per month		•	nravx	X 1L5NC	\$0.00										
Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat. Per month - Zone 1	d - 2-Wire Voice th - Zone 1	Local Channel - Dedicated - 2. Wire Voice Grade [shown here deaveraged]	-	ALDVX	X ULDR2	\$17.18	\$199.33	\$24.16	25.81	\$4.80			\$20.35	\$10.54	\$13.30	8 0.00
Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat. Per Month - Zone 2	d - 2-Wire Voice th - Zone 2	Local Channel - Dedicated - 2. Wire Voice Grade [shown here deaveraged]	8	ULDVX	X ULDR2	\$22.44	\$199.33	\$24.16	\$54.81	8.7			\$20.35	\$10.54	\$13.30	\$0.00
Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat. Per Month - Zone 3	d - 2-Wire Voice th - Zone 3	Local Channel - Dedicated - 2- Wire Volce Grade (shown here deaveraged)	6				\$199.33	\$24.16	\$54.81	\$4.80			\$20.35	\$10.54	\$13.30	\$ 0.0 \$
Local Channel - Dedicated - 4-Wire Voice Grade Per Mile per month	d - 4-Wire Voice		•	NEDDX	X 1L5NC	\$0.00						_				
Local Channel - Dedicated - 4-Wire Volce Grade per month - Zone 1	d - 4-Wire Volce	Local Channel - Dedicated - 4- Wire Voice Grade (shown here deaveraged)	-	חרםסא	X ULDV4	\$18.18	\$201.53	\$24.83	\$55.52	\$5.51			\$20.35	\$10.54	\$13.30	\$0.00
Local Channel - Dedicated - 4-Wire Volce Grade per month - Zone 2	d - 4-Wire Volce	Local Channel - Dedicated - 4. Wire Voice Grade [shown here deaveraged]	2		חרססא חרסא	\$23.74	\$201.53	\$24.83	\$55.52	\$5.51			\$20.35	\$10.54	\$13.30	00 0 5
		18		٦.		1		١.								

Unbundled Network Elements TENNESSEE

Exhibit 1-TN Attachment 2 Exhibit C

				F	-	-		Va	BATES		ŀ			OSS	OSS RATES		
								-				-				Incremental Cherse -	Cherge -
		A THE LEGISLANCE STATISTICS									Ī	Svc Order	Svc Order	Incremental	Incremental	Manual Sve	Manual Svc
_	UNBUNDLED NETWORK ELEMENT	STATED IN DOCKET 97-01262	<u> </u>	Zone BC\$	5 5 5	ý.				Norrecuring		Submitted	Submitted	Charge - Manuel	<u> </u>	- 5	Order vs.
								Nonrecurring	 E	Disconnect	Ţ	•	LSA PE	Electronic-1st	Electronic-Add7		Add1
				_	_		ž	First	VQQ.I	Fired	Pdd1	Н	SOMAN	SOMAN	SOMAN	SOHAN	SOMAN
	Local Channel - Dedicated - 4-Wire Voice	Local Channel - Dedicated - 4- Wire Voice Grade [shown here		-	NO III DV4		\$31.05	\$201.53	\$24.83	\$55.52	55.51			\$20.35	\$10.54	\$13.30	\$0.00
	Local Channel - Dedicated - DS1 Per Mile	- Francisco	*	+	-	<u> </u>	-		1								
	Local Channel - Dedicated - DS1 per month - Local Channel - Dedicated DS1	Local Channel - Dedicated DS1			-	_	ļ _	\$ 2777.35	\$233.28	\$33.18	\$22.30			\$45.08	\$1.76	\$21.75	\$1.78
	Local Channel - Dedicated - DS1 per month -			_		1_		+	+-	┼				945.00	61.70	431.75	61.78
1	Zone 2 [Shown here deaveraged] [Shown here deaveraged]	[shown here deaveraged]		z nrpp1	DI ULDFI	4	\$47.33	\$277.35	\$233.26	\$33.18	322.30			943.00	9/16	01.126	2
	Zone 3	[shown here deaveraged]		3 0,00	DO1 ULDF1		\$61.89	\$277.35	\$233.28	\$33.18	\$22.30			\$45.68	\$1.78	\$21.75	\$1.78
Z I	Will TIPI EXERS			-	-												
	Channelization - DS1 to DS0 Channel System	Channelization - Channel System DS1 to DSO		UXTD1	D1 MQ1		\$80.77	\$141.87	\$77.11	\$14.51	\$13.46			\$20.35	\$9.80	\$11.49	\$1.18
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kbs)	Interlace Unit - Interlace DS1 to DSO - OCU - DP Card		UDI). 10100		\$1.82	\$6.07	\$4.66					\$20.35	\$9.80	\$11.49	\$1.18
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per month	Interface Unit - Interface DS1 to DSO - Brite Card		5	UDN UC1CA		\$3.10	\$6.07	\$4.66					\$20.35	\$9.80	\$11.49	\$1.18
	Voice Grade COCI - DS1 to DS0 Channel System - per month	Interface Unit - Interface DS1 to DSO - Voice Grade Card		5	UEA 1D1VG		\$.91	\$6.07	\$4.66					\$20.35	\$9.80	\$11.49	\$1.18
	DS3 to DS1 Channel System per month	Channelization - Channel System DS3 to DS1		ň	UXTD3 MQ3		\$222.98	\$308.03	\$108.47	\$44.47	\$42.62			\$20.35	\$9.80	\$11.49	\$1.18
	STS1 to DS1 Channel System per month	Channelization - Channel System DS3 to DS1		š	UXTS1 MO3		\$222.98	\$308.03	\$108.47	\$44.47	\$42.62			\$20.35	\$9.80	\$11.49	\$1.18
	DS3 Interface Unit (DS1 COCI) used with Logo per month	Interface Unit - Interface DS3 to DS1		ISO	St. UC1D1		\$17.58	\$6.07	\$4.66					\$20.35	\$9.80	\$11.49	\$1.18
	DS3 Interface Unit (DS1 COCI) used with Local Channel per month	Interface Unit - Interface DS3 to DS1		ਤ	ULDD1 UC1D1		\$17.58	\$6.07	\$4.66					\$20.35	\$9.80	\$11.49	\$1.18
	DS3 Interface Unit (DS1 COCI) used with Interoffice Channel per month	Interlace Unit - Interlace DS3 to DS1		5	UITDI UCIDI		\$17.58	\$6.07	\$4.66					\$20.35	\$9.80	\$11.49	\$1.18
				\parallel	+	_			\dagger								
DARKE	DARK FIBER Con Elber Stroots Day Boule			+	+	+				1							
	Dark Floor, Four Floor Strands, For House Mile or Fraction Thereof per month - Local Channel	Dark Fiber, per 4 fiber strands, per route mile or fraction thereof			UDF 1L5DC		\$53.23										
	NRC Dark Fiber - Local Channel	Dark Fiber, per 4 fiber strands, per route mile or fraction thereof		귀	UDF UDFC	5	~	\$1,219.22	\$169.75	\$453.22	\$339.34						
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Internetine Channel	Dark Fiber, per 4 fiber strands, per route mile or fraction thereof			UDF 1150F		\$53.23		1.1								
	NRC Dark Fiber - Interoffice Channel	Dark Fiber, per 4 fiber strands, per route mile or fraction thereof		ח	UDF UDF14			\$1,219.22	\$169.75	\$453.22	\$339.34						
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Local Loop	Dark Fiber, per 4 fiber strands, per route mile or fraction thereof	_		UDF 11.5DL		\$53.23										
	NRC Dark Fiber - Local Loop	Dark Fiber, per 4 fiber strands, per route mile or fraction thereof		ם	UDF UDFL4	2	\$	\$1,219.22	\$169.75	\$453.22	\$339.34						
	** *** *******************************			+		+					İ						
	SXX Access Teo Dirti Streeding Per Cell	800 Access Ten digit screening (RD) ATDS) per call				05	\$0.0005192										
	8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number	800 Access Ten digit screening Reservation Charge per 800				+	-										
	Reserved	Number Reserved	1	\dashv	XI-N8H1X	1XI	-	\$5.21	\$0.78					\$20.35			

\parallel			-	H	-			RATES					989	OSS RATES		
_					<u>. </u>						-					Incremental
	ATTENDED TO AN ADMINISTRATION OF PARTY AND ADMINISTRATION	UNBUNDLED NETEORK ELEMENT AS	Not in TRA	-	<u>.</u>						Sve Order		Incremental	Incremental	Charge . Menual Svc	Charge -
	UNBONOLED NE I WORK ELEMEN I	STATED IN DOCKET 97-01262	01262						Nonrecurring	urring	Submitted	Z Z		6	Order vs. Electronic-Disc	Order vs. Electronic-Disc
					<u>.</u>	ž	Norrecurring First A	Addi	Firet	Addi	SOMEC	SOMAN	SOMAN	SOMAN	BOWAN	SOMAN
	8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translations	800 Access Ten digit screening per 800 # established w/o POTS Translations					\$11.47	\$1.46	\$7.34	\$0.7602			\$20.35		\$13.28	
	8XX Access Ten Digit Screening, Per 8XX No. Established With POTS Translations	800 Access Ten digit screening per 800 # established with POTS Translations		Z	NBFTX		\$11.47	\$1.46	\$7.34	\$0.7802			\$20.35		\$13.28	
	8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Number	800 Access Ten digit screening Customized Area of Service per 800 Number	and the state of t	Ž	NBFCX		\$4.47	\$2.24								
	8XX Access Ten Digit Screening, Multiple 800 ATDS, Multiple InterLATA InterLATA CXR Routing Per CXR Requested CXR Routing per CXR Requested Per 8XX No.	800 ATDS, Multiple InterLATA CXR Routing per CXR Requested per 800 #		Ž	NBFMX		\$5.23	\$3.00								
	8XX Access Ten Digit Screening, Change Charge Per Request	800 Access Ten digit screening, Change Charge per Request		Z	NBFAX		\$5.97	\$0.78					\$20.35			
	8XX Access Ten Digit Screening, Call Handling and Destination Features	800 Access Ten digit screening, Call Handling and Destination Features		Ž	NBFDX		\$4.47									
- INE	LINE INFORMATION DATA BASE ACCESS (LIDB)															
-	I IDB Common Transport Per Query	LIDB Common Transport per Query			100	\$0,0000354										
	LIDB Validation Per Query	LIDB Validation per Query				\$0.0117403										
	LIDB Originating Point Code Establishment or Change	LIDB Originating Point Code Establishment or Change					\$49.03				·		\$20.35			
H					\dagger											
SIGNAL	SIGNALING (CCS7)	CCS7 Signaling Termination per														
\dashv	CCS7 Signaling Termination, Per STP Port	STP Port			+	\$138.41										
	CCS7 Signaling Usage, Per TCAP Message	CCS/ Signaring Osage per I CAP Message			9	\$0.0000916										
	CCS7 Signaling Connection, Per link (A link)	CCS7 Signaling Connection per 56 kbps facility				\$17.84	\$130.84						\$20.35			
	CCS7 Signaling Connection, Per link (B link) (also known as D link)	CCS7 Signaling Connection per 56 kbps facility				\$17.84	\$130.84						\$20.35			
	CCS7 Signaling Usage, Per ISUP Message	CCS7 Signaling Usage, per call setup message			•	\$0.0000373										
	CCS7 Signaling Usage Surrogate, per link per LATA	CCS7 Signaling Usage Surrogate, per 58 kbps facility per LATA per month				\$352.30							\$20.35			
	OEI ECTIVE BOLITING															
-		Selective Routing (Interim Solution														
	Selective Routing Per Unique Une Class Code Per Request Per Switch	Line Class Codes) per Unique Line Class Code per Request per Switch		ă	USRCR		\$179.80						\$20.35			
	AIN DELICOTED AND ONG ACCESS OF DUCCES															
	AIN SMS Access Service - Service Establishment. Per State, Initial Setup	AIN SMS Access Service - Service Establishment, per State, Initial Setup		3	CAMSE		\$135.56									
	AIN SMS Access Service - Port Connection - AIN SMS Access Service - Port Connection - Dial/Shared Access	AIN SMS Access Service - Port Connection - Dial/Shared Access		ò	CAMDP		\$41.75									o mora da capacidado de la composição de la capacidado de
	rvice - Port Connection -	AIN SMS Access Service - Port Connection - ISDN Access		Ö	CAM1P		\$41.75									

Page 10 of 29

03/03/01

					-		RATES					SSO	OSS RATES		
			j					_						_	Charge -
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 87-01262	Docket 97- Zone 01262	2	8			2	Nomecuning	Svo Order Submitted	Svc Order Submitted	Incremental Charge - Manual	Incremental Charge - Manuel	Manual Sve Order vs.	Manuel Svc Order vs.
			!				Nonseuring		Disconnect	Der LSR	Manually per	Svc Order vs.	Sve Order vs. Dectronic-Add'l	Sectronic-Disc 1st	Electronic-Disc Add1
						Pec	First Add'I	Frest	LPPY	SOMEC	SOMAN	SOMAN	BOMAN	SOMAN	SOMAN
	AIN SMS Access Service - User Identification Codes - Per User ID Code	AIN SMS Access Service - User Identification Codes - Per User ID Code			CAMAU		\$96.63								-
	AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement	AIN SMS Access Service - Security Card, per User ID Code, Initial or Replacement			CAMRC		\$113.67								
	AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)					\$0.0024									
	AIN SMS Access Service - Session, Per Minute.	AIN SMS Access Service -				\$0.0920123									
	AIN SMS Acrese Service - Company	AIN SMS Access Service -													
	Performed Session, Per Minute	minute		_		\$2.27		_							
AN-B	AN - BELLSOUTH AIN TOOLKIT SERVICE														
	AIN Tookit Service - Service Establishment Charge, Per State, Initial Setup	AIN TS - Service Establishment Charge per State, Initial Setup			BAPSC		\$132.04								
	AIN Toolkit Service - Training Session, Per Customer	AIN 15 - Training Session, per Customer			BAPVX		\$7,915								
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt				BAPTT		\$31.21								
	AIN Töckit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay				BAPTD		\$31.21								
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hock Immediate	AIN TS - Trigger Access Charge, per Trigger per DN, Off Hook immediate			BAPTM		\$31.21								
	AIN Toolkit Service - Trigger Access Charge, AIN TS - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP per Trigger, Per DN, 10-Digit PODP	AIN TS - Trigger Access Charge, per Trigger per DN, 10-Digit PODP			BAPTO		\$85.24	-							
	AIN Toolkil Service - Trigger Access Charge, AIN TS - Trigger Access Charge, Per Trigger, Per DN, CDP	AIN 1S - Ingger Access Charge, per Ingger per DN, CDP			BAPTC		\$85.24								
	AIN Toolkit Service - Trigger Access Charge, AIN TS - Trigger Access Charge, Per DN, Feature Code per Trigger per DN, Feature Code	, AIN TS - Trigger Access Charge, per Trigger per DN, Feature Code			BAPTF		\$85.24								
	AiN Toolkit Service - Query Charge, Per Query	AIN TS - Query Charge, Per Query				\$.0211882									
	AIN Tookit Service - Type 1 Node Charge, Per AIN Tookit Subscription, Per Node, Per Query	AIN TS - Type 1 Node Chargé, per AIN Tookit subscription, per Node, per query				\$.0054774									
	AIN Toolkit Service - SCP Storage Charge, Per SMS Access Account, Per 100 Kilobytes					\$1.50									
	AIN Tookit Service - Monthly report - Per AIN TS - Monthly Report - per AII AIN Tookit Service Subscription Tookit Service Subscription	AIN TS - Monthly Report - per AIN TS Subscription			BAPMS	\$17.43	\$33.52								
	AiN Tookit Service - Special Study - Per AIN Tookit Service Subscription	AIN 15 Special Study - per AIN 15 Subscription			BAPLS	\$0.1321116	\$36.23								
	Ain Tookkit Service - Call Event Report - Per Ain Tookkit Service Subscription	AIN TS - Call Event Report - per AIN TS Subscription			BAPDS	\$17.35	\$33.52								
	AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit Service Subscription	AIN TS - Call Event Special Study - per AIN TS Subscription			BAPES	\$0.0511435	\$36.23								
ODUF	ODUE/EDOUE/ADUE/CMDS							-	-						
ō	OPTIONAL DAILY USAGE FILE (ODUF)							4							

							RATES				980	OSS RATES		
			1											Incremental Charge
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Docker 97- Ze 01262	Zone BCS	0060			Nonrecurring	Svc Order Submitted	Svc Order Submitted Manually per	Incremental Charge - Manual Sun Order vs.	Incremental herge - Manuel Sue Order on	Manual Svc Order vs. Flantinote-Disc	Manual Sve Order vs. Flectrools-Disc
						2	Norrecuring First Add1	Disconnect First Add1	╅		Electronio-1st SOMAN	SOMAN	SOMAN	SOWAN
	ODUF: Recording, per message	OSS OLEC Daily Usage File Recording per message				\$0.0000044								
	ODUF: Message Processing, per message	OSS OLEC Daily Usage File Message Distribution per message				\$0.0027368								
	ODUF: Message Processing, per Magnetic Tape provisioned	OSS OLEC Daily Usage File Message Distribution per magnetic tape provisioned				\$52.75								
	ODUF: Data Transmission	OSS OLEC Daily Usage File Data Transmission (Connect: Direct),				\$0 00000330								
	3839991 DA V. 25111111111111111111111111111111111111	2				\$0000000								
0	UNBUNDLED PORT/LOOP COMBINATIONS - COST BASED RATES Cost Based Rates are applied where BellSouth is	OST BASED RATES												
z 0~-	required by FCC and/or State Commission rule to provide Unbunded Local Switching or Switch Ports. BellSouth is required to offer unbunded port/loop combinations that are Currently Combined in Tennessee unless the end user has lour or more DSO equivalent lines and is in Zone 1 of one of the Top 8 MSAs in BellSouth's region.													
	In the absence of ordered rates by a State Commission, the recurring rates for combinations of porthoop network elements will the sum of the recurring rates for the UNEs which make up the combinations, and the nonrecurring rates shall be as set forth in this section.													
	Vertical features shall apply to the Unbundled Por/Loop Combination - Cost Based Rate section in the same manner as they are applied to the Stand-Alone Unbundled Port section of this Rate Exhibit.													
	End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section of this rate earbits shall apply to all combinations of loop/port network elements except for UNE Coin Port/Loop Combinations which have a flat rate usage charge (USOC: WHICH).													
C ±	For Currently Combined Combos in TN, the nonrecurring charges shall be those identified in the Nonrecurring Charges - Currently Combined sections.													
c	S. WISE VOICE COADE LOOP WITH 3 WISE LINE BOBY (SES)	E 6A64 (6E6)												
4	WINE VOICE GRADE LOOP WITH E-WINE LIT	E roni (neo)												
	ONE POTALOND COMBINATION NAMES	2-Wire Voice Grade Loop with 2-		<u> </u>										
	2-Wire VG LooyPort Combo - Zone 2	2-Wire Voice Grade Loop with 2-Wire Line Port - Zone 2				613.10								
	2-Wire VG Loop/Port Combo - Zone 3	2-Wire Voice Grade Loop with 2- Wire Line Port - Zone 3	3			\$23.02								
	UNE Loop Rates			$\perp \mid$	Ш									

		_					-	RATES				SSO	OSS RATES		
					٠					-				Incremental	Incremental
UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORIX ELEMENT AS 8TATED IN DOCKET 97-01262	Docker 97-	Zove	S	2020				Noncecurine	Submitted	Submitted	Incremental Charge - Manuel	Incremental Charge - Manual	Manual Sve Order vs.	Manual Sve Order vs.
					<u> </u>		Normacurring	2	Disconnect	Elec Ser Lan		Svc Order vs.	Svc Order vs.	Electronic-Disc	Electronic-Disc
					<u> </u>	Je L	7.	L PPY	First Add'l	SOMEC	BOMAN	SOWAN	-	SOMAN	SOMAN
2-Wire Voice Grade Loop (SL1) - Zone 1) EPRX I	UEPLX	\$12.29							L		
2-Wire Voice Grade Loop (SL1) - Zone 2			2	JEPHX	UEPLX	\$16.12									
Z-Wire Voice Grade Loop (SL1) - Zone 3		:	\neg	JEPHX	NEPLX	\$21.13			-						
2-Wire Voice Grade I Ine Bort Bates (Bes)			İ												
2.Wite policy and particular action and a		:	Ť	1 ICDOV	10001	00.5						7000			
2-Wire voice unbundled not with Caller ID -					ן ו	69.14	1	1							
es.		:	_	UEPRX	UEPRC	\$1.89			- · · · <u>-</u>					-	
2-Wire voice unbundled port outgoing only -		:		-	0001	8									
2.Wis voice Grade inhinded anneces			1	OEPHX OEPHX	OEPHO DEPHO	88.18									
extended local dialing parity port with Calter ID - res		:		UEPRX	UEPAQ	\$1.89									
2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (F2R)		:		UEPRX	UEPAK	\$1.89									
2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACER)		:		UEPRX	UEPAL	\$1.89									
2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACSR)		:		UEPRX	UEPAM	81.83									
2-Wire voice unbundled Tennessee Area		:		-	NA OR	8									
2-Wire voice unbundled Tennessee Area			T		2	60.19									
Calling port with Caller ID - res (2MR)		:	_	UEPRX	UEPAO	\$1.89									
z-wire voce uncuroles res, low usage line port with Caller ID (LUM)		:		UEPRX	UEPAP	\$1.89									
4 4 4			_	+	1										
FEATURES	4		†		1										
All Available Vertical Features	(Exchange Ports includes all Applicable Features.)	:		UEPRX	UEPVF	\$0.00	\$0.00	\$0.00							
OCAL MIMBED BODTABILITY			+												
Local Number Portability (1 per port)		:	Ť	UEPRX	LNPCX	\$0.35									
NONRECURRING CHARGES (NRCs) - CURRENTLY COMBINED	NTLY COMBINED														
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is	Port Combo - Switch-as-is			UEPRX	USAC2		\$1.03	\$0.29				\$30.89	\$7.03		
2-Wire Voice Grade Loop / Line Port Combination • Conversion • Switch with		•		×000	00401		8	00				90			
2-Wire Voice Grade Loop / Line Port	2-Wire Voice Grade Loop/Line			$\overline{}$	2		3	670				60.20	3		
Combination - Conversion - Subsequent Database Update	Port Combo - Subsequent Database Update						\$0.76					\$7.97			
ADDITIONAL MBC.			\dagger		+										
2-Wire Volce Grade Loop/Line Port Combination - Subsequent Activity		:	1-1	UEPRX	USAS2		\$10.00	\$10.00							
			\dashv												
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)	NE PORT (BUS)		+	1	+										
UNE Port/Loop Combination Rates			\dagger		-										
2-Wire VG Loop/Port Combo - Zone 1	2-Wire Voice Grade Loop with 2- Wire Line Port - Zone 1		-			\$14.18									
	2-Wire Voice Grade Loop with 2- Wire Line Port - Zone 2		2	-		\$18.01									
	2-Wire Voice Grade Loop with 2- Wire Line Port - Zone 3		6			\$23.02									
			1												200

			\vdash	\vdash			٤	RATES				SSO	OSS RATES		
		F								_					Incremental Charge -
UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET \$7-01262	Docket 97-	Zone	50g	nsoc			-	Nonrecurring	Submitted	Submitted	Charge - Manuel	Incremental Charge - Manual	Manuel Svc Order vs.	Memuel Svc Order vs.
							Nonrecurring	auju.	Disconnect	Per LSR		Electronic-1st	Electronic-Add1	Telectronic-Diac	Add1
	0.00		\dashv	\dashv	+	ž	First	Addi	First Add'l	SOMEC	SOMAN	SOWAN	BOMAN	"	SOMAN
			1												
ONE Loop Rates		:	=	ABGSI	X IOU	640.00									
2.Wire Voice Grade (con (SI 1) - Zone 2		:	+		\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \	618 13								-	
2-Wire Voice Grade Loop (SL1) - Zone 3		:	3 5	_	NEP C	\$21.13									
			†		-										
2-Wire Voice Grade Line Port (Bus)															
2-Wire voice unbundled port without Caller ID - bus		:	5	UEPBX	UEPBL	\$1.89									
2-Wire voice unbundled port with Caller + E484 ID - bus		:	5		UEPBC	\$1.89									
2-Wire voice unbundled port outgoing only - bus		:	35	+	UEPBO	\$1.89									
2-Wire voice Grade unbundled Tennessee extended local dialing parity port with Caller ID - bus		:.	5		UEPAV	\$1.89									
2-Wire volce unbundled incoming only port with Catter ID - Bus		:	5	UEPBX	UPEB1	\$1.89									
2-Wire voice unbundled Tennessee Bus 2- Way Area Calling Port Economy Option (TACC1)		:	3	UEPBX U	UEPAC	\$1.89									
2-Wire voice unbundled Tennessee Bus 2- Way Area Calling Port Standard Option (TACC2)		:	3		UEPAD	\$1.89									
2-Wire voice unbundled Tennessee Bus 2- Way Collerville and Memphis Local Calling Port (B2F)		:	3	UEPBX	UEPAE	\$1.89									
T HONTED GOOD IN TACK			+	+	\dagger										
Local Number Portability (1 per port)		:	15	UEPBX L	LNPCX	\$0.35									
			H	$\left \cdot \right $	H										
FEATURES All Available Vertical Features	[Exchange Ports Includes all Applicable Features.]	:	5	UEPBX	UEPVF	\$0.00	\$0.00	\$0.00							
NONBECUBRING CHARGES (NRCs) - CURRENTLY COMBINED	NTLY COMBINED		+	-											
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is	2-Wire Voice Grade Loop/Line Port Combo - Switch-as-is		5	UEPBX	USAC2		\$1.03	\$0.29				\$30.89	\$7.03		
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with change		•	<u>#</u>	UEPBX	USACC		\$1.03	\$0.29				\$30.89	\$7.03		
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update	2-Wire Voice Grade Loop/Line Port Combo - Subsequent Database Update						\$0.76					\$7.97			
ADDITIONAL NRCs			+	\vdash	\dagger										
2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity		:	5	UEPBX	USAS2		\$10.00	\$10.00							
2-WIRE VOICE GRADE LOOP WITH 3-WIRE LINE PORT (RES - PRX)	NE PORT (RES - PRX)		-		+										
S 5111111111111111111111111111111111111				$\frac{1}{1}$											
UNE POTILOOP COmbination Rates	2-Wire Voice Grade Loop with 2-		+	+	+										
2-Wire VG Loop/Port Combo - Zone 1	Wire Line Port - Zone 1		-	\dashv	1	\$14.18									

			-					RATES				088	OSS RATES		
					_									Incremental	Incremental
	At President of June Contract	Not in TRA						1		į,	1	la brancher la	la constitue de la constitue d	Charge.	Charge .
UNBUNDLED NETWORK ELEMENT	STATED IN DOCKET 97-01262	Docket 97- 01252	Zone	S S	<u></u>				Norrecuring	Submitted	Submitted Manually per	5	ᇴ	Order vs.	Order vs.
							Nomecurring	gulling	Disconnect	per LSR	ES.	Electronic-1st	Electronic-Add1		ğ
			\dashv	+	+	ž	First	Mdd:1	First Add't	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
2-Wire VG Loop/Port Combo - Zone 2	2-Wire Voice Grade Loop with 2- Wire Line Port - Zone 2		7			\$18.01			-						
2-Wire VG Loop/Port Combo - Zone 3	2-Wire Voice Grade Loop with 2- Wire Line Port - Zone 3		6			\$23.02									
			Н												
UNE Loop Rates															
2-Wire Voice Grade Loop (SL 1) - Zone 1		:	<u>Р</u>		UEPLX	\$12.39									
2-Wire Voice Grade Loop (SL 1) - Zone 2		:	2		UEPLX	\$16.22			-						
2-Wire Voice Grade Loop (SL 1) - Zone 3		:	9	UEPRG	UEPLX	\$21.23									
2. With Volce Grade Inte Both Bates (BES BRS	REY		\dagger		1			T							
2-Wire VG Unbundled Combination 2-Way		:	-	5897	Cadai	61 70									
Sec. 15. Vibra Var			-	_	1										
LOCAL NUMBER PORTABILITY															
Local Number Portability (1 per port)		:	ر	UEPRG L	LNPCP	\$3.15									
			\dagger												
FEATURES	4		1												
All Available Vertical Features	[Exchange Ports includes all Applicable Features.]	:	اد_	UEPRG (UEPVF	\$0.00	\$0.00	\$0.00							
	A PARTICULAR OF THE PARTICULAR		\dagger	\dagger											
NONHECCHAING CHANGES (NACS) - CUNNENILY COMBINED	ENILT COMBINED		\dagger	\dagger											
2-Wire Voted Grade Loop Line Port Combination (PBX) - Conversion - Switch-As-2-Wire Voice Grade Loop/Line Is Port Combo - Switch-as-Is	As-2-Wire Voice Grade Loop/Line Port Combo - Switch-as-is			UEPRG	USAC2		\$1.03	\$0.29				\$30.89	\$7.03		
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch		,			-							900			-
with Change			7	UEPHG	USACC		2 1.03	\$0.29				\$30.68	\$7.03		
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update	2-Wire Voice Grade Loop/Une Port Combo - Subsequent Database Update						\$0.76			-		\$7.97			
			Н												
ADDITIONAL NRCs			\dagger	1	1										
Combination (PBX) - Subsequent Activity		:	٥	UEPRG	USAS2		\$10.00	\$10.00				\$30.89	\$7.03		
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)	LINE PORT (BUS - PBX)		$\dagger \dagger$												
11NE Dard American Dates			\dagger	\dagger											
ONE TOTAL OF THE PRINCIPLE OF THE PRINCI	2-Wire Voice Grade Loop with 2-		-			617.10									
Caville VG LOOPPOIL CORRIO - CORR 1	2-Wire Une Port - Zone 1 Wire Line Port - Zone 2		- "			61801									
1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	2-Wire Voice Grade Loop with 2-		+												
2-Wire VG Loop/Port Combo - Zone 3	Wire Line Port - Zone 3		6			\$23.02									
UNE Loop Rates			H												
2-Wire Voice Grade Loop (SL 1) - Zone 1		:	_	UEPPX	UEPLX	\$12.39									
2-Wire Voice Grade Loop (SL 1) - Zone 2		:	N 6	_	X	\$21.22				+					
2 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -			_												
2-Wire Voice Grade Line Port Rates (BUS - PBX)	BX)		H	П											

Unbundled Network Elements TENNESSEE

Exhibit 1-TN Attachment 2 Exhibit C

	4	_	-	_	å	_
	Incremental In	Charge -	Manual Sve	Order vs.	Electronic-Dis	1
OSS RATES			Incremental	Charge - Menuel	Swc Order va.	Plantonic-Add"
SSO			Svc Order Svc Order Incremental Incremental Manual Svc	Submitted Submitted Charge - Manual Charge - Manual Order vs.	Elec Manually per Svc Order vs. Svc Order vs. Electronic-Disc Elec	1 S.P. Flactmode, 1st Plantmode, 4.4673 1st
			Sve Order	Submitted	Manually per	
			8vc Order	Submitted	20	100
				ecurring		Cleananan
				Norrecurring		ž
RATES						Monetacuerino
					<u> </u>	3
			50	3		
			MOLIN TRA	ĵ		
		-	<u> </u>	*/- KOL		
_			2 2	DOCKET I	95	
			UNBUNDLED NETEORK ELEMENT AS	STATED IN DOCKET 97-01262		
				UNBUNDLED NETWORK ELEMENT		

				L	-	-		à	RATES				SSO	OSS RATES		
						<u> </u>			-						Charge -	Cherge -
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS	5 %	Zone	5	nsoc			-	No.	Svc Order Submitted	Submitted	Incremental Charge - Manual	Incremental Charge - Manuel	Manual Sve Order vs.	Manual Svc Order vs.
			01282			1_		Morececuring	8	Disconnect	Elec Der I.S.	Manually per	Sve Order vs.	Sve Order vs. Electronic-Add?		Electronic-Diac
				\dashv			ž	First	Vdd1	First Add1	SOMEC	BOMAN	SOMAN	\$OMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update	2-Wire Voice Grade Loop/Line Port Combo - Subsequent Database Update						\$0.78					\$7.97			
	ADDITIONAL MBCs			-												
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activity		:	5	UEPPX	USAS2		\$10.00	\$10.00				\$30.89	\$7.03		
	2-WIRE VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK PORT	TH 2-WIRE DID TRUNK PORT														
	UNE Port/Loop Combination Rates	C this year of the County of t		+												
	Combo - Zone 1	Wire DID Trunk Port - Zone 1		_			\$18.38									
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - Zone 2	2-Wire Voice Grade Loop with 2- Wire DID Trunk Port - Zone 2		2			\$19.87									
<u> </u>	2-Wire VG Loop/2-Wire DID Trunk Port Combo - Zone 3	2-Wire Voice Grade Loop with 2- Wire DID Trunk Port - Zone 3		6			\$25.52									
L																
Ĺ	NONRECURRING CHARGES - CURRENTLY COMBINED	OMBINED														
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination - Switch-as-is	2-Wire Volce Grade Loop with 2- Wire DID Trunk Port Combo - Switch-as-is		an Bi	UEPPX	USAC1		\$8.76	\$5.75				\$41.43	\$9.80		
	2-Wire Voice Grade Loop / 2-Wire DID			_												
	I runk Port Combination Conversion with Changes		•	ᄬ	UEPPX US	USA1C		\$8.76	\$5.75		\$3.50		\$41.43	\$9.80	\$20.00	\$20.00
				-												
<i>-</i> 0	Telephone Number/Trunk Group Establisment Charges															
	DID Trunk Termination (One Per Port)		:	5	UEPPX	TON	\$0.00	\$0.00	\$0.00		\$3.50		\$41.43	\$9.80	\$20.00	\$20.00
	DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbers		:		VEPPX	ZQN	\$0.00	\$0.00	\$0.00		\$3.50		\$41.43	\$9.80	\$20.00	\$20.00
	Additional DID Numbers for each Group of 20 DID Numbers		:	5	UEPPX	Š	\$0.00	\$0.00	\$0.00		\$3.50		\$41.43	\$9.80	\$20.00	\$20.00
	DID Numbers, Non- consecutive DID Numbers, Per Number		:	5	-	SON	\$ 0.00	00:00	\$0.00		\$3.50		\$41.43	\$9.80	\$20.00	\$20.00
	LOCAL NUMBER PORTABILITY			-	-											
	Local Number Portability (1 per port)		:	3	UEPPX LN	LNPCP	\$3.15	\$0.00	\$0.00							
	2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT	MRE ISDN DIGITAL LINE SIDE POR	<u>.</u>													
	History Combinetion Deter			+												
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - Zone 1	2-Wire ISDN Digital Grade Loop with 2-Wire ISDN Digital Line Side Port - Zone 1		-			\$32.27						\$19.99	\$19.99	\$20.00	\$20.00
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port . Zone 2	2-Wire ISDN Digital Grade Loop with 2-Wire ISDN Digital Line Side Port - Zone 2		2			\$34.78						\$19.99	\$19.99	\$20.00	\$20.00
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - Zone 3	2-Wire ISDN Digital Grade Loop with 2-Wire ISDN Digital Line Side Port - Zone 3		၈			\$44.32						\$19.99	\$19.99	\$20.00	\$20.00
				+				1								
	NONRECURRING CHANGES - CURRENTLY COMBINED	OMBINED		\dashv	\dashv	1			1							

100	8.	nte-Disc	SOWAN	\$20.00		\$20.00										П						800	250.00	\$20.00		\$20.00		
	Charge - Manual Svc		₹ 8	258		\$20							_				1		1		-	-	\$ 65	\$20		\$20		
a di di di di di di di di di di di di di	Charge -		SOMAN			\$20.00																8 904	\$20.00	\$20.00		\$20.00		
ATES	Incremental	Charge - Manuel Sve Order vs.	Electronic-Add 1	\$19.99		\$19.99																800	96018	\$19.99		\$19.99		
OSS RATES	Incremental	-	Electronic-1st SOMAN			\$19.99																80,	900	\$19.99		\$19.99		
-	Svc Order	Submitted O	SOMAN														1											•
			SOMEC SOME	\$3.50		\$3.50																						
		Norrecurring	Disconnect																									
		Nonrek	First																									
RATES			Norrecurring st Add't	\$117.23				\$0.00		\$0.00	\$0.00	\$0.00		\$0.00	\$0.00	\$0.00	\$0.00		\$0.00							\$328.53		
			Norrec	\$117.23		\$212.88		\$0.00		\$0.00	\$0.00	\$0.00		\$0.00	\$0.00	\$0.00	\$0.00		\$0.00							\$328.53		
			Rec					\$0.35		\$0.00	\$0.00	\$0.00		\$0.00	\$0.00	\$0.00	\$0.00		\$0.00			613.50	\$150.25	\$173.44				
	30	3	!	USACB	+	USASB		LNPCX		UTUCA	U1UCB	UIUCC		UTUCD	U1UCE	UTUCF	U1UMA		UEPVF							USACP		
	5	3		UEPPB	+	UEPPB L		UEPPR		UEPPB		UEPPB U		UEPPB	UEPPB	UEPPB	UEPPB U		UEPPB	$\parallel \parallel$						UEPPP U		
	2 02			ر	\perp	2		د			د	7		3		7			+			-	. 2	က		n		
	Not in TRA Doctor 97.	01262						:		:	:	:		:	:	:	**		:									
	UNBUNDLED NETEORK ELEMENT AS			2-Wire ISDN Digital Grade Loop 2- Wire ISDN Line Side Port Combo - Switch-as-is		2-Wire ISDN Digital Grade Loop 2-Wire ISDN Line Side Port Combo- Non Feature Subsequent Activity			The state of the s											TOO NOT IN THE	N DSI DIGILAL INGIN PONI	4-Wire DS1 Orgital Loop with 4- Wire ISDN DS1 Digital Trunk Port -	4-Wire DS1 Digital Loop with 4. Wire ISDN DS1 Digital Trunk Port - Zone 2	4-Wire DS1 Digital Loop with 4- Wire ISDN DS1 Digital Trunk Port - Zone 3	OMBINED	4-Wire DS1 Digital Loop with 4- Wire ISDN DS1 Digital Trunk Port Combo - Switch-as-is		4-Wire DS1 Digital Loop with 4-Wire ISDN DS1 Digital Trunk Port
	THE RELEASE OF CANDEND			2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination - Conversion	ADDITIONAL NRCs	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination-Non Feature Subs Activity	LOCAL NUMBER PORTABILITY	Local Number Portability (1 per port)	B-CHANNEL USER PROFILE ACCESS:	CVS/CSD (DMS/SESS)	CVS (EWSD)	CSD	B-CHANNEL AREA PLUS USER PROFILE ACCESS: (AL KY) A MS SC MS & TM)	CVS/CSD (DMS/SESS)	CVS (EWSD)	- OSO	USER TERMINAL SERVICE PROFILE (EWSD	VEDTION CEATIDED	One per Channel B User Profile	THE POT THE PAST OF THE POT TH	THE DELICATION AND THE ISO	UNE Port/Loop Combination Rates 4W DS1 Digital Loop/4W ISDN DS1 Digital Trans Doc 27004	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - Zone 2	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - Zone 3	ONRECURRING CHARGES - CURRENTLY CO	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 4-Wire DS1 Digital Loop with 4- Digital Trunk Port Combination - Conversion- Wire ISDN DS1 Digital Trunk Port Switch-as-is Combo - Switch-as-is	ADDITIONAL NRCs	A Wish DC 1 Pro 1 and 1

10/60/20

				L	-			RATES				SSO	OSS RATES		
			1							_					Incremental Charge -
	UNDUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Docket 97- Zc 01262	Zone BCS	nsoc	6	,		Nonrecurring	Submitted	Svc Order Submitted	Incremental Charge - Manuel	Incremental Charge - Manuel		Manual Svc Order vs.
							Nonrecuring	urring	Disconnect	per LSR	LSR	Svc Order vs.	Sve Order vs. Dectronic-Add1	2	Electronic-Unic
1			1	4	1	Rec	First	I.PPV	First Add1	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	4-Wire DS1 Dig Loop / 4-Wire ISDN DS1 Dig Wire ISDN DS1 Digital Trunk Port Trunk Port Combination-Sub Inward/2-Way Combo - Subsequent Inward/2-Telephone Numbers	4-Wire DS1 Digital Loop with 4-Wire ISDN DS1 Digital Trunk Port Combo - Subsequent Inward/2-Way Telephone Numbers		UEPPP	PP PR7TG	<u> </u>	9					80 00 00 00 00	66.61	\$20.00	\$20.00
	4-Wire DS1 Dig Loop / 4-Wire ISDN DS1 Digital Loop with 4-Trunk Port Combination-Sub Outward Combo - Subsequent Outward Telephone Numbers Telephone Numbers	4.Wire DS1 Digital Loop with 4. Wire ISDN DS1 Digital Trunk Port Combo - Subsequent Outward Telephone Numbers		UEPPP		<u>a</u>	\$22.36					\$19.99	\$19.99	\$20.00	\$20.00
	4-Wire DS1 Dig Loop / 4-Wire ISDN DS1 Dig Wire ISDN DS1 Digital Trunk Port Trunk Port Combination-Subsequent Inward Telephone Numbers Telephone Numbers	4-Wire DS1 Digital Loop with 4- Wire ISDN DS1 Digital Trunk Port Combo - Subsequent Inward Telephone Numbers		UEPPP	PP PR7ZT	L .	\$44.71					\$19.99	\$19.99	\$20.00	\$20.00
	4-Wire DS1 Dig Loop / 4-Wire ISDN DS1 Dig Wire ISDN DS1 Digital Trunk Port Trunk Port Trunk Port Combinesion-Subsequent Service Combo - Subsequent Service Order Per Order	4-Wire DS1 Digital Loop with 4- Wire ISDN DS1 Digital Trunk Port Combo - Subsequent Service Order Per Order		UEPPP	PP USASP	ą.	\$189.78					\$19.99	\$19.99	\$20.00	\$20.00
	LOCAL NUMBER POR ABILLITY LOCAL Number Portability (1 per port)		:	UEPPP	PP LNPCN	S1.75									
	INTERFACE (Provsioning Only)			_	+										
	Voice/Data		:	UEPPP	PR71V	× \$0.00									
	Digital Data		:	UEPPP											
	Inward Data		•	UEPPP	PH71E	\$0.00									
Ĭ	CALL TYPES			-											
	Inward		•	UEPPP		\$0.00									
	Outward		:	UEPPP		Ц									
	Two-way		:	UEPPP	PR7CC	00.0 \$									
	4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT	S TRUNK PORT													
					_										
	i *	4-Wire DS1 Digital Loop With 4- Wire DID Trunk Port - Zone 1		-		\$93.28									
	1 -	4-Wire DS1 Digital Loop With 4- Wire DID Trunk Port - Zone 2		7		\$110.95									
	4W DS1 Digital Loop/4W DDITS Trunk Port Zone 3	4-Wire DS1 Digital Loop With 4- Wire DID Trunk Port - Zone 3		е		\$134.14									
	NONRECURRING CHARGES - CURRENTLY COMBINED	MBINED													
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Switch-83-is	4-Wire DS1 Digital Loop/4-Wire DID Trunk Port Combo - Switch-as- is		UEPDC	XC USAC4	4	\$312.91	\$312.91		\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
	_		•	UEPDC	C USAWA	₹	\$312.91	\$312.91		\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
	igital Loop / 4-Wire DDITS inclustion - Conversion with k	4-Wire DS1 Digital Loop/4-Wire DID Trunk Port Combo - Switch-as- is		UEPDC	C USAWB	<u>e</u>	\$312.91	\$312.91		\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
	ADDITIONAL NRCs														
					-										

USAS4 \$94.88 UDTTA \$108.67 UDTTB \$108.67 UDTTC \$108.67 UDT				-	_				RATES				088	OSS RATES		
Unique Comparison Compari				j											$\overline{}$	incremental Charge -
DSI Dig Loop 44-Wire DDIS Trank		UNBUNDLED NETWORK FLEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Docket 97- Zor		nsoc				Nonrecurring	Svc Order Submitted	_	Incremental Charge - Menual	Incremental Charge - Manual		Manuel Sve Order vs.
e DSI Dig Loop (4 Wile DDITS Truck DDIS Digital Coop 4 Wire To District Dis				78				Home	:	Disconnect	E E	Manually per	Svc Order vs. Plectronic-1st	Sve Order vs.	Electronic-Disc 1	Electronio-Disc
6 St Dig Loop 4 Wire DOITS Trunk DUID Trunk Port DUID T							3	First	Addi	First Add1	Н	11		_	SOMAN	SOMAN
10 10 10 10 10 10 10 10		-Wire DS1 Dig Loop / 4-Wire DDITS Trunk ort Combination-Subsequent Service ctivity Per Svc Order	4-Wire DS1 Digital Loop 4-Wire DID Trunk Port Combo - Subsequent Service Order Per Order		UEPDC			\$94.88	\$94.88		\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
		-Wire DS1 Dig Loop/4-Wire DDITS Trk Port combination-NRC-Sub Channel Activation- er Chanl-2-Way Trk	4-Wire DS1 Digital Loop/4-Wire DID Trunk Port Combo • Subsequent Channel Activation Per Channel		UEPDC			\$108.67			\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
### 6 Sti Dig Loog/4-Wire DDITS Trk Port DID Trunk Port Combo **Ostb Chan Activation** Per Chan**. **Description** DDITS Trk Port DID Trunk Port Combo **Ostb Chan Activation** Per Chan**. **Description** DDITS Trk Port DDITS Tr		-Wire DS1 Dig Loop/4-Wire DDITS Trk Port combo-Sub Channel Activation-Per Chanl-1- 1ay Outward Trk	4-Wire DS1 Digital Loop/4-Wire DID Trunk Port Combo - Subsequent Channel Activation Per Channel		UEPDC			\$108.67			\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
Ost Dig Locot4-Wire DDITS Trk Pord IDID Trunk Pord Combo Ost Dig Locot4-Wire DDITS Trk Pord IDID Trunk Pord Combo Ost Dig Locot4-Wire DDITS Trk Pord IDID Trunk Pord Combo Ost Dig Locot4-Wire DDITS Trk Pord IDID Trunk Pord Combo Ost Dig Locot4-Wire DDITS Trk Pord IDID Trunk Pord Combo Ost Dig Locot4-Wire DDITS Trk Pord IDID Trunk Pord Combo Ost Dig Locot4-Wire DDITS Trk k Pord Combo Ost Dig Locot4-Wire DDITS Trk Pord Combo Ost Doctore Pord Combo Ost Doctore Pord Trk DOCOTE Ost DOCTORED Trk DOCTORED Ost DOCTORED Ost DOCTORED Trk DOCTORED Ost DOCTORED Ost DOCTORED Ost DOCTORED Ost DOCTORED Ost DOCTORED Ost DOCTORED Ost DOCTORED Ost DOCTORED Ost DOCTORED Ost DOCTORED Os		-Wire DS1 Dig Loop/4-Wire DDITS Trk Port ombo-Sub Chan Activation-Per Chan-1- tay Inwrd Trk w/o DID	4-Wire DS1 Digital Loop/4-Wire DID Trunk Port Combo · Subsequent Channel Activation Per Channel		UEPDC	рттс		\$108.67		-1	\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
A With DS 1 Dig Loop 4-Wire DDITS Trk Port DID Trush Per Combones about Chan Activition Did Loop 4-Wire DDITS Trk Port DID Trush Per Combones and Use of DID Trush Per Combones and D	, , , ,	Wire DS1 Dig Loop/4-Wire DDITS Trk Port ombo-Sub Chan Activation-Per Chan-1- sy Inwd Trk with DID	4-Wire DS1 Digital Loop/4-Wire DID Trunk Port Combo - Subsequent Channel Activation Per Channel		UEPDC			\$108.67			\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
Fort Combination - Subsequent DDITS 4-Wire DSI Digital Loop 4-Wire DSI Digital Loop / 4-Wire DDITS 4-Wire DDI Trunk Port Combo - UEPDC USASA \$100 PDC \$100 P	, , ,	Wire DS1 Dig Loop/4-Wire DDITS Trk Port ombo-Sub Chan Activ'n-Per Chan-2-Way 10 with User Trans	4-Wire DS1 Digital Loop/4-Wire DID Trunk Port Combo - Subsequent Channel Activation Per Channel		UEPDC			\$108.67		_	\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
e DS1 Digital Loop / 4-Wire DDITS 4-Wire DDITS DID Trunk Port Combo. 1 EPDC USASA \$94.88 Port Combination - Subsequent Subsequent Subsequent Subsequent Dorder 1 EPDC USASA \$94.88 e DS1 Dig Loop/4-Wire ISDN DS1 Digital Loop 4-Wire DS2 DS1 Digital Loop 4-Wire DS2 DS2 DS2 DS3 DS3 DS3 DS3 DS3 DS3 DS3 DS3 DS3 DS3		-Wire DS1 Digital Loop / 4-Wire DDITS runk Port Combination - Subsequent ignaling Changes	4-Wire DS1 Digital Loop 4-Wire DID Trunk Port Combo - Subsequent Signaling Changes		UEPDC			\$22.92			\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
## DS1 Dig Loop/4-Wire ISDN DS1 ### DS1 Digital Loop 4-Wire box of Combonation - Subsequent Policy and Combonation - Subsequent Policy and Combonation - Subsequent Policy and Combonation - Subsequent Policy and Combonation of New Format - Conversion or New Install	1 - 0	-Wire DS1 Digital Loop / 4-Wire DDITS nuck Port Combination - Subsequent endo Order Per Order	4-Wire DS1 Digital Loop 4-Wire DID Trunk Port Combo - Subsequent Service Order Per Order		UEPDC			\$94.88	\$94.88		\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
Name Format - Conversion or New 10 10 10 10 10 10 10 1	7 3 6	Wire DS1 Dig Loop/4-Wire ISDN DS1 igital Trunk Port Combination - Subsequent elephone Numbers	4-Wire DS1 Digital Loop 4-Wire DID Trunk Port Combo - Subsequent Telephone Numbers		UEPDC			\$88.68			\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
ded Superframe Format - Conversion ••• UEPDC CCOEF \$0.00 \$0.00 w Install ••• UEPDC CCOSF \$0.00 \$590.00 riame Format - Subsequent Activity ••• UEPDC CCOEF \$0.00 \$590.00 Mark Inversion ••• UEPDC CCOEF \$0.00 \$590.00 Mark Inversion ••• UEPDC MCOSF \$0.00 ded SuperFrame Format ••• UEPDC MCOSF \$0.00 ded SuperFrame Format ••• UEPDC MCOSF \$0.00	748 1	LAR & ZERO SUBSTITUTION uperframe Format - Conversion or New stall		:	UEPOC			\$0.00	\$0.00							
ded SuperFrame Format *** UEPDC CCOSF \$0.00 \$590.00 ded SuperFrame Format *** UEPDC CCOEF \$0.00 \$590.00 Mark Inversion *** UEPDC MCOSF \$0.00 ded SuperFrame Format *** UEPDC MCOPP \$0.00 ded SuperFrame Format *** UEPDC MCOPP \$0.00		xtended Superframe Format - Conversion r New Install		:	UEPDX			\$0.00	\$0.00							
Mark Inversion ** UEPDC MCOSF \$0.00 ded SuperFrame Format ** UEPDC MCOPO \$0.00 e Number/Trunk Group Establisment ** UEPDC MCOPO \$0.00		uperframe Format - Subsequent Activity xiended Superframe Format - Subsequent civity		: :	UEPDC			\$590.00	\$590.00		\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
ded SuperFrame Format ** UEPDC MCOSF \$0.00 ded SuperFrame Format ** UEPDC MCOPO \$0.00	Alter	late Mark Inversion														
ded SuperFrame Format Oed SuperFrame Format NUEPDC MCOPO \$0.00		uperframe Format		:	UEPDC			\$0.00	\$0.00							
e Number/Trunk Group Establisment		xtended SuperFrame Format		:	UEPDX			\$0.00	\$0.00							
	Char	hone Number/Trunk Group Establisment jes														
Telephone Number for 2-Way Trunk Group		elephone Number for 2-Way Trunk Group		:	UEPDX	YDTGX					\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
Trunk Group		ereprodue nomber for 1-vray conward runk Group		:	UEPDX	UDTQY	_				\$3.50		\$19.99	\$19.99	\$20.00	\$20.00

Unbundled Network Elements TENNESSEE

						BATES				OSS	OSS RATES		
												Incremental Charge -	Charge -
	UNBUNDLED NETEORK ELEMENT AS	Not in TRA	2	9				ğ		Incremental	Incremental	Manual Svc	Manual Svc
UNBUNCED NET WORK ELEMENT	STATED IN DOCKET 87-01282	01262		} 			Nonrecurring	Submitted	3 5	Svc Order vs.	Svc Order va.	Electronic-Disc	Electronic-Disc
					ž	Norrecuring First Add1	First Add1	Per LSR	SOMAN	SOMAN	SOMAN	SOMAN	BOWAN
Telephone Number for 1-Way Inward Trunk		:	2001	70101	5			\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
DID Numbers, Establish Trunk Group and			j j	·-									
Provide First Group of 20 DID Numbers		:	UEPDC	ZQN	\$0.00			\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
Additional DID Numbers for each Group of 20 DID Numbers		:	UEPDC	Š	\$0.00			\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
DID Numbers, Non- consecutive DID Numbers , Per Number		:	UEPDC	NDS	\$0.00			\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
Decicated DS (Interoffice Channel Mileage)													
FXFCO for 4-Wire DS1 Dighal Loop with 4-Wire DDTS Trunk Port													
Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Termination)		:	UEPDC	1LNO1	\$75.83			\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
Interoffice Channel Mileage - Additional rate per mile - 0-8 miles		:	UEPDC	1LNOA	\$0.3525			\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Termination)		:	UEPDC	1LNO2	\$0.000								
Interoffice Channel Mileage - Additional rate per mile - 9-25 miles		:	UEPDC	1LNOB	\$0.3525			\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Termination)		:	UEPDC	1LNO3	\$0.000								
Interoffice Channel Mileage - Additional rate per mile - 25+ miles		:	UEPDC	1LNOC	\$0.3525	-		\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
Local Number Portability, per DS0 Activated		:	UEPDC		\$3.15								
Central Office Termininating Point		•	UEPDC	010	\$0.00								
ENHANCED EXTENDED LINK (EELs)													
		-											
NOTE: New (not currently combined) loop/transport (EEL) combinations are only evallable in density Zone 1 of the Nashville MSA in TN for customers w/ 4 or more lines.													
For Currently Combined EELs, network element recurring and Switch As is Charges apply. For new EELs, network element recurring and processuring and Switch As is Charmal processuring apply (no Switch As is Charma).									j		ļ		
da and and and and and and and and and a													
2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPO	H DEDICATED DS1 INTEROFFICE	TRANSPORT	(EEL)										
First 2-Wire Analog Voice Grade Loop - SL2/DS1 Interofficed Transport Combination - Zone 1	2-Wire Voce Grade Extended Loop With DS1 Dedicated Interoffice Transport (First 2-Wire Voice Grade With DS1 excluding mileage) - Zone 1		UNCVX	UEAL2	\$176.10								
First 2:Wire Analog Voice Grade Loop - SL2/DS1 Interofficed Transport Combination 2 one 2		~	XAONO ONO	UEAL2	\$181.17								
		, n	XXXX	UEAL2					<u> </u>				
be Transport - Dedicated - DS1	Interoffice Transport Dedicated DS1 - Per Mile		_										
				ė.	ł								Page 21 of 29

03/09/01

Exhibit 1-TV Attachment 2 Exhibit C

	_		_					C.	RATES					Ö	OSS RATES			
_																Incrementa	F	ncremental
		UNBUNDLED NETEORK ELEMENT AS	Not in TRA									Svc Order	Swc Order	Incremental	lat Incremental	Charge -		Manual Sve
	ONBONDLED NETWORK ELEMENT	STATED IN DOCKET 97-01262	01262 01262	Zove BCZ	20 25 	<u> </u>				Norre	Norrecurring	Submitted	Submitted	8,	Β.		Š	Order va.
								Norrecurring	ring	Diece	Disconnect	per LSR	LSR		Vit. Svc Order Vit.	de'i Electronia	7	Add1
				4	\dashv	Ц	ž	First	- PAG1	First	Addi	SOMEC	SOMAN	Н	_	Ц	Н	SOMAN
	Each Additional 2-Wire Analog VG Loop- SL2 in the same DS1 interoffice Transport Combination-Zone 1	2-Wire Vote Grade Extended Loop With DS1 Dedicated Interdise Transport (Additional 2- Wire Votes Grade Loop in same DS1 excluding mileage) - Zone 1	-	UNCVX	VX UEAL2		\$17.47											
	Each Additional 2-Wire Analog VG Loop- SL2 in the same DS1 Interoffice Transport Combination-Zone 2	2-Wire Voice Grade Extended Loop With DS1 Dedicated Interoffice Transport (Additional 2- Wire Voice Grade Loop in same DS1 exduding mileage) - Zone 2	2	NCVX	/X VEAL2		\$22.54											
	Each Additional 2-Wire Analog VG Loop- SL2 in the same DS1 interoffice Transport Combination-Zone 3	2-Wire Voice Grade Extended Loop With DS1 Dedicated Interoffice Transport (Additional 2- Wire Voice Grade Loop in same DS1 excluding mileage) - Zone 3	8	3 UNCVX	XX UEAL2		\$29.19									· · · · · · · · · · · · · · · · · · ·		
	Nonrecurring Currently Combined Network Elements Switch - As-is Charge	Non-Recurring Cost for Extended Loop or Local Channel and Interoffice Combination - Switch-asis		UNC1X	OOON XI	၁		\$52.73	\$24.62	\$9.12	\$9.12							
	4-WINE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)	H DEDICATED DS1 INTEROFFICE	TRANSPORT	(EEL)		-												
	First 4-Wire Analog Voice Grade Loop/DS1 Interoffice Transport Combination - Zone 1	4-Wire Voice Grade Extended Loop With DS1 Dedicated Interoffice Transport (First 4-Wire 56 or 64 kbps Digital Grade Loop with DS1 excluding mileage) - Zone 1	-	ONCX	VEAL4		\$184.24											
	First 4-Wire Analog Voice Grade Loop/DS1 interoffice Transport Combination - Zone 2	4-Wire Voice Grade Extended Loop With DS1 Dedicated Interoffice Transport (First 4-Wire 56 or 64 kbps Digital Grade Loop with DS1 excluding mileage) - Zone 2	2	 		<u> </u>	\$191.80											
	First 4-Wire Analog Voice Grade Loop/DS1 Interoffice Transport Combination - Zone 3	4-Wire Voice Grade Extended Loop With DS1 Dedicated Interoffice Transport (First 4-Wire 56 or 64 kbps Digital Grade Loop with DS1 excluding mileage) - Zone 3	6	UNCVX	X UEAL4		\$201.72											
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month	Interoffice Transport - Dedicated - DS1 - Per Mile		UNC1X	IX 1L5XX		\$0.3562											
1	Additional 4-Wire Analog Voice Grade Loop in same DS1 interoffice Transport Combination - Zone 1	4-Wire Voice Grade Extended Loop With DS1 Dedicated Interdifice Transport Additional 4- Wire 56 or 64 kbps in same DS1 excluding mileage) - Zone 1	-	UNCVX	/X UEAL4		\$25.61											
	Additional 4-Wire Analog Votce Grade Loop in same DS1 Interoffice Transport Combination - Zone 2	4-Wire Voice Grade Extended Loop Wirth DSI Dedicated Interoffice Transport (Additional 4- Wire 56 or 64 kbps in same DSI excluding mileage). Zone 2	2		UNCVX DEAL4		\$33.17											

Exhibit 1-TN Attachment 2 Exhibit C

			-	-	+		N.	RATES		r			SSO	OSS RATES		
					İ						-				Incremental Charge -	Incremental Charge -
UNBUNDLED NETWORK ELEMENT	UNBUNDLED METEORIK ELEMENT AS 8TATED IN DOCKET 97-01262	Docket \$7: 2	Zone BCS		nsoc			1	Nomecuring	Ę	Submitted	Submitted	Incremental Charge - Manual	Charge - Manual	Manual Svc Order va.	Manual Svc Order va.
				_			Normecurring	Ž.	Disconnect	128	per LSR	LSR	Electronic-1st	Electronic-Add1	1	Add1
			4	\dashv	\dashv	ž	First	Addi	Firet	LPP4	SOMEC	SOMAN	NYMOS	SOMAN	SOMAN	SOLLAN
Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport	4-Wire Voice Grade Extended Loop With DS1 Dedicated Interdifice Transport (Additional 4- Wire 56 or 64 kbps in same DS1 excluding mileage) -Zone 3		3 CNCVX	VX UEAL4		\$43.09										
Nonrecuring Currently Combined Network	Non-Recurring Cost for Extended Loop or Local Channel and Interoffice combination Switch-as-		1	X	CCC		\$52.73	\$24.62	\$9.12	\$9.12						
							Н									
4-WIRE 56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)	WTH DEDICATED DS1 INTEROFFI	CE TRANSP	ORT (EE	2												
First 4-Wire 56Kbps Digital Grade Loop/DS1	4-Wire 28 or 64 ktps: Extended Digital Loop With Dedicated DSJ Intercrition Transport (First 4-Wire 56 or 64 ktps: Digital Grade Loop with DSJ excluding mileage) - Zone 1		- CNCDX		UDLS8	90.09.			······							
First 4-Wire 56Kbps Digital Grade Loop/DS1 Intercelloe Transport Combination - Zone 2			2 UNCDX	DX UDL56		\$200.16										
First 4-Wire 56Kbps Digital Grade Loop/DS1 Interoffice Transport Combination - Zone 3			3 UNCDX		nDL58	\$212.85										
Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month	Interoffice Transport - Dedicated - DS1 - Per Mile		UNC1X	-	1L5XX	\$0.3562										
Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interdifice Transport Combination - Zone 1	4-Wire 56 or 84 kbps Extended Obgies Loop With Dedicated DS1 Interoffice Transport (Additional 4-Wire 56 or 64 kbps Digital Grade Loop in same DS1 excluding mileage) -Zone 1		1 UNCDX		9S TON	\$32.01										
Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 2	4-Wire 55 or 84 kbps Extended Digital Loop With Dedicated DS1 Interoffice Transport (Additional 4-Wire 56 or 64 kbps Digital Grade Loop in same DS1 excluding mileage) -Zone 2		2 UNCDX		UDLS&	\$41.52										
Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 3	4-Wire 56 of 54 kpps Extended Orgital Loop With Dedicated DS1 Interoffice Transport (Additional 4- Wire 56 or 64 kbps Digital Grade Loop in same DS1 excluding mileage) -Zone 3		3 UNCDX		UDLS8	\$54.02										
Nonrecurring Currently Combined Network Elements Switch - As-is Charge	Nonrecurring Cost for Extended Loop or Local Channel and interoffice Combination Switch-as- is		UNC1X		UNCCC		\$52.73	\$24.62	\$9.12	\$9.12						
4-WRE 64 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)	MTH DEDICATED DS1 INTEROFF	ICE TRANSP	ORT (EI	()	-											

Unbundled Network Elements

Exhibit 1-TN Attachment 2 Exhibit C

TENNESSEE	

f			-	-	-	-		74	RATES		-			OSS	OSS RATES		
						1_		-			-	-				Incremental Charge	Incremental Charge -
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01282	¥ .	Zone BC\$		2005				Norrecurring	Π	1 Y	Suc Order Submitted	=	Incremental Charge - Manuel		Mamual Svo Order vs.
			29216					1	1	2		_		Svc Order vs. Flectronics (et	Svc Order vs.	Electronic-Disc	Electronic-Disc Add1
							Plec	First Ac	Addi	First	ş	Ħ	SOMAN	SOMAN	$\overline{}$	\$OMAN	SOMAN
├		4-Wire 56 or 64 kbps Extended Digital Loop With Dedicated DS1													-		
		Interoffice Transport (First 4-Wire 56 or 64 kbps Digital Grade Loop															
	First 4-Wire 64Kbps Digital Grade Loop/DS1			- C	UNCDX	UDL64	\$190.64										
+-		4-Wire 56 or 64 kbps Extended Digital Loop With Dedicated DS1											-				
		Interoffice Transport (First 4-Wire															
	Clear 4. Wise 64V box Digital Grade 1	56 or 64 kbps Ligital Grade Loop with DS1 excluding mileson) -				-											
	Interoffice Transport Combination - Zone 2			2 	UNCDX UE	UDL64	\$200.16										
-		4-Wire 56 or 64 kbps Extended Digital I am With Pedicated DS1															
		Interoffice Transport (First 4-Wire															-
	First 4-Wire 64kbps Ligital Grade Look/UST Interoffice Transport Combination - Zone 3			3	UNCDX UE	UDL64	\$212.65										
\vdash	Interoffice Transport - Dedicated - DS1	Interoffice Transport - Dedicated -															
-	combination - Per Mile Per Month	DS1 - Per Mile		3	UNC1X 1L	1.5xx	\$0.3562										
		4-Wire 56 or 54 kbps Extended Digital Loop With Dedicated DS1															
	Additional A.WiseEdaKhos Osisies Oresie	Interoffice Transport (Additional 4-	•												_		
	Loopin same DS1 Interoffice Transport	Loop in same DS1 excluding															
_	Combination - Zone 1	mileage) -Zone 1		Š	NCDX	NDL64	\$32.01										
		4-Wire 56 or 64 kbps Extended Digital Loop With Dedicated DS1 interoffice Transport (Additional 4-															
	Additional 4-Wire546Kbps Digital Grade	Wire 56 or 64 kbps Digital Grade			-												
	Combination - Zone 2	mileage) -Zone 2		2	UNCDX UC	UDI.64	\$41.52										
		4-Wire 56 or 64 kbps Extended Digital Loop With Dedicated DS1															
		Interoffice Transport (Additional 4-	-														
	Additional 4-Wireb46Kbps Digital Grade Loopin same DS1 Interoffice Transport	Loop in same DS1 excluding															
	Combination - Zone 3	mileage) -Zone 3		3	UNCDX	UDLE	\$54.00										
	Acoustic Contraction Contraction	Nonrecuring Cost for Extended Loop or Local Channel and interoffice Combination Switch-as-										•					
	Elements Switch - As-Is Charge			Š	UNC1X UNCCC	ÇCC		\$52.73	\$24.62	\$9.12	\$9.12						
+-1																	
Ä	2-WIRE VOICE GRADE DEDICATED EXTENDED LOCAL CHANNEL WITH DEDICATED	D LOCAL CHANNEL WITH DEDI	CATED DS1	MERO	FFICE 1	HANSP	DS1 INTEROFFICE TRANSPORT (EEL)										
		Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local															
	First 2-Wire Local Channel with/DS1 Interoffice Transport Combination - Zone 1	Channel with DS1 excluding miteage) - Zone 1		-			\$176.72										
		Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local	140 18 14														
	First 2-Wire Local Channel with/DS1	Channel with DS1 excluding					8		_								
\dashv	Intercribe Transport Combination - 20ne 2	miesge) - core c		7	-		201.90										

Unbundled Network Elements
TENNESSEE

Exhibit 1-TN Attechment 2 Exhibit C

				-	\vdash	-		à	RATES					SSO	OSS RATES		
												-				Incremental Charge -	Incremental Charge -
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Docket 97- Zc	Zone BCS		2081				Norrecurring		Submitted	Sve Order Submitted		Incremental Charge - Manual	Manual Sve Order vs.	Manuel Svc. Order vs.
			78710					Normecurring	ring	Disconnect	J#Cf	Elec per LSR	Manuelly per LSR	Svc Order vs. Electronic-1st		Electronic-Disc 1st	Electronic-Disc Add1
				\dashv	\dashv	4	ş	First	Add1	First	Add'i	SOMEC	NYMOS	SOMAN	SOMAN	SOMAN	SOMAN
	First 2:Wire Local Channel with/DS1 Interoffice Transport Combination - Zone 3	Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with DS1 excluding mileage) - Zone 3			.		\$1 88.88										
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month	Interoffice Transport - Dedicated - DS1 - Per Mile		UNC1X	1X UITE		\$0.3562										
	Additional 2-Wire Voice Grade Channel in Same DSI Interoffice Transport Combination						8										
	Additional 2-Wire Voice Grade Channel in Same DSI interoffice Transport Combination Zone 2	Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated Local Channel with Dedicated DS1 Interoffice Transport (Additional 2-Wire Voice Grade Channel in same DS1 excluding mileage) -Zone 2		2			\$23.35										
	Additional 2-Wire Voice Grade Channel in Same DSI interoffice Transport Combination Zone 3						\$30.25										
	Norrecuring Currently Combined Network Elements Switch -As-is Charge	Nonrecuring Cost for Extended Loop or Local Channel and Interoffice combination Switch-as- is		ON S	UNC1X UNCCC	8		\$52.73	\$24.62	\$9.12	\$9.12						
4	WIRE VOICE GRADE DEDICATED EXTENDE	D LOCAL CHANNEL WITH DEDIC		TEROF	FICE TR	ANSPO	DS1 INTEROFFICE TRANSPORT (EEL)										
	Extended 4-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 interoffice Transport Channel with DS1 Interoffice Transport Combination - Zone 1 Interoffice Transport Combination - Zone 1 Interoffice Transport Combination - Zone 1	Extended 4-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 interoffice Transport (First 4-Wire Local Channel with DS1 excluding mileage) - Zone 1		_			\$177.73										
	First 4-Wire Local Channel with/DS1	Extended 4-Wire Voice Grade Dedicated Local Channel with Dedicated Local Interdition Transport (First 4-Wire Local Channel with DS1 excluding mileage) - Zone 2		~			\$183.29										
	First 4-Wire Local Channel with/DS1 Interdfice Transport Combination - Zone 3	Extended 4-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 4-Wire Local Channel with DS1 excluding mileage) - Zone 3		9		**	\$190.59										
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month	Interoffice Transport - Dedicated - DS1 - Per Mie		UNC1X	1X UITE	_	\$0.3562										

Unbundled Network Elements TENNESSEE

Exhibit 1-TN Attachment 2 Exhibit C

-				-				RATES					SO	OSS RATES		
									-					-	Incremental Charge	Charge
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORIX ELEMENT AS	Not in TRA Docket 97- Zone	9C\$	ngoc	Ĺ	_			Mooracurring	Svc Order Submitted		Incremental Charge - Manu	Incremental usi Charge - Manuel		
			01262						_		200	Memually per	r Sve Order vs.	Bve Order va.	Electronic-Disc	Electronic-Disc
						ž		Norsecurring First Add1	\mathbb{H}	Disconnect First Add'1	SOMEC	_	SOMAN	BOWAN	П	SOMAN
1-	Additional A.Wira Volve Grade Channel in	Extended 4-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transcort Additional 4-Wire Voice														
	Ē					\$19.09	8									
	al 4.Wire Voice Grade Channel in ISI Interoffice Transport Combination	Exfended 4-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (Additional 4-Wire Voice Grade Channel in same DS1 excluding mileage) -Zone 2		2		\$24.66	8									
	al 4-Wire Voice Grade Channel in ISI Interoffice Transport Combination	Extended 4-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (Additional 4-Wire Voice Grade Channel in same DS1 excluding milesce) -Zone D		6		\$31.96	8									
1	uring Currently Combined Network Is Switch -As-is Charge	Nonrecuring Cost for Extended Loop or Local Channel and Interoffice Combination Switch-as-		UNCIX	1X UNCCC	Ŋ	ις •9	\$52.73 \$24.62		\$9.12 \$9.12						
††	4 WARE DOT INCITED EXTENDED LOOP WITH DEDICATED DS INTEROFFICE TRANSPORT (EEL)	DEDICATED DS1 INTEROFFICE T	RANSPORT	(EEL)												
 	First 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1	Extended 4-Wire DS1 Digital Loop With Dedicated DS1 Interoffice Transpor (First 4-Wire DS1 Digital Loop with DS1 excluding mileage) - Zone 1		1 UNC1X	XX USLXX	XX \$135.59	69:									
†	First 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport -			2 UNC1X	1X USLXX	XX \$153.26	1.26									
1	First 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3	Extended 4-Wire DS1 Digital Loop With Dedicated DS1 Interoffice Transport (First 4-Wire DS1 Digital Loop with DS1 excluding mileage) Zone 3		3 UNC1X	X USLXX	XX \$176.45	1.45									
	Interoffice Transport - Dedicated - DS1 Interoffice Tran combination - Facility Termination Per Month DS1 - Per Mile	Interoffice Transport · Dedicated · DS1 · Per Mile		UNC1X	1X UITE1	F1 \$0.3562	282									
	Additional 4-Wire DS1 Digital Loop in same same DS1 Interoffice Transport Combination . Zone 1	Extended 4-Wire DS1 Digital Loop With Dedicated DS1 Interoffice Transport (Additional 4-Wire DS1 Digital Loop in same DS1 excluding mileage) - Zone 1		1 UNC1X	XXISU XX	XX \$57.73	57.									
	Extended 4-Wire DS1 Dig With Dedicated DS1 Interval Additional 4-Wire DS1 Digital Loop in same Transport (Additional 4-Wisame DS1 Intervalice Transport Combination Digital Loop in same DS1 - Zone 2 excluding mileage) - Zone	Extended 4-Wire DS1 Digital Loop With Dedicated DS1 Interoffice Transport (Additional 4-Wire DS1 Digital Loop In same DS1 excluding milesge) - Zone 2		2 UNC	UNC1X USLXX	XX \$75.40	04.									

Unbundled Network Elements TENNESSEE

03/09/01

				-			RATES					988	OSS RATES		
		į								_				Incremental Charge -	Incremental Charge -
UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS 8TATED IN DOCKET 97-01282	Docket 97-	\$00e BC\$	naoc		1		Norrecuring	urring	2 2	Sve Order Submitted	Incremental Charge - Manual	Incremental herge - Manual	Manual Sve Order vs.	Manuel Svc Order vit.
						Nonrecurring	gunna	Disconnect	mect		Menuany per LSA	Electronio-1st	Sections-Add	1st	Add1
			+	\downarrow	ž	Fire	Addi	First	Addi	SOMEC	SOMAN	SOMAN	SOMAN	BOMAN	SOMAN
Additional 4-Wire DS1 Digital Loop in same same DS1 interoffice Transport Combination . Zone 3			3 UNC1X	X NSLXX	\$98.59										
Nonrecurring Currently Combined Network Elements Switch - As-Is Charge	Nonrecurring Cost for Extended Loop or Local Channel and Interoffice Combination Switch-as-		UNC1X	OD ON O		\$52.73	\$24.62	\$9.12	\$9.12						
			-												
Currently Combined Network Transport Elements (Non-Switched Combinations Resulting	ents (Non-Switched Combinations		from a Conversion)	iversion)											
Local Channel - Dedicated - 2-Wire VG Monthly Recurring per month		:	UNCVX	/X ULDV2	\$19.43	\$199.33	\$24.18	\$54.81	\$4.80			\$20.35	\$10.54	\$13.30	
Monthly Recurring per mile per month		:	UNCVX		Ш										
oral Channel - Dedicated - 3-Wire VG - Ray			+												
Monthly Recurring per month		:	UNCVX	/X ULDR2	\$19.43	\$199.33	\$24.16	\$54.81	\$4.80			\$20.35	\$10.54	\$13.30	
Monthly Recurring per mile per month		:	UNCVX	/X 1L5NC	\$0.00										
Local Channel - Dedicated - 4-Wire VG															
Monthly Recurring per month		:	UNCVX	/X ULDV4	\$20.58	\$201.53	\$24.83	\$55.52	\$5.51			\$20.35	\$10.54	\$13.30	
Monthly Recurring per mile per month		:	UNCVX	/X 1LSNC	\$0.00										
			+												
Local Channel - Dedicated - US1		:	1 INC 1 X	X 111 DF1	\$20.58	\$201.53	\$24.83	\$55.52	\$5.51			\$20.35	\$10.54	\$13.30	
Monthly Recurring per mile per month		:	UNC1X		Ш										
				-											
Interoffice Channel - Dedicated - 2-wire VG			+												
per mile per month		:	UNCVX	/X 1L5XX	\$0.0174										
Interoffice Channel - Dedicated 2-wire VG - Facility Termination per month		:	UNCVX	SVT1U X/	\$18.58	\$55.39	\$17.37	\$27.96	\$3.51			\$20.35	\$21.09	\$9.80	\$10.54
Interoffice Channel - Dedicated - 2-wire VG Rev Battery	ev Battery														
Interoffice Channel • Dedicated 2-wire VG • per mile per month		:	UNCVX	/X 1LSXX	\$0.02										
nteroffice Channel - Dedicated 2-wire VG - Facility Termination per month		:	UNCVX	/X UITR2	\$18.58	\$55.39	\$17.37	\$27.96	\$3.51			\$20.35	\$21.09	\$9.80	\$10.54
			+												
Interoffice Channel - Dedicated - Sokops Interoffice Channel - Dedicated - 56 kbps - ber mile per month		:	UNCDX)X 1L5XX	\$0.0174										
Interoffice Channel - Dedicated - 56 kbps - Facility Termination per month		:	CUNCDX	OX UITDS	5 \$17.98	\$55.39	\$17.37	\$27.96	\$3.51			\$20.35	\$21.09	\$9.80	\$10.54
interoffice Channel - Dedicated - 64kbps				+											
Interoffice Channel - Dedicated - 64 kbps - per mile per month		:	NCDX	X 1L5XX	\$0.02										
Interoflice Channel - Dedicated - 64 kbps - Facility Termination per month		:	UNCDX	X U1TD6	817.98	\$55.39	\$17.37	\$27.96	\$3.51			\$20.35	\$21.09	\$9.80	\$10.54
			H		Ц										

Unbundled Network Elements	TENNESSEE

03/09/01

				F	-	-		ğ	RATES					088	OSS RATES		
			į			<u> </u>						-				Incremental	Incremental Charge -
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97- Z 01262	g gove	BCS US	2082				Nonrecurring				Incremental Charge - Manuel Svc Order vs.		Manual Sve Order vs. Electronic-Disc	Manual Sve Order vs. Electronic-Disc
						_		Norrecuring	Bull	Disconnect	Jac.	_		Electronic-1st	Electronic-Add1		LDQQ.
1			1	+	+	+	2	Į.	- Vadd	i i	F AG	SOMEC	SOMAN	SOMAN	BOWAN	SONAN	SOMAN
	Interoffice Channel - Dedicated - DS1	A service of the first of the state of the s		+	+	+											
	mile per month		:	ś	UNC1X 1L5	1L5XX	\$0.3562										
	Interoffice Channel - Dedicated - DS1 - Facility Termination per month		:	3	UNC1X U1	U1TF1	\$77.86	\$112.40	\$78.27	\$19.55	\$14.99			\$20.35	\$21.09	\$9.80	\$10.54
				+	+	+	+										
	Local Loop - Dedicated - 2-Wire VG		1	+		+						Ì					
	Local Loop - Dedicated - Z-Wire VG - Monthly Recurring per month - Zone 1		:	- S	UNCVX UE	UEAL2	\$16.56	\$75.08	\$48.20	\$28.70	\$17.64			\$20.35	\$10.54	\$13.32	
	Local Loop - Dedicated - 2-Wire VG -		:	2	XACAN	I IEAL 2	63163	£100 07	\$140.72	428 70	\$17.84			\$20.35	\$10.54	\$13.32	
	Local Loop - Dedicated - 2-Wire VG -			+	_	\perp	$^{+}$	+	+-		5			95.0.00			
	Monthly Recurring per month - Zone 3		:	S 0	UNCVX UE	UEAL2	\$28.28	\$192.97	\$140.72	\$28.70	\$17.64			\$20.35	\$10.54	\$13.32	
	Local Loop - Dedicated - 2-Wire VG - Monthly Recurring per mile per month		:	5	UNCVX 1L5	1L5ND	\$0.00										
				+	_	\dashv											
1	Local Loop - Dedicated - 2-Wire VG - Rev Bat.				+	+					İ						
	Bat - Monthly Recurring per month - Zone 1		:	2	UNCVX UE	UEAR2	\$16.56	\$75.06	\$48.20	\$28.70	\$17.64			\$20.35	\$10.54	\$13.32	
	Local Loop - Dedicated - 2-Wire VG - Rev		:	-	AVONI	COVO	631.83	600	64.00.73	628 70	£17.64			£20 35	£10 54	61333	
1	Local Loco - Dedicated - 2-Wire VG - Rev	ender som meder der et der der geste der stade stade de de delenge green de de service de la service		+	-	ļ.,	t	1	1								
	Bat - Monthly Recurring per month - Zone 3		:	8	UNCVX UE	UEAR2	\$28.28	\$192.97	\$140.72	\$28.70	\$17.64			\$20.35	\$10.54	\$13.32	
****	Monthly Recurring per mile per month		:	Š	UNCVX 1L5	1L5ND	\$0.00										
٩	Local Loop - Dedicated - 4-Wire VG			_		+		1									
	Local Loop - Dedicated - 4-Wire VG - Monthly Recurring per month - Zone 1		:	<u>s</u>	UNCVX	UEAL4	\$24.70	\$122.78	\$85.57	\$78.35	\$39.16			\$20.35	\$10.54	\$13.32	
	Local Loop - Dedicated - 4-Wire VG -		:	2	XAONI	I IF A! 4	36 663	8122 7B	£85.57	£78.35	\$39.16	-		\$20.35	\$10.54	\$13.32	
	Local Loco - Dedicated - 4-Wire VG -			Ť	_	L	╁	╁	200								
	Monthly Recurring per month - Zone 3	***************************************	:	5	UNCVX UE	UEAL4	\$42.17	\$122.78	\$85.57	\$76.35	\$39.16			\$20.35	\$10.54	\$13.32	
	Monthly Recurring per mile per month		:	3	UNCVX 1L5	1L5ND	\$0.00			-							
	Local Loop - Dedicated - 2-Wire ISDN Digital			H													
	Local Loop - Dedicated - 2-Wire ISDN -		:	<u> </u>	INCVX UI	U11.2X	\$22.00	\$142.78	\$88.88	\$76.35	\$39.16			\$20.35	\$10.54	\$13.32	
	Local Loop - Dedicated - 2-Wire ISDN -			1	+	_										3	
	Monthly Recurring per month - Zone 2		:	2	ONCVX	NICZX OICZX	\$29.02	\$142.78	\$88.88	\$78.35	\$39.16			\$20.35	\$10.5	313.32	
	Monthly Recurring per month - Zone 3		:	Σ 5	UNCVX U1	U1L2X	\$37.95	\$142.76	\$88.88	\$76.35	\$39.16			\$20.35	\$10.54	\$13.32	
	Monthly Recurring per mile per month		:	5	UNCVX 1L5	1L5ND	\$0.00										
	Dedicated A Wise ER Phone	in the state of th		+	+	+											
1	Local Loop - Dedicated - 4-Wire 56 KDps [Local Loop - Dedicated - 4-Wire 56 Kbps			+	+	+					Г						
	Digital - Monthly Recurring per month - Zone		:	ž 5	UNCVX	UDL56	\$31.10	\$207.01	\$141.38	\$90.70	\$44.18			\$20.35	\$10.54	\$13.32	
	Local Loop - Dedicated - 4-Wire 56 kbps Digital - Monthly Recurring per month - Zone		:		ZI XV	4	100	6207	614138	02 20	444 18			\$20.35	2105	\$13.32	
	Local Loop - Dedicated - 4-Wire 56 kbps			\top	+-		\dagger	+									
	Digital - Monthly Recurring per month - Zone		:	ž 5	ONCVX UD	UDLS6	\$53.11	\$207.01	\$141.38	\$90.70	\$44.18			\$20.35	\$10.54	\$13.32	
1	***************************************				4	l	1										

Unbundled Network Elements TENNESSEE

03/09/01

Exhibit 1-TN Attachment 2 Exhibit C

				-	-	-		C	RATES					980	OSS RATES		
						<u> </u>						-				Incremental Cherge -	Incremental Charge
UNBUNDLED NETWORK ELEMENT	LEWENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Docket 97-2	Zone	5 50	2080			'	Nonrecuring	urting	Submitted	Submitted	<u>~</u> _		. 3	Manual Sve Order vs.
	•				-			Nonrecuring	ming	Disconnect	mect	Elec per LSR	Manuelly per LSR	Svc Order vs.	Electronic-Add ¹		Addi
				\dashv	\dashv	+	ž	First	PQ41	Frest	VQQ.	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
4-Wire 56 kbps Digital Grade Loop per mile per month	Loop per mile		:	5	UNCVX 1L	1L5ND	\$0.00										
2				1		+											
Local Loop - Dedicated - 4-Wire 54 lops	St Ichis			+	1	+											
Local Loop - Dedicated - 4-Virie 64 Kops Digital - Monthly Recurring per month - Zone	re o4 kops r month - Zone		:	<u> </u>	UNCVX	UDL64	\$31.10	\$207.01	\$141.38	\$90.70	\$44.18			\$20.35	\$10.54	\$13.32	
Local Loop - Dedicated - 4-Wire 64 kbps Digital - Monthly Recurring per month - Zone	re 64 kbps r month - Zone		:	8 8	UNCVX	UDL64	\$40.61	\$207.01	\$141.38	\$90.70	\$44.18			\$20.35	\$10.54	\$13.32	
Local Loop - Dedicated - 4-Wire 64 kbps Digital - Monthly Recurring per month - Zone	re 64 kbps r month - Zone		:	<u>န</u>	ONCVX	UDL64	\$53.11	\$207.01	\$141.38	\$90.70	\$44.18			\$20.35	\$10.54	\$13.32	
4-Wire 64 kbps Digital Grade Loop per mile per month	Loop per mile		:	5	UNCVX 1L	1L5ND	\$0.00										
Local Loop - Dedicated - DS1																	
Local Loop - Dedicated - DS1 - Monthly Recurring per month - Zone 1	- Monthly		:	5	UNC1X US	nscxx	\$57.73	\$313.08	\$219.72	\$96.86	\$40.45			\$18.98	\$8.43	\$11.95	
Local Loop - Dedicated - DS1 - Monthly Recurring per month - Zone 2	- Monthly		:	2 U	UNC1X	USLXX	\$75.40	\$313.08	\$219.72	\$96.86	\$40.45			\$18.98	\$8.43	\$11.95	
Local Loop - Dedicated - DS1 - Monthly Recurring per month - Zone 3	· Monthly		:	5	UNC1X US	ascxx	\$98.59	\$313.08	\$219.72	\$96.86	\$40.45			\$18.98	\$8.43	\$11.95	
Local Loop - Dedicated - DS1 - Monthly Recurring per mile per month	· Monthly		:	5	UNC1X 1L	1LSND	\$0.00										
DS1 Channelization														30,000	9	07.14	44 40
DS1 to DS0 Channel System per month	per month		:) 	CNC1X	5	\$80.77	\$141.87	\$77.11	514.51	\$13.46			\$20.35	99.00	6	91.10
OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kbs)	S DS0 Channel		:	5	UNCDX 10	10100	\$1.82	\$6.07	27.00					\$20.35	\$9.80	\$11.49	\$1.18
2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System - per month	0\$1 to D\$0		:	5	UNCNX	UC1CA	\$3.10	\$6.07	\$4.68					\$20.35	\$9.80	\$11.49	\$1.18
Voice Grade COCI - DS1 to DS0 Channel System - per month	S0 Channet		:	ā	UNCVX 10	101VG	\$0.91	\$6.07	\$4.66					\$20.35	\$9.80	\$11.49	\$1.18
MULTIPLEXERS				+	4	1		2000	27		40.00			400.36	0000	611.70	61 1B
DS3 to DS1 Channel System per month	per month		: :) 	4	g g	82228	\$308.03	\$106.47	344.4/	20.764			\$50.33	00.00	611 40	
STS1 to DS1 Channel System per month	per month			7	ONCSX	Š	\$222.98	\$308.03	4.00.47	14.44	20.04			50.039	200		
Loop per month	i) used with		:	5	UNC1X	UC1D1	\$17.58	\$8.07	\$4.66					\$20.35	\$9.80	\$11.49	\$1.18
* Rates for these elements, products or a	vervices were not or	Rates for these elements, products or services were not ordered by the TRA in Docket No. 97-01292. However, auch elements, products or services are necessary for the provision of other elements, products or services for which permanent alless were ordered	However, s	uch elem	ents, produ	Cts or servi	ICES SATE DECEMB	any for the pro	wiston of other	er elements.	anducts or se	evices for whi	ch permanent	rates were ordere	4.		
** Rates for these elements, products or t	services are related	** Rates for these elements, products or services are related to a combination(s) ordered by the TRA in Docket No. 97-01262.	n Docket No.	97-0126	1	4											

LOCAL INTERCONNECTION Tennessee

10/60/00

			\vdash	H			RATES	-		_		So	OSS RATES		
						-	_	\mid	L		L			Incremental	Incremental
Frank is advantaged as straight	UNBUNDLED NETWORK ELEMENT AS STATED IN	Not in TRA	200	*5	2005			\dagger		1			Incremental	Manual Svc	Manual Svc
	DOCKET 97-01262						:		Nonrecurring	Sub-mitted Flee	Submitted Manually per	Charge - Manual Svc Order va.	Charge - Manuel Svc Order vs.	Order vs. Electronic-Disc	Order vs. Electronic-Disc
					=	2	First A	Add:1	First Add'	SOMEC	_		SOMAN	SOMAN	SOMAN
			H	H											
LOCAL INTERCONNECTION (CALL TRANSPORT AND TERMINATION)	AND TERMINATION)			+											
END OFFICE SWITCHING															
End Office Switching Function, Per MOU	End office switcing function				\$0.00	\$0.0008041	1	1		1					
TANDEM SWITCHING			\dagger	+	-	-	-	-	-						
Tandem Switching Function Per MOU	Tandem switching function				\$0.00	\$0.0009778									
			\dagger				1	+	1						
LOCAL INTERCONNECTION (TRANSPORT)							-	+							
COMMON TRANSPORT (Shared)	\neg		H												
Common Transport - Per Mile, Per MOU	-		1		\$0.0¢	\$0.0000064		+							
Common Transport - Facilities Termination Pe	er Common Transport - Facilities Termination per MOU				\$0.00	\$0.0003871									
TO BOOK TO MOTOR APPROACH STANKED TO THE STANKED TO	TOPOGE STATE		\dagger	+		-		+	+						
Interoffice Channel - Dedicated Transport - 2-	Interoffice Transport - Dedicated - Voice		=	7/4	1 50	22100									
Wire Voce Grade - Per Mile per month Internetice Change - Dedicated Transport 2.	Grade		+	+	4	*		1	-						
Wire Voice Grade - Facility Termination per month	Interoffice Transport - Dedicated - 2-Wire Voice Grade - Facility Termination		٥	1 XVT IU	1L5XF \$18.	82	\$55.39 \$1	\$17.37 \$2	\$27.96 \$3.51			\$20.35	\$21.09	\$9.80	\$10.54
NACTOR CHANNEL DENCATED	TOCOON TAKEN		+	+				+		_					
INTEROFFICE CHANNEL - DEDUCATED INANSTON - 3494 NOTS	Interesting Transport Dedicated DEC		-	1			1	+	I	-					
Interoffice Channel - Dedicated Transport - 56 kbcs - ner mile per month	6 56/64 kbps - Interoffice Transport - Dedicated DSO - Dedicated - DSO - Der mile			- XQTIN	1L5XK \$0.(\$0.0174	•								
Interoffice Channel - Dedicated Transport - 56			-												
kbps - Facility Termination per month	Dedicated - DSO - Facility Termination		د	E XQL	1L5XK \$17.	8	\$55.39	\$17.37	\$27.96 \$3.51	-		\$20.35	\$21.09	\$9.80	\$10.54
Interoffice Channel - Dedicated Transport - 64 kbps - per mile per month			⊃	L XQT IU	1L5XK \$0.0	\$0.0174									
Interoffice Channel · Dedicated Transport · 64 kbps · Facility Termination per month				×QLID	1L5XX \$11	\$17.98	\$55.39	\$17.37	\$27.96 \$3.51			\$20.35	\$21.09	29.60	\$10.54
			H	H											
INTEROFFICE CHANNEL - DEDICATED TRANSPORT - DS1	4SPORT - DS1		\dagger	+					1						
Interoffice Channel - Dedicated Channel - DSI - Interoffice Transport - Dedicated - DSI Per Mile per month	per mile.		د	UITDI	1L5XIL \$0.	\$0.3562									
Interoffice Channel - Dedicated Tranport - DS1 - Interoffice Transport - Dedicated - DS1 Facility Termination per month	51 - Interoffice Transport - Dedicated - DS1 - Facility, Termination		ے	11701	1L5X1L \$7	\$77.86 \$112.	\$	\$78.27	\$19.55 \$14.	8		\$20.35	\$21.09	\$9.80	\$10.54
LOCAL CHANNEL - DEDICATED TRANSPORT			+	+	+		+		-						
Local Channel - Deldicated - 2-Wire Voice				-	_		\vdash	_				50	***		800
Grade per month - Zone 1 Local Channel - Deldicated - 2-Wire Voice	Local Channel - Dedicated - 2-Wire Voice		+	- - - - -	\perp	D1./14	+-	924.10	+-			\$50.50	5	8.5	33
Grade per month · Zone 2	_		7	E XAOD	TEFV3 \$2	\$22.44	\$199.33	\$24.16	\$54.81 \$4.80	0		\$20.35	\$10.5	\$13.30	20 00
Local Cahnnel - Dedicated 2-Wire Voice Grade per month - Zone 3			3	ULDVX	TEFV3 \$2	\$29.34 \$19	\$199.33	\$24.16	\$54.81 \$4.8	98		\$20.35	\$10.54	\$13.30	\$0.00
Local Channel - Dedicated - 4-Wire Voice Grade per month - Zone 1	Local Channel - Dedicated - 4-Wire Voice Grade (shown here desveraged)			T XOON	TEFV4 \$1	\$18.18	\$201.53	\$24.83	\$55.52 \$5.51	_		\$20.35	\$10.54	\$13.30	80.00
Local Channel - Dedicated - 4-Wire Voice	Local Channel - Dedicated - 4-Wire Voice		T	-	L		├	_	-						
Grade per month · Zone 2	Grade (shown here deaveraged)		7 7	T XOON	TEFV4 \$2	\$23.74 \$24	\$201.53	\$24.83	\$55.52 \$5.5	15		\$20.35	\$10.54	\$13.30	80.03
Grade per month - Zone 3			3	T	TEFV4 \$3	\$31.05	\$201.53	\$24.83	\$55.52 \$5.51	-		\$20.35	\$10.54	\$13.30	\$0.00
Local Channel - Dedicated - DS1 per month - Zone 1				ULDO1	TEFHG \$3	\$36.24 \$2	\$277.35	33.28	\$233.26 \$33.18 \$22.30	8		\$45.68	\$1.78	\$21.75	\$1.76
			1			1									

LOCAL INTERCONNECTION Tennessee

Exhibit 1-TM Attachment 3 Exhbit A

+				\vdash				RA	RATES					SSO	OSS RATES		
			i									F				Incremental Charge	Incremental Charge
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETWORK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97- Zone	Zone	5	2085				Nonrecurring		Sve Order	Submitted (Manual Svc Order vs.	Menuel Svc Order vs.
			26.		_	1		Moorecuring	irthe	Disconnect		_	Menually per LSR	Svc Order vs. Electronic-1st	Svc Order vs. Electronic-Add1	Electronic-Disc Electronic-Disc 1st Add1	Sectronic-Disc
						<u> </u>	ž	First	L PPY	First	LPPY	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
+				\dagger		l											
+	Local Channel - Dedicated - DS1 per month -	Local Channel - Dedicated - Dedicated		-		-									;		-
_	Zone 2	DS1 [shown here deaveraged]		2	10010	드라	\$47.33	\$277.35	\$233.26	\$33.18	\$22.30			\$45.68	\$1.76	\$21.75	0/.
+	hannel - Dedicated - DS1 per month -	Local Channel - Dedicated - Dedicated		_	_										1		,
		DS1 [shown here deaveraged]		3	ULDD1 T	TEFHG	\$61.89	\$277.35	\$233.26	\$33.18	\$22.30			\$45.68	\$1.78	\$21.75	\$1.78
\vdash				-	1												
1	MULTIPLEXERS			1													
+-		Channelization - Channel System DS1 to			_									30 000	9	•	9
	Channelization - DS1 to DS0 Channel System DSO	oso		-	S TOUR	SATC1	\$80.77	\$141.87	\$77.11	\$14.51	\$13.46			\$20.35	20.00	DT	0
+-	OCU-DP COCI (data) - DS1 to DS0 Channel	Interface Unit - Interface DS1 to DSO -			_			;						30 004	2	9	9
_		OCU - DP Card		-	UDL	SATSA	\$1.82	\$6.07	8.56					\$20.35	33.00	D	9
+-	2-wire ISON COCI (BRITE) - DS1 to DS0	Interface Unit - Interface DS1 to DSO -			-	-				_	_			10 000		•	•
_	Channel Systsem - per month	Brite Card		 	S NO	SATSA	\$3.10	\$6.07	25 26					\$20.35	28.00	P11.43	0
\vdash	Voice Grade COCI - DS1 to DS0 Channel	Interface Unit - Interface DS1 to DSO -			_									;	-		
_	System - per month	Voice Grade Card			NEA S	SATSA	\$.91	\$6.07	2 8					\$20.35	29.80	\$11.49	91.16
} −		Channelization - Channel System DS3 to		_						:	- 1				9		9
_	DS3 to DS1 Channel System per month	DS1		ر	SECTIVE	SATCS	\$222.98	\$308.03	~	244.47	\$42.62			\$20.35	29.00	54.1.43	0
╁	DS3 Interface Unit (DS1 COCI) per month	Interface Unit - Interface DS3 to DS1			USL S	SATCO	\$17.58	\$6.07	2					\$20.35	\$9.80	\$11.49	\$1.18
\vdash																	
+					-	-					_	-		_	_	_	

Exhibit 1-TN Attachment 7 Exhbit A

ODUF/ADUF/CMDS Tennessee

				H		_	Ą	RATES		L			OSS RATES	TES		
							-	L			-		-		Incremental	Incremental
			Mar in Till									_	•			5
		UNBUNDLED NETWORK ELEMENT AS STATED IN	Doctor 87. 7nns		1800	۶.				Q ox		PCret	mental	Incremental	Manual Svc	Manual Svc
	UMBUMULED MEI WORN ELEMEN	DOCKET 97-01262	01282						Nonrecurring	Submitted		Serge Serge	- Menuel G	uarge - Manual	Submitted Charge - Manual Charge - Manual Order vs. Order vs.	Order vs.
										_		O Sác Sác	rder vs. S	w Order va.	Sectronic Disc	Electronic-Disc
_				_	_		Ę.	Nonrecurring	Disconnect	ect per LSR		Electric	ONIC-181	ectronic-Add1	٤	100
						Pec	First	Add1	First	Add1 SOMEC		SO.	SOMAN	SOMAN	SOMAN	SOMAN
l				-	-	_										
OPLIFIE	ODUF/EDOUF/ADUF/CMDS			-	_			L								
					<u> </u>											
ō	OPTIONAL DAILY USAGE FILE (ODUF)															
		OSS OLEC Daily Usage File Recording									. =					
	ODUF: Recording, per message	per message			Ż	N/A \$0.0000044	9044					-				
		OSS OLEC Daily Usage File Message		_		┝										
	ODUF: Message Processing, per message	Distribution per message		_	z	N/A \$0.0027366	7366									
		OSS OLEC Daily Usage File Message		_												
	ODUF: Message Processing, per Magnetic	Distribution per magnetic tape	_						_							
	Tape provisioned	provisioned			Ż	NA \$52.75	75	\downarrow								
		OSS OLEC Daily Usage File Data														
	ODUF: Data Transmission	Transmission (Connect: Direct), per														
_	(CONNECT:DIRECT), per message	пеззаде			Ż	NA \$0.0000339	9339						-			
			_	_		_	_	_		-		_				

2-WIRE ANALOG VOICE GRADE LOOP

4-WIRE ANALOG VOICE GRADE LOOP

2-WIRE ISDN DIGITAL GRADE LOOP

2-WIRE ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOOP

2-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP

4-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP

4-WIRE DS1 DIGITAL LOOP

4-WIRE 56 OR 64 KBPS DIGITAL GRADE LOOP

SUB-LOOP DISTRIBUTION

SUB-LOOP FEEDER

NETWORK INTERFACE DEVICE (NID)

UNBUNDLED LOOP CONCENTRATION

UNBUNDLED SUB-LOOP CONCENTRATION (OUTSIDE CO)

UNBUNDLED LOCAL EXCHANGE SWITCHING(PORTS)

UNBUNDLED LOCAL SWITCHING, PORT USAGE

COMMON TRANSPORT (SHARED)

INTEROFFICE CHANNEL - DEDICATED TRANSPORT - VOICE GRADE

INTEROFFICE CHANNEL - DEDICATED TRANSPORT- 56/64 KBPS

INTEROFFICE CHANNEL - DEDICATED TRANSPORT - DS1

LOCAL CHANNEL - DEDICATED TRANSPORT

MULTIPLEXERS

DARK FIBER

8XX ACCESS TEN DIGIT SCREENING

LINE INFORMATION DATA BASE ACCESS (LIDB)

SIGNALING (CCS7)

SELECTIVE ROUTING

AIN - BELLSOUTH AIN SMS ACCESS SERVICE

AIN - BELLSOUTH AIN TOOLKIT SERVICE

OPTIONAL DAILY USAGE FILE (ODUF)

UNBUNDLED PORT/LOOP COMBINATIONS - COST BASED RATES

2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)

2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)

2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)

2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)

2-WIRE VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK PORT

2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT

4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK PORT

4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT

2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)

4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)

4-WIRE 56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)

4-WIRE 64 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)

2-WIRE VOICE GRADE DEDICATED EXTENDED LOCAL CHANNEL WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)

4-WIRE VOICE GRADE DEDICATED EXTENDED LOCAL CHANNEL WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)

4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)

·	·.		

INTERCONNECTION AGREEMENT BETWEEN BELLSOUTH TELECOMMUNICATIONS INC. AND NEWSOUTH COMMUNICATIONS CORP.

TABLE OF CONTENTS

General Terms and Conditions

- 1. Definitions
- 2. Term of the Agreement
- 3. Operational Support Systems
- 4. Parity
- 5. White Pages Listings
- 6. Bona Fide Request/New Business Request Process for Further Unbundling
- 7. Court Ordered Requests for Call Detail Records and Other Subscriber Information
- 8. Liability and Indemnification
- 9. Intellectual Property Rights and Indemnification
- 10. Proprietary and Confidential Information
- 11. Assignments
- 12. Resolution of Disputes
- 13. Taxes
- 14. Force Majeure
- 15. Adoption of Agreements
- 16. Modification of Agreement
- 17. Non-waiver of Legal Rights
- 18. Severability
- 19. Waivers
- 20. Governing Law
- 21. Arm's Length Negotiations
- 22. Notices
- 23. Rule of Construction
- 24. Headings of No Force or Effect
- 25. Multiple Counterparts
- 26. Implementation of Agreement
- 27. Filing of Agreement
- 28. Compliance with Applicable Law
- 29. Necessary Approvals
- 30. Good Faith Performance
- 31. Nonexclusive Dealings
- 32. Survival
- 33. Entire Agreement

Attachment 1 - Resale

- Attachment 2 Network Elements and Other Services
- Attachment 3 Network Interconnection
- Attachment 4 Physical Collocation
- Attachment 5 Access to Numbers and Number Portability
- Attachment 6 Pre-Ordering, Ordering and Provisioning, Maintenance and Repair
- Attachment 7 Billing and Billing Accuracy Certification
- Attachment 8 Rights-of-Way, Conduits and Pole Attachments
- Attachment 9 Performance Measurements
- Attachment 10- Agreement Implementation Template
- Attachment 11- BellSouth Disaster Recovery Plan
- Attachment 12-Bona Fide Request and New Business Requests Process

AGREEMENT

THIS AGREEMENT is made by and between BellSouth Telecommunications, Inc., ("BellSouth"), a Georgia corporation, and NewSouth Communications, Corp., ("NewSouth") a Delaware corporation, and shall be deemed effective as of the date of the last signature of both Parties ("Effective Date"). This Agreement may refer to either BellSouth or NewSouth or both as a "Party" or "Parties."

WITNESSETH

WHEREAS, BellSouth is an Incumbent Local Exchange Telecommunications Company (ILEC) authorized to provide telecommunications services in the states of Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina and Tennessee; and

WHEREAS, NewSouth is or seeks to become a Competitive Local Exchange Telecommunications Company ("CLEC") authorized to provide telecommunications services in the states of Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina, and Tennessee; and

WHEREAS, the Parties wish to resell BellSouth's telecommunications services and/or interconnect their facilities, purchase network elements and other services, and exchange traffic specifically for the purposes of fulfilling their obligations pursuant to sections 251 and 252 of the Telecommunications Act of 1996 ("the Act").

NOW THEREFORE, in consideration of the mutual agreements contained herein, BellSouth and NewSouth agree as follows:

1. Purpose

The Parties agree that the rates, terms and conditions contained within this Agreement, including all Attachments, comply and conform with each Parties' obligations under sections 251 and 252 of the Act. The resale, access and interconnection obligations contained herein enable NewSouth to provide competing telephone exchange service to residential and business subscribers within the territory of BellSouth. The Parties agree that NewSouth will not be considered to have offered telecommunications services to the public in any state within BellSouth's region until such time as it has ordered services for resale or interconnection facilities for the purposes of providing business and/or residential local exchange service to customers.

2. Term of the Agreement

2.1 The term of this Agreement shall be two years, beginning on the Effective Date and shall apply to the states of Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina, and Tennessee. If as of the expiration of this Agreement, a Subsequent Agreement (as defined in Section 2.2 below) has not been executed by the Parties, this Agreement shall continue on a month-to-month basis while a Subsequent Agreement is being negotiated. The

Parties' rights and obligations with respect to this Agreement after expiration shall be as set forth in Section 2.4 below.

- 2.2 The Parties agree that by no later than one hundred and eighty (180) days prior to the expiration of this Agreement, they shall commence negotiations with regard to the terms, conditions and prices of resale and/or local interconnection to be effective beginning on the expiration date of this Agreement ("Subsequent Agreement").
- If, within one hundred and thirty-five (135) days of commencing the negotiation referred to in Section 2.2 above, the Parties are unable to satisfactorily negotiate new resale and/or local interconnection terms, conditions and prices, either Party may petition the Commission to establish appropriate local interconnection and/or resale arrangements pursuant to 47 U.S.C. 252. The Parties agree that, in such event, they shall encourage the Commission to issue its order regarding the appropriate local interconnection and/or resale arrangements no later than the expiration date of this Agreement. The Parties further agree that in the event the Commission does not issue its order prior to the expiration date of this Agreement, or if the Parties continue beyond the expiration date of this Agreement to negotiate the local interconnection and/or resale arrangements without Commission intervention, the terms, conditions and prices ultimately ordered by the Commission, or negotiated by the Parties, will be effective retroactive to the day following the expiration date of this Agreement.
- 2.4 Notwithstanding the foregoing, in the event that as of the date of expiration of this Agreement and conversion of this Agreement to a month-to-month term, the Parties have not entered into a Subsequent Agreement and either no arbitration proceeding has been filed in accordance with Section 2.3 above, or the Parties have not mutually agreed (where permissible) to extend the arbitration window for petitioning the applicable Commission(s) for resolution of those terms upon which the Parties have not agreed, then either Party may terminate this Agreement upon sixty (60) days notice to the other Party. In the event that BellSouth terminates this Agreement as provided above, BellSouth shall continue to offer services to NewSouth pursuant to the terms, conditions and rates set forth in BellSouth's Statement of Generally Available Terms (SGAT) to the extent an SGAT has been approved by the applicable Commission(s). If any state Commission has not approved a BellSouth SGAT, then upon BellSouth's termination of this Agreement as provided herein, BellSouth will continue to provide services to NewSouth pursuant to BellSouth's then current standard interconnection agreement. In the event that the SGAT or BellSouth's standard interconnection agreement becomes effective as between the Parties, the Parties may continue to negotiate a Subsequent Agreement, and the terms of such Subsequent Agreement shall be effective retroactive to the day following expiration of this Agreement.

3. Ordering Procedures

- 3.1 NewSouth shall provide BellSouth its Carrier Identification Code (CIC),
 Operating Company Number (OCN), Group Access Code (GAC) and Access
 Customer Name and Address (ACNA) code as applicable prior to placing its first order.
- 3.2 The Parties agree to adhere to the BellSouth Local Interconnection and Facility Based Ordering Guide and Resale Ordering Guide, as appropriate for the services ordered.
- 3.3 NewSouth shall pay charges for Operational Support Systems (OSS) as set forth in this Agreement in Attachment 1 and/or in Attachment 2, 3, 5 and 7 as applicable.

4. Parity

When NewSouth purchases, pursuant to Attachment 1 of this Agreement, telecommunications services from BellSouth for the purposes of resale to end users, BellSouth shall provide said services so that the services are equal in quality, subject to the same conditions, and provided within the same provisioning time intervals that BellSouth provides to its affiliates, subsidiaries and end users. To the extent technically feasible, the quality of a Network Element, as well as the quality of the access to such Network Element provided by BellSouth to NewSouth shall be at least equal in quality to that which BellSouth provides to itself. The quality of the interconnection between the networks of BellSouth and the network of NewSouth shall be at a level that is equal to that which BellSouth provides itself, a subsidiary, an Affiliate, or any other party. The interconnection facilities shall be designed to meet the same technical criteria and service standards that are used within BellSouth's network and shall extend to a consideration of service quality as perceived by end users and service quality as perceived by NewSouth.

5. White Pages Listings

BellSouth shall provide NewSouth and their customers access to white pages directory listings under the following terms:

- Listings. NewSouth shall provide all new, changed and deleted listings on a timely basis and BellSouth or its agent will include NewSouth residential and business customer listings in the appropriate White Pages (residential and business) or alphabetical directories. Directory listings will make no distinction between NewSouth and BellSouth subscribers.
- 5.2 Rates. BellSouth and NewSouth will provide to each other subscriber primary listing information in the White Pages for a non-recurring charge.
- 5.3 Procedures for Submitting NewSouth Subscriber Information are found in BellSouth's Ordering Guide for manually processed listings and in the Local Exchange Ordering Guide for mechanically submitted listings.

- Notwithstanding any provision(s) to the contrary, NewSouth agrees to provide to 5.3.1 BellSouth, and BellSouth agrees to accept, NewSouth's Subscriber Listing Information (SLI) relating to NewSouth's customers in the geographic area(s) covered by this Interconnection Agreement. NewSouth authorizes BellSouth to release all such NewSouth SLI provided to BellSouth by NewSouth to qualifying third parties via either license agreement or BellSouth's Directory Publishers Database Service (DPDS), General Subscriber Services Tariff, Section A38.2, as the same may be amended from time to time. Such CLEC SLI shall be intermingled with BellSouth's own customer listings of any other CLEC that has authorized a similar release of SLI. Where necessary, BellSouth will use good faith efforts to obtain state commission approval of any necessary modifications to Section A38.2 of its tariff to provide for release of third party directory listings, including modifications regarding listings to be released pursuant to such tariff and BellSouth's liability therunder. BellSouth's obligation pursuant to this Section shall not arise in any particular state until the commission of such state has approved modifications to such tariff.
- No compensation shall be paid to NewSouth for BellSouth's receipt of NewSouth SLI, or for the subsequent release to third parties of such SLI. In addition, to the extent BellSouth incurs costs to modify its systems to enable the release of NewSouth's SLI, or costs on an ongoing basis to administer the release of NewSouth SLI, NewSouth shall pay to BellSouth its proportionate share of the reasonable costs associated therewith.
- 5.3.3 BellSouth shall not be liable for the content or accuracy of any SLI provided by NewSouth under this Agreement. NewSouth shall indemnify, hold harmless and defend BellSouth from and against any damages, losses, liabilities, demands claims, suits, judgments, costs and expenses (including but not limited to reasonable attorneys' fees and expenses) arising from BellSouth's tariff obligations or otherwise and resulting from or arising out of any third party's claim of inaccurate NewSouth listings or use of the SLI provided pursuant to this Agreement. BellSouth shall forward to NewSouth any complaints received by BellSouth relating to the accuracy or quality of NewSouth listings.
- 5.3.4 Listings and subsequent updates will be released consistent with BellSouth system changes and/or update scheduling requirements.
- 5.4 <u>Unlisted/Non-Published Subscribers</u>. NewSouth will be required to provide to BellSouth the names, addresses and telephone numbers of all NewSouth customers that wish to be omitted from directories.
- Inclusion of NewSouth Customers in Directory Assistance Database. BellSouth will include and maintain NewSouth subscriber listings in BellSouth's Directory Assistance databases at no recurring charge and NewSouth shall provide such Directory Assistance listings at no recurring charge. BellSouth and NewSouth will formulate appropriate procedures regarding lead-time, timeliness, format and content of listing information.

- Listing Information Confidentiality. BellSouth will accord NewSouth's directory listing information the same level of confidentiality that BellSouth accords its own directory listing information, and BellSouth shall limit access to NewSouth's customer proprietary confidential directory information to those BellSouth employees who are involved in the preparation of listings.
- 5.7 Optional Listings. Additional listings and optional listings will be offered by BellSouth at tariffed rates as set forth in the General Subscriber Services Tariff.
- 5.8 <u>Delivery.</u> BellSouth or its agent shall deliver White Pages directories to NewSouth subscribers at no charge or as specified in a separate BAPCO agreement.

6. Bona Fide Request/New Business Request Process for Further Unbundling

If NewSouth is a facilities based provider or a facilities based and resale provider, this section shall apply. BellSouth shall, upon request of NewSouth, provide to NewSouth access to its network elements at any technically feasible point for the provision of NewSouth's telecommunications service where such access is necessary and failure to provide access would impair the ability of NewSouth to provide services that it seeks to offer. Any request by NewSouth for access to a network element, interconnection option, or for the provisioning of any service or product that is not already available shall be treated as a Bona Fide Request/New Business Request, and shall be submitted to BellSouth pursuant to the Bona Fide Request/New Business Request process set forth in Attachment 12 of this Agreement

7. Court Ordered Requests for Call Detail Records and Other Subscriber Information

- 7.1 To the extent technically feasible, BellSouth maintains call detail records for NewSouth end users for limited time periods and can respond to subpoenas and court ordered requests for this information. BellSouth shall maintain such information for NewSouth end users for the same length of time it maintains such information for its own end users.
- 7.2 NewSouth agrees that BellSouth will respond to subpoenas and court ordered requests delivered directly to BellSouth for the purpose of providing call detail records when the targeted telephone numbers belong to NewSouth end users. Billing for such requests will be generated by BellSouth and directed to the law enforcement agency initiating the request.
- 7.3 Where BellSouth is providing to NewSouth telecommunications services for resale or providing to NewSouth the local switching function, then NewSouth agrees that in those cases where NewSouth receives subpoenas or court ordered requests regarding targeted telephone numbers belonging to NewSouth end users, if

NewSouth does not have the requested information, NewSouth will advise the law enforcement agency initiating the request to redirect the subpoena or court ordered request to BellSouth. Where the request has been forwarded to BellSouth, billing for call detail information will be generated by BellSouth and directed to the law enforcement agency initiating the request.

7.4 In all other instances, NewSouth will provide NewSouth end user and/or other customer information that is available to NewSouth in response to subpoenas and court orders for their own customer records. When BellSouth receives subpoenas or court ordered requests regarding targeted telephone numbers belonging to NewSouth end users, BellSouth will advise the law enforcement agency initiating the request to redirect the subpoena or court ordered request to NewSouth.

8. Liability and Indemnification

- 8.1 <u>BellSouth Liability</u>. BellSouth shall take financial responsibility for its own actions in causing or its lack of action in preventing, unbillable or uncollectible NewSouth revenues.
- 8.2 NewSouth Liability. In the event that NewSouth consists of two (2) or more separate entities as set forth in the preamble to this Agreement, all such entities shall be jointly and severally liable for the obligations of NewSouth under this Agreement.
- 8.3 <u>Liability for Acts or Omissions of Third Parties</u>. Neither BellSouth nor NewSouth shall be liable for any act or omission of another telecommunications company providing a portion of the services provided under this Agreement.
- 8.4 <u>Limitation of Liability</u>.
- 8.4.1 Each Party's liability to the other for any loss, cost, claim, injury or liability or expense, including reasonable attorney's fees relating to or arising out of any negligent act or omission in its performance of this Agreement whether in contract or in tort, shall be limited to a credit for the actual cost of the services or functions not performed or improperly performed.
- Limitations in Tariffs. A Party may, in its sole discretion, provide in its tariffs and contracts with its Customer and third parties that relate to any service, product or function provided or contemplated under this Agreement, that to the maximum extent permitted by Applicable Law, such Party shall not be liable to Customer or third Party for (i) any Loss relating to or arising out of this Agreement, whether in contract, tort or otherwise, that exceeds the amount such Party would have charged that applicable person for the service, product or function that gave rise to such Loss and (ii) Consequential Damages. To the extent that a Party elects not to place in its tariffs or contracts such limitations of liability, and the other Party incurs a Loss as a result thereof, such Party shall indemnify and reimburse the other Party for that portion of the Loss that would have been limited had the first

Party included in its tariffs and contracts the limitations of liability that such other Party included in its own tariffs at the time of such Loss.

- 8.4.3 Neither BellSouth nor NewSouth shall be liable for damages to the other's terminal location, POI or other company's customers' premises resulting from the furnishing of a service, including, but not limited to, the installation and removal of equipment or associated wiring, except to the extent caused by a company's negligence or willful misconduct or by a company's failure to properly ground a local loop after disconnection.
- 8.4.4 Except in cases of gross negligence, willful or intentional misconduct, under no circumstance shall a Party be responsible or liable for indirect, incidental, or consequential damages, including, but not limited to, economic loss or lost business or profits, damages arising from the use or performance of equipment or software, or the loss of use of software or equipment, or accessories attached thereto, delay, error, or loss of data. In connection with this limitation of liability, each Party recognizes that the other Party may, from time to time, provide advice, make recommendations, or supply other analyses related to the Services, or facilities described in this Agreement, and, while each Party shall use diligent efforts in this regard, the Parties acknowledge and agree that this limitation of liability shall apply to provision of such advice, recommendations, and analyses.
- Indemnification for Certain Claims. The Party providing services hereunder, its affiliates and its parent company, shall be indemnified, defended and held harmless by the Party receiving services hereunder against any claim, loss or damage arising from the receiving company's use of the services provided under this Agreement pertaining to (1) claims for libel, slander or invasion of privacy arising from the content of the receiving company's own communications, or (2) any claim, loss or damage claimed by the customer of the Party receiving services arising from such company's use or reliance on the providing company's services, actions, duties, or obligations arising out of this Agreement.
- Disclaimer. EXCEPT AS SPECIFICALLY PROVIDED TO THE CONTRARY IN THIS AGREEMENT, NEITHER PARTY MAKES ANY REPRESENTATIONS OR WARRANTIES TO THE OTHER PARTY CONCERNING THE SPECIFIC QUALITY OF ANY SERVICES, OR FACILITIES PROVIDED UNDER THIS AGREEMENT. THE PARTIES DISCLAIM, WITHOUT LIMITATION, ANY WARRANTY OR GUARANTEE OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE, ARISING FROM COURSE OF PERFORMANCE, COURSE OF DEALING, OR FROM USAGES OF TRADE.

9. Intellectual Property Rights and Indemnification

9.1 No License. No patent, copyright, trademark or other proprietary right is licensed, granted or otherwise transferred by this Agreement. NewSouth is strictly prohibited from any use, including but not limited to in sales, in marketing or

advertising of telecommunications services, of any BellSouth name, service mark or trademark.

- 9.2 Ownership of Intellectual Property. Any intellectual property which originates from or is developed by a Party shall remain in the exclusive ownership of that Party. Except for a limited license to use patents or copyrights to the extent necessary for the Parties to use any facilities or equipment (including software) or to receive any service solely as provided under this Agreement, no license in patent, copyright, trademark or trade secret, or other proprietary or intellectual property right now or hereafter owned, controlled or licensable by a Party, is granted to the other Party or shall be implied or arise by estoppel. It is the responsibility of each Party to ensure at no additional cost to the other Party that it has obtained any necessary licenses in relation to intellectual property of third Parties used in its network that may be required to enable the other Party to use any facilities or equipment (including software), to receive any service, or to perform its respective obligations under this Agreement.
- Indemnification. The Party providing a service pursuant to this Agreement will defend the Party receiving such service or data provided as a result of such service against claims of infringement arising solely from the use by the receiving Party of such service and will indemnify the receiving Party for any damages awarded based solely on such claims in accordance with Section 8 of this Agreement.
- Claim of Infringement. In the event that use of any facilities or equipment (including software), becomes, or in reasonable judgment of the Party who owns the affected network is likely to become, the subject of a claim, action, suit, or proceeding based on intellectual property infringement, then said Party shall promptly and at its sole expense, but subject to the limitations of liability set forth below:
- 9.4.1 modify or replace the applicable facilities or equipment (including software) while maintaining form and function, or
- 9.4.2 obtain a license sufficient to allow such use to continue.
- In the event 9.4.1 or 9.4.2 are commercially unreasonable, then said Party may, terminate, upon reasonable notice, this contract with respect to use of, or services provided through use of, the affected facilities or equipment (including software), but solely to the extent required to avoid the infringement claim.
- 9.5 Exception to Obligations. Neither Party's obligations under this Section shall apply to the extent the infringement is caused by: (i) modification of the facilities or equipment (including software) by the indemnitee; (ii) use by the indemnitee of the facilities or equipment (including software) in combination with equipment or facilities (including software) not provided or authorized by the indemnitor provided the facilities or equipment (including software) would not be infringing if used alone; (iii) conformance to specifications of the indemnitee which would necessarily result in infringement; or (iv) continued use by the indemnitee of the

affected facilities or equipment (including software) after being placed on notice to discontinue use as set forth herein.

9.6 Exclusive Remedy. The foregoing shall constitute the Parties' sole and exclusive remedies and obligations with respect to a third party claim of intellectual property infringement arising out of the conduct of business under this Agreement.

10. Proprietary and Confidential Information

- Proprietary and Confidential Information: It may be necessary for BellSouth and 10.1 NewSouth, each as the "Discloser," to provide to the other party, as "Recipient." certain proprietary and confidential information(including trade secret information) including but not limited to technical, financial, marketing, staffing and business plans and information, strategic information, proposals, request for proposals, specifications, drawings, prices, costs, procedures, processes, business systems, software programs, techniques, customer account data, call detail records and like information (collectively the Discloser's "Information"). All Information shall be provided to Recipient in written or other tangible or electronic form, clearly marked with a confidential and, proprietary notice. Information orally or visually provided to Recipient must be designated by Discloser as confidential and proprietary at the time of such disclosure and must be reduced to writing marked with a confidential and proprietary notice and provided to Recipient within thirty (30) calendar days after such oral or visual disclosure.
- 10.2 Use and Protection of Information. Recipient shall use the Information solely for the purpose(s) of performing this Agreement, and Recipient shall protect Information from any use, distribution or disclosure except as permitted hereunder. Recipient will use the same standard of care to protect Information as Recipient uses to protect its own similar confidential and proprietary information, but not less than a reasonable standard of care. Recipient may disclose Information solely to the Authorized Representatives of the Recipient who (a) have a substantive need to know such Information in connection with performance of the Agreement; (b) have been advised of the confidential and proprietary nature of the Information; and (c) have personally agreed in writing to protect from unauthorized disclosure all confidential and proprietary information, of whatever source, to which they have access in the course of their employment. "Authorized Representatives" are the officers, directors and employees of Recipient and its Affiliates, as well as Recipient's and its Affiliates' consultants, contractors, counsel and agents. " Affiliates" means any company that is owned in whole or in part, now or in the future, directly or indirectly through a subsidiary, by a party hereto.
- Ownership, Copying & Return of Information. Information remains at all times the property of Discloser. Recipient may make tangible or electronic copies, notes, summaries or extracts of Information only as necessary for use as authorized herein. All such tangible or electronic copies, notes, summaries or

extracts must be marked with the same confidential and proprietary notice as appears on the original. Upon Discloser's request, all or any requested portion of the Information (including, but not limited to, tangible and electronic copies, notes, summaries or extracts of any information) will be promptly returned to Discloser or destroyed, and Recipient will provide Discloser with written certification stating that such Information has been returned or destroyed.

10.4

Exceptions. Discloser's Information does not include: (a) any information publicly disclosed by Discloser; (b) any information Discloser in writing authorizes Recipient to disclose without restriction; (c) any information already lawfully known to Recipient at the time it is disclosed by the Discloser, without an obligation to keep confidential; or (d) any information Recipient lawfully obtains from any source other than Discloser, provided that such source lawfully disclosed and/or independently developed such information. If Recipient is required to provide Information to any court or government agency pursuant to written court order, subpoena, regulation or process of law, Recipient must first provided Discloser with prompt written notice of such requirement and cooperate with Discloser to appropriately protect against or limit the scope of such disclosure. To the fullest extent permitted by law, Recipient will continue to protect as confidential and proprietary all Information disclosed in response to a written court order, subpoena, regulation or process of law.

10.5

Equitable Relief. Recipient acknowledges and agrees that any breach or threatened breach of this Agreement is likely to cause Discloser irreparable harm for which money damages may not be an appropriate or sufficient remedy. Recipient therefore agrees that Discloser or its Affiliates, as the case may be, are entitled to receive injunctive or other equitable relief to remedy or prevent any breach or threatened breach of this Agreement. Such remedy is not the exclusive remedy for any breach or threatened breach of this Agreement, but is in addition to all other rights and remedies available at law or in equity.

10.6

Survival of Confidentiality Obligations. The parties' rights and obligations under this Section 10 shall survive and continue in effect until two (2) years after the expiration or termination date of this Agreement with regard to all Information exchanged during the term of this Agreement. Thereafter, the parties' rights and obligations hereunder survive and continue in effect with respect to any Information that is a trade secret under applicable law.

11. Assignments

Any assignment by either Party to any non-affiliated entity of any right, obligation or duty, or of any other interest hereunder, in whole or in part, without the prior written consent of the other Party shall be void, and such consent shall not be unreasonably withheld. A Party may assign this Agreement or any right, obligation, duty or other interest hereunder to an Affiliate company of the Party without the consent of the other Party. All obligations and duties of any Party under this Agreement shall be binding on all successors in interest and assigns of such Party. No assignment or delegation hereof shall relieve the assignor of its obligations under this Agreement in the event that the assignee fails to perform such obligations.

12. Resolution of Disputes

Except as otherwise stated in this Agreement, the Parties agree that if any dispute arises as to the interpretation of any provision of this Agreement or as to the proper implementation of this Agreement, either Party may petition the Commission for a resolution of the dispute. However, each Party reserves any rights it may have to seek judicial review of any ruling made by the Commission concerning this Agreement.

13. Taxes

- Definition. For purposes of this Section, the terms "taxes" and "fees" shall include but not limited to federal, state or local sales, use, excise, gross receipts or other taxes or tax-like fees of whatever nature and however designated (including tariff surcharges and any fees, charges or other payments, contractual or otherwise, for the use of public streets or rights of way, whether designated as franchise fees or otherwise) imposed, or sought to be imposed, on or with respect to the services furnished hereunder or measured by the charges or payments therefore, excluding any taxes levied on income.
- 13.2 Taxes and Fees Imposed Directly On Either Providing Party or Purchasing Party.
- Taxes and fees imposed on the providing Party, which are not permitted or required to be passed on by the providing Party to its customer, shall be borne and paid by the providing Party.
- Taxes and fees imposed on the purchasing Party, which are not required to be collected and/or remitted by the providing Party, shall be borne and paid by the purchasing Party.
- 13.3 <u>Taxes and Fees Imposed on Purchasing Party But Collected And Remitted By Providing Party.</u>

- Taxes and fees imposed on the purchasing Party shall be borne by the purchasing Party, even if the obligation to collect and/or remit such taxes or fees is placed on the providing Party.
- To the extent permitted by applicable law, any such taxes and/or fees shall be shown as separate items on applicable billing documents between the Parties. Notwithstanding the foregoing, the purchasing Party shall remain liable for any such taxes and fees regardless of whether they are actually billed by the providing Party at the time that the respective service is billed.
- 13.3.3 If the purchasing Party determines that in its opinion any such taxes or fees are not payable, the providing Party shall not bill such taxes or fees to the purchasing Party if the purchasing Party provides written certification, reasonably satisfactory to the providing Party, stating that it is exempt or otherwise not subject to the tax or fee, setting forth the basis therefor, and satisfying any other requirements under applicable law. If any authority seeks to collect any such tax or fee that the purchasing Party has determined and certified not to be payable, or any such tax or fee that was not billed by the providing Party, the purchasing Party may contest the same in good faith, at its own expense. In any such contest, the purchasing Party shall promptly furnish the providing Party with copies of all filings in any proceeding, protest, or legal challenge, all rulings issued in connection therewith, and all correspondence between the purchasing Party and the taxing authority.
- In the event that all or any portion of an amount sought to be collected must be paid in order to contest the imposition of any such tax or fee, or to avoid the existence of a lien on the assets of the providing Party during the pendency of such contest, the purchasing Party shall be responsible for such payment and shall be entitled to the benefit of any refund or recovery.
- 13.3.5 If it is ultimately determined that any additional amount of such a tax or fee is due to the imposing authority, the purchasing Party shall pay such additional amount, including any interest and penalties thereon.
- Notwithstanding any provision to the contrary, the purchasing Party shall protect, indemnify and hold harmless (and defend at the purchasing Party's expense) the providing Party from and against any such tax or fee, interest or penalties thereon, or other charges or payable expenses (including reasonable attorney fees) with respect thereto, which are incurred by the providing Party in connection with any claim for or contest of any such tax or fee.
- Each Party shall notify the other Party in writing of any assessment, proposed assessment or other claim for any additional amount of such a tax or fee by a taxing authority; such notice to be provided, if possible, at least ten (10) days prior to the date by which a response, protest or other appeal must be filed, but in no event later than thirty (30) days after receipt of such assessment, proposed assessment or claim.
- 13.4 Taxes and Fees Imposed on Providing Party But Passed On To Purchasing Party.

- Taxes and fees imposed on the providing Party, which are permitted or required to be passed on by the providing Party to its customer, shall be borne by the purchasing Party.
- To the extent permitted by applicable law, any such taxes and/or fees shall be shown as separate items on applicable billing documents between the Parties. Notwithstanding the foregoing, the purchasing Party shall remain liable for any such taxes and fees regardless of whether they are actually billed by the providing Party at the time that the respective service is billed.
- 13.4.3 If the purchasing Party disagrees with the providing Party's determination as to the application or basis for any such tax or fee, the Parties shall consult with respect to the imposition and billing of such tax or fee. Notwithstanding the foregoing, the providing Party shall retain ultimate responsibility for determining whether and to what extent any such taxes or fees are applicable, and the purchasing Party shall abide by such determination and pay such taxes or fees to the providing Party. The providing Party shall further retain ultimate responsibility for determining whether and how to contest the imposition of such taxes and fees; provided, however, that any such contest undertaken at the request of the purchasing Party shall be at the purchasing Party's expense.
- In the event that all or any portion of an amount sought to be collected must be paid in order to contest the imposition of any such tax or fee, or to avoid the existence of a lien on the assets of the providing Party during the pendency of such contest, the purchasing Party shall be responsible for such payment and shall be entitled to the benefit of any refund or recovery.
- 13.4.5 If it is ultimately determined that any additional amount of such a tax or fee is due to the imposing authority, the purchasing Party shall pay such additional amount, including any interest and penalties thereon.
- 13.4.6 Notwithstanding any provision to the contrary, the purchasing Party shall protect indemnify and hold harmless (and defend at the purchasing Party's expense) the providing Party from and against any such tax or fee, interest or penalties thereon, or other reasonable charges or payable expenses (including reasonable attorney fees) with respect thereto, which are incurred by the providing Party in connection with any claim for or contest of any such tax or fee.
- Each Party shall notify the other Party in writing of any assessment, proposed assessment or other claim for any additional amount of such a tax or fee by a taxing authority; such notice to be provided, if possible, at least ten (10) days prior to the date by which a response, protest or other appeal must be filed, but in no event later than thirty (30) days after receipt of such assessment, proposed assessment or claim.
- Mutual Cooperation. In any contest of a tax or fee by one Party, the other Party shall cooperate fully by providing records, testimony and such additional

information or assistance as may reasonably be necessary to pursue the contest. Further, the other Party shall be reimbursed for any reasonable and necessary out-of-pocket copying and travel expenses incurred in assisting in such contest.

14. Force Majeure

In the event performance of this Agreement, or any obligation hereunder, is either directly or indirectly prevented, restricted, or interfered with by reason of fire, flood, earthquake or like acts of God, wars, revolution, civil commotion, explosion, acts of public enemy, embargo, acts of the government in its sovereign capacity, labor difficulties, including without limitation, strikes, slowdowns, picketing, or boycotts, unavailability of equipment from vendor, changes requested by Customer, or any other circumstances beyond the reasonable control and without the fault or negligence of the Party affected, the Party affected, upon giving prompt notice to the other Party, shall be excused from such performance on a day-to-day basis to the extent of such prevention, restriction, or interference (and the other Party shall likewise be excused from performance of its obligations on a day-to-day basis until the delay, restriction or interference has ceased); provided however, that the Party so affected shall use diligent efforts to avoid or remove such causes of non-performance and both Parties shall proceed whenever such causes are removed or cease.

15. Network Maintenance and Management

- The Parties shall work cooperatively to implement this Agreement. The Parties shall exchange appropriate information (e.g., maintenance contact numbers, network information, information required to comply with law enforcement and other security agencies of the Government, etc.) as reasonably required to implement and perform this Agreement.
- Each Party hereto shall design, maintain and operate their respective networks as necessary to ensure that the other Party hereto receives service quality which is consistent with generally accepted industry standards at least at parity with the network service quality given to itself, its Affiliates, its End Users or any other Telecommunications Carrier.
- Neither Party shall use any service or facility provided under this Agreement in a manner that impairs the quality of service to other Telecommunications Carriers' or to either Party's End Users. Each Party will provide the other Party notice of any such impairment at the earliest practicable time.
- BellSouth agrees to provide NewSouth prior notice consistent with applicable FCC rules and the Act of changes in the information necessary for the transmission and routing of services using BellSouth's facilities or networks, as well as other changes that affect the interoperability of those respective facilities and networks. This Agreement is not intended to limit BellSouth's ability to

Page 15 upgrade its network through the incorporation of new equipment, new software or otherwise so long as such upgrades are not inconsistent with BellSouth's obligations to NewSouth under the terms of this Agreement.

16. Modification of Agreement

- BellSouth shall make available, pursuant to 47 USC § 252(i), and the FCC rules and regulations and Court Orders regarding such availability, to NewSouth any interconnection, service, or network element provided under any other agreement filed and approved pursuant to 47 USC § 252 (e).
- If NewSouth changes its name or makes changes to its company structure or identity due to a merger, acquisition, transfer or any other reason, it is the responsibility of NewSouth to notify BellSouth of said change and request that an amendment to this Agreement, if necessary, be executed to reflect said change.
- No modification, amendment, supplement to, or waiver of the Agreement or any of its provisions shall be effective and binding upon the Parties unless it is made in writing and duly signed by the Parties.
- 16.4 Execution of this Agreement by either Party does not confirm or infer that the executing Party agrees with any decision(s) issued pursuant to the Telecommunications Act of 1996 and the consequences of those decisions on specific language in this Agreement. Neither Party waives its rights to appeal or otherwise challenge any such decision(s) and each Party reserves all of its rights to pursue any and all legal and/or equitable remedies, including appeals of any such decision(s).
- In the event that any effective legislative, regulatory, judicial or other legal action materially affects any material terms of this Agreement, or the ability of NewSouth or BellSouth to perform any material terms of this Agreement, NewSouth or BellSouth may, on thirty (30) days' written notice require that such terms be renegotiated, and the Parties shall renegotiate in good faith such mutually acceptable new terms as may be required. In the event that such new terms are not renegotiated within ninety (90) days after such notice, the Dispute shall be referred to the Dispute Resolution procedure set forth in Section 12.
- If any provision of this Agreement, or the application of such provision to either Party or circumstance, shall be held invalid, the remainder of the Agreement, or the application of any such provision to the Parties or circumstances other than those to which it is held invalid, shall not be effective thereby, provided that the Parties shall attempt to reformulate such invalid provision to give effect to such portions thereof as may be valid without defeating the intent of such provision.

17. Waivers

A failure or delay of either Party to enforce any of the provisions hereof, to exercise any option which is herein provided, or to require performance of any of the

Page 16

provisions hereof shall in no way be construed to be a waiver of such provisions or options, and each Party, notwithstanding such failure, shall have the right thereafter to insist upon the specific performance of any and all of the provisions of this Agreement.

18. Governing Law

This Agreement shall be governed by, and construed and enforced in accordance with, the laws of the State of Georgia, without regard to its conflict of laws principles.

19. Arm's Length Negotiations

This Agreement was executed after arm's length negotiations between the undersigned Parties and reflects the conclusion of the undersigned that this Agreement is in the best interests of all Parties.

20. Notices

Every notice, consent, approval, or other communications required or contemplated by this Agreement shall be in writing and shall be delivered in person or given by postage prepaid mail, address to:

BellSouth Telecommunications, Inc.

CLEC Account Team 9th Floor 600 North 19th Street Birmingham, Alabama 35203

and

General Attorney - COU Suite 4300 675 W. Peachtree St. Atlanta, GA 30375

NewSouth Communications, Corp.

Senior Vice President of Network Planning & Provisioning NewSouth Center Two N. Main Street Greenville, SC 29601

and

Vice President of Regulatory Affairs NewSouth Center Two N. Main Street Greenville, SC 29601

or at such other address as the intended recipient previously shall have designated by written notice to the other Party.

- Where specifically required, notices shall be by certified or registered mail.

 Unless otherwise provided in this Agreement, notice by mail shall be effective on the date it is officially recorded as delivered by return receipt or equivalent, and in the absence of such record of delivery, it shall be presumed to have been delivered the fifth day, or next business day after the fifth day, after it was deposited in the mails.
- 20.3 BellSouth shall provide NewSouth notice via Internet posting of price changes and of changes to the terms and conditions of services available for resale.

21. Rule of Construction

No rule of construction requiring interpretation against the drafting Party hereof shall apply in the interpretation of this Agreement.

22. Headings of No Force or Effect

The headings of Articles and Sections of this Agreement are for convenience of reference only, and shall in no way define, modify or restrict the meaning or interpretation of the terms or provisions of this Agreement.

23. Multiple Counterparts

This Agreement may be executed multiple counterparts, each of which shall be deemed an original, but all of which shall together constitute but one and the same document.

24. Implementation of Agreement

If NewSouth is a facilities based provider or a facilities based and resale provider, this section shall apply. Within 60 days of the execution of this Agreement, the Parties will adopt a schedule for the implementation of the Agreement. The schedule shall state with specificity time frames for submission of including but not limited to, network design, interconnection points, collocation arrangement requests, pre-sales testing and full operational time frames for the business and residential markets. An implementation template to be used for the implementation schedule is contained in Attachment 10 of this Agreement.

25. Filing of Agreement

- Provided that NewSouth is certified as a CLEC in all applicable states, upon execution of this Agreement it shall be filed with the appropriate state regulatory agency pursuant to the requirements of Section 252 of the Act. If the regulatory agency imposes any filing or public interest notice fees regarding the filing or approval of the Agreement, NewSouth shall be responsible for publishing the required notice and the publication and/or notice costs shall be borne by NewSouth.
- For electronic filing purposes in the State of Louisiana, the CLEC Louisiana Certification Number is required and must be provided by NewSouth prior to execution of the Agreement. The CLEC Louisiana Certification Number for NewSouth is TSP00231.

26. <u>Changes In Subscriber Carrier Selection</u>

- Both Parties hereto shall apply all of the principles set forth in 47 C.F.R. § 64.1100 to the process for End User selection of a primary Local Exchange Carrier. BellSouth shall not require a disconnect order from an NewSouth Customer or another LEC in order to process an NewSouth order for Resale Service for an NewSouth End User. Until the FCC or the Commission adopts final rules and procedures regarding a Customer's selection of a primary Local Exchange Carrier, unless already done so, NewSouth shall deliver to BellSouth a Blanket Representation of Authorization that applies to all orders submitted by NewSouth under this Agreement that require a primary Local Exchange Carrier change. Both Parties hereto shall retain on file all applicable documentation of authorization, including letters of authorization, relating to their End User's selection as its primary Local Exchange Carrier, which documentation shall be available for inspection by the other Party hereto upon reasonable request during normal business hours.
- If an End User denies authorizing a change in his or her primary Local Exchange Carrier selection to a different local exchange carrier ("Unauthorized Switching"), the Party receiving the End User complaint shall switch or caused to be switched that End User back to his preferred carrier in accordance with Applicable Law.

27. Additional Fair Competition Requirements

In the event that either Party transfers facilities or other assets to an Affiliate which are necessary to comply with its obligations under this Agreement, the obligations hereunder shall survive and transfer to such Affiliate.

BellSouth shall allow local exchange customers of NewSouth to select BellSouth for the provision of intraLATA toll services on a nondiscriminatory basis; provided, however, that prior to establishment of BellSouth as the intraLATA toll carrier for NewSouth local exchange customers, the Parties shall negotiate a billing and collections agreement on commercially reasonable terms whereby NewSouth shall bill the customer on BellSouth's behalf and shall collect from the customer and remit to BellSouth intraLATA toll revenues. NewSouth agrees to bill its customers on BellSouth's behalf for both presubscribed and "dial around" intraLATA toll traffic. The Parties shall exchange customer record data on a timely basis as necessary to bill such customers for intraLATA toll usage.

27.3 BellSouth shall not use information derived from providing services or facilities to NewSouth to create a lead or other information base for a "winback" sales program.

28. Operational Support Systems (OSS) Rates

BellSouth has developed and made available the following mechanized systems by which NewSouth may submit LSRs electronically.

LENS Local Exchange Navigation System
EDI Electronic Data Interchange

TAG Telecommunications Access Gateway

RoboTAG

or such other mechanical systems BellSouth may support for LSRs

LSRs submitted by means of one of these interactive interfaces will incur an OSS electronic ordering charge as specified in the Table below. An individual LSR will be identified for billing purposes by its Purchase Order Number (PON). LSRs submitted by means other than one of these interactive interfaces (mail, fax, courier, etc.) will incur a manual order charge as specified in the table below:

OPERATIONAL SUPPORT SYSTEMS (OSS) RATES	Electronic Per LSR received from the CLEC by one of the OSS interactive interfaces	Manual Per LSR received from the CLEC by means other than one of the OSS interactive interfaces
OSS LSR Charge	\$3.50	\$19.99
USOC	SOMEC	SOMAN

Page 20

Note: In addition to the OSS charges, applicable discounted service order and related discounted charges apply per the tariff.

28.1 Denial/Restoral OSS Charge

In the event NewSouth provides a list of customers to be denied and restored, rather than an LSR, each location on the list will require a separate PON and, therefore will be billed as one LSR per location.

28.2 Cancellation OSS Charge

NewSouth will incur an OSS charge for an accepted LSR that is later canceled by NewSouth.

Note: Supplements or clarifications to a previously billed LSR will not incur another OSS charge.

28.3 Threshold Billing Plan (Resale and Number Portability only)

The Parties agree that NewSouth will incur the mechanized rate for all LSRs, both mechanized and manual, if the percentage of mechanized LSRs to total LSRs meets or exceeds the threshold percentages shown below:

Year	Ratio: Mechanized/Total LSRs
2000	80%
2001	90%

The threshold plan will be discontinued in 2002.

BellSouth will track the total LSR volume for each CLEC for each quarter. At the end of that time period, a Percent Electronic LSR calculation will be made for that quarter based on the LSR data tracked in the LCSC. If this percentage exceeds the threshold volume, all of that CLEC's future manual LSRs will be billed at the mechanized LSR rate. To allow time for obtaining and analyzing the data and updating the billing system, this billing change will take place on the first day of the second month following the end of the quarter (e.g. May 1 for 1Q, Aug 1 for 2Q, etc.). There will be no adjustments to the amount billed for previously billed LSRs.

28.4 Network Elements and Other Services Manual Additives

The Commissions in some states have ordered per-element manual additive non-recurring charges (NRC) for Network Elements and Other Services ordered by means other than one of the interactive interfaces. These ordered Network Elements and Other Services manual additive NRCs will apply in these states, rather than the charge per LSR. The per-element charges are listed on the Rate Tables in Attachment 2 of this agreement.

29. Entire Agreement

This Agreement and its Attachments, incorporated herein by this reference, sets forth the entire understanding and supersedes prior Agreements between the Parties relating to the subject matter contained herein and merges all prior discussions between them, and neither Party shall be bound by any definition, condition, provision, representation, warranty, covenant or promise other than as expressly stated in this Agreement or as is contemporaneously or subsequently set forth in writing and executed by a duly authorized officer or representative of the Party to be bound thereby.

This Agreement may include attachments with provisions for the following services:

Network Elements and Other Services
Local Interconnection
Resale
Collocation

The following services are included as options for purchase by NewSouth. NewSouth shall elect said services by written request to its Account Manager if applicable.

Optional Daily Usage File (ODUF)
Enhanced Optional Daily Usage File (EODUF)
Access Daily Usage File (ADUF)
Line Information Database (LIDB) Storage
Centralized Message Distribution Service (CMDS)
Calling Name (CNAM)

IN WITNESS WHEREOF, the Parties have executed this Agreement the day and year above first written.

BellSouth Telecommunications, Inc.	NewSouth Communications, Corp.
The Fallenslu	Sele E. 1 -7
Signature	/ Signature /
Greg Follensbee	Jake E. Jennings
Name	Mame
Senior Director	Vice President of Regulatory Affairs
- Title	'Citle
5/18/01	5/18/01
Date	/ Date

IN WITNESS WHEREOF, the Parties have executed this Agreement the day and year above first written.

BellSouth Telecommunications, Inc.	NewSouth Communications, Corp.
Sk Sollander	Sole E. 1:
Signature	Signature /
Greg Follensbee	Jake E. Jennings
Name	Name
Senior Director	Vice President of Regulatory Affairs
Title	Title
5/18/01	5/18-101
Date	Date

Definitions

Affiliate is defined as a person that (directly or indirectly) owns or controls, is owned or controlled by, or is under common ownership or control with, another person. For purposes of this paragraph, the term "own" means to own an equity interest (or equivalent thereof) of more than 10 percent.

Centralized Message Distribution System is the Telcordia (formerly BellCore) administered national system, based in Kansas City, Missouri, used to exchange Exchange Message Interface (EMI) formatted data among host companies.

Commission is defined as the appropriate regulatory agency in each of BellSouth's nine state region, Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina, and Tennessee.

Daily Usage File is the compilation of messages or copies of messages in standard Exchange Message Interface (EMI) format exchanged from BellSouth to a CLEC.

Exchange Message Interface is the nationally administered standard format for the exchange of data among the Exchange Carriers within the telecommunications industry.

Information Service means the offering of a capability for generating, acquiring, storing, transforming, processing, retrieving, utilizing, or making available information via telecommunications, and includes electronic publishing, but does not include any use of any such capability for the management, control, or operation of a telecommunications system or the management of a telecommunications service.

Intercompany Settlements (ICS) is the revenue associated with charges billed by a company other than the company in whose service area such charges were incurred. ICS on a national level includes third number and credit card calls and is administered by Telcordia (formerly BellCore)'s Calling Card and Third Number Settlement System (CATS). Included is traffic that originates in one Regional Bell Operating Company's (RBOC) territory and bills in another RBOC's territory.

Intermediary function is defined as the delivery of traffic from NewSouth; a CLEC other than NewSouth or another telecommunications carrier through the network of BellSouth or NewSouth to an end user of NewSouth; a CLEC other than NewSouth or another telecommunications carrier.

Local Interconnection is defined as 1) the delivery of local traffic to be terminated on each Party's local network so that end users of either Party have the ability to reach end users of the other Party without the use of any access code or substantial delay in the processing of the call; 2) the LEC network features, functions, and capabilities set forth in this Agreement; and 3) Service Provider Number Portability sometimes referred to as temporary telephone number portability to be implemented pursuant to the terms of this Agreement.

Local Traffic is defined in Attachment 3.

Message Distribution is routing determination and subsequent delivery of message data from one company to another. Also included is the interface function with CMDS, where appropriate.

Multiple Exchange Carrier Access Billing ("MECAB") means the document prepared by the Billing Committee of the Ordering and Billing Forum ("OBF:), which functions under the auspices of the Carrier Liaison Committee of the Alliance for Telecommunications Industry Solutions ("ATIS") and by Telcordia (formerly BellCore) as Special Report SR-BDS-000983, Containing the recommended guidelines for the billing of Exchange Service access provided by two or more LECs and/or CLECs or by one LEC in two or more states within a single LATA.

Network Element is defined to mean a facility or equipment used in the provision of a telecommunications service. Such term may include, but is not limited to, features, functions, and capabilities that are provided by means of such facility or equipment, including but not limited to, subscriber numbers, databases, signaling systems, and information sufficient for billing and collection or used in the transmission, routing, or other provision of a telecommunications service. BellSouth offers access to the Network Elements, unbundled loops; network interface device; sub-loop elements; local switching; transport; tandem switching; operator systems; signaling; access to call-related databases; dark fiber as set forth in Attachment 2 of this Agreement.

Non-Intercompany Settlement System (NICS) is the Telcordia (formerly BellCore) system that calculates non-intercompany settlements amounts due from one company to another within the same RBOC region. It includes credit card, third number and collect messages.

Percent of Interstate Usage (PIU) is defined as a factor to be applied to terminating access services minutes of use to obtain those minutes that should be rated as interstate access services minutes of use. The numerator includes all interstate "non-intermediary" minutes of use, including interstate minutes of use that are forwarded due to service provider number portability less any interstate minutes of use for Terminating Party Pays services, such as 800 Services. The denominator includes all "non-intermediary", local, interstate, intrastate, toll and access minutes of use adjusted for service provider number portability less all minutes attributable to terminating Party pays services.

Percent Local Usage (PLU) is defined as a factor to be applied to intrastate terminating minutes of use. The numerator shall include all "non-intermediary" local minutes of use adjusted for those minutes of use that only apply local due to Service Provider Number Portability. The denominator is the total intrastate minutes of use including local, intrastate toll, and access, adjusted for Service Provider Number Portability less intrastate terminating Party pays minutes of use.

Revenue Accounting Office (RAO) Status Company is a local exchange company/alternate local exchange company that has been assigned a unique RAO code. Message data exchanged among RAO status companies is grouped (i.e. packed) according to From/To/Bill RAO combinations.

Service Control Points ("SCPs") are defined as databases that store information and have the ability to manipulate data required to offer particular services.

Signal Transfer Points ("STPs") are signaling message switches that interconnect Signaling Links to route signaling messages between switches and databases. STPs enable the exchange of Signaling System 7 ("SS7") messages between switching elements, database elements and STPs. STPs provide access to various BellSouth and third party network elements such as local switching and databases.

Signaling links are dedicated transmission paths carrying signaling messages between carrier switches and signaling networks. Signal Link Transport is a set of two or four dedicated 56 kbps transmission paths between NewSouth designated Signaling Points of Interconnection that provide a diverse transmission path and cross connect to a BellSouth Signal Transfer Point.

Telecommunications means the transmission, between or among points specified by the user, of information of the user's choosing, without change in the form or content of the information as sent and received.

Telecommunications Service means the offering of telecommunications for a fee directly to the public, or to such classes of users as to be effectively available directly to the public, regardless of the facilities used.

Telecommunications Act of 1996 ("Act") means Public Law 104-104 of the United States Congress effective February 8, 1996. The Act amended the Communications Act of 1934 (47, U.S.C. Section 1 et. seq.).

Attachment 1

Resale

TABLE OF CONTENTS

1.	Discount Rates	3
2.	Definition of Terms	3
3.	General Provisions	4
4.	Bellsouth's Provision of Services To NewSouth	8
5.	Maintenance of Services	9
6.	Establishment of Service	10
7.	Payment and Billing Arrangements	12
8.	Discontinuance of Service	14
9.	Line Information Database (LIDB)	15
10.	RAO Hosting	15
11.	Optional Daily Usage File (ODUF)	16
12.	Enhanced Optional Daily Usage File (EODUF)	16
Exhi	nibit A – Applicable Discounts/OSS Rates	17
Exhi	nibit B – Resale Restrictions	20
Exhi	nibit C – Line Information Database (LIDB) Storage Agreement	22
Exhi	nibit D – CMDS/ROA Hosting	28
Exhi	nibit E – Optional Daily Usage File ODUF)	
Exhi	nibit F – Enhanced Option Daily Usage File (EODUF)	37
Exhi	tibit G – ODUF/EODUF/CMDS Rates	Rate Tahle

RESALE

1. Discount Rates

The discount rates applied to NewSouth purchases of BellSouth Telecommunications Services for the purpose of resale shall be as set forth in Exhibit A. Such discount shall reflect the costs avoided by BellSouth when selling a service for wholesale purposes.

2. Definition of Terms

- 2.1 COMPETITIVE LOCAL EXCHANGE COMPANY (CLEC) means a telephone company certificated by the public service commissions of BellSouth's franchised area to provide local exchange service within BellSouth's franchised area.
- 2.2 CUSTOMER OF RECORD means the entity responsible for placing application for service; requesting additions, rearrangements, maintenance or discontinuance of service; payment in full of charges incurred such as non-recurring, monthly recurring, toll, directory assistance, etc.
- 2.3 DEPOSIT means assurance provided by a customer in the form of cash, surety bond or bank letter of credit to be held by BellSouth.
- 2.4 END USER means the ultimate user of the telecommunications services.
- 2.5 END USER CUSTOMER LOCATION means the physical location of the premises where an end user makes use of the telecommunications services.
- 2.6 NEW SERVICES means functions, features or capabilities that are not currently offered by BellSouth. This includes packaging of existing services or combining a new function, feature or capability with an existing service.
- 2.7 RESALE means an activity wherein a certificated CLEC, such as NewSouth subscribes to the telecommunications services of BellSouth and then offers those telecommunications services to the public
- 2.8 RESALE SERVICE AREA means the area, as defined in a public service commission approved certificate of operation, within which a CLEC, such as NewSouth, may offer resold local exchange telecommunications service.

3. General Provisions

- NewSouth may resell the tariffed local exchange and toll telecommunications services of BellSouth contained in the General Subscriber Service Tariff and Private Line Service Tariff subject to the terms, and conditions specifically set forth herein. Notwithstanding the foregoing, the exclusions and limitations on services available for resale will be as set forth in Exhibit B, attached hereto and incorporated herein by this reference.
- All of the negotiated rates, terms and conditions set forth in this Attachment pertain to the resale of BellSouth's retail telecommunications services and other services specified in this Attachment. BellSouth shall make available telecommunications services for resale at the discount rates set forth in Exhibit A to this Agreement and subject to the exclusions and limitations set forth in Exhibit B to this Agreement. BellSouth does not however waive its rights to appeal or otherwise challenge any decision regarding resale that resulted in the discount rates contained in Exhibit A or the exclusions and limitations contained in Exhibit B. BellSouth reserves the right to pursue any and all legal and/or equitable remedies, including appeals of any decisions. If such appeals or challenges result in changes in the discount rates or exclusions and limitations, the parties agree that appropriate modifications to this Agreement will be made promptly to make its terms consistent with the outcome of the appeal.
- 3.3 NewSouth may purchase resale services from BellSouth for their own use in operating their business. The resale discount will apply to those services under the following conditions:
- 3.3.1 NewSouth must resell services to other end users.
- 3.3.2 NewSouth must order services through resale interfaces, i.e., the Local Carrier Service Center (LCSC) and/or appropriate Resale Account Teams pursuant to Section 3 of the General Terms and Conditions.
- 3.3.3 NewSouth cannot be a competitive local exchange telecommunications company for the single purpose of selling to themselves.
- The provision of services by BellSouth to NewSouth does not constitute a joint undertaking for the furnishing of any service.
- 3.5 NewSouth will be the customer of record for all services purchased from BellSouth. Except as specified herein, BellSouth will take orders from, bill and expect payment from NewSouth for said services.

- 3.6 NewSouth will be BellSouth's single point of contact for all services purchased pursuant to this Agreement. BellSouth shall have no contact with the end user except to the extent provided for herein.
- 3.7 BellSouth will continue to bill the end user for any services that the end user specifies it wishes to receive directly from BellSouth.
- 3.8 BellSouth maintains the right to serve directly any end user within the service area of NewSouth. BellSouth will continue to directly market its own telecommunications products and services and in doing so may establish independent relationships with end users of NewSouth.
- 3.9 Neither Party shall interfere with the right of any person or entity to obtain service directly from the other Party.
- 3.10 Current telephone numbers may normally be retained by the end user and are assigned to the service furnished. However, neither Party nor the end user has a property right to the telephone number or any other call number designation associated with services furnished by BellSouth, and no right to the continuance of service through any particular central office. BellSouth reserves the right to change such numbers, or the central office designation associated with such numbers, or both, whenever BellSouth deems it necessary to do so in the conduct of its business and in accordance with BellSouth practices and procedures on a nondiscriminatory basis.
- NewSouth, BellSouth will provide NewSouth with an on line access to telephone numbers for reservation on a first come first serve basis. Such reservations of telephone numbers, on a pre-ordering basis shall be for a period of forty-five (45) days. NewSouth acknowledges that there may be instances where there is a shortage of telephone numbers in a particular Common Language Location Identifier Code (CLLIC) and in such instances BellSouth may request that NewSouth cancel its reservations of numbers. Any such request shall be made at parity and in a non-discriminatory manner. NewSouth shall comply with such request.
- Further, upon NewSouth's request, and for the purpose of the resale of BellSouth's telecommunications services by NewSouth, BellSouth will reserve up to 100 telephone numbers per CLLIC, for NewSouth's sole use. Such telephone number reservations shall be valid for forty-five (45) days from the reservation date. NewSouth acknowledges that there may be instances where there is a shortage of telephone numbers in a particular CLLIC and in such instances BellSouth shall use its best efforts to reserve for a forty-five (45) day period a sufficient quantity of NewSouth's reasonable need in that particular CLLIC.

- 3.13 Service is furnished subject to the condition that it will not be used for any unlawful purpose.
- 3.14 Service will be discontinued if any law enforcement agency advises that the service being used is in violation of the law.
- 3.15 BellSouth can refuse service when it has grounds to believe that service will be used in violation of the law.
- 3.16 BellSouth accepts no responsibility to any person for any unlawful act committed by NewSouth or its end users as part of providing service to NewSouth for purposes of resale or otherwise.
- 3.17 BellSouth will cooperate fully with law enforcement agencies with subpoenas and court orders for assistance with BellSouth's end users, pursuant to Section 7 of the General Terms and Conditions
- The characteristics and methods of operation of any circuits, facilities or equipment provided by any person or entity other than BellSouth shall not:
- 3.18.1 Interfere with or impair service over any facilities of BellSouth, its affiliates, or its connecting and concurring carriers involved in its service; or
- 3.18.2 Cause damage to BellSouth's plant;
- 3.18.3 Impair the privacy of any communications; or
- 3.18.4 Create hazards to any BellSouth employees or the public.
- Facilities and/or equipment utilized by BellSouth to provide service to NewSouth remain the property of BellSouth.
- 3.20 White page directory listings will be provided in accordance with Section 5 of the General Terms and Conditions.
- BellSouth provides electronic access to customer record information. Access is provided through the Local Exchange Navigation System (LENS), the Telecommunications Access Gateway (TAG), and RoboTAG. Customer Record Information includes but is not limited to, customer specific information in CRIS and RSAG. In addition, NewSouth shall provide to BellSouth access to customer record information including electronic access where available. Otherwise, upon request by BellSouth, NewSouth shall provide paper copies of customer record information within a reasonable period of time. Customer Record Information is equivalent to but not limited to the type of customer specific information contained in CRIS and

RSAG. The Parties agree not to view, copy, or otherwise obtain access to the customer record information of any customer without that customer's permission, and further agrees that NewSouth and BellSouth will obtain access to customer record information only in strict compliance with applicable laws, rules, or regulations of the State in which the service is provided.

- 3.22 All costs incurred by BellSouth to develop and implement operational interfaces shall be recovered from Resellers who utilize the services. Charges for use of Operational Support Systems (OSS) shall be as set forth in Exhibit A of this Attachment.
- 3.23 Where available to BellSouth's end users, BellSouth shall provide the following telecommunications services at a discount to allow for voice mail services:
 - Message Waiting Indicator ("MWI") stutter dialtone and message waiting light feature capabilities
 - Call Forward on Busy ("CF/B")
 - Call Forward Don't Answer ("CF/DA")

Further, BellSouth messaging services set forth in BellSouth's Messaging Service Information Package shall be made available for resale without the wholesale discount.

- 3.23.1 BellSouth shall provide branding for, or shall unbrand, voice mail services to NewSouth per the Bona Fide Request/New Business Request process as set forth in Attachment 12 of this Agreement.
- 3.24 BellSouth's Inside Wire Maintenance Service Plans may be made available for resale at rates, terms and conditions as set forth by BellSouth and without the wholesale discount.
- 3.25 If NewSouth requires a special assembly NewSouth agrees to pay the costs incurred by BellSouth for providing the requested special assembly. The costs will be provided to NewSouth prior to providing the service. Such costs could include both recurring and non-recurring charges and shall exclude any cost attributable to any marketing, billing collection or other costs that will be avoided by BellSouth in providing service to NewSouth.
- 3.26 Recovery of charges associated with implementing Number Portability through monthly charges assessed to end-users has been authorized by the FCC. This end user line charge will be billed to Resellers of BellSouth's telecommunications services and will be as filed in FCC No. 1. This charge is not discounted.

- 3.27 BellSouth shall provide 911/E911 for NewSouth customers in the same manner that it is provided to BellSouth customers. BellSouth shall provide and validate NewSouth customer information to the PSAP. BellSouth shall use its service order process to update and maintain, on the same schedule that it uses for its customers, the NewSouth customer service information in the ALI/DMS (Automatic Location Identification/Location Information) databases used to support 911/E911 services.
- 3.28 Pursuant to 47 CFR Section 51.617, BellSouth will bill NewSouth end users common line charges identical to the end user common line charges BellSouth bills its end users.
- CSAs shall be available for resale at the wholesale discount set forth in Exhibit A of this Attachment; provided, however, that in the event the Commission establishes a specific discount for CSAs such discount shall apply thereafter. NewSouth may resell a CSA to the end user for whom the CSA was constructed or to end users similarly situated to the specific end user for whom the CSA was constructed. Customers shall be deemed to be similarly situated when the quantity of use; time of use; manner of service; and costs of rendering the service are the same. In cases where NewSouth resells an existing CSA, no termination or rollover charges shall apply to the assignment of the CSA to NewSouth provided that NewSouth assumes the obligations set forth within the CSA. Notwithstanding the foregoing, BellSouth may impose a single service order charge (not to exceed the level of tariffed service order charges for comparable services) to recover the cost of changing the billing name on the account.
- 3.30 BellSouth shall notify NewSouth in advance of long term promotions (offered for longer than ninety (90) days) by posting a notice on its website.
- New Resale Services; Changes in Provision of Resale Services. BellSouth shall use best efforts to provide NewSouth forty-five (45) days advance notice via Internet posting of changes to the prices, terms or conditions of services available for Resale. To the extent that revisions occur between the time BellSouth notifies NewSouth of changes under this Agreement and the time the changes are scheduled to be implemented, BellSouth will notify NewSouth of such revisions consistent with its internal notification process; provided that, NewSouth shall not utilize any notice given under this subsection to market resold offerings of that service in advance of BellSouth. In addition, upon request BellSouth shall furnish NewSouth with copies of publicly available service descriptions regarding the Resale Services.

 Notwithstanding the foregoing, NewSouth shall not utilize any such BellSouth service descriptions as part of its own sales or marketing efforts.

4. BellSouth's Provision of Services to NewSouth

- 4.1 NewSouth agrees that its resale of BellSouth services shall be as follows:
- 4.1.1 The resale of telecommunications services shall be limited to users and uses conforming to the class of service restrictions.
- 4.1.2 Hotel and Hospital PBX services are the only telecommunications services available for resale to Hotel/Motel and Hospital end users, respectively. Similarly, Access Line Service for Customer Provided Coin Telephones is the only local service available for resale to Independent Payphone Provider (IPP) customers. Shared Tenant Service customers can only be sold those local exchange access services available in BellSouth's A23 Shared Tenant Service Tariff in the states of Florida, Georgia, North Carolina and South Carolina, and in A27 in the states of Alabama, Kentucky, Louisiana, Mississippi and Tennessee.
- 4.1.3 BellSouth reserves the right to periodically audit services purchased by NewSouth to establish authenticity of use. Such audit shall not occur more than once in a calendar year. NewSouth shall make any and all records and data available to BellSouth or BellSouth's auditors on a reasonable basis. BellSouth shall bear the cost of said audit.
- 4.2 Resold services can only be used in the same manner as specified in BellSouth's Tariffs. Resold services are subject to the same terms and conditions as are specified for such services when furnished to an individual end user of BellSouth in the appropriate section of BellSouth's Tariffs. Specific tariff features (e.g. a usage allowance per month), shall not be aggregated across multiple resold services.
- 4.3 NewSouth may resell services only within the specific resale service area as defined in its certificate.
- 4.4 Telephone numbers transmitted via any resold service feature are intended solely for the use of the end user of the feature. Resale of this information is prohibited.

5. Maintenance of Services

- 5.1 NewSouth will adopt and adhere to the standards contained in the applicable reasonable and non-discriminatory CLEC Work Center Operational Understanding Agreement regarding maintenance and installation of service.
- 5.2 Services resold pursuant to this Attachment and BellSouth's General Subscriber Service Tariff and Private Line Service Tariff and facilities and equipment provided by BellSouth shall be maintained by BellSouth.

- NewSouth or its end users may not rearrange, move, disconnect, remove or attempt to repair any facilities owned by BellSouth, other than by connection or disconnection to any interface means used, except with the written consent of BellSouth.
- NewSouth accepts responsibility to notify BellSouth of situations that arise that may result in a service problem.
- NewSouth will be BellSouth's single point of contact for all repair calls on behalf of NewSouth's end users. The parties agree to provide one another with toll-free contact numbers for such purposes.
- NewSouth will contact the appropriate repair centers in accordance with procedures established by BellSouth.
- 5.7 For all repair requests, NewSouth accepts responsibility for adhering to BellSouth's prescreening guidelines prior to referring the trouble to BellSouth.
- BellSouth will bill NewSouth for handling troubles that are found not to be in BellSouth's network pursuant to its standard time and material charges. The standard time and material charges will be no more than what BellSouth charges to its retail customers for the same services.
- 5.9 BellSouth reserves the right to contact NewSouth's end users, if deemed necessary, for maintenance purposes.

6. Establishment of Service

- After receiving certification as a local exchange company from the appropriate regulatory agency, NewSouth will provide the appropriate BellSouth service center the necessary documentation to enable BellSouth to establish a master account for NewSouth's resold services. Such documentation shall include the Application for Master Account, proof of authority to provide telecommunications services, an Operating Company Number ("OCN") assigned by the National Exchange Carriers Association ("NECA") and a tax exemption certificate, if applicable. When necessary deposit requirements are met, BellSouth will begin taking orders for the resale of service.
- 6.2 Service orders will be in a standard format designated by BellSouth.
- When notification is received from NewSouth that a current end user of BellSouth will subscribe to NewSouth's service, standard service order intervals for the appropriate class of service will apply.

- BellSouth will not require end user confirmation prior to establishing service for NewSouth's end user customer. NewSouth must, however, be able to demonstrate end user authorization upon request.
- NewSouth will be the single point of contact with BellSouth for all subsequent ordering activity resulting in additions or changes to resold services except that BellSouth will accept a request directly from the end user for conversion of the end user's service from NewSouth to BellSouth or will accept a request from another CLEC for conversion of the end user's service from NewSouth to the other LEC. BellSouth will notify in writing NewSouth within five (5) business days that such a request has been processed.
- 6.6 If BellSouth determines that an unauthorized change in local service to NewSouth has occurred, BellSouth will reestablish service with the appropriate local service provider and will assess NewSouth as the CLEC initiating the unauthorized change, the unauthorized change charge described in F.C.C. Tariff No. 1, Section 13 or applicable state tariff. Appropriate nonrecurring charges, as set forth in Section A4 of the General Subscriber Service Tariff, will also be assessed to NewSouth. These charges can be adjusted if NewSouth provides satisfactory proof of authorization.
- 6.7 In order to safeguard its interest, BellSouth reserves the right to secure the account with a suitable form of security deposit, unless satisfactory credit has already been established.
- 6.7.1 Such security deposit shall take the form of an irrevocable Letter of Credit or other forms of security acceptable to BellSouth. Any such security deposit may be held during the continuance of the service as security for the payment of any and all amounts accruing for the service.
- 6.7.2 If a security deposit is required, such security deposit shall be made prior to the inauguration of service.
- 6.7.3 Such security deposit may not exceed two months' estimated billing.
- 6.7.4 The fact that a security deposit has been made in no way relieves NewSouth from complying with BellSouth's regulations as to advance payments and the prompt payment of bills on presentation nor does it constitute a waiver or modification of the regular practices of BellSouth providing for the discontinuance of service for non-payment of any sums due BellSouth.
- 6.7.5 BellSouth reserves the right to increase the security deposit requirements when, in its reasonable and non-discriminatory judgment, circumstances so warrant and/or gross monthly billing has increased beyond the level initially used to determine the security deposit.

- 6.7.6 In the event that NewSouth defaults on its account, service to NewSouth will be terminated and any security deposits held will be applied to its account.
- 6.7.7 Interest on a security deposit shall accrue and be paid in accordance with the terms in the appropriate BellSouth tariff.
- Orders to switch services "as is" shall be treated as a change of service and shall not be treated as a disconnection and subsequent reconnection of service.

7. Payment And Billing Arrangements

- 7.1 Prior to submitting orders to BellSouth for local service, a master account must be established for NewSouth. NewSouth is required to provide the following before a master account is established: proof of PSC/PUC certification, the Application for Master Account, an Operating Company Number ("OCN") assigned by the National Exchange Carriers Association ("NECA") and a tax exemption certificate, if applicable.
- 7.2 BellSouth shall bill NewSouth on a current basis all applicable charges and credits.
- Payment of all charges will be the responsibility of NewSouth. NewSouth shall make payment to BellSouth for all services provided. BellSouth is not responsible for payments not received by NewSouth from NewSouth's end user. BellSouth will not become involved in billing disputes that may arise between NewSouth and its end user. Payments made to BellSouth as payment on account will be credited to an accounts receivable master account and not to an end user's account.
- 7.4 BellSouth will render bills each month on established bill days for each of NewSouth's accounts.
- 7.5 BellSouth will bill NewSouth in advance charges for all services to be provided during the ensuing billing period except charges associated with service usage, which will be billed in arrears. Charges will be calculated on an individual end user account level, including, if applicable, any charge for usage or usage allowances. BellSouth will also bill NewSouth, and NewSouth will be responsible for and remit to BellSouth, all charges applicable to resold services including but not limited to 911 and E911 charges, telecommunications relay charges (TRS), and franchise fees.
- 7.6 The payment will be due by the next bill date (i.e., same date in the following month as the bill date) and is payable in immediately available funds. Payment is considered to have been made when received by BellSouth.
- 7.6.1 If the payment due date falls on a Sunday or on a Holiday which is observed on a Monday, the payment due date shall be the first non-Holiday day following such

Sunday or Holiday. If the payment due date falls on a Saturday or on a Holiday which is observed on Tuesday, Wednesday, Thursday, or Friday, the payment due date shall be the last non-Holiday day preceding such Saturday or Holiday. If payment is not received by the payment due date, a late payment penalty, as set forth in section 7.8 following, shall apply.

- 7.6.2 If NewSouth requests multiple billing media or additional copies of bills, BellSouth will provide these at an appropriate charge to NewSouth.
- 7.6.3 Billing Disputes
- 7.6.3.1 Each Party agrees to notify the other Party upon the discovery of a billing dispute. In the event of a billing dispute, the Parties will endeavor to resolve the dispute within sixty (60) calendar days of the Bill Date on which such disputed charges appear. Resolution of the dispute is expected to occur at the first level of management resulting in a recommendation for settlement of the dispute and closure of a specific billing period. If the issues are not resolved within the allotted time frame, the following resolution procedure will begin:
- 7.6.3.2 If the dispute is not resolved within sixty (60) days of the Bill Date, the dispute will be escalated to the second level of management for each of the respective Parties for resolution. If the dispute is not resolved within ninety (90) days of the Bill Date, the dispute will be escalated to the third level of management for each of the respective Parties for resolution
- 7.6.3.3 If the dispute is not resolved within one hundred and twenty (120) days of the Bill Date, the dispute will be escalated to the fourth level of management for each of the respective Parties for resolution.
- 7.6.3.4 If a Party disputes a charge and does not pay such charge by the payment due date, such charges shall be subject to late payment charges as set forth in the Late Payment Charges provision of this Attachment. If a Party disputes charges and the dispute is resolved in favor of such Party, the other Party shall credit the bill of the disputing Party for the amount of the disputed charges along with any late payment charges assessed no later than the second Bill Date after the resolution of the dispute. Accordingly, if a Party disputes charges and the dispute is resolved in favor of the other Party, the disputing Party shall pay the other Party the amount of the disputed charges and any associated late payment charges assessed no later than the second bill payment due date after the resolution of the dispute. BellSouth shall only assess interest on previously assessed late payment charges in a state where it has authority pursuant to its tariffs.

- 7.7 Upon proof of tax exempt certification from NewSouth, the total amount billed to NewSouth will not include any taxes due from the end user to reflect the tax exempt certification and local tax laws. NewSouth will be solely responsible for the computation, tracking, reporting, and payment of taxes applicable to NewSouth's end user.
- If any portion of the payment is received by BellSouth after the payment due date as set forth preceding, or if any portion of the payment is received by BellSouth in funds that are not immediately available to BellSouth, then a late payment penalty shall be due to BellSouth. The late payment penalty shall be the portion of the payment not received by the payment due date times a late factor and will be applied on a per bill basis. The late factor shall be as set forth in Section A2 of the General Subscriber Services Tariff and Section B2 of the Private Line Service Tariff. NewSouth will be charged a fee for all returned checks as set forth in Section to A2 of the General Subscriber Services Tariff or in applicable state law.
- 7.9 Any switched access charges associated with interexchange carrier access to the resold local exchange lines will be billed by, and due to, BellSouth. No additional charges are to be assessed to NewSouth.
- 7.10 BellSouth will not perform billing and collection services for NewSouth as a result of the execution of this Agreement. All requests for billing services should be referred to the appropriate entity or operational group within BellSouth.
- 7.11 In general, BellSouth will not become involved in disputes between NewSouth and NewSouth's end user customers over resold services. If a dispute does arise that cannot be settled without the involvement of BellSouth, NewSouth shall contact the designated Service Center for resolution. BellSouth will make every effort to assist in the resolution of the dispute and will work with NewSouth to resolve the matter in as timely a manner as possible. NewSouth may be required to submit documentation to substantiate the claim.

8. Discontinuance of Service

- 8.1 The procedures for discontinuing service to an end user are as follows:
- Where possible, BellSouth will deny service to NewSouth's end user on behalf of, and at the request of, NewSouth. Upon restoration of the end user's service, restoral charges will apply and will be the responsibility of NewSouth.
- 8.1.2 At the request of NewSouth, BellSouth will disconnect a NewSouth end user customer.

- 8.1.3 All requests by NewSouth for denial or disconnection of an end user for nonpayment must be in writing.
- 8.1.4 NewSouth will be made solely responsible for notifying the end user of the proposed disconnection of the service.
- 8.1.5 BellSouth will continue to process calls made to the Annoyance Call Center and will advise NewSouth when it is determined that annoyance calls are originated from one of their end user's locations. BellSouth shall be indemnified, defended and held harmless by NewSouth and/or the end user against any claim, loss or damage arising from providing this information to NewSouth. It is the responsibility of NewSouth to take the corrective action necessary with its end users who make annoying calls. Failure to do so will result in BellSouth's disconnecting the end user's service.
- 8.1.6 BellSouth may disconnect and reuse facilities when the facility is in a denied state and BellSouth has received an order to establish new service or transfer of service from an end user or an end user's CLEC at the same address served by the denied facility.
- 8.2 The procedures for discontinuing service to NewSouth are as follows:
- 8.2.1 BellSouth reserves the right to suspend or terminate service in the event of prohibited, unlawful or improper use of the facilities or service, abuse of the facilities, or any other violation or noncompliance by NewSouth of the rules and regulations of BellSouth's Tariffs. BellSouth shall provide notice and an opportunity to cure, not to exceed five business days.
- 8.2.2 BellSouth reserves the right to suspend or terminate service for nonpayment if payment of account is not received by the bill day in the month after the original bill day, BellSouth may provide written notice to NewSouth, that additional applications for service will be refused and that any pending orders for service will not be completed if payment is not received by the fifteenth day following the date of the notice. In addition BellSouth may, at the same time, give thirty days notice to the person designated by NewSouth to receive notices of noncompliance, and discontinue the provision of existing services to NewSouth at any time thereafter.
- 8.2.3 In the case of such discontinuance, all billed charges, as well as applicable termination charges, shall become due.
- 8.2.4 If BellSouth does not discontinue the provision of the services involved on the date specified in the thirty days notice and NewSouth's noncompliance continues, nothing contained herein shall preclude BellSouth's right to discontinue the provision of the services to NewSouth without further notice.

- 8.2.5 If payment is not received or arrangements made for payment by the date given in the written notification, NewSouth's services will be discontinued. Upon discontinuance of service on a NewSouth's account, service to NewSouth's end users will be denied. BellSouth will also reestablish service at the request of the end user or NewSouth upon payment of the appropriate connection fee and subject to BellSouth's normal application procedures. NewSouth is solely responsible for notifying the end user of the proposed disconnection of the service.
- 8.2.6 If within fifteen days after an end user's service has been denied no contact has been made in reference to restoring service, the end user's service will be disconnected.

9. Line Information Database (LIDB)

- 9.1 BellSouth will store in its Line Information Database (LIDB) records relating to service only in the BellSouth region. The LIDB Storage Agreement is included in this Attachment as Exhibit C.
- 9.2 BellSouth will provide LIDB Storage upon written request to NewSouth Account Manager stating requested activation date.

10. RAO Hosting

- 10.1 The RAO Hosting Agreement is included in this Attachment as Exhibit D. Rates for BellSouth's Centralized Message Distribution System (CMDS) are as set forth in Exhibit G of this Attachment.
- BellSouth will provide RAO Hosting upon written request to its Account Manager stating requested activation date.

11. Optional Daily Usage File (ODUF)

- The Optional Daily Usage File (ODUF) Agreement with terms and conditions is included in this Attachment as Exhibit E. Rates for ODUF are as set forth in Exhibit G of this Attachment.
- BellSouth will provide Optional Daily Usage File (ODUF) service upon written request to its Account Manager stating requested activation date.

12. Enhanced Optional Daily Usage File (EODUF)

12.1 The Enhanced Optional Daily Usage File (EODUF) service Agreement with terms and conditions is included in this Attachment as Exhibit F. Rates for EODUF are as set forth in Exhibit H of this Attachment.

12.2 BellSouth will provide Enhanced Optional Daily Usage File (EODUF) service upon written request to its Account Manager stating requested activation date.

EXHIBIT A
Page 1

APPLICABLE DISCOUNTS

The telecommunications services available for purchase by NewSouth for the purposes of resale to NewSouth end users shall be available at the following discount off of the retail rate. If NewSouth cancels an order for telecommunications services for the purpose of resale, any costs incurred by BellSouth in conjunction with the provisioning of that order will be recovered in accordance with the applicable sections of the GSST and the PLST.

DIS	ഗ	TI	UT+	•
			•	

	DIOCOC	<u> </u>	
STATE	RESIDENCE	BUSINESS	CSAs***
ALABAMA	16.3%	16.3%	
FLORIDA	21.83%	16.81%	
GEORGIA	20.3%	17.3%	
KENTUCKY ,	16.79%	15.54%	· · · · · · · · · · · · · · · · · · ·
LOUISIANA	20.72%	20.72%	9.05%
MISSISSIPPI	15.75%	15.75%	
NORTH CAROLINA	21.5%	17.6%	
SOUTH CAROLINA	14.8%	14.8%	8.98%
TENNESSEE**	16%	16%	

- * When a CLEC provides Resale service in a cross boundary area (areas that are part of the local serving area of another state's exchange) the rates, regulations and discounts for the tariffing state will apply. Billing will be from the serving state.
- ** In Tennessee, if a CLEC provides its own operator services and directory services, the discount shall be 21.56%. CLEC must provide written notification to BellSouth within 30 days prior to providing its own operator services and directory services to qualify for the higher discount rate of 21.56%.
- *** Unless noted in this column, the discount for Business will be the applicable discount rate for CSAs.

Attachment 1
Page 19
EXHIBIT B

Exclusions and Limitations On Services Available for Resale

TN	Dir unt		\ \	Vec	S	Mote	COION		0 Zi		i L	\ \ \	3	S :	წ 	1		ટ	N.S.	2	ů		ŝ 		\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \	55		
L	Resale		Yes	> >	S .	3	ິນ	,	X es		S L	\$	3	Yes	Yes	3	S L	Yes	7.5	ICS	Yes		Yes		\$	S		
SC	Resale Discount		Yes	\ \ \	Ü	7	ıcs	;	o Z	;	x es	No.	- ()	Z .	Yes	1	o Z	ž	1	ONI	Yes		Š.		1	0 		
Ø	Resale		Yes	3,	ខ	3	E E	;	Yes	;	X CS	3,	153	Note 8	∠es	,	Yes	25	1	I CS	Yes		Yes			S		
N.	Discount		Yes	35	S L	3	S C		o Z	;	8	1	23	Yes	Yes	;	ę Z	^o Z		o Z	Yes		ž		;	res		
Z	Resale		Yes		X S	;	8	:	Yes		Yes	;	Yes	Yes	Yes	;	Yes	¥ ×	;	چ 	Yes		Yes];	Yes		
MS	Discount		Yes	1	S X	:	Z Z		o Z		Yes	,	Yes	Yes	Yes		ç Ž	ž		ĝ 	χes		No		;	X CS		
2	Resale		Yes	;	ສ 		X X		\ \ \	,	& }	;	Yes	Yes	Yes		Yes	y X	3 :	\ 	×es		Yes			Yes		
F 4	scount		Yes	ļ	S X		Yes		ž 		Υcs			Note 8	Yes		ĝ	2		2 	×ex		ŝ			Yes		
	Resale		Yes	,	<u>જ</u>		Xes		Yes		Yes	ľ	Note 7	Note 8	Yes		Υes	\ }	3	ပ	۲ >	} 	Yes			Yes		
X	Resale Discount		Yes		Xes		Yes		%		Note 4		Yes	Note 8	Yes		°Ž	ž	2	ջ 	2 >	3	ટ્ટ			Z X		
			Yes		Yes		Yes		Xes		Note 4		Yes	Note 8	Yes		Yes	3	G	્ર 	2	3	35			Yes		
 	Discount	The second	Yes		Yes		Yes		ž		Yes		Yes	Yes	Yes		ž	Į.	2	ž	2	3	^o Z			Yes		
	Pecolo	21170	Yes		Yes		Yes		χ		Yes		Yes	Yes	Yes		Yes	12,2	3	Yes		3 	Yes			દ્ર		
	December December Discount Receipt Discount	TO SACOURIE	Yes		Yes		Yes		Νo		Yes		Yes	Yes	Yes		%	1	2	ž	,	S	ů			Yes		
	Deserte	The Sall	Yes		Yes		Yæ		Yes		Yes		Yes	Yes	Yes		Yes	,	2	Yes	3	<u> </u>	Yes			Yes		
14		N Second	Yes		Yes		Yes		No		Yes		Yes	Yes	Yes		å	;	S Z	ž	1	ß	ર			Yes		
	,	Nesale	Yes		۲œ		Yes		Yes		Yes		Yes	χœχ	× S		Yes		Z Z	Yes		<u> </u>	×			Yes		
Turn of Country	Type of Set vice		1 Grandfathered	Services (Note 1)	2 Contract Service	Arrangements	3 Promotions - > 90	Days(Note 2)	4 Promotions - < 90	Days (Note 2)	5 Lifeline/Link Up	Services	6 911/E911 Services	7 N11 Services	8 AdWatch SM Svc (See	Note 6)	9 MemoryCall®	Service	10 Mobile Services	11 Federal Subscriber	Line Changes	12 Non-Kecurring	13 End User Line	Charge Number	Portability	14 Public Telephone	Access Service	(PTAS)

On Services Available for Resale **Exclusions and Limitations**

Applicable

Notes:

- Grandfathered services can be resold only to existing subscribers of the grandfathered service.
- Where available for resale, promotions will be made available only to end users who would have qualified for the promotion had it been provided by BellSouth directly.
- In Tennessee, long-term promotions (offered for more than ninety (90) days) may be obtained at one of the following rates: 3 5 -
 - (a) the stated tariff rate, less the wholesale discount;
- (b) the promotional rate (the promotional rate offered by BellSouth will not be discounted further by the wholesale discount rate)
- Lifeline/Link Up services may be offered only to those subscribers who meet the criteria that BellSouth currently applies to subscribers of these services as set forth in Sections A3 and A4 of the BellSouth General Subscriber Services Tariff.
 - Some of BellSouth's local exchange and toll telecommunications services are not available in certain central offices and areas.
 - AdWatchSM Service is tariffed as BellSouth AIN Virtual Number Call Detail Service.
- In Louisiana 911/E911 services are not available to be resold. NewSouth's customers, however, will be able to contact the appropriate 911/E911 service provider when service is provided on a resold basis. . 6 °
 - N11 services are not available to be resold, however, NewSouth's customers will be able to make calls to N11 providers.

LINE INFORMATION DATA BASE (LIDB) STORAGE AGREEMENT

I. SCOPE

- A. This Agreement sets forth the terms and conditions pursuant to which BellSouth agrees to store in its LIDB certain information at the request of NewSouth and pursuant to which BellSouth, its LIDB customers and NewSouth shall have access to such information. NewSouth understands that BellSouth provides access to information in its LIDB to various telecommunications service providers pursuant to applicable tariffs and agrees that information stored at the request of NewSouth, pursuant to this Agreement, shall be available to those telecommunications service providers. The terms and conditions contained in the attached Addendum(s) are hereby made a part of this Agreement as if fully incorporated herein.
- B. LIDB is accessed for the following purposes:
 - 1. Billed Number Screening
 - 2. Calling Card Validation
 - 3. Fraud Control
- C. BellSouth will provide seven days per week, 24-hours per day, fraud monitoring on Calling Cards, bill-to-third and collect calls made to numbers in BellSouth's LIDB, provided that such information is included in the LIDB query. BellSouth will establish fraud alert thresholds and will notify NewSouth of fraud alerts so that NewSouth may take action it deems appropriate. NewSouth understands and agrees BellSouth will administer all data stored in the LIDB, including the data provided by NewSouth pursuant to this Agreement, in the same manner as BellSouth's data for BellSouth's end user customers. BellSouth shall not be responsible to NewSouth for any lost revenue which may result from BellSouth's administration of the LIDB pursuant to its established practices and procedures as they exist and as they may be changed by BellSouth in its sole discretion from time to time.

NewSouth understands that BellSouth currently has in effect numerous billing and collection agreements with various interexchange carriers and billing clearing houses. NewSouth further understands that these billing and collection customers of BellSouth query BellSouth's LIDB to determine whether to accept various billing options from end users. Additionally, NewSouth understands that presently BellSouth has no method to differentiate between BellSouth's own billing and line data in the LIDB and such data which it includes in the LIDB on NewSouth's behalf pursuant to this Agreement. Therefore, until such time as BellSouth can and does implement in its LIDB and its supporting systems the means to differentiate

NewSouth's data from BellSouth's data and the Parties to this Agreement execute appropriate amendments hereto, the following terms and conditions shall apply:

- (a) NewSouth agrees that it will accept responsibility for telecommunications services billed by BellSouth for its billing and collection customers for NewSouth's end user accounts which are resident in LIDB pursuant to this Agreement. NewSouth authorizes BellSouth to place such charges on NewSouth's bill from BellSouth and agrees that it shall pay all such charges. Charges for which NewSouth hereby takes responsibility include, but are not limited to, collect and third number calls.
- (b) Charges for such services shall appear on a separate BellSouth bill page identified with the name of the entity for which BellSouth is billing the charge.
- (c) NewSouth shall have the responsibility to render a billing statement to its end users for these charges, but NewSouth's obligation to pay BellSouth for the charges billed shall be independent of whether NewSouth is able or not to collect from NewSouth's end users.
- (d) BellSouth shall not become involved in any disputes between NewSouth and the entities for which BellSouth performs billing and collection. BellSouth will not issue adjustments for charges billed on behalf of an entity to NewSouth. It shall be the responsibility of NewSouth and the other entity to negotiate and arrange for any appropriate adjustments.

II. TERM

This Agreement will be effective as of ______, and will continue in effect for one year, and thereafter may be continued until terminated by either Party upon thirty (30) days written notice to the other Party.

III. FEES FOR SERVICE AND TAXES

- A. NewSouth will not be charged a fee for storage services provided by BellSouth to NewSouth, as described in Section I of this Agreement.
- B. Sales, use and all other taxes (excluding taxes on BellSouth's income) determined by BellSouth or any taxing authority to be due to any federal, state or local taxing jurisdiction with respect to the provision of the service set forth herein will be paid by NewSouth. NewSouth shall have the right to have BellSouth contest with the imposing jurisdiction, at NewSouth's expense, any such taxes that NewSouth deems are improperly levied.

IV. INDEMNIFICATION

To the extent not prohibited by law, each Party will indemnify the other and hold the other harmless against any loss, cost, claim, injury, or liability relating to or arising out of negligence or willful misconduct by the indemnifying Party or its agents or contractors in connection with the indemnifying Party's provision of services. provided, however, that any indemnity for any loss, cost, claim, injury or liability arising out of or relating to errors or omissions in the provision of services under this Agreement shall be limited as otherwise specified in this Agreement. The indemnifying Party under this Section agrees to defend any suit brought against the other Party for any such loss, cost, claim, injury or liability. The indemnified Party agrees to notify the other Party promptly, in writing, of any written claims, lawsuits. or demands for which the other Party is responsible under this Section and to cooperate in every reasonable way to facilitate defense or settlement of claims. The indemnifying Party shall not be liable under this Section for settlement by the indemnified Party of any claim, lawsuit, or demand unless the defense of the claim. lawsuit, or demand has been tendered to it in writing and the indemnifying Party has unreasonably failed to assume such defense.

V. LIMITATION OF LIABILITY

Neither Party shall be liable to the other Party for any lost profits or revenues or for any indirect, incidental or consequential damages incurred by the other Party arising from this Agreement or the services performed or not performed hereunder, regardless of the cause of such loss or damage.

VI. MISCELLANEOUS

- A. It is understood and agreed to by the Parties that BellSouth may provide similar services to other companies.
- B. All terms, conditions and operations under this Agreement shall be performed in accordance with, and subject to, all applicable local, state or federal legal and regulatory tariffs, rulings, and other requirements of the federal courts, the U. S. Department of Justice and state and federal regulatory agencies. Nothing in this Agreement shall be construed to cause either Party to violate any such legal or regulatory requirement and either Party's obligation to perform shall be subject to all such requirements.
- C. NewSouth agrees to submit to BellSouth all advertising, sales promotion, press releases, and other publicity matters relating to this Agreement wherein BellSouth's corporate or trade names, logos, trademarks or service marks or those of BellSouth's affiliated companies are mentioned or language from which the connection of said names or trademarks therewith may be inferred or implied; and NewSouth further agrees not to publish or use advertising, sales promotions, press releases, or publicity matters without BellSouth's prior written approval.

- D. This Agreement constitutes the entire Agreement between NewSouth and BellSouth which supersedes all prior Agreements or contracts, oral or written representations, statements, negotiations, understandings, proposals and undertakings with respect to the subject matter hereof.
- E. Except as expressly provided in this Agreement, if any part of this Agreement is held or construed to be invalid or unenforceable, the validity of any other Section of this Agreement shall remain in full force and effect to the extent permissible or appropriate in furtherance of the intent of this Agreement.
- F. Neither Party shall be held liable for any delay or failure in performance of any part of this Agreement for any cause beyond its control and without its fault or negligence, such as acts of God, acts of civil or military authority, government regulations, embargoes, epidemics, war, terrorist acts, riots, insurrections, fires, explosions, earthquakes, nuclear accidents, floods, strikes, power blackouts, volcanic action, other major environmental disturbances, unusually severe weather conditions, inability to secure products or services of other persons or transportation facilities, or acts or omissions of transportation common carriers.
- G. This Agreement shall be deemed to be a contract made under the laws of the State of Georgia, and the construction, interpretation and performance of this Agreement and all transactions hereunder shall be governed by the domestic law of such State.

RESALE ADDENDUM TO LINE INFORMATION DATA BASE (LIDB) STORAGE AGREEMENT

This is a Resale Addendum to the Line Information Data Base Storage Agreement dated, 2000, between BellSouth Telecommunications, Inc.									
("Bell! 2000.	South"), and NewSouth ("NewSouth"), effective the day of,								
I.	GENERAL								
	This Addendum sets forth the terms and conditions for NewSouth's provision of billing number information to BellSouth for inclusion in BellSouth's LIDB. BellSouth will store in its LIDB the billing number information provided by NewSouth, and BellSouth will provide responses to on-line, call-by-call queries to this information for purposes specified in Section I.B. of the Agreement.								
II.	DEFINITIONS								
Α.	Billing number - a number used by BellSouth for the purpose of identifying an account liable for charges. This number may be a line or a special billing number.								
B.	Line number - a ten-digit number assigned by BellSouth that identifies a telephone line associated with a resold local exchange service, or with a SPNP arrangement.								
C.	Special billing number - a ten-digit number that identifies a billing account established by BellSouth in connection with a resold local exchange service or with a SPNP arrangement.								
D.	Calling Card number - a billing number plus PIN number assigned by BellSouth.								
E.	PIN number - a four digit security code assigned by BellSouth which is added to a billing number to compose a fourteen digit calling card number.								
F.	Toll billing exception indicator - associated with a billing number to indicate that it is considered invalid for billing of collect calls or third number calls or both, by the NewSouth.								
G.	Billed Number Screening - refers to the activity of determining whether a toll billing exception indicator is present for a particular billing number.								

- H. Calling Card Validation refers to the activity of determining whether a particular calling card number exists as stated or otherwise provided by a caller.
- I. Billing number information information about billing number or Calling Card number as assigned by BellSouth and toll billing exception indicator provided to BellSouth by the NewSouth.

III. RESPONSIBILITIES OF PARTIES

- A. BellSouth will include billing number information associated with resold exchange lines or SPNP arrangements in its LIDB. The NewSouth will request any toll billing exceptions via the Local Service Request (LSR) form used to order resold exchange lines, or the SPNP service request form used to order SPNP arrangements.
- B. Under normal operating conditions, BellSouth shall include the billing number information in its LIDB upon completion of the service order establishing either the resold local exchange service or the SPNP arrangement, provided that BellSouth shall not be held responsible for any delay or failure in performance to the extent such delay or failure is caused by circumstances or conditions beyond BellSouth's reasonable control. BellSouth will store in its LIDB an unlimited volume of the working telephone numbers associated with either the resold local exchange lines or the SPNP arrangements. For resold local exchange lines or for SPNP arrangements, BellSouth will issue line-based calling cards only in the name of NewSouth. BellSouth will not issue line-based calling cards in the name of NewSouth's individual end users. In the event that NewSouth wants to include calling card numbers assigned by the NewSouth in the BellSouth LIDB, a separate agreement is required.
- C. BellSouth will provide responses to on-line, call-by-call queries to the stored information for the specific purposes listed in the next paragraph.
- D. BellSouth is authorized to use the billing number information to perform the following functions for authorized users on an on-line basis:
- 1. Validate a 14 digit Calling Card number where the first 10 digits are a line number or special billing number assigned by BellSouth, and where the last four digits (PIN) are a security code assigned by BellSouth.
- 2. Determine whether the NewSouth has identified the billing number as one which should not be billed for collect or third number calls, or both.

RAO Hosting

- 1. RAO Hosting, Calling Card and Third Number Settlement System (CATS) and Non-Intercompany Settlement System (NICS) services provided to NewSouth by BellSouth will be in accordance with the methods and practices regularly adopted and applied by BellSouth to its own operations during the term of this Agreement, including such revisions as may be made from time to time by BellSouth.
- 2. NewSouth shall furnish all relevant information required by BellSouth for the provision of RAO Hosting, CATS and NICS.
- 3. Applicable compensation amounts will be billed by BellSouth to NewSouth on a monthly basis in arrears. Amounts due from one Party to the other (excluding adjustments) are payable within thirty (30) days of receipt of the billing statement.
- 4. NewSouth must have its own unique RAO code. Requests for establishment of RAO status where BellSouth is the selected Centralized Message Distribution System (CMDS) interfacing host, require written notification from NewSouth to the BellSouth RAO Hosting coordinator at least eight (8) weeks prior to the proposed effective date. The proposed effective date will be mutually agreed upon between the Parties with consideration given to time necessary for the completion of required Telcordia (formerly BellCore) functions. BellSouth will request the assignment of an RAO code from its connecting contractor, currently Telcordia (formerly BellCore), on behalf of NewSouth and will coordinate all associated conversion activities.
- 5. BellSouth will receive messages from NewSouth that are to be processed by BellSouth, another LEC or CLEC in the BellSouth region or a LEC outside the BellSouth region.
- 6. BellSouth will perform invoice sequence checking, standard EMI format editing, and balancing of message data with the EMI trailer record counts on all data received from NewSouth.
- 7. All data received from NewSouth that is to be processed or billed by another LEC or CLEC within the BellSouth region will be distributed to that LEC or CLEC in accordance with the agreement(s) which may be in effect between BellSouth and the involved LEC or CLEC.
- 8. All data received from NewSouth that is to be placed on the CMDS network for distribution outside the BellSouth region will be handled in accordance with the

- agreement(s) which may be in effect between BellSouth and its connecting contractor (currently Telcordia (formerly BellCore)).
- 9. BellSouth will receive messages from the CMDS network that are destined to be processed by NewSouth and will forward them to NewSouth on a daily basis.
- 10. Transmission of message data between BellSouth and NewSouth will be via CONNECT:Direct.
- 11. All messages and related data exchanged between BellSouth and NewSouth will be formatted in accordance with accepted industry standards for EMI formatted records and packed between appropriate EMI header and trailer records, also in accordance with accepted industry standards.
- 12. NewSouth will ensure that the recorded message detail necessary to recreate files provided to BellSouth will be maintained for back-up purposes for a period of three (3) calendar months beyond the related message dates.
- 13. Should it become necessary for NewSouth to send data to BellSouth more than sixty (60) days past the message date(s), NewSouth will notify BellSouth in advance of the transmission of the data. If there will be impacts outside the BellSouth region, BellSouth will work with its connecting contractor and NewSouth to notify all affected Parties.
- In the event that data to be exchanged between the two Parties should become lost or destroyed, both Parties will work together to determine the source of the problem. Once the cause of the problem has been jointly determined and the responsible Party (BellSouth or NewSouth) identified and agreed to, the company responsible for creating the data (BellSouth or NewSouth) will make every effort to have the affected data restored and retransmitted. If the data cannot be retrieved, the responsible Party will be liable to the other Party for any resulting lost revenue. Lost revenue may be a combination of revenues that could not be billed to the end users and associated access revenues. Both Parties will work together to estimate the revenue amount based upon historical data through a method mutually agreed upon. The resulting estimated revenue loss will be paid by the responsible Party to the other Party within three (3) calendar months of the date of problem resolution, or as mutually agreed upon by the Parties.
- 15. Should an error be detected by the EMI format edits performed by BellSouth on data received from NewSouth, the entire pack containing the affected data will not be processed by BellSouth. BellSouth will notify NewSouth of the error condition.

NewSouth will correct the error(s) and will resend the entire pack to BellSouth for processing. In the event that an out-of-sequence condition occurs on subsequent packs, NewSouth will resend these packs to BellSouth after the pack containing the error has been successfully reprocessed by BellSouth.

- In association with message distribution service, BellSouth will provide NewSouth with associated intercompany settlements reports (CATS and NICS) as appropriate.
- 17. In no case shall either Party be liable to the other for any direct or consequential damages incurred as a result of the obligations set out in this agreement.
- 18. RAO Compensation
- 18.1 Rates for message distribution service provided by BellSouth for NewSouth are as set forth in Exhibit A to this Attachment.
- 18.2 Rates for data transmission associated with message distribution service are as set forth in Exhibit A to this Attachment.
- Data circuits (private line or dial-up) will be required between BellSouth and NewSouth for the purpose of data transmission. Where a dedicated line is required, NewSouth will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. NewSouth will also be responsible for any charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to NewSouth. Additionally, all message toll charges associated with the use of the dial circuit by NewSouth will be the responsibility of NewSouth. Associated equipment on the BellSouth end, including a modem, will be negotiated on a case by case basis between the Parties.
- All equipment, including modems and software, that is required on the NewSouth end for the purpose of data transmission will be the responsibility of NewSouth.
- 19. <u>Intercompany Settlements Messages</u>
- This Section addresses the settlement of revenues associated with traffic originated from or billed by NewSouth as a facilities based provider of local exchange telecommunications services outside the BellSouth region. Only traffic that originates in one Bell operating territory and bills in another Bell operating territory is included.

Traffic that originates and bills within the same Bell operating territory will be settled on a local basis between NewSouth and the involved company(ies), unless that company is participating in NICS.

- 19.2 Both traffic that originates outside the BellSouth region by NewSouth and is billed within the BellSouth region, and traffic that originates within the BellSouth region and is billed outside the BellSouth region by NewSouth, is covered by this Agreement (CATS). Also covered is traffic that either is originated by or billed by NewSouth, involves a company other than NewSouth, qualifies for inclusion in the CATS settlement, and is not originated or billed within the BellSouth region (NICS).
- Once NewSouth is operating within the BellSouth territory, revenues associated with calls originated and billed within the BellSouth region will be settled via Telcordia (formerly BellCore)'s, its successor or assign, NICS system.
- 19.4 BellSouth will receive the monthly NICS reports from Telcordia (formerly BellCore), its successor or assign, on behalf of NewSouth. BellSouth will distribute copies of these reports to NewSouth on a monthly basis.
- 19.5 BellSouth will receive the monthly Calling Card and Third Number Settlement System (CATS) reports from Telcordia (formerly BellCore), its successor or assign, on behalf of NewSouth. BellSouth will distribute copies of these reports to NewSouth on a monthly basis.
- 19.6 BellSouth will collect the revenue earned by NewSouth from the Bell operating company in whose territory the messages are billed (CATS), less a per message billing and collection fee of five cents (\$0.05), on behalf of NewSouth. BellSouth will remit the revenue billed by NewSouth to the Bell operating company in whose territory the messages originated, less a per message billing and collection fee of five cents (\$0.05), on behalf on NewSouth. These two amounts will be netted together by BellSouth and the resulting charge or credit issued to NewSouth via a monthly Carrier Access Billing System (CABS) miscellaneous bill.
- 19.7 BellSouth will collect the revenue earned by NewSouth within the BellSouth territory from another CLEC also within the BellSouth territory (NICS) where the messages are billed, less a per message billing and collection fee of five cents (\$0.05), on behalf of NewSouth. BellSouth will remit the revenue billed by NewSouth within the BellSouth region to the CLEC also within the BellSouth region, where the messages originated, less a per message billing and collection fee of five cents (\$0.05). These two amounts will be netted together by BellSouth and the resulting charge or credit

issued to NewSouth via a monthly Carrier Access Billing System (CABS) miscellaneous bill.

BellSouth and NewSouth agree that monthly netted amounts of less than fifty dollars (\$50.00) will not be settled.

PAGE INTENTIONALLY LEFT BLANK

Optional Daily Usage File

- 1. Upon written request from NewSouth, BellSouth will provide the Optional Daily Usage File (ODUF) service to NewSouth pursuant to the terms and conditions set forth in this section.
- 2. NewSouth shall furnish all relevant information required by BellSouth for the provision of the Optional Daily Usage File.
- The Optional Daily Usage Feed will contain billable messages that were carried over the BellSouth Network and processed in the BellSouth Billing System, but billed to a NewSouth customer.

Charges for delivery of the Optional Daily Usage File will appear on NewSouths' monthly bills. The charges are as set forth in Exhibit A to this Attachment.

- 4. The Optional Daily Usage Feed will contain both rated and unrated messages. All messages will be in the standard Alliance for Telecommunications Industry Solutions (ATIS) EMI record format.
- Messages that error in NewSouth's billing system will be the responsibility of NewSouth. If, however, NewSouth should encounter significant volumes of errored messages that prevent processing by NewSouth within its systems, BellSouth will work with the to determine the source of the errors and the appropriate resolution.
- 6. The following specifications shall apply to the Optional Daily Usage Feed.
- 6.1 <u>Usage To Be Transmitted</u>
- 6.1.1 The following messages recorded by BellSouth will be transmitted to NewSouth:
 - Message recording for per use/per activation type services (examples: Three Way Calling, Verify, Interrupt, Call Return, ETC.)
 - Measured billable Local
 - Directory Assistance messages
 - IntraLATA Toll
 - WATS & 800 Service
 - N11

- Information Service Provider Messages
- Operator Services Messages
- Operator Services Message Attempted Calls (UNE only)
- Credit/Cancel Records
- Usage for Voice Mail Message Service
- Rated Incollects (originated in BellSouth and from other companies) can also be on Optional Daily Usage File. Rated Incollects will be intermingled with BellSouth recorded rated and unrated usage. Rated Incollects will not be packed separately.
- 6.1.3 BellSouth will perform duplicate record checks on records processed to Optional Daily Usage File. Any duplicate messages detected will be deleted and not sent to NewSouth.
- 6.1.4. In the event that NewSouth detects a duplicate on Optional Daily Usage File they receive from BellSouth, NewSouth will drop the duplicate message (NewSouth will not return the duplicate to BellSouth).
- 6.2 Physical File Characteristics
- The Optional Daily Usage File will be distributed to NewSouth via an agreed medium with CONNECT:Direct being the preferred transport method. The Daily Usage Feed will be a variable block format (2476) with an LRECL of 2472. The data on the Daily Usage Feed will be in a non-compacted EMI format (175 byte format plus modules). It will be created on a daily basis (Monday through Friday except holidays). Details such as dataset name and delivery schedule will be addressed during negotiations of the distribution medium. There will be a maximum of one dataset per workday per OCN.
- Data circuits (private line or dial-up) may be required between BellSouth and NewSouth for the purpose of data transmission. Where a dedicated line is required, NewSouth will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. NewSouth will also be responsible for any charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to NewSouth. Additionally, all message toll charges associated with

the use of the dial circuit by NewSouth will be the responsibility of NewSouth. Associated equipment on the BellSouth end, including a modem, will be negotiated on a case by case basis between the parties. All equipment, including modems and software, that is required on NewSouth end for the purpose of data transmission will be the responsibility of NewSouth.

6.3 <u>Packing Specifications</u>

- 6.3.1 A pack will contain a minimum of one message record or a maximum of 99,999 message records plus a pack header record and a pack trailer record. One transmission can contain a maximum of 99 packs and a minimum of one pack.
- The OCN, From RAO, and Invoice Number will control the invoice sequencing. The From RAO will be used to identify to NewSouth which BellSouth RAO that is sending the message. BellSouth and NewSouth will use the invoice sequencing to control data exchange. BellSouth will be notified of sequence failures identified by NewSouth and resend the data as appropriate.

THE DATA WILL BE PACKED USING ATIS EMI RECORDS.

6.4 Pack Rejection

NewSouth will notify BellSouth within one business day of rejected packs (via the mutually agreed medium). Packs could be rejected because of pack sequencing discrepancies or a critical edit failure on the Pack Header or Pack Trailer records (i.e. out-of-balance condition on grand totals, invalid data populated). Standard ATIS EMI Error Codes will be used. NewSouth will not be required to return the actual rejected data to BellSouth. Rejected packs will be corrected and retransmitted to NewSouth by BellSouth.

6.5 Control Data

NewSouth will send one confirmation record per pack that is received from BellSouth. This confirmation record will indicate NewSouth received the pack and the acceptance or rejection of the pack. Pack Status Code(s) will be populated using standard ATIS EMI error codes for packs that were rejected by NewSouth for reasons stated in the above section.

6.6 Testing

Upon request from NewSouth, BellSouth shall send test files to NewSouth for the Optional Daily Usage File. The parties agree to review and discuss the file's content and/or format. For testing of usage results, BellSouth shall request that NewSouth set up a production (LIVE) file. The live test may consist of NewSouth's employees

making test calls for the types of services NewSouth requests on the Optional Daily Usage File. These test calls are logged by NewSouth, and the logs are provided to BellSouth. These logs will be used to verify the files. Testing will be completed within 30 calendar days from the date on which the initial test file was sent.

Enhanced Optional Daily Usage File

- 1. Upon written request from NewSouth, BellSouth will provide the Enhanced Optional Daily Usage File (EODUF) service to NewSouth pursuant to the terms and conditions set forth in this section. EODUF will only be sent to existing ODUF subscribers who request the EODUF option.
- 2. The NewSouth shall furnish all relevant information required by BellSouth for the provision of the Enhanced Optional Daily Usage File.
- 3. The Enhanced Optional Daily Usage File (EODUF) will provide usage data for local calls originating from resold Flat Rate Business and Residential Lines.
- 4. Charges for delivery of the Enhanced Optional Daily Usage File will appear on NewSouths' monthly bills. The charges are as set forth in Exhibit A to this Attachment.
- 5. All messages will be in the standard Alliance for Telecommunications Industry Solutions (ATIS) EMI record format.
- 6. Messages that error in the billing system of NewSouth will be the responsibility of NewSouth. If, however, NewSouth should encounter significant volumes of errored messages that prevent processing by NewSouth within its systems, BellSouth will work with NewSouth to determine the source of the errors and the appropriate resolution.
- 7. The following specifications shall apply to the Optional Daily Usage Feed.
- 7.1 <u>Usage To Be Transmitted</u>
- 7.1.1 The following messages recorded by BellSouth will be transmitted to NewSouth:

Customer usage data for flat rated local call originating from NewSouth's end user lines (1FB or 1FR). The EODUF record for flat rate messages will include:

Date of Call
From Number
To Number
Connect Time
Conversation Time
Method of Recording
From RAO
Rate Class
Message Type

Billing Indicators Bill to Number

- 7.1.2 BellSouth will perform duplicate record checks on EODUF records processed to Optional Daily Usage File. Any duplicate messages detected will be deleted and not sent to NewSouth.
- 7.1.3 In the event that NewSouth detects a duplicate on Enhanced Optional Daily Usage File they receive from BellSouth, NewSouth will drop the duplicate message (NewSouth will not return the duplicate to BellSouth).

7.2 Physical File Characteristics

- 7.2.1 The Enhanced Optional Daily Usage Feed will be distributed to NewSouth over their existing Optional Daily Usage File (ODUF) feed. The EODUF messages will be intermingled among NewSouth's Optional Daily Usage File (ODUF) messages. The EODUF will be a variable block format (2476) with an LRECL of 2472. The data on the EODUF will be in a non-compacted EMI format (175 byte format plus modules). It will be created on a daily basis (Monday through Friday except holidays).
- 7.2.2 Data circuits (private line or dial-up) may be required between BellSouth and NewSouth for the purpose of data transmission. Where a dedicated line is required, NewSouth will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. NewSouth will also be responsible for any charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to NewSouth. Additionally, all message toll charges associated with the use of the dial circuit by NewSouth will be the responsibility of NewSouth. Associated equipment on the BellSouth end, including a modem, will be negotiated on a case by case basis between the parties. All equipment, including modems and software, that is required on NewSouth's end for the purpose of data transmission will be the responsibility of NewSouth.

7.3 <u>Packing Specifications</u>

- 7.3.1 A pack will contain a minimum of one message record or a maximum of 99,999 message records plus a pack header record and a pack trailer record. One transmission can contain a maximum of 99 packs and a minimum of one pack.
- 7.3.2 The Operating Company Number (OCN), From Revenue Accounting Office (RAO), and Invoice Number will control the invoice sequencing. The From RAO will be used to identify to NewSouth which BellSouth RAO that is sending the message.

BellSouth and NewSouth will use the invoice sequencing to control data exchange. BellSouth will be notified of sequence failures identified by NewSouth and resend the data as appropriate.

THE DATA WILL BE PACKED USING ATIS EMI RECORDS.

ODUF/EDOUF/CMDS	BELLSOUTHWEWSOUTH RATES
-----------------	-------------------------

Exhibit G

Attachment 2

Network Elements and Other Services

TABLE OF CONTENTS

1.	INTRODUCTION3
2.	UNBUNDLED LOOPS, INTEGRATED DIGITAL LOOP CARRIERS, NETWORK INTERFACES DEVICE, UNBUNDLED LOOP CONCENTRATION (ULC) SYSTEM, SUB LOOPS AND DARK FIBER
3.	SWITCHING22
4.	ENHANCED EXTENDED LINK (EEL)32
5.	PORT/LOOP COMBINATIONS38
6	TRANSPORT AND DARK FIBER41
7	BELLSOUTH SWA 8XX TOLL FREE DIALING TEN DIGIT SCREENING SERVICE47
8	LINE INFORMATION DATABASE (LIDB)48
9	SIGNALING51
10.	OPERATOR CALL PROCESSING, INWARD OPERATOR SERVICES AND DIRECTORY ASSISTANCE SERVICES
11.	CALLING NAME (CNAM) DATABASE SERVICE65
12.	BASIC 911 AND E91166
13.	TRUE-UP68
LII	OB Storage AgreementExhibit A
CN.	AM Database ServicesExhibit B
	es

ACCESS TO NETWORK ELEMENTS AND OTHER SERVICES

1. Introduction

- 1.1. This Attachment sets forth the unbundled network elements and combinations of unbundled network elements that BellSouth agrees to offer to NewSouth in accordance with its obligations under Section 251(c)(3) of the Act. The specific terms and conditions that apply to the unbundled network elements are described below in this Attachment 2. The price for each unbundled network element and combination of unbundled Network Elements are set forth in Exhibit A of this Agreement. As an option, deaveraged rates, where available, are included in Exhibit A.
- 1.2. For purposes of this Agreement, "Network Element" is defined to mean a facility or equipment provided by BellSouth on an unbundled basis as is used by the CLEC in the provision of a telecommunications service. These unbundled network elements will be consistent with the requirements of the FCC 319 rule. For purposes of this Agreement, combinations of Network Elements shall be referred to as "Combinations."
- 1.2.1. Except as otherwise permitted by law, BellSouth shall not impose limitation restrictions or requirements or request for the use of the network elements or combinations that would impair the ability of NewSouth to offer telecommunications service in the manner NewSouth intends.
- 1.2.2. Except upon request by NewSouth, BellSouth shall not separate requested network elements that BellSouth currently combines.
- 1.3. BellSouth shall, upon request of NewSouth, and to the extent technically feasible, provide to NewSouth access to its network elements for the provision of NewSouth's telecommunications services. If no rate is identified in the contract, the rate for the specific service or function will be as ordered by the Commission. If the Commission has not ordered a rate then the rates will be as set forth in applicable BellSouth tariff or as negotiated by the Parties upon request by either Party.
- 1.4. NewSouth may purchase network elements and other services from BellSouth for the purpose of combining such network elements in any manner NewSouth chooses to provide telecommunication services to its intended users, including recreating existing BellSouth services. With the exception of the sub-loop elements, which are located outside of the central office, BellSouth shall deliver the network elements purchased by NewSouth for combining to the designated NewSouth collocation space or any other technically feasible point. The network elements shall be provided as set forth in this Attachment.

- 1.5. Subject to applicable and effective FCC Rules and Orders as well as effective State Commission Orders, BellSouth will offer combinations of network elements pursuant to such orders. In addition to the combinations set forth in Sections 4 and 5 BellSouth will provide the following combined network elements for purchase by NewSouth. The rate of the following combined network elements is the sum of the individual element prices as set forth in this Attachment. Except as specified below, Order Coordination as defined in Section 2 of Attachment 2 of this Agreement is available for each of these combinations:
 - SL1 Loop and cross connect
 - SL2 loop and cross connect
 - Port and cross connect
 - Port and cross connect and common (shared) transport
 - Port and vertical features
 - SL2 Loop with loop concentration
 - Port and common (shared) transport
 - SL1 Loop and LNP
 - SL2 Loop and LNP
- 1.6. NewSouth will adopt and adhere to the reasonable and non-discriminatory standards contained in the applicable CLEC Work Center Operational Understanding Agreement regarding maintenance and installation of service. Provided, however, nothing herein, shall override the Parties rights or obligations under this agreement.
- 1.7. Standards for Network Elements
- 1.7.1 BellSouth shall comply with the requirements set forth in the technical references, as well as any performance or other requirements identified in this Agreement, to the extent that they are consistent with the greater of BellSouth's actual performance or applicable industry standards.
- 1.7.2 If one or more of the requirements set forth in this Agreement are in conflict, the parties shall mutually agree on which requirement shall apply. If the parties cannot reach agreement, the dispute resolution process set forth in Section 12 of the General Terms and Conditions of this Agreement, incorporated herein by this reference, shall apply.
- 2. Unbundled Loops, Integrated Digital Loop Carriers, Network Interfaces Device, Unbundled Loop Concentration (ULC) System, Sub loops and Dark Fiber
- 2.1 Unbundled Loops

2.1.1 Definition

- 2.1.2 The local loop network element ("Loop(s)") is defined as a transmission facility between a distribution frame (or its equivalent) in BellSouth's central office and the loop demarcation point at an end-user customer premises, including inside wire owned by BellSouth. The local loop network element includes all features, functions, and capabilities of the transmission facilities, including dark fiber and attached electronics (except those used for the provision of advanced services, such as Digital Subscriber Line Access Multiplexers) and line conditioning. The loop shall include the use of all test access functionality, including without limitation, smart jacks, for both voice and data. NewSouth shall be entitled to order all loops set forth in Exhibit C of this Attachment. Unless otherwise requested, all loops will be provisioned with the appropriate Network Interface Device (NID).
- 2.1.3 The provisioning of service to NewSouth will require cross-office cabling and cross-connections within the central office to connect the loop to a local switch, or to other transmission equipment. in collocation space. These cross-connects are a separate element and are not considered a part of the loop.
- 2.1.4 BellSouth Order Coordination referenced in Attachment 2 includes two types: "Order Coordination" and "Order Coordination Time Specific."
- 2.1.5 "Order Coordination" refers to standard BellSouth service order coordination involving the reuse of facilities where NewSouth is requesting that their loop order be provisioned over an existing circuit that is currently providing service to the end-user. Order coordination for physical conversions will be scheduled at BellSouth's discretion during normal working hours on the committed due date and NewSouth will be advised. OC will be provided as a standard item on SL2 voice grade loops and all Unbundled Digital Loops (UDLs). OC will be provided as a chargeable option on SL1 voice grade loops, and all Unbundled Copper Loops.
- 2.1.5.1 For a coordinated conversion, i.e., stand alone INP, INP or LNP with loop, or stand alone loop where order coordination is provided for in this agreement, BellSouth shall verbally coordinate the disconnect with NewSouth and perform any switch translations so as to limit end user service outage. BellSouth and NewSouth will mutually agree upon a cut-over time 24 to 48 hours prior to the actual conversion. NewSouth may designate the conversion time when the conversion involves a loop with ILNP or LNP by ordering time specific conversion at rates designated in this agreement. For time specific conversions, BellSouth will verify the cut-over time designated by NewSouth 24 to 48 hours in advance to ensure that the conversion is to be completed as ordered. Both parties will use best efforts to ensure mutually agreed to conversion times, as identified in this paragraph, will commence within 15 minutes of the agreed time. For coordinated conversions, BellSouth's target intervals for service disruption to the end-user is 15 minutes or less for each loop.

2.1.6 "Order Coordination - Time Specific" refers to service order coordination in which NewSouth requests a specific time for a service order conversion to take place. BellSouth will make every effort to accommodate NewSouth's specific conversion time request. However, BellSouth reserves the right to negotiate with NewSouth a conversion time based on load and appointment control when necessary. Loops on a single service order of 14 or more loops will be provisioned on a project basis. This is a chargeable option for any coordinated order and is billed in addition to the OC charge. NewSouth may specify a time between 9:00 a.m. and 4:00 p.m. (location time) Monday through Friday (excluding holidays). If NewSouth specifies a time outside this window, or selects a time or quantity of loops that requires BellSouth technicians to work outside normal work hours (as specified in Section 1.2.1 of Attachment 6 to this Agreement), overtime charges will apply in addition to the OC and OC-TS charges. Overtime charges will be applied according to actual costs based on type of force group required to perform the work, overtime hours worked, and any special circumstances.

If NewSouth requests work to be done for any UNE loop that requires BellSouth technicians to work outside normal work hours, overtime charges will be applied according to actual costs based on type of force group required to perform the work, overtime hours worked and any special circumstances

	Order Coordinatio n (OC)	Order Coordination – Time Specific (OC-TS)	Test Points	DLR	Charge for Dispatch and Testing if No Trouble Found
SL-1	Chargeable option	Chargeable Option*	Not available	Chargeable Option ordered as Engineering Information Document	Charged for Dispatch inside & outside Central Office
SL-2	Included	Chargeable Option*	Included	Included	Charged for Dispatch outside Central Office
Unbundled Digital Loop	Included	Chargeable Option* (except on Universal Digital Channel)	Included (where appropriate)	Included	Charged for Dispatch outside Central Office
Unbundled Copper Loop	Chargeable Option	Not available	Included	Included	Charged for Dispatch outside Central Office

- *Order Coordination-Time Specific charge for orders due on same day at same location will be applied on a per LSR basis. For UVL-SL1, NewSouth must order OC when requesting OC-TS.
- Where facilities are available, BellSouth's targeted installation interval, for up to 5 2.1.7 DS1 Loops on a single order, will be 5 business days. Such interval shall include BellSouth's return of the Firm Order Confirmation. For 6 to 14 DS1 Loops on a single order, BellSouth's targeted installation interval will be 7 business days. Such interval shall include BellSouth's return of the Firm Order Confirmation. For orders of 14 or more DS1 loops, the installation will be handled, in a reasonable and nondiscriminatory manner, on a project basis and the intervals will be set by the BellSouth project manager for that order. All other Loops will be provisioned in accordance with the intervals set forth in BellSouth's Products and Services Interval Guide. Some loops require a Service Inquiry (SI) to determine if facilities are available prior to issuing the order. The interval for the SI process is separate from the installation interval. For expedite requests by NewSouth, expedite charges will apply for intervals less than 5 days. The charges outlined in Exhibit C, will apply. If NewSouth cancels an order for network elements and other services, any costs incurred by BellSouth in conjunction with the provisioning of that order will be recovered in accordance with Attachment 6 of this Agreement.
- 2.1.8 If NewSouth modifies an order after being sent a Firm Order Confirmation (FOC) from BellSouth, any costs incurred by BellSouth to accommodate the modification will be reimbursed by NewSouth.
- 2.1.9 BellSouth will offer Unbundled Voice Loops (UVL) in two different service levels Service Level One (SL1) and Service Level Two (SL2).
- 2.1.10 SL1 loops are 2-wire loop start circuits, and will be non-designed and will not have test points. OC will be offered as a chargeable option on SLI loops when reuse of existing facilities has been requested by NewSouth. NewSouth may also order OC-TS when a specified conversion time is requested. OC-TS is a chargeable option for any coordinated order and is billed in addition to the OC charge. An Engineering Information (EI) document can be ordered as chargeable option. The EI document provides loop make up information which is similar to the information normally provided in a Design Layout Record. Upon issuance of a non-coordinated order in the service order system, SL1 loops will be activated on the due date in the same manner and time frames that BellSouth normally activates POTS-type loops for its customers. If NewSouth requests work to be done for SL1s that requires BellSouth technicians to work outside normal work hours, overtime charges will be applied according to actual costs based on type of force group required to perform the work, overtime hours worked and any special circumstances. Such overtime charges shall be as set forth in BellSouth's applicable Tariff.

- SL2 loops are either 2-wire or 4-wire circuits, with test points, with or without conditioning, and will be designed with a design layout record provided to NewSouth. SL2 circuits can be provisioned with loop start, ground start or reverse battery signaling. OC is provided as a standard feature on SL2 loops. The OC feature will allow NewSouth to coordinate the installation of the loop with the disconnect of an existing customer's service and/or number portability service. In these cases, BellSouth will perform the order conversion with standard order coordination at the time designated by BellSouth during normal work hours.
- 2.1.12 BellSouth will also offer Unbundled Digital Loops (UDL). They will be designed, will be provisioned with test points (where appropriate), and will come standard with Order Coordination and a Design Layout Record (DLR).
- As a chargeable option on all loops except UVL-SL1 and UCL, BellSouth will offer Order Coordination Time Specific (OC-TS), as described in Section 2.1.6 of this Attachment. This will allow NewSouth the ability to specify the time that the coordinated conversion takes place. The OC-TS charge for orders due on the same day at the same location will be applied on a per Local Service Request (LSR) basis.
- 2.1.14 NewSouth will be responsible for testing and isolating troubles on the loops. Once NewSouth has isolated a trouble to the BellSouth provided loop, NewSouth will issue a trouble to BellSouth on the loop. BellSouth will take the actions necessary to repair the loop if a trouble actually exists. BellSouth will repair these loops in the same time frames that BellSouth repairs similarly situated loops to its customers.
- 2.1.15 If NewSouth reports a trouble on SL1 loops and no trouble actually exists, BellSouth will charge NewSouth for any dispatching and testing (both inside and outside the CO) required by BellSouth in order to confirm the loop's working status.
- 2.1.16 If NewSouth reports a trouble on SL2 loops and no trouble actually exists, BellSouth will charge NewSouth for any dispatching and testing (outside the CO) required by BellSouth in order to confirm the loop's working status. There will be no charges in the event BellSouth fails to isolate the trouble in the initial dispatch, but isolates the trouble on a subsequent dispatch.
- In addition to the UVLs and UDLs, BellSouth shall make available an Unbundled Copper Loop (UCL). The UCL will be a copper twisted pair loop that is unencumbered by any intervening equipment (e.g., filters, load coils, range extenders, digital loop carrier, or repeaters). The UCL will be offered in two versions Short and Long. A short UCL (18 kft or less) will be provisioned according to Resistance Design parameters. The long UCL (beyond 18kft) will be used when a CLEC wants to condition copper loops longer than 18kft by removing load coils and other intervening equipment. BST will only ensure electrical continuity and balance relative to tip and ring on UCLs.

- 2.1.18 The UCL will be a designed circuit, with or without conditioning, provisioned with a test point and come standard with a DLR. OC will be offered as a chargeable option on all UCL loops. Order Coordination Time Specific (OC-TS) will not be offered on UCLs.
- 2.1.19 The UCL is a dry cooper loop and is not intended to support any particular telecommunications service. NewSouth may use the UCL loop for a variety of services, including xDSL (e.g., ADSL and HDSL) services, by attaching appropriate terminal equipment of NewSouth's choosing. NewSouth will determine the type of service that will be provided over the loop.
- 2.1.20 Because the UCL loop is an unbundled loop offering that is separate and distinct from BellSouth's ADSL and HDSL capable loop offerings, NewSouth agrees that BellSouth's UCL loop will not be held to the service level and performance expectations that apply to its ADSL and HDSL unbundled loop offerings. BellSouth shall only be obligated to maintain copper continuity and provide balance relative to tip and ring on UCL loops.
- 2.1.21 The UCL loop shall be provided to CLEC in accordance with BellSouth's Technical Reference 73600, or other industry guidelines.
- 2.1.22 <u>Technical Requirements</u>
- 2.1.22.1 To the extent available within BellSouth's Network at a particular location, BellSouth will offer loops capable of supporting telecommunications services such as: POTS, Centrex, basic rate ISDN, analog PBX, voice grade private line, ADSL, HDSL, DS1 and digital data (up to 64 kb/s). If a requested loop type is not available, then the CLEC can use the Special Construction process to request that BellSouth place facilities or otherwise modify facilities in order to meet NewSouth's request.
- 2.1.22.2 NewSouth will be responsible for providing BellSouth with a Service Profile Identifier (SPID) associated with a particular ISDN-capable loop and end user. With the SPID, BellSouth will be able to adequately test the circuit and ensure that it properly supports ISDN service.
- 2.1.22.3 The loop will support the transmission, signaling, performance and interface requirements of the services described in 2.1.3 above. It is recognized that the requirements of different services are different, and that a number of types or grades of loops are required to support these services.
- 2.1.22.4 NewSouth may utilize the unbundled loops to provide any telecommunication service it wishes. However, BellSouth will only provision, maintain and repair the loops to the standards that are consistent with the type of loop ordered. For example, if NewSouth orders an ISDN-capable loop but wants to use the loop for a service other than ISDN, BellSouth will only support that the loop is capable of providing ISDN service. For non-service specific loops (e.g. UCL, loops modified by NewSouth using

- the Special Construction process), BellSouth will only support that the loop has copper continuity and balanced tip-and-ring.
- 2.1.22.5 In those cases where NewSouth has requested that BellSouth modify a loop so that it no longer meets the technical parameters of the original loop type (e.g., voice grade, ISDN, ADSL, etc.) the resulting modified loop will be ordered and maintained as a UCL.
- 2.1.22.9 The loop shall be provided to NewSouth in accordance with BellSouth's TR73600 Unbundled Local Loop Technical Specification and applicable industry standard technical references.
- 2.1.23 Universal Digital Channel (UDC) Loop
- 2.1.23.1 Due to technical limitations associated with certain DLC systems, some ISDN-capable loops that are provisioned using DLC systems may not support IDSL service. Effective with this agreement, BellSouth will no longer reconfigure its ISDN-capable loop to support IDSL service.
- 2.1.23.2 Instead, BellSouth agrees to offer the Universal Digital Channel (UDC) loop as a part of their Unbundled Digital Loop offerings. The UDC loop is intended to be compatible with IDSL service and has the same physical characteristics and transmission specifications as BellSouth's ISDN-capable loop. These specifications are listed in BellSouth's TR73600.
- 2.1.23.3 Like the ISDN-capable loop, the UDC loop may be provisioned on copper or through a DLC system. However, when UDC loops are provisioned using a DLC system, BellSouth will ensure that they are only provisioned on time slots that are compatible with data-only services such as IDSL.

2.1.24. <u>Testing</u>

2.1.24.1 BellSouth will perform the appropriate pre-service tests to ensure NewSouth dial tone is delivered to the appropriate connecting point. The timing of the test is based on the overall interval and type of the service being provisioned. Under normal intervals, testing for designed services are normally completed 24 hours in advance of the conversion. For non-designed services, dial tone is verified in time frames consistent with the same time frames that BellSouth uses to activate POTS services for it's own end users. In any event, BellSouth will advise NewSouth whenever connectivity cannot be verified with NewSouth and will work cooperatively with NewSouth to correct the problem. BellSouth will advise NewSouth at completion of the conversion or turn up of new services in order for NewSouth to accept, reject or open a maintenance ticket to BellSouth on the services being provisioned. BellSouth will work cooperatively with NewSouth to ensure end user service outage is minimal.

- Where a field visit is required to provision the loop, BellSouth will test the loop ordered by NewSouth to the NID. Testing requested by NewSouth to points beyond the NID will be billed a time and material charge at the same increments BellSouth charges it's own end users. Requests for field-testing where a dispatch is not required may be made by NewSouth and where mutually agreed to, BellSouth will dispatch to perform additional field testing at rates billed on a time and material basis as mentioned in the previous paragraph.
- 2.1.24.3 BellSouth will place a tag on all unbundled loops that require a technician to be dispatched to the end user's premises during the provisioning process. The loop tag will include the CLEC's name and the circuit ID number. Otherwise, the loop will be tagged by BellSouth during the next scheduled maintenance or repair visit to the customer's location for that loop; or the loop may be tagged by the CLEC during their dispatch to that customer's location.
- 2.1.24.4 Cut-over intervals for ILNP, ILNP with loop and LNP with loop will be at parity with the intervals experienced by BellSouth end users, BellSouth itself or any other NewSouth as indicated in the results of the Service Quality Measurements published by BellSouth. In any event, BellSouth will use best efforts to convert each loop within fifteen (15) minutes.
- 2.1.24.5 BellSouth and NewSouth will jointly develop additional processes or procedures as the need arises to improve service delivery during the life of the agreement.
- 2.2 Unbundled Loop Modifications (ULM) / Loop Conditioning
- 2.2.1 Subject to applicable and effective FCC rules and orders, BellSouth shall condition loops, as requested by NewSouth, whether or not BellSouth offers advanced services to the End User on that loop.
- 2.2.2 Loop conditioning is defined as the removal from the loop of any devices that may diminish the capability of the loop to deliver high-speed switched wireline telecommunications capability, including xDSL service. Such devices include, but are not limited to, bridge taps, load coils, repeaters, low pass filters, and range extenders.
- 2.2.3 BellSouth shall recover the cost of line conditioning requested by NewSouth through a recurring charge and/or nonrecurring charge(s) in accordance with the FCC's forward-looking pricing principles promulgated pursuant to section 252 (d) (1) of the Act and in compliance with FCC Rule 52.507 (e).
- 2.3. Integrated Digital Loop Carriers

Where BellSouth uses Integrated Digital Loop Carrier (IDLC) systems to provide the local loop and BellSouth has a suitable alternate facility available, BellSouth will make arrangements to permit NewSouth to order a contiguous local loop. To the extent it is technically feasible, these arrangements will provide NewSouth with the capability to serve end users at a level that is at parity with the level of service BellSouth provides its customers. If no alternate facility is available, BellSouth will utilize its Special Construction (SC) process to determine the additional costs required to provision the loop facilities. NewSouth will then have the option of paying the one-time SC rates to place the loop facilities or NewSouth may chose some other method of providing service to the end-user (e.g., Resale, private facilities, etc.).

2.4 Network Interface Device

2.4.1 Definition

The NID is defined as any means of interconnection of end-user customer premises wiring to BellSouth's distribution plant, such as a cross-connect device used for that purpose. The NID is a single-line termination device or that portion of a multiple-line termination device required to terminate a single line or circuit at the premises. The NID features two independent chambers or divisions that separate the service provider's network from the End User's on-premises wiring. Each chamber or division contains the appropriate connection points or posts to which the service provider and the End User each make their connections. The NID provides a protective ground connection and is capable of terminating cables such as twisted pair cable.

- 2.4.2. BellSouth shall permit NewSouth to connect NewSouth's loop facilities to onpremises wiring through the BellSouth NID or at any other technically feasible point.
- 2.4.3 Access to Network Interface Device (NID)
- 2.4.3.1. Due to the wide variety of NIDs utilized by BellSouth (based on subscriber size and environmental considerations), NewSouth may access the on-premises wiring by any of the following means: BellSouth shall allow NewSouth to connect its loops directly to BellSouth's multi-line residential NID enclosures that have additional space and are not used by BellSouth or any other telecommunications carriers to provide service to the premise. NewSouth agrees to install compatible protectors and test jacks and to maintain the protection system and equipment and to indemnify BellSouth pursuant to Section 8 of the General Terms and Conditions of this Agreement.
- 2.4.3.2. Where an adequate length of on-premises wiring is present and environmental conditions permit, either Party may remove the on-premises wiring from the other Party's NID and connect that wire to that Party's own NID; or
- 2.4.3.3. Enter the subscriber access chamber or "side" of "dual chamber" NID enclosures for the purpose of extending a connecterized or spliced jumper wire from the on-premises wiring through a suitable "punch-out" hole of such NID enclosures; or

- 2.4.3.4. Request BellSouth to make other rearrangements to the on-premises wiring terminations or terminal enclosure on a time and materials cost basis to be charged to the requesting Party (i.e., NewSouth, its agent, the building owner or the subscriber). Such charges will be billed to the requesting Party.
- 2.4.3.5. In no case shall either Party remove or disconnect the other Party's loop facilities from either Party's NIDs, enclosures, or protectors, without state regulatory requirement, without providing prior notice to the other Party, and without appropriately capping off and guarding the other Party's loop. In such cases, it shall be the responsibility of the disconnecting party to properly ground the other party's loop, maintain the NID, and assume full liability for its action and any adverse consequences.
- 2.4.3.6. In no case shall either Party remove or disconnect ground wires from BellSouth's NIDs, enclosures, or protectors.
- 2.4.3.7. In no case shall either Party remove or disconnect NID modules, protectors, or terminals from BellSouth's NID enclosures.
- 2.4.3.8. Due to the wide variety of NID enclosures and outside plant environments BellSouth will work with NewSouth to develop specific procedures to establish the most effective means of implementing this Section 2.4.3.
- 2.4.4 <u>Technical Requirements</u>
- 2.4.4.1 The NID shall provide an accessible point of interconnection and shall maintain a connection to ground.
- 2.4.4.2 The NID shall be capable of transferring electrical analog or digital signals between the subscriber's inside wiring and the Distribution Media and/or cross connect to NewSouth's NID, consistent with the NID's function at the Effective Date of this Agreement.
- 2.4.4.3 When NewSouth orders a NID at a particular location, NewSouth will get the NID that is currently located at such location. If NewSouth requires additional work or modification to the NID then BellSouth shall do such additional work or modification to the NID in accordance with Section 2.4.3.
- 2.4.4.4 When NewSouth deploys its own local loops with respect to multiple-line termination devices, NewSouth shall specify the quantity of NIDs connections that it requires within such device.
- 2.4.5 <u>Interface Requirements</u>
- 2.4.5.1 The NID shall be equal to or better than all of the requirements for NIDs set forth in the applicable industry standard technical references.

2.5 Unbundled Loop Concentration (ULC) System

- 2.5.1 BellSouth will provide to NewSouth Unbundled Loop Concentration (ULC). Loop concentration systems in the central office concentrate the signals transmitted over local loops onto a digital loop carrier system. The concentration device is placed inside a BellSouth central office. BellSouth will offer ULC with a TR008 interface or a TR303 interface.
- ULC will be offered in two sizes. System A will allow up to 96 BellSouth loops to be concentrated onto multiple DS1s. The high-speed connection from the concentrator will be at the electrical DS1 level and may connect to NewSouth at NewSouth's collocation site. System B will allow up to 192 BellSouth loops to be concentrated onto multiple DS1s. System A may be upgraded to a System B. A minimum of two DS1s is required for each system (i.e., System A requires two DS1s and System B would require an additional two DS1s or four in total). All DS1 interfaces will terminate to the CLEC's collocation space. ULC service is offered with or without concentration and with or without protection. A Line Interface element will be required for each loop that is terminated onto the ULC system. Rates for ULC are as set forth in this Attachment.

2.6 Sub-loop Elements

- Where facilities permit and subject to applicable and effective FCC rules and orders, BellSouth shall offer access to its Unbundled Sub Loop (USL), Unbundled Subloop Concentration (USLC) System and Unbundled Network Terminating Wire (UNTW) elements. BellSouth shall provide non-discriminatory access on a unbundled basis to the subloop, in accordance with 47 C.F.R. 51.311 and section 251(c) (3) of the Act, and pursuant to the following terms and conditions and the rates approved by the Commission and set forth in this Attachment.
- 2.6.2 Subloop components include but are not limited to the following:
- 2.6.2.1 Unbundled Sub-Loop Distribution;
- 2.6.2.2 Unbundled Sub-Loop Concentration/Multiplexing Functionality; and
- 2.6.2.3 Unbundled Network Terminating Wire; and
- 2.6.2.4 Unbundled Sub-Loop Feeder.
- 2.6.3 Unbundled Sub-Loop (distribution facilities)
- 2.6.3.1 Definition

- 2.6.3.2 Subject to applicable and effective FCC rules and orders, the unbundled sub-loop distribution facility is dedicated transmission facility that BellSouth provides from a customer's point of demarcation to a BellSouth cross-connect device. The BellSouth cross-connect device may be located within a remote terminal (RT) or a stand-alone cross-box in the field or in the equipment room of a building. There are two offerings available for Unbundled Sub-Loops (USL):
- 2.6.3.3 Unbundled Sub-Loop Distribution (USL-D) will include the sub-loop facility from the cross-box in the field up to and including the point of demarcation.
- 2.6.3.4 BellSouth will also provide sub-loop interconnection to the intrabuilding network cable (INC) (riser cable). INC is the distribution facility inside a subscriber's building or between buildings on one customer's same premises (continuous property not separated by a public street or road). USL-INC (riser cable) will include the facility from the cross-connect device in the building equipment room up to and including the point of demarcation.
- 2.6.4. Requirements for Unbundled Sub-Loop Distribution Facilities
- 2.6.4.1 Unbundled Sub-Loop distribution facilities were originally built as part of the entire voice grade loop from the BellSouth central office to the customer network interface. Therefore, the Unbundled Sub-Loop may have load coils, which are necessary for transmission of voice grade services. The Unbundled Sub-Loops will be provided in accordance with technical reference TR73600. However, NewSouth may request such sub-loop be conditioned in accordance with Section 2.2 of this Attachment.
- 2.6.4.2 Unbundled Sub-Loop distribution facilities shall support functions associated with provisioning, maintenance and testing of the Unbundled Sub-Loop. In a scenario that involves connection at a BellSouth cross-box located in the field, NewSouth would be required to deliver a cable to the BellSouth remote terminal or cross-box to provide continuity to NewSouth's feeder facilities. This cable would be connected, by a BellSouth technician, to a cross-connect panel within the BellSouth RT/cross-box. NewSouth's cable pairs can then be connected to BellSouth's USL within the BellSouth cross-box by the BellSouth technician. In a scenario that requires connection in a building equipment room, BellSouth will install a cross connect panel on which access to the requested sub-loops will be connected. The CLEC's cable pairs can then be connected to the Unbundled Sub-Loop pairs on this cross-connect panel by the BellSouth technician.
- 2.6.4.3 BellSouth will provide Unbundled Sub-Loops where technically feasible. Through the firm order Service Inquiry (SI) process, BellSouth will determine if it is feasible to place the required facilities where NewSouth has requested access to Unbundled Sub-Loops. If existing capacity is sufficient to meet the CLEC demand, then BellSouth will perform the set-up work as described in the next section 2.6.4.4. If any work must be done to modify existing BellSouth facilities or add new facilities (other than

adding the cross-connect panel in a building equipment room as noted in 2.6.4.4) to accommodate NewSouth's request for Unbundled Sub-Loops, BellSouth will use its Special Construction (SC) process to determine the additional costs required to provision the Unbundled Sub-Loops. NewSouth will then have the option of paying the one-time SC charge to modify the facilities to meet NewSouth's request.

- 2.6.4.4 During the initial set-up in a BellSouth cross-connect box in the field, the BellSouth technician will perform the necessary work to splice the CLEC's cable into the cross-connect box. For the set-up inside a building equipment room, BellSouth will perform the necessary work to install the cross-connect panel that will be used to provide access to the requested USLs. Once the set-up is complete, the CLEC requested sub-loop pairs would be provisioned through the service order process based on the submission of a LSR to the LCSC.
- 2.6.5 <u>Interface Requirements</u>
- 2.6.5.1 Unbundled Sub-Loop shall be equal to or better than each of the applicable requirements set forth in the applicable industry standard technical references.
- 2.6.6 Unbundled Sub-Loop Concentration System (USLC)
- 2.6.6.1 Where facilities permit, BellSouth will provide to NewSouth with the ability to concentrate its sub-loops onto multiple DS1s back to the BellSouth Central Office. The DS1s will then be terminated into NewSouth's collocation space. TR-008 and TR303 interface standards are available.
- USLC, using the Lucent Series 5 equipment, will be offered in two different systems. System A will allow up to 96 of NewSouth's sub-loops to be concentrated onto multiple DS1s. System B will allow an additional 96 of NewSouth's sub-loops to be concentrated onto multiple DS1s. One System A may be supplemented with one System B and they both must be physically located in a single Series 5 dual channel bank. A minimum of two DS1s is required for each system (i.e., System A requires two DS1s and System B would require an additional two DS1s or four in total). The DS1 level facility that connects the RT site with the serving wire center is known as a Feeder Interface. All DS1 Feeder Interfaces will terminate to the CLEC's collocation space within the SWC that serves the RT where the CLEC's sub-loops are connected. USLC service is offered with or without concentration and with or without a protection DS1.
- In these scenarios NewSouth would be required to place a cross-box, remote terminal (RT), or other similar device and deliver a cable to the BellSouth remote terminal. This cable would be connected, by a BellSouth technician, to a cross-connect panel within the BellSouth RT/cross-box and would allow NewSouth's sub-loops to then be placed on the ULSC and transported to their collocation space at a DS1 level.

2.6.7 Unbundled Network Terminating Wire (UNTW)

2.6.7.1 BellSouth agrees to offer its Unbundled Network Terminating Wire (UNTW) to NewSouth pursuant to the following terms and conditions at rates as set forth in this Attachment.

2.6.7.2 Definition

2.6.7.2.1 Subject to applicable and effective FCC rules and orders, UNTW is a dedicated transmission facility that BellSouth provides from the Wiring Closet /Garden Terminal (or other type of cross-connect point) at the point of termination of BellSouth's loop distribution facilities to the end user's point of demarcation.

2.6.7.3 Requirements

- 2.6.7.3.1 BellSouth will offer spare pairs that are available to an end user's premises to NewSouth. Available spare pairs are defined as pairs that are not being utilized by BellSouth or by a third party to provide an end user with working service at the time of NewSouth's request for UNTW. If no spare pairs are available and the end user is no longer using BellSouth's local service, BellSouth will relinquish the first pair to NewSouth. If after BellSouth has relinquished the first pair to NewSouth and the end user decides to change local service providers to BellSouth, NewSouth will relinquish the first pair back to BellSouth.
- 2.6.7.3.2 Notwithstanding the foregoing, should BellSouth subsequently require the use of additional pair(s) to provide for the activation of additional lines in an end users premises in response to a request from such end user, NewSouth agrees to surrender their spare pair(s) upon request by BellSouth.
- 2.6.7.3.3 If an end user of NewSouth desires to receive local exchange service from a service provider who is not a Party to this Agreement, and such third party service provider needs access to the BellSouth UNTW to provide local exchange service to the end user, then NewSouth agrees to surrender the requisite number of its inactive spare pair(s) if no other spare pair is available and upon request by BellSouth.
- 2.6.7.3.4 If NewSouth has placed NTW at a location and an end user desires to receive local exchange service from BellSouth and BellSouth needs access to NewSouth's NTW to provide local exchange service to the end user, then NewSouth agrees to make available the requisite number of its spare pair(s) upon request by BellSouth, at rates, terms and conditions to be negotiated by the Parties.
- 2.6.7.3.5 In new construction, where possible, both Parties may at their option and with the property owner's agreement install their own NTW. In existing construction, BellSouth shall not be required to install new or additional NTW beyond existing NTW to provision the services of NewSouth.

2.6.8 <u>Technical Requirements</u>

2.6.8.1 In these scenarios, BellSouth will connect the requested UNTW pairs to a single point of interconnection (SPOI) designed for NewSouth access to BellSouth's NTW. The SPOI will be installed either near BellSouth's garden terminal or wiring closet. NewSouth will be required to place a cross-box, terminal or other similar device and deliver a cable to this SPOI. NewSouth will then connect their cable to the cross-connect panel to access the requested UNTW pairs.

2.7 Dark Fiber

2.7.1 <u>Definition</u>

Dark Fiber is optical transmission facilities without attached multiplexing, aggregation or other electronics that connects two points within BellSouth's network. Dark Fiber is unused strands of optical fiber. It may be strands of optical fiber existing in aerial or underground structure. No line terminating elements terminated to such strands to operationalize its transmission capabilities will be available. No regeneration or optical amplification will be included with this element.

2.7.2 Requirements

- 2.7.2.1 BellSouth shall make available in a reasonable and non-discriminatory manner, Dark Fiber where it exists in BellSouth's network and where, as a result of future building or deployment, it becomes available. If BellSouth has bona fide plans to use the fiber within a two year planning period, there is no requirement to provide said fiber to NewSouth. BellSouth shall provide access to Dark Fiber at any technically feasible point.
- 2.7.2.2 If the requested dark fiber has any lightwave repeater equipment interspliced to it, BellSouth will remove such equipment at NewSouth's request subject to time and materials charges.
- 2.7.2.3 NewSouth may test the quality of the Dark Fiber to confirm its usability and performance specifications.
- 2.7.2.4 BellSouth shall use its best efforts to provide to NewSouth information regarding the location, availability and performance of Dark Fiber within ten (10) business days for a records based answer and twenty (20) calendar days for a field based answer, after receiving a request from NewSouth ("Request"). Such request shall not be denied based on the fact that designated locations are not BellSouth end-offices or NewSouth's collocation space. Within such time period, BellSouth shall send written confirmation of availability of the Dark Fiber ("Confirmation"). From the time of the Request to forty-five (45) days after Confirmation, BellSouth shall hold such requested Dark Fiber for NewSouth's use and may not allow any other party to use such media, including BellSouth.
- 2.7.2.5 BellSouth shall use its best efforts to make Dark Fiber available to NewSouth within thirty (30) business days after it receives written confirmation from NewSouth that the Dark Fiber previously deemed available by BellSouth is wanted for use by NewSouth.

This includes identification of appropriate connection points (e.g., Light Guide Interconnection (LGX) or splice points) to enable NewSouth to connect or splice NewSouth provided transmission media (e.g., optical fiber) or equipment to the Dark Fiber.

- 2.7.2.6 Dark Fiber shall meet the manufacturer's design specifications.
- 2.7.2.7 NewSouth may splice and test Dark Fiber obtained from BellSouth using NewSouth or NewSouth designated personnel. BellSouth shall provide appropriate interfaces to allow splicing and testing of Dark Fiber. BellSouth shall provide an excess cable length of 25 feet minimum (for fiber in underground conduit) to allow the uncoiled fiber to reach from the manhole to a splicing van.
- 2.8 Preordering Loop Makeup (LMU)
- 2.8.1 <u>Description of Service</u>
- 2.8.1.1 BellSouth shall make available to NewSouth loop makeup (LMU) data for BellSouth's network facilities. This section addresses LMU as a preordering transaction, distinct from NewSouth ordering any other service(s). Loop Makeup Service Inquiries (LMUSI) for preordering loop makeup are likewise unique from other preordering functions with associated service inquiries (SI) as described in this Agreement.
- 2.8.1.2 BellSouth will provide NewSouth with loop makeup information consisting of the composition of the loop material (copper/fiber); the existence, location and type of equipment on the loop, including but not limited to digital loop carrier or other remote concentration devises, feeder/distribution interfaces, bridge taps, load coils, pair-gain devices; the loop length; and the wire gauge. The LMUSI may be utilized by NewSouth for the purpose of determining whether the loop requested is capable of supporting DSL service or other advanced data services. The determination shall be made solely by NewSouth and BellSouth shall not be liable in any way for the performance of the advanced data services provisioned over said loop.
- 2.8.1.3 BellSouth's LMU information is provided to NewSouth as it exists either in BellSouth's databases or in its hard copy facility records. BellSouth does not guarantee accuracy or reliability of the LMU information provided.
- 2.8.1.4 BellSouth offers LMU information for the sole purpose of allowing NewSouth to determine whether, in NewSouth's judgment, BellSouth's loops will support the specific services that NewSouth wishes to provide over those loops. NewSouth may choose to use equipment that it deems will enable it to provide a certain type and level of service over a particular BellSouth loop; however, such configurations may not match BellSouth's or the industry's standards and specifications for the intended type

and level of service. Accordingly, NewSouth shall be responsible for insuring that the specific loop type (ADSL, HDSL, or otherwise) ordered on the LSR matches the LMU of the facility requested. NewSouth bears full responsibility for being knowledgeable of BellSouth's technical standards and the specifications of BellSouth's loops. NewSouth bears full responsibility for making the appropriate ordering decisions of matching BellSouth loops with NewSouth's equipment for accomplishing NewSouth's end goal for the intended service it wishes to provide its end-user(s). NewSouth is fully responsible for any of its service configurations that may differ from BellSouth's technical standard for the loop type ordered.

2.8.2 <u>Submitting Loop Makeup Service Inquiries</u>

- 2.8.2.1 NewSouth will be able to obtain LMU information by submitting a LMUSI mechanically or manually. Mechanized LMUSIs should be submitted through BellSouth's Operational Support Systems interfaces. After obtaining the resulting loop data from the mechanized LMUSI process, if NewSouth determines that it needs further loop data information in order to make a determination of loop service capability, NewSouth may initiate a separate manual LMUSI for a separate nonrecurring charge as set forth in Exhibit C hereto.
- 2.8.2.2 Manual LMUSIs shall be submitted on the preordering manual LMUSI form by means of fax or electronic-mail to BellSouth's Complex Resale Support Group (CRSG)/Account Team utilizing the Preordering Loop Makeup Service Inquiry form. The standard service interval for the return of a Loop Makeup Manual Service Inquiry is seven business days. This service interval is distinct from the interval applied to the subsequent service order. Manual LMUSIs are not subject to expedite requests.

2.8.3 <u>LMUSI Types & Associated Charges</u>

NewSouth may request LMU information by submitting LMUSIs in accordance with the rate elements in Exhibit C. LMU information is available for "working" loops (i.e., those loops that are currently serving a particular end user) and "spare" loops (i.e., those loops that are available to serve a particular end user but are not currently in service). NewSouth may request LMU information for up to three (3) spare facilities per Manual LMUSI and ten (10) spare facilities per Mechanized LMUSI. NewSouth shall submit its loop criteria when placing the LMUSI, and the response shall provide NewSouth with information of each loop (up to the total number of facilities queried) that meet the criteria specified by NewSouth.

- 2.8.3.1 NewSouth will be assessed a nonrecurring charge for each facility queried as specified in Exhibit C. Rates for all states are interim and subject to true-up pending approval of final rates by the respective State Commissions. True-ups will be retroactive to the effective date of this Agreement.
- 2.8.3.2 NewSouth may reserve spare facilities for up to four (4) days in connection with a LMUSI. Reservations are not available for working facilities. Reserved facilities for

- which NewSouth does not plan to place a UNE local service request (LSR) should be cancelled by NewSouth. Should NewSouth wish to cancel a reservation on a spare facility, the cancellation will require a facility reservation number (RESID/FRN).
- 2.8.3.3 The reservation holding timeframe is a maximum of four days from the time that BellSouth's LMU data is returned to NewSouth for the facility queried. During this holding time and prior to NewSouth's placing an LSR, the reserved facilities are rendered unavailable to other customers, whether for CLEC(s) or for BellSouth. Notwithstanding the foregoing, if multiple loops meet NewSouth's criteria as specified in the LMUSI and NewSouth does not order all of such loops, NewSouth shall not be entitled to specify which of the loops contained in the query response BellSouth will actually provision to complete NewSouth's order.
- 2.8.3.4 If NewSouth does not submit an LSR for a UNE service order on a reserved facility within the four-day reservation timeframe, the reservation of that spare facility will become invalid and the facility will be released.
- 2.8.3.5 Charges for preordering LMUSI are separate from any charges associated with ordering other services from BellSouth.
- 2.8.4 Ordering of Other UNE Services
- 2.8.4.1 Whenever NewSouth has reserved a facility through BellSouth's preordering LMU service, should NewSouth seek to place a subsequent UNE LSR on a reserved facility, NewSouth shall provide BellSouth the RESID/FRN of the single spare facility on the appropriate UNE LSR., NewSouth will be billed the appropriate rate element for the specific type UNE loop ordered by NewSouth as set forth in this Attachment. NewSouth will not be billed any additional Loop Makeup charges for the loop so ordered. Should NewSouth choose to place a UNE LSR having previously submitted a request for preordering LMU without a reservation, NewSouth will be billed the appropriate rate element for the specific UNE loop ordered as well as additional Loop Makeup charges as set forth in this Attachment. Rates are provided in the UNE Rate Exhibits for Attachment 2.
- Where NewSouth submits an LSR to order facilities reserved during the LMUSI process, BellSouth will use its best efforts to assign to NewSouth the facility reserved as indicated on the return of the LMU. Multi-facility reservations per single RESID/FRN as provided with the mechanized LMUSI process are less likely to result in the specific assignment requested by NewSouth. For those occasions when BellSouth's assignment system cannot assign the specific facility reserved by NewSouth during the LMU pre-ordering transaction, BellSouth will assign to NewSouth, subject to availability, a facility that meets the BellSouth technical standards of the BellSouth type loop as ordered by NewSouth. If the ordered loop type is not available, NewSouth may utilize the Unbundled Loop Modification process or the Special Construction process, as applicable, to obtain the loop type ordered.

2.8.4.3 BellSouth offers LMU information for the sole purpose of allowing NewSouth to determine whether, in CLEC's judgment, BellSouth's loops will support the specific services that NewSouth wishes to provide over those loops. NewSouth may choose to use equipment that it deems will enable it to provide a certain type and level of service over a particular BellSouth loop; however, such configurations may not match BellSouth's or the industry's standards and specifications for the intended type and level of service. Accordingly, NewSouth shall be responsible for insuring that the specific loop type (ADSL, HDSL, or otherwise) ordered on the LSR matches the LMU of the facility requested. NewSouth bears full responsibility for being knowledgeable of BellSouth's technical standards and the specifications of BellSouth's loops. NewSouth bears full responsibility for making the appropriate ordering decisions of matching BellSouth loops with NewSouth's equipment for accomplishing NewSouth's end goal for the intended service it wishes to provide its end-user(s). NewSouth is fully responsible for any of its service configurations that may differ from BellSouth's technical standard for the loop type ordered.

3. Switching

All of the negotiated rates, terms and conditions set forth in this Section pertain to the provision of local and tandem switching.

3.1 Local Switching

BellSouth shall provide non-discriminatory access to local circuit switching capability, and local tandem switching capability, on an unbundled basis, except as set forth below in Section 3.1.3 to NewSouth for the provision of a telecommunications service. BellSouth shall provide non-discriminatory access to packet switching capability on an unbundled basis to NewSouth for the provision of a telecommunications service only in the limited circumstance described below in Section 3.5.6

- 3.1.1. Except as otherwise provided herein, BellSouth shall not impose any restrictions on NewSouth regarding the use of Switching Capabilities purchased from BellSouth provided such use does not result in demonstrable harm to either the BellSouth network or personnel or the use of the BellSouth network by BellSouth or any other telecommunication carrier.
- 3.1.2. Local Circuit Switching Capability, including Tandem Switching Capability

3.1.2.1 Definition

Local Circuit Switching Capability is defined as: (A) line-side facilities, which include, but are not limited to, the connection between a loop termination at a main distribution frame and a switch line card; (B) trunk-side facilities, which include, but are not limited to, the connection between trunk termination at a trunk-side cross-connect panel and a switch trunk card; and (C) All features, functions, and

capabilities of the switch, which include, but are not limited to: (1) the basic switching function of connecting lines to lines, line to trunks, trunks to lines, and trunks to trunks, as well as the same basic capabilities made available to BellSouth's customers, such as a telephone number, white page listings, and dial tone; and (2) all other features that the switch is capable of providing, including but not limited to customer calling, customer local area signaling service features, and Centrex, as well as any technically feasible customized routing functions provided by the switch; (D) switching provided by remote switching modules.

- 3.1.2.2 When utilizing BellSouth's local circuit switching capability, local traffic shall be defined as set forth in Attachment 3 of this Agreement.
- Notwithstanding BellSouth's general duty to unbundle local circuit switching, BellSouth shall not be required to unbundle local circuit switching for NewSouth when NewSouth serves end-users with four (4) or more voice-grade (DS-0) equivalents or lines in locations served by BellSouth's local circuit switches, which are in the following MSAs: Atlanta, GA; Miami, FL; Orlando, FL; Ft. Lauderdale, FL; Charlotte-Gastonia-Rock Hill, NC; Greensboro-Winston Salem-High Point, NC; Nashville, TN; and New Orleans, LA, and BellSouth has provided non-discriminatory cost based access to the Enhanced Extended Link (EEL) throughout Density Zone I as determined by NECA Tariff No. 4 as in effect on January 1, 1999.
- In the event that NewSouth orders local circuit switching for a single end user account name at a single physical end user location with four (4) or more two (2) wire voice-grade loops from a BellSouth central office in Density Zone 1 as determined by NECA Tariff No. 4 as in effect on January 1, 1999 that is also located in one of the MSAs listed above in Section 3.1.3, BellSouth shall charge NewSouth the market-based switching rates set forth in Exhibit C or, to the extent a particular rate is not set forth in Exhibit C, the parties shall negotiate such rate for use of the local circuit switching functionality for the affected facilities.
- A featureless port is one that has a line port, switching facilities, and an interoffice port. A featured port is a port that includes all features then capable or a number of then capable features specifically requested by NewSouth. Any features that are not currently then capable but are technically feasible through the switch can be requested through the BFR process.
- 3.1.6 BellSouth will provide to NewSouth customized routing of calls: (i) to a requested directory assistance services platform; (ii) to an operator services platform pursuant to Section 10 of Attachment 2; (iii) for NewSouth's PIC'ed toll traffic in a two (2) PIC environment to an alternative OS/DA platform designated by NewSouth. NewSouth customers may use the same dialing arrangements as BellSouth customers.
- 3.1.7 Remote Switching Module functionality is included in Switching Capability. The switching capabilities used will be based on the line side features they support.

- 3.1.8 Switching Capability will also be capable of routing local, intraLATA, interLATA, and calls to international customer's preferred carrier; call features (e.g. call forwarding) and Centrex capabilities.
- 3.1.9 Where required to do so in order to comply with an effective Commission order, BellSouth will provide to NewSouth purchasing local BellSouth switching and reselling BellSouth local exchange service under Attachment 1, selective routing of calls to a requested directory assistance services platform or operator services platform. NewSouth customers may use the same dialing arrangements as BellSouth customers, but obtain a NewSouth branded service.

3.2 <u>Technical Requirements</u>

- 3.2.1 The requirements set forth in this Section apply to Local Switching, but not to the Data Switching function of Local Switching.
- 3.2.1.1 Local Switching shall be equal to or better than the requirements for Local Switching set forth in the applicable industry standard technical references.
- 3.2.1.2 When applicable, BellSouth shall route calls to the appropriate trunk or lines for call origination or termination.
- 3.2.1.3 Subject to this section, BellSouth shall route calls on a per line or per screening class basis to (1) BellSouth platforms providing Network Elements or additional requirements (2) Operator Services platforms, (3) Directory Assistance platforms, and (4) Repair Centers. Any other routing requests by NewSouth will be made pursuant to the Bona Fide Request/ New Business Request Process as set forth in General Terms and Conditions.
- 3.2.1.4 BellSouth shall provide unbranded recorded announcements and call progress tones to alert callers of call progress and disposition.
- 3.2.1.5 BellSouth shall activate service for an NewSouth customer or network interconnection on any of the Local Switching interfaces. This includes provisioning changes to change a customer from BellSouth's services to NewSouth's services without loss of switch feature functionality as defined in this Agreement.
- 3.2.1.6 BellSouth shall perform routine testing (e.g., Mechanized Loop Tests (MLT) and test calls such as 105, 107 and 108 type calls) and fault isolation on a mutually agreed upon schedule.
- 3.2.1.7 BellSouth shall repair and restore any equipment or any other maintainable component that may adversely impact Local Switching.

- 3.2.1.8 BellSouth shall control congestion points such as those caused by radio station callins, and network routing abnormalities. All traffic shall be restricted in a non-discriminatory manner.
- 3.2.1.9 BellSouth shall perform manual call trace and permit customer originated call trace.
- 3.2.1.10 Special Services provided by BellSouth will include the following:
- 3.2.1.10.1 Telephone Service Prioritization;
- 3.2.1.10.2 Related services for handicapped;
- 3.2.1.10.3 Soft dial tone where required by law; and
- 3.2.1.10.4 Any other service required by law.
- 3.2.1.11 BellSouth shall provide Switching Service Point (SSP) capabilities and signaling software to interconnect the signaling links destined to the Signaling Transfer Point Switch (STPS). These capabilities shall adhere to the technical specifications set forth in the applicable industry standard technical references.
- 3.2.1.12 BellSouth shall provide interfaces to adjuncts through Telcordia (formerly BellCore) standard interfaces. These adjuncts can include, but are not limited to, the Service Circuit Node and Automatic Call Distributors.
- 3.2.1.13 BellSouth shall provide performance data regarding a customer line, traffic characteristics or other measurable elements to NewSouth, upon a reasonable request from NewSouth. CLEC will pay BellSouth for all costs incurred to provide such performance data through the Business Opportunity Request process.
- 3.2.1.14 BellSouth shall offer Local Switching that provides feature offerings at parity to those provided by BellSouth to itself or any other Party. Such feature offerings shall include but are not limited to:
- 3.2.1.14.1 Basic and primary rate ISDN;
- 3.2.1.14.2 Residential features;
- 3.2.1.14.3 Customer Local Area Signaling Services (CLASS/LASS);
- 3.2.1.14.4 CENTREX (including equivalent administrative capabilities, such as customer accessible reconfiguration and detailed message recording); and
- 3.2.1.14.5 Advanced intelligent network triggers supporting NewSouth and BellSouth service applications.

3.2.2 BellSouth shall offer to NewSouth all AIN triggers in connection with its SMS/SCE offering which are supported by BellSouth for offering AIN-based services. Triggers that are currently available are: 3.2.2.1 Off-Hook Immediate 3.2.2.2 Off-Hook Delay 3.2.2.3 Termination Attempt 3.2.2.4 6/10 Public Office Dialing Plan 3.2.2.5 Feature Code Dialing 3.2.2.6 Customer Dialing Plan 3.2.3 When the following triggers are supported by BellSouth, BellSouth will make these triggers available to NewSouth: 3.2.3.1 Private EAMF Trunk 3.2.3.2 Shared Interoffice Trunk (EAMF, SS7) 3.2.3.3 N11 3.2.3.4 Automatic Route Selection 3.2.3.5 9XX Blocking 3.2.3.6 Toll Blocking Where capacity exists, BellSouth shall assign each NewSouth customer line the class 3.2.4 of service designated by NewSouth (e.g., using line class codes or other switch specific provisioning methods), and shall route directory assistance calls from NewSouth customers to NewSouth directory assistance operators at NewSouth's option. 3.2.5 Where capacity exists, BellSouth shall assign each NewSouth customer line the class of services designated by NewSouth (e.g., using line class codes or other switch specific provisioning methods) and shall route operator calls from NewSouth customers to NewSouth operators at NewSouth's option. For example, BellSouth may translate 0- and 0+ intraLATA traffic, and route the call through appropriate trunks to an NewSouth Operator Services Position System (OSPS). Calls from Local Switching must pass the ANI-II digits unchanged.

Local Switching shall be offered in accordance with the technical specifications set

forth in the applicable industry standard references.

3.2.6

- 3.2.7 <u>Interface Requirements</u>
 3.2.7.1 BellSouth shall provide the following interfaces to loops:
- 3.2.7.1.1 Standard Tip/Ring interface including loopstart or groundstart, on-hook signaling (e.g., for calling number, calling name and message waiting lamp);
- 3.2.7.1.2 Coin phone signaling;
- 3.2.7.1.3 Basic Rate Interface ISDN adhering to appropriate Telcordia (formerly BellCore) Technical Requirements;
- 3.2.7.1.4 Two-wire analog interface to PBX;
- 3.2.7.1.5 Four-wire analog interface to PBX;
- 3.2.7.1.6 Four-wire DS1 interface to PBX or customer provided equipment (e.g. computers and voice response systems);
- 3.2.7.1.7 Primary Rate ISDN to PBX adhering to ANSI standards Q.931, Q.932 and appropriate Telcordia (formerly BellCore) Technical Requirements;
- 3.2.7.1.8 Switched Fractional DS1 with capabilities to configure Nx64 channels (where N=1 to 24); and
- 3.2.7.1.9 Loops adhering to Telcordia (formerly BellCore) TR-NWT-08 and TR-NWT-303 specifications to interconnect Digital Loop Carriers.
- 3.2.7.2 BellSouth shall provide access to the following but not limited to:
- 3.2.7.2.1 SS7 Signaling Network or Multi-Frequency trunking if requested by NewSouth;
- 3.2.7.2.2 Interface to NewSouth operator services systems or Operator Services through appropriate trunk interconnections for the system; and
- 3.2.7.2.3 Interface to NewSouth Directory Assistance Services through the NewSouth switched network or to Directory Assistance Services through the appropriate trunk interconnections for the system; and 950 access or other NewSouth required access to interexchange carriers as requested through appropriate trunk interfaces.
- 3.3 Tandem Switching
- 3.3.1 <u>Definition</u>

Tandem Switching is the function that establishes a communications path between two switching offices through a third switching office (the Tandem switch).

3.3.2 <u>Technical Requirements</u>

- 3.3.2.1 Tandem Switching shall have the same capabilities or equivalent capabilities as those described in Bell Communications Research TR-TSY-000540 Issue 2R2, Tandem Supplement, 6/1/90. The requirements for Tandem Switching include, but are not limited to the following:
- 3.3.2.1.1 Tandem Switching shall provide signaling to establish a tandem connection;
- 3.3.2.1.2 Tandem Switching will provide screening as jointly agreed to by NewSouth and BellSouth;
- 3.3.2.1.3 Tandem Switching shall provide Advanced Intelligent Network triggers supporting AIN features where such routing is not available from the originating end office switch, to the extent such Tandem switch has such capability;
- 3.3.2.1.4 Tandem Switching shall provide access to Toll Free number portability database as designated by NewSouth;
- 3.3.2.1.5 Tandem Switching shall provide all trunk interconnections discussed under the "Network Interconnection" section (e.g., SS7, MF, DTMF, DialPulse, PRI-ISDN, DID, and CAMA-ANI (if appropriate for 911));
- 3.3.2.1.5.1 Tandem Switching shall provide connectivity to PSAPs where 911 solutions are deployed and the tandem is used for 911; and
- 3.3.2.1.5.2 Where appropriate, Tandem Switching shall provide connectivity to transit traffic to and from other carriers.
- 3.3.2.1.6 Tandem Switching shall accept connections (including the necessary signaling and trunking interconnections) between end offices, other tandems, IXCs, ICOs, CAPs and CLEC switches.
- 3.3.2.1.7 Tandem Switching shall provide local tandeming functionality between two end offices including two offices belonging to different CLEC's (e.g., between a CLEC end office and the end office of another CLEC).
- 3.3.2.1.8 Tandem Switching shall preserve CLASS/LASS features and Caller ID as traffic is processed.
- 3.3.2.1.9 Tandem Switching shall record billable events and send them to the area billing centers designated by NewSouth. Tandem Switching will provide recording of all billable events as jointly agreed to by NewSouth and BellSouth.
- 3.3.2.1.10 Upon a reasonable request from NewSouth, BellSouth shall perform routine testing and fault isolation on the underlying switch that is providing Tandem Switching and all its interconnections. The results and reports of the testing shall be made immediately available to NewSouth.

- 3.3.2.1.11 BellSouth shall maintain NewSouth's trunks and interconnections associated with Tandem Switching at least at parity to its own trunks and interconnections.
- 3.3.2.1.12 BellSouth shall control congestion points and network abnormalities. All traffic will be restricted in a non-discriminatory manner.
- 3.3.2.1.13 Selective Call Routing through the use of line class codes is not available through the use of tandem switching. Selective Call Routing through the use of line class codes is an end office capability only. Detailed primary and overflow routing plans for all interfaces available within BellSouth's switching network shall be mutually agreed to by NewSouth and BellSouth.
- 3.3.2.1.14 Tandem Switching shall process originating toll-free traffic received from NewSouth's local switch.
- 3.3.2.1.15 In support of AIN triggers and features, Tandem Switching shall provide SSP capabilities when these capabilities are not available from the Local Switching Network Element, to the extent such Tandem Switch has such capability.
- 3.3.2.2 Interface Requirements
- 3.3.2.2.1 Tandem Switching shall provide interconnection to the E911 PSAP where the underlying Tandem is acting as the E911 Tandem.
- 3.3.2.2.2 Tandem Switching shall interconnect, with direct trunks, to all carriers with which BellSouth interconnects.
- 3.3.2.2.3 BellSouth shall provide all signaling necessary to provide Tandem Switching with no loss of feature functionality.
- 3.3.2.2.4 Tandem Switching shall interconnect with NewSouth's switch, using two-way trunks, for traffic that is transiting via BellSouth's network to interLATA or intraLATA carriers. At NewSouth's request, Tandem Switching shall record and keep records of traffic for billing.
- 3.3.2.2.5 Tandem Switching shall provide an alternate final routing pattern for NewSouth's traffic overflowing from direct end office high usage trunk groups.
- 3.3.2.2.6 Tandem Switching shall be equal or better than the requirements for Tandem Switching set forth in the applicable technical references.
- 3.4 AIN Selective Carrier Routing for Operator Services, Directory Assistance and Repair Centers

- 3.4.1 BellSouth will provide AIN Selective Carrier Routing at the request of NewSouth. AIN Selective Carrier Routing will provide NewSouth with the capability of routing operator calls, 0+ and 0- and 0+ NPA (LNPA) 555-1212 directory assistance, 1+411 directory assistance and 611 repair center calls to pre-selected destinations.
- 3.4.2 NewSouth shall order AIN Selective Carrier Routing through its Account Team. AIN Selective Carrier Routing must first be established regionally and then on a per central office, per state basis.
- 3.4.3 AIN Selective Carrier Routing is not available in DMS 10 switches.
- Where AIN Selective Carrier Routing is utilized by NewSouth, the routing of NewSouth's end user calls shall be pursuant to information provided by NewSouth and stored in BellSouth's AIN Selective Carrier Routing Service Control Point database. AIN Selective Carrier Routing shall utilize a set of Line Class Codes (LCCs) unique to a basic class of service assigned on an "as needed" basis. The same LCCs will be assigned in each central office where AIN Selective Carrier Routing is established.
- Upon ordering of AIN Selective Carrier Routing Regional Service, NewSouth shall remit to BellSouth the Regional Service Order non-recurring charges set forth in Exhibit A of this Attachment. There shall be a non-recurring End Office Establishment Charge per office due at the addition of each central office where AIN Selective Carrier Routing will be utilized. Said non-recurring charge shall be as set forth in Exhibit A of this Attachment. For each NewSouth end user activated, there shall be a non-recurring End User Establishment charge as set forth in Exhibit A of this Attachment, payable to BellSouth pursuant to the terms of the General Terms and Conditions, incorporated herein by this reference. NewSouth shall pay the AIN Selective Carrier Routing Per Query Charge set forth in Exhibit A of this Attachment.
- 3.4.6 This Regional Service Order non-recurring charge will be non-refundable and will be paid with 1/2 coming up-front with the submission of all fully completed required forms, including: Regional Selective Carrier Routing (SCR) Order Request-Form A, Central Office AIN Selective Carrier Routing (SCR) Order Request Form B, AIN_SCR Central Office Identification Form Form C, AIN_SCR Routing Options Selection Form Form D, and Routing Combinations Table Form E. BellSouth has 30 days to respond to the client's fully completed firm order as a Regional Service Order. With the delivery of this firm order response to the client, BellSouth considers that the delivery schedule of this service commences. The remaining 1/2 of the Regional Service Order payment must be paid when at least 100% of the Central Offices listed on the original order have been turned up for the service.
- 3.4.7 The non-recurring End Office Establishment Charge will be billed to the client following our normal monthly billing cycle for this type of order.

- 3.4.8 End-User Establishment Orders will not be turned-up until the 2nd payment is received for the Regional Service Order. The non-recurring End-User Establishment Charges will be billed to the client following our normal monthly billing cycle for this type of order.
- 3.4.9 Additionally, the AIN Selective Carrier Routing Per Query Charge will be billed to the client following the normal billing cycle for per query charges.
- 3.4.10 All other network components needed, for example, unbundled switching and unbundled local transport, etc, will be billed according per contracted rates.

3.5 Packet Switching Capability

3.5.1 <u>Definition</u>

Packet Switching Capability. The packet switching capability network element is defined as the basic packet switching function of routing or forwarding packets, frames, cells or other data units based on address or other routing information contained in the packets, frames, cells or other data units, and the functions that are performed by Digital Subscriber Line Access Mulitplexers, including but not limited to:

- 3.5.2 The ability to terminate copper customer loops (which includes both a low band voice channel and a high-band data channel, or solely a data channel);
- 3.5.3 The ability to forward the voice channels, if present, to a circuit switch or multiple circuit switches;
- 3.5.4 The ability to extract data units from the data channels on the loops, and
- 3.5.5 The ability to combine data units from multiple loops onto one or more trunks connecting to a packet switch or packet switches.
- 3.5.6 BellSouth shall be required to provide non-discriminatory access to unbundled packet switching capability only where each of the following conditions are satisfied:
- 3.5.6.1 BellSouth has deployed digital loop carrier systems, including but not limited to, integrated digital loop carrier or universal digital loop carrier systems; or has deployed any other system in which fiber optic facilities replace copper facilities in the distribution section (e.g., end office to remote terminal, pedestal or environmentally controlled vault);
- 3.5.6.2 There are no spare copper loops capable of supporting the xDSL services NewSouth seeks to offer;
- 3.5.6.3 BellSouth has not permitted NewSouth to deploy a Digital Subscriber Line Access Multiplexer at the remote terminal, pedestal or environmentally controlled vault or other interconnection point, nor has the NewSouth obtained a virtual collocation

arrangement at these subloop interconnection points as defined by 47 C.F.R. § 51.319 (b); and

- 3.5.6.4 BellSouth has deployed packet switching capability for its own use.
- 3.5.7 If there is a dispute as to whether BellSouth must provide Packet Switching, such dispute will be resolved according tot the dispute resolution process set forth in Section 12 of the General Terms and Conditions of this Agreement, incorporated herein by this reference.

4. Enhanced Extended Link (EEL)

- For purposes of this Section, references to "Already Combined" network elements shall mean that such network elements are in fact already combined by BellSouth in the BellSouth network to provide service to a particular end user at a particular location.
- Where necessary to comply with an effective FCC and/or State Commission order, or as otherwise mutually agreed by the Parties, BellSouth shall offer access to loop and transport combinations, also known as the Enhanced Extended Link ("EEL") as defined in Section 4.3 below.
- Subject to Section 4.2.3 below, BellSouth will provide access to the EEL in the combinations set forth in 4.3 following. This offering is intended to provide connectivity from an end user's location through that end user's SWC to NewSouth's POP serving wire center. The circuit must be used for the purpose of provisioning telecommunications services, including telephone exchange service, to NewSouth's end-user customers. Except as provided for in paragraph 22 of the FCC's Supplemental Order Clarification, released June 2, 2000, in CC Docket No. 96-98 ("June 2, 2000 Order"), the EEL will be connected to NewSouth's facilities in NewSouth's collocation space at the POP SWC. NewSouth may purchase BellSouth's access facilities between NewSouth's POP and NewSouth's collocation space at the POP SWC.
- 4.2.3 BellSouth shall provide EEL combinations to NewSouth in the state of Georgia regardless of whether or not such EELs are Already Combined. In all other states, BellSouth shall make available to NewSouth those EEL combinations described in Section 4.3 below only to the extent such combinations are Already Combined.
- 4.2.4 BellSouth will make available EEL combinations to NewSouth in density Zone 1, as defined in 47 C.F.R. 69.123 as of January 1, 1999, in the Miami, Orlando, Fort Lauderdale, Charlotte, New Orleans, Greensboro and Nashville MSAs, regardless of whether or not such EELs are Already Combined.

Additionally, BellSouth shall make available to NewSouth a combination of an unbundled loop and tariffed special access interoffice facilities. To the extent NewSouth will require multiplexing functionality in connection with such combination, BellSouth will provide access to multiplexing within the central office pursuant to the terms, conditions and rates set forth in its Access Services Tariffs. The combination of an unbundled loop and tariffed special access interoffice facilities and any associated tariffed services, including but not limited to multiplexing, shall not be eligible for conversion to UNEs as described in Section 4.5 below. Where multiplexing functionality is required in connection with loop and transport combinations, such multiplexing will be provided at the rates and on the terms set forth in this Agreement.

4.3 <u>EEL Combinations</u>

- 4.3.1 DS1 Interoffice Channel + DS1 Channelization + 2-wire VG Local Loop
- 4.3.2 DS1 Interoffice Channel + DS1 Channelization + 4-wire VG Local Loop
- 4.3.3 DS1 Interoffice Channel + DS1 Channelization + 2-wire ISDN Local Loop
- 4.3.4 DS1 Interoffice Channel + DS1 Channelization + 4-wire 56 kbps Local Loop
- 4.3.5 DS1 Interoffice Channel + DS1 Channelization + 4-wire 64 kbps Local Loop
- 4.3.6 DS1 Interoffice Channel + DS1 Local Loop
- 4.3.7 DS3 Interoffice Channel + DS3 Local Loop
- 4.3.8 STS-1 Interoffice Channel + STS-1 Local Loop
- 4.3.9 DS3 Interoffice Channel + DS3 Channelization + DS1 Local Loop
- 4.3.10 STS-1 Interoffice Channel + DS3 Channelization + DS1 Local Loop
- 4.3.11 2-wire VG Interoffice Channel + 2-wire VG Local Loop
- 4.3.12 4-wire VG Interoffice Channel + 4-wire VG Local Loop
- 4.3.13 4-wire 56 kbps Interoffice Channel + 4-wire 56 kbps Local Loop
- 4.3.14 4-wire 64 kbps Interoffice Channel + 4-wire 64 kbps Local Loop
- 4.4 Other Network Element Combinations

In the state of Georgia, BellSouth shall make available to NewSouth, in accordance with Section 4.6 below: (1) combinations of network elements other than EELs that are Already Combined; and (2) combinations of network elements other than EELs that are not Already Combined but that BellSouth ordinarily combines in its network. In all other states, BellSouth shall make available to NewSouth, in accordance with Section 4.5 below, combinations of network elements other than EELs only to the extent such combinations are Already Combined.

- 4.5 Special Access Service Conversions
- 4.5.1 NewSouth may not convert special access services to combinations of loop and transport network elements, whether or not NewSouth self-provides its entrance facilities (or obtains entrance facilities from a third party), unless NewSouth uses the combination to provide a "significant amount of local exchange service" (as described in Section 4.5.2 below), in addition to exchange access service, to a particular customer. Such conversions of existing special access services pursuant to this section may include facilities within a single density zone (as described in 47 C. F. R. 69.123) or across Density Zones.
- 4.5.1.2 For the purpose of special access conversions under Section 4.5.1, a "significant amount of local exchange service" is as defined in the FCC's June 2, 2000 Order. The Parties agree to incorporate by reference paragraph 22 of the June 2, 2000 Order. When NewSouth requests conversion of special access circuits, NewSouth will self-certify to BellSouth in the manner specified in paragraph 29 of the June 2, 2000 Order that the circuits to be converted qualify for conversion. In addition there may be extraordinary circumstances where NewSouth is providing a significant amount of local exchange service, but does not qualify under any of the three options set forth in paragraph 22 of June 2, 2000 Order, or under a fourth option set forth below in Section 4.5.2. In such case, NewSouth may petition the FCC for a waiver of the local usage options set forth in the June 2, 2000 Order. If a waiver is granted, then upon NewSouth's request the Parties shall amend this Agreement to the extent necessary to incorporate the terms of such waiver for such extraordinary circumstance.
- 4.5.1.3 The recurring charges for such combinations shall be the sum of the recurring charge for the applicable UNE loop and transport segments (including multiplexing, if applicable), as set forth in Exhibit C to this Attachment. The nonrecurring charges for such combinations shall be an amount equal to all applicable conversion charges set forth in Exhibit C to this Attachment for conversion of special access circuits to EELs, plus all applicable nonrecurring cross connect charges (set forth in Attachment 4 to this Agreement) required to connect the facility to NewSouth's collocation arrangement. EELs that terminate in NewSouth collocation arrangements may be connected by NewSouth via cross-connects to BellSouth services used by NewSouth to transport traffic between NewSouth's collocation space and NewSouth's POP.

- Upon request for conversions of up to 15 circuits from special access to EELs, BellSouth shall perform such conversions within seven (7) days from BellSouth's receipt of a valid, error free service order from NewSouth. Requests for conversions of fifteen (15) or more circuits from special access to EELs will be provisioned on a project basis. Except as set forth in Section 4.5.3 below, conversions should not require the special access circuit to be disconnected and reconnected because only the billing information or other administrative information associated with the circuit will change when NewSouth requests a conversion. Submission of a spreadsheet identifying the circuits to be converted shall serve as a substitute for submission of a local service request (LSR), only until such time as the LSR process is modified to accommodate such requests.
- 4.5.1.5 BellSouth may, at its sole expense, and upon thirty (30) days notice to NewSouth, audit NewSouth's records not more than once in any twelve month period, unless an audit finds non-compliance with the local usage options referenced in the June 2, 2000 Order, in order to verify the type of traffic being transmitted over combinations of loop and transport network elements. If, based on its audits, BellSouth concludes that NewSouth is not providing a significant amount of local exchange traffic over the combinations of loop and transport network elements, BellSouth may file a complaint with the appropriate Commission, pursuant to the dispute resolution process set forth in this Agreement. In the event that BellSouth prevails, BellSouth may convert such combinations of loop and transport network elements to special access services and may seek appropriate retroactive reimbursement from NewSouth.
- 4.5.2 In addition to the circumstances under which NewSouth may identify special access circuits that qualify for conversions to EELs (referenced in Section 4.5.1.2 above), NewSouth also shall be entitled to convert special access circuits to unbundled network elements pursuant to the terms of this section 4.5.2 et seq.
- Upon request by NewSouth, BellSouth will convert special access circuits to 4.5.2.1 combinations of an unbundled loop connected to special access transport provided that: (1) the combination terminates to a NewSouth collocation arrangement; and (2) NewSouth certifies, in the manner set forth in Section 4.5.2 above, that at least 75% of the unbundled network element(s) component of the facility is used to provide originating and terminating local voice traffic. The recurring charges for such combinations shall be the sum of the recurring charge for the applicable UNE loop, as set forth in Exhibit C to this Attachment, and all applicable recurring charges for the special access transport facility, as set forth in the BellSouth tariff under which such facilities were ordered. The nonrecurring charges for such combinations shall be an amount equal to all applicable conversion charges set forth in Exhibit C to this Attachment for conversion of special access circuits to EELs, plus the applicable nonrecurring cross connect charges (set forth in Attachment 4 to this Agreement) required to connect the facility to NewSouth's collocation arrangement. Such combinations that terminate in NewSouth collocation arrangements may be connected

- by NewSouth via cross-connects to BellSouth services used by NewSouth to transport traffic between NewSouth's collocation space and NewSouth's POP.
- 4.5.2.2 Upon request from NewSouth to convert special access circuits pursuant to Section 4.5.2, BellSouth shall have the right, upon 10 business days notice, to conduct an audit prior to any such conversion to determine whether the subject facilities meet local usage requirements set forth in Section 4.5.2. An audit conducted pursuant to this Section shall take into account a usage period of the past three (3) consecutive months, and shall be subject to the requirements for audits as set forth in the June 2, 2000 Order, except as expressly modified herein.
- 4.5.3 In consideration of Section 4.5.2.1 above, and subject to Section 4.5.7 below, for those special access circuits identified by NewSouth in writing as of January 19, 2001 as being eligible for conversion pursuant to the terms of this Agreement, BellSouth will provide to NewSouth a credit in an amount equal to three times the difference between the monthly special access rates for such circuits and the monthly rates for the combinations to which those circuits are converted.
- 4.5.3.1 For circuits converted pursuant to one of the three options made available to NewSouth in Section 4.5.1, the credit will be in an amount equal to three times the difference between the monthly special access rates for such circuits and the monthly UNE recurring charges for the loop, transport and multiplexing (if applicable), as set forth in Exhibit C to this Attachment, that, in combination, form an EEL.
- 4.5.3.2 For circuits converted pursuant to the fourth option made available to NewSouth in Section 4.5.2, the credit will be in an amount equal to three times the difference between the monthly special access rates for such circuits and the sum of the monthly UNE recurring charges for the loop, as set forth in Exhibit C to this Attachment, and the monthly recurring charge for the special access transport facility, as set forth in the BellSouth tariff under which such facility was ordered.
- 4.5.3.3 Such credits will be applied to NewSouth's bill within sixty (60) days following execution of this Agreement.
- 4.5.3.4 Within ten (10) days following execution of this Agreement, NewSouth shall certify to BellSouth in writing that the circuits designated as of January 19, 2001 meet significant local use requirements of one of the four conversion options set forth above. Such certification shall include a designation by NewSouth of which of the particular four conversion options specified herein is applicable to each of the individual circuits designated as of January 19, 2001.
- 4.5.3.5 BellSouth shall assign a project management team and designate a project manager to facilitate the timely conversion of special access circuits. BellSouth and NewSouth will participate in a joint implementation meeting within fifteen (15) days following execution of this Agreement, or within 15 days of any subsequent request for

conversion, to establish a schedule for conversion of the identified special access circuits. BellSouth shall complete conversions of all circuits identified by NewSouth as of January 19, 2001 within 3 months of the joint implementation meeting, unless an alternative completion date is agreed to by the Parties. For purposes of conversion of the circuits identified by NewSouth as of January 19, 2001, NewSouth's spreadsheet identifying the circuits to be converted shall serve as a substitute for submission of a local service request (LSR). For subsequent conversion requests pursuant to Sections 4.5.1 and 4.5.2 above, submission of a spreadsheet identifying the circuits to be converted shall serve as a substitute for submission of a local service request (LSR), only until such time as the LSR process is modified to accommodate such requests.

- 4.5.4 For all special access circuits converted under this Agreement, NewSouth shall pay BellSouth any termination charges applicable to the special access circuits converted, as specified in BellSouth's tariffs.
- 4.5.5 The Parties acknowledge that the conversion option described in Section 4.5.2 and the credits offered NewSouth in Section 4.5.3 constitute a reasonable negotiated alternative to those developed by the FCC in the June 2, 2000 Order. However, BellSouth has agreed to the terms of Sections 4.5.2 and 4.5.3 based upon the assumption that the FCC's current rules regarding special access conversions will remain in effect throughout the 2001 calendar year. In the event that the FCC modifies its rules regarding conversion of special access circuits in a manner that is inconsistent with BellSouth's stated position on the issue, then BellSouth cannot realize the value of the alternative option made available to NewSouth hereunder. In the event that the FCC rules regarding special access conversions are modified in the manner described herein with an effective date prior to January 1, 2002, NewSouth will reimburse BellSouth one-seventh of the credits extended to NewSouth under Section 4.5.3 above for each month or portion thereof prior to January 1, 2002, that such modified FCC rules are in effect.
- 4.6 Rates
- 4.6.1 Georgia
- 4.6.1.1 The non-recurring and recurring rates for the EEL Combinations of network elements set forth in 4.3, whether Already Combined or new, are as set forth in this Attachment.
- 4.6.1.2 On an interim basis, for combinations of loop and transport network elements not set forth in Section 4.3, where the elements are not Already Combined but are ordinarily combined in BellSouth's network, the non-recurring and recurring charges for such UNE combinations shall be the sum of the stand-alone non-recurring and recurring charges of the network elements which make up the combination. These interim rates shall be subject to true-up based on the Commission's review of BellSouth's cost studies.

- 4.6.1.3 To the extent that NewSouth seeks to obtain other combinations of network elements that BellSouth ordinarily combines in its network which have not been specifically priced by the Commission when purchased in combined form, NewSouth, at its option, can request that such rates be determined pursuant to the Bona Fide Request/New Business Request (NBR) process set forth in this Agreement.
- 4.6.2 All Other States
- 4.6.2.1 Subject to Section 4.2.3 and 4.4 preceding, all other states, the rates for (1) Already Combined EEL combinations set forth in Section 4.3, and (2) other combinations of network elements that are Already Combined in the network will be the sum of the recurring rates for the individual network elements plus a nonrecurring charge as specified in Exhibit C of this Attachment.
- 4.6.2.2 Rates for new EEL combinations in Density Zone 1 in the Miami, Orlando, Fort Lauderdale, Charlotte, New Orleans, Greensboro and Nashville MSAs shall be as set forth in Exhibit C hereto; provided, however, that to the extent a rate is not established in Exhibit C, the rate shall be the sum of the recurring and nonrecurring charges for the individual network elements as set forth in Exhibit C to this Attachment, unless otherwise established by the Commission.

5. Port/Loop Combinations

- For purposes of this Section, references to "Already Combined" network elements shall mean that such network elements are in fact already combined by BellSouth in the BellSouth network to provide service to a particular end user at a particular location. For purposes of this Section, "soft dial tone" (i.e., where network elements are connected through from the end user premises to the BellSouth end office and no dispatch is required to initiate service) shall be considered "Already Combined".
- At NewSouth's request, BellSouth shall provide access to combinations of port and loop network elements, as set forth in Section 5.5 below, that are Already Combined in BellSouth's network except as specified in Sections 5.2.1 and 5.2.2 below, consistent with the requirements of 47 C.F.R. 315(b) and all applicable FCC and Commission rules and policies.
- 5.2.1 BellSouth shall not provide access to combinations of unbundled port and loop network elements in locations where, pursuant to FCC rules, BellSouth is not required to provide circuit switching as an unbundled network element.
- In accordance with effective and applicable FCC rules, BellSouth shall not provide unbundled circuit switching in density Zone 1, as defined in 47 C.F.R. 69.123 as of January 1, 1999, of the Atlanta, Miami, Orlando, Fort Lauderdale, Charlotte, New Orleans, Greensboro and Nashville MSAs to NewSouth if NewSouth's customer has 4 or more DS0 equivalent lines.

- Combinations of port and loop network elements provide local exchange service for the origination or termination of calls. BellSouth shall make available the following loop and port combinations at the terms and at the rates set forth below:
- In Georgia, BellSouth shall provide to NewSouth combinations of port and loop network elements to NewSouth on an unbundled basis regardless of whether or not such combinations are Currently Combined except in those locations where BellSouth is not required to provide circuit switching, as set forth in Section 5.2.2 above. The rates for such combinations shall be the cost based rates set forth in Exhibit C of this Attachment.
- In all other states, BellSouth shall provide to NewSouth combinations of port and loop network elements on an unbundled basis if such combinations are Currently Combined, except in those locations where BellSouth is not required to provide unbundled circuit switching, as set forth in Sections 5.2.1 and 5.2.2 above. The rates for such combinations shall be the cost based rates set forth in Exhibit C of this Attachment.
- In all states other than Georgia, except in those locations where BellSouth is not required to provide unbundled circuit switching, as set forth in Sections 5.2.1 and 5.2.2, BellSouth shall provide to NewSouth combinations of port and loop network elements that are not Currently Combined. The rates for such combinations shall be negotiated by the Parties.
- In those locations where BellSouth is not required to provide unbundled circuit switching, as set forth in Sections 5.2.1 and 5.2.2, BellSouth shall provide to NewSouth combinations of port and loop network elements whether or not such combinations are Currently Combined. The rates for Currently Combined combinations are the market based rates as set forth in Exhibit C. The rates for not Currently Combined combinations shall be negotiated by the Parties.
- When NewSouth orders loop/port combinations, and identifies to BellSouth the type of telecommunications service it intends to deliver to its end user customer through that combination (e.g., POTS, ISDN), BellSouth will provide the requested elements with all the functionality, and with at least the same quality of performance and operations systems support (ordering, provisioning, maintenance, billing and recording), that BellSouth provides through its own network to its local exchange service customers receiving equivalent service, unless NewSouth requests a lesser or greater quality of performance through the Bona Fide Request process. BellSouth will provide ordering, provisioning and maintenance services, including intervals, at parity with the same services BellSouth provides to its own end users or resold services as measured in Attachment 9 Performance Measures. The intervals that BellSouth provides for its products and services are as set forth in the Products and Services Interval Guide which can be found on the BellSouth Interconnection website at

www.interconnection.BellSouth.com. The Products and Services Interval Guide may be amended from time to time. Any intervals contained in The Products and Service Interval Guide will not be increased unless ordered to do otherwise by the appropriate regulatory or judicial body. BellSouth's intervals begin with the receipt of an error free local service request (LSR). At the time of this interconnection agreement, not all combinations can be ordered electronically. All residence, business, and PBX port loop services can be electronically ordered. BellSouth will provide manual ordering processes for loop port combinations which cannot be electronically processed. BellSouth will provide notice of additional electronic ordering functionality via the Change Control Process.

- 5.4 Rates for Combinations of Loop and Port Network Elements
- Rates for combinations of loop and port network elements, as set forth in Section 5.4, are provided in Exhibit A of this Attachment.
- 5.4.2 Rates for Circuit Switching
- 5.4.2.1 Rates for circuit switching, where BellSouth is not required, pursuant to Sections 5.1.1 and 5.1.2, to provide circuit switching are as set forth in Exhibit A of this Attachment.
- 5.5 <u>Port/Loop Combination Offerings</u>
- 5.5.1 2-wire voice grade port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 2-wire voice grade DID port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 2-wire CENTREX port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 5.5.4. 2-wire ISDN Basic Rate Interface, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 5.5.5 2-wire ISDN Primary Rate Interface, DS1 loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.

- 4-wire DS1 Trunk port, DS1 Loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 4-wire DS1Loop with normal serving wire center channelization interface, 2 wire voice grade ports (PBX), 2-wire DID ports, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.

6. Transport and Dark Fiber

BellSouth shall provide nondiscriminatory access, in accordance with FCC Rule 51.311 and Section 251(c)(3) of the Act, tc interoffice transmission facilities on an unbundled basis to NewSouth for the provision of a telecommunications service. All of the negotiated rates, terms and conditions set forth in this Section pertain to the provision of unbundled transport and dark fiber.

6.1. Transport

6.1.1 <u>Definition of Common (Shared) Transport</u>

Common (Shared) Transport is an interoffice transmission path between two BellSouth end-offices, BellSouth end-office and a local tandem, or between two local tandems. Where BellSouth Network Elements are connected by intra-office wiring, such wiring is provided as a part of the Network Elements and is not Common (Shared) Transport. Common (Shared) Transport consists of BellSouth inter-office transport facilities and is unbundled from local switching.

- 6.1.2 <u>Technical Requirements of Common (Shared) Transport</u>
- 6.1.2.1 Common (Shared) Transport provided on DS1 or VT1.5 circuits, shall, at a minimum, meet the performance, availability, jitter, and delay requirements specified for Central Office to Central Office ("CO to CO") connections in the appropriate industry standards.
- 6.1.2.2 Common (Shared) Transport provided on DS3 circuits, STS-1 circuits, and higher transmission bit rate circuits, shall, at a minimum, meet the performance, availability, jitter, and delay requirements specified for CO to CO connections in the appropriate industry standards.
- 6.1.2.3 BellSouth shall be responsible for the engineering, provisioning, and maintenance of the underlying equipment and facilities that are used to provide Common (Shared) Transport.
- 6.1.2.4 At a minimum, Common (Shared) Transport shall meet all of the requirements set forth in the applicable industry standard technical references.

- 6.2 Interoffice transmission facility network elements include:
- 6.2.1 Dedicated transport, defined as BellSouth's transmission facilities, including all technically feasible capacity-related services including, but not limited to, DS1, DS3 and OCn levels, dedicated to a particular customer or carrier, that provide telecommunications between wire centers or switches owned by BellSouth, or between wire centers and switches owned by BellSouth and NewSouth.
- Dark Fiber transport, defined as BellSouth's optical transmission facilities without attached multiplexing, aggregation or other electronics;
- 6.2.3 Shared transport, defined as transmission facilities shared by more than one carrier, including BellSouth, between end office switches, between end office switches and tandem switches, and between tandem switches, in BellSouth's network.
- 6.2.4 BellSouth shall:
- 6.2.4.1 Provide NewSouth exclusive use of interoffice transmission facilities dedicated to a particular customer or carrier, or shared use of the features, functions, and capabilities of interoffice transmission facilities shared by more than one customer or carrier;
- 6.2.4.2 Provide all technically feasible transmission facilities, features, functions, and capabilities that NewSouth could use to provide telecommunications services;
- 6.2.4.3 Permit, to the extent technically feasible, NewSouth to connect such interoffice facilities to equipment designated by NewSouth, including but not limited to, NewSouth's collocated facilities; and
- 6.2.4.4 Permit, to the extent technically feasible, NewSouth to obtain the functionality provided by BellSouth's digital cross-connect systems in the same manner that BellSouth provides such functionality to interexchange carriers.

6.3 Dedicated Transport

- 6.3.1 Definitions
- 6.3.2 Dedicated Transport is defined as BellSouth transmission facilities dedicated to a particular customer or carrier that provide telecommunications between wire centers owned by BellSouth or requesting telecommunications carriers, or between switches owned by BellSouth or requesting telecommunications carriers.
- 6.3.3 Unbundled Local Channel
- Unbundled Local Channel is the dedicated transmission path between NewSouth's Point of Presence and the BellSouth Serving Wire Center's collocation.
- 6.3.5 Unbundled Interoffice Channel.

- 6.3.6 Unbundled Interoffice Channel is the dedicated transmission path that provides telecommunication between BellSouth's Serving Wire Centers' collocations.
- 6.3.7 BellSouth shall offer Dedicated Transport in each of the following ways:
- 6.3.7.1 As capacity on a shared UNE facility.
- 6.3.7.2 As a circuit (e.g., DS0, DS1, DS3, OCn) dedicated to NewSouth. This circuit shall consist of an Unbundled Local Channel or an Unbundled Interoffice Channel or both.
- 6.3.8 When Dedicated Transport is provided it shall include:
- 6.3.8.1 Transmission equipment such as, line terminating equipment, amplifiers, and regenerators;
- 6.3.8.2 Inter-office transmission facilities such as optical fiber, copper twisted pair, and coaxial cable.
- Rates for Dedicated Transport are listed in this Attachment. For those states that do not contain rates in this Attachment the rates in the applicable State Access Tariff will apply as interim rates. When final rates are developed, these interim rates will be subject to true up, and the Parties will amend the Agreement to reflect the new rates.
- 6.3.10 <u>Technical Requirements</u>
- 6.3.10.1 This Section sets forth technical requirements for all Dedicated Transport.
- 6.3.10.2 When BellSouth provides Dedicated Transport, the entire designated transmission service (e.g., DS0, DS1, DS3, and OCn) shall be dedicated to NewSouth designated traffic.
- 6.3.10.3 BellSouth shall offer Dedicated Transport in all technologies that become available including, but not limited to, (1) DS0, DS1, DS3, and OCn transport services, and (2) SONET at available transmission bit rates.
- 6.3.10.4 For DS1 or VT1.5 circuits, Dedicated Transport shall, at a minimum, meet the performance, availability, jitter, and delay requirements specified for Customer Interface to Central Office ("CI to CO") connections in the appropriate industry standards.
- 6.3.10.5 Where applicable, for DS3, Dedicated Transport shall, at a minimum, meet the performance, availability, jitter, and delay requirements specified for CI to CO connections in the appropriate industry standards.
- 6.3.10.6 SONET, OC-3, OC-12, and OC-48 Dedicated Transport shall, at a minimum meet the performance, availability, jitter, and delay requirements specified for CI to CO connections in the appropriate industry standards.

- 6.3.10.7 BellSouth shall offer the following interface transmission rates for Dedicated Transport:
- 6.3.10.7.1 DS0 Equivalent;
- 6.3.10.7.2 DS1
- 6.3.10.7.3 DS3
- 6.3.10.7.4 OC-3,
- 6.3.10.7.5 OC-12,
- 6.3.10.7.6 OC-48
- 6.3.10.7.6 SDH (Synchronous Digital Hierarchy) Standard interface rates in accordance with International Telecommunications Union (ITU) Recommendation G.707 and Plesiochronous Digital Hierarchy (PDH) rates per ITU Recommendation G.704.
- 6.3.10.7.8 When Dedicated Transport is provided, BellSouth shall design it according to BellSouth's network infrastructure to allow for the termination points specified by NewSouth.
- 6.3.11 At a minimum, Dedicated Transport shall meet each of the requirements set forth in the applicable industry technical references.
- 6.3.11.1 BellSouth Technical References:
- 6.3.11.2 TR-TSY-000191 Alarm Indication Signals Requirements and Objectives, Issue 1, May 1986.
- 6.3.11.3 TR 73501 LightGate®Service Interface and Performance Specifications, Issue D, June 1995.
- 6.3.11.4 TR 73525 MegaLink®Service, MegaLink Channel Service & MegaLink Plus Service Interface and Performance Specifications, Issue C, May 1996.
- 6.4 Unbundled Channelization
- 6.4.1 BellSouth agrees to offer access to Unbundled Channelization when available pursuant to following terms and conditions and at the rates set forth in the Attachment.
- 6.4.2 Definition

- Unbundled Channelization (UC) provides the multiplexing capability that will allow a DS1 (1.544 Mbps) or DS3 (44.736 Mbps) or STS-1 Unbundled Network Element (UNE) or collocation cross-connect to be multiplexed or channelized at a BellSouth central office. This can be accomplished through the use of a stand-alone multiplexer or a digital cross-connect system at the discretion of BellSouth. Once UC has been installed, NewSouth can have channels activated on an as-needed basis by having BellSouth connect lower level UNEs via Central Office Channel Interfaces (COCIs).
- 6.4.3 Channelization capabilities will be as follows:
- 6.4.3.1 DS3 Channelization System: An element that channelizes a DS3 signal into 28 DS1s/STS-1s.
- 6.4.3.2 DS1 Channelization System: An element that channelizes a DS1 signal into 24 DS0s.
- 6.4.3.3 Central Office Channel Interfaces (COCI): Elements that can be activated on a channelization system.
- DS1 Central Office Channel Interface elements can be activated on a DS3 Channelization System.
- Voice Grade and Digital Data Central Office Channel Interfaces can be activated on a DS1 Channelization System.
- 6.4.6 AMI and B8ZS line coding with either Super Frame (SF) and Extended Super Frame (ESF) framing formats will be supported as options.
- 6.4.7 COCI will be billed on the lower level UNE order that is interfacing with the UC arrangement and will have to be compatible with those UNEs.
- 6.4.8 Channelization may be incorporated within dedicated transport or ordered as a standalone capability, which requires either the high or low speed side to be connected to collocation.
- 6.4.9 Technical Requirements
- In order to assure proper operation with BST provided central office multiplexing functionality, the customer's channelization equipment must adhere strictly to form and protocol standards. Separate standards exist for the multiplex channel bank, for voice frequency encoding, for various signaling schemes, and for subrate digital access.
- 6.4.9.2 DS0 to DS1 Channelization

6.4.9.2.1 The DS1 signal must be framed utilizing the framing structure defined in ANSI T1.107, Digital Hierarchy Formats Specifications and ANSI T1.403.02, DS1 Robbed-bit Signaling State Definitions. DS0 to DS1 Channelization requirements are essential the same as defined in BellSouth Technical Reference 73525, MegaLink® Service, MegaLink® Channel Service, MegaLink® Plus Service, and MegaLink® Light Service Interface and Performance Specification.

6.4.9.3 DS1 to DS3 Channelization

6.4.9.3.1 The DS3 signal must be framed utilizing the framing structure define in ANSI T1.107, Digital Hierarchy Formats Specifications. DS1 to DS3 Channelization requirements are essentially the same as defined in BellSouth Technical Reference 73501, LightGate Service Interface and Performance Specifications. The asynchronous M13 multiplex format (combination of M12 and M23 formats) is specified for terminal equipment that multiplexes 28 DS1s into a DS3.

6.4.9.4 DS1 to STS Channelization

6.4.9.4.1 The STS-1 signal must be framed utilizing the framing structure define in ANSI T1.105, Synchronous Optical Network (SONET) – Basic Description Including Multiplex Structure, Rates and Formats and T1.105.02, Synchronous Optical Network (SONET) – Payload Mappings. DS1 to STS Channelization requirements are essentially the same as defined in BellSouth Technical Reference TR 73501, LightGate® Service Interface and Performance Specifications

6.5 Dark Fiber

6.5.1 Definition

Dark Fiber is optical transmission facilities without attached multiplexing, aggregation or other electronics that connects two points within BellSouth's network. Dark Fiber is unused strands of optical fiber. It may be strands of optical fiber existing in aerial or underground structure. No line terminating elements terminated to such strands to operationalize its transmission capabilities will be available. No regeneration or optical amplification will be included with this element.

6.5.3 Requirements

- 6.5.3.1 BellSouth, on a reasonable and non-discriminatory basis, shall make available Dark Fiber where it exists in BellSouth's network and where, as a result of future building or deployment, it becomes available. If BellSouth has bona fide plans to use the fiber within a two-year period, there is no requirement to provide said fiber to NewSouth. BellSouth shall provide access to Dark Fiber at any technically feasible point.
- 6.5.3.2 If the requested dark fiber has any lightwave repeater equipment interspliced to it, BellSouth will remove such equipment at NewSouth's request subject to time and materials charges.

- 6.5.3.3 NewSouth may test the quality of the Dark Fiber to confirm its usability and performance specifications.
- BellSouth shall use its best efforts to provide to NewSouth information regarding the location, availability and performance of Dark Fiber within ten (10) business days for a records based answer and twenty (20) calendar days for a field based answer, after receiving a request from NewSouth ("Request"). Such request shall not be denied based on the fact that designated locations are not BellSouth end-offices or NewSouth's collocation space. Within such time period, BellSouth shall send written confirmation of availability of the Dark Fiber ("Confirmation"). From the time of the Request to forty-five (45) days after Confirmation, BellSouth shall hold such requested Dark Fiber for NewSouth's use an may not allow any other party to use such media, including BellSouth.
- 6.5.3.5 BellSouth shall use its best efforts to make Dark Fiber available to NewSouth within thirty (30) business days after it receives written confirmation from NewSouth that the Dark Fiber previously deemed available by BellSouth is wanted for use by NewSouth. This includes identification of appropriate connection points (e.g., Light Guide Interconnection (LGX) or splice points) to enable NewSouth to connect or splice NewSouth provided transmission media (e.g., optical fiber) or equipment to the Dark Fiber.
- 6.5.3.6 Dark Fiber shall meet the manufacturer's design specifications.
- 6.5.3.7 NewSouth may splice and test Dark Fiber obtained from BellSouth using NewSouth or NewSouth designated personnel. BellSouth shall provide appropriate interfaces to allow splicing and testing of Dark Fiber. BellSouth shall provide an excess cable length of 25 feet minimum (for fiber in underground conduit) to allow the uncoiled fiber to reach from the manhole to a splicing van.
- 7. BellSouth Switched Access (SWA) 8XX Toll Free Dialing Ten Digit Screening Service

All of the negotiated rates, terms and conditions set forth in this Section pertain to the provision of 8XX Access Ten Digit Screening Services.

- 7.1 BellSouth SWA 8XX Toll Free Dialing Ten Digit Screening Service database
- 7.1.1 The BellSouth SWA 8XX Toll Free Dialing Ten Digit Screening Service database (herein known as 8XX SCP) is a SCP that contains customer record information and functionality to provide call-handling instructions for 8XX calls. The 8XX SCP IN software stores data downloaded from the national SMS and provides the routing instructions in response to queries from the SSP or tandem. The BellSouth SWA 8XX Toll Free Dialing Ten Digit Screening Service (herein know as 8XX TFD), utilizes the 8XX SCP to provide identification and routing of the 8XX calls, based on the ten digits dialed. 8XX TFD is provided with or without POTS number delivery,

dialing number delivery, and other optional complex features as selected by NewSouth. BellSouth shall provide 8XX TFD in accordance with the following:

7.1.2 <u>Technical Requirements</u>

- 7.1.2.1 BellSouth shall provide NewSouth with access to the 8XX record information located in the 8XX SCP. The 8XX SCP contains current records as received from the national SMS and will provide for routing 8XX originating calls based on the dialed ten digit 8XX number.
- 7.1.2.2 The 8XX SCP is designated to receive and respond to queries using the American National Standard Specification of Signaling System Seven (SS7) protocol. The 8XX SCP shall determine the carrier identification based on all ten digits of the dialed number and route calls to the carrier, POTS number, dialing number and/or other optional feature selected by NewSouth.
- 7.1.2.3 The SCP shall also provide, at NewSouth's option, such additional feature as described in SR-TSV-002275 (BOC Notes on BellSouth Networks, SR-TSV-002275, Issue 2, (Telcordia (formerly BellCore), April 1994)) as are available to BellSouth.

 These may include but are not limited to:
- 7.1.2.3.1 Network Management;
- 7.1.2.3.2 Customer Sample Collection; and
- 7.1.2.3.3 Service Maintenance.

8 Line Information Database (LIDB)

- 8.1 All of the negotiated rates, terms and conditions set forth in this Section pertain to the provision of LIDB.
- 8.2 BellSouth will store in its LIDB only records relating to service in the BellSouth region. The LIDB Storage Agreement is included in this Attachment.
- 8.2.1 Definition
- 8.2.2 The Line Information Database (LIDB) is a transaction-oriented database accessible through Common Channel Signaling (CCS) networks. It contains records associated with end user Line Numbers and Special Billing Numbers. LIDB accepts queries from other Network Elements and provides appropriate responses. The query originator need not be the owner of LIDB data. LIDB queries include functions such as screening billed numbers that provides the ability to accept Collect or Third Number Billing calls and validation of Telephone Line Number based non-proprietary calling cards. The interface for the LIDB functionality is the interface between

BellSouth's CCS network and other CCS networks. LIDB also interfaces to administrative systems.

- 8.2.3 <u>Technical Requirements</u>
- 8.2.4 BellSouth will offer to NewSouth any additional capabilities that are developed for LIDB during the life of this Agreement.
- 8.2.4.1 BellSouth shall process NewSouth's Customer records in LIDB at least at parity with BellSouth customer records, with respect to other LIDB functions. BellSouth shall indicate to NewSouth what additional functions (if any) are performed by LIDB in the BellSouth network.
- 8.2.4.2 Within two (2) weeks after a request by NewSouth, BellSouth shall provide NewSouth with a list of the customer data items, which NewSouth would have to provide in order to support each required LIDB function. The list shall indicate which data items are essential to LIDB function, and which are required only to support certain services. For each data item, the list shall show the data formats, the acceptable values of the data item and the meaning of those values.
- 8.2.4.3 BellSouth shall provide LIDB systems for which operating deficiencies that would result in calls being blocked shall not exceed 30 minutes per year.
- 8.2.4.4 BellSouth shall provide LIDB systems for which operating deficiencies that would not result in calls being blocked shall not exceed 12 hours per year.
- 8.2.4.5 BellSouth shall provide LIDB systems for which the LIDB function shall be in overload no more than 12 hours per year.
- 8.2.4.6 All additions, updates and deletions of NewSouth data to the LIDB shall be solely at the direction of NewSouth. Such direction from NewSouth will not be required where the addition, update or deletion is necessary to perform standard fraud control measures (e.g., calling card auto-deactivation).
- 8.2.4.7 BellSouth shall provide priority updates to LIDB for NewSouth data upon NewSouth's request (e.g., to support fraud detection), via password-protected telephone card, facsimile, or electronic mail within one hour of notice from the established BellSouth contact.
- 8.2.4.8 BellSouth shall provide LIDB systems such that no more than 0.01% of NewSouth customer records will be missing from LIDB, as measured by NewSouth audits. BellSouth will audit NewSouth records in LIDB against DBAS to identify record mismatches and provide this data to a designated NewSouth contact person to resolve the status of the records and BellSouth will update system appropriately. BellSouth will refer record of mis-matches to NewSouth within one business day of audit. Once reconciled records are received back from NewSouth, BellSouth will update LIDB the same business day if less than 500 records are received before 1:00PM Central Time.

- If more than 500 records are received, BellSouth will contact NewSouth to negotiate a time frame for the updates, not to exceed three business days.
- 8.2.4.9 BellSouth shall perform backup and recovery of all of NewSouth's data in LIDB including sending to LIDB all changes made since the date of the most recent backup copy, in at least the same time frame BellSouth performs backup and recovery of BellSouth data in LIDB for itself. Currently, BellSouth performs backups of the LIDB for itself on a weekly basis and when a new software release is scheduled, a backup is performed prior to loading the new release.
- 8.2.4.10 BellSouth shall provide NewSouth with LIDB reports of data, which are missing or contain errors, as well as any misrouted errors, within a reasonable time period as negotiated between NewSouth and BellSouth.
- 8.2.4.11 BellSouth shall prevent any access to or use of NewSouth data in LIDB by BellSouth personnel that are outside of established administrative and fraud control personnel, or by any other Party that is not authorized by NewSouth in writing.
- 8.2.4.12 BellSouth shall provide NewSouth performance of the LIDB Data Screening function, which allows a LIDB to completely or partially deny specific query originators access to LIDB data owned by specific data owners, for Customer Data that is part of an NPA-NXX or RAO-0/1XX wholly or partially owned by NewSouth at least at parity with BellSouth Customer Data. BellSouth shall obtain from NewSouth the screening information associated with LIDB Data Screening of NewSouth data in accordance with this requirement. BellSouth currently does not have LIDB Data Screening capabilities. When such capability is available, BellSouth shall offer it to NewSouth under the Bona Fide Request/New Business Process as set forth in General Terms and Conditions.
- 8.2.4.13 BellSouth shall accept queries to LIDB associated with NewSouth customer records, and shall return responses in accordance with industry standards.
- 8.2.4.14 BellSouth shall provide mean processing time at the LIDB within 0.50 seconds under normal conditions as defined in industry standards.
- 8.2.4.15 BellSouth shall provide processing time at the LIDB within 1 second for 99% of all messages under normal conditions as defined in industry standards.
- 8.2.5 <u>Interface Requirements</u>
- 8.2.6 BellSouth shall offer LIDB in accordance with the requirements of this subsection.
- 8.2.6.1 The interface to LIDB shall be in accordance with the technical references contained within.
- 8.2.6.2 The CCS interface to LIDB shall be the standard interface described herein.

8.2.6.3 The LIDB Data Base interpretation of the ANSI-TCAP messages shall comply with the technical reference herein. Global Title Translation shall be maintained in the signaling network in order to support signaling network routing to the LIDB.

9 Signaling

- 9.1 All of the negotiated rates, terms and conditions set forth in this Section pertain to the provision of Signaling Transport Services.
- 9.2 BellSouth agrees to offer access to signaling and access to BellSouth's signaling databases subject to compatibility testing and at the rates set forth in this Attachment. BellSouth may provide mediated access to BellSouth signaling systems and databases. Available signaling elements include signaling links, signal transfer points and service control points. Signaling functionality will be available with both A-link and B-link connectivity.

9.3 Signaling Link Transport

- 9.3.1 Definition Signaling Link Transport is a set of two or four dedicated 56 Kbps. transmission paths between CLEC-designated Signaling Points of Interconnection (SPOI) that provides appropriate physical diversity.
- 9.3.2 <u>Technical Requirements</u>
- 9.3.2.1 Signaling Link Transport shall consist of full duplex mode 56 kbps transmission paths.
- 9.3.3 Of the various options available, Signaling Link Transport shall perform in the following two ways:
- 9.3.3.1 As an "A-link" which is a connection between a switch or SCP and a home Signaling Transfer Point Switch (STP) pair; and
- 9.3.3.2 As a "B-link" which is a connection between two STP pairs in different company networks (e.g., between two STP pairs for two Competitive Local Exchange Carriers (CLECs)).
- 9.3.4 Signaling Link Transport shall consist of two or more signaling link layers as follows:
- 9.3.4.1 An A-link layer shall consist of two links.
- 9.3.4.2 A B-link layer shall consist of four links.
- 9.3.5 A signaling link layer shall satisfy a performance objective such that:
- 9.3.5.1 There shall be no more than two minutes down time per year for an A-link layer; and

- 9.3.5.2 There shall be negligible (less than 2 seconds) down time per year for a B-link layer.
- 9.3.5.3 A signaling link layer shall satisfy interoffice and intraoffice diversity of facilities and equipment, such that:
- 9.3.5.3.1 No single failure of facilities or equipment causes the failure of both links in an A-link layer (i.e., the links should be provided on a minimum of two separate physical paths end-to-end); and
- 9.3.5.3.2 No two concurrent failures of facilities or equipment shall cause the failure of all four links in a B-link layer (i.e., the links should be provided on a minimum of three separate physical paths end-to-end).
- 9.3.5.4 Interface Requirements
- 9.3.5.4.1 There shall be a DS1 (1.544 Mbps) interface at the NewSouth designated SPOIs. Each 56 kbps transmission path shall appear as a DS0 channel within the DS1 interface.
- 9.4 Signaling Transfer Points (STPs)
- 9.4.1 <u>Definition</u> Signaling Transfer Points is a signaling network function that includes all of the capabilities provided by the signaling transfer point switches (STPs) and their associated signaling links which enable the exchange of SS7 messages among and between switching elements, database elements and signaling transfer point switches.
- 9.4.2 <u>Technical Requirements</u>
- 9.4.2.1 STPs shall provide access to Network Elements connected to BellSouth SS7 network. These include:
- 9.4.2.1.1 BellSouth Local Switching or Tandem Switching:
- 9.4.2.1.2 BellSouth Service Control Points/DataBases;
- 9.4.2.1.3 Third-party local or tandem switching;
- 9.4.2.1.4 Third-party-provided STPs.
- 9.4.2.2 The connectivity provided by STPs shall fully support the functions of all other Network Elements connected to the BellSouth SS7 network. This explicitly includes the use of the BellSouth SS7 network to convey messages which neither originate nor terminate at a signaling end point directly connected to the BellSouth SS7 network (i.e., transient messages). When the BellSouth SS7 network is used to convey transient messages, there shall be no alteration of the Integrated Services Digital Network User Part (ISDNUP) or Transaction Capabilities Application Part (TCAP) user data that constitutes the content of the message.

- 9.4.2.3 If a BellSouth tandem switch routes calling traffic, based on dialed or translated digits, on SS7 trunks between an NewSouth local switch and third party local switch, the BellSouth SS7 network shall convey the TCAP messages that are necessary to provide Call Management features (Automatic Callback, Automatic Recall, and Screening List Editing) between NewSouth local STPs and the STPs that provide connectivity with the third party local switch, even if the third party local switch is not directly connected to BellSouth STPs.
- 9.4.2.4 STPs shall provide all functions of the MTP as defined in the applicable industry standard technical references.
- 9.4.2.5 STPs shall provide all functions of the SCCP necessary for Class 0 (basic connectionless) service, as defined in Telcordia (formerly BellCore) ANSI Interconnection Requirements. In particular, this includes Global Title Translation (GTT) and SCCP Management procedures, as specified in T1.112.4. In cases where the destination signaling point is a NewSouth or third party local or tandem switching system directly connected to BellSouth SS7 network, BellSouth shall perform final GTT of messages to the destination and SCCP Subsystem Management of the destination. In all other cases, BellSouth shall perform intermediate GTT of messages to a gateway pair of STPs in an SS7 network connected with BellSouth SS7 network, and shall not perform SCCP Subsystem Management of the destination. If BellSouth performs final GTT to a NewSouth database, then NewSouth agrees to provide BellSouth with the Destination Point Code for the NewSouth database.
- 9.4.2.6 STPs shall provide on a non-discriminatory basis all functions of the OMAP commonly provided by STPs, as specified in the reference in Section 12.4.5 of this Attachment. All OMAP functions will be on a "where available" basis and can include:
- 9.4.2.6.1 MTP Routing Verification Test (MRVT); and
- 9.4.2.6.2 SCCP Routing Verification Test (SRVT).
- 9.4.2.7 In cases where the destination signaling point is a BellSouth local or tandem switching system or database, or is an NewSouth or third party local or tandem switching system directly connected to the BellSouth SS7 network, STPs shall perform MRVT and SRVT to the destination signaling point. In all other cases, STPs shall perform MRVT and SRVT to a gateway pair of STPs in an SS7 network connected with the BellSouth SS7 network. This requirement shall be superseded by the specifications for Internetwork MRVT and SRVT if and when these become approved ANSI standards and available capabilities of BellSouth STPs, and if mutually agreed upon by NewSouth and BellSouth.
- 9.4.2.8 STPs shall be on parity with BellSouth.
- 9.4.2.9 SS7 Advanced Intelligent Network (AIN) Access

- 9.4.2.9.1 When technically feasible and upon request by NewSouth, SS7 Access shall be made available in association with switching. SS7 AIN Access is the provisioning of AIN 0.1 triggers in an equipped BellSouth local switch and interconnection of the BellSouth SS7 network with the NewSouth SS7 network to exchange TCAP queries and responses with an NewSouth SCP.
- 9.4.2.9.2 SS7 AIN Access shall provide NewSouth SCP access to BellSouth local switch in association with switching via interconnection of BellSouth SS7 and NewSouth SS7 Networks. BellSouth shall offer SS7 access through its STPs. If BellSouth requires a mediation device on any part of its network specific to this form of access, BellSouth must route its messages in the same manner. The interconnection arrangement shall result in the BellSouth local switch recognizing the NewSouth SCP as at least at parity with BellSouth's SCP's in terms of interfaces, performance and capabilities.
- 9.4.3 <u>Interface Requirements</u>
- 9.4.3.1 BellSouth shall provide the following STPs options to connect NewSouth or NewSouth-designated local switching systems or STPs to the BellSouth SS7 network:
- 9.4.3.1.1 An A-link interface from NewSouth local switching systems; and,
- 9.4.3.1.2 A B-link interface from NewSouth local STPs.
- 9.4.3.2 Each type of interface shall be provided by one or more sets (layers) of signaling links.
- 9.4.3.3 The Signaling Point of Interconnection (SPOI) for each link shall be located at a cross-connect element, such as a DSX-1, in the Central Office (CO) where BellSouth STP is located. There shall be a DS1 or higher rate transport interface at each of the SPOIs. Each signaling link shall appear as a DS0 channel within the DS1 or higher rate interface. BellSouth shall offer higher rate DS1 signaling for interconnecting NewSouth local switching systems or STPs with BellSouth STPs as soon as these become approved ANSI standards and available capabilities of BellSouth STPs. BellSouth and NewSouth will work jointly to establish mutually acceptable SPOIs.
- 9.4.3.4 BellSouth CO shall provide intraoffice diversity between the SPOIs and BellSouth STPs, so that no single failure of intraoffice facilities or equipment shall cause the failure of both B-links in a layer connecting to a BellSouth STP. BellSouth and NewSouth will work jointly to establish mutually acceptable SPOIs.
- 9.4.3.5 STPs shall provide all functions of the MTP as defined in the applicable industry standard technical references.
- 9.4.3.6 Message Screening

- 9.4.3.6.1 BellSouth shall set message screening parameters so as to accept valid messages from NewSouth local or tandem switching systems destined to any signaling point within BellSouth's SS7 network where the NewSouth switching system has a legitimate signaling relation.
- 9.4.3.6.2 BellSouth shall set message screening parameters so as to pass valid messages from NewSouth local or tandem switching systems destined to any signaling point or network accessed through BellSouth's SS7 network where the NewSouth switching system has a legitimate signaling relation.
- 9.4.3.6.3 BellSouth shall set message screening parameters so as to accept and pass/send valid messages destined to and from NewSouth from any signaling point or network interconnected through BellSouth's SS7 network where the NewSouth SCP has a legitimate signaling relation.
- 9.4.4 STPs shall be equal to or better than all of the requirements for STPs set forth in the applicable industry standard technical references.

9.5 Service Control Points/Databases

9.5.1 <u>Definition</u>

- 9.5.1.1 Databases are the Network Elements that provide the functionality for storage of, access to, and manipulation of information required to offer a particular service and/or capability. Databases include, but are not limited to: Local Number Portability, LIDB, Toll Free Number Database, Automatic Location Identification/Data Management System, Calling Name Database, access to Service Creation Environment and Service Management System (SCE/SMS) application databases and Directory Assistance.
- 9.5.2 A Service Control Point (SCP) is a specific type of Database functionality deployed in a Signaling System 7 (SS7) network that executes service application logic in response to SS7 queries sent to it by a switching system also connected to the SS7 network. Service Management Systems provide operational interfaces to allow for provisioning, administration and maintenance of subscriber data and service application data stored in SCPs.

9.5.3 <u>Technical Requirements for SCPs/Databases</u>

9.5.3.1 Requirements for SCPs/Databases within this section address storage of information, access to information (e.g. signaling protocols, response times), and administration of information (e.g., provisioning, administration, and maintenance). All SCPs/Databases shall be provided to NewSouth in accordance with the following requirements.

- 9.5.3.2 BellSouth shall provide physical access to SCPs through the SS7 network and protocols with TCAP as the application layer protocol.
- 9.5.3.3 BellSouth shall provide physical interconnection to databases via industry standard interfaces and protocols (e.g. SS7, ISDN and X.25).
- 9.5.3.4 The reliability of interconnection options shall be consistent with requirements for diversity and survivability.

9.5.4 Database Availability

- 9.5.4.1 Call processing databases shall have a maximum unscheduled availability of 30 minutes per year. Unavailability due to software and hardware upgrades shall be scheduled during minimal usage periods and only be undertaken upon proper notification to providers, which might be impacted. Any downtime associated with the provision of call processing related databases will impact all service providers, including BellSouth, equally.
- 9.5.4.2 The operational interface provided by BellSouth shall complete Database transactions (i.e., add, modify, delete) for NewSouth customer records stored in BellSouth databases within 3 days, or sooner where BellSouth provisions its own customer records within a shorter interval.

9.6 Local Number Portability Database

9.6.1 Definition

9.6.2 The Permanent Number Portability (PNP) database supplies routing numbers for calls involving numbers that have been ported from one local service provider to another. PNP is currently being worked in industry forums. The results of these forums will dictate the industry direction of PNP. BellSouth agrees to provide access to the PNP database at rates, terms and conditions as set forth by BellSouth and in accordance with an effective FCC or Commission directive.

9.7 SS7 Network Interconnection

9.7.1 Definition.

9.7.2 SS7 Network Interconnection is the interconnection of NewSouth local Signaling Transfer Point Switches (STP) and NewSouth local or tandem switching systems with BellSouth STPs. This interconnection provides connectivity that enables the exchange of SS7 messages among BellSouth switching systems and databases (DBs), NewSouth local or tandem switching systems, and other third-party switching systems directly connected to the BellSouth SS7 network.

9.7.3 Technical Requirements

- 9.7.3.1 SS7 Network Interconnection shall provide connectivity to all components of the BellSouth SS7 network. These include:
- 9.7.3.1.1 BellSouth local or tandem switching systems;
- 9.7.3.1.2 BellSouth DBs; and
- 9.7.3.1.3 Other third-party local or tandem switching systems.
- 9.7.4 The connectivity provided by SS7 Network Interconnection shall fully support the functions of BellSouth switching systems and DBs and NewSouth or other third-party switching systems with A-link access to the BellSouth SS7 network.
- 9.7.5 If traffic is routed based on dialed or translated digits between an NewSouth local switching system and a BellSouth or other third-party local switching system, either directly or via a BellSouth tandem switching system, then it is a requirement that the BellSouth SS7 network convey via SS7 Network Interconnection the TCAP messages that are necessary to provide Call Management services (Automatic Callback, Automatic Recall, and Screening List Editing) between the NewSouth local STPs and BellSouth or other third-party local switch.
- 9.7.6 When the capability to route messages based on Intermediate Signaling Network Identifier (ISNI) is generally available on BellSouth STPs, the BellSouth SS7 Network shall also convey TCAP messages using SS7 Network Interconnection in similar circumstances where the BellSouth switch routes traffic based on a Carrier Identification Code (CIC).
- 9.7.7 SS7 Network Interconnection shall provide all functions of the MTP as specified in ANSI T1.111. This includes:
- 9.7.7.1 Signaling Data Link functions, as specified in ANSI T1.111.2;
- 9.7.7.2 Signaling Link functions, as specified in ANSI T1.111.3; and
- 9.7.7.3 Signaling Network Management functions, as specified in ANSI T1.111.4.
- 9.7.8 SS7 Network Interconnection shall provide all functions of the SCCP necessary for Class 0 (basic connectionless) service, as specified in ANSI T1.112. In particular, this includes Global Title Translation (GTT) and SCCP Management procedures, as specified in T1.112.4. Where the destination signaling point is a BellSouth switching system or DB, or is another third-party local or tandem switching system directly connected to the BellSouth SS7 network, SS7 Network Interconnection shall include final GTT of messages to the destination and SCCP Subsystem Management of the destination. Where the destination signaling point is an NewSouth local or tandem switching system, SS7 Network Interconnection shall include intermediate GTT of messages to a gateway pair of NewSouth local STPs, and shall not include SCCP Subsystem Management of the destination.

- 9.7.9 SS7 Network Interconnection shall provide all functions of the Integrated Services Digital Network User Part (ISDNUP), as specified in ANSI T1.113.
- 9.7.10 SS7 Network Interconnection shall provide all functions of the TCAP, as specified in ANSI T1.114.
- 9.7.11 If and when Internetwork MTP Routing Verification Test (MRVT) and SCCP Routing Verification Test (SRVT) become approved ANSI standards and available capabilities of BellSouth STPs, SS7 Network Interconnection shall provide these functions of the OMAP.
- 9.7.12 SS7 Network Interconnection shall be equal to or better than the following performance requirements:
- 9.7.12.1 MTP Performance, as specified in ANSI T1.111.6;
- 9.7.12.2 SCCP Performance, as specified in ANSI T1.112.5; and
- 9.7.12.3 ISDNUP Performance, as specified in ANSI T1.113.5.
- 9.7.13 Interface Requirements
- 9.7.13.1 BellSouth shall offer the following SS7 Network Interconnection options to connect NewSouth or NewSouth-designated local or tandem switching systems or STPs to the BellSouth SS7 network:
- 9.7.13.1.1 A-link interface from NewSouth local or tandem switching systems; and
- 9.7.13.1.2 B-link interface from NewSouth STPs.
- 9.7.13.2 The Signaling Point of Interconnection (SPOI) for each link shall be located at a cross-connect element, such as a DSX-1, in the Central Office (CO) where the BellSouth STP is located. There shall be a DS1 or higher rate transport interface at each of the SPOIs. Each signaling link shall appear as a DS0 channel within the DS1 or higher rate interface. BellSouth shall offer higher rate DS1 signaling links for interconnecting NewSouth local switching systems or STPs with BellSouth STPs as soon as these become approved ANSI standards and available capabilities of BellSouth STPs. BellSouth and NewSouth will work jointly to establish mutually acceptable SPOI.
- 9.7.13.3 BellSouth CO shall provide intraoffice diversity between the SPOIs and the BellSouth STP, so that no single failure of intraoffice facilities or equipment shall cause the failure of both B-links in a layer connecting to a BellSouth STP. BellSouth and NewSouth will work jointly to establish mutually acceptable SPOI.

- 9.7.13.4 The protocol interface requirements for SS7 Network Interconnection include the MTP, ISDNUP, SCCP, and TCAP. These protocol interfaces shall conform to the applicable industry standard technical references.
- 9.7.13.5 BellSouth shall set message screening parameters to accept messages from NewSouth local or tandem switching systems destined to any signaling point in the BellSouth SS7 network with which the NewSouth switching system has a legitimate signaling relation.
- 9.7.13.6 SS7 Network Interconnection shall be equal to or better than all of the requirements for SS7 Network Interconnection set forth in the applicable industry standard technical references.
- 10. Operator Call Processing, Inward Operator Services and Directory Assistance Services
- All of the negotiated rates, terms and conditions set forth in this Section pertain to the provision of Operator Call Processing, Inward Operator Services and Directory Assistance Services.
- 10.2 Operator Systems
- 10.2.1 <u>Definition.</u> Operator Systems is the Network Element that provides operator and automated call handling and billing, special services, end user telephone listings and optional call completion services. The Operator Systems, Network Element provides two types of functions: Operator Service functions and Directory Assistance Service functions, each of which are described in detail below.

10.3 Operator Service

- 10.3.1 <u>Definition</u>. Operator Service provides: (1) operator handling for call completion (for example, collect, third number billing, and manual credit card calls), (2) operator or automated assistance for billing after the end user has dialed the called number (for example, credit card calls); and (3) special services including but not limited to Busy Line Verification and Emergency Line Interrupt (ELI), Emergency Agency Call, Operator-assisted Directory Assistance, and Rate Quotes.
- 10.3.2 Requirements
- 10.3.2.1 When NewSouth requests BellSouth to provide Operator Services, the following requirements apply:
- 10.3.2.1.1 BellSouth shall complete 0+ and 0- dialed local calls.
- 10.3.2.1.2 BellSouth shall complete 0+ intraLATA toll calls.

- 10.3.2.1.3 BellSouth shall process calls that are billed to NewSouth end user's calling card that can be validated by BellSouth.
- 10.3.2.1.4 BellSouth shall complete person-to-person calls.
- 10.3.2.1.5 BellSouth shall complete collect calls.
- 10.3.2.1.6 BellSouth shall provide the capability for callers to bill to a third party and complete such calls.
- 10.3.2.1.7 BellSouth shall complete station-to-station calls.
- 10.3.2.1.8 BellSouth shall process emergency calls.
- 10.3.2.1.9 BellSouth shall process Busy Line Verify and Emergency Line Interrupt requests.
- 10.3.2.1.10 BellSouth shall process emergency call trace, as they do for their End users prior to the Effective Date. Call must originate from a 911 provider.
- 10.3.2.1.11 BellSouth shall process operator-assisted directory assistance calls.
- 10.3.2.1.12 BellSouth shall adhere to equal access requirements, providing NewSouth local end users the same IXC access as provided to BellSouth end users.
- 10.3.2.1.13 BellSouth shall exercise at least the same level of fraud control in providing Operator Service to NewSouth that BellSouth provides for its own operator service.
- 10.3.2.1.14 BellSouth shall perform Billed Number Screening when handling Collect, Personto-Person, and Billed-to-Third-Party calls.
- 10.3.2.1.15 BellSouth shall direct customer account and other similar inquiries to the customer service center designated by NewSouth.
- 10.3.2.1.16 BellSouth shall provide a feed of customer call records in "EMI" format to NewSouth in accordance with CLEC ODUF standards specified in Attachment 7.
- 10.3.3 <u>Interface Requirements</u>
- 10.3.3.1 With respect to Operator Services for calls that originate on local switching capability provided by or on behalf of NewSouth, the interface requirements shall conform to the then current established system interface specifications for the platform used to provide Operator Service and the interface shall conform to industry standards.
- 10.4 Directory Assistance Service

10.4.1 <u>Definition.</u> Directory Assistance Service provides local end user telephone number listings with the option to complete the call at the callers direction separate and distinct from local switching.

10.4.2 Requirements

- Directory Assistance Service shall provide up to two listing requests per call. If available and if requested by NewSouth's end user, BellSouth shall provide caller-optional directory assistance call completion service at rates contained in this Attachment to one of the provided listings, equal to that which BellSouth provides its end users. If not available, NewSouth may request such requirement pursuant to the Bona Fide Request/New Business Process as set forth in General Terms and Conditions.
- 10.4.4 Directory Assistance Service Updates
- 10.4.4.1 BellSouth shall update end user listings changes daily. These changes include:
- 10.4.4.1.1 New end user connections: BellSouth will provide service to NewSouth that is equal to the service it provides to itself and its end users;
- 10.4.4.1.2 End user disconnections: BellSouth will provide service to NewSouth that is equal to the service it provides to itself and its end users; and
- 10.4.4.1.3 End user address changes: BellSouth will provide service to NewSouth that is equal to the service it provides to itself and its end users;
- 10.4.4.1.4 These updates shall also be provided for non-listed and non-published numbers for use in emergencies.
- 10.4.5 Branding for Operator Call Processing and Directory Assistance
- 10.4.5.1 The BellSouth Operator Systems Branding Feature provides a definable announcement to NewSouth end users using Directory Assistance (DA)/Operator Call Processing (OCP) prior to placing them in queue or connecting them to an available operator or automated operator system. This feature allows NewSouth to have its calls custom branded with NewSouth's name on whose behalf BellSouth is providing Directory Assistance and/or Operator Call Processing. Rates for Custom Branding, Operator Call Process and Directory Assistance are set forth in this Attachment.
- 10.4.5.2 BellSouth offers four service levels of branding to NewSouth when ordering Directory Assistance and/or Operator Call Processing.
- 10.4.5.2.1 Service Level 1 BellSouth Branding
- 10.4.5.2.2 Service Level 2 Unbranded
- 10.4.5.2.3 Service Level 3 Custom Branding

- 10.4.5.2.4 Service Level 4 Self Branding (applicable only to NewSouth for Resale or use with an Unbundled Port when routing to an operator service provider other than BellSouth).
- 10.4.6 For Resellers and Use with an Unbundled Port
- 10.4.6.1 BellSouth Branding is the Default Service Level.
- 10.4.6.2 Unbranding, Custom Branding, and Self Branding require NewSouth to order selective routing for each originating BellSouth end office identified by NewSouth. Rates for Selective Routing are set forth in this Attachment.
- 10.4.6.3 Customer Branding and Self Branding require NewSouth to order dedicated trunking from each BellSouth end office identified by NewSouth, to either the BellSouth Traffic Operator Position System (TOPS) or NewSouth Operator Service Provider. Rates for trunks are set forth in applicable BellSouth tariffs.
- 10.4.6.4 Unbranding Unbranded Directory Assistance and/or Operator Call Processing calls ride common trunk groups provisioned by BellSouth from those end offices identified by NewSouth to the BellSouth TOPS. These calls are routed to "No Announcement."
- 10.4.7 For Facilities Based Carriers
- 10.4.7.1 All Service Levels require NewSouth to order dedicated trunking from their end office(s) point of interface to the BellSouth TOPS Switches. Rates for trunks are set forth in applicable BellSouth tariffs.
- 10.4.7.2 Customized Branding includes charges for the recording of the branding announcement and the loading of the audio units in each TOPS Switch, IVS and NAV equipment for which NewSouth requires service.
- 10.4.8 Directory Assistance customized branding uses:
- 10.4.8.1 the recording of the name;
- 10.4.8.2 the front-end loading of the Digital Recorded Announcement Machine (DRAM) in each TOPS switch.
- 10.4.9 Operator Call Processing customized branding uses:
- 10.4.9.1 the recording of the name;
- 10.4.9.2 the front-end loading of the DRAM in the TOPS Switch;
- 10.4.9.3 the back-end loading in the audio units in the Automated Alternate Billing System (AABS) in the Interactive Voice Subsystem (IVS);

- 10.4.9.4 the 0- automation loading for the audio units in the Enhanced Billing and Access Service (EBAS) in the Network Applications Vehicle (NAV).
- 10.4.9.5 BellSouth will provide to NewSouth purchasing local BellSouth switching and reselling BellSouth local exchange service, selective routing of calls to a requested directory assistance services platform or operator services platform. NewSouth end users may use the same dialing arrangements as BellSouth end users, but obtain a NewSouth branded service.

10.5 Directory Assistance Database Service (DADS)

- BellSouth shall make its Directory Assistance Database Service (DADS) available solely for the expressed purpose of providing Directory Assistance type services to NewSouth end users. The term "end user" denotes any entity which obtains Directory Assistance type services for its own use from a DADS customer. Directory Assistance type service is defined as Voice Directory Assistance (DA Operator assisted and Electronic Directory Assistance (Data System assisted)). NewSouth agrees that Directory Assistance Database Service (DADS) will not be used for any purpose which violates federal or state laws, statutes, regulatory orders or tariffs. Except for the permitted users, NewSouth agrees not to disclose DADS to others and shall provide due care in providing for the security and confidentiality of DADS. Further, NewSouth authorizes the inclusion of NewSouth Directory Assistance listings in the BellSouth Directory Assistance products.
- BellSouth shall provide NewSouth initially with a base file of subscriber listings which reflect all listing change activity occurring since NewSouth's most recent update via magnetic tape, and subsequently using electronic connectivity such as Network Data Mover to be developed mutually by NewSouth and BellSouth. NewSouth agrees to assume the costs associated with CONNECT: Direct TM connectivity, which will vary depending upon volume and mileage.
- 10.5.3 BellSouth will require approximately one month after receiving an order to prepare the Base File. BellSouth will provide daily updates which will reflect all listing change activity occurring since CLEC's most recent update. BellSouth shall provide updates to NewSouth on a Business, Residence, or combined Business and Residence basis. NewSouth agrees that the updates shall be used solely to keep the information current. Delivery of Daily Updates will commence the day after NewSouth receives the Base File.
- 10.5.4 BellSouth is authorized to include NewSouth Directory Assistance Listing Information in its Directory Assistance Database Service (DADS). Any other use by BellSouth of NewSouth Directory Assistance Listing Information is not authorized and with the exception of a request for DADS, BellSouth shall refer any request for such information to NewSouth.

- 10.5.5 Rates for DADS are as set forth in this Attachment.
- 10.6 Direct Access to Directory Assistance Service
- Direct Access to Directory Assistance Service (DADAS) will provide NewSouth's directory assistance operators with the ability to search all available BellSouth's subscriber listings using the Directory Assistance search format. Subscription to DADAS will allow NewSouth to utilize its own switch, operator workstations and optional audio subsystems.
- 10.6.2 BellSouth will provide DADAS from its DA location. NewSouth will access the DADAS system via a telephone company provided point of availability. NewSouth has the responsibility of providing the physical links required to connect to the point of availability. These facilities may be purchased from the telephone company as rates and charges billed separately from the charges associated with this offering.
- A specified interface to each NewSouth subsystem will be provided by BellSouth. Interconnection between NewSouth's system and a specified BellSouth location will be pursuant to the use of NewSouth owned or NewSouth leased facilities and shall be appropriate sized based upon the volume of queries being generated by NewSouth.
- 10.6:4 The specifications for the three interfaces necessary for interconnection are available in the following documents:
- 10.6.4.1 DADAS to Subscriber Operator Position System—Northern Telecom Document CSI-2300-07; Universal Gateway/ Position Message Interface Format Specification;
- 10.6.4.2 DADAS to Subscriber Switch—Northern Telecom Document Q210-1 Version A107; NTDMS/CCIDAS System Application Protocol; and AT&T Document 250-900-535 Operator Services Position System Listing Service and Application Call Processing Data Link Interface Specification;
- 10.6.4.3 DADAS to Audio Subsystem (Optional)—Directory One Call Control to Audio Response Unit system interface specifications are available through Northern Telecom as a licensed access protocol—Northern Telecom Document 355-004424 and Gateway/Interactive Voice subsystem Protocol Specification.
- 10.6.5 Rates for DADAS are as set forth in this Attachment.
- 10.7 Automatic Location Identification/Data Management System (ALI/DMS)
- 10.7.1 The ALI/DMS Database contains end user information (including name, address, telephone information, and sometimes special information from the local service provider or end user) used to determine to which Public Safety Answering Point (PSAP) to route the call. The ALI/DMS database is used to provide more routing flexibility for E911 calls than Basic 911. BellSouth shall provide the Emergency Services Database in accordance with the following:

10.7.2 <u>Technical Requirements</u>

- 10.7.2.1 BellSouth shall offer NewSouth a data link to the ALI/DMS database or permit NewSouth to provide its own data link to the ALI/DMS database. BellSouth shall provide error reports from the ALI/DMS database to NewSouth immediately after NewSouth inputs information into the ALI/DMS database. Alternately, NewSouth may utilize BellSouth, to enter end user information into the data base on a demand basis, and validate end user information on a demand basis.
- 10.7.2.2 The ALI/DMS database shall contain the following end user information:
- 10.7.2.2.1 Name;
- 10.7.2.2.2 Address;
- 10.7.2.2.3 Telephone number; and
- 10.7.2.2.4 Other information as appropriate (e.g., whether a end user is blind or deaf or has another disability).
- 10.7.2.3 When BellSouth is responsible for administering the ALI/DMS database in its entirety, ported number NXXs entries for the ported numbers should be maintained unless NewSouth requests otherwise and shall be updated if NewSouth requests, provided NewSouth supplies BellSouth with the updates.
- 10.7.2.4 When Remote Call Forwarding (RCF) is used to provide number portability to the local end user and a remark or other appropriate field information is available in the database, the shadow or "forwarded-to" number and an indication that the number is ported shall be added to the customer record.
- 10.7.2.5 If BellSouth is responsible for configuring PSAP features (for cases when the PSAP or BellSouth supports an ISDN interface) it shall ensure that CLASS Automatic Recall (Call Return) is not used to call back to the ported number. Although BellSouth currently does not have ISDN interface, BellSouth agrees to comply with this requirement once ISDN interfaces are in place.
- 10.7.3 Interface Requirements

The interface between the E911 Switch or Tandem and the ALI/DMS database for NewSouth end users shall meet industry standards.

11. Calling Name (CNAM) Database Service

11.1 The Agreement for Calling Name (CNAM) with standard pricing is included as Exhibit B to this Attachment. NewSouth must provide to its account manager a

written request with a requested activation date to activate this service. If NewSouth is interested in requesting CNAM with volume and term pricing, NewSouth must contact its account manager to request a separate CNAM volume and term Agreement. BellSouth provisioning of CNAM shall be compliant with all applicable industry standard technical references.

- SCPs/Databases shall be equal to or better than all of the requirements for SCPs/Databases set forth in the applicable industry standard technical references.
- 11.3 Service Creation Environment and Service Management System (SCE/SMS)
 Advanced Intelligent Network (AIN) Access
- 11.3.1 BellSouth's Service Creation Environment and Service Management System (SCE/SMS) Advanced Intelligent Network (AIN) Access shall provide NewSouth the capability that will allow NewSouth and other third parties to create service applications in a BellSouth Service Creation Environment and deploy those applications in a BellSouth SMS to a BellSouth SCP. The third party service applications interact with AIN triggers provisioned on a BellSouth SSP.
- BellSouth's SCE/SMS AIN Access shall provide access to SCE hardware, software, testing and technical support (e.g., help desk, system administrator) resources available to NewSouth. Scheduling procedures shall provide NewSouth equivalent priority to these resources.
- BellSouth SCP shall partition and protect NewSouth service logic and data from unauthorized access, execution or other types of compromise.
- When NewSouth selects SCE/SMS AIN Access, BellSouth shall provide training, documentation, and technical support to enable NewSouth to use BellSouth's SCE/SMS AIN Access to create and administer applications. Training, documentation, and technical support will address use of SCE and SMS access and administrative functions, but will not include support for the creation of a specific service application.
- When NewSouth selects SCE/SMS AIN Access, BellSouth shall provide for a secure, controlled access environment in association with its internal use of AIN components. NewSouth access will be provided via remote data connection (e.g., dial-in, ISDN).
- When NewSouth selects SCE/SMS AIN Access, BellSouth shall allow NewSouth to download data forms and/or tables to BellSouth SCP via BellSouth SMS without intervention from BellSouth (e.g., service customization and end user subscription).
- 12. Basic 911 and E911

- All of the negotiated terms and conditions set forth in this Section pertain to the provision of Basic 911 and E911.
- 12.2 If NewSouth orders network elements and other services, then NewSouth is also responsible for providing E911 to its end users. BellSouth agrees to offer access to the 911/E911 network pursuant to the following terms and conditions set forth in this Attachment.

12.3 <u>Definition</u>

Basic 911 and E911 is an additional requirement that provides a caller access to the applicable emergency service bureau by dialing a 3-digit universal telephone number (911).

12.5 <u>Requirements</u>

- 12.5.1 Basic 911 Service Provisioning. For Basic 911 service, BellSouth will provide to NewSouth a list consisting of each municipality that subscribes to Basic 911 service. The list will also provide, if known, the E911 conversion date for each municipality and, for network routing purposes, a ten-digit directory number representing the appropriate emergency answering position for each municipality subscribing to 911. NewSouth will be required to arrange to accept 911 calls from its end users in municipalities that subscribe to Basic 911 service and translate the 911 call to the appropriate 10-digit directory number as stated on the list provided by BellSouth. NewSouth will be required to route that call to the appropriate tandem or end office. When a municipality converts to E911 service, NewSouth will be required to discontinue the Basic 911 procedures and being using E911 procedures.
- 12.5.2 E911 Service Provisioning. For E911 service, NewSouth will be required to install a minimum of two dedicated trunks originating from the NewSouth serving wire center and terminating to the appropriate E911 tandem. The dedicated trunks shall be, at a minimum, DS-0 level trunks configured either as a 2-wire analog interface or as part of a digital (1.544 Mb/s) interface. Either configuration shall use CAMA-type signaling with multifrequency ("MF") pulsing that will deliver automatic number identification ("ANI") with the voice portion of the call. If the user interface is digital, MF pulses, as well as other AC signals, shall be encoded per the u-255 Law convention. NewSouth will be required to provide BellSouth or the appropriate designated vendor daily updates to the E911 database. NewSouth will be required to forward 911 calls to the appropriate E911 tandem, along with ANI, based upon the current E911 end office to tandem homing arrangement as provided by BellSouth. If the E911 tandem trunks are not available, NewSouth will be required to route the call to a designated 7-digit local number residing in the appropriate Public Service Answering Point ("PSAP"). This call will be transported over BellSouth's interoffice network and will not carry the ANI of the calling party. NewSouth shall be responsible for providing BellSouth with complete and accurate data for submission to the 911/E911 database for the purpose of providing 911/E911 to its end users.

- 12.5.3 Rates. Charges for 911/E911 service are borne by the municipality purchasing the service. BellSouth will impose no charge on NewSouth beyond applicable charges for BellSouth trunking arrangements.
- 12.5.4 Basic 911 and E911 functions provided to NewSouth shall be at least at parity with the support and services that BellSouth provides to its end users for such similar functionality.
- 12.5.5 Detailed Practices and Procedures. The detailed practices and procedures contained in the E911 Local Exchange Carrier Guide For Facility-Based Providers as amended from time to time during the term of this Agreement will determine the appropriate practices and procedures for BellSouth and NewSouth to follow in providing 911/E911 services.
- 12.5.6 NewSouth shall have access to BellSouth's 911/E911 records database so that it can update its end users' 911 records in order to ensure accuracy. BellSouth will generate the disconnect order for the porting telephone number(s) within 36 hours, excluding weekends and published holidays, after the porting has been activated by NewSouth. The completion of the disconnect order will unlock the 911/E911 record to allow NewSouth to facilitate such updates. BellSouth will not be obligated to meet such interval as described in this paragraph as a result of a failure on NewSouth's part. Such failures shall include: (1) NewSouth's failure to port all numbers on the LSR resulting in an incomplete port; (2) instances where NewSouth activates a port without an FOC; (3) instances of NewSouth porting a number without an FOC, where a LSR was submitted to BellSouth and subsequently clarified to NewSouth for inaccurate or incomplete information; (4) NewSouth stops porting activity for less than all numbers required to be ported and subsequently reinstates porting activity at a later time or date; and (5) failure on New South's part to adhere to NPAC or other applicable industry standard processes for porting telephone numbers.

13 Rates

- The prices that NewSouth shall pay to BellSouth for Network Elements and Other Services are as set forth in Exhibit C to this Attachment. It is the intent of the Parties that where applicable state commissions have approved rates for network elements and other services set forth in this Agreement as of the date of the date hereof, such rates have been included in Exhibit C.
- The prices that NewSouth shall pay to BellSouth for Operational Support Systems are as set forth in General Terms and Conditions of this Agreement.

14. True-Up

This section applies only to Tennessee and other rates that are interim or expressly subject to true-up under this attachment.

- 14.1 The interim prices for Network Elements and Other Services and Local Interconnection shall be subject to true-up according to the following procedures:
- The interim prices shall be trued-up, either up or down, based on final prices determined either by further agreement between the Parties, or by an effective order of the Commission which final order meets the criteria of (3) below. The Parties shall implement the true-up by comparing the actual volumes and demand for each item, together with interim prices for each item, with the final prices determined for each item. Each Party shall keep its own records upon which the true-up can be based, and any final payment from one Party to the other shall be in an amount agreed upon by the Parties based on such records. In the event of any disagreement as between the records or the Parties regarding the amount of such true-up, the Parties agree that the body having jurisdiction over the matter shall be called upon to resolve such differences, or the Parties may mutually agree to submit the matter to the Dispute Resolution process in accordance with the provisions of Section 12 of the General Terms and Conditions.
- The Parties may continue to negotiate toward final prices, but in the event that no such Agreement is reached within nine (9) months, either Party may petition the Commission to resolve such disputes and to determine final prices for each item. Alternatively, upon mutual agreement, the Parties may submit the matter to the Dispute Resolution Process set forth in Section 12 of the General Terms and Conditions.
- A final order of this Commission that forms the basis of a true-up shall be the final order as to prices based on appropriate cost studies, or potentially may be a final order in any other Commission proceeding which meets the following criteria:
 - (a) BellSouth and NewSouth are entitled to be a full Party to the proceeding;
 - (b) It shall apply the provisions of the federal Telecommunications Act of 1996, including but not limited to Section 252(d)(1) (which contains pricing standards) and all then-effective implementing rules and regulations; and,
 - (c) It shall include as an issue the geographic deaveraging of network element and other services prices, which deaveraged prices, if any are required by said final order, shall form the basis of any true-up.

EXHIBIT A

LINE INFORMATION DATA BASE (LIDB) STORAGE AGREEMENT

I. SCOPE

- A. This Agreement sets forth the terms and conditions pursuant to which BellSouth agrees to store in its LIDB certain information at the request of NewSouth and pursuant to which BellSouth, its LIDB customers and NewSouth shall have access to such information. NewSouth understands that BellSouth provides access to information in its LIDB to various telecommunications service providers pursuant to applicable tariffs and agrees that information stored at the request of NewSouth, pursuant to this Agreement, shall be available to those telecommunications service providers. The terms and conditions contained in the attached Addendum(s) are hereby made a part of this Agreement as if fully incorporated herein.
- B. LIDB is accessed for the following purposes:
 - 1. Billed Number Screening
 - 2. Calling Card Validation
 - 3. Fraud Control
- C. BellSouth will provide seven days per week, 24-hours per day, fraud monitoring on Calling Cards, bill-to-third and collect calls made to numbers in BellSouth's LIDB, provided that such information is included in the LIDB query. BellSouth will establish fraud alert thresholds and will notify NewSouth of fraud alerts so that NewSouth may take action it deems appropriate. NewSouth understands and agrees BellSouth will administer all data stored in the LIDB, including the data provided by NewSouth pursuant to this Agreement, in the same manner as BellSouth's data for BellSouth's end user customers. BellSouth shall not be responsible to NewSouth for any lost revenue which may result from BellSouth's administration of the LIDB pursuant to its established practices and procedures as they exist and as they may be changed by BellSouth in its sole discretion from time to time.

NewSouth understands that BellSouth currently has in effect numerous billing and collection agreements with various interexchange carriers and billing clearing houses. NewSouth further understands that these billing and collection customers of BellSouth query BellSouth's LIDB to determine whether to accept various billing options from end users. Additionally, NewSouth understands that presently BellSouth has no method to differentiate between BellSouth's own billing and line data in the LIDB and such data which it includes in the LIDB on NewSouth's behalf pursuant to this Agreement. Therefore, until such time as BellSouth can and does implement in its LIDB and its supporting systems the means to differentiate NewSouth's data from BellSouth's data and the Parties to this Agreement execute appropriate amendments hereto, the following terms and conditions shall apply:

- (a) NewSouth agrees that it will accept responsibility for telecommunications services billed by BellSouth for its billing and collection customers for NewSouth's end user accounts which are resident in LIDB pursuant to this Agreement. NewSouth authorizes BellSouth to place such charges on NewSouth's bill from BellSouth and agrees that it shall pay all such charges. Charges for which NewSouth hereby takes responsibility include, but are not limited to, collect and third number calls.
- (b) Charges for such services shall appear on a separate BellSouth bill page identified with the name of the entity for which BellSouth is billing the charge.
- (c) NewSouth shall have the responsibility to render a billing statement to its end users for these charges, but NewSouth's obligation to pay BellSouth for the charges billed shall be independent of whether NewSouth is able or not to collect from NewSouth's end users.
- (d) BellSouth shall not become involved in any disputes between NewSouth and the entities for which BellSouth performs billing and collection. BellSouth will not issue adjustments for charges billed on behalf of an entity to NewSouth. It shall be the responsibility of NewSouth and the other entity to negotiate and arrange for any appropriate adjustments.

II. TERM

This Agreement will be effective as of ______, and will continue in effect for one year, and thereafter may be continued until terminated by either Party upon thirty (30) days written notice to the other Party.

III. FEES FOR SERVICE AND TAXES

- A. NewSouth will not be charged a fee for storage services provided by BellSouth to NewSouth, as described in Section I of this Agreement.
- B. Sales, use and all other taxes (excluding taxes on BellSouth's income) determined by BellSouth or any taxing authority to be due to any federal, state or local taxing jurisdiction with respect to the provision of the service set forth herein will be paid by NewSouth. NewSouth shall have the right to have BellSouth contest with the imposing jurisdiction, at NewSouth's expense, any such taxes that NewSouth deems are improperly levied.

IV. INDEMNIFICATION

To the extent not prohibited by law, each Party will indemnify the other and hold the other harmless against any loss, cost, claim, injury, or liability relating to or arising out of negligence or willful misconduct by the indemnifying Party or its agents or contractors in connection with the indemnifying Party's provision of services, provided, however, that any indemnity for any loss, cost, claim, injury or liability arising out of or relating to errors or omissions in the provision of services under this Agreement shall be limited as otherwise specified in this Agreement. The indemnifying Party under this Section agrees to defend any suit brought against the other Party for any such loss, cost, claim, injury or liability. The indemnified Party agrees to notify the other Party promptly, in writing, of any written claims, lawsuits, or demands for which the other Party is responsible under this Section and to cooperate in every reasonable way to facilitate defense or settlement of claims. The indemnifying Party shall not be liable under this Section for settlement by the indemnified Party of any claim, lawsuit, or demand unless the defense of the claim, lawsuit, or demand has been tendered to it in writing and the indemnifying Party has unreasonably failed to assume such defense.

V. LIMITATION OF LIABILITY

Except in cases of gross negligence, willful or intentional misconduct, neither Party shall be liable to the other Party for any lost profits or revenues or for any indirect, incidental or consequential damages incurred by the other Party arising from this Agreement or the services performed or not performed hereunder, regardless of the cause of such loss or damage.

VI. MISCELLANEOUS

- A. It is understood and agreed to by the Parties that BellSouth may provide similar services to other companies.
- B. All terms, conditions and operations under this Agreement shall be performed in accordance with, and subject to, all applicable local, state or federal legal and regulatory tariffs, rulings, and other requirements of the federal courts, the U. S. Department of Justice and state and federal regulatory agencies. Nothing in this Agreement shall be construed to cause either Party to violate any such legal or regulatory requirement and either Party's obligation to perform shall be subject to all such requirements.
- C. NewSouth agrees to submit to BellSouth all advertising, sales promotion, press releases, and other publicity matters relating to this Agreement wherein BellSouth's corporate or trade names, logos, trademarks or service marks or those of BellSouth's affiliated companies are mentioned or language from which the connection of said names or trademarks therewith may be inferred or implied; and NewSouth further agrees not to publish or use advertising, sales promotions, press releases, or publicity matters without BellSouth's prior written approval.

- D. This Agreement constitutes the entire Agreement between NewSouth and BellSouth which supersedes all prior Agreements or contracts, oral or written representations, statements, negotiations, understandings, proposals and undertakings with respect to the subject matter hereof.
- E. Except as expressly provided in this Agreement, if any part of this Agreement is held or construed to be invalid or unenforceable, the validity of any other Section of this Agreement shall remain in full force and effect to the extent permissible or appropriate in furtherance of the intent of this Agreement.
- F. Neither Party shall be held liable for any delay or failure in performance of any part of this Agreement for any cause beyond its control and without its fault or negligence, such as acts of God, acts of civil or military authority, government regulations, embargoes, epidemics, war, terrorist acts, riots, insurrections, fires, explosions, earthquakes, nuclear accidents, floods, strikes, power blackouts, volcanic action, other major environmental disturbances, unusually severe weather conditions, inability to secure products or services of other persons or transportation facilities, or acts or omissions of transportation common carriers.
- G. This Agreement shall be deemed to be a contract made under the laws of the State of Georgia, and the construction, interpretation and performance of this Agreement and all transactions hereunder shall be governed by the domestic law of such State.

FACILITIES BASED ADDENDUM TO LINE INFORMATION DATA BASE (LIDB) STORAGE AGREEMENT

Адтее	This is a Facilities Based Addendum to the Line Information Data Base Storage nent dated
_	mmunications, Inc. ("BellSouth"), and
	("NewSouth"), effective the day of
	······································
I.	GENERAL
	This Addendum sets forth the terms and conditions for NewSouth's provision of billing number information to BellSouth for inclusion in BellSouth's LIDB. BellSouth will store in its LIDB the billing number information provided by NewSouth, and BellSouth will provide responses to on-line, call-by-call queries to this information for purposes specified in Section I.B. of the Agreement.
II.	DEFINITIONS
A.	Billing number - a number that NewSouth creates for the purpose of identifying an account liable for charges. This number may be a line or a special billing number.
B.	Line number - a ten digit number that identifies a telephone line administered by NewSouth.
C.	Special billing number - a ten digit number that identifies a billing account established by NewSouth.
D.	Calling Card number - a billing number plus PIN number.
E.	PIN number - a four digit security code assigned by NewSouth which is added to a billing number to compose a fourteen digit calling card number.
F.	Toll billing exception indicator - associated with a billing number to indicate that it is considered invalid for billing of collect calls or third number calls or both, by NewSouth.
G.	Billed Number Screening - refers to the activity of determining whether a toll billing exception indicator is present for a particular billing number.

- H. Calling Card Validation refers to the activity of determining whether a particular calling card number exists as stated or otherwise provided by a caller.
- Billing number information information about billing number, Calling Card number and toll billing exception indicator provided to BellSouth by NewSouth.

III. RESPONSIBILITIES OF PARTIES

- A. NewSouth will provide its billing number information to BellSouth's LIDB each business day by a method that has been mutually agreed upon by both Parties.
- B. BellSouth will store in its LIDB the billing number information provided by NewSouth. Under normal operating conditions, BellSouth shall include NewSouth's billing number information in its LIDB no later than two business days following BellSouth's receipt of such billing number information, provided that BellSouth shall not be held responsible for any delay or failure in performance to the extent such delay or failure is caused by circumstances or conditions beyond BellSouth's reasonable control. BellSouth will store in its LIDB an unlimited volume of NewSouth's working telephone numbers.
- C. BellSouth will provide responses to on-line, call-by-call queries to the stored information for the specific purposes listed in the next paragraph.
- D. BellSouth is authorized to use the billing number information provided by NewSouth to perform the following functions for authorized users on an on-line basis:
 - 1. Validate a 14 digit Calling Card number where the first 10 digits are a line number or special billing number assigned by NewSouth, and where the last four digits (PIN) are a security code assigned by NewSouth.
 - Determine whether NewSouth or the subscriber has identified the billing number as one which should not be billed for collect or third number calls, or both.
- E. NewSouth will provide its own billing number information to BellSouth for storage and to be used for Billed Number Screening and Calling Card Validation. NewSouth will arrange and pay for transport of updates to BellSouth.

IV. COMPLIANCE

Unless expressly authorized in writing by NewSouth, all billing number information provided pursuant to this Addendum shall be used for no purposes other than those set forth in this Addendum.

CALLING NAME DELIVERY (CNAM) DATABASE SERVICES

1. Definitions

For the purpose of this Attachment, the following terms shall be defined as:

CALLING NAME DELIVERY DATABASE SERVICE (CNAM) - The ability to associate a name with the calling party number, allowing the end user subscriber (to which a call is being terminated) to view the calling party's name before the call is answered. This service also provides NewSouth the opportunity to load and store its subscriber names in the BellSouth CNAM SCPs.

CALLING PARTY NUMBER (CPN) - The number of the calling party that is delivered to the terminating switch using common channel signaling system 7 (CCS7) technology, and that is contained in the Initial Address Message (IAM) portion of the CCS7 call setup.

COMMON CHANNEL SIGNALING SYSTEM 7 (CCS7) - A network signaling technology in which all signaling information between two or more nodes is transmitted over high-speed data links, rather than over voice circuits.

SERVICE CONTROL POINTs (SCPs) - The real-time data base systems that contain the names to be provided in response to queries received from CNAM SSPs.

SERVICE MANAGEMENT SYSTEM (SMS) - The main operations support system of CNAM DATABASE SERVICE. CNAM records are loaded into the SMS, which in turn downloads into the CNAM SCP.

SERVICE SWITCHING POINTs (SSPs) - Features of computerized switches in the telephone network that determine that a terminating line has subscribed to CNAM service, and then communicate with CNAM SCPs in order to provide the name associated with the calling party number.

SUBSYSTEM NUMBER (SSN) - The address used in the Signaling Connection Control Part (SCCP) layer of the SS7 protocol to designate an application at an end signaling point. A SSN for CNAM at the end office designates the CNAM application within the end office. BellSouth uses the CNAM SSN of 232.

2. Attachment

2.1 This Attachment contains the terms and conditions where BellSouth will provide to the NewSouth access to the BellSouth CNAM SCP for query or record storage purposes.

NewSouth shall submit to BellSouth a notice of its intent to access and utilize
BellSouth CNAM Database Services pursuant to the terms and conditions of this
Attachment. Said notice shall be in writing, no less than 60 days prior to NewSouth's
access to BellSouth's CNAM Database Services and shall be addressed to
NewSouth's Account Manager.

3. Physical Connection and Compensation

- 3.1 BellSouth's provision of CNAM Database Services to NewSouth requires interconnection from NewSouth to BellSouth CNAM Service Control Points (SCPs). Such interconnections shall be established pursuant to Attachment 3 of this Agreement. The appropriate charge for access to and use of the BellSouth CNAM Database service shall be as set forth in this Attachment.
- In order to formulate a CNAM query to be sent to the BellSouth CNAM SCP, NewSouth shall provide its own CNAM SSP. NewSouth's CNAM SSPs must be compliant with TR-NWT-001188, "CLASS Calling Name Delivery Generic Requirements".
- 3.3 If NewSouth elects to access the BellSouth CNAM SCP via a third party CCS7 transport provider, the third party CCS7 provider shall interconnect with the BellSouth CCS7 network according to BellSouth's Common Channel Signaling Interconnection Guidelines and Telcordia (formerly BellCore)'s CCS Network Interface Specification document, TR-TSV-000905. In addition, the third party provider shall establish CCS7 interconnection at the BellSouth Local Signal Transfer Points (LSTPs) serving the BellSouth CNAM SCPs that NewSouth desires to query.

3.4 Out-Of-Region Customers

If the customer queries the BellSouth CNAM SCP via a third party national SS7 transport provider, the third party SS7 provider shall interconnect with the BellSouth CCS7 network according to BellSouth's Common Channel Signaling Interconnection Guidelines and Telcordia's (formerly BellCore's) CCS Network Interface Specification document, TR-TSV-000905. In addition, the third party provider shall establish SS7 interconnection at one or more of the BellSouth Gateway Signal Transfer Points (STPs). The payment of all costs associated with the transport of SS7 signals via a third party will be established by mutual agreement of the Parties in writing and shall, by this reference become an integral part of this Agreement.

4. CNAM Record Initial Load and Updates

4.1 The mechanism to be used by NewSouth for initial CNAM record load and/or updates shall be determined by mutual agreement. The initial load and all updates shall be provided by NewSouth in the BellSouth specified format and shall contain records for every working telephone number that can originate phone calls. It is the

responsibility of NewSouth to provide accurate information to BellSouth on a current basis.

- 4.2 Updates to the SMS shall occur no less than once a week, reflect service order activity affecting either name or telephone number, and involve only record additions, deletions or changes.
- NewSouth CNAM records provided for storage in the BellSouth CNAM SCP shall be available, on a SCP query basis only, to all Parties querying the BellSouth CNAM SCP. Further, CNAM service shall be provided by each Party consistent with state and/or federal regulation.

					2	RATES BY STATE				
DESCRIPTION NO.	nsoc	7	Z	Ψ	ځ	3	£	ž	သို့	Z
RED to MED Cross Connect, 2-Whis or 4-Whis, NESC	UNDCZ	\$11.63	\$8.15	¥¥	\$11.79	\$11.72	ž	\$11.68	ž	2
PMD to MMD Crose Connect, 2-Why or 4-Win, MMC	UNDC	\$11.63	\$6.15	NA.	\$11.79	\$11.72	NA	\$11.68	¥	ž
NIO, 1-2 lines, per month	UND12	ž	ž	٧N	¥	٧N	٧×	Ş	¥	¥
MRC - 18	2005	Ē	24.50	2	95 168	06:05\$	OEL.	180	TBD	TBO
NINC - ACT	2105	9	22/22	180	\$57.28	\$58.67	8	180	180 0	180
MMC: DROWNED CAPPS - 18	2015	2	≨:	8	≨:	ş	2	180	180	9
	STATE	35	ž Š	3	ž S	4	3 17	200		
MPC - Service Order automitted Electronically, per LSR - Discornect	SOMEC	. ₹	200	ž	3 2	2	5 3	2 3	£ \$	2 2
The sale	SOMAN	ž	\$21.56	Ş	\$29.24	Ş	ž	ž	ź	9
Manualy,	SOMAN	₹	13.04 14.04	MA	20.02	ž	ž	ž	ź	TBD
MRC - Incremental Overge - Menual Service Order - 1st	SOMAN	OBL	ž	DBT	¥۲	٧×	Œ	CBT	TB0	180
NRC - Incremental Charge - Manual Service Order - Add1	SOMAN	OBT	ž	780	ž	¥	TBO	TBD	TBO	TBD
MFIC - Incremental Charge - Manual Service Order - Decorrect	SOMAN	ğ	ž	52	ž	ž	ξ	TBO	180	TBD
NO. 1-4 lines, per morelli	200	≨	ž	¥	¥	¥	¥ å	≨ į	ž	ž
BIT CAN	100	3 6	5136 /3	3	136.81	135.00	3 5	3 5	3	2
NAC Decomed Chans - 1st	910W1	3 3	AN AN	3 5	20.02	NA.	3	35	3 5	3 6
MRC - Decorrect Charge - Acch	\$ CAN	2	. ₹	2	ž X	2 3	3 2	2 2	1	1 E
Electronically.	SOMEC	05.53	22.73	ž	8 23	05 23	ž	05.53	ž	021
NFC - Selvice Order submitted Electronically, per LSR - Discovnect	SOMEC	¥	20.03	ž	٧×	ž	ž	¥	¥	180
MRC - Service Order automated Menually, per LSR	SOMAN	≨	85128	ž	\$29.24	ž	¥	¥	٧N	180
MMC - Service Order authorities, per LSR, Discorned	SOMAN	≨ į	Z .	ž	ă.	≨:	ž	ž	¥	180
MARY: Incommended Charms - Mary and Carter - Addition	SCHLAN	3 E	86 AN	OF CE	ž	ž	2 E	OB C	200	
NRC - Incremental Charge - Manual Service Order - Decorrect	SOMAN	2	100	180	2	§ ≨	88	9	281	TBD
Honvourthing Charge - customer transfer, feature additions, charges (1)		98.00	ž	ź	٧¥	ž	88	¥	ž	ž
P. West Analyse VA Loren (Newstern) and month		1	1	1	50.00	1	417	1	,	
March 1		2 3	5 3	2 3	200	Š	\$ 12	2 2	2 3	2 3
NPC-X68		2	2	1	15 55	2	ž	1	2	5 3
2-Whs Analog VG Leop (Custombad), per month		ž	ž	≨	17.12	ž	₹	ž	ž	ž
MAC-1st		¥	ž	ž	\$238.73	ž	ž	¥	٧×	ž
MYC - Add		ž	¥	Ş	\$177.10	ž	ź	٧×	ž	ž
4-18hr Anadog VG Loop (Bandard), per month		ž	ž	ž	85.36	ž	ž	ž	ž	¥
KIT CAN		<u>S</u>	ž	≨:	\$457.14	≨ :	ž:	≨ :	≨:	ž
2-Wes MOM District Loco (Mandard), our month		٤٤	£ Z	źź	3 4 82	2	ž	ž	źź	≨ \$
NATC-14		ž	ž	ž	\$541.28	ž	ž	¥	ž	¥
HATC: Add		ž	¥	¥	\$431.61	ž	ž	¥	VN	Ş
2-Who ADBL Loop (Blandard), per month		ž	Y.	¥	\$10.63	į	ž	WA	¥	٧×
MC-18		Ş	ž	ž	\$713.50	ž	ž	ź	ž	ž
The Man William Part of the American		S S	ž ž	2 3	47.40	\$ \$	ž i	ž	ž	¥ i
MC-1st		ž	≨ ≨	Ę	713.50	₹ ≨	≨ ¥	ź	ž	ž
NATC - Acti		Ş	ž	≨	\$609.44	ž	ž	¥X	\$	42
A Who HDRL Loop (Nenderd) per month		ž	ź	ž	\$9.70	≨	ž	¥	¥	¥
		ž	≨ 5	≨ \$	DE 97.3	≨ 3	ž	≨ 3	≨ :	≨ :
LOOP, PICLUDING HID			1	5		٤	٤	5	٤	٤
2-We Analog VG Loop-BL1										
RC-Statements, per month	UEA 2	ž	≨ !	≨ .	¥	¥	≨.	\$15.86	¥	¥
AC. Zona 2 per month (Note 2)	UEAL?	24.75	200	214.61	200	2 2	8 8	3 2	27.67	2 2 2
RC - Zone 3, per month (Note 2)	UEAL2	E	4.45	80 923	27.7	\$52.07	15.63	8	16 903	= 1/2I
RC - Zone 4, per month (Note 2)	UEAL2	Ş	ž	ž	ž	ž	28.92	NA.	¥	ş
MPC-11	UEAL2	8898	QZ C93	\$2.54	90 963	25 CA2	22.658	\$57.99	\$70.44	\$78.90
WHC - Add	UEA 2	3	2 2 2	K 123	25.053	8	3	123	3	96 95 95 95 95 95 95 95 95 95 95 95 95 95
	UEALZ	203	20 S	٤٤	٤٤	200	8 2	٤ž	٤ź	≨ ≨
Poncaly.	SOMEC	95	2.23	98	88	85	950	05 23	88.83	8
MRC - Salvice Order submitted Dectrologity, per LSR - Decorned	SOMEC	≨:	27.02	≨:	ž	≨:	≨ ;	≨:	≨:	8
ANG: Service Order eutomated Manually, per LSR, Discorrect	SOMAN	≨ ≨	2 2	\$ \$	Z Z	≨ ≨	₹ Ž	≨ <u>≨</u>	žž	2
MHC - Incremental Charge - Manual Service Order - 191	SOMAN	27.37	ž	\$16.94	ź	\$18 14	225.52	¥82	211	ž
Service Order - Addi	NAMOS	\$12.97	≨ :	27 83	≨:	8	7 2	\$12.76	33.55	≨:
INFC - Loop Mate-Lb	SOMAN	77.77	≨ Ę	≨ É	¥ £	141	808	ž Ę	¥₹	≱ £
The state of the s	UEAVER	Ē	3	3	33	36	32	Ē	36	3

NETWORK ELEMENTS	AND OTHER BERWICES
Ē	9

					2	TES BY STA	_			
DESCRIPTION	USOC	7	ź	49	K	5	3	¥	2	ž
Z-Wire Analog VG Loop-SL2 whoop or pround start algneling	IIEM 9	M	AM	42	4M	4N	AM	219.50	MA	NA.
RC-Zone 1, Der month, Plote 2)	UEA12	\$17.96	\$18.40	\$16.84	\$17.27	\$17.65	\$18.35	2	25 125	\$15.92
RC - Zone 2, per month (Note 2)	UEAL	239.16	22.43	\$19.45	2 223	2003	213	130	55.03	20078
RC - Zone 3, per morth (Note 2)	UEA12	282. PE	1872	25 053	856.78	\$61.93	24.77	8	\$43.00	\$27.18
RC - Zone 4, per month (Note 2)	UEAL2	٧×	ž	¥	П	٧×	\$46.68	ž	¥	ž
NRC-14	UEAL2	\$145.46	25.98	\$104.17	Т	63 663	10.01	8142.97	\$170.12	\$192.97
MARC: Addi	UEM2	\$108.40	136.4	\$78.10	т	277.73	\$107.70	20.00	8128.80	\$140.72
MACC DECORDED Charge - 18	JEAN 2	5	19 60	\$ \$	Т	2 24 25	3 8	2 2	ž	2 2
MHC - Service Order extrahted Electronically, per LSR	SOMEC	82	27.27	05.03	Т	853	350	95 EZ	05.03	05.03
ARC - Selvice Order autmitted Bechanically, per LSR - Discornect	SOMEC	≨	3	ž	ž	≨	ž	ž	ž	180
MRC - Service Order authmitted Manualty, per LSR	SOMAN	≨:	21.56	≨ :	28.5	≨ :	≨:	≨ :	ž	\$19.99
MEC. Service Cook atomistic Manual Cooks Order (18	SOLAN	£27.37	3 2	2 8 8 2	2 2	818 14	25.53	2 2	27	Z
HRTC - Incremental Charge - Menual Service Order - Add1	SOMAN	\$12.97	Ş	28.42	ž	\$9.08	\$11.34	\$12.76	\$13.55	¥
1990 - Incremental Charge - Menuel Service Order - Decorned	SOMAN	\$17.77	¥	¥	ž	\$11.41	28.53	ž	٧٧	ž
A MATERIAL MODEL OF THE COORDINATION OF THE SECRETARY OF	88	8 3	200	27.72	92	// XX	X5.27	2	200	388
Ħ	UEAR	≨	ž	ž	ž	ž	¥	819.50	ž	ž
RC - Zone 1, per month (Note 2)	UEAR	\$17.98	\$18 48	20 SE	\$17.27	\$17.65	\$18.35	2	\$21.57	\$15.82
AC-Zone 2, per month (Note 2)	UEAR	200	3 :	\$19.45	2	2 2	2 5	2	22.53	8
INC. Zone A per receib Note 2)	DEAR	ž	ē. ≨	X X	2 2	ž ž	2 3	₹	3 ≥	XX.
101 - CHBY	UEAR	\$145.46	2218.96	\$104.17	\$7.96.28	23 853 23 623	\$144.01	\$142.97	\$178.12	\$192.97
MPC - Add	UEAR	\$108.40	1796.44	01 878	\$177.10	\$74.73	\$107.70	\$108.56	\$128.60	\$140.72
NPC - Discornect Charge - 1st	LEAR	503	113.41	≨ :	≨ 3	200	88 033	≨ ;	ž	ž
NACCUMENTAL CARDS - ACCU	CALC.	5 5	2 2	Ş	5 5	9 12	95.53	2 5	82	05.53
WRC - Beinto Order authriting Electronically, per LSR - Decorrect	SOME	ž	20.42	į	¥	Υ¥	ž	₹	¥	OBT
Menalty, Pa	SOLAN	Ş	\$21.56	¥	129 24	≨.	≨:	≱	۲.	\$19.99
NAC - Service Order automitted Menuelly, per LSR, Discornact	SONAN	≨ ŝ	3 7	≨ .	Z ;	¥	¥	≨	¥ 33	
MACCO TOTAL CHANGE CHANGE SERVICE CHANGE SERVICE CHANGE AND CHANGE	NAME OF THE PERSON	\$12.07	2 3	20.00	2 2	208	2 2	2127	813 56	Ž
NATO - Ingremental Change - Manual Service Order - Decorract	SOMM	\$17.77	≨	2	≨	\$11.41	28.85	¥	NA	ž
MPC - Incremental Charge - Order Coordination - Time Specific (per LSFI)	good.	88	928	242	\$36.16	1223	\$45.27	F6.24	145.43	\$85.00
4-Whe Analog VG Loop	7,10,21	111	12	12	1	Y.	477	23.74		42
DC. Zone I nor morth Dista 24	UEAL4	10401	24.24	82.23	200	24.36	22.38	2	129.47	215 32
PC-Zone 2, per mortin (hose 2)	UEALA	\$39.00	16.53	55.73	\$39.14	2.8	129.67	ğ	3	\$20.78
AC - Zone 3, per month (Note 2)	UEALA	\$70.67	878.35	98 074	867.58	27 998	04.50	æ	25.8 85	\$27.18
RC-Zone 4, per month (Note 2)	UEALA	¥ \$	≨ \$	¥ \$0.00	¥ (¥)	¥ 5	200.78	NA CT	90 UP.3	¥ 95
ACC. ACC.	7735	241.76	338	170.67	23 44 63	\$163.24	87.96.78	\$27.48	12 88 77	20100
MRC - Discornect Charge - 1st	UEALA	\$106.96	\$122.15	ž	ž	\$74.27	\$108.14	3	ž	¥
NAC-Discorned Charge - Add	THE COURT	27.0	27.00 F 00	≨ §	≨ 5	200	27/53	Ş	2 5	S 52
NAC - Sevice Order automated Decirculary, per USA - Decornect	SOME	ž	79	. ≥	≨	ž	¥X	≨	Į	180
APIC - Service Order automitted Menually, per LSR	BOMAN	ž	85 158 85 158	≨	229.24	≨:	≨:	ž	≱:	619.99
INRC - Service Order automitted Menually, per LSR, Decornect	NAME OF THE PERSON	¥ €	7 7	≨ 3	X 12	5	2 83	Z Z	3	2 2
NATC - Increment Charge - Manual Sevice Order - Add	SOMAN	\$12.0	ž	242	≨	90 93	\$11.34	\$12.78	\$13.55	ž
NFC - Incremental Charge - Manual Service Order - Discormed	SOMAN	117.77	ž	Y.	¥	11.41	\$16.06	ž	× ;	¥¥
NATC - Incremental Charge - Order Coordinaton - Time Specific (per LSA)	OCOS	2		77.57	2	2	770	1	2	3
Portre School per morti	U1L2X	ž	¥	ž	¥	¥.	NA	257.38	ž	¥
RC-Zone 1, per month (Note 2)	עבוות	2223	K22.34	\$21.89	82.2 BB	\$1.15	2012	2	992	23.52
RC-Zone Z, per morth (Note 2)	U112X	27.73	2,73	120.27	25.5	27.5	2 2	2 2	2033	81.753 SX 18
RC - Zone 4, per month (Note 2)	מונא	¥	3	¥	ž	3	2	ž	ž	ž
NAC-18	חורכא	831.55	00,9003	\$233.30	\$541.28	12 22	828.38 80.55	200	2 2 2	3 5
MAC - AOT	2011	20.5	3 2 2 2	2 1	2 3	574.77	1.00	2	ž	₹
NATC - Discornact Orange - Addi	אבויט	\$57.01	\$16.28	Ş	ž	\$39.44	\$57.27	ž	ž	NA
NFIC - Service Order submitted Electronically, per LSR	SOMEC	05 23	223	95	8.5	852	8 2	2 2	3 3	25.50
NACO Service Order automated blenually, per USA - Usecommon	SOMME	٤ž	2 2	≨ ≩	1262	₹ 2	ž	ž	ž	819 59
NRTC - Service Order submitted Manually, per LSR, Disconnect	SOLUM	ž	200	¥	3	ž	¥	NA.	≨.	180
WRC - Incremental Cherge - Manual Service Order - 1st	SOMAN	27.37	ž	X 95	\$	2 2 2	2 2	2 2 2	355	¥ \$
2	Strange			*	5					

LBOUTHMEWBOUTH RATES	METWORK ELEMENTS	AND OTHER BEINGAMES
Ž	I	1

Anactement 2 Earlist C Fames - Page 3

					4	RATES BY STATE	<u></u>			
DESCRIPTION	USOC	₹	Z	δĀ	٤	4	83	¥	ဋ္ဌ	2
NRC - tronsmental Charge - Manual Service Order - Discornect	SOMAN	11.71	≨	≨	≨	\$11.41	\$16.06	ž	¥	¥
NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	15000	8	\$36.18	Z Z	836.18	122.77	2527	7. S. S.	2	8
2-Wine Universal Digital Carrier (UDC), etatewids, per month	ž	┪	≨	ž	Ş	¥	¥	274 20	Y S	¥
Zone 1, per month	ž	22.23	7.22	8	38	21.13	25.53	2	2 0	210 %
Zone 2, per month	X S S S	7	S S	Š	2	200.2	20.97	2	.,	2
Zone 3, per month	Š	7	104.47	2	\$76.42	2/4/3	2	3	2	2
Zone 4, per morth	J	7	т	ž	2	5	200	٤	200	45.0
WRC - 1st	1	2	8 8	200	27170	1000	200	e co	2013	8 2
IMC Acci	Т	7	Т	3	2	2076	11 40	Į,	1	2
WHICH DESCRIPTION OF THE	T	7	7	2	Y Z	17 023	\$67.27	٧×	ž	2
MACCONDUCTION AND	I	S	Т	S	05.63	95 63	0568	83.50	05.53	33.55
MAN - CONTROL OF SALITION CONTROL OF LOST	Τ	1	₹¯	¥	ž	ž	ž	ž	≨	98
THE PARTY CONTRACTOR C	NAME	1	Т	¥	229 24	¥	¥	ž	ž	\$19.99
MANUAL CONTROL OCCUPANTA SAME AND PART	SOLAN	2	Т	2	3	ž	ž	ž	¥	081
Man Man Comment Character Character Control - 1st	SOMAN	15.153	≨	818 PK	ž	\$18.14	\$25.52	\$ 92	\$44.42	2
Nation incremental Channel Service Order - Action	SOMAN	\$12.87	ž	24.22	ž	8 5	\$11.34	\$12.78	\$13.55	ΚA
MARC: Jovennester Ohers - Maruel Service Order - Decornect	SOMAN	11.71	≨	¥	ž	\$11.41	\$16.06	¥	¥	NA
NRC - Incremental Charge - Order Coordination - Time Specific (per USR)	000gr	845.99	11.903	242	81.903	17.70	\$45.27	3	25.23	\$55.00
2-Who ADEL Competition Loop and Man Svs Inquiry & Fed Reservation	V6 141	1	3	V.	V.	47	A	\$14.60	ž	¥
INC. SERVICE OF INCHES	IM 2X	\$12.09	\$12.78	\$11.23	2	811.90	\$10.87	180	\$17.10	\$15.80
RC-2ms permon Acts 2	UM2X	2962	\$18.72	\$12.97	\$16.46	\$20.43	\$14.40	TBO	\$25.79	20 G
HC: Zone 3 per morah Okole 2	UM-2X	\$36.59	21.23	20.02	\$28.40	57.17	\$20.56	TBO	\$2.453	228.74
PC-Zona 4, per morth (Note 2)	UAL2X	¥	ž	Υ¥	ž	٧×	\$27.16	¥	ž	ž
NAC: 100	UAL2X	\$514.21	\$113.00	\$369.73	13.50	\$343 13	20.00	200	1900	200
IMPC - Addri	UM 2X	3	1986	\$325.16	1909	2310.83	72 57	2	2507.33	X
MRC - Decorrect Charge - 1st	UM2X	3.08.88	27.2	ž	≨	\$72.54	\$100.E8	ž	ž	5
NPC - Discorrect Charge - Add	W2X	8.93	236.23	≨ .	ž	27.85	57,00	2	2 5	\$ 5
	SOMEC	80	27.73	2	3	2	2	2		200
MAC - Service Order automated Manually, per LSR	SOMAN	S	N S	ž	20.50	Ş	ž	2	ž	C C
RATIC - Service Order submitted Menualty, per LSM, Decompan	SCHOOL	20,13	2 3	14.6	4	218.14	2 22	\$12.76	27.77	¥
March Strommer Character March Coults Order - And	SOMAN	\$12.97	ž	2	ž	80 5	7 = 7	₹	\$13.55	ž
Marchine Cana Service Order - Discount	SOMAN	817.77	2	ž	¥	\$11.41	\$16.06	¥	ž	Ş
NATC - Incremental Charge - Order Coordination - Time Specific (per USP)	00008	25.55	81.903	24.22	\$36.18	125 22	12.843	¥6.34	\$45.43	\$65.00
2-Whe ADSL Compatible Loop without like Sive Inquiry & Fee Reservation										
RC - Statewick, per morth	WZ TVO	≨	ž	≨	ž	ž	¥	814.60	≨ .	≨ .
Zone I, per north	UAL 2W	\$12.08	112.73	2112	2		\$10.6	3	2 76.7	2 8
Zone 2, per morth	NA ZW	300	27872	2	2	20.5	43 000	9	1 72	2 65
Zone 3, per month	AZ IVI	80 07	2 2	3 3	2 12	¥.	27.16	ž	ž	ž
Zone 4, per morth	WC WIT	100	20.00	22073	98 0963	255.34	29903	8 9903	2461 80	8201.79
	UM 2W	225.55	\$175.48	\$188.15	\$176.21	\$170.72	\$217.24	\$317.17	\$368.33	1 00 1
M-C - Decornect Charge - 1st	WZ WI	\$106.66	8108.29	¥	\$112.86	æ311\$	\$105.86	≨	ž	¥
MPC - Discornect Charge - Addi	MZ WN	\$56.96	15.46	≨	\$15.88	\$15.81	\$67.25	3	≨.	¥ S
MAC - Service Order submissed Decreatedly, per LSR	SOMEC	8	ž.	S.	8	2	9	2	2	28.2
AARC - Selvice Order submitted Electronically, per LSR - Decormed	SOME	2 3	2 2	\$	200	2	2 3	≨	ž	818
MAC - Service Order a formation for the Contract of	NAMOS	3	200	ž	33	Ş	ź	ž	≨	OBL
MAC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	ž	70.5	ž	\$18.14	33 83	\$12.78	2	ž
NATC - Incremental Charge - Manual Service Order - Add1	SOMM	\$12.57	¥	2	Ş	80	7	ž	2	ž
WRO - Incremental Cherge - Manual Service Order - Decomed	NAME OF THE PERSON	12.5	≨	\$ 2	¥ 35	20.00	26.50	2 2	£ 52	888
INTER-INCENSION CHIEF CONCENSION I THE SOCIETY OF LOT	3			1						
CHITTE HOSE COMMISSED LOOK BYO MEN SHE INQUIT & FAS NEWS	UHLZX	ş	ž	¥	ž	ž	¥	\$11.98	≨	ž
I BC - Zona 1 car moral) (Note 2)	UH 2X	17.88	98 83	22.08	\$6.29	\$8.97	9.5	TBO	\$12.21	\$1162
RC - Zone Z, per morth (Note Z)	UHZX	\$15.29	614.38	88	611.78	\$15.41	\$11.38	6	\$1841	814 62
RC - Zone 3, per month (Note 2)	XZ HO	627.70	\$31.65	\$14.46	Ω. (2)	44.53	916 10	2	2	80
RC - Zone 4, per fronth (Note 2)	X2+5	≨	≨ .	≨ \$	¥ .	ž	2 2 2	2 23	2	200
MRC-1st	XZ ST	181421	213	2	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2		2 2 2	2 2 2 2 2	100	3
MAC-Add	5 =	20.5	2 7518	2	ž	7.22	18	2	ž	¥¥
INSC. Discorned Charge - Add	UH.2X	\$66.90	135.23	≨	ž	23.62	867.25	≨	≱	¥
MRC - Service Order automitted Bacstonically, per LSA	SOMEC	05 ES	5 73	05 E3	S	35.0	82	8	8	82
NRC - Service Order authritised Electronically, per LSR - Disconnect	SOME	≨ 3	3 2	≨ ≨	¥ 60	\$ 3	\$	≨ ≨	2 3	96 61
MACC Service Cross against Handley, per CAT Therefore	1	5 5	30	≨	2	ş	≨	≨	ž	TBO
MACC incremental Change - Menual Service Order - 1st	SOMAN	127.37	ž	\$18.2	≨	\$18 14	23 ES	13 K	200	ž
					İ	İ				

MELLECUTHWIEWBOUTH RATES
ANTWORK ELEMENTS
AND OTHER REPACES

Attachment 2 Earlot C Fasts - Page 4

The column Column	Color Colo											
The control of the	The control of the	DESCHUTTON	OSO	¥	ŭ	3	2	LA ST	2	¥	S	2
The control of the	Colored State State State	MRC - Incremental Charge - Manual Service Order - Add i	SOMAN	\$12.97	ž	\$8.42	ž	86.08	201.5	\$12.76	\$13.44	47
Control Cont	Control Cont	MMC - Incremental Charge - Manual Service Order - Discornact	SOLLAN	\$17.77	ž	ž	ž	\$11.41	\$16.06	¥	¥	ž
Comparison	Comparison	IMPLC - Incremental Charge - Order Coordination - Time Specific (per LSR)	3000	\$45.99	\$36.18	23.22	\$36.18	125.11	\$45.27	£15.34	\$45.43	\$55.00
1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1,	March Marc	Chairman Table Longa Without Men Sve Inquiry & Fire Reserv									L	
Control Cont	Contact Cont	Zona 1 ne month	**	ž	¥	ž	٧×	٧×	¥	8615	¥	٧X
U. C. U. C	Unity Unit	None 2 ner month	¥.	29.41	08.63	\$7.86	\$2.58 \$2.00	\$8.97	S 55	180	\$12.21	\$11.62
U. C. U. C	Control Cont	Zone if our month	¥.	\$15.29	\$14.35	88	\$11.78	\$15.41	\$11.26	180	\$18.41	\$14.62
March Marc	Mary 1975		¥2.	27.73	\$31.65	\$14.48	\$20.33	\$31.48	\$16.10	780	\$24.39	96 Oct
Complement	Control Cont	Core 4, per moran	UHLZW	≨	ş	Ş	¥	2	21.25	ž	ž	ž
Comparison Com	Comparison	MI - CAM	U#2W	\$375.21	\$276.19	\$220.73	\$278.28	1272.64	\$365.82	8368.90	\$461.60	501.79
United State Colored State United State Uni	United States United State	IDDY: CLAN	UHL2W	\$325.58	\$192.81	\$1.86.15	25 25 25	\$188.02	\$317.24	\$317.17	\$368.33	\$400.94
Column C	Control Cont	MRC - Decornect Charge - 1st	UHLZW	\$106.65	\$106.29	¥	\$112.86	\$112.32	\$105.86	ž	¥	¥
Control Cont	Control Cont	NHC - DECORPER CHENS - ACC	CH 2W	\$56.98	\$15.46	¥	\$15.88	\$15.81	\$57.25	ž	2	¥
Common	Comparison	MRC - Service Order autimated Bechonically, per LSR	SOMEC	8,53	22.73	350	95 63	88	25.53	S	63.65	21.5
March Marc	Part Part	WRC - Selvice Order automitted Electronically, par LSR - Disconnect	SOMEC	≨	20.02	ž	≨	ž	ž	¥	Y.	2
## 1975 1975	March Marc	MMC - Service Order sucmitted Menuelly, per LSR	SOLLAN	≨	95128	Ş	229.24	ž	ž	¥	2	410 00
No. 10, 10, 10, 10, 10, 10, 10, 10, 10, 10,	The control Colore Color	MMC - Service Order authoritied Menualty, per LSR, Dienomect	SOLLAN	ž	19 53	ž	3	4×	Y.	1	1	Tar.
School School State St	March 1985 March 1987 March 1982 Mar	MMC - Incremental Charge - Manual Service Order - 1st	SOMAN	27.37	ž	419	٧×	418 14	23.50	2000	2	
March Chee, December SOUTH 1777 No. 1	March Chem. March Chem.	MMC - Incremental Charge - Manual Service Order - Addi	SOMAN	\$12.97	2	27.63	Y.Y	50.53	12			٤
March Mark March Mark March Mark March Mark March Mark March M	March Mark March	MPC - Incremental Charge - Manuel Service Order - Disconnect	SOMAN	17.77	ž	AM	42	411.41	20.0			1
March Far Name March M	March Marc	MRC - Incremental Charge - Onter Coordination - Time Specific (per LSR)	15000	245.99	838.18	22 123	£36.18	11 03	25.23	3	57.75	2
Market M	United State Unit	4-Who HDSL Compatible Loop, Inc. Mee Sve Inquiry & Fee Reserv									2	3
	UNIVERSITY UNI	RC - Statewids, per month	CH4X	ž	≨	ş	ž	≨	ž	\$13.97	ş	\$17.01
United Health United Healt	Heart	HC - Zone 1, per month. Oldes 2)	UK KX	\$11.52	\$14.75	\$10.39	\$7.66	\$12.97	\$10.36	180	\$16.21	\$15.46
		RC - Zone 2, per month (Note 2)	UPE 4X	\$18.71	65123	\$12.00	\$14.38	221.78	513 73	OBL	27.45	\$19.48
UNIVERSITY STATES	Health H	HC - Zone 3, per month (Note 2)	UHF4X	\$33.80	\$47.64	\$19.07	28.45	24.42	519.62	<u>1</u>	\$32.38	88 /23
Colored Colo	March 1914 1915 1915 1916	INC. Jane 4, per morth (Note 2)	H.	¥	¥	¥	¥	¥	85	ž	ž	ž
Control Cont	Contact Cont		Ž.	541.13	\$116.91	\$378.66	\$748.93	\$361.45	\$531.21	85,153	1525	02.9998
UMAK	UMAIN \$106.65 \$161.19 NA	M-C - Add	×¥35	\$491.50	\$101.71	\$344.28	\$646.17	53.85	2462.63	\$462.62	\$532.76	\$566.86
SOMEC \$15.0 \$28.10 NA NA NA NA NA NA NA N	SOMEC 153.0 153.	MFC - Decorned Onege - 1st	¥.	\$106.65	\$161.19	W	ž	\$ 22\$	\$106.86	ž	≨	ž
SOMEC RES RES NA NA <t< td=""><td>SOMEC R3 0 R2 0 N N NA <t< td=""><th>c١</th><td>UFLAX</td><td>\$56.96</td><td>\$26.10</td><td>₹</td><td>ş</td><td>27 603</td><td>\$57.25</td><td>ž</td><td>ş</td><td>ž</td></t<></td></t<>	SOMEC R3 0 R2 0 N N NA NA <t< td=""><th>c١</th><td>UFLAX</td><td>\$56.96</td><td>\$26.10</td><td>₹</td><td>ş</td><td>27 603</td><td>\$57.25</td><td>ž</td><td>ş</td><td>ž</td></t<>	c١	UFLAX	\$56.96	\$26.10	₹	ş	27 603	\$57.25	ž	ş	ž
SOMEC IN SD 42 IN INA INA </td <td> SOMEC NA</td> <th>21</th> <td>SOMEC</td> <td>63.50</td> <td>\$2.75</td> <td>63.50</td> <td>83.50</td> <td>05.03</td> <td>85</td> <td>88.88</td> <td>83.53</td> <td>05 53</td>	SOMEC NA	2 1	SOMEC	63.50	\$2.75	63.50	83.50	05.03	85	88.88	83.53	05 53
SOMEC NA	SOMEC NA \$0.95											
SOMM	SOMM NA STATE NA NA NA NA NA NA NA N	NAC - Selvice Order submitted Electronically, per LSB - Discoverant	STATE	*	\$	*	\$:	;	:	;	1
SOMM HA \$1544 \$1544 \$1544 \$1544 \$1544 \$1544 \$1544 \$1544 \$1544 \$1546 \$1546 \$1547 \$1546 \$1547 <t< td=""><td>SOMM HA \$154 \$155 \$154 \$154 \$154 \$155 \$154 \$155 \$154 \$155 \$154 \$155 \$154 \$155 \$154 \$155</td><th>MAIC - Service Order automitted Menually, per LSR</th><td>SOMAN</td><td>Ş</td><td>8</td><td>2</td><td>72 823</td><td>2</td><td>£ 3</td><td>٤</td><td>S :</td><td>2</td></t<>	SOMM HA \$154 \$155 \$154 \$154 \$154 \$155 \$154 \$155 \$154 \$155 \$154 \$155 \$154 \$155 \$154 \$155	MAIC - Service Order automitted Menually, per LSR	SOMAN	Ş	8	2	72 823	2	£ 3	٤	S :	2
SOMM \$17.37	SOWM \$77.77 NA \$16.94 NA \$16.44 \$24.00 \$11.34 \$12.76 \$14.00 SOWM \$17.27 NA NA \$11.41 \$11.34 \$12.76 \$14.00 \$11.55 \$14.00 \$14.00 \$11.55 \$14.00 \$14.00 \$11.34 \$12.76 \$14.00 </td <th>MRC - Service Order automitted Manually, per LSR, Diecornaci</th> <td>SOMAN</td> <td>ž</td> <td>Z</td> <td>2</td> <td>30</td> <td>¥</td> <td>5 2</td> <td>1</td> <td>2 2</td> <td></td>	MRC - Service Order automitted Manually, per LSR, Diecornaci	SOMAN	ž	Z	2	30	¥	5 2	1	2 2	
SOMM \$12.57	SOMAN \$12.57 NA \$14.60 \$17.37 \$15.25 NA \$15.25 COCOS \$15.79 \$24.18 \$14.22 \$24.18 \$15.27 \$15.34 \$15.27 \$15.34 \$15.27 \$15.34 \$15.27 \$15.34 \$15.25 \$15.34 \$15.35 \$15.34 \$15.35 \$15.34 \$15.35 \$15.34 \$15.35 \$15.34 \$15.35 \$15.34 \$15.35 \$15.34 \$15.35 \$15.34 \$15.35 \$15.34 \$15.35 \$15.34 \$15.35 \$15.34 \$15.35 \$15.34 \$15.35 \$15.34 \$15.35 \$15.34 \$15.35 \$15.34 \$15.35 \$15.34 \$15.35 \$15.34 <t< td=""><th>NRC - Incremental Charge - Manual Service Order - 1st</th><td>SOMAN</td><td>\$27.37</td><td>ž</td><td>818.94</td><td>¥</td><td>S18.14</td><td>2 83</td><td>20.00</td><td>2</td><td>3</td></t<>	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	ž	818.94	¥	S18.14	2 83	20.00	2	3
COMM \$17.77 NA NA \$11.41 \$16.00 NA NA UMAW \$10.29 \$24.10 \$22.77 \$46.27 \$46.34 \$46.31 \$46.32 \$46.31 \$46.32 \$46.31 \$46.32 \$46.32 \$46.32 \$46.32 \$46.32 \$46.32 \$46.32 \$46.32 </td <td>COOSE \$11.77 NA NA \$11.41 \$10.00 NA \$11.41 \$10.00</td> <th>MRC - Incremental Charge - Menuel Service Order - Add1</th> <td>SOMAN</td> <td>\$12.97</td> <td>≨</td> <td>58.42</td> <td>ž</td> <td>90 93</td> <td>\$113</td> <td>\$12.76</td> <td>55 C18</td> <td>Y.</td>	COOSE \$11.77 NA NA \$11.41 \$10.00 NA \$11.41 \$10.00	MRC - Incremental Charge - Menuel Service Order - Add1	SOMAN	\$12.97	≨	58.42	ž	90 93	\$113	\$12.76	55 C18	Y.
Decorate States S	DECOSE \$45.59 \$58.10 \$12.27 \$45.77 \$45.57 \$46.54 \$16.54 DELAW NA	WRC - Incremental Charge - Manual Service Order - Discornact	SOMAN	17.71	ž	ş	Ş	\$11.41	\$16.06	¥	Y.	Y.
UHAW STORE STATE STORE STATE STORE STATE STORE STATE	UHAW NA N	MRC - Incremental Charge - Orbir Coordination - Time Specific (per LSR)	3000	853	81.903	242	91 903	\$32.77	245 77	3	57.97	846 m
UH4AW \$11.52 \$14.75 \$10.36 \$12.97 \$10.34 \$1	UH4AW \$11.52 \$14.75 \$10.50 \$17.50 \$12.77 \$10.55 \$1	4-With HDSL Compatible Loop, without Man Svc Inquiry & Fac Res		Ī								
UHAM \$10.00 \$10	UHLAW \$18.71 \$19.75 \$19.50 \$	Superior, per mount	¥	ž	ž	ž	ş	¥	¥	26.01\$	¥	1661
UHAW \$100.00 \$17.00 \$17.00 \$17.00 \$17.00 \$24.44 \$19.77 \$19.00 \$22.00 \$24.44 \$19.77 \$19.00 \$22.00 \$24.44 \$19.00 \$22.00 \$20.00 \$24.44 \$19.00 \$20.00 \$20.00 \$24.44 \$19.00 \$20.00 \$20.00 \$24.44 \$19.00 \$20.00 \$20.00 \$24.44 \$19.00 \$20.00 \$20.00 \$24.44 \$20.00 \$20.00 \$20.00 \$24.44 \$20.00 \$2	UHANN \$50.00 \$47.00 \$15.00 \$24.00 \$15.00 \$25.00 \$24.00 \$25.00 \$25.00 \$24.00 \$25.00 \$25.00 \$26.00 \$2	Zone 2 our month	37, 24		214.00	R	2/ 66	112 97	810.38	Q.	18 21	\$15.46
UHLAW STOSE STOS	UHLAW SECSIA SE	Zone 3, per month	THE COL		20.57	3	2 2	2	213.63	3 4	2 2 2	519 46
UHLAM \$100.00 \$200.00 \$200.00 \$200.7	UHLAM \$100.00 \$200.40 \$270.70 \$192.70 \$292.71 \$190.35 \$440.11 UHLAM \$100.00 \$200.01 \$200.20 \$192.70 \$294.51 \$190.35 \$190.35 UHLAM \$100.00 \$200.01 \$200.20 \$292.10 \$210.00 UHLAM \$100.00 \$210.00 \$210.00 \$210.00 \$210.00 UHLAM \$100.00 \$110.00 \$100.00 \$100.00 UHLAM \$100.00 \$110.00 \$100.00 \$100.00 UHLAM \$100.00 \$100.00 \$100.00 \$100.00 UHLAM \$100.00 \$100.00 \$100.00 UHLAM \$100.00 \$200.00 \$100.00 UHLAM \$100.00 \$200.00 \$100.00 UHLAM \$100.00 \$200.00 \$100.00 UHLAM \$100.00 \$100.00 \$100.00 UHLAM \$100.00 \$200.00 \$100.00 UHLAM \$100.00 UHLAM \$100.00 \$100.00 UHLAM \$100.0	Zone 4, per morth	CH.AW	. ₹	2	2	1	Į.	8 8	S M	977	8 2
UHAM \$100.05 \$114.30 NA \$112.17 \$124.19 \$100.05 \$140.02 \$150.07 \$100.05 \$114.30 NA \$112.17 \$124.19 \$100.05 \$140.02 \$150.07 \$100.05 \$140.05	UMAN \$100.66 \$114.30 NA \$12.17 \$124.19 \$105.68 NA NA NA NA NA NA NA N	NHC-18	UH-4W	13	97 1323	20.00	87 A 70	27.00.72	16.000	30.00	2	2000
March Marc	March Marc	MACC - Add	UPLAW	05 750	F250 01	205.28	\$192.02	245.15	E343.63	23 (75	8380 78	2429 86
March Marc	March Marc		-									
Adm USLACK \$19.56 \$19.56 \$10.56 <td>Adm Light Set \$19.56 \$19.56 \$19.56 \$15.55 NA <th< td=""><th>MPC - Disconnect Charge - 1st</th><td>CH.CW</td><td>\$108.65</td><td>2112.30</td><td>2</td><td>\$122.17</td><td>\$124.10</td><td>20.5</td><td>\$</td><td>\$</td><td>¥</td></th<></td>	Adm Light Set \$19.56 \$19.56 \$19.56 \$15.55 NA NA <th< td=""><th>MPC - Disconnect Charge - 1st</th><td>CH.CW</td><td>\$108.65</td><td>2112.30</td><td>2</td><td>\$122.17</td><td>\$124.10</td><td>20.5</td><td>\$</td><td>\$</td><td>¥</td></th<>	MPC - Disconnect Charge - 1st	CH.CW	\$108.65	2112.30	2	\$122.17	\$124.10	20.5	\$	\$	¥
Out Description of the LSA SOURCE \$1.90 \$2.75 \$1.50	Option Description of the Confection of the	NFIC - Decornect Charge - Addr	CF.	256.88	\$19.56	ž	19 02	500	× 153	¥	1	5 3
Mode Decornect COMEC NA RM NA	One Description NA FAN DESCRIPTION NA FAN DESCRIPTION NA FAN DESCRIPTION NA FAN DESCRIPTION	MRC - Service Order automitted Electronically, per LSR	SOMEC	\$3.50	20.23	33	\$3.50	05.03	3.50	0913	95.03	95.03
Order Coordination - Time Specific part LSPQ COMBAN NA EV1544 NA EV1544 NA EV1544 NA NA <td>Order Conclusion: Time Specific part LSQ SSACH NA EVELSA NA EVELSA NA EVELSA NA NA</td> <th>MRC - Selvice Order eubmitted Decernically, per LSR - Dieconnect</th> <td>SOMEC</td> <td>Ş</td> <td>20 05</td> <td>ş</td> <td>ž</td> <td>ž</td> <td>¥</td> <td>¥</td> <td>ž</td> <td>150</td>	Order Conclusion: Time Specific part LSQ SSACH NA EVELSA NA EVELSA NA EVELSA NA	MRC - Selvice Order eubmitted Decernically, per LSR - Dieconnect	SOMEC	Ş	20 05	ş	ž	ž	¥	¥	ž	150
Manual Service Order - 18	Marine Service Octave - 18	MRC - Service Order automitted Manually, per LSR	SOMAN	Ş	85128	٧¥	\$29.24	ž	Ş	ş	ş	20.5
SOMAN \$27.37	Manual Service Order Add1 SCOMM ST277	MMC - Sevice Urbar authoritied Manually, per LSR, Discorried	SOMM	ž	7	ş	13.94	ž	٧V	Ş	ž	180
Married Service Order Decorporated SOMANY STOTE NA NA STOTE STOT	Married Services Order-Disconnect SOMANY \$17.77	Mile Investment Change Hand Coulty Add	2000	16 /21	≨ :	20.00	ž	\$10.14	23.53	Z 82	8	ž
One-Confinition: The Specific part LSQ Document SET OF SE	Code Condition This Specific part LSQ	Mile American Charles Charles Charles Charles Charles		312.3/	¥ :	2	≨ I	8	2	\$12.78	23.55	ž
USLX USLX	USLVX NA NA NA NA NA SEC S	MFC - Incremental Charge - Order Coordination - Time Specific face LSRI	8000	3	2 80	¥ 22	2 2		216 06	¥	¥	≨ \$
OSLXX	OSLOCK USLOCK SAL 10 S	4 Win DS1 Digital Loop								5	2	3
150 150	USLICK 261.74 564.68 555.51 556.72 550.99 TBO 559.61		DSI DO	ž	¥	KX	¥	X	Υ×	\$62.73	ž	¥
USLXX 88.40 89.17 854.15 834.06 888.73 857.96 180 849.00 8	USLXX 884 05 89471 86413 894 06 806 73 847 96 180 848 90 818 91 91 91 91 91 91 91 91 91 91 91 91 91	륁	NS XX	121	28 735	\$55.53	\$20.28	£ 83	86 OS	180	129 61	22,73
USLUX \$84.05 \$94.71 \$64.13 \$54.05 \$66.73 \$47.94 180 \$48.90 USLUX \$152.29 \$7040 \$107.91 \$152.74 \$107.97 \$107.91	USLXX \$84 06 \$84 71 \$64 13 \$16 73 \$47 14 \$16 75 \$48 15 \$18 05 \$1											
USLXX 819.29 \$70110 \$10124 \$197.57 \$99.54 180 \$119.05 USLXX NA NA NA NA NA 187.77 NA NA NA	USLXX 115.22 PORD 517.12 117.27 135.44 TBO 517.05 117.17 NA NA NA NA NA NA NA N	RC - Zone 2, per month (Nose 2)	NSLXX.		-	\$64.13	254.08	27.362	\$67.58		8	\$73
USLXX MA NA NA B12747 NA NA NA	USLXX NA NA NA NA 127.47 NA NA NA NA 157.77 NA NA NA NA NA NA NA NA NA NA NA NA NA	FRC - Zone 3, per month (Note 2)	xxrsn	П	┰	\$101.8G	\$162.34	\$197.57	85 56	П	\$119.06	\$96.59
	USLUCK \$610.13 \$540.00 \$473.90 \$410.90 \$559.00 \$714.64 \$715.77	FIC - Zone 4, per month (Note 2)	XIXI	7	+	≨	ž	ž	\$127.47	М	¥	ş

RELEGITIMMENMOLTH AKTES
NETWORK REPARCES
AND OTHER STREETS

Anachmen 2 Evibr C Rabe - Page 5

					5	RATES BY STATE	<u> </u>			
IND. AND	200	7	z	3	۷	5	ş	¥	၁	X
MAC Decorrect Charas 1st	XX 891	22.25	B 98	2	12 523	255	8273	\$121	¥21.50	2219.72
MAC - Discorrect Charge - Addi	XX ISI	\$KA 97	21.60	4	2 2	27 26	200	\$:	Š	200
MAC - Service Order eutomitted Electronically, per LSR	SOME	Ş	2 20	\$ 5	63 63	03 63	0700	Y S	ž	5
MRC - Service Order automitted Electronically, per LSR - Decorment	SOME	ş	279	N N	2	3	3 2	2	2	3
MRC - Service Order automitted Manuelly, per LSR	SOMM	ž	9110	ž	76 663	42	1	1		28
MRC - Service Order authralited Menselly, per LSR, Disconnect	SOMAN	ž	200	ž	3	¥	¥N	¥2	Ž	C C
MRC - Intremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	ž	\$18.9	¥	\$18.14	22.53	200	41.17	2
MHC - Incremental Cherge - Manual Service Order - Add1	SOMAN	\$12.97	¥	28.5	ş	80.8	¥.1.	\$12.76	\$13.55	2
MHC - Incremental Charge - Manuel Service Order - Discorrect	SOMM	817.77	ž	ž	≨	\$11.41	\$16.06	ž	ž	ž
MRC - Incremental Charge - Order Ocerchiston - Time Specific (per LSR)	19000	86.18	\$36.18	23 K3	\$28.18	\$33.06	\$48.17	\$45.34	\$48.47	\$85.00
HC. Statestin, or month	27.00	1	1	177		1				
PC-Zone 1, per month (Note 2)	75 101	200	28.5	¥ 36	2	20.50	<u>ا</u>	è	ž	22.22
RC - Zone 2, per month (Note 2)	101.56	977	157.23	72 627	200	27.22	3 2	8	2 3	20.43
MC-Zone 3, per morth (Note 2)	UCKS6	\$50.45	\$128.22	1211	06.753	896.40	\$48.51	2	C7-99	\$65.75
INC. Zone 4, per month (Note 2)	95700	ş	ΝA	YN	¥	ž	\$64.02	¥	ž	ž
	833	8	7 79	8	23.052	\$200 S	2469.00	\$489.04	\$602.73	\$643.00
MPC - Decomed Chara - 1g	25	200	\$428.45	R7172	\$178.11	96.00	537.83	1337.51	95 SEC	\$21.26
ANC. Decornect Charge - Addi	2503	25.23	27.72	€ ₹	06/23	2 2	X 73	4	3 2 2 2	\$ 3
NRC: Service Order automitted Electronically, per LSR	SOMEC	95.53	27.23	88	88	\$3.50	05.03	\$3.50	\$3.50	0513
	ASSESS OF THE PARTY OF THE PART	Ş	2	ž	ž	≨	≨	ž	ž	8
MCC. Service Order is furthered Manually over 1999	711	1		:		:		;	;	
MPC - Service Order automitted Manually, per LSR, Discornect	SOMAN	2 3	8 3	2 3	× 25	ž	2 3	5 5	≨ 5	66618
MRC - Incremental Charge - Manual Service Order - tet	SOMAN	15 123	≨	282	ž	\$1814	28.50	18 923	\$ 3	Ž
MRC - Incremental Charge - Manual Service Order - Add?	SOMAN	\$12.97	ž	58.42	ž	88	¥ 15	\$12.78	ž	ž
MARC - Incremental Charge - Manual Service Order - Discornect MARC - Incremental Charge - Order - Charge - Transfer - Charge - Ch	SOMAN	11.11	¥	ž	¥	\$11.41	\$16.06	\$	¥	ž
A Way 64 Kipes Dis Grade Logs	333	R	20.10	XX.22	206.18	222.77	25.27	Š	25.23	855.00
HC - Suserick, per month	UOLEA	ž	ž	ş	ž	¥	¥	400	27.73	200
RC-Zone I, per month (Note 2)	LOUBA	27.33	ED 603	52.73	25 523	\$27.50	19 62	5	87473	55.50
HC-Zone 2, per morth (Note 2)	MO	24.60	12721	72 623	20.02	M7.24	15 B St	180	151.67	\$45.07
IAC Zone 4 car month (Note 2)	200	200	-	21.27	86.72	27 FE	3	180	268.43	55.73
NATC 1st	700	883	1	25.65	7	#C128	3 5	7 20 7	Z 22	¥¥
MRC: Addi	UDLEA	\$343.70	\$4.29.45	24120	\$176.11	\$230.60	537 60	137.61	05 080	20.3
ANC - Discerned Charge - 1st	DOL64	312362	\$122.15	\$	\$127.74	\$87.95	\$128.36	ž	25	¥
NYC. Service Order submitted Businessey, car (SR	COMPC	Q S	27.0	≨ \$	8/2	2 5	2 2	¥	\$13.56	ž
MRC - Setvice Order submitted Becacolically, per LSR - Decornect	SOMEC	2	3	₹ 2	2	₹ 2	3 ₹	2	2 2	2 2
MRC - Service Order externities Manually, per LSR	SOMAN	ž	21.56	ž	12924	ž	ž	¥	ž	\$19.99
MAC - Service Order such densely, per LSR, Decornect	SOMAN	ž	¥.	ş	Z S	ž	ž	ž	ž	¥
WRC - Incremental Charge - Manual Service Onter - Auch	SOMAN	\$12.57	٤٤	200	٤ ≥	105	X Z	2 2	ž ž	1
NRC - Incremental Charge - Manual Service Order - Discornact	SOMAN	11.77	ž	ş	ž	\$11.41	\$16.06	¥	ž	5 ₹
2. When I lish Consent Long Burge - Cross Coordination - Time Specific for USR)	3000 3000 3000 3000 3000 3000 3000 300	8.93	236.18	Z Z	91 963	£32.77	12 945	72.74	145.43	855.00
RC - Statewick, per month	T	815.11	ı	13.97	2	₹	≸	ž	8023	¥
RC-Zone 1, per month. (Note 2)	П	2	\$18.60	919 80	214.94	\$16.34	\$16.85	\$12.27	816.90	819 86
HC - Zone 2 per month (Note 2)	1	2	1	22.88	\$15.16	\$17.99	22.34	C9 02\$	05 923	827.38
RC Zone 4, per month (Note 2)	T	8 4	- 1	70.00	7	28.2	2 5	23.00	5,72	
HRC-1st	La Pe	\$614.21	1	91.0623	23.93	5306.13	20.03	2508	19 0098	127001
MRC: Add	П	27.25	1 1	217.39		246.38	\$456.24	\$390,00	\$507.33	23463
MAC Decorate Chara Add	T	2 5		\$142.27	\blacksquare	\$150.29	8105.06	≨:	ž	2.2
MRC - Service Order submitted Electronically, per LSR	Τ	8.23		8 8	7	3 8 8	C 9	× 95	5 G	3 5
NATC - Service Choir submitted Electronically, per LSR - Discornact	П	≨	1 1	ž	7	ž	ž	ž	ž	≨
MRC - Service Order authorities Manually, per LSR, Decornage	T	≨ ≱	- 1	2 3	282	4 4	2 2	≨ :	ž	219 96
NAC - Incremental Charge - Manual Service Order - 1st	П	87.00		\$10	Т	\$18 14	25.52	3 83	2,3	₹ ≨
MRC - Incremental Charge - Manual Service Order - Add1	NYWOS	8	¥.	798	≨	88	X E	\$12.76	23.53	\$
EL CHINES CALLES OF THE CONTRACT CONTRA	٦	\$17.77	- 1	\$142.27	٦	\$1141	S16 GB	ž	818	ž

MELLSOUTHWEWBOUTH RATE: METWORK BLEMENTS AND OTHER BERNCES

Anachment 2 Earbr C Rates - Page 6

ŀ		٤		ξ.		ł					
-	MCC - Incremental Channe - Manual Service Order - Discourse - Add	OSOS.		٤	ð	¥	3	£	¥	ဌ	Ţ
t	HAPC - Incremental Charge - Manual Order Coordination - per loss	SALO.	W 913	NA 4	90.75	2	2	809	¥	8	¥
1	Und Copper Localitated (4 at a 1865), without then \$1.0 feet flast	3	3	10.014	20.40	216.31	11 22	24 5.27	\$16.00	25.43	27.5
ģ	Statewick, per mon	AND ICE		•	4.7						
L	Zone 1, per month		5 6	٤	2	2	ž	ž	≨.	200	ž
t	Zone 2, per month	30.0	1	603.00			2016	200	\$1227	\$18.90	\$19 65
t	Zone 3, per morth			27.72		213.13	8	K.52	2	28.20	3
1	Zona 4, ner morth			0	3	613.73	\$ 10.63	22 122	22.0	\$37.75	\$38.8
t	NPC . 14		ž	ž	≨	ž	ž	\$42.13	ž	Ş	ž
ŧ	W77 : X07	200	5275.21	827.00	\$154.13	80 892 8	\$253.48	\$366.B2	831.8	\$461.61	\$131.0
‡		UC PW	852.58	\$173.62	5.39.75	\$174.35	\$166.86	\$317.24	8.58	00 9903	388
1	And Checomed Charge - 1st	NOT DA	081	\$108.29	6140.73	\$112.86	\$112.32	\$105.86	ž	MA.	£74 €.
	MRC - Decornect Charge - Addi	₹	180	\$15.46	237.65	\$15.88	615.81	K17 24	12	1	
	NFC - Service Order automitted Electronically, per LSR	SOME	2	× 0	S	S	3			3	
E	MEC. Septem Order as formers for programs, new 100. December 2	2000		27.7	2	2	3	3	2	8	2
t	AND Course Annual Lange and Princers		ž	200	ş	¥	ž	¥	¥	MA	ž
+	AND COLUMN THE LOCAL PROPERTY AND THE PARTY	SOMAN	ž	38	≨	\$29.24	¥	ž	≨	ž	\$19.9
1	W. C. CONVENCION BUSINESS AND LIGHT, MICHAEL	SOMAN	NA	23.64	≨	ı	ž	¥	¥	≨	₹
Ⅎ	Pert Intremental Orange - Manual Service Order - 1st	SOMAN	27.8	ž	ž	ž	\$18.14	20.00	20.80	£13 M	3
	MPC - Incremental Charge - Manual Service Order - Addi	NAMOS	21.00	ž	Ą	AM.	5	78 0.00	61278		
	MPC - Incremental Charge - Manual Service Order - Decornact - 1st	SOLLAN	41777	Ž	¥N	12				2	
	NFIC - Incremental Charge - Manual Service Order - Disconnect - Add1	SOMAN	\$17.77	¥	Y.	MA		20,000		3 8	
	NPC - Incremental Charge - Manual Order Coordination - per loop	NOWC	816.00	\$16.31	S78.4A	£16.31	1 62	646.97			
											5
4	John Conser Leonf, one (> 18ke), Incl Man St. & Fee Plan										
F	1	3	20,000			1					
Ł	RC - Zone I ner morth. Abba 21		3	٤		ž	ž	Ş	ž	¥	≨
Ł	RC. Jone 2 Am month. White 21	4 4 4	2	2	8	20 13	3	\$16.65	839.65	\$18.90	\$19.8
ĺ	The table of the factor of the	A COL	2	528.13	22.08	24931	\$59.76	22.34	8	528.50	6.43
ſ	Communication (Communication)	Maz	<u>8</u>	\$71.17	236.34	\$80.78	\$104.74	25.52	\$73.89	\$37.75	\$35.8
$\frac{1}{1}$	PAC-CANTER, PACIFICIAN (PACIFIC)	K DA	≨	¥	≨	≨	V.V	\$42.13	ž	Ş	ž
	WRC - 1st	Z Z	\$614.21	\$331.86	\$326.16	14 023	u au	6504.63	2003	Seconda.	W.63
	WHC - Add	× 01	2777	410127	2017	2013	8		4200		
E	MRC - Decorment Charge - 1st	60			***					3	
E	ABC - Decorate Chara - AAri	1	٤	3	12.0	90.00	200	31(0) (1)	ž	ž	Š
\mathbf{f}		3	Ş	ž	99/22	236.20	836.82	\$57.25	≨	٧×	\$39.1
$\frac{1}{1}$	MANA - CHEVILLE LANGE BLOTHERS EMCCROTHOUSEY, DAY LOAN	SOMEC	23.50	22.73	88	8	05:03	83.50	05.03	8 2	3
$\frac{1}{2}$	MHC - Selvice Order suchrelled Electronically, per LSR - Discovned	SOMEC	ž	50.42	٨×	ž	ž	≨	ž	ž	2
-	MFIC - Service Order automitted Menuelly, per LSR	SOMAN	ž	\$21.56	ž	\$29.24	ž	ž	ž	ž	613
	MRC - Service Order submitted Manually, per LSR, Dieconnect	SORAN	Ş	19.03	ź	S E	Ž	Y.	7	3	
	NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	27.8	2	385	647.00	,	638.80	2		
Ē	NRC - Incernated Charte - Manual Service Order - Arth	COMMAN	8							3	٤
F	NAC - Incompanie Chans - Manual Service Order - Decreases - 145	777700				5	8		3/5/0	X 2	ž
F	ADD: Investment Change Library Course Order Change	NAME OF THE PERSON		Į.	3142.27	ş	17.	816.08	ž	8	≨
Ŧ		SCHOOL STATE	///	ž	8 /2	ž	\$11.41	\$16.06	¥	8.18	ž
Ŧ		UCMC	8.9	\$16.31	\$36.46	\$16.31	120.71	\$45.27	\$16.00	\$46.43	87
-											
٤	1										
Ŧ	Same and the same	E 3	B	≨	27.00	ž	≨	ž	ž	Ş	≨
Ŧ		WZ CS	QE.	248.73	9	\$36.19	273 82	_	99.903	\$18.90	\$19 86
-	CONB Z, Dar Morte)	NG 2W	2	\$58.13	9	563	22.00	_	84.8	828.50	574
-	2016 3, per morfi)	NG.2W	OBT	\$71.17	8	\$80.78	\$104.74	۲	87.2 69	K 153	2.8
	Zone 4, per month	WZ CO	≨	≨	t	₹	ž	r	3	ž	1
F	HPC-1#	NC SW	1375.21	10001	117.13	th west	Sam As	636.63	8		2
E	#C· Add	LC 2W	2 20	2115	٠			+	_		
E	APC - Discornect Change - 1st	76.51		1	۲.	2		+	_1	3	S
F	MIC. Denomer Change, AAC	170	2	28.5	1	3112.00	21.12.22	7	_1	ž	2
$ar{1}$		* T	Į.	2.48	ł	\$15.88	\$15.81	-1		¥	\$39.14
$\frac{1}{1}$	THE SERVES ONLY SUCHES BESTER SHOULD BY LIST	SOMEC	82 50	22.73	-	8	53.50	_	05.53	05:03	3
-	HRC - Selvice Order authmitted Electronically, per LSRI - Discornact	SOMEC	¥	27.08	┝	≨	×	t	Ş	ž	ž
	ARC - Service Order automated Manuelly, per LSR	SOMAN	2	21.0	╆	620 34		1	1	1	5
F	ARC - Service Order submitted Manually, ner LSR, Dienoment	NV TO			+		٤	1	٤	٤:	
F	APC - Ingemental Oberna - Manual Service Onter . 1-4	1	2	5	┥	2	Ş.	7	ž	2	ž
Ŧ		SCHOOL	B/X	ž	-	≨	\$18.14	П	£ 92	87.8	Ş
1	WHICH THE CHARGE CHARGES MANUAL SERVICE CHOST - ACCT	SOMM	21.00	ž		ž	90.88	Н	\$12.76	23.52	ž
7	MFC - Incremental Charge - Manual Service Order - Decornact - 1st	SOMAN	\$17.77	Ş	ı	≨	\$11.41	\$16.06	¥	812	ž
-	WRC - Incremental Charge - Manual Service Order - Disconnect - Add1,	SOLAN	11.77	Ş	l	Į	\$11.41	1	2	813	2
_	WRC - Incremental Charge - Manual Order Doordination - per topp	LIGHT CO.	8,68	10.00	۲	618.31	1 22	t	2		
Š	Und Copper Local Short (4 or a 1882) Inch Shan St & Fac Shan				+			1	3	2	2
į	the car availt	94 23	5		1						
F	And I ne month	2		Š.	20.00	ď.	┪	ž	ž	7	¥
Ŧ		UCA S	2	e G	1	ę S	╗	83.8	\$1848		8
7	COLD C. DALLICIEU	UCL4S	2	23.053	-	123 00	_	27.50	22 23	H	32.53
7	Cont. 3, per month	2	9	\$32.24	H	819.08	r	00 003	574 14	t	500
	Zone 4, per month	10.00	į	MA	t	47	477	8			3
F	4FC - Sa	200	5 6	5	+	Т	+	3 2 2	7	+	ž
Ŧ	USC. Add	3	3	┪	00 77	20 33	2	7	2400 00	888	\$4000
Ŧ	17 - W. C. C. C. C. C. C. C. C. C. C. C. C. C.	SYCI	130	_	_	_	-	۲	Г	ŀ	*
7	THE CHARGE CHARLES THE			ľ	t	Ī	t	1		4	3

MILES MOS (DOPS

METHOCH METHOCH MATES
METHOCH ELEMENTS
AND OTHERS BEFANCES

\$710.71 \$710.72 77.07.7 72.37 74.65
 USBSA
 \$617.43
 \$11.17
 \$42.10
 \$46.21
 \$46.22
 TBD
 \$46.04
 \$40.05
 \$46.22
 TBD
 \$46.04
 \$40.05
 \$46.22
 TBD
 \$46.04
 \$40.05
 \$46.22
 TBD
 \$46.04
 \$40.05
 \$46.22
 TBD
 \$46.04
 \$40.05
 \$46.22
 TBD
 \$46.04
 \$40.05
 \$46.02
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$40.00
 \$ 9 \$7.16.32 \$7.16.32 \$ \$716.32 \$ \$716.32 \$65.24 25 25 25 25 25 25 26 25 25
 \$69.28
 \$65.40
 \$66.30
 \$60.55
 \$60.56
 \$66.10

 \$77.70
 \$710.73
 \$700.71
 \$600.00
 \$718.53

 \$23.49
 \$23.74
 \$23.74
 \$23.50
 \$23.66
 \$709.71 \$480.00 \$23.74 \$27.30 \$66.35 \$121.14 \$710.73 \$710.73 \$23.77 \$757.04 \$757.04 \$70.59 \$710.71 \$710.71 \$716.70 \$710.71 \$23.55 \$23.77 \$66.28 \$66.44 \$7.8.73 \$7.8.73 \$23.56 NAMOS NAMOS NAMOS NAMOS NAMOS NAMOS NAMOS NAMOS NAMOS NAMOS NAMOS NAMOS NAMOS NAMOS NAMOS UNION OF THE O ULIMO ULIMO ULIMO ULIMO 2087 1 PATC. OCAS - Facility Territorion. 18

PATC. OCAS - Facility Territorion. 18

PATC. OCAS - Facility Territorion. 18

PATC. OCAS - Facility Territorion. Document 18

PATC. OCAS - Facility Territorion. Document 18

PATC. OCAS - Facility Territorion. Document 18

PATC. OCAS - Facility Territorion. Document 18

PATC. OCAS - Facility Territorion. Document 18

PATC. OCAS - Facility Territorion. Document 18

PATC. OCAS - Facility Territorion. Document 18

PATC. OCAS - Facility Territorion. Document 18

PATC. OCAS - Facility Territorion. Document 18

PATC. OCAS - Facility Territorion. Document 18

PATC. OCAS - Facility Territorion. Document 18

PATC. OCAS - Facility Territorion. Document 18

PATC. OCAS - Facility Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCAS - Parallel Territorion. Document 18

PATC. OCA NPC - Load Colificulationant Plantoval per 2 Wite pair - Locya bes than or equal to 18th -FPC - Load Colificulationant Removal per 2 Wite pair - Locya greater from 18th - 1st -WIC - Load Colificulationant Removal per 2 Wite pair - Locya greater from 18th - Actif -if MPC - Load Colficulament Personal per 4 Wire pair - Loops least fren or equal to 18et = MPC - Load Colficulament Personal per 4 Wire pair - Loops greater than 1861 - 181 ^{st.} MPC - Load Colficulament Personal per 4 Wire pair - Loops greater than 1861 - 181 ^{st.} MPC - Bridge Tap Personal per pair unbodied ^{st.} the Leap Modification/Conditioning

_		
2		
ž	e	2
Ē	E	ğ
3	3	ĺ
Ē	교	2
į	ž	b
Ž	£	Ě
3	6	Ž
١	z	2
ď		

Service Order - December Scholary Tigo No. N	Comparison Com		SCHOOL	30			į		RATES BY STATE				
Section Sect	Color Companies Color Colo	-	NHC - Incremental Charge - Manual Service Order - Addit	S			\$	2	4	3	ž	၁	2
Court Convention Part	Copyrig Gales 1965		MRC - Incremental Charge - Manual Service Order - Decorrance	MANOS	8 8	٤	ź:	ž	88	ē	≨	≨	TBO
Control Cont	Colore Location Colore		MRC - Incremental Charge - Manual Order Coordination - per loco	J. K. BAA	2	2	¥	ž	21.41	<u>a</u>	۲ ۲	780	180
Color Colo	Colore C	2	to Debthytion per 2 Wire Unbundled Copper Sub-Logs, per month	Noon.	Q :	\$16.31	27.53	\$16.31	\$16.29	CEL	\$45.34	\$45.t3	8
Color Colo	Colored Heat Colo		Zone 1, per month		8	≨	2	ž	¥	TBD	ž	\$6.82	ž
No. 2017 Column	Colored Color Colored Color Colored Color Colo		Zone 2, per month	3	2	57.91	<u></u>	\$8.01	25	180	\$7.33	¥	GEL
Colore C	Comparison Com	F	Zone 3, per month	XXXI	9	\$10.37	Œ	\$9.18	\$11.43	B E	\$30.88	ž	185
Colore C	Colore C		Zone 4, per month	3	2	\$12.76	욢	\$11.02	\$14.08	OBT	\$12.36	¥	G
Colored Decision Colored Dec	Comparison Com	F	N-C-18	3	٤	4	ž	≨	≨	780	Ş	ž	Cer
Color Colo	Company Comp	L	N.P.C. Add	XZSZ	52.85	4	OBT	\$139.19	\$136.96	180	\$137.10	25:24	Į,
Colored Colo	Colore C	r	INTC - Discourse Obers - 14	XXXX	888	4	DBT	\$61.90	10.00	180	76 055	\$70 mg	É
The color of the	Vicable Vica	ļ	March Character	XXXX	13.68	4	567	\$101.18	5100 70	QE.	878 AA		
Colore C	Court Cour	‡	W.C. Carrier Carrier State of the Control of the Co	XZSON	\$14.06	Н	98	\$13.44	\$13.37	QE P	100		
March Marc	Compact March Ma	1	THE SELLING CHARLES THE SELLING THE SELLIN	SOMEC	8 23	┡	9563	ş	Ş	E			
March Marc	Service Order - 1	1	Trans. Demon Cross Automated Electronically, per USA - Diecornect	SOMEC	≨	╄	4M	1		3	3	2	2
Marie December South No. South N	Service Order - 16 16 17 17 17 17 17 17	#	MMC - Berytol Order automated Manually, per LSR	SOLAN	MA	+		2	٤	3	ž	ž	
Service Ober Arial Service Ober Ober Ober Ober Ober Ober Ober Obe	Service Other Land County 1970 14.9 10.0 14.0 19.0 14.0 14.0 14.0 14.0 14.0 14.0 14.0 14		IMRC - Service Order automated Menuelly, per LSR, Dieconnect	Taylor Co		4	٤.	17.00	§	2	≨	ž	180
Colore March Colore Tribo March Marc	Compare Sub-Legge From the Compare Compare Compare Sub-Legge From the Com		MRC - Incremental Charge - Manual Service Order - 1st	111111111111111111111111111111111111111	٤	4	٤	X C	ž	Ę	≨	¥	180
Compact Section Control Contro	Control Cont		MMC - Incremental Change - Manual Service Order - Adm	20000	2	4	2	ž	\$18.14	OBT.	₹	ž	180
Ober Confidence in Fig. 1 Control Cont	Cooker Controlled to Fig. 17 17 17 17 17 17 17 17	E	MRC - Incremental Charge - Manual Servine Order - Demonstra	S CHANGE	08	4	2	Ş	80 OB	TBO	₹	≨	180
Comparison	Cooper back-toop per mode CCSAX TRO STAT TRO	t	NFC - normanial Orana - Manual Arter Presidentes	SOMM		4	TBO	¥	\$11.41	SE OB	2	ž	GEL
Color Colo	Colore C	Ŀ	Distribution over 1 Wiles Land of Contract of Contra	CERT	\$51.29	_	087	\$16.31	\$16.29	QEL	25.25	54.53	Cat.
CLEAN TRO STATE TRO	CLEAR TROP STATE TROP STATE TROP STATE TROP STATE TROP STATE TROP STATE TROP STATE TROP STATE TROP STATE TROP STATE TROP STATE TROP STATE TROP TROP STATE TROP STATE TROP STATE TROP STATE TROP STATE TROP STATE TROP STATE TROP STATE TROP STATE TROP STATE TROP STATE TROP STATE TROP STATE TROP STATE TROP STATE TROP STATE STATE TROP STATE STATE TROP STATE STATE TROP STATE STATE TROP STATE STATE STATE TROP STATE		Total has most	CSAX	808	_	CBT	¥	ž	3	44	3	
UCSAY THO \$11.55 THO	UCSAY TRO S112 TRO S112 TRO S112 TRO	†	2000 0 000	CSAX	180	L	480	\$10.66	\$10.44	TES.	*1		5
USSAT 110.00 11	UCSAY 110 11	‡	Pros. 1 care months	UCSAX	180	⊢	2	12.63	\$13.53	ş	9		200
USSAY 116.04 11	U.S.M. 1.03.4 1.0	#		NCS4X	9	-	Q	77 75	20.17	3 4	3	٤	3
USSAY \$110.50 1	UCSAY \$16.50 119.00 116.50 11	1	COLUMN AL DEL MICHEL	NC34X	≨	-	2	¥N	4	3	3	§ :	2
USSAT 1965	UCSAY \$10.00 \$1	1		XXXX	\$161.69	L	g	4166 A7	27.0	3	2	ž	180
Colore Strict S	Control Cont	1	MAC-Add	XXXX	3	1.		100		3	12.24	23646	8
March Marc	Company Comp	1	MRC - Decornect Charps - 1st	173	26 CH 27	1	2 2	000	Ç.	082	82.38	\$97.61	Œ
Comparison	Marchelle, per LSR		IMPC - Diezonnect Cherps - Adril	A7921		1	3	3	2111.78	2	\$78.56	Ş	180
March Marc	Control Cont	Н	MRC - Service Order submitted Electronically, per LSR		200	1	3	808	818.36	180	\$13.53	¥	180
Mail per List Mail per Lis	Section Code Action	Н	NYC - Service Order pubmined Electronicals, ner (SR - Diegyman)	2000	2	e X	S.	93.53	8	180	05 E\$	8 8	081
March Cheer LEAT Decorated SOMAN	Service Octor - 148 SOAMH	\vdash	MRC - Service Order submitted Manually, ner (SR	3	Į.	2	≨	ž	¥	180	¥	≱	3
Section Colors 18 19 19 19 19 19 19 19	Service Order - 1887 Service O	\vdash	INPC - Service Order authoritied Manuals, per 159 Plantonia	2	Ş	8	ş	\$29.24	¥	OBT	₹	ž	TBO
Series Code: Addition	Service Order - Add)	۱	NVC - Incremental Charge - Marviel Service Order - 1-4	2	ş	3	ş	2	Ş	180	Ş	ž	180
Second Color Decembed South TBO	Section Code Description School Teach Teach Teach Teach NA Teach T	\vdash	NRC - Incremental Change - Manual Service Order - A44	2	2	ž	2	Ş	\$18.14	180	¥	ž	180
Company Comp	Cide Coordination pair top Cide Coordination pair top USBRZ \$1.30 \$1.81 TRD NA \$1.14 TRD Code Coordination pair top USBRZ \$1.30 \$1.81 \$1.50 \$1.69	H	NAC - Incremental Change - Manual Service Code - Changes	SOM N	082	ž	OEF	₹	\$0.88	180	¥	≨	TBO
Control Cont	Control Cont	H	MMC - Incremental Charge - Manual Conference Conference Conference	SOMAN	2	≨	Œ	Ş	17'11\$	08 1	ž	ş	1BO
Company Comp	USBRZ \$113.00 \$113.0	1		Sales	\$51.29	\$16.31	OE.	\$16.31	\$16.29	180	26.32	45.43	2
USBRZ \$11.50 \$1	USBRZ \$113.00 \$113.00 \$103.01 TDD	1	MACC. 14	Saga Cagas	\$1.30	23.67	19.18	\$3.23	81.69	180	9 53	3	
USBNZ \$50.04 \$11.05 \$10.01 \$10.00 \$	USBPR \$59.94 \$59.94 \$59.95 \$11.10 \$10.11 \$1	+		ZE BES	\$113.60	\$113.62	00'4618	\$113.61	\$108.11	180	\$117.00	6160 63	20 021
Control Cont	Colorable part SA	╁	MOC. Charles	Zaesn Ceers	\$50.44	9C 9C\$	\$41.59	28.36	21.58	QEL	27.30	67,700	3
Control (a) EST (a) \$11.00 \$19.17 \$13.44 \$13.87 TBD \$10.67 \$10.	Control Cont	+	MOC. Manual A. Carrier A.	2882	\$99.54	\$26.49	\$115.86	\$101.18	\$10.08	Q.	27.87.2		
Control Cont	SOURCE SESSION SESSI	╀	The contract of the contract o	CAB231	\$14.05	\$13.06	\$19.17	\$13.44	8138	SEL	640 64		20.00
March Marc	## 1250 Discovered SOAMY NA \$12.54 NA NA TED	+	MICH. CANADA CALLE MATTERED CHESTON, DR. LSM.	SOMEC	\$3.50	22.75	\$3.50	85	95 53	(BC)	92.53	5 5	1
Part Part	March Marc	+	MAC SERVICE COM MATERIAL CHESTONICARY, DRI LISH - DECORNECT	SOMEC	Ş	27 OS	ž	≨	≨	Œ	Ž		2 2
SOURCE S	SOMAN NA KISA NA KISA NA KISA NA KISA KISA KISA TEO	+	MED. Sentra Order a American Land.	SOMAN	ž	95123	Ş	23 24	¥	OB.	ž	2	8 0
SOLICIO SOLI	SOLICE Chest-karrier SOMAN TBD	+	N.C browned Course, the set of	SOMM	¥	23 64	Ž	200	Ş	88	ş	Ž	GEL
### Solution SOLAM TBD NA NA NA NA NA NA NA N	SOUND SOUN	╀	MGC - bryandental Channel Control Control Control	SOMAN	180	¥	Ş	ş	11811	180	ž	ž	Ž
Company Comp	Control of the Cont	+	ASC. Investorial Chairs of the Control of the Contr	SOMAN	9	ž	¥	ź	80.93	9	ş	ž	Ž
Coulob W mining per more USBNA E17.17 E17.10 E17.25 E17.21 E17.25	Coulob, 178 smile, per mon. COSBMC S1129 S1121 S1122 S1121 S1122 S1121 S1122 S1121 S1122 S1121 S1122	╀	MC: Ingeneral Chane, Manual Control Control	SOMAN	Ē	ž	¥	¥	\$11.41	180	ž	ž	ž
Control of the cont	Content of the cont	19	Accomplished the Cable (see seed.) As	USBINC	82128	\$16.31	13422	\$16.31	\$16.29	OEL	145.34	25.62	888
USBN4 \$107.07 \$185.0 \$	USBN4 \$1777 \$188.0 \$18	-	NRC - 1st	THE COURT	4	82.28	8	\$6.29	1283	9	5.53	22.23	55.03
Comparison	CARC Confined con	+	NRC - Adri	HOOD.	11/21	0.32	8178.48	\$128.10	8118.69	TBO	\$127.67	\$204.20	39.0818
Control Cont	Control Cont	H	ARC - Decorred Charse - (at	NOON.	2	ž	1288	28.2	57.73	180	\$50.82	09.095	\$60.47
Out-Mily, part LSR Colored by the contract Colored by the colored by	Original LRT Colored (1.50) 61.50 61.61 180.08 81.61 180.00 81.61 180.00 81.61 180.00 81.61 180.00 81.61 180.00 81.61 180.00 81.61 180.00 81.61 180.00 180	1	NFC - Decorned Chans - April	100	\$10/45	\$104.31	\$122.17	¥ 69:3	\$101.65	180	\$78.71	Ş	\$135.88
Compact Comp	Company Comp	╀	WPC - Service Order automated Bactering also new 1 RB		814.30	\$17.15	\$1957	818 06	\$14.61	180	\$10.00	ž	20.78
March Marc	10	H	WIC - Service Order a Aprillacy Description, par 1 60 - Number 1	SOMEC	8	E C	350	05.03	05.03	081	05:03	80	35
March Marc	March Marc	+	MC - Sevice Onice a doubted Manage, see 1 60	200	ž	20.02	ž	ž	¥	OBJ	Ş	Ę	780
### SOLUTION SOLUTIO	### 1894 1894 1895	H	ASC. Service Order to Ambred Use of 18th Part Land	SOMAN	ž	\$21.56	\$	\$29.24	ž	98	ž	ž	8 518
CACC Described by Betty CACC Described	### STATE OF THE PART OF THE P	╀	MEC. Investmental Character Manual Section Chairs	SOMAN	≨	3	ş	15.54	ş	081	Z.	≨	8
### NA NA NA NA NA NA NA NA NA NA NA NA NA	## 50 Part	+	MC-Incomence Chanse Mana Sector Onto - And	SOM		ž	≨	ž	\$18.14	081	Y.	ž	¥
Control Existing set up USBNZ \$11.25 \$14.21 \$14.21 \$10.41 TBD NA NA Fige Total Existing set up USBNZ \$15.29 \$16.31 \$10.41 \$10.31 \$10.31 \$10.31 \$10.31 \$10.31 \$10.32 \$10.31 \$10.31 \$10.32 \$10.31 \$10.32	Control Cont	F	MC - Incremental Chema - Manual Course Order - Processing	SOFTWA	8	ž	¥	ž	80 85	OBL	¥X	ž	¥
600 - G.E.C. Dustrbuson Facility set-up U.SBFW 517.43 711.78 5421.08 427.16 8439-64 TBD 45.34 510.15 610.15	6001 - CLEC Districtation Facility set-up USBFW 517.45 711.78 5421.08 627.16 545.25 1100 1000 - Per 25 pets printed set-up USBFW 517.45 711.78 5421.08 627.16 545.95 1100 1000 - Per 25 pets printed set-up USBFW 517.45 711.78 5421.08 627.16 545.95 1100 1001 - Per 25 pets printed set-up USBFW 517.45 711.78 5421.08 627.16 545.95 1100 1001 - Per 25 pets printed set-up USBFW 517.45 71.78 5421.08 627.16 545.95 1100	F	MPC - Ingremental Change - Manual Order Coordination - one from	1	2	≨.	ž	≨	\$11.41	TBO	MA	ž	≨
600 - CLEC Describation Facility set-up USBFW 517.43 711.78 \$421.08 42.74 180 4809 5101.5 601 - per 25 pair primed set-up USBFY 44.87 45.52 807.16 42.74 43.74 43.74 18.77 8421.08 127.16 843.94 180 4809 5101.5 602 - CLEC Describation Facility set-up USBFY 517.43 711.77 8421.08 127.16 8439.94 170 4809 5101.5	600 - CAEC Destrution featility set-up USBFW 517.43 711.78 5421.08 627.18 5439.44 TBD 600 - CAEC Destrution featility set-up USBFW 517.43 711.78 5421.08 627.18 543.27 TBD 600 - CAEC Destrution featility set-up USBFW 517.43 711.78 5421.08 627.68 543.29 TBD 600 - CAEC Destrution featility set-up USBFW 517.43 71.78 5421.08 627.69 543.78 FBD 600 - CAEC DESTRUCION FEATILITY 5421.08 627.69 543.78 FBD 600 - CAEC DESTRUCION FEATILITY 5421.08 627.69 543.78 FBD 600 - CAEC DESTRUCION FEATILITY 5421.08 627.69 543.78 FBD 600 - CAEC DESTRUCION FEATILITY 5421.08 627.69 543.78 FBD 600 - CAEC DESTRUCION FEATILITY 5421.08 627.69 543.78 FBD 600 - CAEC DESTRUCION FEATILITY 5421.08 627.69 543.78 FBD 600 - CAEC DESTRUCION FEATILITY 5421.08 627.69 543.78 FBD 600 - CAEC DESTRUCION FEATILITY 5431.78 FBD 600 - CAEC DESTRUCION FE	F		3	2 2	\$16.31	27.23	\$1631	\$16.29	180	\$534	25.63	88.58
601 CLC Destruction Facility set-up USBFW 517.43 711.78 \$421.00 627.16 \$659.04 150 480.09 51015 601 per 25 pair paramet set-up USBFW 4.4 ft 4.5 25 87.10 44.70 6.27.16 \$659.04 1700 44.8 18.10 4.0 17.10 17	600 - CAEC Distriction Facility set-up USBFW 517.43 711.78 \$421.00 427.16 \$439.44 180 and the control of the co	擅	OOF FEEDEN										
801 CLEO Distribution Figuility set-up USBFY 617.43 711.78 8421.00 627.16 845.02 180 45.09 510.15 81	800 - GLEO benchaeor Facility set-up U-SBFY 517.43 711.73 \$421.06 (27716 \$439.44 180 180 180 180 180 180 180 180 180 180	ō	No Box Set-Up			1	1						
801 per 25 pair pervis letter 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	601 par & pa	L	NFIC - DS0 Set-Up per Cross Box tocation - CLEC Distriction Facility set-up	NG65W		2	12.5						
ton CLEC Dertoution Facility setup (LSBPY 517.43 711.77 \$421.00 627.16 \$639.64 TBD 45.09 510.15	ton CLEC Destruction Feating setup USBPY 517.43 71.78 6.27.10 6.37 16 6.39 60 TBD 600 PP pair pair pair pair pair pair pair pair	Г	NFC - D50 Set-Up per Cross Sick location - per 25 pair cernal set-up	A CORD	3	2 2 2	8	67.18	25.39 6.3	8	28 08 00	51015	99
Marie Carrier 1980 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	407 per pair pair per per per per per per per per per pe	F	MPC - DS1 Set Lib car Cross for incedem - CI FC Destruction Explained	Y LOCK	•	45.26	01 /94	45.28	252	<u>6</u>	45.04	2	8
	00 10 10 10 10 10 10 10 10 10 10 10 10 1	£	MRC - IX1 Km I in not Order for includen, one not never an	1	517.03	211.78	861.28	627.16	85.39 GE	TBO	60 96 9	\$10.15	9

MILLEOUTHWATWACUTH AATES METWORK ELEMENTS AND OTHER BETWACES

1	DESCRIPTION	20\$0	1	-	100	*	RATES BY STATE	[
4	2-Whe Analog VG Ground Start Unburnded Buth one Send						5	2	¥	S	Ē
	Zone 1, per month	LKSBFA	\$13.66	ž	88.58	≨	ž	4	1	120	
	Zone 2, per month	LS9FA	06	\$10.75	1	\$10.36	10 115	4	5 2	8002	ž
	Zone 3, per month	USBFA	180	\$11.57	2	2118	2	4	×	ž	180
E	Zone 4. car month	USBFA	Œ	\$13.51	5	99 91		4	10	ž	180
E	WCC 14	USBFA	≨	ž	¥	1	2	4	\$16.11	Ş	180
E	WC.Ad	USBFA	27 96 18	3180 60	77 800	1000	2 2	4	≨	≨	180
E	MBC: Designed Chart it	VZ8EA	\$118.64	W 11.5	W V.	6.7	3	4	\$122.62	\$199.27	OBL D
	WC-Degree Over 144	LISBEA	\$119.55	8116.59	GE	8 0110	8		1997	\$60.33	180
E	MC Sevin One of the Control of the C	USBFA	27.01	26.70	9	2	10.57	1	8,7,8	ž	180
E	acrecia, per LSA	SOMEC	95 63	S X	200	200	92.0	1	814.68	¥	980
E	MAC: Sevine Once 1	SOMEC	Ş	3		2	8	1	85.55	83.50	180
E	Man Control Annual Manual Control Cont	SOMAN	NA.	23 100		٤	ş	4	ž	ž	382
E	Mary Leaves Marriage Marriage, per LSR, Discornaci	SOMAN	V.		٤	27.5	ž	┙	¥	ž	2
Ī	Annual Course Manual Service Order - 1st	S	5	2	<u> </u>	20.00	≨		ž	ž	9
Ī	ANY INCOMPANIE CHAPTER Service Order - Addi	S	É			ž	\$18.14		200	ž	180
1	PATC - PETERSON CHARGE - Manual Service Order - Decornect	1		٤	2	Ş	90.53	L	\$12.76	ž	6
1			3	ž	9	¥	\$11,41	L	\$15.12	1	3 6
4	The Analog VG Lace dear Unbunded Sub-Loop Feeder, per month	900		1				L			2
1		200	2	Ş	2	¥	¥	2	ş	110.00	7.7
	Zone 2, per month		2	\$10.75	9	\$10.34	\$11.01	2	20 15	1	Š
1	Zone 3, per month	2003		\$11.57	8	\$13.62	\$13.36	180	\$34.10	2	
1	ZONO 4, per month.	200	3	1221	2	\$19.69	\$5128	180	\$18	M	
	MAC: 14	a constant	ž	ş	ž	NA.	¥	9	ž	17	3 4
	MRC - Add	2003	2	\$183.62	208.44	1192.57	1122.30	8	1120	1100 27	
1	NRC - Decorned Charge - 1st		8	31138	\$170.05	\$111.96	\$111.80	8	17 975	14.43	3
1	NRC - Decornect Charge - Adril			\$116.59	780	\$119.14	\$118.57	8	00 1/2	17	26
1	NRC - Service Order a Combined Electronically, per LSR		2	28.2	9	£26.79	\$28.66	180	314 64	1	2 12
	MMC - Service Order automitted Electronically, per LSR - Discorners		8	22	23.50	83.50	85.55	1BO	Ş	43 63	3
	NPC - Service Order externised Manually, per LSR	2	≨ I	20	¥	ž	¥	180	¥	2 2	
	NHC - Service Order automitted Menually, per LSR, Decorrect	S CHANK	§	21.58	¥	23 24	ž	9	ž	1	26
	MRC - Incremental Charge - Menuel Service Order - Les		≨ Ž	13.04	Į	2	ž	2		5	200
	NFIC - Incremental Charge - Manual Service Order - AARI	SOMAN	2	2	18.94	ž	\$18.14	S S		\$	3
	NAC - Incremental Cherge - Menuel Service Onte - Decomposition	SOMAN	98	¥	28.42	ž	5	2	100	٤	08
		SOMAN	8	₹	98	ž	\$11.4	9		5	2
i i	To Anadog VO Reverse Bettery Use Sub-Lane Coult.									٤	2
	Zone 1, per month	2	813.68	¥	87.88	ž	2	A P	4.3	2	
	Zone 2 per month	USBFC	<u>6</u>	\$10.75	180	\$10.36	10118	9	69.53	3	≨ ¢
	Zone 1, per month	2000	2	\$11.57	£	113.62	81336	8	814.10	1	
	Zone 4, per month	2	2	\$13.51	TBIO	\$19.69	2512	2			
	NAC - 1*	USBFC	ž	¥	H	¥	≨	2		£	3
	MAC: Add	SEC.	_	\$180.62	-	\$192.57	00 2018	2	613563	5	ž
	MRC - Decomed Chans - In	2		\$113.00	t	361118	9	8		/ / /	3
	WAC - Decorract Charge - Add	USBEC	8119	\$116.59	081	\$119.14	\$118.57	9	8		3
	MIC - Savice Order automised Electronicals, par 1 55	2	-	26.70	۲	\$ P.	28.85	9	1271		
	ARC - Serios Order automitiand Decreasing our ISB - Decreases	2	4	27.73	Н	83.50	88	180	Ş	59.63	3 6
	MMC - Service Order automated Menuelly, per LSA	2000	1	200	Н	ş	Ş	180	Ş	2	3
	MRC - Service Order eutomitted Menually, per LSR, Discornect	2000	†	8	7	29.24	ž	98	ž	ž	2
1	MMC - Incremental Charge - Manual Service Order - 1st	NAMO	Ş	2	+	2	٧¥	TBO	ž	ž	180
+	PAC - Incremental Charge - Manual Service Order - Add	SOMAN	1		+	ź.	818.14	0	126 54	¥	9
#	ATC - INCREMENDA CANDO - Manual Service Order - Discornaci	SOMAN	98	1	149	٤	8	9	\$12.76	¥	180
1					1	٤	17.17	2	21.511	¥	TBO
	Cons. 1. der morth	USBFD		3	16 618	+	172		13	1	
	Zone 2. der fromfi	CARRA	Н	\$23.38	1	+-	85.53	1	1000	77.	ž
	Zone 3, per month	CE SS	9 <u>8</u>	27.22	180	\$36.12	87.23	8	98.5	£ 2	3 5
	Zone 4, per month	210	1	1503	ı	⊢	19.5	98	141.37	¥	3 6
	WC-18	CLESS	+	Н		-	ž	8	Ş	Y.	3
	W.C. Adri	2	┪	Н		-	49023	180	┿	14 67-63	2
	APC - Decornect Charge - 1st	2	┪	-	- 1	Н	\$136.47	8	╀	691.69	1
	MC - Decorrect Charge - Add	CE SEL	4	15.21		-	\$135.74	08	╁	Y.	
	MC - Savice Order automitted Electronicals, par 158	CERTO	+	Н		1	24.23	2	200	1	
		SOME	4	Н	_	⊢	35.55	180	5 5	150	
		SOMEC	4	Н		_	ş	8	2	3 2	3 2
	MPC - Sevice Order automitted Manually, per LSR, Discornaci	1	¥.	8	ş	Н	¥	88	Y.Y	Ž	3
-	#1C - Incremental Charge - Manual Service Order - 1st	AVIACA	4	+	7	_	٧×	180	¥	ž	180
-	M.C. Incremental Charge - Manual Service Order - Addr	MANOS	4	+	7	_	\$18 14	<u>8</u>	¥	ş	180
-	M.C Incremental Charge - Manual Service Order - Discornect	SOLLAN	4	t	т	_1	88	200	¥	Ž	98
	M THE STATE OF THE		╀	╁	7		\$1141	2	¥	¥	180
	Average Viol Loop-Mart Unfounded Bub-Loop Feeder, per month	USBFE	20,52	¥	16 613	2	422	6			
				1			1	-	\dashv	21072	¥

						1	BATTE BY CTATE				
\$	DESCHAFTION	nsoc	₹	2	П	¥	ıls		3	5	77.
1	COMB 1 per month	USBFE	TBO	SE 523	OB1	\$30.69	\$25.90	題	16123	2	N C
1	COURT, DEF PROPERTY	USBFE	TBO	27.94	1	\$36.12	\$27.00	9	23.5	ž	2
+	COME 3, DEF FROM	USBFE	9	15043	Г	\$22.90	1952	8	1	12) cat
+	ZOTB 4, par month	USBFE	ž	≨	T	¥	2	£	1		
<u> </u>	N-C IR	USBFE	12723	17.22	T	\$221.19	2220.87	ā	35 X	27 67.03	5 5
1	i sovi - Sand	USBFE	\$142.45	\$140.22	Г	\$138.67	\$129.47	TRO.	6144 28	100	36
	MFC - Discornact Charge - 1st	USBFE	\$131.15	\$127.64	Т	\$170.64	4136.74	ķ	1 20	2	2
	MFC - Discormect Charge - Add?	USBFE	\$3.1.66	16 253	Т	£10 69	X	GE .	2 2	2 3	200
1	PMIC - Service Order automated Electronically, per LSR	SOMEC	35.53	27.03	Т	25.50	S	2	100	5 5	2
	MMC - Setvice Order automitted Electronically, per LSR - Decorved	SOMEC	≨	20.03	Т	V.		1		3	
	MMC - Service Order automitted Menually, per LSR	SOMAN	ž	95.103	Т	2003	1	2	٤	٤	
	MRC - Service Order authoritied Menually, per LSR, Disconnect	SOMAN	ž	200	Т	200		1	2	٤	Z X
	MHC - Incremental Charge - Manual Service Order - 1st	SOMAN	9	*	Т	1		3 8	٤	٤	2
	NRC - Incremental Charge - Manual Service Order - Addi	SOMAN	9	YN	Т	1	2		٤	٤	2
	MRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	2	12	§ 3	2 2	8	2	ž	≦:	180
					т	5	11.0	2	٤	٤	9
7 M	e ISDN Unburded Beb Looe Feader, ser storth	110000	77 152	117	****]	ļ				
		I Chee		2 50		ž	ž	Š	ž	224.33	¥
E	Zona 2 ger repres	10000	3 6		3	2///	1	3	23.62	≨	180
	Zone 3, per month	HODE		3 2	2 2		3	2	2	ž	2
	Zona 4 per month		3	70.15	3	300	27.7	2	36.27		TBO
E		1	ž	×	ž	ž	≨	087	≨		¥
	W.C. Ach		2	777	05 BOX	888	\$21859	9	500 00		180
	MPC December Change 148	100	5	27.072	E 23	\$136.39	\$136.19	2	\$105.88	_	đđ
		USBFF	8.0	\$127.64	\$119.68	\$121.40	\$120.62	G B1	15.263	L	CBL
I		LESBFE	£3.5	16 223	\$29.58	10.92	16.55	130 081	821.58	1_	180
-	Conversion of the Conversion o	SOMEC	99 23	22.78	\$3.50	83.55	85.53	780	95 63	<u>ı</u> .	É
-	MYC Beryto Under Blommed Electronically, per LSR - Disconnect	SOMEC	٧V	\$0.42	Ş	ž	ž	180	2	MA.	TRO
	THAT - SOUTH BATTANED MATLENY, per LSA	SOMAN	ž	85128	≨	2324	ž	TBD	YN.	_	SET.
	WHICH SHAVE ONE SAMMEN METABY, per LSR, Decorrect	SOMAN	≨	200	2	3	Ą	Va.	12	_L	35
	NRC - Incremental Charge - Manuel Service Order - 1st	SOMAN	98	¥	2	¥		3	1	_£.	3
	ARC - Incremental Charge - Manual Service Order - Add1	SOMAN	9	2	ž	2	2	1	5	_I	2
	NPC - Incremental Charge - Manual Service Order - Discorpance	SCALAN	ğ	1			8	3	5		3
			3	٤	٤	٤		3	Ş	- 1	180
+	4-Whn Dâl Unbundled Sub-Loop Feeder, ner grouph	KRES	+		Ka						
	Zone 1, per month	KARA	Ť	W 700		5	2	Ş	2	٤	ž
	Zone 2, per morth	INGREG	+	3		20.07	19 19	3	20 65	2	2
	Zone 3, per month	IKBEO	+	200		5	/0 700	8	8	2	2
	Zone 4, per morth	IKBEO	+		3		200	3	3/8/5	2	2
	ZC - 7Z	2000	+	2	٤	ž	٤	8	≨	ş	ş
	WEC-AM	2000	+	8	8	8	2209.47	TBO	\$390 01	1 <u>B</u> 0	180
	Mary Character Comment Character Comment Character Chara	D. POSO	7	2002	9	\$127.11	\$127.07	180	\$163.37	9	180
Ï		USBFG	-	\$127.76	TBO T	\$130.43	\$135.74	180	¥X	92	9
	MHC. URCOTHEC CAMPS - ACCT	USBRO	Н	\$33.06	2	830.59	24.23	OBL	WA	TBO	G
	WHI Service Order Bushmided Electronically, per LSA	SOMEC	-	27.23	8 23	05.53	53.50	TBO	95 63	911	9
	MHC - Service Order pubmitted Electronically, per LSR - Discornact	SOMEC	-	27.03	ž	ž	¥	28	Ž	¥2	1
	MFIC - Service Order automated Manually, per LSR	SOMM	Н	\$21.58	ž	228.24	ž	9	ž	Ž	Car.
	PRIC - SHIVES CHOR'S RICHARDS Manually, per LSR, Discornact	SOMAN	Н	10.00	ž	1	ž	8	ž	¥	9
	M-C. HETWENTER CANDS - Manual Sevice Order - 14	SOMAN		MA	2	ž	\$18.14	180	\$42.10	180	OB.
	MAN CONTRACTOR CONTRACTOR DESIVERS CARDIN CARDON	SOMAN	Q	¥	OB1	ž	88	9	\$12.76	180	192
	ATC - EXPERIENCE CARROL - MATLE Service Order - Decorrect	SOMAN	\dashv	į	130	ź	\$11.41	180	ž	8	180
2	Conner Underwiter Sand Ann Starter are seen	1000	7		- 1						
	1. Der month	LL COLOR	e Out	¥ .	- 1	ž	ž	ž	¥	\$10.78	ş
	Zone 2, per month	- Febru	+	24.03	ŧ	2 42			210 66	Ş	9
	Cone 3, per month	70	+		1	200	***			ž	2
	Zone 4, per month	FEBER	╁	1	1	3 42	2		2	2	3
Ē	#C-1#	I SAFA	╁	1 K.	ı	¥:0	200	3 5	2 2	2	≨ ¦
	UPC - Add	Heasi	t	23 65	1	40.00	200	2	8000	3	2
	MC - Discornect Charge - 1st	K-BE-H	٠	27113	1	74.77	11.12	3 5		8	2
	MC - Disconnect Charge - Add	HOUSE.	+	72.00	ı	27.75	1	3 6	1700	\$	2
		0.00	+	200	ł	2 5	2 2 2	2	210/2	Ş	2
		2	t	25	ł	3 3	2 2	2	2	8	2
		SOMAN	t	75	1	2000	٤ :	2	ž	≨ :	2
	4P.C Service Order eutomitted Menually, per LSR, Discornact	SOME	t	1963	1	J G	ž	3 2	Š	5 3	3 5
		SOMAN	t	ž	1	2	\$1814	9	1	£ 12	3
		SOMAN	Н	¥	1	ş	88	2	HA	3	2 2
#	MFIC - Incremental Charge - Manual Service Order - Discornace	SOMM	180 1	≨	180	¥	\$11.41	180	¥	ž	180
1	PFC - Incremental Charge - Menual Order Coordination - per toop	USBAC	Н	816.31	H	1631	\$16.23	9	245.54	35.53	180
	Comment Indiana de La Lacar Contact de La Co		+		ı				Ī		
	The terminal authors, per month	USBFJ	\$18.94	≨	¥	≨	MA	¥	¥	\$15.63	ž
1	The House	USBEJ	┥	6502	TBO	816.55	95 028	8	\$1466	¥	00

And the sent inves

MELLBOUTHMETWROUTH RATES
NETWORK ELEMENTS
AND TITLES OF BROADERS

#	DESCRIPTION Zone 2 per month	OSO	7	Ш	QA	2	MIES BY STATE		2		2
Ė	Zone 3 Der month	USBE	180	\$21.48	OBT.	\$15.35	\$14.96	8	223 74	2	Ser
Ė	Zone 4, Der fronth	USBE	<u>e</u>	_	OEL	\$12.52	\$13.15	8	\$27.38	ž	2
	NPC - 18	Sas:	≱	_	≨	≨	¥	9 <u>0</u>	2	¥Z	ž
	MAC. Add	- Car	200	_	9	500 500 500 500 500 500 500 500 500 500	\$209.31	TBO	\$207.14	\$294.59	180
1	MRC - Discorract Charge - 1st	USBE.	1220	٠.	3 5	212	\$126.91	2	23.7	\$152.62	180
#		USBEJ	23.623	1	1	17076	\$126.36		\$30 12	ž	180
#	å,	SOMEC	82.55	1_	95 125	9	200		1	ž	2
#	MRC - Service Order excepted Electronically, per LSR - Discornect	SOMEC	ž	11	ž	≨	₹		2 4	2 2	2 2
E	NAC - Service Order sucmitted Manually, ner (SR Decrees)	SOMAN	ž	Į	ž	\$29.24	¥	8	₹	ž	9
	MMC - Incremental Charge - Manual Service Order - 1st	SOMAN S	≨ ģ	2	≨	82	ž	5	₹	ž	9
1	MRC - Incremental Charge - Manual Service Order - Addi	SOMAN	2 2	•	3 6	≨ :	\$18.14	2	¥	ž	ŤBO
1	NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	9	1	3 2	2 2	8		ž	ž	180
	PWIC - Incrimental Chiege - Manual Order Coordination - per top	USBINC	\$61.29	1	1	\$16.31	\$18.29	2 2	≨ } 3	¥ ¥	B §
Ī	4-Mrs 2.4 KBPS Deltal Unbandled Sub-Local Seater per security	200									3
		Y DES	/6 DC	ž	ž	≨	۲×	≱	¥	17.77	≨
	Zone 2, per month	USBEK	2	2 2	2	2,2	27.18	9	1,923	ž	TBO
1	Lore J. per morth	USBFK	180	\$29.65	2	24.47	2 ×	3 5	5 60	≨	TBO
E		XJesh	Ş	¥	≨	≨	ž	18	×	2 2	2
E	MC. Add	LSB-X	\$212.90	द्धाव	<u>6</u>	250977	17 6023	8	2015.00	\$309.57	\$ 9
	NRC - Decorned Charge - 1st	Y SBPK	\$131.14	128 81	180	\$127.28	\$127.07	<u>1</u>	28:20.19	\$157.90	3
	L	SBEK	27.72	10/21		23.53	\$135.74	2	05 963	ž	180
	MRC - Service Order automated Electronically, per LSR	SOME	9 2	2 2	3 5	3 2	27.33	8	25.82	¥	180
1	MRC - Setrice Order autimated Electronically, per LSR - Discornect	SOMEC	ž	2	2 ₹	2 2	2 2	2	3	8	8
\pm	MIC Service Order submidded Manually, per LSR	SOMAN	ž	85158	≨	28.24	2	ğ	ž	X	2
E	MRC Insumental Charms - Maria Sector Note: 154	NAMOS	ž	20	¥	a E	ž	2	Y.	1	2
	NAC Incremental Olarge - Manual Service Order - Add	SOMAN	9	≨:	OB1	ž	\$18.14	180	¥	≨	2
	MRC - Incremental Charge - Manual Service Order - Discornact	SOUTH A	3 5	Ź.	9	ž	\$6.08) Oğl	ΝĄ	Ş	TBC
		Seemon .	3	٤	8	≨	\$11.41	2	¥	ΝA	TBO
1	+Why 4.8 KDPS Digital Unbundled Sub-Loop Feeder, per month	USBA.	4,063	¥	2	¥	12	VIII	172	\$7.55	
I	Zone) per mode	HBSH	130	\$24.88	OBT.	\$27.38	27.16	9	26.75	1	ž á
E	Zone 3, per morali	USBA.	œ	2012	180	133.41	24.50	8	£ G	€ ≨	2 2
	Zone 4, per month	Less:	Q2 :	200	OBJ.	25.47	\$25.06	180	\$50.83	≨	2
	MIC: 184	1983	¥ 6:60	¥ .	≨	≨	≨	99	ž	ź	ş
	WRC - Add	USBR	131.12	¥ = X	3 2	2008.17	4 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	2	\$215.00	49 6003	TBO
1	MRC - Discorrect Charge - 1st	USBR.	\$131.15	\$127.64	3 2	1111	12/0/	3 5	28 25 25	\$157.93	8
Ī	MAC Series Add	USBR	\$34.68	16 22	8	23.00	27.73	8	2 2 2	ž	
İ	MAC-Service Order automated Personalists over 1843 - Navonalist	SOMEC	95.53	52.2	3.50	\$3.50	63.53	180	83	80	
	NATC - Service Order automated Manually, par LSR	2000	≨ :	2	ž	≨	≨	TB0	¥	ž	180
	MRC - Service Order automitted Manually, per LSR, Decorned	SOLAN	2 2	2 2 2	≨ i	282	≨ :	<u>1</u>	ž	ž	18 0
1	NVC - Incremental Charge - Manual Service Order - 1st	SOMAN	8	ž	2 2	2 3	2 2	3 5	£ 5	≨ :	92
1	MAC Incremental Charae Macual Service Order - Add	SOLLAN	180	¥	180	≨	88	2	ı ş	٤ ≥	2
		SOMAN	8	¥	180	≱	\$11.41	180	≨	ž	3 2
1	4-Whe 9.8 KBPS Digital Unbended Sub-Loop Feeder, per month	USBFW	45003	¥	VIV	12	1				
‡	Zoos S	USBFM	OBT	87.53	2	X (2)	27.75	2 2	2 2	× × ×	≨ g
ŧ	Zone 1. oer more	USBFIA)BO	129 P.C.	9	17.53	224.93	9	10.55	¥ ¥	2
	Zone 4, per month	CSBLA	OS :	2 23	<u>6</u>	15.4.7	\$25.06	Q E	\$50.60	₹	200
	NATO-100	NS850	20.00	¥ ::6	≨ É	≱ \$	≱	2	ź	ź	ž
#	MRC-Add	USBFM	\$131.14	128 81	2 2	127.28	127.07	3 5	88	25 802	8
1	MSC. Departed Chara. Add	USBFM	\$131.15	\$127.64	98	13.23	\$135.74	28	05 863	X X	9 9
ŧ	WRC - Service Order submitted Electronically our LSB	USBFN	\$34.66	16 ZZ 9	2	83 ES	24.2	OEL.	29.62	¥	3 2
	NRC - Service Order submitted Electronically, per LSR - Discornect	SOME	2 2	2 2	82	82	8	180	3 2 3	3 09 03	180
#	NARC - Service Order extended Manually, per LSR	SOMMAN	5 3	20	2 2	¥ 000	X 5	2	¥:	≨ :	180
#	WEC - Service Order pubmitted Manually, per LSR, Disconnect	SOMM	¥	1803	ž	200	٤٤	3 5	¥ 3	\$ 5	O K
ŧ	MRC - Promised Once - Manual Service Once - 1gt	SOLLAN	081	ž	2	¥	\$10.14	38	٤ ≨	žž	38
	MRC - Incremental Charge - Manual Service Order - Discornacy	SOMAN	5 5	≱:	8	ž	8	98	ž	¥	8
	1 1	Domino.	4-	ž	2	≨	1 4	2	≨	Ş	90
生	Zone 1, per month	USBFN	H	¥	≨	ž	Υ¥	¥	ž	17.02	¥
	Zona 2, per month	USBFN	92	812	8	27.38	27.15	180	1, 923	¥	8
		Maga	1	23 63	5E	133.41	24.83	180	\$44.07	¥	380

ALECUTHMENBOUTH SATES METWORK ELEMENTS

Ξ.	NO.148C)930						RATER BY OTATE				
Ξ	Zone 3, per month	2005	ž	7	ď	K	5	.1	¥.	78	77.
Ξ	Zone 4, per month	7000	2 3	2	+	\$24.47	\$25.06	2	18083	ž	Ç.
=	ARC 18	Koch	¥	ž	4	≨	Н	TBO	Ş	╄	2
Ξ	MRC - Add		2	21.22	4	\$209.77	Н	TBD	\$215.00	╁	TBA
=	MRC - Decomed Charge - 1st	N. SEL	\$131.14	\$120.01	_	\$127.26	┝		800	╁	
Ξ	MFIC - Discorned Charge - Act	E 183	\$131.15	\$127.64	L	\$133.53	۲	Ē	1 90	+	
Ε	NATC - Service Order a company Resemblads, see, 188	USBFN	83.42	16 753	L	\$33.69	╁		2	ž	GBI
E		SOMES	\$3.50	23.23	L	8	╁			ž	2
E	211	SOMEC	2	20.03	Ž	1	+		2	25.53	180
Ė	MAC. Savios Ories abundan Hansah.	SOMAN	¥	\$21.58	ž	76 063	1		¥:	ž	TBO
E	MC Insurant Chart III	SOMAN	¥	200	¥	a	╀	3 4	٤	ž	9
Ė	ALC: Property Come. Manual Company	SOMAN	081	ž	1			2	≨	ž	TBO
Ė	March Instrumental Charles and Property Control of the Control of	SOMAN	98	≨	9		+		ž	ž	130
E	CHARGE CALL COLOR CALL	SOMAN	98	Ş	180	ž			٤	ž	8
Ē							+		ž	ž	1 80
E	Zone 1 per month	USBFO	26.003	ž	ž	2	13				
Ė	Zinak 2 nor minor	USBFÖ	081	224.69	ğ	\$27.26	637.46	٤	ž	_1	¥
L	Zone 1 on more	USBFO	2	28.83	E	1	22.5		Ş	_1	8
Ė	Zone 4, per month	USBFO	780	\$29.16	180	24.47	× ×		9	ź	180
Ė	XAC . 18	USBFO	¥	¥	ž	¥N			2	- 1	TB0
Ė	WRC. Ach	USBRO	2212.80	A IIA	9	1000	6300	3	ž	- 1	≨
Ė	WRC - Decrease Opens - 14	OSero	\$131.14	\$128 81	9	\$127.56	+	2	8 2	- 1	<u>8</u>
Ė	M.C. Channel Chan	USBFO	\$131.15	\$127.64	9	1	+		317.50		780 200
Ė	M.C. Service Order a year of the	USBFO	99.753	16.55	2	F13 R8	+		8 2	¥	2
-	NAC Serve Order a comment Description	SOMEC	82.50	52.23	ž	S	3 5		S S	1	<u>g</u>
E	NAC - Sevice Order a Arribad Learner, 198 LSM - Uncornect	SOMEC	Ź	27.03	≨	Ž	3		2	1	8
Ė	NRC Service Order to produce of the con-	SOMAN	ž	\$21.58	ž	200	١		2	1	180
E	MAC - Insumental Charas - Macade Common Comm	SOMAN	ž	200	ž	2	2	8 2	\$	1	2
Ŀ	M.C. Programme Course, Character Service Course, 18	SOMAN	QE.	ž	98	¥	2	3 5	≨ :	-	90
	Marchinerae Character Language Control Control	SOMAN	98	ž	180	2	2	3	٤	1	9
Ė	DRUGGET - BOLD STATE	SOMAN	180	ž	9	ž			٤	٤	2
1	4-Wing 64 (OPPS Diethal Unbanodied S. e.) see E. e.									Š	180
Ł	Zona I par month	USBFP	4903	≨	ž	ž	VA	3	120		
E	Zone 2 car secusion	USBFP	180	\$24.89	2	15 / 63	100	Į Į	£ 2	22.47	ž
E	Zone 3 for examin	USBFP	180	\$28.83	8	100	10.40	9		Š	9
E	Zone 4, oer moreh	USBFP	OET	\$29.16	280	24.47	X	3	200	٤	2
E	APIC - 1st	USBr	П	ž	٧Z	ž	Ş	180		5 2	2
	NRC - Add	J8S0	7	\$11.35	98	17.605	\$209.47	2	60.5.00	12.00	2 6
	INRC - Decorrect Charge - 1st		┪	\$128.61	TBO	\$127.28	\$127.07	8	व्हें दर । इ	157.67	3 6
	MRC - Discornect Charge - Addi	1000	~	\$127.64	Q.	\$133.63	\$135.74	98	05 963	77	3
	INTC - Service Order submitted Electronicals, per 159	3	7	16.23	rBo	£33.69	24.23	180	23.83	¥	Ç.
	NRC - Betvice Order submitted Electronically, par LSR - Discounters	SOURCE OF STREET	7	2,2	ž	\$3.50	35.50	180	850	93 63	9
	NPC - Service Order automitted Manually, per LSR	SCALAN.	-	200	Ş	٧×	¥	TBO	≨	ž	28
	MMC - Service Order euthritized Manually, per LSR, Discornect	7777	-+	8	ž	229.24	\$29.24	TBO	¥	ž	180
1	MMC - Incremental Charge - Manual Service Order - tel	NAMOS	+		ž	2	3	OBT.	¥	ž	2
1	MAC - Incremental Charge - Manual Service Order - Addi	SOMAN	081	2 3		ž	\$18.14	2	ž	¥	780
I	THE STATE OF THE SERVICE ONCE - DECORDED	SOMAN	+-	ž	9	NA	8,1	2	Į.	≨:	9
ľ	medical Such Lover Medicalism		-					3	ž	Š.	98
	NPC - Load Col/Coutoners Person if our 3 Miss cut. 14		Н						T	1	
	MFC - Load Colificationent Removal per 2 Was pair - Add	NACY.	+	20,00	2	237.82	E122.79	98	\$360.33	\$358.71	68
1	MACC - Load Col/Equipment Removal per 4 Wire pair - 1st	UMAX	08 853	1000	2 2	64.15	27.48	2	11 20	19	180
I	ARTO: Betne 1 on Demonstration of the Wine part - Add)	UCMAX	٠.	\$8.15	98	1	97.63	2	7	1/8872	2
	MACC-British Tay Service for the control of the	ULMBT	Н	\$542.71	22	55.03	52100	6		10.00	980
		ULMBT	+	\$10.19	8	\$10.19	\$6.08	2	\$10.14	208	C C C
5	Loop Mates Up										2
	NRC - Losp Makeup - Preordering Wilthout Reservation, per worlding facility quanted		1	1							
1	(Namusa)	UNACH	8348	8134.88	833	2.5	5		-		
=	Medimum number of spans facilities on manual tales in a		┝				3	3	3	8	8
E	WTC-Losp Making - Preordering With Reservation, per spare facility queried (Manual)	COMMITTAL	8 8	8248	8.2	8134.88	813480	813488	8134.00	8134 00	8100 00
1	Mitto number of spere tacilities per manual UALVSI is 3 ** NRC - Loso Makaio - Prescharine Mittors 6 Description	CAMPLE	\$14000	\$140.00	\$140.00	8140 00	\$140.00	\$140.00	8140.00	\$140 m	8
	(Mechanized) **		2		12:						
Ξ	Loop Makeup - Preordering Without Reservation, per apera lacitiy queried (Mechanized)		8	R	8	8	8	8	8	20 20	\$0 60em
Ė	Loop Makeup - Prendering With Reservation, December to the August Machinery		8 2	80.5	\$1.08	81.06	80 18	8	80 50	8	80 60 kg
	Max number of sperv facilities per mechaniced LMJSI is 10.7		8	8			2	1	:	╁	
=					8	8	80 15	8	8	8	20 696
					-	-			-	-	_

Colored State 15th	Column C					-						
Color Colo	Corner C	Unbundled Network Terminating Wire, per pair, per month	USOC	₹5	<u>ر</u> و	a	È	≤	1 1	¥	28	N.
SOME SOME	SOME SOME	NRC - UNTW Pair, per pair	UFNP	2007	2 2	2 3	8	\$0.35	8	3 1	\$0.46	180
SOME SOME	SOME SOME	NAC - Decorned Charge, per pair	UENP	29 05	N N	2 2	9	2847	8	28.83 29.83	26 09 3	180
SOME N. 1874	SOME	NHC - Service Order authoritied Electronically, per LSR	SOMEC	95.63	S	5	2	ž	00	ž	ž	180
SOLIMA N.M. SETAM N.M.	SOLIMA N.M. \$2.5 M. \$	MAC - Service Order Bushmitted Electronically, per LSR - Deconnect	SOMEC	2	8	2	2 2	2	2	\$350	3 3	98
SOME SOME N. STA N. STA	SCORNE BAN B	ALTO SELVE CHOR BLOTHING MAINTAIN, per LSR	SOMAN	2	37.50		5	٤	2	2	1 <u>10</u> 0	180
SOME BM B	SOME BLAG REG SEG SEG TOTAL TAT	1994 - Service Order eutherlied Menually, per LSR, Discornect	SOMAN	ž	ä	1	200	200		2	<u>6</u>	Ē
EÓMEC BAY ENA BAY BAY </th <th>SOMEC BM EDA REG BM CAD TOD TOD<th>4.0</th><th></th><th></th><th></th><th></th><th>2</th><th>2</th><th>2</th><th>9</th><th><u>a</u></th><th>2</th></th>	SOMEC BM EDA REG BM CAD TOD TOD <th>4.0</th> <th></th> <th></th> <th></th> <th></th> <th>2</th> <th>2</th> <th>2</th> <th>9</th> <th><u>a</u></th> <th>2</th>	4.0					2	2	2	9	<u>a</u>	2
SOMEC REAL DEAT NA ESTA TOTO	SOME REAL BEAL REAL REAL <th< th=""><th>Sub-Long Concentration - Chewnellosidon Sys (Outside CO)</th><th></th><th></th><th></th><th></th><th></th><th>1</th><th></th><th></th><th></th><th></th></th<>	Sub-Long Concentration - Chewnellosidon Sys (Outside CO)						1				
SOME NA SIGN SIGN SIGN NA NA SIGN SIGN NA NA NA SIGN SIGN SIGN NA NA NA SIGN SIGN SIGN NA NA NA SIGN SIGN SIGN NA NA SIGN SIGN SIGN NA NA NA SIGN SIGN NA	SOMM NA SIGN SIGN TRID	LICK BATTO OTHER BLOMBED BECKENICARY, per LSR	SOMEC	95 52	S S	100	3					
SOMM N.M. STISE N.M. STISE N.M. STISE N.M. STISE TIDD	SOMM NA SURSAL SURSAL TIDO NA NA SOMM TIDO NA SURSAL TIDO TIDO TIDO NA NA SOMM TIDO NA SURSAL TIDO TIDO NA NA SOMM TIDO NA SURSAL TIDO TIDO NA NA LOCIAN SERSAL SURSAL TIDO TIDO TIDO NA NA LOCIAN SERSAL SERSAL SERSAL SERSAL SERSAL TIDO TIDO NA NA LOCIAN SERSAL SERSAL SERSAL SERSAL SERSAL TIDO TIDO NA NA LOCIAN SERSAL SERSAL SERSAL SERSAL SERSAL TIDO TIDO NA NA LOCIAN SERSAL SERSAL<	AND SERVICE UND SECRETARIES ESCRIPTION, per LSR - Decornect	SOMEC	ž	2	3	3 3	2	8	9	8	\$3.50
SOMMA Th. SEA, A. Th. SEA, A. Th. SEA, A. Th.	SOMM TIME STATE TIME TIME <t< th=""><th>LOCK SERVICE SLOWING Manually, per LSR</th><th>SOMAN</th><th>2</th><th>23.10</th><th></th><th>2</th><th>2</th><th>2</th><th>2</th><th>8</th><th>ž</th></t<>	LOCK SERVICE SLOWING Manually, per LSR	SOMAN	2	23.10		2	2	2	2	8	ž
SOMMA TIO	SOMMA TIDD NA STIGN TIDD NA STIGN TIDD NA NA CUTTAL SECOLAT MATTAL STIGN	PMTC - Service Urder (stornified Menuelly, per LSR), Decornect	SOMAN	¥	120	5 3		202	2	e E	2	\$19.99
Cuttor C	Cuttor C	TOTAL - MICHIGAN Charge - Manual Service Order - 1st	SOMAN	QII.	1	5	2	2	2	2	200	ž
UCTAN MACHINE	UCTON MERGY MATCH MATCH <th< th=""><th>MATC - Incremental Charge - Marrial Service Order - Addit</th><th>NAL S</th><th>3 6</th><th>5</th><th>4</th><th>2</th><th>2</th><th>ĝ</th><th>4</th><th>¥</th><th>¥</th></th<>	MATC - Incremental Charge - Marrial Service Order - Addit	NAL S	3 6	5	4	2	2	ĝ	4	¥	¥
UCTAN MANA MANA <t< th=""><th>UCTAL FAVA FAVA STATA STATA TOTAL FAVA STATA STATA TOTAL FAVA TOTAL FAVA TOTAL STATA TOTAL TOTAL TOTAL STATA TOTAL TOTA</th><th>TYXXX - System A (96 channel capacity - channels 1-86), per month</th><th>INTRA</th><th>20.00</th><th>2</th><th>2</th><th>3</th><th>2</th><th>180</th><th>¥</th><th>≨</th><th>ž</th></t<>	UCTAL FAVA FAVA STATA STATA TOTAL FAVA STATA STATA TOTAL FAVA TOTAL FAVA TOTAL STATA TOTAL TOTAL TOTAL STATA TOTAL TOTA	TYXXX - System A (96 channel capacity - channels 1-86), per month	INTRA	20.00	2	2	3	2	180	¥	≨	ž
UCTON ENGLY MAN	LCTIA PARAM HANZA FARAM FARAM <th< th=""><th>MRC-18</th><th></th><th></th><th>2/1/1</th><th>\$7.24.73</th><th>22</th><th>1546.21</th><th>780</th><th>\$578 63</th><th>\$571.42</th><th>\$683.78</th></th<>	MRC-18			2/1/1	\$7.24.73	22	1546.21	780	\$578 63	\$571.42	\$683.78
UCINA ENGL	UCINA EXACT EXAC	INPC - Addi	3	5	7202	8	28	29.70	8	\$567.83	\$586.42	\$634.31
UCTON \$15.00 </th <th>UCTON \$15.00<!--</th--><th>INPC-Decorated, 1st</th><th>4</th><th>X 00 X</th><th>\$222.37</th><th>\$310.62</th><th>\$22.37</th><th>\$222.08</th><th>8</th><th>27.78.60</th><th>16 1123</th><th>82112</th></th>	UCTON \$15.00 </th <th>INPC-Decorated, 1st</th> <th>4</th> <th>X 00 X</th> <th>\$222.37</th> <th>\$310.62</th> <th>\$22.37</th> <th>\$222.08</th> <th>8</th> <th>27.78.60</th> <th>16 1123</th> <th>82112</th>	INPC-Decorated, 1st	4	X 00 X	\$222.37	\$310.62	\$22.37	\$222.08	8	27.78.60	16 1123	82112
UCINA \$14.00 \$17.40 </th <th>UCTOM \$7.40 \$7.44 <th< th=""><th>WRCOlessomes, Addi</th><th>3</th><th>229623</th><th>22,802</th><th>ž</th><th>\$253.67</th><th>\$258.08</th><th>200</th><th>ž</th><th>2</th><th></th></th<></th>	UCTOM \$7.40 \$7.44 <th< th=""><th>WRCOlessomes, Addi</th><th>3</th><th>229623</th><th>22,802</th><th>ž</th><th>\$253.67</th><th>\$258.08</th><th>200</th><th>ž</th><th>2</th><th></th></th<>	WRCOlessomes, Addi	3	229623	22,802	ž	\$253.67	\$258.08	200	ž	2	
UCTOR 587.34 586.72 582.34 586.73 786.73 586.73 786.73 </th <th>UCTION 987.4 987.1 <t< th=""><th>THOS - System & All channel neverth . Account of the</th><th>AS IS</th><th>874.80</th><th>\$74.84</th><th>Į</th><th>\$80.50</th><th>591.63</th><th>OEL</th><th>¥.</th><th></th><th>1</th></t<></th>	UCTION 987.4 987.1 <t< th=""><th>THOS - System & All channel neverth . Account of the</th><th>AS IS</th><th>874.80</th><th>\$74.84</th><th>Į</th><th>\$80.50</th><th>591.63</th><th>OEL</th><th>¥.</th><th></th><th>1</th></t<>	THOS - System & All channel neverth . Account of the	AS IS	874.80	\$74.84	Į	\$80.50	591.63	OEL	¥.		1
UCTOR \$20.04 \$40.04 </th <th>UCTOR \$1,00.00 <t< th=""><th>MAC 1st Marine</th><th>UCTRB</th><th>267.34</th><th>\$85.12</th><th>16.263</th><th>\$12.15</th><th>S#7 #8</th><th>Ę</th><th>99</th><th></th><th>٤</th></t<></th>	UCTOR \$1,00.00 <t< th=""><th>MAC 1st Marine</th><th>UCTRB</th><th>267.34</th><th>\$85.12</th><th>16.263</th><th>\$12.15</th><th>S#7 #8</th><th>Ę</th><th>99</th><th></th><th>٤</th></t<>	MAC 1st Marine	UCTRB	267.34	\$85.12	16.263	\$12.15	S#7 #8	Ę	99		٤
UCTOR EXAMA EXAMA <th< th=""><th>UCTOR EXAMA TED MA MA EXAMA EXAMA<!--</th--><th>INSC. Acti</th><th>OCT 88</th><th>\$404.8</th><th>2408.22</th><th>\$632.36</th><th>\$408.23</th><th>\$407.65</th><th>S</th><th>2607</th><th></th><th></th></th></th<>	UCTOR EXAMA TED MA MA EXAMA EXAMA </th <th>INSC. Acti</th> <th>OCT 88</th> <th>\$404.8</th> <th>2408.22</th> <th>\$632.36</th> <th>\$408.23</th> <th>\$407.65</th> <th>S</th> <th>2607</th> <th></th> <th></th>	INSC. Acti	OCT 88	\$404.8	2408.22	\$632.36	\$408.23	\$407.65	S	2607		
UCTOR EXACT EXACA NA EXACA EXACA NA EXACA UCTOA \$200.00 \$10.00 \$10.00 \$10.00 \$10.00 NA NA UCTOA \$10.00 \$10.00 \$10.00 \$10.00 NA	UCTOR EXACT EXACT <th< th=""><th>MOC. Newseard 14</th><th>UCTAB</th><th>\$550.55</th><th>\$222.57</th><th>\$310 62</th><th>\$ 0003</th><th>222</th><th>3 5</th><th>3 2</th><th>7 8 8</th><th>7</th></th<>	MOC. Newseard 14	UCTAB	\$550.55	\$222.57	\$310 62	\$ 0003	222	3 5	3 2	7 8 8	7
UCTAN 59.10 57.10 57.10 57.10 57.10 57.10 57.10 57.10 57.10 57.10 MA NA VA UCTAN 500.02	UCTOR \$10.00 </th <th>Mark August 12 and 12 a</th> <th>UCTBB</th> <th>23623</th> <th>\$236.02</th> <th>WA</th> <th>S X</th> <th>2</th> <th></th> <th>2</th> <th>is in</th> <th>821178</th>	Mark August 12 and 12 a	UCTBB	23623	\$236.02	WA	S X	2		2	is in	821178
UCTAA MORTA MORTA <th< th=""><th>UCTAA Meet T \$17.00<!--</th--><th></th><th>UCT88</th><th>\$74.90</th><th>\$74.84</th><th>A.A</th><th>5</th><th>1</th><th>3 5</th><th>٤</th><th>Ş</th><th>ž</th></th></th<>	UCTAA Meet T \$17.00 </th <th></th> <th>UCT88</th> <th>\$74.90</th> <th>\$74.84</th> <th>A.A</th> <th>5</th> <th>1</th> <th>3 5</th> <th>٤</th> <th>Ş</th> <th>ž</th>		UCT88	\$74.90	\$74.84	A.A	5	1	3 5	٤	Ş	ž
UCTAL MORIGA MORIA MORIA <t< th=""><th>UCTAL GOLD FOLKE FOLKE</th><th>Thoras - symmetry and chemically - chemicals 1-60, per manch</th><th>VCT3A</th><th>2,000</th><th>\$512 AE</th><th>675.43</th><th>2 5000</th><th>2</th><th>2</th><th>ž</th><th>ž</th><th>ž</th></t<>	UCTAL GOLD FOLKE	Thoras - symmetry and chemically - chemicals 1-60, per manch	VCT3A	2,000	\$512 AE	675.43	2 5000	2	2	ž	ž	ž
Victim V	UCTION ECONOMIC		NCTIN	40.48	200	2000		200	2	200	29 1 23	\$726.87
Variety Vari	Victory Vict	rest. Addi	UCTA	200	222	2	27.00	8		289,83	\$506.42	634.31
Off-Control (1977) \$15,000 NA \$25,000 \$35,000	Victor V	PARC-UNICOTABLE, 186	ACT 3A	2262		20.00	200	3 77	2	\$278.63	27.97	\$2,1162
Vir. Challender Frield, para month LUCTOR 51.87.20 17.72.9 51.87.20 17.72.9 51.87.20 51.87.50 51.90.0 71.90.0	Official States 57.00 at 10.00 at 1	MMC-Disconnect, Addi	ACT TA	18		٤:	Š	8	2	¥	≨	≨
UCTOS \$250.00 \$17.00 \$	UCTOS \$250.00 \$10.00 \$	TROCG - Bystem B (Ne channel capacity - channels 87-162), per security	1000	Т			200.50	20.183	780 08	¥	ş	ž
UCTOR 120.00 12	UCTOR 150.00 15	NAC - 18	0.1.2.1	7	200		\$119.01	\$125.86	OBT	\$122.16	\$127.76	\$145.21
UCTOR \$25.52 \$10.00 \$25.52 \$10.00 \$1	UCTOR 1970 174 1	NAC: Add	1	т	7		5409.23	\$407.68	180	\$567.63	\$586.42	\$634.31
Cut State NA State NA State NA State NA State NA State NA State NA State NA State NA State NA State NA State NA State NA State NA State NA State NA State NA State S	UCTOR \$7.50 \$7.5	NFC-Discornsect, 1st	8	7	15.00		\$222.37	\$222.08	180 081	23 87 52	16 11 57	\$111.78
Cutting	Cutting	MRCDecomect, Add	8	┱	2.8.B		\$253.87	\$258.08	081	ž	≨	Ž
UCTFS MAY MA	Cutties	D&1 Feeder Interface, ser month	8133	27.80	274.62	ž	\$80.50	281.62	8	Ş	¥	1
UCTFS	UCTY No. 190	Zone 1, per month	ST.	8/9	ž	\$72.12	٧×	≨) JE	564.63	\$69.24	276 77
Curing NA \$65.00 TRD \$100.11 \$100.	UCTTS NA \$45.00 190	Zone 2, per month	S. C.	\$	888	2	\$60.01	\$73.29	OBL	180	2	1BO
Cutting 1925	Cutties Nat Max Nat	Zone 3, per month	200	٤	8	2	200	248.72	æ	180	180	CEL CEL
UCTO STATE	UCTFS STREET ST	Zone 4, per month	200	٤:	80./08	8	\$106.11	8187.11	TBO	OEL	98	180
UCTFS \$105.73 \$15.24 \$15.05 \$15.27 \$15.24 \$15.27 \$15.24 \$15.27 \$15.24 \$15.25 \$	UCTTS \$15.24 \$10.00 \$1	MAC 1st	Sire.	ş	ş	ž	¥	ž	780 087	ž	ş	ĄN
UCTOR \$100,73 \$100,73 \$112,04 \$112,04 \$113,14 \$113,14 \$123,17 \$123,17 \$116,01 \$115,03 \$116,01 \$115,03 \$115,04 \$113,14 \$133,1	UCITIS \$100,73 \$123.04 \$123.04 \$123.04 \$123.04 \$123.07 \$16.01 \$11.601 \$11.53 March of Chound Start, par man UCICS \$2.34 \$127.04 NA \$15.07 \$10.00 <th>MCAAM</th> <th>SCITS</th> <th>15.251</th> <th>2012</th> <th>23.7</th> <th>29 60 CE</th> <th>47.8023</th> <th>OBL</th> <th>27.00</th> <th>20.00</th> <th></th>	MCAAM	SCITS	15.251	2012	23.7	29 60 CE	47.8023	OBL	27.00	20.00	
UCTOR \$12.50 W. \$133.51 \$13.51	UCTOR 62.94 512.75 NA 513.54 513.74 TBD NA NA	MAC Decomed: 14	UCIFS	5105.73	\$129.04	\$196.06	\$127.11	\$127.07	180	116.01	27 31 12	
Beart or Ground Start, par man U.CZS \$12.81 \$2.90 NA \$35.59 \$14.25 \$12.90 NA \$15.59 \$15.50 TBO \$2.75 \$2.00 Month U.CZZ \$10.00 \$20.00 \$10.0	Bound or Chrowned Shart, page mane U.C.CTS \$12.81 \$12.90 NA \$33.56 \$14.25 TRD \$7.57 NA Month U.C.CZ \$10.80 \$12.90 \$1.00 \$2.00 \$1.00 \$2.00 \$1.00 \$2.00 \$1.00 \$2.00 \$1.00 \$2.00 \$1.00 \$2.00 \$1.00	MC Discouns: Acc	UCTFS	F 958	\$127.78	ž	\$133.43	\$135.74	TBO		2	i P
U.C.2 \$2.30 \$2.15 \$2.30 \$2.24 \$2.2	U.C.2 \$50.00 \$1	Change blackers 3 the Value one feet - A	UCTF6	812.81	10 Cts	ž	65058	24.23	130	47		5
U.C.2 \$20.00 \$41.00 \$20.00 \$15.00 \$1	U.C.2 \$70.00 \$41.00 \$20.00 \$41.00 \$20.00 \$2	IMPC 14	883	8C 28	12 12	\$2.36	223	22	2	200	170	3
U.C.2 10.00 10.0	U.C.2 \$10.00 \$1	MPC A4#	2035	850.08	2012	2 2	8012	20103	CBL	20 64		1
March Marc	Marcheller	WCOscored 1st	200	28077	96.023	\$41.58	98 023	28 023	180	Sec	27.53	2 3
Marcol M	Marcollane	MACDIONNAC AND	MCC2	\$10.00	89.89	¥	\$10.73	\$10.90	TBO	Ž	1	
Marcol M	UCC \$50.00 \$1.00	Channel Printers . 9 Mar 1874	MCC2	20	C3 63	ž	810.68	\$10.86	S	1	2	
UCC 2004 2010 20	ULCC \$70.00 \$10.	MACKE	NOC!	18031	\$9.48	12.83	8 5	15.63	081	W 53	20 22	610.44
UCC \$1000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000	U.C.C. \$10.00 \$	MACAAN	133	2 8 8 8	251.07	3	80123	80.153	780	\$21.24	528.58	2
U.C.C 5594 459.50 NA 410.75 610.60 NA NA 10.05 610.60 NA NA NA NA NA NA NA N	ULCCA \$15.00 MA \$10.05 \$10.00 MA MA MA MA MA MA MA	MRC Decomed, 1st	1300	202	80.00	84.50	\$50.96	ESO 83	081	\$21.13	23 62	i
Companies	UCCS \$14.14 \$12.51 \$14.17 \$13.56 \$10.66 \$10.06 \$10.	MPC Discomect, Add	1333	808	86.03	ž	\$10.75	\$10.92	180	ž	ž	42
ULCOR \$20.00 \$21.01 \$11.01 \$11.05 \$11.00 \$1	ULCOR \$20.00 \$41.00 \$10.00 \$1	Cheerel Interface - 2 West Voice - Research Business	5	X Q	20.03	≨	810.68	210.86	28	ž	2	1
ULCOR \$20.00 \$21.00 \$41.50 \$21.00 \$2	ULCOR \$20.00 \$41.50 \$21.00 \$2	MARC 1st	NCO.	\$14.14	\$12.61	\$14.17	\$13.36	\$13.69	Ę	8 ::	617.46	
ULCOR \$20.96 \$41.54 \$70.96 \$41.54 \$70.96 \$40.97 \$10.97 \$10.00 \$40.97 \$10.00 \$1	ULCOR \$10.00 \$29.96 \$41.54 \$70.05 \$10.00 \$20.96 \$41.54 \$70.05 \$10.00 \$20.96 \$41.54 \$70.05 \$10.00 \$20.96 \$20.05 \$20.95 \$2	MRC Adm	HCC3	850.08 0.23	7012	2 2	22.08	8	2	20 862	2	1
U.C.Cri \$19.00 NA \$10.05 \$10.	MCCA \$110.00 \$19.99 NA \$10.97 \$10.92 180 NA NA NA NA NA NA NA N	Merchanics to	MOON	230.77	\$5.02\$	\$41.58	26 O.2	25 023	2	986	5	
Main Curve, parameter ULCC4 \$15.45 \$15.45 \$10.66 \$10.66 \$10.66 \$10.60 \$10.6	March March March 1954 1954 1954 1956 1756 1750 1757 1952 1952 1954	March March 1995	ECC3	\$10.00	86.03	ž	\$10.73	210 82	2	1		
UCC4 \$8.43 \$7.52 \$8.44 \$7.56 \$8.75	U.C.C.d 20.00 221.07 24.18 57.28 24.48 57.56 24.18 57.61 57.61 57.61 57.61		ROOM.	23.54	\$9.83	ž	810 68	98 OF 8	2	5 3	5 3	\$
UCC2 \$20.00 \$21.07 \$21.05 \$21	UCC4 \$20.08 \$21.07 \$41.02 \$21.04 \$21.04 \$20.04 \$21.04 \$21.04 \$20.04 \$21.04 \$21.04 \$20		700	\$8.43	\$7.52	\$8.45	\$7.96	2	8	100	2	2
U.CC4 E20.77 E20.58 \$41.54 E20.56 E20.50 TBO E20.50 E20.5	U.CCA \$20.00 \$41.54 \$20.95 </th <th></th> <th>roon r</th> <th>80 023</th> <th>\$21.07</th> <th>28 22</th> <th>813</th> <th>81.0</th> <th>3</th> <th>2000</th> <th>2 2</th> <th>2</th>		roon r	80 023	\$21.07	28 22	813	81.0	3	2000	2 2	2
UCC4 \$10.00 \$19.99	ULCC4 \$10.00 \$29.99 NA \$10.75 \$10.90 TBD NA NA NA NA NA NA NA N	TO THE PARTY OF TH	roon	1201	85.028	3	36.023	20 00	3 2	8 9	8 3	2
UCCIC 59.94 \$9.50 NA \$10.68 \$10.06 TBD NA NA ULTIC 50.08 E10.7 \$20.70 \$20.90 \$40.49 TBD \$20.47 \$42.14 UCTIC \$20.77 \$20.08 \$41.52 \$20.08 \$20.05 TBD \$26.65 \$26.66 \$26.66 \$26.66 \$26.66 \$20.05 TBD \$26.60 \$26.60 \$20.42 NA NA NA NA NA NA NA NA NA NA NA NA NA	UCC4 \$3.94 \$3.50 NA \$10.06 \$10.06 \$10.07 NA NA	M. Year Market	NCC4	\$10.00	\$ 63	ž	F. 012	3	1	2	2	2
CTTC \$41.21 \$59.76 \$41.30 \$58.90 \$40.49 \$100 \$24.14 \$42.14 \$10.7	Cutto	FUNCTORING ACT	MCC4	3.23	28 83	¥	10 68		2 8	٤	Š	ž
UCTTC \$20.69 \$21.07 \$41.57 \$21.05 \$71.05 \$71.05 \$10.00 \$28.94 \$42	UCTTC \$20.66 \$21.07 \$41.56 \$21.05 \$10.07 \$48.41 \$42.41 \$4	I WAT CATCUM, per MOTEN	UCTTC	241.21	╁	5	8 8	8 9	8 8	ž	≨	¥
UCTTC \$10,07 \$20,98 \$41,58 \$20,95 \$20,93 \$18.00 \$28.98 \$28.98 \$1.58 \$20,93 \$18.00 \$28.99 \$20,93 \$20,	UCTTC \$70.77 \$70.78 \$41.58 \$20.58 \$20.57 \$10.00 \$28.50 \$28.42 \$20.57 \$10.00 \$28.42 \$20.57 \$10.55 \$10.57 \$10.57 \$10.55 \$1		UCTIC	\$20.68	╁	3	2 2		3		2	22.53
10071C 81000 8999 NA 81075 81087 190 XX 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	MA SIGN STATE STAT	MATC ACC	UCTTC	12.05	t		3 2	3 2	3	8	95 925	ž
MA KA 1805 1805 KA KA 1805 1806 1800 KA KA KA 1805 1806 1800 KA KA KA KA KA KA KA KA KA KA KA KA KA	Per mandel UCTTC 5554 5555 NA S1045 51056 TBD NA NA NA NA NA NA NA N	MACCORCUMENT, 180	UCTTC	810 PG	+		R	2 2		0.82	23.42	7 2
mental NA NA STUDE STUDE THO NA NA	PRO PROPERTY ST. 14 S12.57 S1.77 S1.77 S1.27 TBO S1.56 S1.77		1 CTTC	200	t	\	2000	25.0	9	ž	ş	¥
	\$12.00 \$11.00 \$11.00 \$12.27 180 \$11.27		18005	╀	+	٤	8	210	2	ž	¥	¥

	MELLACUTHANEWBOUTH NATES METWORK ELEMENTS AND OTHER SERVICES	MOUTH NATE ELEMENTS SERVICES									Anathrapit 2 Embit C Rates - Page 16
BCMPTION					2	AATES BY STAT	<u></u>				
MRC 18	T COST	¥ 83	+	8	П	3	1 1	¥	\$6	Z	
NRC Discornect 191	M.CCS	2077	Н	25.55	7	203	2 2	2 2	88 58	8	
MRCDecomect Addi		808	4	≨:	П	\$10.80	TB0	¥	¥	Y.	
NAC 14	n cos	\$12.49	+-	15 21	_	20.00	2	ž	ž	Ş	
MRC Add	200	88 023	251.07	21.82	т	\$21.05	8	8 88	21277	2 3	
MCCOeconec(is	n co	10.00	4.	2 2	_	05023	<u>2</u>	28.50	28.22	141.71	
Op Concentration System (Neide C.O.)	NCOS	3	₩	2	\$10.68	\$10.86	2 2	≨≨	≨ ≨	¥	
MRC - Service Order autmitted Electronically, per LSA	SOMEC	95.03	2 6	5		1 1					
MRC - Service Order is continued Electronically, per LSR - Discovnince	SOMEC	ž	33	7	2 4	8 2	05.22	8	8	S 53	
MRC - Service Order automitted Manually, per U.S.P. Disconners	BOWAN	₹	Ιi	ž	1	1		ž	5 5	¥	
MRC - Incremental Charge - Manual Service Order - 1st	SOLAN.	ž	- 1	_		ļΙ	1	ž	2		
INTC - Incremental Charge - Menual Service Order - Addit	MANOS	8 8	_	_		П	17	1 90	110	180	
MACO (as manual capacity - channels 1-80), per month	UCTBA	27.72		-	7	Т	- T	OB.	\$13.55	780 08	
MPG - Addi	UCTBA	\$1,115.10		-	50.53	1 0 0 0 0	2 5 5	86.0	2662	\$300.06	
MPC-Decorrect, 1st	X .	≨ :	_	-	ž	ΥN	≨	≱	2	9 4	
PATCOBCONNECt, Addit		2	≨ s	-	≨.	ş	¥	ž	ş	\$	
MANY TOWN IS (THE CHIEF OF CHIEFLY - Chemick 67-182), per month	ı	\$67.41	25.52	-	¥2.55	Į.	ş	≨.	≨	ž	
MC Adm		\$464.57	121121		21.0	100	27.30	862.98	18.1	\$68.71	
NPC Discornect 1st	П	¥	ž	_	ž	N N	1	2	R 99	2787	
MRC-Decomed, Addi	Т	≨	٧V	ž	¥	ž	1 ≥	£ 2	2 3	5	
RJCD - Bytthem A (16 charred capacity - channels 1-46), per mores	Т	ž	ž		į	ş	≨		1	2 2	
NAC. 18	Г	41 114 10	201037		\$567.21	\$478.24	\$506.70	\$422.68	\$450.13	\$428.73	
MRC/Connect	П	ž	3 ≨		Т	1000	115 10			1,114.05	
WRC/Decorated Acti	П	ž	ž	_	┱	2 3	2 3			\$	
POG - Pytem & (10 channel amounts - channels (7) tern	П	ž	ž	¥	Т	1 ≤	1 2			ž	
NAC 1st	Т	811130		À	т.	\$ 100 89	7	Т		Y	
NAC. Add	Т	264.57			П		Т	\$463.74		277	
MARC-Decorner, 1st	Т	Į.	Т	≨:	≨.	П		1		¥	
MACCHINECT ACCT	Т	2	т	ž	T	_				¥	
NRC 14	П	22	Т		7	┰	_	т		₹	
WPCAed	П	57,78CS	Т		+	7	┰	7	_	\$6.49	
NPC-Discornact, 1ix	Т	8132.03	\$2.17	_	Н	1	\$132.07	\$131.79	-	2	
MRCDiscornect, Addi	8150	٤į	_	ž	97	27.00	-	П	ş	≨	
HARC 14	Т	852	1	-	+	+	- 1	7	_	ž	
MCAdd	H	125.77	Н	1	+-	+	1	+		3	
MRCDiscement (at	†	883	8 8	Н	Н	₩	\$35.37	+		2 50	
MRCDecomec, Addi	+	ž ž	+	+	+	H	l i	Н		¥	
Manual Interface - 2 Wire MDM, per month	t	\$10.19	╁	╀	╀	+	- 1	+	11	ž	
WC VA		17.853	+	┿	+	+	-1	+		\$10.30	
MRCDecembed (at	H	33 SS	Н	┿	╁	╁	7	+	1	8	
MRC-Discornact, Addit	†	≨:	Н	Н	Н	1	1-	+-	1	ă ¥	
Annel Interface - 2 Why Voice - Revense Bettery, per moves	T	ž	╅	+	\$10.66	\$10.06	¥	Н	T	ž	
MCC (#	T	1987	+	+	+	+	П	Н	Н	15.32	
MC Authority (w	Г	38.83	╁	╁	3 3	+	Т	+	7	18.74	
MPC-Decorned, Add	MOCH	≨	86 83	¥	+	\$10.92	Т	- A	+	3 4	
Annel Meritice - 4 Why Volce, per south	7	≱ š	Н	Н	Н	Н	П	+-	╈	5 3	
MRC 1 M	T	2 2 2 2	+	Т	+	H	П	Н	Н	51.53	
MAC Adm	Π	П		19.5	3 8	+	2 2 2	+	16 92	135.74	
WCCOmment And	П	П		Т	╀	+	Т	4	+	7	
d Croat, per moreh	Ħ	П			╁	+	Т	4-	+	¥:	
NRC 14	7			П	╁	╁	+	+	+	¥ 2	
MRC Addi	†	Т		83 85	Н	Н		↓_	╀	2 2	
AFIC Discornacy, 1st	†	Т		┪	Н	Н	Н	Ц	╀	2	
MCChecomed, Addi	Ť	Т		_	+	+	7	Ц	Н	ž	
event interface - Digital SAChes, per month	t	Т	51.5	Т	+	+	+	┙	H	¥.	
MTC 18	T	П	1012	T	+	+	+	1	╅	8	
WC Appr		Т	85028	T	┿	+	+	_	+	2	
II TOTAL PROPERTY IN	Н	П	66.63	≨	┿	28.018	+		2	2 2	
					1	1	1		$\frac{1}{2}$	3	

NEWSCUTH RATES IN ELEMENTS ER SERVICES

Attachment 3 Emert C Retes - Page 17

NA

DELLECTIVAEWSCUTH RATES NETWONE ELEMENTS AND OTHER SERVICES

Anachmen 2 Embit C Raen - Paps 18

	FESCHIPTION.	O O O									
T _C	A Was Analog Line Port (San Line)				5	٤	3	3	¥	ပ္တ	Ĭ
Н	2-Wire Voice Grade Line Port (Residence), per month								1		
+	2- wire volce unbumbed port - residence	UEPR	60 63	1						1	
+	A-WW VOICE Unbunded port with caller ID - residence	DEPRO	3 8		00 · CO	K 61 - Note	2	11.24	2 X	25.33	77. 17.
+	2 The Machine of Conform only - residence	UEPRO	200	2	B	192	2	123	80.03	\$2.35	22.73
	Confer ID				3	10.74	ai	23	23	£23	54.73
-	2 Hits voice grade unbunded Kentucky entended local district party party auto	UEPAR	2002	ş	¥	¥	¥.	≨	*		3
4	Company	UEPRIM	*	3	3					٤	٤
	caller ID				Ş	K.B.	ş	≨	≨	≨	¥
ļ_	2-wire voice grade unbunded Managerol entended local dialing party on with	UEPAS	ž	≨	ž	¥	\$2.30	ž	≨	¥	3
4	2-may 10	UEPAT	3	ş	\$;					-
	with caller ID				5	٤	ž	122	ž	ž	¥
	2-wire voice grade unbunded Ternames enerded local deling parity port with	UEPAU	≨	ž	≨	¥	ž	¥	≨	20	2
I		UEPAO	3	3	•	:					
Ι	2-WTO VOCO Unbunded Portch pres calling with caller 10 - mercence	UEPAF	2	¥ !	ž	ž	ž	NA	≨	ž	7
Ι	K-WTB VOICE LEGUNDED LOUGHEND And Pive with cases 10 - residence (RUL)	UEPAG	S M	2	≨ :	Ş	¥	¥	ş	ž	2
Τ	5 me vitto utturtoet Loudena Ana Plue with caler ID - residence (ACT)	UEPAH	ž	1	2 3	2	2	ž	ž	¥	ş
	residence (LWS)				5	٤	2	≨	≨	¥	≨
_	2-wire volce unburded Terrasses Area Calling port with Caller ID - maintena	UEPAJ	≨	ş	ž	¥	ž	ž	ž	2	3
Т	7 mm water	UEPAK	ž	\$	3	3	;	:			
┪	(TACER)					5	Ě	≨	≨	ž	2
_	2-wire voice unburided Termense Area Calling port with Caler 10 - residence	¥ LYS	≨	2	ž	ž	ž	Ş	ş	ž	2
1	2-wite voice unburger Terranean Area Californ	UEPAM	ž	ž		ž	3	;	;		
-	(1MF2X)	UEPAN	×	;	:			•	٤	ž	Z
_	CAMPS WICH WILDINGS Termesses Area Calling port with Caller ID - neithbridge			5	§	≨	≨	ž	ž	ş	2.2
т	2-with votes unbanded me, tow useps less port with Caller ID (134)	LEPAO	¥	ž	ž	Ş	\$	\$	2	2	1
┿		2	/n/X	2015	£	1923	88	<u>=</u>	82.00	15 X	2 2
+	ACCOUNT NUMBER POSTINBILITY (PECUARES CHE PER PORT)	UNPCX		T		1					
+-	A Who Voice Grade Line Profilement										
₩	2-wire voice unbundled port without Caller ID	100	2								
+	A Here wood unburded port with unburded port with Caller+ E484 ID	UEPBC	100	2015	8 3	2	8	12	88	22.35	27.72
+	2 with video grade untended Automa account in the	UEPBO	202	2	8	200	2 2	2 2 2	8.8	8	2.73
-+	Company (Company) (Company	UEPAW	8	1	:				N X	2	7
_	Caller IO			1	٤	<u> </u>	≨	≨	ş	ž	ž
+-	2-wee votce grade unbunded Louetans entended local dusing party, not wen.	UEPBN	ş	¥	Į.	19.28	\$	ş	≨	1	42
+	Catte votal case actuated in	UEPAX	ž	¥	 ≨		\$	1	:		
-4	Carper 10	À de la la la la la la la la la la la la la	;	-		+		5	į	Ş	ž
	2-sets voice grade unbunded South Carolina entended local dialing parity port		5	ž.	≨	ž	ž	=	ž	ž	ş
1	2 -ets voice grade unbunded Temesses extended frost degree sects and	UEPAZ	ž	ž	ž	ş	ž		2	3	1
4		UEPAV	1		;	-					٤
4	2-see votes unburged incoming only port with Caller ID	UEPBI	102	2013	¥ 34	2 2	¥ £	≨ .	ž	ž	2
₽	2-are votes unturded SC But Are Caling Part and Calar II a Lo.	UEPAA	ž	5	ş	ž	83	1	8 4	2	2
_	2-wee volce unburded TN Bus 2-Way Area Calling Port Economy Option	NEV.	ş	≨	≨	≨	¥	≨	. ≨	2 22	ž
1	(TACCI)	UEPAC	≨	.		1		-			
_	(TACC2)	I ED AN	:	:			٤	Š.	ž.	ž	Y E
	2-wire volce unbundled TN Bus 2-WAY Collecytle and Memphis Local Calling	2	5	1	≨	ž	ž	ž	ž	ž	2
ш		CEPAE	ž	ž	2	¥	ž	2	ş	.	2
1	LOCAL MARKER PORTABLITY (RECURES ONE PER PORT)	CAPCX	1	\dagger	+	+					
Ш	Non-Resurting Change (ARC), 1st (Australia)					+	\dagger	\dagger	1		
П	2- who volce unturnised port - medianos	\dagger	18	H	H	Н	Н	╁		\dagger	
_[2-wire voice unbundled port with caller ID - neitdence	UEPRO	2 5 5	+	+	+	Н	Н	98 i 23	85 725	03 63
	2-wee voice emburded port outgoing only - residence	П	25125	2	\$17.16	27.73	218.43	22.53	3 3	8 7	22 22
	Collection of the control of the collection of t	┝		├	╀	╀	╀	+	5	8 7	8
1		NAVA S		ž	ş	Į.	ž	ş	¥	¥.	¥

Š	DESCRIPTION	ngoc	₹	4	ð	Š	1	S	¥	ဒ္ဓ	T.
	ches voca grass unsurced Kemucky energied local daing penty port with caller ID	UEPRIA	≨	ž	ž	\$37.78	ž	ž	ž	\$	¥
	Z-with voice grade unbundled Louisians entended local dusting parity port with caller ID.	UEPAS	ž	ž	ž	ş	\$16.43	\$	ž	ž	\$
	2-wins volce grade unburndled Médidelppi entended local diabing party port with calter ID	UEPAT	\$	2	2	ž	2	\$22	\$	ž	ž
	2-with voice grade unburnded South Carolina antended local disling parity port with caller ID	UEPAU	\$	ž	ž	\$	\$	ş	2	26 7-23	¥
	2-who volce grade unbunded Tennesias extended local dialing parity port with caller ID.	OKAN	2	1	1	42	3	1	12	1	8
	2-we volce urbunded Portas area calling with caller ID - meidence	UEPAF	ž	2 76	2	ž	ž	ž	5 ≥	ž	ž
1	2-who voice unburded Louislana Ana Pha with caller ID - maidence (RU)	DV430	≨:	≨:	≨:	≨ :	\$16.43	≨:	ž	ž	ž
L	2-Mrs volce unbundled South Carolina Area Calling port with Caller ID -	5	ž	§	٤	٤	27914	Ş	ž	ž	≨
1	Presidence (LWS) 2-sets votes just profest Tempesses Area Calling and with Caller ID. presidence	VEPA	ž	ž	≨	≨	\$16.43	≨	ž	ž	¥
	0.50	UEPAK	ş	¥	ş	ş	ž	ž	\$	8773	\$
	Z-wite volce unbunded Tennissee Area Califing port with Califin ID - residence (TACER)	UEPAL	ş	ž	ş	ş	ž	¥	YN	₩	28 63
	Z-włe volce urbunded Ternesses Ania Calling port with Callier ID - residence (TACSR)	UEPAM	\$	≨		\$	\$	ž	ž	ž	29.82
	Z-wine volce unbundled Tennassas Ania Calling port with Caller ID - residence (116F2X)	NEPAN	¥	¥	\$	ž	ş	\$	Ş	≨	2
	2-wie volce urbundled Ternseses Arie Calling port with Caller ID - residence (2MR)	OFF	**	¥	RA	ž	¥	NA	¥	\$	2 22
1	2-wire voice unturched Res Low Usage Line Port with Caller+ES63 ID &UMA	UEPAP	ž	ž	≨	≨	≨	¥	¥¥	¥	28:53
	HPRC - Adef1 (Neekdence)										
#	2-wire voice unturded port - residence	UEPR	ន្តន	3	\$17.16	E (C)	\$16.43	96 ZZ3	221.60	86 72	9 9
	2-why voice unbundled port outgoing only - martenos	UEPRO	2012	Z	\$17.16	2 10	795	22.23	8 8 22	8 8 Ž	2 22
	2-wire voice grade unbundled Alabama antended botal disling parity port with custom to	WASIN	08128	\$	ş	ž	\$	ž	\$	ş	¥X
	Z-win voice grade unbundled Kentucky extended local dieling party port with collect O	UEPRIM	ž	¥	\$	E. 123	2	¥	W.	44	Y.
	2-wire votes gride unbundled Louiseans entended local disfing parity port with caller ID	UEPAS	ž	ž	ş	ž	\$16.43	\$	ž	¥	ž
	2-who voice grade unbundled Meetamppi entended local dieting partly port with caller to	UEPAT	ž	2	ž	3	¥	20.00	¥	¥.	42
	2-with victie grade unbundled South Carolina extended local dialing parity port with caller ID	UEPAU	2	ž	ş	2	2	ş	\$	22.4 88	42
E	2-wire voice grade unbunded Terresees extended local dialing perty port with	S. C.	:	:				:	:	:	
	2-wire voice unbundled Floride area calling with calter ID - residence	UEPA	§ ≨	Z Z	٤ź	ž	≨≨	≨≱	ž	źź	2 2
	2 with volce unburded Louisians Area Plus with caller ID - residence (RUL)	UEPAG	≨:	≨.	≨:	≨:	\$18.43	≨:	≨:	ž	ž
\pm	Z-win voice urburded South Carolina Area Calling port with Callin ID -	F .	Į.	ž	ž	Ş	\$16.43	§	¥.	Ş	į
+	Z-uira volce urbunded Tennesses Ans Caling port with Calier IO - residence	745	ž	ž	£	£	≨	≨	ž	87.00	\$
1	(F20)	UEPAK	ž	ş	ž	ž	ş	ž	¥	ž	22.02
	(TACER)	UEPAL	ž	ž	ž	Ş	ž	ž	ž	¥	\$9.19
	2-with voice unbundled Tennesses Area Calling port with Calter ID - residence (TACSR)	UEPAM	ş	ş	ž	ş	\$	ž	ž	ž	2
	2-who voice unbunded Tennames Avea Calling port with Caller ID - residence (1MP20)	UEPAN	\$	\$	¥	≨	ž	2	\$	\$	2
	Z-wee vidos unbunded Termasses Area Calling port with Caller ID - neadongs (Charle)	OPAGE	\$	1	2	4	47	1	1	×	
	2-wire volce unbundled Res Low Usage Line Port with Caller ID (LUM)	UEPAP	221.53	7.7	\$17.16	£77.73	\$16.43	8523	80 68	24.98	61.63
	MMC - Subsequent Activity	USASC	\$10.00	810.00	00.013	\$10.00	810 00	00 018	810.00	\$10.00	\$10.00
	2-wire Volce Unburgled Port without Caller ID	CEPBL	21.50	27.72	\$17.16	52.73	\$16.43	\$22.96	\$21.60	\$24.98	69.60
1	2-wine voice unbundled port with Caffer ID 2-wire voice unbundled outgoing only port	OE SEC	2 2 2 2 2 2	77	\$17.16	8/3	2 2 2	2 2 2 3	3 3	8 8 3 3	2 2 2 2
	2-was voce grade urbunded Alabama enanded local dateing perly port with loader to	UEPAW	23.123	ş	ž	\$	\$	ž	ž	\$	ž
	Z-eve volce grade unbundled Kentucky entended local disting party port with leafer D.	(SPBM	ş	\$	2	2.5	ş	\$	42	\$	1
	Z-wire voice grade unbundled Louisbure extended toost dusting party port with	A PROPERTY	3	1	1	;		1	3	:	
1		****	٤	5	ž	§	23.85	Ę.	5	Š	ž

METWORK ELEMENTS AND OTHER BEHAVES

DESCRIPTION	PTION	INOC	,								
* C	whe voice grade unbunded Meelengol emended local dusing parity port with	3	4	2	ΔP.	2	5	3	¥	9	¥
4	with volce grade unbunded South Carolina entended local daling parity por	UEDAY	ž	ş	ž	ž	ş	\$2.2.38	ž	≨	\$
* 1	The caller (O	UEPAZ	ş	ş	≨	2	ž	*	:		
0	May 10	INFPAV	-	;				٤	ž	224.98	≨
	wire voice unbundled incoming only Port with Caller ID	LEPB1	Š	≨ .	ž	ž	≨	≱	¥	≨	8
4 6	2-With voice unburded LA Bus Area Calling Port with Callier ID (BUC)	UEPAA	2 ≥	* *	21.71	\$2,52	16.43	\$22.98	24.04	\$24.98	28.83
+	THE WASHINGTON OF BUILDING FOR WELL CARE TO FEST (TAB)	UEPAB	≨	2	ž	2	2 1	≨ :	≨:	≱	ž
٤	(TACC)	9	-				[٤	٤	8	≨
<u>د د</u>	2-eths voice unburided TN Bus 2-eny Area Culling Port Standard Opeon	2	≨	≨	≨	≨	ž	ž	ş	¥	8
44	2-with voice unburdled TN Bus 2-way Collectifie and Memorial cost Callect	UEPAD	ž	ş	ž	ž	ž	≨	≨	2	2
2	1 (809)	UEPAE	\$	*	5	3	:				2
1	C. Add (Ambres)				٤	5	ž	≨	≨	ş	23.53
ŕ	Me valor urbunded port without Caller ID	3									
	ere value unturalled part selfn Celler ID	CELES C	2 5	3	\$17.16	85/03	\$16.43	\$22.98	89.23	872	20.19
+	- who wice unburded outpoing only port	UEPBO	8	Ž		8 2	16.43	22.98	80.63	872	\$6.19
8	sifer ID	1650	1	:				200	3	22	5
-	Are volce grade unbunded Kentucky extended local defing perty port with		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	ž	≨	≨	≨	≨	¥	≨	ş
6	the votes grade unbunded Louisierns priended local diging bathy root with	UEPBN	ž	≨	ž	\$37.78	¥	¥	≨	≨	¥
1	TO YOUR STATE LIVE STATE LANGE CO.	UEPAK	ş	ş	\$	\$	218.63	2	1	:	
3 6	er 10	UEPAY	≨	\$	***	47	1		1	Š	Ş
	me your grade undurant bouth Carolina arthroded local dusing party port. It caller ID		:			5	٤	R Z	≨	≨	¥
2	tra volce grade untunded Terramese entended local daling parity port with	7	≨	ž	ş	≨	ž	≨	¥	24.38	ž
	to value unbundled incoming and that the Call In	UEPAV	¥	¥	ş	\$	¥	2	3	;	
	re voce urburded LA Bus Area Cultus Pari and Calar IN 181 pc	SEP SE	23.53	2	\$17.16	27.55	\$16.43	8 22	10.65	2 2	
	2-wire voice unbundled &C Bus Ama Calling Port with Calair ID (LAB)	NA PLAN	≨ à	≨ 3	ž	≨:	\$16.43	ž	Ş	Ş	ž
Ě	C-With Voice Unburdled TN Bus 2-way Area Calling Port Economy Option (TACC)			5	٤	Ş	§	¥	ž	3 7 5	\$
2	were voice unbunded TN Bus 2 way Area Calling Port Standard Option	DEFAC	≨	≨	ž	ž	ž	¥	Ş	ž	2
	who voice unburded TN Bus 2-way Collegiville and Memorias local Culton	UEPAO	ž	¥	¥	ž	ž	\$	\$	ž	:
Ž	WI (EZF)	UEPAE	≨	≨	\$	44	3	3	:		
Ĕ	MRC - Subsequent Activity	7848H					٤	£	ž.	≨	2
		3	20.00	210.00	808	8	\$10.00	\$10.00	10 00	\$10.00	\$10.00
	* We vote urburded out - meriens										
2	e volce unbunded port with cales ID - residence	T COOL	2 2	Z Z	≨.	≨	23	25.52	ž	ž	200
1	was voice unburded por outgoing only - reactings	UEPRO		2 2	≨ ≨	ž	Z :	3	ž	ž	8
İ	were vicinal grapps unburided Ambienns extended focal diefing parity port with ler ID					٤	2	200	\$	ž	20 20
,	wire votes grade unbundled Kentucky entended local dieling parity port with ties. ()	E	Ž	ž	ž	ş	¥	\$	\$	\$	\$
7	with voice grade unburded Louisians animoted local dasing parity port with	NE DE	ž	ž	2	Ę	ž	\$	ş	Ş	ş
	are voice grade urbunden Managers artistic in a dist	UEPAS	ž	ž	ş	ž	2,3	<u></u>	*	1	:
1	O i	UEPAT	¥	\$	**	5	;			5	5
•	to caller ID	170	:	;		1	5		ş	\$	\$
	were volce grade unbunded Terressee aniended local dialing party port with	2	٤	§	¥	ž	ž	\$	ş	ş	ş
į	who wice unburded Forith are celling with celler D - markeys	OV.	\$	ž	ş	ž	ş	£	<u></u>	2	3
Į.	a votos urbunded Louislans Ans Plus with cader ID - narcianos (PUL)	DE PAG	2 3	2,3	≨ :	ž	ş	ž	¥	ž	3
	voice unbundled Louisiana Area Plus with caller 10 - residence (ACT)	UEPAH	ž	₹ ≨	≨ ≱	≨ ≱	N Z	≨ :	≨ :	≨ :	≨
1	mediance (LWR)	UEPAU	1	1	3	:	3	٤	٤	ş	\$
Ę	2-errs voice unbundled Tennessee Area Calling port with Caller ID - naidence			1	£	٤	≨	ž	ž	2	ž
1 5	Parks voice unbunded Tennesses Area Calling port with Caller ID - residence	¥ des	\$	≨	ş	ž	ş	ž	2	ž	23.86
4	2-wire voice unburded Termesee Avea Caling port with Caller ID - naidence	LEPA	ž	5	\$	ž	ž	¥	ş	\$	2
	8	UEPAN	\$	¥	\$	ş	\$	2		472	3
											8

ELLBOUTHMEWBOUTH RATES NETWORK ELBRERTS AND OTHER SERVICES

ă	DESCRIPTION	USOC	7	1	43	Š	3	5	2	9	F
3	Z-MMB voice unbunded Tennesses Area Calling port with Caller ID - residence (TMF2X)	UEPAN	ž	ş	¥X	7	ź	42	1	3	:
	2-wire volce unbunded Tennesese Area Caling port with Caller ID - maidence					[9	5	٤	٤	2
Į	Courts when are prefer that the land has been also Court to the	CEPAO	≱	ž	ž	ž	≨	ž	¥	ž	23 68
	(WOT) OF THE CITY OF THE CASE	A-13	282	27.73	¥	ž	3	\$6.56	¥	¥	99.03
	2-wre voice unbunded port without Calay ID	UEPE	\$6.21	82.78	¥	2	2	86 58	Y.	¥N	23.64
1	2-wire voice urburded port with Caller ID	UEPBC	12.98	92.28	¥	ž	\$4.38	\$6.56	¥	ž	20.03
1	2 with with fresh privated Automa Control Automa Control	UEPBO	18621	2.78	ž	ž	2 38	8 88	₹ Ž	¥	23 66
	Calebra ID	UEPAW	\$2	ź	≨	\$	¥	¥.	2	4	3
	2-stre volce grade unbundled Kentucky extended local deling parity port with					i		5	5	٤	٤
1	2-wha votce grade unbunded Louisians entended facility party port with	UELDM	ž	ž	ž	≨	ž	Ş	¥	ž	\$
1	caller 10	UEPAX	≨	ž	ž	Ş	2	¥	ž	ş	\$
	Z-ere voce grose urbunded Managips entended local daining party pon with caller ID.	UEPAY	≨	\$	¥	V.	¥N	5	4	3	1
_	2-wire votes grade unbundled South Caroline extended local dualing parity port							8	Ę	£	Š
1	2-ste vite cate utterded Terrenes erected from delter rate and	UGPAZ	ş	ž	ž	ž	\$	ş	ž	¥	ž
	cafer ID	UEPAV	ž	ź	<u></u>	ş	≨	¥	¥	¥	20
	2-wire voice unbundled Incoming only Port with Caller ID	UEPBI	\$4.21	57.78	ž	ž	23	33	ž	Į.	3
#	A take wide unfamilies for the form full or full or form of the following for the following for the following for full or form of the full or form of the following for full or form of the form of the full or form of the full or form of the full or form of the full or full or form of the full or full o	Web W	≨:	ş	≨:	\$	8,3	¥	ž	¥	ş
	2 were value unbundled TN Bue 2-way Area Calling Port Economy Cotton	2	ž	٤	ş	§	§	¥	≨	Ş	ş
1	(TACS)	UEPAC	ž	ž	¥	ş	Ş	≨	ş	ž	\$3.08
	(TACC)	OCEPAD	\$	£	ž	\$	\$	¥	4	7	27.5
=	2-wire votos unbumbled TN Bus 2-Way Collierville and Memorine Local Calling										2
1	ron (e.g.)	UEDAE	≨	ž	ş	¥	ž	ž	ž	ž	80.03
Ė	MMC - Decorrect Charge - Adril			1							
	2- wire voice unbundled port - residence	UEPPE	\$6.21	65.28	2	≨	2	86.58	¥	¥	200
#	2-vira voice unbunded port with caller ID - neidence	UEPRC	12821	\$2.59	ž	ž	238	86 56	≨	ž	28.03
1	Course water transfer part degrang any - resource	UEP90	\$6.21	9525	ž	ž	2 2	95.98	≨	¥	23
	College ID	UEPAR	\$8.21	≨	≨	£	¥	Ş	2	**	¥N
_	2-wire voice grade unburdled Kentucky antanded local dieling party port with loafer ID.	PERM	1	1	3	1	1	:	:	:	:
	2-wire voice grade unbunded Louisiers enended local dialing party port with					5	٤	٤	٤	ž	٤
1	coder to	UEPAS	ş	ž	≨	ž	2 2	ž	¥	Ş	KA
	cater ID	UEPAT	ş	\$	Ş	≨	Ş	38	Ş	MA	Ž
_	2-ethe voice grade unbundled South Carolina actended local disling parity port with calter ID	ONAG	77	1	44.	4	3	;	:	:	
	2-etre voice grade unbunded Temesese extended local dealing parity port with					•	2	٤	٤	٤	٤
1	Caller 10	UEPAO	ş.	ž	ž	ş	ž	\$	ş	\$	2
-	2-ere votes unturated Louisians Area Plus with cates 10 - markeyse (R.E.)	EPAG	\$ 3	2 1	\$	≨ ;	¥.	≨ :	≨ :	≨ :	≨ :
	2-eins volce unbunded Louisiens Area Plus with caller (D - residence (AC7)	UEDAH	≨	ž	≨	≨ ¥	2 3	§ §	\$ \$	5 ≥	≨ ≨
_	2-white votos urbunded Bouth Carolina Area Calling port with Caller ID - residence (LWS)	UEPAL	170	\$	4	1	1	;	;	1	1
	2-sins volce unbunded Temeseas Area Calling port with Caller ID - residence		:	1					5		5
Ė	2-win votes urbunded Terremes Avec Calling port with Caller 10 - residence	1	į	٤	5	§	ž	£	≨	≨	Q Q
#	(TACEN	UEPAL	ž	ž	\$	ž	ž	ž	ž	ş	33
	(TACSR)	LEPAM	ş	¥	\$	ž	ş	ž	ş	¥	2
_	Z-wire volce urbunded Tennesse Aves Calling port with Callier ID - residence (19872)	LEPAN	**	1	5	1	;	;	:	:	1
	2-wire voice unbunded Termesse Ann Calling port with Caller ID - residence						1	1	5	5	
#	22-eve votce unturnided files Low Usego Line Port with Caller ID & Like	OK AND	≨ Š	¥ 63	1	2 2	¥ 2	20 2	≨ 5	≨ 5	20 2
								R	٤	٤	2
1	2 wine voice unburded port without Caller ID	TEMBE.	17 77	65 Z	≨:	≨ :	2	3	≨	3	33
	2 with votes unbunded outgoing only port	08.63	178	R SR	٤ž	٤ź	3 3	8 3	\$ \$	2 2	200
E	2-eine volce grade unburdled Alabama antended local disting partly port with	300	:	-		;	:				
\pm	2-wire voice grade unbunded Kembody entended local dealing partly port with	UEVAW	22.58	ž	≨	S	ž	\$	£	§	ž
∄.	Cater 10	CEPER	ž	ž	Ş	ş	ž	ş	ş	ş	ş

79	╀	¥.	+	Y.			≨ :	+	¥ 8	NA 22 RD	NA \$2.92	H	╁	\dashv	+	WA 613 32	+	\$6.29 NA	+	+	L	Ц	2	Ц	1	Ц	1	≨ ≨	4	╀	Ц	28 811.11	23 23	╀	Н	+	25 E13	+	58.97	+	╀	Н	35 52 AN	╀	₩	\$130.23 \$120.00	10 be \$80 00 negotiated	Ì
4	+	+	+	5	-	-	≨ 5	+	ž	ž	¥		╀	+	+	ź	╁	∓	\dagger	t	H	\dagger	\dagger	H	+	Н	\dagger	ž	\dagger	t	1	+-	912	+	z Ž	+	╁	┿	\$12.36 \$12	╅	┿	N.	H	NA NA	╁	\$12166 \$13	\$116.59	ł
941	1	ž	3 5	\$	٧	95.98	ž	٤	ş	ž	¥			22.50	22.2	818.08	٤	5,73	2017	8363	\$19.68	25.52	91919	₹	15 031	8	2 5	88.20	22.2	816.00	≨	88	8 1	25.93	35	2 2	816.06	8	\$14.60	88	\$13.46	\$13.48	25 523	\$16.07	416 07	\$146.46	\$117.01	
		8	ž	ž	≨	3	2	[ž	≨	≨			98.23	\$18.14	\$10.39	٤	22 23	ž	. ₹	¥	≨ 3	∑	ž	\$8.26	¥	≨ ≥	ž	ž	Ş	ş	\$10.13	5	123	12	100	3	X X	\$13.12	200	200	\$9.20	71 818	\$10.39	\$10.39	\$149.27	28.63	
2	:	ž	≨ :	ž	≨	≨:	ž į	[ž	ş	ž			S S	≨ ≨	≨ :	٤	≨:	2 2	₹	≨	ž	§ ≨	ž	≨	≨:	≨ ≱	ž	ž	ž	ž	ş	≨ :	ź	≨:	≨ ≨	ž	\$	\$	≨ ≨	. ≨	≨	≨≨	≨	ž	Υ¥	MA	
40	:	ž	≨ :	≨	¥	≨ :	ž		ş	ž	\$			982	28.0	≨ 3	٤	≨ :	2 1	≨	≨	≨ \$	ž	ź	≨	≨:	ž ž	ş	2 2	ž	≨	17 85	817.18	ş	ž	2 2	ž	Ş	21.3	\$ 3	≨	ź	3 2 2	ž	ž	\$120.60	288 44	
		§	≨ :	ž	ž	2	2 3		≨	\$	ž			82	≨ ≨	≨ :	[040	2 2	ž	3	≨ ‡	. ≥	ž	¥	¥	ž ž	¥	2 3	≨	ž	\$6.74	Ž.	20.03	73	2 2	200	2	9C 92	21.0	\$113.28	\$7.12	# 5 20	170	II CI	\$63.31	6413.50	
7	3	ž	≨ :	≨	≨.	202	₹ 3		≨	2	ž			8 :	812.97	11.11		26 55 25 25 25 25 25 25 25 25 25 25 25 25 2	420	\$18.41	\$10.41	12.37	117.77	2	Ş	≨:	ž	≨.	ž	ž	≨	≨	≨≨	ž	≨ :	ž	≨.	٤	\$12.08	8 8	≨	ž	2 2	ž	ş	130.23	860.00	
1300	YEDAY	8	AV430	NEPAZ MEPAZ	UEPAV	UEPB1	UEPAB		UEPAC	UEPAD	UEPAE			SOMEC	SOMAN	NAMOS		UEPVF				NAME OF THE PERSON	SOLLAN	SOMAN	UEPVF				SOMAN	SOMAN	SOMAN	UEA	UEP4A	BFR	BFR	SOMM	SOMM	BURNA	24430	2445	UEPP2	UEPP2	SOMM	SOLLAN	SOMAN	OCHDO	UEPOO	
SCHETION	2-wire voice grade unbundled Louislans extended local desing party port with caster ID	2-wire voice grade unbundled Mesessippi entended local disling parity port with	2-ers in 2-ers grade unbundled South Caroina extended boal dailing parity port with natural to	2-who voice grade unbundled Tennessee extended local desing parity port with	Substitution of the substi	Zeate voice unburdled I A Bus Aves Celling Part with Color in All Co.	2-wire volces unburded &C Bus Area Calling Port with Caller (D. (LAB)	2-wire volce unbunded. Thi Bue 2-way Area Calling Port Economy Option	2-wire vides unburded TN Bus 2-way Area Calling Port Standard Option	(TACCS)	Part (227)	ARC: OBS	AMC - 086 LSR Charge, Electronic, per LSR received from the CLEC by one	14 PB CSS FEBRUAY BEARINGS MARC - Incremental Change - Marcha Service Order - 14	NRC - Incremental Charge - Manual Service Order - Addr	NRC - Incremental Charge - Manual Service Order - Discornact - Let NRC - Incremental Charge - Manual Service Order - Discornact - Additional Service Order - Discornact - Additional Service Order - Discornact - Additional Service Order - Discornact - Additional Service Order - Discornact - Additional Discornact - Addi		Average testure, per mortis	MPC - Add1 (all types)	MRC - Decornect Charge - 1st	MFC - Discorned Charge - Add1	MAC incremental Chaps - Manual Sevice Order - Add	NRC - Incremental Charge - Manual Service Order - Decornact - Ex	NVIC - Incremental Charge - Manual Service Order - Disconnect - Addit	rea available feature, per month	NATION SIX (all hypes)	MRC - Decorrect Charge - 1st	NRC- Disconnect Charge - Add1	MRC - Incremental Charge - Manual Service Order - Add	MRC - Incremental Charge - Manual Service Order - Decornact - 1st	MHC - Incremental Charge - Manual Service Order - Discorpect - Addi	Me Analog VO Port, per month	MCC 18	NAC - Decorrect Charge - 1st	MFC - Decornect Charge - Add UAPC - Incremental Charge - Macrael Resolve Colors - 14		MRC - Incremental Charge - Manual Service Order - Decornacy - 1st		And DID Part, per morally	NHC - Add	NMC - Discorrect Charge - 1st	MRC - Decornect Charge - Add1	MHC - Incremental Chargo - Namual Service Order - 1st MHC - Incremental Chargo - Manual Service Order - Acri	121	21	Ne DS1 Pert w/DiD capability, per month	147C-18	

•

EMPTH .	ENACES
METWORK E	

DESCRIPTION	USOC	7	2	8	Ķ	7	9	ž	33	×
WRC - Decorned Charge - Addi	UEPDO	ž	3	ž	¥	20 85	\$12.94	ş	ž	≨
MACC Programmed Charms Mary Service Order - 18	NAMOS S	ž	2 2	200	≨:	218.14	25.55	3. 8.	≨:	ž
IMC - Incenental Owns - Marvel Service Order - Decorners - 1st	SOLIAM	2 2	2 2	2 2	ž	9	2	\$12.76	≨:	≨ :
NRC - Incremental Charge - Manual Service Order - Decornact - Acci	SOMAN	§ §	2 5	ž	¥ 2	\$10.39	216.06	ž	ž	ž
П					5	2001	300	٤	Š	٤
2-Why ISON Port(2) (3), per month	USPMA	\$16.42	\$10.20	\$13.47	\$12.33	EE 123	161153	25.450	\$30.74	¥¥
WRC 1s	UIPMA	\$63.24	155.34	\$47.37	\$90.46	25.33	\$63.59	862.29	\$65.79	ž
M.C. Add)	SIPMA	\$53.24	\$106.00	\$47.37	\$64.53	145.35	\$63.59	\$62.29	\$65.79	¥
AND DECORPED CAMPS - 18	SPAA	2	\$50.37	ž	ž	1 2	\$7.04	ş	¥	¥
March Commercial Comme	CIPMA	25 BB	88	ž	ž	12.0	\$7.0	ş	ž	NA
COLOR - PECCENTING CANADA - MANAGA CANADA CANADA - 18	SOMAN	62 43	812	8.65	¥	52823	183.67	\$55.30	\$67.52	NA
Mary - stownship Carps - Mary Bery De Chost - April	SOMA	256.15	85128	838	≨:	82 92	18.53	\$65.30	\$67.62	ž
MAC - Insurance Charte - Manual Cardon Calar - Manual - Add	1		2 2	ž	Ý.	20 20 20 20 20 20 20 20 20 20 20 20 20 2	21.34	¥	ž	ž
MCC Line Profits one A Channel 45	SCHOOL	77.7	8 4	\$ 3	YZ	8	\$11.34	Ý.	≨:	ž
		٤	٤	٤		Ş	ž	Š	ž	ž
2-Who IBOH Peri(2) (3) Including all evellable heatures, per month	UIPMA	≨	ş	ş	ž	≨	ž	≨	23 A 60	\$18.40
NPC-1st	CIPALA CIPALA	ž	ž	ž	¥	ž	ş	ž	8106.40	23023
Nec - year	UIPMA	ş	ş	ž	ž	ž	ž	ž	\$108.40	55.63
MRC - Decormed Charps - 1st	UIPNA	¥	٧¥	¥	¥	¥	ž	ž	ž	5
MRC - Disconnect Charge - Addi	UIPWA	¥	W	¥	¥	¥	ž	ž	ž	07.73
NATC - Incremental Charge - Manual Service Order - Int	SOMAN	≨	ž	ž	¥	¥¥	٧×	¥	\$87.52	\$41.43
MATCH INCOMPANIE CHANGE SERVICE CHOICE - ACCT	SOMM	≨:	Į.	ž	ž	ž	ž	ź	\$67.52	\$42.17
MACO Intermedial Chara - Macual Sevice College - Machine Land	W 100	ž	≨ :	≨ ŝ	≨ :	≨ :	≨:	≨ :	≨ :	98
		5	٤	٤	٤	5	ž	<u> </u>	٤	2
2-Who ISOM Port(2) (3) Including three available features, per ments	UIPMA	ž	Ş	ž	¥	≨	ž	ž	10.903	≨
NRC-1#	UIPMA	¥¥	¥	¥	ž	≨	ž	ž	25.03	2
MAC. Adri	UIPMA	ş	≨	¥	¥	¥	ž	ž	\$70.32	ž
MRC - Incremental Charge - Manual Service Order - 1st	SOMAN	ş	¥	ž	¥	MA	ž	ş	\$67.52	ž
MMC - insemental Charge - Manual Sevice Order - Addi	SOMAN	ž	≨	ž	ž	¥	¥	Į	\$67.52	ž
+ The BUH LET For per month	ŒĿ	S . S . S	88.38	\$163.16	ž	27.72	\$213.21	22.22	24.78	800
		8544.88	117.61	\$196.80	≨.	\$1018	244.12	23.52	27837	148 66
Marie Comment Comment of the Comment	METEX FORTY	9	2 2 2	00.00	ž	20191	27.12	2	278.37	\$147.18
MAC. Decreased Chema. Acti	NETEX TERES	9	200	\$:	\$ 3		2 2	ž	≨ :	25 45
MRC - Incremental Change - Manual Service Origo - 100	NAMA	2 7 7 7	23.5	2	2 3		3	2	£	2 2
MHC - hovemental Chanse - Manual Service Order - Acci	NAMOS	2 2	2 2	2 6	2 2	2	3 8	25.00	97 207	200
MRC - Incremental Charge - Manuel Service Order - Decorment - 1st	SOMM	21.23	2	ž	ž	87.73	1583	ž	ž	20 65
MRC - Incremental Charge - Manual Service Order - Disconnect - Addit	SOMAN	811.53	2	ş	ž	\$7.73	\$8.51	\$	≨	\$10.54
A-WAR SECRETARY OF PROPERTY OF PROPERTY OF PROPERTY OF PROPERTY. 14	TENEX.	≨ 5	Ž,	≨ :	27.5	≨ :	≨ :	¥:	818	≨ :
MAC. And	X de la	2 2	1	2 3	77 1913	Š	2 3	23	21	ž
NAC - Incomental Change - Manual Service Order - 1st	NALLO.	1				2	٤	1	44 44	5
MPC - Incremental Charge - Manual Service Order - Add1	SOMAN	Ş	≨	ž	≨	≨	≨	ž	88.48	ž
			1							
C WIND ACRES CHILD FOR THE MAN STATE OF THE		22.50	:		1	1	1		1	***
LINE SIDE LABUNIX ED COMBINATION PWAY PBX TRUME - BUSINESS		10.23	3		ia	200	-	82	2 2	2 2
LINE SIDE UNBUNOLED OUTWAND PBX TRUNK - BUSINESS		\$2.07	\$1.62	31.85	19:28	23	15.53	85.00	85.23	2
CASS CASTALLES INCOMENTS POX TRUME - BUSINESS	CEPT.	8	2	20.	19 23	22	11.23	82	90.23	2
The 2-way Call land by Ala Pay Trainer - All Kalence		3 2	X C	8 4	2 5	3 5	¥ :	3 8	8 2 2	2
TH OLITWARD CALLERS PLAN PRIX TRUMK - BUSINESS		10.23	2	200	500	22	ı a	8 2	2 22	2 2
2-WHE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA										
SALEND PORT	UEPAR	60	ž	ž	ž	ž	Ş	ž	ž	¥
CALING PORT	UEPt2	\$	ş	¥	Ş	8	ž	ş	¥	ž
2:WITE VOICE UNBUNDLED PRIX LD TEHANINA, PORTS	UEPLD	25.07	31.8	31.86	1973	223	12.11	82	22.23	3
CALLING PORT	UEPTZ	\$	\$	¥	\$	2	¥	¥.	2	2
2-WIRE VOICE UND ACTED 1-WAY OUTGOING PIST TENNESSEE										
PANIES VOICE LAGINOLED SAMAY COMBULTICADE PART PART PART PART PART PART PART PART	UEPTO	≨ £	≨	ž	₹.	¥ £	≨ \$	¥¥	¥	2
2-WIPE VOICE UNBLACKED PRX TOLL TEPAINAL HOTEL PORTS	UEPXB	102	81.62	91.89	1923	223	12	85.00	£23	2
2: WHE VOICE LABORACIED PRIX LD DOD TERMINALS PORT 12: WHE VOICE LABORACIED PRIX LD TERMINAL SWITTS BOARD PORT	UEPXC	2003	3 13	8 1	1923	223	1123	88	223	2 2
2-WIPE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOAPD DO	2							3	R	2
CAPABLE PORT	UEPXE	25.07	20 15	21.00	1973	82.23	11.53	80 24	22	3

MELLEGUTHARWEGUTH AATES METWORK ELEMENTS AND OTHER BENNELS

ĽŁ	SCRIPTION STATES OF STATES	USOC	7	ď	43	KY	5	M	¥	3	N
=	CALING PORT WITHOUT UD	DEPAG	7	2	ž	\$,,	;	:		
Н	2-WIRE VOICE UNBUNDLED PRIX KENTUCKY LLID AREA CALLING PORT	UEPXG	€ ₹	≨ ≩	≨ ≨	923	≨≨	\$ \$	¥ ž	ž	ž
±	2-WIFE VOICE INBUINDLED PRIX KENTLICKY PREMIUM CALLING PORT	UEPXH	≨	ž	≨	1923	₹ 3	§ ≥	5 ≥	≨	ž
	WITHOUT LUD	UEPXI	\$	**	¥M	3	1	;	:	:	
=	2-WHE VOICE UMBUNDLED 2-WAY PRY LOUISIANA LOCAL OPTIONAL						2	5	Ş	٤	≨
 -	2-WIFE VOICE UNBUNDLED 2-WAY PBX HOTEL AND SPITAL ECONOMY	UEPXK	≨	ž	≨	ž	83	ž	≨	ž	ž
±	ADMINISTRATIVE CALLING PORT 2-WER VOICE LINE ACT HT 8-WAY BOY LOTTE ACCESS THE RECOLUMN	UEPAL	200	29.55	\$1.85	19.23	823	12.13	82.08	8.3	7
_	ROOM CALING PORT	UEPXM	\$2.07	2	28	15.61	83	3	8	2.4	2
	2-WINE YONGE UNBANDLED 1-WAY OUTGOING PRIX HOTEL/HOSPITAL ECONOMY ADMINISTRATIVE CALLING PORTTENESSEE CALLING PORT										
1	2: WIFE VOICE UNBUNDLED 1: WAY OUTGOING PRY HOTEL ANSFITTAL	CEDION	≨	ž	ž	ş	\$	ž	ž	ş	7
1	AWIND VACE INDIVIDUAL TO TAXA OF TOWN TO BE THE TOWN	CEDICO	20.03	20.5	28	19.23	\$2.20	11.23	8	X	7 8
	DECOUNT CALLING PORT	UEPXP	ž	ş	≨	*	8	¥X	¥	\$	42
=	CALLING PORT.	UEPXO	ž	2	1	3	3	3	:	1	:
	2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL OPTIONAL						٤	i k	٤	٤	Ş
\pm	Z-WIRE VOICE UNBLACUED 1-WAY OUTGOING PROJECTSURED PORT	LEPXS	₹£	¥ S	≱ .	¥	¥	2	ž	ž	\$
	E-WIRE VOICE LABOURDLED E-WAY PBX SOUTH CAROLINA AREA PLUS				3		2	Į.	8	8	2
£	2-WIPE VOICE UNBURDLED PRIX COLLIERVILLE & MEMPHIS CALLING	UEPXT	≨	ž	ž	ž	≨	ş	ž	23	ž
\exists	PORT AND THE RESERVENCE OF THE PERSON OF THE	UEPXO	ş	ž	NA NA	ž	¥	ž	ž	≨	2
\exists	CALING PORT	UEPXV	ş	ž	ž	¥	ş	ž	\$	4	47
Ξ	CONTROL FOLLOW RELIES AND 1800 REPORTED BY SAME RESIDENCE.	V 20 4									
Н		1									
\pm	NOCAL NUMBER PORTABLITY (REQUIRES ONE PER PORT)	UNICE									
E	##C-1#	1 SPPC	5	27 623	:::	27, 52					
Ц	2 WHE VOICE UNBUNDLED COMBINATION 2-WAY PRIX TRUNK - Residen	UEPRO	2010	2 23	\$17.16		200	2 2	2 2 2	8 2	2 2
Ŧ	LINE SIDE UNBURNOLED COMBINATION 2:WAY PRIX TRUME - BUSINESS	S C	8	95.28	\$17.16	126.47	\$16.43	822.98	24.04	24.36	28.63
\coprod	LINE SIDE UNBUNCHED INCOMING PRIX TRUNK - BUSINESS	UEPPI	2 2	20.00		2	25.5	8 23	2404	24.38	28 63
-	LONG DISTANCE TERMINAL PBX THUMK-BUSINESS	UEPLD	251.50	85.29	\$17.16	28.47	25.6	22	240	2 2	2 5
I	TH CHWAY CALING PLAN PRY TRIAK - BUSINESS	CEPTS	22.23	55 286	\$17.18	298.47	\$16.43	8522.88	1010	\$24.36	88
E	2-WIFE VOICE UNBUNDLED 3-WAY COMBINATION PBX ALABAMA	21.23	Z X	200	817.18	28.47	216.43	22.23	5,53	24.38	89.60
\pm	CALLING PORT A WINE VOICE IN THE BANK PER SAME A PARTICIPATION OF THE SAME AND THE	UEPAR	22 122	ž	ş	ş	¥	ž	ž	¥	Ş
\exists	CALLING PORT	20	\$	£	\$	¥	234.63	Ž	4	***	1
1	2-WHE VOICE LAGRACUED PRIX LD TERMINAL PORTS	UEPLD	23.153	\$62.54	\$17.16	\$36.47	\$16.43	822.88	24.0	87.4	€ ≨
	CALING PORT	UENTS	ş	¥	ž	\$	774	1	3	;	8
_	2-WIPE VOICE UNBUNDLED 1-WAY OUTGOING PRY TENNESSEE	the first	:	;					1	5	2
Н	A: WIRE VOICE UNBLAIDLED 2: WAY COMBINATION PBX LISAZE FORT	UEPXA	Q I Z	25.03	817.16	(P 90 3	\$1643	≨ &	₹20	¥ S	2 S
1	2-WHE VOICE UNBUNDED PRIX TOLL TERMINAL HOTEL PORTS	UEPXB	21.2	20.03	\$17.16	\$36.47	\$16.43	\$22.88	20.50	24.38	2 2
I	Z-WHE VOXZ UNEUKOLED PRX LD TENANAL SWITCHBOARD PORT	S C C C C C C C C C C C C C C C C C C C	2 5	2 2	27.2	236.47	\$16.43	822.88	1010	24.36	28 83
Е	AWIFE VOICE UNBLANDLED PRIX LD TERMINAL SWITCHBOARD IDD						2	R	5	× ×	2
1	2-WIRE VOICE UNBLACK FOR SWAY PRINCIPAL PROVIDED ABOVE	UEPXE	8.2	200	817.18	238.07	\$18.43	822.88	124.04	24.38	20 65
4	CALING PORT WITHOUT LID	UEPXF	ž	≨	ž	17.803	¥	¥	2	2	\$
1	2-WHE VOICE UNBUNCALD PRIX KENTICKY LUD ANEA CALLING PORT	GEPXG	ž:	≨:	\$	17903	ž	ž	ž	¥	¥
E	2-WIFE VOICE LINBLINDLED 2-WAY KENTLICKY APEA CALLING PORT	57.03	£	٤	٤	2	\$	ž	ž	ž	ž
Ξ	SAME VACE INDICATE STATE OF SAME STATE STATE STATE SAME VALUE AND SAME STATE SAME SAME SAME SAME SAME SAME SAME SAM	7495	Ş	ž	¥	238.47	¥	≨	ž	ž	ş
	CALING PORT	UEPXX	ž	ş	ž	ž	\$16.43	ž	ž	ž	£
	Z-WPIE YOK-E UNBURELED Z-WAY PBX HOTELAHOSPITAL ECONOMY ADMINISTRATIVE CALLING PORT	d de	8	25 226	\$17.18	28.0	57 915	3	2.5	2,58	:
	2-WIPE VOICE UNBUNDLED 2-WAY PRX HOTEL/HOSPITAL ECONOMY	į	1	╁				R	5	8	2 2
]		CEPXIN	2	85 28	817.18	238.47	\$16.43	22.38	20404	85 4 38	\$ 80

METAGOLTHAREWINGUTH RATE: METWORK ELEMBATS AND STRAIN ASTROCES

8	DESCRIPTION	DI POC	7	2	ð	č	4	577	9	9	71
	2-WIFE VOKE UNBLADLED 1-WAY OUTGOING PBX HOTELANGSPITAL ECONOMY ADMINISTRY CALLING PORTTENESSEE CALLING PORT PRINGS PORT PRING	UEPXN	≨	\$	≨		ž	₹	2	2	S
	2-WIRE VOICE UNBLADLED 1-WAY OUTGOING PBX HOTELFIOSPITAL DIACOUNT ROOM CALLING PORT	UEPXO	\$21.53	95 Z9\$	\$17.16	536.47	\$16.43	22.38	8	87.23	26
Ξ	2-WIPE YORKE UNBUNDLED 1-WAY OUTGOING PBX LOUSLANA LOCAL DISCOUNT CALLING PORT	UEPXP	¥	ž	≨	ž	\$16.43	Ž	ž	ž	42
	2-WIFE VOICE UMBUNDLED 2-WAY PBX MISSISSIPPI LOCAL ECONOMY CALLING PORT	OKER	ş	2	\$	\$	¥	8	2	1	2
	2-WIFE VOICE UNBLINDLED 2-WAY PRX MISSISSIPPI (OCAL OPTIONAL CALLING PORT	UEPXB	\$	\$	1	3	42	5	2	1	\$ \$
	2-WITE VOICE LABOURD ED 1-WAY OUTGOING PRODEKSURED PORT	UEPXS	221.83	85.28	\$17.16	296.47	\$16.43	22.88	15104	85,53	08.63
\exists	CALLING PORT	UEPXT	¥	ş	ž	¥	¥.	ş	ź	\$24.36	≨
	2-WIRE VOICE UNBUNDLED PBX COLLIERVILLE & MEMPHIS CALLING PORT.	UEPYO	§	≨	\$	¥	ş	ž	ž	2	8
	2-WIPE VOICE UNBUNDLED 2-WAY PBX TEMMESSEE PEGIONSERV CALLING PORT	LEPXV	ž	ş	≨	¥	≨	ş	ž	\$	26
#	Scheequari Activity	OSYSU	\$10.00	\$10.00	\$10.00	\$10.00	30.013	\$10.00	\$10.00	810.00	810 00
#											
	2 WITHE VOICE UNBUNDLED COMBINATION 2-WAY PRX TRUNK - Readent	UEPRO	81.23	228.73	\$17.16	\$36.47	\$16.43	\$22.96	521.60	\$24.36	9 9
#	LINE SIDE UNBUNDLED COMBINATION 2-WAY PRIX TRUME - BUSINESS	UEPPC	3.5	28.20	\$17.18	1790	\$16.43	22.98	59 65	\$24.38	61 62
	LINE SIDE UNBUNCHED INCOMING PBX TRUNK - BUSINESS	- Septiment	z S S	2 2 2		1000	218.43	8 22	8 8	8 8	2 Z
1	LONG DISTANCE TETARNAL PBX TRIAN BUSINESS	UEMO	DS 1.23	0.63	\$17.16	17903	\$16.43	22.98	89.05	24.38	61.63
	TH OUTWAND CALLING PLAN PBY THINK - BUSINESS	UEPTO	2 2	2 8 8	817.18	206.47	516.43	\$ 22 \$2 \$2 \$2 \$3 \$3 \$3 \$3 \$3 \$3 \$3 \$3 \$3 \$3 \$3 \$3 \$3	8 8	8 2	60 00
	2-WIPE YOU'S UNBUNDLED 2-WAY COMBINATION PBX ALABAMA	16843	\$	1	;	3	;	1	;	:	
	2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA			1	£ :	§ :	\$	ξ :	Š	§	2
1	Z-WIPE VOICE UNBUNCLED PBX LD TERMAN, PORTS	OF THE	8	¥ 85	¥1,71	¥ %	218.43	¥ 65	≨	¥	¥
	2-WIFE VOICE LABURIOLED 2-WAY COMBAUTION PRX TENNESSEE	CEPTE	ž	ž	ž	1	¥ N	V.	2	72	8
	Z-WIPE VOICE UNBLANDLED 1-WAY OUTGOING PIEX TEINLESSEE	1		:							2
\pm	2-WINE VOICE UNBUNDLED 2-WAY COMBINATION PLIX USAGE PORT	CEPX4	¥ 5	≨ 83	817.18	NA 47	¥ sie	NA CO	¥ 8	¥ 6	9 E
	S-WARE VOICE LABORICAED PRIX TOLL TERMINAL HOTEL PORTS	CEPYOR	8:2	۲ چ	\$17.16	236.47	\$16.43	\$22.98	89.03	\$24.36	89.19
1	LAWRE VOICE UNBUNDED PAY ID DOD TEHNING PORT	CEPTO	8 8	2 E	\$17.16	200	200	22	88	24.38	\$ 65 5
	2-WIFE VOICE UNBUNDLED PAX LD TERMINAL SWITCH BOATD IDD	BANGE !	\$						3	2	2
	2-WINE VOICE UMBUNOLED 2-WAY PBX KENTUCKY ROOM MEA						2	R	3	85 858	
1	DANIE VOCE UNITADI ED PRI KENTICKY I ID MEA CALTIAG PORT	UEPYG	≨ ≨	≨ ≨	\$ \$	4 1	\$ \$	≨ ;	¥	≨ 5	≨ 5
	2 WIRE VOICE UNBURDLED PRIX KENTUCKY PREMUM CALLING PORT	UEPXH	ş	≨	ž	536.47	5 ≱	3	₹ ≨	¥.	ž
_	Z-WIFE VOICE UNBUILDLED Z-WAY KENTUCKY AREA CALLING PORT WITHOUT LLID	UEPXI	≨	ž	≨	17:803	≨	\$	٧¥	ž	ž
	2-WINE VOICE LABURACILED 2-WAY PRX LOUISLANA LOCAL OPTICINAL CALLING POFT	UEPYX	¥	ž	ş	\$	816.43	\$	ž	ž	2
	2-WINE VOICE UNBUNDLED 2-WAY PBX HOTELAIOSPITAL ECONOMY ADMINISTRATIVE CALLING PORT	de ye	8	2	\$17.18	138.47	216.43	22.00	88	W 763	2
	2-WIRE VOICE LINBURIDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY ROOM CALENG PORT	NO. ASIA	81.58	2	817.18	28.47	818.43	22.88	8	\$24.38	2
	2-WIFE VOICE LIBITIONICED 1-WAY OUTGOING PRIX HOTELANDSPITAL. FOONDAY ADMINISTRIC CALLING POPITTEMESSEE CALLING PART.		3	1	3	1	3	1	1	3	:
	A WIPE VOICE UNBLANCED I WAY OUTGOING PEXHOTEL/HOSPITAL DIACOUNT ROOM CALLING PORT	9	8,0	£	8 2	2 2	2,842	2	20	7.76	2 2
	2:WHE VOICE UNBUNCLED 1:WAY OUTGOING PRA LOUISIANA LOCAL PROCESS INT CALL HAS POST	200	43	1	4	42	27 873	42	1	4	4
	2-WIFE VOICE UNBLACKED & WAY PBY MISSISSIPPI LOCAL ECONOMY										
1	2-WIFE VOICE UNBUNCLED 2-WAY PRY LINSSISSIPPI LOCAL OPTIONAL	Q.	S	\$	≨ .	≨	≨	8	ž	<u> </u>	ž
1	CALLING PORT 2-WIRE VOICE UNBLIED 1-WAY OUTGOING PROJECKSURED FORT	LEP XS	¥ a	¥ 83	¥ 25.718	¥ (₹ 90)	N. 8. 6.3	22 23	¥ 50	¥ 70	≨ 5
	Z-WIPE VOICE UNBLANDLED Z-WAY PBX SOUTH CANDLINA APEA PLUS	UEPXT	ş	2	\$	Ş	\$	2	ž	× 72	\$
	2-WIFE VOICE UNBUNDLED PRIX COLLIERVILLE & MEMPHS CALLING	i Kebali	3	3	1	;	1	3	1	:	
1		UELVI	£	٤	٤	٤	1	٤	٤	Ę	2

METWORK ELEMENTS AND OTHER SERVICES	
-------------------------------------	--

2	SCHEPTION	1800	4		10	3	•		-		
E	2-WIRE VOICE LABUADLED 2-WAY POX TENNESSEE REGIONSERV			•	5	Ž	5		2	2	2
Ξ	THE LOWER PARTY OF THE PARTY OF	UEPXV	≨	ž	¥	≨	≨	≨	ž	≨	89.19
	MMC - Disconnect Charge - 1st										
	2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residen	UEPRO	\$6.21	\$26.37	¥	2	413	3,5	VA.	ΨM	37 63
	LINE SIDE UNBUNDLED COMBUNATION 2-WAY PRX TRUNK - BUSINESS	UEPPC	18 21	\$26.37	ž	ž	11 13	3	Ž	2	2 2
1	LINE SIDE UNBUNDLED CUTWARD PBX TRUNK - BUSINESS	UEPPO	\$6.21	\$24.37	¥	ş	23.77	\$6.56	2	¥	3
1	LINE SIDE LINBUNDLED INCOMING PBX TRUNK - BUSINESS	UEPP1	\$8.21	\$26.37	¥	ž	#3.77	95.95	¥	ž	28 (2
1	LONG DISTANCE TERMINAL PBX THUNK-BUSINESS	UEPLD	\$6.21	\$26.37	M	ž	49.77	35.33	ž	Ž	39 03
1	THE 2-WAY CALLING PLAN PEX THUNK - BUSINESS	UEPT2	28	\$26.37	≨:	ž	113	95.95	Ş	≨	99 (3
I	2-WIRE VOICE INDIDATED STATE COMMINATION BOY AL ABOUT	OI LE	1294	E-18.37	≨	≨	121	95.98	ş	¥	99 03
=	CALING PORT	UEPAS	3	*	2	2	×	3	2	•	;
	2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PRX LOUISIANA								٤	ž	2
-	HANDE INC. INC. INC. INC. IN CO. IN C	UEPL2	ž	ž	ž	ş	7.03	¥	¥	Ş	ş
İ	E-WINE VOICE UNEXHALLED PICK LD TERMINAL PORTS E-WINE VOICE UNEXHALLED SAVEY PARAMETERS	OEMO	12 21	£ 33	≨	≨	11.13	3 5	≨	¥	23 28
	CALLING PORT	21.695	ž	≨	≨	₹	ž	¥	¥X	2	8
	2-WINE VOICE UNBUNDLED 1-WAY OUTGOING PEX TEMMESSEE										3
Ī	PARTIES PORTY THAN AND A TAXABLE MANAGEMENT REAL PROPERTY OF THE PARTIES OF THE P	UEPTO	¥	¥	5	≨:	ş	414	ž	ž	20.00
İ	2-WINE VOICE LINERACT FOR TOT THEM AND HOTEL BANGE	LEBYS .	200		≨ :	≨:	1,13	95 93	Ş	ž	3
E	2-WIPE VOICE UNBUNDLED PRIX LD DDD TETAMALS PORT	UEPXC	Ś	4 2 2	2 3	ž	2 2	8 3	2	≨ :	3
	E-WITE VOICE UNBURDLED PRIX LD TERMINAL SWITCHBOARD PORT	Œ	12.8	\$26.57	≨	ž	40.71	3	ž	ž	38 63
_	2-WHE VOICE UNBUNDLED PRIX LD TERMINAL SWITCH-BOARD IDD	ļ				:					
Ė	P.W.F.E. VOICE UNBUNCLED 2-WAY PEX KENTLICKY ROOM AREA	VETAE		Š	Ş	2	11.03	3	≨	¥	200
	CALLING PORT WITHOUT LLID	UEPXE	≨	-	≨	\$	\$	ž	¥	**	2
1	P-WIFE VOICE IMBURDLED PRIX KENTUCKY ULD APEA CALLING PORT	UEPXG	₹	ş	ž	ž	\$	¥¥	ž	≨	ž
1	D. WICE WORK I MAI MAI DO S. WAY DEATH INC. AREA 7 W. LAND DOFT	UEPX	ž	≨	ž	ž	¥	Ş	Υ¥	ž	ž
	WITHOUT LLD	CEPY	ş	ž	ž	\$	ž	ź	2	\$	4
	Z-WIFE VOKCE UMBUNCLED 2-WAY PBX LOUISIANA LOCAL OPTIONAL										
‡	D. WIGE VOICE (MR ANY ET) S.WAY PRIVIDITE LANGUAGE EVANAVIOR	NE NE	§	ž	¥	≨	23.77	ž	ş	ş	¥
	ADAMASTRATIVE CALLING PORT	CEP.20	\$6.2 1	20.37	ź	ž	123.77	25.25	ş	2	8
	AWIFE VOICE UNBUNCLED & WAY PEX HOTEL/HOSPITAL ECONOMY										
丰	PARTIES WHOSE LINES HOLD STILL AND OUTTOOMS PRIX HOTEL ANGEST A	CENTRA	28.21	28.37	ž	ş	12.77	20 20	ž	≨	22 28 28
	ECONOMY ADMINISTRATIVE CALLING PORTTENNESSEE CALLING	NO.	2	24.17	\$	Ş	4	4	\$	3	3
	Z-WIPE VOICE UNBLINDLED 1-WAY OUTGOING PEX HOTEL/HOSPITAL							5		٤	8
1	SWIDE VOOR I ME ANY ED CARACTERISTICS INVITED TO SECURITIONS	OKEN	ž	28.37	≨	≨	12.03	33	≨	≨	8
	DISCOUNT CALLING PORT	UEPXP	12.88	ş	. ≨	ş	12.03	2	¥		ş
_	2-WITE VOICE LINBLAIDLED 2-WAY PRX MSSISSIPPI LOCAL ECONOMY CALLING PORT	CIKEDIO	\$	1	4	1	1	3	1	1	1
	2-WINE VOICE UNBUNDLED EWAY PRO LASSISSIPPI LOCAL OPTIONAL										5
‡	D.W.INC. VOICE INTERNAL PO L.WAY OF PRODUCE DESCRIPTIONS	UEPXH	ž,	ž	≨ :	≨ :	≨ ;	3	≨ :	≨ :	ž
	2-WIRE VOICE UNBUNOLED 2-WAY PBX SOUTH CAPOLINA APEA PLUS	2		0	٤	٤	2	8	<u> </u>	ž	2
1	CALLING PORT	UEPXT	≨	ž	ž	\$	¥	ž	ž	ş	¥
	PORT	CEPYO	≨	¥	2	<u></u>	2	2		2	2
	2-WIRE VOICE UNBUIDDED 2-WAY PRIX TENNESSEE REGIONSERV) BEDAY	1	;	3	;	;	:	1	:	
				1	5	٤	Š.	٤	٤	2	8
	MMC - Discorrect Charge - Add1										
#	2 WIPE VOICE UNBUNOLED COMBINATION 2-WAY PBX TRUNK - PRESENT		2	88	≨:	≨ :	403	95 95	ž	Į.	2
1	LINE SIDE LINEUNICED OUTWAND PRY TRIBBY - IN SUMESS	200	2 3	2 2	2 2	Į į	1 2 2	2 3	≨ á	≨ \$	X 5
Ė	LINE SIDE LYBUNDLED INCOMING PRIX TPLAK - BUSINESS		28.2	2	! ≨	5 ₹	143	2 3	٤٤	٤٤	S K
	LONG DISTANCE TERMINAL PBX TRUMK-BUSINESS		\$6 21	23.53	ž	ž	137	25.93	¥	ž	22.23
<u>‡</u>	TIN 2-WAY CALLING PLAN PRY THURK - BUSINESS	UEPT2	\$4.21	8 5	ž	ž	72	3	ž	ş	28.23
#	THE COTTON AND CALLING PLAN PER TRUME. BLOSINESS DEWINE WORSE LARGEN TO SWAY COMMINATION PRY ALABAMA.		28 21	8	≨	≨	227	38	ž	ž	2
	CALLEG PORT	UEPAZ	12.88	ž	ş	Ş	¥	\$	¥	Ž	¥
	2-WPE VOICE UNBUNDLED 2-WAY COMBINATION PAX LOUISIANA CALLING PORT	2	\$	ş	\$	ş	11.83	¥	\$	\$	¥
	2-WINE VOICE UNBUNCKED PAX LD TEPAMAN, PORTS	UEPLD	12.5	\$1.69	≨	ž	13.77	95 93	ž	ź	æ
=	Z-WPE YOCE UNBUNDLED Z-WAY COMBINATION PBX TEMPESSEE ICALING PORT	CLASS.	1		\$		\$	4	1	7	8
1				<u> </u>	1	2	٤	Š	٤	5	×

LECUTIVICATES NETWORK ELEMENTS AND CTIVES GENERATES

Committee Comm	WITH VOICE I MAN MAN EN 1.WAY OF MODALS PRANCESSEE	2080	2	4	3	₹	3	3	2	2	7
Color Colo	LUNG PORT	UEPTO	ž	ž	Ş	¥	ş	ž	ş	ž	26 23
State Stat	INC. VOICE UNGUNULED 2:WAY COMBINATION PEX USAGE PORT	UEPXA	28.2	22	ž	ž	72	95 95 95	≨	ž	26.23
MACE MACE	INC. VOICE UNBURNOLED PRATICIL LEMMINAL HOTEL POHTS	UEPXB	26.21	88	¥	ž	23.77	86.56	ž	1	25.35
MACED SHIP (TENNING SHIP) COLOR	HE VOICE UNBONIA ED PBA LD DOO TEHMINALS PORT	UEPXC	\$6.21	88	¥	≨:	113	\$6.56	ž	ž	28.23
March Marc	REVOICE UNBUNCLED PEX LD TERMINAL SWITCHEDARD FOR	OETAD	2	28	ž	ž	12	92 92	ž	≨	26 23
NOTE NOTE	WALE PORT	UEPXE	12.31	2	ş	ź	\$3.7	\$6.56	2	≨	25.22
March Marc	IRE VOICE UNBUNCLED & WAY PBX KENTLICKY ROOM AREA	1	;								
MACES SAVIV PRINCES MACE	REVOKS INVESTIGED PRIX REMINISTY (1.0) ARE A CALL BACK PARTY	CENT	≨ ş	≨ :	ž	ž	≨ :	≨:	≨ :	ž	≨:
Note 15 years for the Collect Person Name Nam	THE VOICE UNBURGLED PRIX KENTUCKY PRBAIUM CALLING PORT	CEP XX	£ ≨	₹ ₹	ž ž	Y X	2	2 3	ž	2	¥ 2
MACKED PART PORT LOCATION Corner Corne March March Corne Corne March Corne Corne March Corne Corne Corne March Corne	IRE VOICE UNBUNCLED 2-WAY KENTUCKY AREA CALLING PORT	1		:		:				<u> </u>	5
March Marc	PREVOICE UNBENDLED 2:WAY PBX LOUSINA LOCAL OPTIONAL	CEPA	≨	ž	ź	≨	≨	≨	ž	≨	ş
March Marc	LING PORT	UEPXK	ş	ş	ž	Ş	13.77	≨	ş	ž	ž
MANUEL DE NAVA TREE COLONY RECOGNETAL LEPTON BEST STEED IN NA NA NA NA NA NA NA NA NA NA NA NA NA	NE VONC UNBONDLED Z-NAT PBA NOTELINOSPITAL ECONOMY INSTRATIVE CALLING PORT	DE PO	*	91.5		¥X	2	2	2	1	\$
NATIONALED : NAY OLYGONG PER PORTING AND STATES AND NATIONAL CONTROL OF THE PART OF THE PA	PRE VOICE INFORMELED & WAY PRIX HOTELARSPITAL ECONOMY. NA CALLING PORT	70.4.51	2	8	1	1	# 5			1	
MACE MACE	HE VOICE UNBUADLED 1-WAY OLTGOING PEX HOTELAGOSTIAL. MOAY ADMINATRATIVE CALLING PORTTENNESSER CALLING		:	:	;					1	ž Ž
MACRE 1974 1975	PRE VOICE UNBLACLED 1-WAY OUTGOING PRIX HOTEL HOSPITAL.	Constant	1	2	5	Š i	2 :	2	≨ :	\$	2 2
MACED WAY PAY BASSASIFF COLUMN CEPTON WAY WAY PAY PAY BASSASIFF COLUMN CEPTON WAY WAY PAY BASSASIFF COLUMN CEPTON WAY WAY PAY COLUMN PAY PAY BASSASIFF COLUMN CEPTON WAY WAY PAY COLUMN PAY PAY BASSASIFF COLUMN CEPTON WAY WAY PAY COLUMN PAY PAY BASSASIFF COLUMN CEPTON WAY WAY PAY COLUMN PAY PAY BASSASIFF COLUMN CEPTON WAY WAY PAY TOWN PAY PAY PAY TOWN PAY PAY TOWN PAY PAY TOWN PAY PAY TOWN PAY PAY TOWN PAY PAY TOWN PAY PAY TOWN PAY PAY TOWN PAY PAY TOWN PAY PAY TOWN PAY PAY PAY TOWN PAY PAY PAY TOWN PAY PAY PAY PAY TOWN PAY PAY PAY TOWN PAY PAY PAY PAY PAY PAY PAY PAY PAY PAY	THE VOICE UNBLANDED 1-WAY OUTGOING PRIX LOUISIANA LOCAL				5	£	à	R	٤	٤	***
MACED FWAY PAX MASSESSHIPT COAL OFFICIAL VEPAG	RE VOICE UNBUILD 2-WAY PRIX LASSISSIPPI LOCAL ECONOMY	CENT	ž.	ş	≨	ž	12.02	8	ž	ž	≨
MACED 14/14 PM CARCALERATE CALLED CARCAL STATES STATE CARCAL STATES CARCA STATES CARCAL STATES CARCA STATES CARCA STATES CARCA STATES CARCA STATES C	LING PORT	UEPXO	ž	ş	ş	ž	3	2	ş	ş	¥
MACE STATE	LING FORT	UEPXS	¥	ž	¥	ş	¥	\$6.56	ş	¥	ž
MACLED FAX COLLEGALLE LEMENTS CALLING LEPTY N. N. N. N. N. N. N. N. N. N. N. N. N.	RE VOICE UNBUSILED 1-WAY BUILGOING PROMEASURED PORT RE VOICE UNBUSILED 2-WAY PBX SOUTH CAPOLINA APEA PLUS	UEPXS	28.21	8 -	\$	ž.	13.77	3	≨	ž	क्रक
MACKED FWAY PRATESSER REGIONSENT LEPYN NA NA NA NA NA NA NA NA NA NA NA NA NA	LING PORT	UEPXT	¥	ž	ž	¥	ž	ž	ş	¥	¥
MACLED 2-WAY PRI TENERSEE RECONSERY UEPVY NA NA NA NA NA NA NA N	REVOKE URBURALED FBX COLLIENVILLE & MEMPTHS CALLING	UEPXÜ	¥	ž	ž	ž	ž	¥	≨	ž	X
Particular Par	PRE VOKCE UMBUNDLED 2-WAY PBX TENVESSEE REGIONSERV LING PORT	UEPXV	ž	ş	ş	ž	\$	2	\$	ž	2
### Source											
Table - Internal Service Order-118 SOLAMN \$7.37 \$2.14 \$18.04 \$25.20 \$25.30 \$25.40 \$21.86 \$21.8	- COSA - COSS LEA Cherga, Electroria, per LSR received from the CLEC by one - CSS defeature transferon.	Canada	Ş	8	Ş	8	8	1	1	1	
Targe - Manual Service Order - Clean Control - Couldn's See learner \$15.75 \$15.44 \$15.77 \$15.44 \$15.74 \$15.74 \$15.74 \$15.74 \$15.74 \$15.74 \$15.74 \$15.74 \$15.74 \$15.74 \$15.74 \$15.74 \$15.74	- Incremental Charge - Manual Service Order - 1st	SOMAN	127.37	8512	\$18.94	ž	\$18.14	22.52	2 S	21.00	\$20.35
Part Part	- Incremental Charge - Manual Service Order - Decornect - 1st	SOLAN	817.77	2 2	2 ₹	≨≱	83	2 2 2	\$12.76 NA	\$14.46	51054
### Part of the per mouth HTGUX See leafarms	- Incremental Charge - Manual Service Order - Discornect - Addition	SOMAN	\$0.46	13.14	ş	ž	ş	ž	ž	ž	91.40
Marca data Mar	deg Nanting, per ins per month - i fi		See befare		2 3	\$0.29 \$2.14	32	See features		See laatures See laatures	
### 12.00 ##	- A001	- 1 1	See feet ree		¥	\$2.14	¥	San factures		See leatures	
Section Sect	per more.		X 3	2 7	88	200	95.23	क्रम	**	42.	91.23
No. of the control	- Add		23123	22	\$17.16	16.7	27 91	8523	≨	22.73	22
Party - Manual Service Order - Let SOURM \$2.57 \$2.15 \$1.64 MA \$14.14 \$14.24 Party - Manual Service Order - Let SOURM \$2.52 \$2.15 \$1.64 MA \$14.24 MA \$14.24 Party - Manual Service Order - Charles Order - Charle	Obscornect Overge - 1st		12.08	22.78	ž	ž	\$4.15	95 93	¥¥	ž	39 63
Name of the Part of Decorated Collection SOLACH ST 27 ST 52 54 SA 22 NA SD 56 ST 57 SD 54 NA SD 55 SD 54 NA SD 55 SD 54 NA SD 55 SD 55 SD 54 NA SD 55 SD 55 SD 54 NA SD 55	- Incremental Charge - Manual Service Order - 1st	SOLLAN	28.53	2 2	¥ 8 8 8	\$ \$	2 2	2 2	ž	YX S	2 2
Number of the Colors of State of S	· Incremental Charge - Manual Service Order - Addi	SOMAN	812.87	85.23	24.2	ž	3	× =	\$	\$14.67	2008
	- Incremental Overse - Manual Service Order - Discornact - 1st	SOMAN	CE 91.5	3 3	≨:	≨ :	9863	\$16.06	≨:	≨:	\$13.32
March Marc	i seni taga promot ana ana ana ana ana ana ana ana		2	5 2	٤	٤	ž	٤	Ę	į	20 14
NA	FIATURES			170		1					
11.12 N. N. N. N. N. 15.22 \$0.55 \$11.0	ting Features othered with Port, Per month	WA	¥	cheme	ž	Charge	2	\$	\$	See above	£
## NA NA NA NA \$1.02 \$15.1 \$15	Calling, per stores.		\$1.15	≨ :	≨ :	≨:	ž	20.18	89 03	01.18	≨:
best Culting, per south NA NA NA NA S0 0755 \$0.175 \$0.1247 \$1.00 \$1.00 \$1.00 \$1.00 \$1.51 \$1.51 \$1.00 \$1.00 \$1.00 \$1.51 \$1.51 \$1.00 \$1.00 \$1.51 \$1.51 \$1.00 \$1.00 \$1.51 \$1.51 \$1.00 \$1.00 \$1.00 \$1.51 \$1.00 \$1.00 \$1.00 \$1.51 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00 \$1.00<	- Decorrect		3 3	٤ ≨	≨ ≨	2 3	٤ ٤	200	2	NA.	Į į
11.00 NA NA NA 10.00 11.51 11.	Changeshis Speed Calling, per month		8	ž	ž	¥	ž	\$0.0755	\$0.17	\$0.1247	≱
100 NA NA NA 100 100 100 100 100 100 100 100 100 10	Demmes		8 2	≨ 5	≨ :	≨ :	≨ 5	81.8	\$1.51	\$1.51	≨:
			88	₹ ₹	2 2	ž	≨ ≩	0000	2 0	SPS CE	≨ 2

DESCRIPTION	nsoc	オ	F	Ϋ́	ΚA	≤	SW	¥	36	ĭ
IMPC - Discorrance		\$0.55	ž	٧	ž	٧X	\$0.5466	¥	ž	ž
Remote Activation of Call Fordwarding, per month		2	ž	ž	ž	ž	\$0 4669	28 28	\$0.3743	ž
NBC - Demonstra		8	≨ :	≨ :	ž	≨:	2015	5 5	\$ 51	ž
Court Carlotte		8	ž	ž	ž	¥	30 5466	¥	ž	¥
Central Cell Watching, per month		500	ž	ž	ž	ž	20008	2003	660003	ž
		8	ş	ž	ž	۷×	21.02	\$1.51	\$1.51	ž
INT. LEGATION		8	≨ :	ž	ž	ž	\$0.5466	¥	ž	ž
		2 2	5 5	2	Ž.	Š	128	8	\$0.0015	≨ :
WC-Decorated		2 5	2 2	2 2	£ 2	4	20.00		5	2 3
Automatic Receil, our month		2 5	1	1	2	2 2	20160	2 5	20.5	2 3
MARC		8	ž	ž	ź	ž	2015	15.15	15.55	2
NARC - Discormed		88	≨	ž	ž	ž	\$0.5466	ž	ž	¥
Celling Number Delivery, per month		22.04	ž	ž	ž	¥	\$0.1817	50.33	20.2272	ž
NAC.		818	ž	ž	ž	٧×	\$1.02	15.18	15.18	¥
MMC - Decornect		\$0.56	ž	ž	¥	VV.	\$0.5466	¥	¥	¥
Cathing Number Delivery Blocking, per month		\$1.17	ž	ž	ΥN	٧V	\$0.9913	8	20.3684	ž
IMC		8	ž	ž	ž	¥	\$1.00	1.51	\$1.61	ž
INFC - Decomed		25.05	≨	≨	ž	ž	\$0 5466	ž	Ź	≨
Controver Unitridied Trick, per month		3	ş	×	ž	ž	80.1918	8	201-62	ž
MAD. Automate		8	Į.	≨	≨:	ž	20	191	51.51	ž
I INTO - LABORATION		50.35	ž	≨ :	≨ :	¥.	20.5466	≨.	ž	ž
MACON CAR INSTITUTE OF THE PROPERTY OF THE PRO		2 8	\$	S :	Į.	ž	127.1	20.03	825	¥.
MBC - Newword		3 5	ž	2	<u> </u>	٤.	20.5	6	10.10	ž
Substitute Controlling and march		3 5	5	5 3	ξį	2 3	8 5	¥2.0	2	\$
		3 2	2 3	į	ž	2 3	201.00	2	10.10	£ 3
WRC - Decornect		3 3	4	1	2 2	2 2	2073	2 47	10.13	2
Selective Call Acceptance, ser recrift		2	2	4	ž	1	9	5 5	Die Co	1
Jews .		818	¥	2	ž	MA	2015	3		
WC Decorned		55 05	2	2	ž	¥2	1875	1		1
Mutative Hunt Service (Rosery)										
Service per line, (in addition to part), per month		11.03	ž	¥¥	ž	ž	1221	71.03	1001	42
381		818	2	¥	2	Ş	20 18	\$1.61	15.55	2
MRC - Discormed		25.05	ž	2	ž	Ş	\$0.5466	¥	ž	2
Call Fornanding Variable, per month		\$0.08	ž	≨	≨	ž	2000	50 10	80,076.0	ž
14.5C		818	¥	ž	٧Z	ž	81.8	\$1.51	\$1.61	ž
HVC - Decomen		85.55	¥	ş	ž	ž	9975.03	¥	ş	¥
Cell Forestroing Busy Line, per month		80.03	۲¥	¥	¥	ž	\$0.0279	80.08	50 0603	ž
CHAI		\$1.03	≨	ž	ž	٧×	\$1.02	\$1.51	11:18	ž
NRC - Diacomed		85.08	ž	ž	¥	¥	\$975 O\$	٧¥	ž	¥
Cell Fernending Den't Avener All Calls, per month		8	≨	≨	≨	ž	\$0.0308	80.08	\$590.08	ž
WHC.		8	ž	¥	≨	ž	8.8	\$1.51	\$1.51	ž
IMPC - Decomed		\$3	ž	ž	≨	ž	\$05468	ž	ž	ž
Reacts Call Forwarding, per month		200	ž	ž	Ź	ž	\$1.47	88	21.41	ž
IMC		8	ž	ž	ş	ž	\$100		\$1.51	≨
Met C - Decornect		22	≨	ž	ž	ž	20 5468	٦	ž	ź
		21.02	ž:	Į.	ž	ž	1	Т	20 1382	≨:
Mary Discount	1	3 5	2 3	Ę	ž	Ş	20.00	T	10.14	\$ 5
Call Hold are morth.		3 5	2	1	1	2	8 8	Т	2 2	1
l Med		8.5	ž	¥	ž	ž	2015	Т	15.15	2
MPC - Decomed		\$0.55	₹	¥	ž	MA	\$0.5466	П	¥	2
Toll Recycled Service, per month		10 OF	ş	ž	ž	≨	\$0.0367	П	\$0.0743	ž
M.C		8	≨.	ž	ž	¥	8	T	\$1.51	ž
Met. Departed		8 8	≨ :	\$	ž	≨ :	99 5	T	ž	\$
BI.		3 8	2 3	2	1	2 3	60,00	1	200	
MEC Decorned		3 2	2	₹ ₹	1	2 2	97.00	Т	. ≨	5
Anonymous Call Pajection, per month		28.03	¥	¥	ž	Ş	\$196.03	81.29	\$1.13	\$
I Metic		818	ź	¥	Ź	¥	\$1.02	П	\$1.51	¥
_1		8	Ę	ž	ž	ž	50 5466	¥	ž	ž
Shared Call Appearances of a DN, per month		504	¥	≨ :	ş	ž	\$05015	2	20.00	≨:
	1	8 3	≨ :	≨ :	2	≨ :	8	2		\$ 3
THE PART OF THE PA		RES	5 3	\$ 3		\ { {	8 600	200	500	5
The state of the s		8 8	٤٤	٤٤	≨	₹ ₹	200	2 2	21.47	2 3
MFC - Dacomect		25.03	2	ž	Ş	ž	\$0.5466	ž	ž	ş
MON Bridged Call Exchanion, per month		88	ž	ž	ž	ž	\$0001	1100 08	\$0.000	ž
MHC		8	ş	ž	¥	ž	27.02	\$1.47	\$1.47	≨
NPC - Discornect		33	ž	≨	≨	ž	\$0.5466	2	≨	¥

į

BELL SOUTHWEWSOUTH RATES METWORK BLENEDITS

	Conf. by Conf. Account. But month		64.50		•				1 10014		
			,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	¥	٤	٧×	ž	2	20.5	10.3621	ž
19 12 13 14 15 15 15 15 15 15 15	NAC.		228.94	¥	ž	ž	≨	\$28.61	£10.33	\$33.36	ž
10 10 10 10 10 10 10 10	MRC - Discomnect		22.88	ž	≨	≨	¥	\$5.16	¥	≨	¥
100 100	Privacy Paleana, per month		10.03	¥	ž	ž	ž	\$0.000	\$00.00	\$0.0116	٧×
10 10 10 10 10 10 10 10	MAC		81.8	≨	2	ž	¥	8.18	\$1.51	\$1.61	¥
15 15 15 15 15 15 15 15	MAC - Decornect		95.08	ž	ž	ž	¥	8985 03	¥N	¥	Ą
11.00 11.0	Handle Assessment Chardren th ambier Colle over moreth		9	4	42	4	42	\$ 11.5	S	400	12
				5	2			2			£ :
10 10 10 10 10 10 10 10			3 3	5	٤:	2	٤	3			Ě
100 100			8	٤	٤	٤	Š	8	٤	٤	Š
10 10 10 10 10 10 10 10	Marte Set Busy, per storio		500	Ş	ž	ş	ž	\$0.0013	02000	200101	ž
March Marc			5 .8	≨	ž	ž	ş	\$1.00	\$1.51	\$1.51	ΚA
100 100	MATC - Discorment		3	≨	≨	ź	≨	\$0.5466	≨	ž	¥
March Marc	Trees Service (Res., Dist., Abriting Service), per month		50.15	ž	ž	ž	≨	101 05	92 OS	\$02149	٧×
Street S		_	\$103	2	2	ž	¥	8	\$1.51	\$161	¥
	Mary Designate		22.02	1		1	1	40 6464			3
			8 3	2	\$	5	2		5	5	5
No. 10 N	Cade respiction and Untertion, per mortin		3	≨.	Ş	<u>\$</u>	2	3	8	\$0.070g	≨.
	2		3	٤	ž	ž	Ş	20.5	10.15	2018	ž
	MAKC - Discontised		80.55	¥	ž	¥	ž	\$0 5466	ž	ş	ž
15.00 14.0	Cell Park, per month		20.03	ž	ž	ź	ž	\$0.0443	80 03	\$0.0694	Ź
March Marc	244		8.18	≨	¥	ž	¥	\$1.02	\$1.51	\$1.51	ž
March Marc	MRC - Disconnect		99 08	¥	ž	ž	ž	\$0.5466	ž	ž	ž
15.61 15.62 15.44 15.6	Authorizable (fine may second)		80.08	4	2	3	M		71 53	2	1
March Marc			8 3	3	47	1	3	8	13.13		1
Company	LED. Democrat		20.00		1			22.20			
March Marc			3	٤	٤	5	٤	200	٤	5	4
Company Comp	A COUNTY LAND THE ACT AND ADDRESS.									Ī	
March 1900	CATE TACK STEP FEATURES										
March 1900	Shared Frimery Number High Age On Each Add Termine	DSIF	2	8	8	2	180	2	2	2	8
March 1900	Secondary Only Dn (Shareshop-Shares) First Agreements	ysom T	SE E	2	2	2	<u>e</u>	8	9	ě	2
March 1900	Shared Secondary Only Dri-Frat Agar On Each Adolf Term	081F1	180	2	8	9	8	180	9	TBO	180
March Marc	Shared Non-ISDN DN	8	180	1	6	鱼	1	780	TBO	58	9
Line	Privacy Patence	DS1FU	OBT	081	980	OBL	081	180	180	ORL	DE C
Line	Manual Park atom	D\$150	GEL	9	180	Œ	TBO	CEL	GEL	180	GE.
Line	Call Properties Votes Or Voice/Date	N I	TES.	GEL	GEL	Ę	S	Ę	Œ	Car	1
Line	California Valenta - California	2011	2	i i	1		2	185	S	2	1
Line			3	3 5	2 1	2	3		2 4	3 6	3 5
Lichard The color The co	CONTROL VANIENDS - PARTY BACKS - VOICE	3		2		3	2		3		3
Line	CAN TOTAL VANCO - FEBRUA BUILD - LAG				3		2			9	
MAYO 180	Call Forwarding Busy Line - Young Or Votoe Data	TOCA	2	3	2	8	3	3	3	2	3
MANAY	Call Forwarding Busy Line - Data	00km	8	8	8	2	8	282	2	8	8
March 180 18	Call Frieding Busy Line-Prominds-Voice Or Voice/Data	MEAVA	<u>8</u>	92	2	2	2	180	B	8	£
High color	Call Forwarding Busy Line - Programmable - Data	PEAD	9	2	2	2	2	2	2	2	2
March 1800	Call Forwarding Don't Answer - Voice Or Voice/Data	ITSCV	180	180	<u>1</u>	180	90	180	180	2	20
March	Call Forwarding Don't Answer - Data	CONTI	180	180	180	180	180	TBO	180	180	TBO
Macros	Call Forecho Don't Angerer Promerable Votos Or Votos/Deta	VAR9VV	TBO	061	180	180	180	180	08	280	780
Marcola	Call Forwarding Don't Anamer - Programmable - Data	MEBOF	TEG I	081	081	2	8	180	<u>8</u>	8	TBO
Heart Hear	Call Fraction Mailtine Struttenens - Votes Or Voles-Oats	MBCVS	5	280	081	2	180	587	8	2	587
	Coll Engage to day of the College of the	THE COST	QE	ş	2	9	É	2	TBO	8	9
Hard	Contents Date Hold And Transfer	Neigh	9	1	2	S	QE	9	GE L	180	Q.
Harriston Harr	Ch. Was Codemon Day Med And Tourish	- I VED	2	9	2	2	ORL	CEL	Œ	12	2
HTGSO TRO TRO TRO TRO TRO TRO TRO TRO TRO TR	Later the best Green - Write Or Write Con-	1	2	2	1	2	9	9	180	180	TBO
Laborate Laborate	Table 1 to the One . Date	Children	G	1	Ş	Sat	9	Ę	Œ	É	SE C
Market 180 1		100	E E	3			9	ş	2	٤	2
No. No.	MALL AND AND AND AND AND AND AND AND AND AND	974	ş	1	1	2	Ę	1	18	S	GEL
PSIFICATION TRO	Lateral Material Waller Industria	ACAST.	2	2	2	E	CELL	130	٤	QEL C	180
March Marc	TABLE TO SERVICE STATE OF THE	50:50	1	Ş	3	É	1		1	ş	Ş
155 150	ACCRECATION FOR CALL	N. Co.						3 5	3 5	3 5	3
N. C. 180 18		2	2	8	3	8	2		3 8	3 8	
March Marc		200					2		3 5		
March Marc	Preferred Call Forwarding	Ž	2	200	2	2	2		2	2 5	
March Marc	Call Shock	YSY	Q	2	8	2	2	2	8	2	2
March Marc	Report Dialing	SS Y	2	2	2	2	2	2	8	2	2
Marchelle Marc	Per Line Blocking For Agencies/Line Enforcement	90¥	180	200	130	9	<u>g</u>	8	2	œ	8
	Per Line Blocking For Non-Pub Customens	NOBIN	180	55	8	ğ	1	180	180	180	180
March New Listed Customers March New List	Par Line Brothing For General Public	CORPC	9	2	180	081	180	380	180	CBT	190
Marchelle Marc	Per I se Rivetino For Nov-Puti. And Nov. I steel Customer	808PP	180	180	180	92	981	180	180	Œ	180
Conference	Day I has Brooken for through Comment	PAGON	CELL CELL	2	CEL	CEL CEL	180	38	8	8	180
Activities Act		Grand Co.	i i	F	1	ş	É	S	ZE C	8	5
Activation Act	Per Line Booking For North Condition	1		3 6	3		F	1	2	Car	182
March Marc		\$ 1	2	3 4	3 5	3 6	3			ş	2
Columbia Columbia	Proper During, Dental Of, Per Activation	2	2	2	2		8		3 5	3 5	
MACKING TBO TBO TBO TBO TBO TBO TBO TBO TBO TBO	Automatic LineOract Cornect	MEGINA	2	2	2		3		8 8		1
Machine Mach		2	2	3	3	3	2	3		3 5	2
Material Material	Selective Call Acceptance	MGK16	Q	8	8	2	8	3	3		3
Medico	Car PeriCal Rentwo	MSHP6	TBO	2	3	2	9	2	28	8	2
180 180 180 180 180 180 180 180 180 180	Call Transfer System Excitation	QTQ5M	180	180	180	90	780 88	180	9	8	2
180 180 180 180 180 180 180 180 180 180	Make Set Busy - Intracroup	COMSM	8	130	28	TBO	190	8	780 28	28	8
	I All Dustratural Code Restrictions	SPEC	B	OET	OET C	₽ 28	TBO	8	785 —	8	<u>6</u>

DESCRIPTION OF THE PROPERTY OF	3	ł		3	7	5	9	2	Ç,	=
Additional Limitings	Z.	1B0	Œ	5	2	180	780	68	TRO	185
Additional Listing No Rate	FLT	180	Œ	180	180	180	180	8	160	9
Cross Reference Listing	ΕŤ	081	780	180	180	180	TBO	180	8	180
Non-Pub Lasing No Rate	AP3	TBO	OBI	780	TBO	TBO	180	180	180	180
Non-Lies Liebra	ž	TBO	OBT	TBO	TBO	1 80	180	180	180	180
Non-Liet Leting No Pare	ĭ	TBO	OBT	180	180	180	180	780	8	OBT
Allomate Call Listing	¥	Ð	180	1 80	TBO	OEL	180	180	180	180
Menual Service Order Charge	SOMM	TBO	081	TBO	TBO	TBO	OBL	180	3	180
All Selective Clear Of Call Screening	\$RG↔	180	061	TBO	TBO	180	TBO	T80	TBO	TBD
ISLUM Mesesso Wrelling Indication-Lang, per manch		50.03	ž	Ş	Ž	¥Α	\$0.0105	\$0.0107	\$0.0136	ž
		8.18	¥	¥	ž	¥	\$1.00	\$1.47	\$1.47	ž
MFG - Decornect		\$0 55°	¥	¥¥	¥	¥	\$0.5466	¥	ž	ž
(SON Feature Function Buttoms		≨	ž	¥	¥	٧×	¥	≨		
		8	ž	MA	WA	٧¥	\$1.02	\$1.51	\$1.51	¥
ret C - Uncontract		\$0.56	≨	ž	ž	ž	\$0.5468	Ş	٧¥	Ν
Supplement (Annual Column - Control of the Bost)		ş	ž	ž	ž	ž	ž	ž		
Mary Charter And		8 2	≨ :	ž	ž	≨.	200	\$3.42	9213	ž
No. Land.		8 8	\$ \$	\$	≨:	≨:	8	8	120.71	≨
N-TC - Manual - Acti		8 8	2	2	\$ 3	\$ 5	2 8	20 12	27.25	¥
MRC - Decomed		10	2	**	4	5	3 2	2 3	8 2	٤ 3
						5	5	5		Š
Unbundled Port Usage Charges										
End Office Settathing (Part Usego)										
End Office Switching Function, per mou	¥	\$0.0018	\$0,0008846	\$0.0016333	50,000,00	10000	17775000	000010000	SO COL BOOK	and of
End Office Instruction Trunk Port—Shared, per mou	Υ×	П	\$0,0001683	\$0,0001564		20000	\$0,0001927	ž	\$0,0002581	¥

Tenden Substant (Fort Unique of Account (Account)	479	100000	20000000	1				- 1		
Tanken Internifica Trush Bod - Chanal nor man.	2	20000		/6/8000	900000	20000	NO 000/854		27 000 OF	\$0,000
			PLANTA I	W.WWC150	<u> </u>	2000	10.0002504	≨	10.0004034	ž
Common (Sharad) Transport										
Common (Shared) Transport per mile per mou	٧A	_	61/0000000\$ 6000000 05	80,00000	\$0,0000049	\$0,0000083	\$0,0000091	1	12 0000121	10 0000084
Common (Shared) Intragon Facilities Termination per mou	≨ Ž	\$0000	80 0004579	60 0004152	\$0,000.08	\$0,00047	\$0,0004281	П		1/2000000
NOTES				Ī						
American resistes subject to true-up.										
2. Transmission/waqp charpes associated with PDTS chout switched usage will also apply to obtain switched video switch chouf switched data transmission by IB-Charmes associated with 2-we SCM cont.										
Access to 8 Cherval or 0 Cherval Packet copatifies will be avail, able only intrody EFFNew Basimes Request Process. Resea for the packet copatifies will be determined via the Bons Fibe Requestive Busimes Faquest Process.										
4 This rate element is for those states which have a specific rate for User Profile loan B Demnet.										
5 This rate element is for use in those above with a different rate for additional manages of use										
Sifetes in TN and FL are interim and shall be trued-up when final nase are ordered.										
Where the state Commission has actored rates for the rate elements										
and to apply the same consistent with applicable PCC and Commission rules										
land orders.										

United Street S	U.D.C. E	\pm	INDUSTRICATED TO AND AND AND AND AND AND AND AND AND AND	2080	7	æ	8	٤	5	ş	ž	36	N.
U.D.P. HANG END SUO	159C 1600 1500	╃┥	Local Cherval - Dedicated - 3-Wee VG										
UDOP 814.66 825.23 813.99 825.85 815.99 815	Line March	Н	2-wre VG per mile	LSNC	00 OS	00.00	00.03	80.03	00 03	00.03	9008	80 63	W 93
UDDR SEAR	UDDY First	4-	2-wire VG Monthly Recurring per month AMC - 2-wire VG - 1-e	novz	\$14.61	28.23	13.91	823	8118	\$17.83	\$1483	\$16.83	20 61\$
U.D.P. ST.78 SEST N. N. N. ST.58 ST.78 N. N. ST.78 SS.24 SS.2444 N. S.153 N. S.154 ST.78 ST.78 ST.78 N. S.74 SS.2444 N. S.154 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 SS.2444 N. S.24 ST.78 ST.78 ST.78 ST.78 ST.78 SS.2444 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 SS.2444 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 SS.2444 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 SS.2444 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 SS.2444 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 SS.2444 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 SS.2444 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 SS.2444 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 SS.2444 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 SS.2444 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 SS.2444 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 SS.2444 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 SS.2444 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 SS.2444 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 SS.2444 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 ST.78 SS.2444 ST.78 ST.7	Unity \$77.81 \$25.91 N. N. \$25.93 \$77.95 N. N. N. \$25.94 N. \$17.95 N. N. N. N. N. N. N. N	ш	MHC - 2-wea VG - Addr		2 2 2	95.95 2.95 2.95 2.95 2.95 2.95 2.95 2.95	9 3	2	2 × 2 × 3		646.60	8878	\$19933
COMMAN STATE NA NA STATE S	Common N. C. Comm	_	NRC - 2-Wire VG - Disconnect Chg - 1st	aDv2	\$77.61	16.78	ž	≨	\$53.69	\$77.69	¥	Ž	\$54 B
SOWER 110 </td <td> SOME M. M. M. M. M. M. M. </td> <th>ш</th> <td>NRC - Manual Svc Order, per LSR</td> <td></td> <td>2 12</td> <td>32.03</td> <td>ž</td> <td>\$ 800</td> <td>3</td> <td>8 3</td> <td>≨:</td> <td>≨:</td> <td>8</td>	SOME M. M. M. M. M. M. M.	ш	NRC - Manual Svc Order, per LSR		2 12	32.03	ž	\$ 800	3	8 3	≨ :	≨:	8
SOUNCE 13.0 12.0 13.0	SOUNCE 15.00 12.00 13.	Ц	NRC - Manual Svc Order, per LSR disconnect	SOLLAN	¥	13.54	¥	ž	<u> </u>	<u> </u>	ž	≨ ≨	\$ ≥
Comm String Str	Column First Fir	4	NAC - Decion Seconds per St Assessed	SOMEC	8.5	X .	05.03	05.23	S 23	95.53	9503	95.53	95.53
1300 1300	1990 1900	Ц.	NRC 2 Wre VO - Incomertal Charge Manual Svc Order - 1st	MANOS	27.37	ž	16 91	2 2 3	\$18.14	25.50	M2.17	13 75 E	¥
COMM S177 NA	1906 500	┸		NO NO	\$18.37	≨	2	8	88	2	\$12.76	\$13.55	¥
U.D.P. First Fir	U.D.P. String S	_	NPC - 2-Whe VG - Incremental Charge-Manual Svc Order-Disconnect	SOMAN	\$17.75	ž	¥	ž	\$11.40	\$16.05	ž	≨	ž
UDPE FATH	U.D.P. 1976 1970	1	Local Channel - Dudleshed - 5/W/n VO Nov. But										
U.D.P. 549.46 1873. 18	U.D.P.	ш	2-wire VG par mile	1.5MC	90.03	90.00	80.03	90.03	8008	90 05	00 05		W og
UDPE 177 1865.4 1828 1834 1843	ULDRE SEAT	_1	Z-ertre VG Monthly Recurring per month.	29	114.61	22.53	\$13.91	82.23	817.8	\$17.83	\$14.83		\$19.00
UDPS FFF	UDPS FFFF FF		NRC: 2-46 VG - Act	3	8	27 27	8 20 5	1586.15 me :	67.79	28.2	3653.80		11.69
COUNTY STATE STA	COLUNY STATE RAS RA	Ш	MRC - 2-Wire VG - Decormed Org - 1st	1200	19 11	16.79	ž	Z ×		3 62	2 7		2
SOWANI NA RESIS NA	SOWM NA N	┙	MPC - 2-Wire VG - Decornect Chg - Addi	200	\$7.63	26.52	ž	¥	98.65	58.85	ΚĀ		28
SOMEC 150 253 150 </td <td>SOME 150 253 150<th>L</th><td>NRC - Manual Sec Order, per 15th demonstra</td><td></td><td>≨ :</td><td>81.55</td><td>ž</td><td>818.88</td><td>≨:</td><td>≨</td><td>¥</td><td>≱</td><td>86 66 66 66 66 66 66 66 66 66 66 66 66 6</td></td>	SOME 150 253 150 <th>L</th> <td>NRC - Manual Sec Order, per 15th demonstra</td> <td></td> <td>≨ :</td> <td>81.55</td> <td>ž</td> <td>818.88</td> <td>≨:</td> <td>≨</td> <td>¥</td> <td>≱</td> <td>86 66 66 66 66 66 66 66 66 66 66 66 66 6</td>	L	NRC - Manual Sec Order, per 15th demonstra		≨ :	81.55	ž	818.88	≨:	≨	¥	≱	86 66 66 66 66 66 66 66 66 66 66 66 66 6
SOUNCE W. St. 2 W. W. W. W. W. W. W.	SOURCE W. \$0.22 W. W. W. W. W. W. W.		NRC - Bechant Svc Order, per LSR		250	2 2 2 2	V 13	2 2	¥ 5	YY C	AN S	¥	¥
SOWAY 87.37 NA 518.94 \$18.14 \$18.14 \$18.26 \$17.26	SOWM \$17.77 NA \$18.84 \$16.44 \$16.14 \$15.76 \$15.26 \$15.35 \$15.76 \$15.35 \$15.36	Ш	NRC - Beckonic Svc Order, per LSR decorned	SOMEC	≨	27.03	ž	ž	ž	2 2	2 ≥	2 2	2 2
TEME REGION REG	COMM \$17.75 NA	Ш	NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - 1st NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - Add1	NAMOS SORMA	16.13	≨≨	18.5	4 8	\$18.14	88.83	11217	25.73	≨ 5
15MC 1600 1800	SOMM 817.75 NA NA NA 811.40 \$16.00 NA NA NA NA NA NA NA NA \$11.40 \$10.0								3	5,	916./9	85.5	Ę
LBMC REGO	LBMC REGG	1	MAC - 2-WITS VG - MOTEMBRIES CHENGE - MENLES Dic Order-Discounsed	SOM	K E	ž	¥	ž	\$11.40	816.05	ž	≨	ž
LUDN4 REGA	UDV4 REG	П	1										
ULDV4 \$15.77 \$50.50 \$14.50 \$25.30 \$15.57 \$15.65 \$15.65 \$1	ULDV4 \$15.77 \$50.50 \$14.99 \$25.30 \$15.77 \$15.57 </td <th></th> <td>Ιì</td> <td>1 SMC</td> <td>90 00</td> <td>80.00</td> <td>\$0.00</td> <td>\$0.00</td> <td>00.03</td> <td>80.03</td> <td>88.00</td> <td>80.03</td> <td>80</td>		Ιì	1 SMC	90 00	80.00	\$0.00	\$0.00	00.03	80.03	88.00	80.03	80
UDDY FREE MAT MA	ULDV4 \$55.00 W. W. \$55.00 W. W. W. \$55.00 W. W. W. \$55.00 W. W. W. \$55.00 W. W. W. W. \$55.00 W. W. W. W. W. W. W.	1		Ž	113.77	95 92	8	0C (22)	\$16.21	819.03	\$15.87		\$ 0.00 \$20 \$20 \$20 \$20 \$20 \$20 \$20 \$20 \$20 \$
UDN \$15.57 \$44.78 WA WA \$15.50 WA WA \$15.50 WA WA \$15.50 WA WA WA \$15.50 WA WA WA WA WA WA WA W	UDN \$18.57 \$40.78 \$17.5 \$15.50 \$17.5 \$15.50 \$17.5 \$1	1	- [2 2 2 2 2	22/20	1 1	25.00 20.15	252.75	28.2	22.23	Т	2 2 2 2 3
COUNTY 64.33 56.73 14.4 15.75 55.94 14.4 14.4 14.4 15.5 14.4 14.4 15.5 1	COUNTY 64.33 56.73 14.4 15.74 55.94 14.4 1	1]	1	NOV4		1 13	¥	2	36.35	\$78.56	à ≱	т	2 2
SOME NA ST ST ST ST ST ST ST S	SOLWIN IN STATE NA	1	1 1	Ž Ž	25	25.73	ž	¥	87.28	23.PA	ž	П	18.51
SOME READ <th< td=""><td>SOMEC NY EXAM REST</td><th>1</th><td>Ŧ</td><td>NAME OF THE PERSON</td><td>2</td><td>2</td><td>Į.</td><td>\$19.99</td><td>≨:</td><td>¥</td><td>≨:</td><td>П</td><td>20.0</td></th<>	SOMEC NY EXAM REST	1	Ŧ	NAME OF THE PERSON	2	2	Į.	\$19.99	≨:	¥	≨:	П	20.0
SOURCE \$15.00 \$10.00 \$11.00 \$11.20 \$17.70 \$	SOURCE \$15.00 \$10.00 \$11.00 \$	1	Т	2	§ ≨	2 22	200	25.53	200	12.13	2 5	Т	₹ 5
SOMMN SERTATION SE	SOWMY \$17.37 NA \$16.94 \$17.14 \$16.75 \$17.25 NA \$17.25 NA NA \$17.25 NA NA \$17.25 NA NA \$17.25 NA NA \$17.25 NA NA \$17.25 NA NA \$17.25 NA NA \$17.25 NA NA \$17.25 NA NA \$17.25 NA NA \$17.25 NA			SOLEC	\$3.50	\$0.42	¥	¥	¥	¥	¥	Т	≨
LBMC REGGS	159C 800 810 800 800 800 8173 817.8 MA MA 811.40 817.25 MA MA 811.40 817.25 MA MA MA 811.40 817.25 MA MA MA MA MA MA MA M	1		NAMOS	11.12	ž	Z (2	¥ 8	\$18.14	25.52	42.13	П	ž
1154C 8100	1154C 8100	ı -	т-						3			R	٤
LISK 86.00 85.00	U.DF E.S.	T	MFC - 4-WTB VG - Incremental Charge - Manual Svc Order-Disconnect	¥	27.72	ž	\$	ž	\$11.40	\$17.25	ş	≨	ş
UDF RES. REG. R	U.DFT EAST	1		T									
U.DFT 853.4 853.5 853.6 853.6 854.6 855.6 855.7 852.6 853.7 852.6 853.7 852.6 853.7 852.6 853.7 852.6 853.7 852.6 853.7 852.6 853.7 852.6 853.7 852.6 853.7 852.6 853.7 852.7 853.	U.DFT 855.42 855.15 855.15 854.56 544.56 545.05 855.65 855.15 85	ıT	per mile	1LENC	90.00	80.00	20.03	80.03	80.00	_	80.08		800
	U.DFT 142.24 122.44 122.45 142.44 142.44 142.44 12	Т	per morth	190	75 973	25.53	\$5.0C3	143.80	143.60	_	89 S2		240.27
U.D. 14.2 14.2 11.2 11.2 14.6 14.6 14.7 14.4 14.6	UDF1 14128 1413 1723 NA 12415 14466 NA NA NA NA NA NA NA	T	Т		988	\$ 7 K	996	8 3	35 65 CS	_	2 K.34		277.36
U.D.F. £2.18 £21.28 H.A. £13.3 £3.32 H.A.	COUNT R.2.16 SEL28 S15.14 NA SEL31 EAST EAST NA NA NA NA NA NA NA N	П	1		K 313	2 3	1521	¥	24.15	-	2 ≥		1100
SOMMY NA ST 56 NA ST 59 NA NA NA NA NA NA NA N	SOMMY NA ST 56 NA ST 57 NA NA NA NA NA NA NA N	T	П		92	28.28	1119.14	≨	1512	_	ž		223
SOURC 13.50 13.70 13.5	SOMEC 8150 R73 8150 8150 8150 8150 8150 8150 8150 8150	Т		NAMO SOMAN	≨ ≨	2 2	≨ ≨	5	2 3	-	≨ ≱		86.7
SOME MA MA MA MA MA MA MA	SOMM SOMM SOLAH	П	П	Source	65.53	2.3	23.50	23.50	23.50	-	8130	8.63	13.50
SOMAN SOLICE No. 1942 SOLICE	Column Solution Column Solution Column Solution Column Solution Column Co	Т	Н	SOVEC	ž	27 Q	≱ :	≨	ž	•	ź	ž	ž
U.D.S. State Sta	COUNTY EST N.Y. STORE EST COUNTY EST COUNTY EST EST COUNTY EST COUNTY EST	Т	1		2 2	≨ 3	3	7 19	7.00	-	S .	86.	≨ :
11-90C 54.4 56.16 56.20 55.05.4 NA NA 512.06 U.O.F.3 55.08 55.05.4 S.05.3 S.	1154C 1914 1916 1852 15400 15054 104 105 1540 1550	П	1	SOUNN	12.62	≨	8.0	≨	\$19.44	-	¥	2 2	\$ \$
	11.54C 19.4 19.16 18.52 15.40 15.03 14.4 15.10 15.15	П					-			-			
UDP3 8558 8657 1515 1 655.0 5650 1 559.0 1 659	U.DP3 \$5.58 \$2 \$4.54 \$7 \$51.51 \$4.58.50 \$4.56.50 \$1.50.70 \$4.59.50 \$4.5	T		115WC	17 15	90.00	→-	88	75.053	2			27. 17.0
U.D.** 140.5 140.5 11.6 13.0 14.0 15.0	ULD ² SeA S. 1902 JT SEA S. 11.001 STAN 14 S.SE ST 1822 STAN 14 SEA S. 1833 SEA S. 1833 SEA S. 1833 SEA S. 1833 SEA S. 1833 SEA S. 1833 SEA S. 1833 SEA S. 1834	ı		ubri	35.38 82	1288	-	SK.35.08	10 6998	07 (07.31)	7		82 (09)
UDP3 1428 1528 CS 1428 1 1428 1428 1 1428	UDF3 14.8.78 15.24 15.	1		690	75 OF 95	£ 1063	-	1001	\$709.14	12.867	\Box		37.26.16
UDS 518.15 518.	UD53 5112.14 512.17 514 515.05 5112.11 514 515.05 5112.11 514 515.05 5112.11 514 515.05 5112.11 514 515.05 515.			S Z	22	8	-	22.59	3	282	_		3
SOMUN NA ELISE NA 81999 NA NA NA NA NA NA NA NA NA NA NA NA NA	SOME HA ELE HA BISS HA HA HA HA HA HA HA HA HA HA HA HA HA	Т			77 711	2 2	-	≨ ≥	\$1.00 16 \$4.00 16	2 3	Т		8 8
SOMEN NA 1354 NA NA NA NA NA NA NA NA NA NA NA NA NA	SOMEC NA 534 NA NA NA NA NA NA NA NA NA NA NA NA NA	1		1310	ž	87.153	1	819.99	₹	¥	T		66 61
SOMEC \$3.90 \$2.75 \$3.50 \$3.50 \$3.50 \$3.50 \$3.50	SOMEC RISO RISO RISO RISO RISO RISO RISO RISO	П		NAMO	₹	19 63	ş	ž	¥	KX	Γ		ž
	SCAREC NA 18042 NA NA NA NA NA NA	Т		ONEC	9 12 12	27.5	05.63	83.53	82 83	83.53	П		25.23

ПП	П		11		
2 2 2 2	1 1	15.53.11 15.63.57 15.63.57 15.63.59 15.63.59 15.59 15.50 15.	\$ \$	15.25	NA OS
¥3.3 2	3	25.25.25 25.25.25.25 25.25.25.25 25.25 25.25	5 5	100 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	¥¥ ¥ €
¥ 55 55 55 55 55 55 55 55	3	X X X X X X X X X X X X X X X X X X X	5 5	25.25.25.25.25.25.25.25.25.25.25.25.25.2	¥¥8
2 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	\$5. 35.	153.39 153.39 15.004.33 15.02.13 15.22.11 15.00.87 10.83 10.	x x	18.8.7.0 15.4.1	¥ ¥ 65
₹055 \$055 \$055 \$055 \$055 \$055 \$055 \$055	¥.02	18.77 18.58.00 18.58.00 11.13.75 11.13.75 11.13.75 11.10.00 11.10.	\$16.77		¥ ¥ 53
\$50.12 \$90.12 NA	ž	EA-00 11.097 11.097 14.097 11.097 11.097 11.097 11.097 11.097 11.097 11.097 11.097 11.097 11.097 11.097 11.097 11.097	\$ \$	15.43	819 88 KA KA KA KA
9 23 23 23 755 28 25 25	81800	16 RP 16.75 Se 16.75 Se 16.75 Se 17.75	\$1800		₹\$ ³³
4333	NA.	25.65.45 25.65.45 25.65.65 25.65.65 25.75.45 27.55 27.	\$ \$		
20 45 45 45 45 45 45 45 45 45 45 45 45 45	\$19.03	115NC 844 UDFS 85340 UDFS 811817 UDFS 811817 UDFS 811817 UDFS 811817 SOWN NA SOWN NA SOWN SOWN NA SOWN SOWN	\$19.03	10 10 10 10 10 10 10 10	₹ ₹
SOMWN SOMWN SOMWN SOMWN	SOMAN	LUDES ULDES ULDES SOUMN SOUMN SOUMN SOUMN SOUMN	SOMAN	TIBRO COMMA	NAME OF THE PROPERTY OF THE PR
NRC DS3 incremental Charpe-Menual Svc Order - 1st NRC DS3 - Incremental Charpe-Menual Svc Order - Add NRC DS3 - Incremental Charpe-Menual Svc Order - Add NRC DS3 - Incremental Charpe-Menual Svc Order Decornect - 1st		Lead Channes Designated - \$11.51 S16 Facility Termination per month S16 Facility Termination (it is NRC - 515.51 - Facility Termination (it is NRC - 515.51 - Facility Termination - Add NRC - 515.51 - Facility Termination - Discornect - Add NRC - 515.51 - Facility Termination - Discornect - Add NRC - Excitotic Size Order, per LSR decornect - NRC - Excitotic Size Order, per LSR decornect - NRC - Excitotic Size Order, per LSR decornect - NRC - Size - Add NRC - 515.51 - Accornected - Add NRC - 515.51 - According - Add NRC - 515.	Zerge-Meruel Svo Order-Decornect -1se Zerge-Meruel Svo Order-Decornect-Addit	COST Facility Control of the Cost of the C	$\Pi\Pi$
				┞╏╘╘┋┪┪╏┊╏╘╘┪┝╃┩╸┈╏╒╬╃╏╃╏┩╏┩╇╇╇╇	╽

F ZZZZZ	35252	┼┈╏┈╏┈╏┈╏┈╏┈╏ ┈	23 200173 25 39 20 20 20 20 20 20 20 20 20 20 20 20 20	### ### ### ### ### ### ### ### ### ##
2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	M	* ***********************************		
MA NA NA NA NA NA NA NA NA NA NA NA NA NA	362		23 180 28 28 180 28 28 28 28 28 28 28 28 28 28 28 28 28	2000 00 00 00 00 00 00 00 00 00 00 00 00
HA NA NA NA NA NA NA NA NA NA NA NA NA NA	8983 9683 8983 9683 8983 9683 8983 9683 8983 9683 8983 9683 8983 9683 8983 9683	╏╏┩┪╏╏╏╏	2000 20 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2000 H H H H H H H H H H H H H H H H H H
NA S20	╘╼╏╋╏╏╏┩┩╏			2000年 2000年
MA NA NA NA NA NA NA NA NA NA NA NA NA NA		NA NA NA NA NA NA NA NA NA NA NA NA NA N	100000 100 100 100 100 100 100 100 100	100 CO CO CO CO CO CO CO CO CO CO CO CO CO
7 X X X X X	╼╂╂╂╂╂╂╂╂	2	80008 800008 80008 80008 80008 80008 80008 80008 80008 80008 80008 80008 8000000	100000 1000000 100000 100000 100000 100000 100000 100000 1000000 100000 100000 100000 100000 100000 100000 1000000 100000 100000 100000 100000 100000 100000 1000000 100000 100000 100000 100000 100000 100000 1000000 100000 100000 100000 100000 100000 100000 1000000 100000 100000 100000 100000 100000 100000 1000000 100000 100000 100000 100000 100000 100000 1000000 100000 100000 100000 100000 100000 100000 1000000 100000 100000 100000 100000 100000 100000 1000000 100000 100000 100000 100000 100000 100000 1000000 100000 100000 100000 100000 100000 100000 1000000 100000 100000 100000 100000 100000 100000 1000000 100000 100000 100000 100000 100000 100000 1000000 100000 100000 100000 100000 100000 100000 1000000 100000 100000 100000 100000 100000 100000 1000000 100000 100000 100000 100000 100000 100000 1000000 100000 100000 100000 100000 100000 100000 1000000 100000 100000 100000 100000 100000 100000 1000000 100000 100000 100000 1000000 1000000 1000000 1000000 1000000 1000000 1000000 1000000 1000000 10000000 1000000 10000000 10000000 100000000
MA MA MA MA MA MA MA MA MA MA MA MA MA M		8.5 88.2 8.5 8.5 8.5 8.5 8.5 8.5 8.5 8.5 8.5 8.5	20,000 210,140 210,	200068
USOC SOMAN SOMAN SOMAN SOMAN SOMAN	SOMAN SOMAN	SOULT SOULT	LESS UITRE UITRE UITRE UITRE SOULM S	MA SECTION AND SEC
MRC - Bectronic Svc Order, per USH decorned MRC - DCAR - Incremental Charge - Marcal Svc Order - 1st MRC - DCAR - Incremental Charge - Marcal Svc Order - 1st MRC - DCAR - Institute - Charge - Marcal Svc Order - Marcal - Marcal Svc Order - Marcal Svc Order - Marcal - Marcal - Mar	NRC - CC46 - Incremental Charge - Martual She Order - Discornaces Addition - CC48 - Talefricas - Incremental Cost - Martual She Chiefer vs. NRC - CC48 - Talefricas - Incremental Cost - Martual She Chiefer vs. NRC - CC48 - Talefricas - Incremental Cost - Martual She Chiefer vs. UMSUMOLE DESERVATED TRANSPORTE - InferroPhica Charmed Theoretics - Transport - Cardeding - 2, -drie V/d 2-Wire V/G - Facility - Termination per month NRC - 2-wire V/G - Facility - Termination per month NRC - 2-wire V/G - Facility - Termination - Record - Cost - C	ve Order o	2-Wes VG - Fer hale per morth 2-Wes VG - Fer hale per morth 2-Wes VG - Feath V eministen - lef WC - Z-wes VG - Feath V eministen - lef WC - Z-wes VG - Feath V eministen - Discorned Charge - Lef WC - Z-wes VG - Feath V eministen - Discorned Charge - Lef WC - Level VG - Feath V eministen - Discorned Charge - Lef WC - Level VG - Feath V eministen - Discorned Charge - Lef WC - Z-wes VG - Feath V eministen - Discorned Charge - Lef WC - Z-wes VG - Feath V eministen Charge - Manual Sec Order - Lef WC - Z-wes VG - Feath W eministen Charge - Manual Sec Order - Discorned - Lef WC - Z-wes VG - Feath W eministen Charge - Manual Sec Order - Discorned - Lef WC - Z-wes VG - Feath W eministen Charge - Manual Sec Order - Discorned - Lef WC - Z-wes VG - Feath W eministen Charge - Manual Sec Order - Discorned - Lef WC - Z-wes VG - Feath W eministen Charge - Manual Sec Order - Discorned - Lef WC - Z-wes VG - Feath W eministen Charge - Manual Sec Order - Discorned - Lef WC - Z-wes VG - Feath W eministen Charge - Manual Sec Order - Discorned - Lef WC - Z-wes VG - Feath W eministen Charge - Manual Sec Order - Discorned - Lef WC - Z-wes VG - Feath W eministen W eministe	Current State of Lineage Leave VG

25 25 25 25 25 25 25 25 25 25 25 25 25 2	10.17	20.55 20.55	8 23 05 05 05 05 05 05 05 05 05 05 05 05 05	#####################################
25 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	89 6073 81.847 81.847 81.847 81.847 81.847 81.84		▗ ▐▆▐▆▐▆▐▆▐▆▐▆▊▊	╉╒╃╒╏┈┇╌┇╌┩╌┩╌┋╌┧╌┩╌┩╌┩╌┪╺╇ ╍╂╌╁┈╏
			┍┩╋┋╇┋┋ ╋╫	2.000
N 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	817.74 8137.74 8137.74 822.54 NA NA NA NA NA NA NA NA NA NA NA S38.07 839.07 800.07 80	10.078 17.178 18.17.17 18.17.17 18.178 18.178 18.50 18	25.2 26.2 26.2 26.2 26.2 26.2 26.2 26.2	20024 20024
27.23 NA NA NA NA NA NA NA NA NA NA NA NA NA N	## MA WAS SEED AND	87.440 81.440 81.153 81.655 821.61 83.50 83 83.50 83.50 83.50 83.50 83.50 83.50 83.50 83.50 83.50 83.50 83.5	115.02 1464.34 1477.74 1125.56 113.56 114.77 115.56 115.50	113.4 2626.5 2626.5 2626.5 2526.5 211.7 31.6 31.5 31.5 31.5 31.5 31.5 31.5 31.5 31.5
2 X X X X X X X X X X X X X X X X X X X	20.004 276.20 27	80 760 816 49 816 89 816 34 816 34 NA NA NA NA NA NA NA NA NA NA NA NA NA	11.04 151.101 150.101 150.10 150.10 150.10 150.10 150.23 150.23 150.23 150.24 150.24 150.24 150.24	101.18 101.14 101.14 101.14 101.14 101.16 10
1 1 1 1 1 1 1 1 1 1	10.0001 10.000	80.4500 82.05 82.05 82.05 83.10 83.10 84.4 81.50 84.4 81.50 84.4 84.50 84.4 84.4 84.4 84.4 84.4 84.4 84.4 84.	112.05 1844.25 1844.25 1816.05 1816.05 1819.06	11.204 11.204 13.404 73 13.404 73 13.404 74 13.404 74 13.404 74 14
4 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	25.00 27.00	80.4523 878.47 811.70 811.70 N.X N.X N.X N.X 815.90 816.94 816.94 816.94 816.94 816.94 816.94 816.94 816.94 816.94 816.94 816.94	82.75 57.86 00 53.00 70 53.00 70 51.22 31 51.22 31 51.22 31 51.22 31 51.22 31 51.32 40 51.33 40 51.33 41 51.34 51 51.34	27.2 27.2 27.2 27.2 27.2 27.2 27.2 27.3 27.3
R X X X X X X X X X X X X X X X X X X X	2000年 1000	80.83 188.73 188.73 18.73 18.73 18.73 18.73 18.73 18.73 18.73 18.73 18.73 18.73	11.11.18 20.17.18 20.17.18 20.18.18 20.	11.10.00 10.00 10.10.00 10.00 10.10.00 10.10.00 10.10.00 10.10.00 10.10.00 10.10.00 10.10.00 10.10.00 10.10.00 10.10.00 10.10.00 10.00 10.00 10.00 10.00 10.00 10.00 10.00 10.00 10.00 10.00 10.00 10.00 10.00 10.00 10.00
E 288 RAA RAA RASS RASS RASS RESS 812.97	11570 46 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	57.5 (5) (5) (5) (5) (5) (5) (5) (5) (5) (5	MA NA NA NA NA NA NA NA NA NA NA NA NA NA	K K K K K K K K K K K K K K K K K K K
LIBOC UITDS SOMAN SOMAN SOMAN SOMAN SOMAN SOMAN SOMAN SOMAN	HESON UTITAS UTITAS UTITAS UTITAS SOURCE SOU	ILEXX INTEN INTEN OULAN OULAN OULAN OULAN OULAN OULAN	115XX 41.98 U1755 8695.15 U1775 8695.15 U1775 8767.75 U1775 816.75 U17	15.00 HWW.022 HW.02 HW.02 HW.02 HW.02 HW.02 HW.02 HW.02 HW.02 HW.02 HW.02 HW.02 HW.02 HW.02 HW.02 HW.0
	Wilder Control of the Control of the Control of Control	1551 per mis pai ricorth 1551 - Der mis pai ricorth 1551 - Der Lichtig I dermission 1 at 1552 - M. C. Lichtig I dermission 1 Add 1553 - M. C. Lichtig I dermission 1 Add 1553 - M. C. Lichtig I dermission 1 Add 1554 - M. C. Lichtig I dermission 1 Add 1555 - M. C. Manual Sec Order, por LSR decorned 1555 - M. C. Lichtig I dermission 2 Add 1555 - M. C. Lichtig I dermission 2 Add 1555 - M. C. Lichtig I dermission 2 Add 1555 - M. C. Lichtig I dermission 2 Add 1555 - M. C. Lichtig I dermission 2 Add 1555 - M. C. Lichtig I dermission 2 Add 1555 - M. C. Lichtig I dermission 2 Add 1555 - M. C. M.	Efficient Chaps - 18 connect Chaps - 18 connect Chaps - 18 decornect und Sto Chap - 18 und Sto Chie - 1801 und Sto Chie - 1801 und Sto Chie - 1800 nuil Sto Chie - 1800	Historifies Tensiport Disclosinid - \$T\$. 1515.1 Anolity Stringland ST\$. 1515.1 Anolity Stringland ST\$. 1515.1 Anolity Stringland ST\$. 1515.1 Anolity Stringland ST\$. 1515.1 ST\$. Stringly Stringland ST\$. 1515.1 ST\$. Stringly Stringland ST\$. 1515.1 ST\$. Stringly Stringland ST\$. 1515.1 ST\$. ST\$. ST\$. 1515.1 ST\$. ST\$. ST\$. 1515.1 ST\$. ST\$. ST\$. ST\$. 1515.1 ST\$. ST\$. ST\$. ST\$. 1515.1 ST\$. ST\$. ST\$. ST\$. 1515.1 ST\$. ST\$. ST\$. ST\$. 1515.1 ST\$. ST\$. ST\$. ST\$. ST\$. 1515.1 ST\$. ST\$. ST\$. ST\$. ST\$. 1515.1 ST\$.

TRANCINCHI

113 12 151 12 151 12 151 12 151 13	#19.80 #1.178 #1.178 #1.178 #19.90 #19.90 #19.90 #19.90 #19.90 #19.90	¥ ¥	2011年 201
2012年 2012	15.25 14.13 14.1	\$ \$	26 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2
814.10 12.071 12.071 15.1381 15.1381 17.185 17.185 17.185 18.50 18	H48.8 17.676 12.081 12.08 02 11.1166 11.2166 1	E E	11.08.08.08.08.08.08.08.08.08.08.08.08.08.
818 35 31 852 00 81 81 81 81 81 82 82 83 81 82 83 81 81 81 81 81 81 81 81 81 81 81 81 81	2002 21.187 21.087 21.083 21.0	85 85 85 85 85 85 85 85 85 85 85 85 85 8	11.00
2018 2018 2018 2018 2018 2018 2018 2018	11.13.1 10.14.7 11.13.1 11.13.	# # 00 20 20 20 20 20 20 20 20 20 20 20 20	11.24.59 11.14.50 11.
E 1399 11,399 11	12.14 12.14 12.14 12.15 12.15 13.15	£ £	11.10.00 11.1
14.77 14.73 14.17.36 14.17.36 14.73.31 14.14.14 14.14 14.14 14.14 14.14 14.14 14.14 14.14 14.14 14.14 14.14 14.14 14.14	115.06 11.06.00 11.06.00 10.17.39 11.19.14 11.19.14 11.19.14 11.19.14 11.19.14 11.19.14 11.19.14 11.19.14	818 88	### 11/2/20 ##################################
E 23.74 20.05.05 20.05.05 20.05.05 20.05.75 20.05 20.05 20.05.75 20.05.75 20.05.75	AN AN AN AN AN AN AN AN AN AN AN AN AN A	2 2	11.00
12 12 12 12 12 12 12 12 12 12 12 12 12 1	19.78 19.78	B 88	100
NAMES OF THE PROPERTY OF THE P	I EXX OLUMN	SOMAN	X
OCS - per mide per month. OCS - per mide per month. NEC - OCS - Facility 1 campusion - 184 NEC - OCS - Facility 1 campusion - 184 NEC - OCS - Facility 1 campusion - 184 NEC - OCS - Facility 1 campusion - 184 NEC - OCS - Facility 1 campusion - 184 NEC - OCS - Facility 1 campusion - 184 NEC - Secret Exp 1 Campusion - 184 NEC - Exception - 184 NEC - OCS - Facility 1 campusion - 184 NEC - OCS - Facility 1 cam	200 Sept. 200 Se	HEC. OC12 - Incremental Cost - Manual See Order VI. Elect-Disconnect	en en bottomas - OCOS Finally Termination of the month o

TRACAC

		3081	4	-	5	3	:	3			
T	Channel System - Incremental Cost - Manuel Suc. Order vs. Electronic - set	SOMAN	\$15.61	≨	\$14.75	3	\$10 74	8	28 13	2 X	F 16
Т	Channel System - Incremental Cost - Manual Syc. Order vs. Electronic - Admi	SOMM	57.38	ž	\$6.55	\$11.99	28.77	28	\$13.33	26 83	\$10.46
T	Incremental Test Manual DVC. Critical PC. Critical PC.	N S	\$11.67	ž	\$10.60	NA	\$12.43	\$16.97	\$18.26	¥	\$14.21
Т	CONTRACTOR OF COURT AND CONTRAC	SOMAN	200	≨	ž	≨	¥	≨	\$1.48	¥	\$1.46
Г	DS1 Chennelization (D81 to D90)										
П	per Chamaited System (24 DSJ) per month	ğ	3136.82	61515	13.25	200.00	62.00	\$8.77.78	13.00.7		
П	ATC-18	Q	8197 98	11 0115	104 25	51	1000	2371 63	27,772	\$14/51	1283
T	NRC - Add	ğ	\$1.22.12	\$125.18	\$123.59	\$184.20	\$118.37	20.75	161 41		13/21
Т	MRC - Ist - Diecomed	Q	\$30.18	\$1952	531 dd	¥	17.84	12.83		, A	27.40
Т	NHC Add1 - Decorned	ğ	81E B6	\$18.14	\$19.75	¥	\$16.83	\$11.98	71.153	¥	\$15.61
Т	per OOU-OP(dea) card per mornin (2.4-Edition)	AN PA	22	*	97.0	7 7 7	11.4				
г	APC-18	3 5	8 25	2	80.18	X I	53.12	85 BB	88 23	X C	3 4
П	NRC - Add	0010	2 5	20.21	27.50	\$15.00	\$12.29	200	\$15.76	\$12.08	\$12.61
П	per BRITE card per month	Š	170		3 5	2	20.00	2	21.5	33	36
П	MACHE	S S	\$12.08	\$13.16	412.00	315 86	613.20	2 2 2	20,70	210.00	
7	MRC - Add	CICA	28.69	28 43	33.33	\$11.36	08.83	\$11.35	1	77.5	24 64
7	per VG cand per month (DSO)	Dividi	90 65 08	\$1.45	\$1.17	81.40	\$1.62	\$1.48	25.52		31.25
+	F1. 124	٥ ٧	\$12.05	\$13.16	\$12.02	\$15.86	\$12.29	\$15.66	\$15.76	\$12.08	\$12.61
1	MC - Maria Sec Date no. 150	DA KO	2	22.00	95 83 83 83 83 83 83 83 83 83 83 83 83 83	\$11.36	26.80	\$11,35	\$11.26	\$9.68	\$9.03
+-	MIC: Manual Sec Order on 1 St Announced		ž	8	ž	819 99	¥	ž	¥	ž	\$19.90
т	MRC - Electronic Suc Darles nee 1 88	2000	¥	2	ź.	ž	ž	٧Z	¥	¥	ž
1	MPC - Electronic Syc Order Cer LSR decorated		2	X	3	8	82	82	95 22	8	99.53
П	Channel System - Incremental Cost - Manual Svc. Order vs. Electronic - Ist			2 12	2	ž.	¥2.	Ž	ž	≱	ž
	Channel System - Incremental Cost - Manuel Svc. Order vs. Electronic - Acot	NAMO S	27.39	1	27.23	8	443	8 2	2	S S	99 52
7	Incremental Cost-Manual Svc. Order vs. Elect - Discompct - 1st	SOMM	19115	≱	\$10.70	ž	\$17.43	£16.07	3 8		10.00
+	Incremental Cost Manual Svo. Order vs. Elect -Discornect - Addit	SOMAN	6946.08	ž	\$0.00	¥	₹	ž	27.5	ź	97
+	The bas to have take										
+	Carl Flux http://pe.flox.flox.flox.flox.flox.flox.flox.flox										
	MONTH THE STREET STREET, THE STREET STREET STREET STREET, DATE STREET,	Š	8	3	3	1	1		-		
+	NRC - Per each lous-liber dark liber amendement - 1st	_	7	61 776 63	W	-	200	200	20.00	2 2	8
Н	NHC - Per each lour-liber dark fiber arrangement - Add1		880.53	27.00	202	# 1.57.5	200	7	B 67.03	8///8	277500
Н	ARC-Decornect-1st	-	≨	13 (95)	ž	-	2	177	1	2 12	3
+	Mrc-DecomactAdd	UDF14	ž	15 3953	ž	≨	ş	Ş	×	2 2	1
+	Date Flore Corn Change Share described to the control of the con	***									
+-	NFC - Per each lour-ther dark the arrenament - Let			20.00	23.5	149.07			17 998	10013	90'09'
Н	NHC - Per each lour-floar dark floar amangement - Add1	2	86.03	_			21./40.UC	37.741.00	882	21,747.00	21,72,8
Н		2	₹	15.67	¥	т	2	т-		2 2	N 17
+	NHC-Depomed-Add	10401	≱	1,5953	ž	₹	₹	¥Χ	ž	¥	1
+-	Dent Floer - Local Loop (four floer strands) per route ente or transfor shared	197	- BW3	3	11.13	67.6 A.S	100	44.788	2.56		
Н	MHC - Per each lour-floer dark fiber arrangement - tal		80	_	_	-	_		_	L	9000
Н	MRC - Per each four-fiber dark fiber arrangement - Add)	UDALA \$563.00	5563.00	٠.		\$563.75	•	86.78	900		27.77
+		100	ž	ш	-	-	¥		¥	Т	¥
+		ZEOS	≨	2002	¥	ž	ž	₹	¥	≨	\$
+-		1	1								
+	WOTES:	T	1	\dagger							
н	Interim rates audient to true-up.	 	1	T	1					1	T
-	Where the state Commission has adopted rates for the rate elements contained		İ							1	
	hershi, it is the intent of the parties to reflect such raise in this exhibit and to applicable FCC and Commission rules and										
4	COORT			_			_				_

DESCRIPTION	OSOC	4	10	40	A.37		947	5		
UNBUNDLED LOOP COMBINATIONS NAME SEED SHAFFLETS	3									
Unbinded Londbox Combinedore Mose A CO.										
MARKET RATES: MACTIDORS ALL VERTICAL SEATTIBLES NAME	-									
	1							Greensboro- Winston Saler		
	NA ST		Ortanda, Ft.					Highpoint/ Charlotte-		
				Attents		New Orleans		Gestonie-Roci		Mechanic
Currently Combined (Noted)										
2-Wire Voice Grade Loop with 2-Wire Line Port (Res. and Bus.)										T
2-With William Life Cold (The), per month.	4-	17	877	2	,			!		
2-whs voice unbundled port with caller ID - residence	UEPRC	₹	8	8	Ž	214.00	≨	8 1 8	ž	88
2-wire valoe unbunded part outpoing anly - residence	Н	ž	814.00	\$14.00	¥	\$14.00	¥	\$1100	ž	807
A HYPER YOCKS GENOW UNDURCHING ARTHURISH BETHINDED TOOM GLARING DARRY DON WITH CARRY.	UEPAR	¥	ž	. ≨	¥	ĄŅ	Ž	V.	42	1
2-wire voice grade unbundled Kentucky extended local dialing party port with called	Medali	1	47	ă	1	3	1	1		
2-who voice grade unbundled Louisians entended local dailing party port with							Ę	٤	٤	Š.
2-wire voice grade unbundled Messesippi extended local dealing partly port with	24430	ž	≨	≨	≨	214.00	ş	ž	≨	¥Z.
2-ere vote grade unburded South Carolina extended from deline centre and with	UEPAT	¥	KA	ž	ž	¥	¥¥	≨	٧V	ž
	UEPAU	ş	\$	¥	ž	Ν	ž	ž	ž	WA
C-three young grade unduration i emissions determined total desirig partity port with	UEPAO	MA	¥	¥	¥	ž	ž	ž	V.	\$14.00
2-wife votce unturded Plotte gree calling with caller ID - residence	UEPA	≨ :	814.00	ž	≨ :	٧V	¥	¥	ş	ž
2-wire yolde unbundled Louisians Area Plus with caller 10 - residence (ACT)	UEPAH	Ş	Į.	ž	≨ <u>≨</u>	8 8	ž	ž	≨ ≨	Y Y
2-wine voice unbundled South Cardina Ana Calling pon with Carler ID - residence (1.W8)	(New Y	7	VA	7.7	11	2	***	;		
2-whe volce unbundled Termessee Area Calling port with Caller ID - neidence	7.00	1				5	£ :	E :	§ :	XX.
2-ere votes unbundled Termesee Area Cathry port with Cather ID - realdence	¥	E	ž	¥	٧×	ž	ž	X	ş	814 00
(TACETO 2-ade units introduct Terresons has California att California . residence	UEPAL	≨	ş	ž	ž	¥	ž	ž	ş	\$14.00
(TACSEN)	UEPAH	ž	¥	NA.	ž	ž	MA	Y.	ž	\$14.00
	UEPAN	¥	ş	NA	ž	ž	ž	ž	ž	814.80
2-wire volce unbundled Termesee Area Calling port with Caller ID - residence (2049)	UEPAO	ž	₹X	¥	44	44	4	42	:	8
A WALL VALUE LEADING THE DAY LEADS THE DOT WITH CARLY TO ALLIAM	UEPAP	ž	\$14.00	\$14.80	ź	\$14.80	ž	\$14.00	ź	8
2-wire voice unbundled port without Calver ID	UEPB.	¥	\$14.00	814.00	ž	871	2	8148	ş	\$14.00
2-vers votes unburded port with unburded port with Caller+E464 ID	UEPBC	≨ :	14 00	214.00	≨:	8	≨:	\$14.00	ž	8
2-wire voice grade unbundled Alabama artended local dusing parity port with calles	2	£ :	3		5 :	8	S	214.00	Ş	80 713
2-whe viola grade unbundled Kentucky entended toost daling party port with calles	1	5	٤	٤	ž	٤	ž	Ş	Ş	<
2-449 vides grade unbundled Louistana entended local duling party port with	18435	ž	≨	≨	ş	ž	≨	ž	\$	¥
Cutter IO	NEPAX	¥	\$	¥	¥	8148	¥	ž	≨	MA
Ol selec	UEPAY	ž	ş	٤	ş	ž	ž	ş	ž	¥
Z-ents vidos grada untoutoled south Ceroma ententiad focal dashing partly port with ceffer IQ.	UEPAZ	¥	¥	ş	¥	ž	¥	ş	ž	.
2-who vides grade unbundled Tennesses entended local desing parity port with caster ID.	VEPAV	¥W	44	4M	2	4	1	42	4	8
2-wire votce unburded incoming only port with Caller ID	ŒP91	¥	\$14.00	814.00	ź	81400	5 ≨	814.00	ž	88
2-wes votes unbunded LA Bus Area Califorg Port with Califor ID (BUC) 2-wes votes unbunded SC Bus Area Califor Info with Califor ID (LAB)	WEPAN	≨≨	≨≩	\$	\$ \$	8 2	3 3	≱ ≸	¥ X	3 3
Control of the Contro	J40301	3	3	4	1	3	;		1	8
	2	5	٤	٤	٤	٤	5	٤	2	ON BILL
2-wire volce unburdled TN Bus 2-Way Area Calling Port Standard Option (TACC2) 2-wire volce unburdled TN Bus 2-WAY Collegy de and Memorite Local Calling Puri	CEPAO	ž	ž	¥	ž	ž	ž	¥	ž	8376
(UZZ)	CEPAE	ž	ş	¥	Y.	¥	ž	ž	ž	81480
RC - 2-Wire Voice Grabe Loop - Steewide	UEPLX	ž	ž	MA	٧¥	٧×	ž	\$14.18	¥	ž

LDOP PORT COLIS

LIBOUTHMENINGUM RATES	METINORIX ELLEMENTS	AND OTHER BENVICES
=	-	•

믭	DESCRIPTION	2080	7	19	46	3	•	1	!		
±	PC-2-Wire Voice Grade Loop Zone 1	Ž	≨	\$14.90	\$10.60	ž	\$14.05	2 ≥	ž	2 2	Z (0 %)
1	RC - 2-Wire Votre Grade Long Zone 9	- 1	¥	\$18.51	\$12.47	¥	\$24.14	ž	ž	Y X	520.55
L	Combination Rates		≨	\$24.25	\$19.83	₹	00 645	ž	ž	¥×	\$1.12
Н	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Statewide	ł	Į,	12							
\pm	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 1 (Note 6)	1	≨	CS 823	20.00	2 2	¥ S	2	\$28.18	ž	¥
1	HC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 2 (Note 6)	ł	ž	15 0.3	20.00		3 :	٤	≨ :	ž	25 623
1	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3, Plose 6)	Note 8	≨	23.23	233.63	£ 2	2 5	2	2 3	≨:	82.428
1	Manneduring Charges								5	ž	2
Ł	AND THE POT (MR. AND BUR.)										
Ŧ	M.C. 2. was come of the contraction of the contraction of the contraction		ş	22.50	\$41.50	ž	\$41.80	ž	2 2 3	3	5
Ł	MC. 2. was write grade or the control provided control of the cont		ž	28	\$41.50	٧V	81.50	ž	21.50	Ž	3
£	MACCO Bette votes grade sets order conference productions.		≨	3	\$1.50	¥	\$41.50	ž	2 50	ž	25
L	Company of the Company		≨	8	25.50	ž	81.50	Ş	\$41.50	ş	\$41.50
Н	MMC - 2-Wine Volce Grade Loop/Line Port Combination - Subsequent		AM.	200							
Ē	NPC - 2-Wire Votos Grade Loop/Line Port Combination - OSS LSA Charge.		٤	Biologia	00.014	ž	\$10.00	ž	\$10.00	ž	\$10.00
_	Becruit, per LSR received from the CLEC by one of the OSS interactive										
E	IMPC - 2-Wire Voice Grade Local Ins Part Combination , parameter C.	SOMEC	≨	22	8 8 8	¥	8	≨	82 80	ş	\$3.50
\exists	Menual Svc. Order vt. Electronic - 1er	SOMAN	≨	25.58	# 43	3	\$	1			
_	MHC - 2-Wine Volce Grade Loop/Line Port Combination - Incremental Cost -					2	20.100	٤	20.18	Ş	530.03
Ŧ	Marrier Sycurptor VI. Electronic - April	BOMAN	≨	\$21.58	87.86	ž	\$7.32	ž	\$2.62	**	8
_	Notice - Superports Notice - Superports		;	i							3
E	HRC-2 Wire Voice Grade Loop/Live Port Combination - Scheen and Database		2	780	8	ž	1128	ž	21.43	¥	TBO
\exists	Update - Menuel Service Order		ž	52	2	2	\$	3		;	1
\exists	WHICH ENGRONIC Service Order Discornect		Ş	27.03	ş	ž	VN.	2	170	5	8
Ŧ	ANT Incremental Menual Service Order Osconnect		ž	23.84	\$20.00	¥	220.00	ž	800	§ <u>\$</u>	S C C C C C C C C C C C C C C C C C C C
_											
\perp											
\equiv	2-Whre Volce Grade Loop with 2-Whe Line Port PBX			_							
	2-Who Aming Line Port (PBX), per month.										
	2 WIFE VOICE UNBLADLED COMBINATION 2-WAY PBX TRUNK - PRESENCE		4	875	8	3		:			
1	LIME SIDE UNBLINDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS	UEPPC	≨	8148	87.8	2 2	214.00	2 3	8 5	≨ :	24.80
\pm	LINE SIDE UNBURGLED OUTWAND PEX TRUMK - BUSINESS		ž	\$14.00	\$14.00	≨	8148	ž	875	S S	200
E	2-WIFE VOICE UNBUNCED 2-WAY COMBRATION PRY ALABATA PATTER		≨	\$14.00	\$14.00	≨	\$14.00	ž	\$14.00	٧×	81.8
	PORT	2430	ş	≨	\$	2	Ą	3	\$;	:
_	2-WHE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING PORT	1							5	٤	Š.
	2-WHE VOKE UNBUNCLED PRIX LD TERMINAL PORTS	Z G	≨ ≥	¥.	≨ .	≨ :	8	ž	ž	ž	¥
Ш	2-WIFE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE		5	3	3	Ş	8	≨	8718	ž	\$14.00
E	2-WHE VOICE UNBUNCLED 1-WAY DUTGOING PEX TENNESSEE CALLING	2	ž	\$	≨	ž	ž	ş	٧¥	ş	814 00
#	PORT	OEPTO	į	ş	ź	\$	ž	2	2	1	
1	2-WHE VOICE UNBURLED 2-WAY COMBINATION PRY USAGE PORT	DEPTA	ş	\$14.00	8148	¥	\$11.00	ž	\$14.00	≨ ≩	87.5
Ħ	2-WIRE VOICE UNBLACKED PRIX LD DOD TEPARINALS PORT	CEPYC	\$ \$	8 8	88	≨ :	8	≨:	814 00	¥	\$14.00
	2-WIPE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD PORT	UEPXO	ş	875	871	\$ 3	38	2 3	8 5	ž	88
	2-WIFE VOICE LIMBURICLED PBX LD TERMINAL SWITCHEGARD IND CAPABLE PORT	1	;					1	3	٤	B
F	2-WIFE VOICE UNBUNCLED 5-WAY PBX KENTLICKY ROOM APEA CALLING	NET AR	٤	m *18	B	≨	8	ž	3148	ş	\$14.00
#	PORT WITHOUT LLID	UEPXE	Ş	≨:	\$	2	¥	¥	ž	¥	¥
	2-WIRE VOICE UNBUNDLED PRIX KENTUCKY PREMILIA CALLING PORT	DE PAR	2 3	\$ 3	≨ :	≨ :	ž	¥	¥	٧×	ΥY
	2-WIRE VOICE LINBUNDLED 2-WAY KENTLUCKY AREA CALLING PORT	2	5	5	٤	٤	ž	ž	ž	ž	\$
#	E-WIPE VOICE LARE AND ED 2-WAY PAY I OF REMAINS I YOUR	UEPX	ž	ž	\$	ş	≨	ž	¥.	NA	NA
#	CALING PORT	UEPXX	ž	¥	ş	¥	814.8	ş	3	¥	\$
_	ADMINISTRATIVE CALLING PORT	UEP/R	\$	8 3	\$14.00	ş	8118	\$	874.00	4	873
	C-WIFE VOICE UNBURDLED 2: WAY PBX HOTELHOSPITAL ECONOMY ROOM CALLING PORT	UEPXX	¥	814.00	875	2	8	3	8	:	
_	2-WIRE VOICE UNBLINGLED 1-WAY OUTGOING PRX HOTEL/HOSPITAL					1	3	£	8	£	814 00
1	EXCRAMENTATION OF COLUMN TWO I ENTENDED CALIFIES FURT	CEPYON	≨	¥.	ž	2	¥	¥	ž	ž	\$14.00

DESCRIPTION	2085	7	1	49	2		3	1		
2-WINE VOICE UNBUNOLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL DOACOUNT ROOM CALLING PORT	1					5		2	3	¥.
2-WIRE VOICE UNBUNDLED I WAY OUTGOING PRIX LOUISIANA LOCAL	R	≨	8	8	ž	814.00	≨	814.00	ž	\$14.00
2-WIRE VOICE LINE MONT	UEPXP	ž	\$	ž	ž	\$14.00	\$	≨	\$	¥V.
CALING PORT	UEPXO	ž	\$		ž	2	Ž	· A	3	
CALLING PORT	INCPYO	2	1	:	:			5	٤	Y _N
2-WIRE VOICE UNBUNOLED 1-WAY OUTGOING PBYMEASURED PORT	UEPXS	≨ ≩	87.5	8	\$ \$	¥ 14	≨ 2	₹.	≨:	NA.
2-WIPE VOICE LABUADLED 2-WAY PBX SOUTH CAROLINA AFEA PLUS CALLING PORT	IX430	\$	1	2	;	1	:		5	00.514
2-WIPE VOICE UNBUNDLED PRIX COLLIFTINGE & MEMPHS CALLING PORT	0.43	1	1	§ ;	E :	≨ :	ž	ž	≨	¥
2-WIFE VOICE UPBLACKED 5-WAY PBX TENNESSEE REGIONSERV CALLING PORT	UEPX	470	2	£ \$	\$ 3	£ :	ž	¥	\$	814.80
LOCAL NEMBER POSTANS ITY REAL SEES ONE SEES SACRES	2001			٤	Š	Š	ž	ž	≨	814.8
	5									
1 1										
P.C 2- Wire Vises Grade Logo - Salewide P.C 2- Wire Vises Grade Logo - Zone i	VEPLX	≨:	¥	≨	ž	¥	¥	\$14.18	≨	ž
HC - 2- Wire Valor Grade Loop - Zone 2	2 2	§ ž	\$14.90	\$10.80	ž	3.58	ž	YN	≨	\$15.92
PC - 2- Wire Voice Grade Loop - Zone 3	JEP X	≨	24.25	\$19.83	ž	76.30	ž	§ ž	\$ \$	27.16
Combination Name		≨	≨	¥	≨	ž	≨	¥	ž	ž
RC - 2-Wire Votos Grade Loop with 2-Wire Line Port, Stasswide	1 1	¥	¥	¥	ž	¥	WA	426.18	1	
RC - Never Works Care Lond with 2-Wire Line Port, Zone 1 (Note 6)		ž	06 823	27.80	ž	28.05	ΝA	ž	YN.	28 823
RC - 2-Whe Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6)	Note 1	\$ \$	15.23	2 2	¥	128.14	¥	\$	¥	\$34.79
Novementring Charges	11				<u> </u>	20.00	Ş	≨	≨	2 2
NATC - 2: www votes grade unbunded portroit combination - Add, with change		≨≨	3 3	3 3	¥	8.5	ž	3	ž	8 13
NFC - 2- who votce grade unbundled porytopp combination - 1st, no change		≨	141.50	3 2	٤ź	2 2	ž ž	8 8	≨ ≨	2 2
MTL: 2. www voca grade urburded portrop combination - Add, no charge		≨	25.50	2 3	ž	8 28	¥	3	ş	3
NRC - 2-Win Voice Grade Logo/Line Port Combination - Subsequent		MA	\$10.00	\$1000	\$10.00	\$10.00	ž	\$10.00	2	4000
M.C. 2. With Young Grade Local-Line Put Continued n. 055 LSR Charge, Eschoric, per LSR received from the CLEC by one of the OSS Interactive. Interactions (New 3)				,						3
199C - 2-Wire Votes Grade Loop/Line Port Combination - Incremental Cost -	30,000	ş	22.23	22.50	ž	950	ž	53.50	ž	8 8 8
Menual Svc. Order vs. Electronic - 1st IABC - 2-Wire Vivine Carda part Ine Syc. Communication	SOMAN	ş	85128	19 63	\$	23 153	ş	\$40.18	MA	8 03
Menual Svs. Order vs. Encirons - Adm	SOMM	ş	85128	\$7.80	ş	27.72	¥¥	89.45	ž	8.2
Update - Electronic		ş	5	OBT	\$	11.03	\$	919	1	ş
RMC-2 Wire Yoo's Grade Loop/Line Purt Continuation - Subsequent Distribuse Update - Manual Service Order		1	į	1	:					
HRC - Bedronic Service Order Disconnect		ž	\$0.42	₹	≨≩	¥ 5.12	\$ \$	2052	≨ ≨	<u>8</u> 3
PARC - Incremental Manual Service Order Disconnect			3	0000	177	w.*	3		:	
							٤	3	5	200
Currently Combined										
2-Wire Voice Grade Loop with 2-Wire Line Port										
2-Wind Votes Grade Line Port (Nee,) per secrets 2- wee votes urburded out - materiae	200	2	7							
2-vira voice unbundled port with caller ID - madence	UEPRE	2 2	2 2	2 2	5 5 X 2	8 22	2 2 2	223	8 3	2 2
2-em value arate urbundled port outgoing only - replaces	UEPRO	e a	S E	£ 15	1973	85.73	21.03	82.28	93	7
Gill Communication of the Comm	UEPAR	8	ž	≱	٧×	¥	ž	≨	\$	≨
C-ere voice grade undunded Kentucky entended total disting penty port with caster.	UEPRI	≨	\$	\$	ā	ş	ş	ş	ž	\$
	UEPAS	ş	2	ž	\$	85.03	ş	ş	1	1
2 with voice grade unbundled Mealestyl extended local dialing parky port with protein.	3	-						5	5	1
	NE LY	Š.	ž	\$	ž	¥	21.23	≨	¥	ž

DESCRIPTION		2080	7	14	49	KY	3	2	S.	8	N.L
2-wire volce grade unbundled South Carolina entended local disling party port with caller ID	alling painty port with	UEPAU	ž	≨	2	ž	¥	¥	ž	\$3.69	ž
ice grade unbunded Temessee actended local	dailing parity port with										
Caller 10		OEPAD.	≨ :	≨ į	ž	ž	ž	ž	¥	ź	2 2
2 were voice unbundled Fioritia area calleng with caller ID - reacts	Series (The B.)	CEPA	¥.	56 13	ž	Ý.	YZ.	ž	¥	¥.	≨ :
2-wre voice unburded Louisana Area Plus with cafer 10 - resid	dence (AC7)	UEPAH	\$ \$	₹ ₹	žž	≨≨	8 22	ž	¥ ×	٤ž	1 2
2-wire volce unbundled South Carolina Area Cabing port with Caller (D - neidence	Mer ID - residence		:	;		;		:	:	1	
2-vire volce unbundled Temesese Area Cathro port with Cather ID - residence	r ID - residence	2	ž	Ş	Ş	Ş	Ş	ž	ž	2	Š.
		UEPAK	ž	ş	٧×	ž	¥	ž	ž	≨	2 2
Z-with voice unbundled Temessee Area Calling port with Carler ID - residence (TACER)	r ID - residence	UEPAL	ž	ş	NA	Y.	NA	¥	\$	¥	2
1	r ID - realdence	UEPAM	ž	≨	ž	\$	ž	YN.	¥	¥	x
2-wire voice unbundled Tennesses Area Carling port with Caller ID - residence (196720	r ID - residence	UEPAW	2	₹	ź	ş	\$	2	\$	ş	2
2-who voice unbunded Terrosese Ana Caling port with Caller ID - residence	r ID - residence					:	:				
2-we vote unburded rea, the usage the port with Caller ID (L.)	TOP O	CEP AS	¥ 23	¥ 5.5	× × ×	¥ 50	K2 55	≨ 2 2	¥2.28	¥ 23	Z Z
2-Whe Valce Grade Line Part (Bus.), per month											
2-with voice unburded port without Calest ID	270		2 2	9 5	2 2	5 S	22.55	2 2 2	22.28	2 2	2 2
2-wire value unbundled autgaing only port		UEPBO	838	81.26	\$1.79	500	22.55	\$2.12	\$2.28	2 22	7 7
2-wire voice grade unbundled Alabema entended local dialing parity port with calls	parity port with caller	CEPAW	02 ZB	ş	¥	ş	ž	ş	\$	\$	\$
2-win voice grade unturnated Kentucky entended local desiring penty port with called	party port with caller	UEPBM	\$	\$	AN.	ā	ν.	2	Ž	ş	\$
2 were voice grade unbundled Louisiers eneroted local deling party port with	partly port with	UEPAK	\$	¥N	2		25.03	₹¥	\$	3	2
2 -errs votos grada unbundied Messesgoi sonended local disting partly port with	perty port with	LEPAY	ž	Ą	¥ Z	2	¥	6.6	*	ş	¥
2 wire volce grade unbundled South Carolina antended local dealing parity port	aling panty port with	440	1	1	;	1	12	3	3	8	4
2-wre volce grade unbundled Termeses extended local disting partly port with	permy port with	1	5	5	5		5				
Callerin		UEPAV	¥	≨ .	ž.	≨ .	¥	¥	¥	¥ ;	X :
2-we vote unturded LA flue Area Caling Port with Caller D (5UC)	Single	VEPA	₹ ¥	9 ₹	2	ž≨	22.23	¥ \$	ž Ž	8 ≨	X X
2-airs volce unbundled SC Bus Area Calling Port with Caller ID ((EME)	UEPAB	ž	ş	ž	\$	≨	ž	¥	\$3.69	¥
2-wire valoe unbundled TN Bus 2-Way Area Calling Port Economy Option (TACC))	amy Option (TACC1)	UEPAC	ž	Y.	¥	ş	ž	ž	NA NA	ş	Z Z
2-wire volce unburded TN Bus 2-Way Area Calling Port Standar	and Option (TACC2)	UEPAD	ž	ş	ž	ş	ž	ş	NA	¥.	2
2-was voice unbunded TN Bus 2-WAY Collecylle and Memoria Local Caffing Port	is Local Califor Port	West.	1	474	1	4	474	¥N	**	42	2
2-Wre Voice Grade Loop (3L1)											
RC - 2- Wire Votos Grade Logo - Szawide		UEPLX	≱ .	¥ .	¥.	≨ .	≨	¥ ;	17.18	NA S	¥ 65
AC - 2. Wire Voice Grade Loop - Zone 2			Ę (23	\$16.51	\$12.47	3 /2	\$24.14	619.33	٧٧	\$25.66	\$14.42
RC - 2- Wire Voice Grade Loop - Zone 3		ZEPZ.	\$25	\$24.28	\$19.00	27.78	649.30	23.63	\$	86 CC3	11.12
IFC - 2- Wire Voice Grade Loop - Zone 4			2	ž	≨	ž	ž	28.47	≨	≨	ş
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Statewide	•	Note 8	≨	¥	ž	¥	≱	¥	816.46	¥	¥
FIC - 2-Wire Volce Grade Loop with 2-Wire Line Port. Zone 1 (N	(Note 6)	Note 6	55 55	\$16.25	\$12.50	217.40	899	\$16.71	×	1207	8 2
FIC. 2-Wire You's Grade Loco with 2-Wire Line Port. Zone 3. Ov.	S S S S S S S S S S S S S S S S S S S	No.	E	8	23.23	203	81.53	22.623	₹ ≵	89753	10.85
FIC. 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)	Note 6)	Note 8	ž	¥	\$	¥	ž	65 903	3	¥	ž
Norwecenting Charges NHC - 2-Wire Vide Grade Logo Line Fort Continuation - 1st, Se	with se is	USAC	8	1961 05	10.23	\$10.00	8 53	\$6.20	123	81 59	\$1.00
NRC - 2-Wine Voice Grade Loop/Line Port Combination - Addt, Switch as is	Switzh as in	USAC	170	7964	\$0.3106	800	50 23 20 23	17:03	9 9	\$ 5	\$0.28.08
AHTC - 2-Wing You'de Lindow Loop/Line Fort Compriseon - 180, Sw	MATCH WITH CHANGE	3	2	*O.1364	5	an one	8	2	// X	201	3
NRC - 2-Wire Votes Grade Loop/Live Port Combination - Add1, Switch with change NRC - 2-Wire Votes Grade Loop/Live Port Combination - Subsequent	Switch with change	SASU SASU	\$10.05	510 00 00 018	\$10.00	800	\$1000	\$10.00	\$10.00	810 00 00 01	\$10.00
NRC - 2-Wire Votes Grade Loop/Live Port Combination - OSS LSA Cherga.	LSA Charga. S Interactive										
- 13		SOPPEC	8	27.22	8 8	952	88	8 23	828	8	8
Marinal Sve. Order vs. Bectonic - 1st		SOMAN	\$40.71	\$21.58	1300	\$19.50	20123	25 075	\$1018	81 534 81 138	\$30.00
HRC - 2-Wire Voice Grade Loop/Line Port Combination - incremental Cost Manual Style Order vs. Electronic - Adm	mental Cost -	SOMM	2	3	2.0	8618	22	8 2	\$ \$	2	82.48
1											

METWORK ELEMENTS

Attechment 2 Eshbar C Rates - Page 41

6		0060	*	14	45	2	1	54	3	3	
<u></u>	NRC-2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database				5		5		2	3	E
+	NGC, 2 With Villa Craft 1 and its Day Perfection C. Access of Contraction		3 2	6	<u>6</u>	92	22.11	2923	\$142	17.08	\$0.76
_	Update - Menual Service Order		\$	ş	Ę	į	:	:			-
Ε	NAC - Electronic Service Order Decornect		S X	20.42	S Z	2	NA NA	8 2	2102/	5 22	27.97
L	NRC - Incremental Manual Service Order Disconnect		9000	12	9000	\$20 m	W OCA	W VG	0000	£ 50.00	200
Н	MRCs for New (not Currently Combined) so ordered in Georgia:									3	30.00
Н	NRC - 2-Wire Votos Grade Loop with 2-Wire Line Port - New - 1st	UEPR	ź	ž	\$22.14	ž	¥N	ž	¥	¥	AN AN
٢	NRC - 2-Wire Voice Grade Losp with 2-Wire Line Port - New - Addi	UEPR	≨	¥	\$15.25	ž	¥	¥	AM	AM	4
4	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPAC	ž	¥	\$22.14	ž	ş	ž	ž	4×	ž
 	MRC - 2-Wire Volce Grade Loop with 2-Wire Line Port - New - Addi	UEPRC	¥	¥	\$15.25	ž	¥	¥	₹	ž	ž
_	MRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	06430	YN .	¥	11 223	ş	ž	¥	ž	Ş	¥
-	MRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Addf	UEPRO	¥	ž	\$15.25	ž	Ş	¥	¥	ž	¥
1	NAC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPAP	¥¥	¥	P1 725	ž	ž	≨	ž	ž	¥
1	MRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - Merr - Addit	UEPAP	¥	Ş	\$2.518	ş	ž	¥X	¥	₹	¥
4	MFC - 2-Wire Yoke Grade Loop with 2-Wire Line Port - New - 1st	UEPBL	¥	ž	\$22.14	ž	ΥN	٧×	¥¥	ž	¥×
_											
_	NPIC - 2-With Voice Grade Loop with 2-Wire Line Port - New - Addit	UEPBL	ş	Ş	\$15.25	ž	ž	¥	¥	2	£
	NFIC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPBC	ž	¥	\$22.14	ž	ž	ž	NA NA	¥	ž
1	MARC - 2-Wins Voice Grade Loop with 2-Wins Line Port - New - Addi	SEPBC LEPBC	ş	٧×	\$15.25	¥	¥	ž	¥	ž	ž
4	MARC - 2:Wire Voice Grade Loop with 2:Wire Line Port - New - 1st	CEP80	Ź	٧×	\$22.14	ž	NA	ž	٧×	ď Z	ş
1	WHICH Z. WING YOO'S GREEK LOOD WITH LINE POTT NEW - ACCT	UEPBO	Ş	ž	\$15.25	≨	٧×	ž	VΑ	٧¥	¥
1	MATC - 2-Wing Votos Grade Long with 2-Wing Line Port - New - 1st	UEPBI	ş	≨	22.14	ž	ž	₹ Z	ΝA	٧V	¥
1	NHC - 2-WTB YOLD GARDE LOOP WER 2-WTB LING FOAT - New - AOOT	UEPB1	≨	≨	\$15.25	ž	ž	ž	¥	٧¥	¥
+	NAC: 2-Wire Value Grade Local Ine Part Combinedon - Subsequent	1 ICARS	472	***	200		• 13				
L	MSC - 2-Wire Votes Grade I no wet 2-Wire I in Brot. Man. Character	Ž	5	5	310.00	\$	Š	ź	¥	ž	ž
£	NACC 2: Wire Voice Grade Loop with 2: Wire Line Port - New - Disconnect - Addi		ž	2	25	2 3	É	2 2	\$ 2	ž	\$ 5
E	MRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge,							5	5	5	5
	Electronic, per LSR received from the CLEC by one of the OSS interactive		:	;	:	;					
Ŧ	(monitored (monitored)		ž	ž	23 50	ş	ž	ž	≨	ž	¥
			≨	Ş	80,728	ž	¥	ž	ž	2	47
	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - Incremental Cost Manual										
-	vs. Bectrarite - New - Add		Ş	ž	\$8.19	\$	¥	ş	¥¥	\$	¥
_	PATCA Z. WITH WOLDS UKING LOGOLUM POTI COMDINATOR - SUDBRIGHT DELIBERA		ž	ş	g	\$	¥	¥M	3	2	1
E	NARC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Despises								1		£
	Update - Menual Service Order		ž	¥	Q.	ž	¥	¥	¥	W	¥
=	MRC - 2-Wite Voice Grade Loop with 2-Wire Line Port - Incremental Cost Manual Ing. Backtonic - New - Disconnect		\$	\$	£11 113	3	1	;	1	1	;
E						4	5	1	§.	٤	٤
	1										
1	2- Wire Voice Grade Loop - Bus Only with 2 -Wire DID Trunk Port										
	2 - We Lie Port - DD Treat Port aw month	, Colon	2	8		100	:	5	*: 6:3	<u> </u>	2
E	1				~				\$16.16	3	2
_	2-Why Yoles Grade Leap (\$1.2)										
	RC - 2- Wins Voice Grade Loop - Statewide	UECD1	¥	¥	ž	≨	≨	¥	\$11.76	¥	ž
1	P.C2-Whe Voice Grade Loop - Zone 1	UECO	17.98	818 48	3	\$17.78	\$17.66	\$18.35	ž	15125	9 60
Ξ	PC. 5. Was Voins Grant Long. 2004.1	5 2		27.53	23.45		2002	R C	≨ :	20.23	8
E	PC - 2- With Voice Grade Loop - Zone 4	1000	1	NA NA	MA	2	2 4	2 3	2 1	8 4	7
	Combination Rates										5
	FIC - 2-Were Volce Grade Loop with 2-Wire DID Port, Statewide	Note 8	¥	MA	ž	ş	ž	¥	23.73	ΥN	₹
	FIC - 2-Wire Volce Grade Loop with 2-Wire DIO Port, Zone (Phose 6)	Note 8	282	19.23	81.852	24.72	530 77	28 72	ş	28 72	\$16.36
1	FIC. 2-Wins Voice Grade Loop with 2-Wins DIO Port, Zone 2 (Note 6)	2	5 X	21.73	800	20.20	2	2.0	Ş	ā	\$19.87
1	HC -2-Wife Voice Grade Loop with 2-Wife DIO Port, Zone 3 (Note 6)		5	27.23	22.27	62	875.08	45.9	Į.	200	22.5
E	MRC 2- Wine Volce Grade Loop with 2- Wine DID Port - Conversion - Switch As is -					٤	5	3		5	
1	(1st port	5	21462	29.118	\$166.06	<u>8</u>	\$14.80	814.80	\$13.26	21162	2 2
	Each Add Port	CASO	22	22.23	\$140.01	Æ	22.03	22	\$6.38	23	r r
	NRC-2: Wire Voice Grade Loop with 2: Wire DID Fort - Convenion with charges -	16.410	27.0	275	61 55 CB	C S	9778	8	* 1.4	27.5	2
	NRIC 2- Wire Voice Grade Loop with 2- Wire DID Port - Conversion with changes -										
<u> </u>	Each Add port	USAIC	2	22.2	P. 00.0	2	22.03	27 13	28.39	E S	25.23
	I MARC - 2-WIND DID SALDERQUEST ACEVITY - PER SWC CAUGE - ACCI LEGITHE, PER LINES.	5	?	2	Ę	- ≨	3	3	ž	2	- ≨

IND. 1874	0 8 0	₹	3	8	2	-	33	5		
Texture Control of the Control of the CAS LSR Charge, Cattornic, part LSR transfer from the CAEC by one of the CSS trianscrive.					_			2	2	=
NRC-2: Wire Voice Grade Loop with 2: Wire DID Port - Incremental Cost - Manual	SON T	8	22	82.80	8	23 55	83.50	S 53	\$3.50	\$3.50
Service Order - 1st	SOMAN	\$19.99	82158	S27 M	4	8				
NATC: 2: Wire Voice Grade Loop with 2: Wire DID Port - Incremental Cost: Manual Service Order - Acts	-			3	6	R	56	SE CS.	86	21.0
MRC - Electronic Service Order Decornect	SOMAN	20.00	3. 2.	298	\$19.99	\$19.99	\$19.99	\$11.34	\$19.99	29 60
MRC - Incremental Manual Service Order Disconnect	-	000	2 2	200	20.52	3 5	3	20.42	20.42	27:05
OD Traft Territology (new conduction)						3	300	B	8	20.00
DIO Numbers, Establish Trunk Group and Provide First Group of 20 DIO Numbers	Q	8	80 03	800	00 0 3	8	88	80.03	\$0.00	00.03
I'M, QA NG A SC OND	20	80.00	80	80.08	8	S	\$	8	1	
(AL, KY, LA, MS, & TN). In addition, Provide Additional Dio Numbers for each							3	3	B	80
States	ğ	00 08	90.00	80.00	90 OS	8	8	8	ş	\$
	Š	88	80.03	00.03	8	8.00	8	00.03	803	88
2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Port							1			
2-Wee ISON Dietal Grafe Loca	CEP78	\$16.42	\$8.51	\$13.47	\$12.89	\$11.42	16.158	\$24.37	220 74	618.51
AC - 2-Whe ISDN Dignal Grade Loop - Statewide	160.00	3				·				
RC - 2-Wire ISDN Digital Grade Loop - Zone 1	USLZX	1212	A 22.5	\$ 163 163 163 163 163 163 163 163 163 163	₹ 2	ž	ž	\$19.06	¥	¥
RC - 2-Win SCIN Digital Grade Loop - Zone 2	USIZX	\$37.74	27.90	25.27	97.153	23/23	28.92	ž	3 3 3 3	\$16.20
RC - 2-Wire ISDN Dignal Grade Logo - Zone 4	VS 23	\$50.30	\$30.78	\$40.17	85.24 25.38	248 42	41.40	2	20.03	2
Combination Pales		ž	ž.	≨	§	ž	25. 25.	\$	≨	NA.
Control of the Contro										
PC - 2-Wis ISON Digital Grade Logo and 2-are ISON Digital Prof - States Co.	2	¥	ž	₹	ž	ž	ž	\$43.45	ž	ž
RC - 2-Wire ISON Digital Grade Loop with 2-wire ISON Digital Port - Zone 2	2	8 2	E 020	205.36	9	2	\$73.77	٧N	\$60.42	134 41
HC - 2-Wire ISON Dignal Grade Loop with 2-wire ISON Dignal Port - Zone 3	NON	8.78	00.653	2 2	2 ×	3 2	200	≨:	\$73.96	26 903
The - 2-With ESUM Digital Grade Loop with 2-wire ISON Digital Port - Zong 4	NO B	≨	¥	ž	≨	2 ≥	\$108.55	Įį	B VN	26.25
NETC - 2-Wite ISON Digital Grade Leggi?-wire ISON Digital Prof - 1st consumers	2,0	:	-							
	3	21.57	2,992	22.0 28	27.12	£79.04	578 12	\$174.35	\$79.12	811723
MATC - 2-Wire ISON Digital Grade Loop/2-wire ISON Digital Port - Add convention MATC - 2-Wire ISON Digital Grade Loop/2-wire ISON Digital Res.	ESVS1	20.00	884.04	\$156.92	20.75	\$53.97	20.00	\$174.35	3	\$117.23
Subsequent Activity	USASB	95.0.50	5	5	50.53					
MFC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSA Charge, Electronic per 158 most and from the CF for one and the CF of the control of the CF of the control of the CF of the control of the CF of the control of the CF of the control of the CF of the control of the CF of the				2	Remark	2	9	653.50	95	\$212 BB
Interfaces (Acre 7)	SOMEC	05:E3	23	80	83	\$	\$	5		;
Menual Service Order - 18	,						3	3	2	2
NFC - 2-Wire ISON Digital Grade Loop/2-wire ISON Digital Port - Incremental Cost			8	8	8658	8 615	818 98	84 61	8 818	\$19.00
Manual Service Order - Add	SOMAN	818 99	82158	\$19.90	\$19.90	\$19.99	\$19.99	213.9	218 28	919
NPC - Incremental Manual Service Order Discornace		88	3 2	200	50.42	300	22.53	20.42	\$ 42	\$0.42
					-	3	8	820.00	88	00 02
Port										
4 - Who ISON DS1 Digital Trans Port	CEPPP	\$196.02	898.39	\$163.16	\$113.21	\$107.55	1321	17901	2,710	07.000
RC - 4- Wire DS I Digital Loop- States dos	97 72	3	*77							04.0
HC - ← Win DS1 Dignal Loop. Zone 1	480	281.74	44 022	25, 52	¥ 2	≨	YN S	\$62.71	ž	٧V
PC - + Wis DSI Detail Loop Zons 2 PC - + Wis DSI Detail Loop Zons 3	180	878	8119	13.21	\$135.15	\$129 12	85.738	٤ž	269	\$75.40
RC - ←Wire DS1 Digital Loop - Zone 4	97.77	62X 24	2 3	\$101.80	20.28	\$344.16	83 96 3	ž	\$119.06	\$96.59
Combination Rates		٤	٤	Ş	≨.	ž	\$127.47	≨	≨	¥
RC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port - Statewide	Note 8	≨.	≨	¥	≨	ž	\$	24172	2	Y
RC - 4 Wire DS1 Digital Loop with 4-wire ISON DS1 Digital Port - Zone 2		27.72	\$167.67	\$218.68	251928	52 802	24 28 28 28	¥	\$274.40	\$136.13
RC - 4-Wire DS1 Digital Loop with 4-wire ISON DS1 Digital Port - Zone 3	NOTE 8	15.903	80 0623	50 8903	200	2000	2,000	≨ :	69 003	8153 80
RC - 4-Wire DS1 Digital Loop with 4-wire ISON DS1 Digital Port - Zone 4 It coul flumber Personality	NOT S	¥	≨	ž	ž	ž	8340 GB	≨≩	g ≨	\$176 99 KA
Local Number Portecting (1 per port)	NO.	¥ 12	* 19	,	7					
Interface (Provatoring Only)				6	2	\$1.75	r ā	ž	r E	2, 73
Vocazius Dodai Data	PR71V	8	90 93	00 Q£	8003	\$0.00	89	08 03	88	80.00
Imend Deta	91434	88	88	88	8	88	8	808	8	80.00
Non-Necembe Charges		***	8.8	8	88	88	8 8	88	80 03	00.03
			-				-		-	

š	CHIPTION	080c	7	2	Ϋ́O	È	3	1	¥	2	3
_	NRC - 4-Whe DS1 Dighal Loop with 4-whe ISDN DS1 Dighal Pon Combination -	800	2	10.75	3	1					
	NRC - 4-Wire DS1 Digital Loop with 4-wire ISON DS1 Digital Port Combination - Add convention	200				2	200	OC ORDER	0.00	00.00	2028.53
1	NRC 4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port -			1	GG GGA	11.761			6.67	\$157.17	\$328.53
#	NRC - 4 Wire DS1 Digital Loco with 4-was ISDN DS1 Digital Port Contribution.	SVS	8	80 623	228.71	88 83	5 82	\$29.08	26 90	23.08 23.08	\$28.39
\rightrightarrows	Subsequent Investigation Municers	PRYTO	8 08	\$0.9804	\$0.9666	96 OS	\$ 6.0 \$	36 Q 3	\$1.17	\$0.96	\$0.9353
\exists	Next. 14 was UST Lighted Loop was 4-wite ISON UST Lighter Part Combination. Subsequent Quineral Telephone numbers.	PRITT	82	82.23	22.73	2052	522.88	823	28.17	22.02	\$22 W
	NRC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port. Combination - Subsequent Interest Telephone Municers.		80.994	20.844	\$ 94 1	868	8	\$46.06	25.83	88	7
	NRC - 4-Whs DS1 Dighal Loop with 4-wire ISDN DS1 Dighal Port Combination - Subsequent Sevice Order Per Order	-	8147.47	\$147.47	\$147.47	\$147.47	\$147.47	20.013	X 546	*****	6.00 Te
_	MRC - 2-Wire Votes Oncise Loopf Lies Port Combination - DSS LSR Changa. Becarach, per Application from the CLEC by one of the OSS treatments of the CRES treatments of the CRES treatments.	l	3								
	NRC - 4-Wine DS1 Digital Loop with 4-wire ISDN Digital Port - Incremental Cost.	SOMEON I	8	ę a	86.23	DG 124	8	8	8	8	23 50
土	Might Service Lycer + 18 MRC - 4-Whe ISON Digital Loop with 4-wire ISON Digital Port - Incremental Cost-	SOMAN	86.5	85.58	26.61	219 30	\$19.99	\$198	\$19.99	\$19.99	\$19.99
\pm	Menual Santos Order - Add	SOMAN	\$19.99	25.53	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99
坩	NRC - Incremental Manual Service Order Decorract		\$20.02	\$3.84	\$20.00	2002	\$20.02	208	25.05	\$ 0.03 \$20.03	\$0.03 \$0.03
\pm	A. Wite Del Chapter over 4 With Drafts Tours and										
#	4 - We DOTS Digks Tunk Part (Formery DD Tunk Port	TOO	\$130.23	\$63.31	\$120.80	\$83.28	\$149.27	\$146.46	\$123.66	180	\$120.00
\pm	4 - Wire DS1 Digital Loop - Stateside	303	≨	ž		¥	AM	42	14 (23)	12	12
Н	4 - Wire DS1 Digital Loop - Zone 1	USUDC	151.74	\$64.69	\$55.53	108.04	258.39	86.053	¥	19 653	2 2
_	4 · Wire D61 Docal Loop · Zons 2	og sa	\$24.05	17.102	\$64.13	\$135.15	536.73	\$67.58	ΥN	06 681	\$75.40
\pm	4 · Wire DS1 Digital Loop · Zone 4		2223	\$206.93	\$101.83	\$1.06.26	\$197.57	98.50	ž	\$119.06	\$96.59
Н	Complination Panes				5	5	٤	10.121	£	٤	Š
1	4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trush Post - Statewide	Note 8	2	ž	ž	ž	ž	ž	\$166.36	ž	N.A
\pm	4 - Wire CS1 Outsid Loco with 4 - Wire DOTS Trust Port - Zone 2		2014.26	8128.00	5176.33	2189.22	25.05.59	\$197.45	ž	8	\$30.28
Н	4 - Wins DS1 Digital Loop with 4 - Wins DD/TS Thurk Port - Zone 3	Note 8	क्रम्बद	277.24	1222 m	12 69.21	23.68	2000	£ ≨	2 2	\$134.14
\pm	4 - Wire CS1 Digital Loop with 4 - Wee COTTS Trusk Port - Zone 4	Note 6	ş	≨	¥	Ş	ž	\$273.83	¥	¥	ž
Н	Local number Portability per DSO Activated	SAC N	81.13	\$1.18	\$0.15	\$3.15	\$3.15	\$3.15	\$3.15	\$1.63	\$1.03
\pm	Central Office Terminaling Point	CTG	80.08	80.03	00.00	00.03	00.03	8	8	80.03	00.0\$
\pm	Telephone Number / Trunk Group setablishment							T			
\parallel	Felectrona Mumber for 2-Way Trunk Group	uotax	88	00.0	80	800	80 05	8	00.03	8 9	00 05
\pm	Telephone Number for 1-77 by Converse I name, unough	20103 20103	8 8 9	88	88	88	88	88	88	88	88
	DIO Numbers, Establish Trunk Group and Provide First Group of 20 DIO Numbers, IRL, GA, NC, & 9C ontol	204	90 08	00 08	8	8	8	8	8	8	\$
	DID Numbers, Establish Truth Group and Provide Frest Group of 20 DID Numbers										
	Group of 20 DID Numbers (Valid in Al Sunes)	Ş	80 00	80 06	88	808	80 00	80 08	\$0.00	00 0\$	8
	DIO Numbers, Non-consecutiva DIO Numbers, Per Number	Ş	00:03	00:03	90.0\$	00:0\$	90 03	00:03	00 03	8	00 03
	Interoffice Chemnel Minespe - (Dedicated DB1) FX/FCO for 4 - Wire DB1 Digital										
\pm	Fixed cost 0-8 miles (Facilities Termination)	10401	87.5	29:261	86.03	88.88	07 05	\$74.40	\$71.29	3	\$75.60
Н	Additional coets per mile 0-8 miles	100A	0289 05	\$0,200	\$0.3068	\$0.4500	10.7831	9699	\$0.0783	\$0.7598	\$0,3525
\pm	Fixed cost 9-25 inflee (Facilities Terraination)	-LNG	888	88	800	800	00.00	00.00	\$0.00	00'05	00 05
\pm	Fixed cost 25 + miles (Facilities emination)	200	00.00	889	20.00	88	388	889	0000	86/08	200
Н	Additional costs 25 + miles	LNOC	\$0.6820	\$0,2000	\$03068	60.4500	\$0.7831	\$0.6696	\$0.0763	\$0 7598	\$0.3525
\pm	Enterned Partyments Charact										
Н	Exhanced Performance Charges - as negotlated in contract	UDTPC	X.	¥.	ZE.	¥8.	TBY.	ABL.	TB.	AB1	Mg
1	Nov-recurring Charges	Ī								T	
	NRC - 4-Who DS1 Diguel Loop with 4-Who DDITS Truth Port Condustrin - OSS LSR Cherge, Epidronic, per LSR received from the CLEC by one of the OSS				-						
+	INTO 4-Wire DS1 Detail Loo with 4-Wire D0118 Trusk Port - Incremental Con-	200	Q; [2]	2	8	85.03	8	8	8	8	8
\exists	Menual Service Order - 1st	SOMM	819 88	25 128	837.88	\$19.99	\$19.99	818 99	218 23	219 33	\$19.89

Arachment 2 Exhibit C Rees - Page 44

\$19.99 \$20.00 \$27.291 \$312.91 \$312.91 \$312.91 \$312.91 \$312.91 \$312.91 \$310.67 \$100.67 \$100.67 **4** ¥ 3

Ē	DESCRIPTION	nsoc	7		40	4.4		33	4	5	
	NRC- 4-Wire DS1 Digital Loop with 4-Wire DDITS Trunk Port - Incremental Cost-										
E	NHC - Bectoric Sevice Order Decorrect	ACTION OF	180	200	180	180 99	8 S	\$19.99 TRO	\$19.99	8 5 Car	
Ц	NRC - Incemental Manual Service Order Disconnect		802	19.03	\$20,00	00023	\$20.00	\$20.00	20.00	\$20.00	
	NHC-4: WHE US! DIGHE! LOOP WITH 4: WHE DO!! S TAIR! FOR -CONVENION: Switch as is - 1st	USACA	TBO	\$268.62	\$269.96	\$261.15	\$266.76	OST	9C 0675	TBD	
	NHC-4 - With DS1 Digital Loop with 4 - Wire DOITS Trusk Port - Conversion - Switch as in - Additional	USACA	180	\$134.07	\$249.96	\$134.08	\$123.16	OBT.	\$490.38	ð	
\Box	NHG-4 - Wire DS1 Dignal Loop with 4 - Wire DD153 Trunk Port - Conversion with DS1 changes - 1st	USAWA	Ę	\$268.02	\$269.96	\$261.15	\$266.76	22	\$490.38	180	
	WRC-4 - Wire DSI Dignat Loop with 4 - Wire DDITS Trunk Port - Convention with IDSI Changes - Additional	USAWA	OET	\$134.07	85 G3C\$	\$134.08	\$123.16	<u>8</u>	8C 064-8	56	
	MPC-4 - Wire D81 Digital Loop with 4 - Wire DD/T8 Trunk Port - Conversion with Change - Trunta - 1st	USAWB	96	22.8952	8.59	\$261.15	12.00 75	92	25.033	P	
	NRC 4. Wis DSI Digital Loop with 4 - Wiss DCRS Trust Port - Convention with Charge - Trusts - Additional	USAWB	OBL	8134.07	8249.96	81.34.08	£123 18	S	25.673	Ę	
	NRC 4 - Wire DS1 Digital Loop with 4 - Wire DDITS Digital Trusk Port - Submanian Section Article Per Section October	16.46.4	ş	8				1			
IJ	WPC-4 - Wire DS1 Digital Loop with 4 - Wire DDTS Trush Port - New - List	TOON	g≱	ã≱	\$658.30	8 ₹	NA S	NA NA	NA NA	2 ≱	
	NRC 4 - Wire DS1 Dights Loop with 4 - Wire DD/TS Trust Port - New - Additional	LOON	ž	¥	\$514.02	¥	Ϋ́	NA	NA	ž	
	NHC 4 · Wire LS1 Dignal Loop with 4 · Wire DD(18 Trunk Port - New - 1st - Obsonned	TOOL	ş	¥	78 0	ž	ž	¥	NA	MA	
\exists	NRG 4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - New - Additional Discornect	roon	OET	¥	CBT	OEL	Ž	TBIO	\$11.96	58	
\exists	ARTC 4 - Wire DS1 Digital Loop with 4 - Wire DDRS Trum Port - Subsequent Cheminal Activation - Per Channal - 2-Way Trum	UDITA	<u>\$</u>	\$21.98	\$28.71	26 96	\$26.60	OBT.	\$146.91	TBD	
\exists	NRC-4 - Wire DSI Digital Loop with 4 - Wire DDITS Trunk Port - Subsequent Cherrel Activation - Per Cherrel - I-Way Outseard Trunk	erron	180	\$21.96	\$28.71	96'92\$	\$26.60	180	\$146.91	9 E	
	IMPC -4 - Who DS1 Digital Loop with 4 - Wire DDITS Trush Port - Subsequent Chernel Activation - Per Chernel - 1-Way Inseed Trush Without DID	DE DE	<u>6</u>	\$28.96	\$28.71	96 92\$	826.60	OBT	8146.91	780	
	NRC 4 - Wire DS1 Digest Loop with 4 - Wire DD/TS Trust Port - Subsequent Chemet Activistion - Per Channel - 1-Way Inward Trust With DID	e e	8	8 22	17.82	86 823	85680	2	8146.91	180	-
	NRC 4 - Wire DS1 Digital Loop with 4 - Wire DOTS Truth Port - Subsequent Chemist Adhvaton - Per Chemist - 2-Way DID with User Transfer	a LLOS	5	80.03	17.823	86.823	8963) E	24.91	8	"
	AMC 4 - Wire DSI Digital Loop with 4 - Wire DD/TS Digital Trush Port - Subsequent Signafing Changes		ĐĘ.	CMET	TBO	OE.	TBO	780	\$29.65	8	
	NARC -4 - Wire DS1 Digital Loop with 4 - Wire DOMS Digital Trunk Port - Subsequent Telephone Numbers		OBT.	OBT.	QE.	Ē	OBT	Ē	\$120.98	18	
	NHC: Intercribe Chemist Missage (Disdonist DS1) FX/FOD for 4 - Wire DS1 Digital Loop with 4 - Wire DOTTS Trust Portifications of the facilities of Territation 1 star New One.	500	26 25	\$	8147.07	\$	ş	ž	2	ž	
	NRC: Inserofroe Chernel Mésage - (Dedicated DS1) FXFCO for 4 - Wire DS1 Digital Loop with 4 - Wire DD18 Tunk PortFaed cost 04 miles (Facilities Terminator) - Additional - See Divis	2	61 69 69	1	K 11.8	1	2	4 2	1	1	1
	NRC: Interoffice Charrel Milesogs - (Dedicated DS1) FXFCD for 4 - Wire DS1 Organ Long Mileson - DOTTS Trans Profession on 0.8 miles (Facilities Tartheten) - Decorated - 14 - Mar One	8	3	3	ž	1	2	1	1	2	
	IMPC: Friendfülber Chemnell Misseger (Diedlanked DSI) FXFCO for 4 - Wins DSI Orgal Loop with Annual Misseger (Diedlanked Cost Oil make filedlikes Termination - Discounted Additional - New Only.	3	3	2	2	2	ž	2	<u> </u>	2	
П	MODIFIED OF THE PARTY OF THE PA										П
I	WRC - Supertrame Format - Conversion or new install 1st	18000	80.03	00.03	80.08	80	808	808	80.08	8	
	NHC - Superhane Format - Conversion or new Install Additional	35000	808	80.08	00 OS	88	800	8	80 OS	88	
1	MRC - Extended Superframe Format - Change or Subsequent Activity - 1st MRC - Extended Superframe Format - Change or Subsequent Activity - Activities	888	88	803	888	88	8 8	800	803	8 8	ſ
H	NRC - Extended Supertrains Format - Conventor or New Install 1st	300€F	80 08	90.00	8008	00 08	\$0.00	80 00	Н	00 00	
	NRC - Extended Superhame Formet - Convention or New Install - Additional	1900G	8	80.00	8	8	8	8	H	00 03	
	NHC - Extended Superform Format - Charge of Subsequent Activity - 18 NHC - Extended Superform Format - Charge or Subsequent Activity - Activities	300	8000	00 5598	00 0095	\$730.00	8608	00 009	\$615.00	00 909	
1	Abenute Mart Invertion (AM)	1	1								1
	MFG - Superhame Fornat - 1st	5007	8.8	88	88	88	88	88	88	8 8	Н
1	MAC - Extended Superirane Format - 1st	0000	883	88	888	88	88	88	88 93	800	
	HRC - Enerded Superhame Formet - Additional	040071	8	80 03	00.03	803	00 05	00 03	00 03	88	
Н											
	2-Wire Volce Grade Loop with 2-Wire Line Port PBX										
j											

Anachment 2 Earloit C Rass Pryc at

100 100			. 86.18							
COMBINATORY FROM RECARDED UEPPO E270 51.35 51.79		4	\$1.35		_		_	_	_	
COMMENTED MESSEE FOLLING CHEPTO E270 51.35 51.75	+	_			18	43.65	3	***	40.00	
COMENATION PAY LONG SECURING COMEN TO NA	╀	+	75.5	200	2	S	K 15	82 28	3 2	3
DEFENDANCE BEST BEST BEST BEST	╀	4	3		2	8	\$2.12	\$2.28	69 23	3
COMBINATION PRY ALGEMEN CALLING UPPAZE R2.70	╀	+	3 2		3	2	K 15	K 18	69	X X
COMBINATION PRY LOUISMUL CALLING UPPLY NA NA NA OT FEALMUL PORTY UPPLY NA NA NA OT FEALMUL PORTY UPPLY NA NA NA OT FEALMUL PORTY UPPLY RA NA NA OT GOARD PRIT TEALER PORTY UPPLY RA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA NA <th>╁</th> <td>+</td> <td>•</td> <td></td> <td>*</td> <td>8</td> <td>27.72</td> <td>K 28</td> <td>20 63</td> <td>2</td>	╁	+	•		*	8	27.72	K 28	20 63	2
		_	≨	ž	¥	¥	ž	2	2	3
OUTGOING PIX TENENSEE CALING UEPTO NA NA NA NA NA NA NA N	L	H								
DEFINITION PART PARTESSEE UEPT2		\dashv	¥	¥	ž	8	ž	ž	ž	ž
UEPTO		4	\$: 35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	89.53	2.2
COMBANTON PEXTENDESSEE CALLING LIEPYD NA NA NA NA NA NA NA N										
CONTROLNED PORT CUEPTO NA	+	+	¥	ž	ž	ž	ž	ž	¥	7
CONGRAVATION PACLUESCIP PORT UEPZA ELZO ELIZA ELIZA	2	_	-	;	:	;	:	;		
VEPTO	1	+	4	ž	ş	¥	ž	ž	ž	2
TERMANUL SPORT UEPYD SE 20 SE 25 SE 27 SE	\dagger	+	9	2/13	19.23	25 25	21.23	\$2.28	53.69	3
TEPABINE SWITCHEOLNED PORT	t	+	3 3		S S	8	2 2	223	89 23	ž
NEW KENTUCKY ROOM AREA CALLING UEPYE R2 20 \$1.35 \$1.79 BAX KENTUCKY ROOM AREA CALLING UEPYG NA NA NA NA STUCKY LUD AREA CALLING PORT UEPYG NA NA NA NA STUCKY LUD AREA CALLING PORT UEPYG NA NA NA NA NA NA BAX LOUISWAN LOCAL OPTOWAL UEPYG R2 20 \$1.35 \$1.79 BAX LOUISWAN LOCAL OPTOWAL UEPYG R2 20 \$1.35 \$1.79 BAX HOTELAGSPITAL ECONOMY ROOM UEPYG R2 20 \$1.35 \$1.79 BAX HOTELAGSPITAL ECONOMY ROOM UEPYG R2 20 \$1.35 \$1.79 BAX HOTELAGSPITAL ECONOMY ROOM UEPYG NA NA NA NA NA BAX MESSISSIPPLICAL ECONOMY UEPYG NA NA NA NA NA NA NA N	t	+	3 2	2 2	3 2	2 2	K 12	22.28	\$3.69	Z
PEXTENTUCKY PROM AMERICALING UEPVE	+	+	3	2	ě	8	22.12	22.28	23 69	2
PRIX KENTUCKY FOOM AVEA CALLING URPYS NA	_	_	81.38	81.78	2	25.55	27.23	2	23 63	2
### STUCKY LUD JAEA CALLING POST LEPYG NA NA NA NA NA NA NA NA NA NA NA NA NA	┝	┞						į.	8	
### PRY LOUIS WALLOCAL COPTONAL UPPYA NA NA NA NA NA NA NA NA NA NA NA NA NA	1	4	ž	ş	19.28	ž	ž	ž	ž	ž
NEW CONTRIBUTE SEE CALLING PORT	+	+	ž	ž	593	≨	¥	¥	¥	ž
NEW HOTELMOSEITAL FOOKOLY NEPTA NA NA NA NA NA NA NA	\dagger	+	ž	ž	192	ž	ž	ž	ž	¥
PRY LOUISIMAL LOCAL OPTONAL PRY LOUISIMAL LOCAL OPTONAL PRY HOTEL-HOSPITAL ECONOMY ROCM OUTGOING PEX HOTEL-HOSPITAL OUTGOING PEX HOTEL-HOSPITAL OUTGOING PEX HOTEL-HOSPITAL OUTGOING PEX HOTEL-HOSPITAL OUTGOING PEX HOTEL-HOSPITAL OUTGOING PEX HOTEL-HOSPITAL OUTGOING PEX HOTEL-HOSPITAL UEPYO NA NA NA NA NA NA NA NA NA NA NA NA NA			ž	≨	50.25	ž	≨	ž		ž
PBY HOTEL-HOSPITAL ECONOLIV NO. UEPYA R2.20 81.35 81.79			:	;						
NEPTON N	\dagger	+	ž	ž	≨	33	≨	Ş	¥	ž
PAX HOTEL-HOSPITAL ECONOMY ROOM OUTGOING PAX HOTEL-HOSPITAL OUTGOING PAX HOTEL-HOSPITAL OUTGOING PAX HOTEL-HOSPITAL OUTGOING PAX HOTEL-HOSPITAL OUTGOING PAX HOTEL-HOSPITAL OUTGOING PAX HOTEL-HOSPITAL UEPYN NA NA NA NA NA NA NA NA NA		-	\$1.36	ž	19.28	33	21.23	22 23	8	7
OUTGOING PRAYHOTELANGSPITAL OUTGOING PRAYHOTELANGSPITAL OUTGOING PRAYHOTELANGSPITAL OUTGOING PRAYHOTELANGSPITAL OUTGOING PRAYLOUSSWALDONL DUTGOING PRAYLOUSSWALDONL UEPYO NA NA NA NA NA NA NA NA NA NA NA NA NA	7	_	*	1	70 50		1	3		
OUTGOING PEX HOTELAND PORT UPPYO E2 20 51.35 51.75 OUTGOING PEX HOTELAND SPITAL UPPYO NA NA NA NA NA NA NA N	╁	╀	3		i k	S X	×	27	2	Z.
DUTGORNG PEXTLOURS ANALOCAL UEPYO	-	4	¥	ş	¥.	¥	¥	ž	¥	7
DUTGORNG PEX LOUISJAWA LOCAL DUTGORNG PEX LOUISJAWA LOCAL FRX MESSISSIPPI LOCAL CAPTONAL BAX MESSISSIPPI LOCAL CAPTONAL UEPYN MASSISSIPPI LOCAL CAPTONAL UEPYN MASSISSIPPI LOCAL CAPTONAL UEPYN MASSISSIPPI LOCAL CAPTONAL UEPYN MASSISSIPPI LOCAL CAPTONAL UEPYN MASSISSIPPI LOCAL CAPTONAL UEPYN MASSISSIPPI LOCAL CAPTONAL UEPYN MASSISSIPPI LOCAL CAPTONAL UEPYN MASSISSIPPI LOCAL CAPTONAL UEPYN MASSISSIPPI LOCAL CAPTONAL UEPYN MASSISSIPPI LOCAL CAPTONAL UEPYN MASSISSIPPI LOCAL CAPTONAL UEPYN MASSISSIPPI LOCAL CAPTONAL UEPYN MASSISSIPPI LOCAL CAPTONAL UEPYN MASSISSIPPI LOCAL CAPTONAL UEPYN MASSISSIPPI LOCAL CAPTONAL UEPYN MASSISSIPPI LOCAL CAPTONAL UEPYN MASSISSIPPI LOCAL CAPTONAL MASSISSIPPI LOCAL UEPYN MASSISSIPPI LOCAL MASSISSIP		_	- E	5	2	2	\$	20	2	313
NA NA NA NA NA NA NA NA	┝	-			4	2		87.74	8	
NA NA NA NA NA NA NA NA	+	\downarrow	ž	ž	ž	\$2.55	ž	ş	ž	٧×
NA NA NA NA NA NA NA NA			ž	≨	\$	ž	21.24	ž	ž	\$
OUTGOING PEXABLASS REST PORT UPPY NA NA NA PEX SOUTH CAROLINA APEA PLUS UPPY NA NA NA NA PEX SOUTH CAROLINA APEA PLUS UPPY NA NA NA NA PEX TEMPOSSE REGIONSERY UPPY NA NA NA NA AURPES CHE PORTI) UPPY NA NA NA NA AURPES CHE PORTI) UPPY NA NA NA NA AURPES CHE PORTI) UPPY NA NA NA NA AURPES CHE PORTI) UPPY NA NA NA NA AURPES CHE PORTI) UPPY NA NA NA NA AURPES CHE PORTI UPPY NA NA NA NA AURPES CHE PORTI UPPY NA NA NA NA AURPES CHE PORTI CHE		L	:							
NA	\dagger	+	ž	¥ 2	\$ 8	¥ 5	21.2	Y S	Y.	¥
NEW TENNESSEE REGIONSERV USPNJ NA NA NA NA NA NA NA		╀				3	* .	8	8 2	ž.
MACHE TO THE PROPERTY CALLING POPT UPPNY IN NA NA NA NA NA NA NA NA NA NA NA NA NA	4	+	ş	ž	2	¥	ž	\$	83.69	¥
MARGE GALE PER PORT)	4	_	ş	Ş	ž	¥	¥	¥	ž	z
Markes ONE PER PORTY LUBPLY NA </td <th></th> <td>-</td> <td>\$</td> <td>ž</td> <td>ž</td> <td>ž</td> <td>\$</td> <td>\$</td> <td>\$</td> <td>3</td>		-	\$	ž	ž	ž	\$	\$	\$	3
Marca Marc										
NA	+									
No. No.		Н								
2 UEPLY \$14.00 \$10.00 \$		╁	ž	ž	ž	ž	ž	\$14.18	ž	ž
3 1057X \$4224 \$2425 \$1850 4 1057X 14224 \$2425 \$1850 4 1057X 1424 143 143 4 10584 105	CLEAN	╁	81651	\$12.47	87.73	24 14	2 2	≨ \$	2 X	\$12.48
4 UEPX NA	OGO.	Н	24.25	\$19.83	\$47.78	\$49.30	\$27.63	ž	86 738	22177
NOB 8 NA NA NA NA NA NA NA NOB 8 816 25 812 26 812	AGN.	+	¥	NA	ž	ž	27.90.3	ş	ž	ž
Nos 8 \$25 51 \$18 55 512 59 Nos 8 \$275 51 \$18 56 51 26 514		H	Ş	ž	ž	ž	ž	\$16.46	≨	ž
NOME 223 1819 1812 1812 1812 1812 1813 1813 1813 1813		_			. !	:		:		
None 8444 \$25.60 \$21.62 None NA NA NA	+	+	218 23	\$12.50	NALUE!	20 818	51671	¥	\$ 5 2 5 2 5	21.00
V. V.	H	Н	09 523	79123	PVALUE!	22.53	\$29.78	\$	\$3.LC\$	15.853
	╀	╁	٤	٤	<u>S</u>	4	2	ž	ž	į
1024 23 18 00 28 22YSN	Н	2 2	\$15.00	102	800	00 E3	65.23	4.0	\$1.59	\$1.03
USAC2 50.41 53.80 50.3108	+	+	08 03	\$0.3108	800	83	3	3	9	\$0.2886
The state of the s	4	4	*10.0K	100	310.00	2	22	***	46 is	3 14

	3	4	4	§	\$	5	3	5	٥	
									,	2
MOTES										
Interim rates subject to may-up.										
CHOUR ENTITING SURFACE TO FOC rules.										
2 in Georgia, raises will apply for Currently Combined as well as not Currently										
Committed topyport combinetions unless otherwise identified.										
Ourselfor Combined contributions of troublest executions and the securing rates for	1									
the recurring rates for the U.M.S. which make up the combinations, and the incorporating rates shall be selected by the line section.	5 5									
4 End Office and Tendem Switzsing Usage and Common Transport Heave present										
The Port section of this rate entitle shed apply to all combinations of loopbort			•••							
network elements.										
		_								
Geographically Desirenged UNE Zones and applicable rates have been.							1			
•										
8	4									
	Some .									
The country. The street of the first country and account figures, estimated to be mid	PE.	_								
A. I.a. Retain for environ partition to that June 9.	200					-				
Acceleration will need be hilled. Ourse helitang acchangement on any and										
Applicable UNE Zone rates reflected to this Agreement will be believed.										
Informat Widelia http://www.intercornaction.ballacush.com/bacoma_class	-									
T a go	- Love									
	er repromertte rates									
8 I Tare to find a unique combination USOC. CLEC should autimit the loop and port USOCa.	ort USOCa				Ī		1			
British in TN and PL are interim and shall be insed-up when final rates are ordered	Ē									
Therefor it is the category of the carties to category care in the category of	- i									
the seme consistent with accidente FCC and Commission name and orders										

METHOUTHWEWBOLD METHORIC ELEMA

17. 18. 18. 18. 18. 18. 18. 18. 18. 18. 18	₹ 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	X X X X X X X X X X	32222 323
8 3 3 3 3 3 3 3 3	33355335 535535555555	22222222222222222222222222222222222222	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2
NC 81128 815.76 83.50 83 83 83 83 83 83 83 83 83 83 83 83 83	100 T T T T T T T T T T T T T T T T T T	120 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	180 180 180 180 180 180 180 181 181 181
2222222	25555555 55555555555555555555555555555	3232323333333333333	\$\$\$\$\$\$\$ \$\$
AN 13.29 8.45 8.0 5.0 5.0 5.0 5.0 5.0 5.0 5.0 5.0 5.0 5	17.75 17	NA NA NA NA NA NA NA NA NA NA NA NA NA N	ES 32 586 33 5197 57 78 780 1 580 40 5150 00
* * * * * * * * *	\$	\$	222222 22
2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	25 PA PA PA PA PA PA PA PA PA PA PA PA PA	MA MA MA MA MA MA MA MA MA MA MA MA MA M	MA
# 1	RA RA RA RA RA RA RA RA RA RA RA RA RA R	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	MA 2008.37 1 200
4 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$ \$\$	\$	222222
SOMMIN SO	LOUGH LICENS COLLAN COL	DE SOUNT SOU	A SAN A SAN
EMALAYCED EXTENDED LIMES (RELs.) INFO-2 wire BRITE(DOC)/Interface card in MFIC-2 wire BRITE(DOC)/Interface card in MFIC-2 wire BRITE(DOC)/Interface card in MFIC-2 in fearching and card in BRITE(DOC) interface card in BRITE(DOC) interface card in BRITE(DOC) interface charmal and 2 wire SCN Local Logy Combination - Member MRC-051 interface charmal and 2 wire SCN Local Logy Combination - Member MRC-051 interface charmal and 2 wire SCN Local Logy Combination - Member MRC-051 interface charmal and 2 wire SCN Local Logy Combination - Member MRC-051 interface charmal and 2 wire SCN Local Logy Combination - Member MRC-051 interface charmal and 2 wire SCN Local Logy Combination - Member MRC-051 interface charmal and 4 wire 86 logie Local Logy Combination - Member MRC-051 interface charmal and 4 wire 86 logie Local Logy Combination - Member MRC-051 interface charmal and 4 wire 86 logie Local Logy Combination - Member MRC-051 interface charmal and 4 wire 86 logie Local Logy Combination - Member MRC-051 interface charmal and 4 wire 86 logie Local Local Local Logy.	Der month Contribution - Electrophysische - Mayu Contribution - Mayu Contribution - Mayu Contribution - Mayu		155 Long per month, Zong (Pong) 156 Long per month, Zong (Pong) 156 Long per month, Zong (Pong) 156 Long per month, Zong (Pong) 156 Long per month, Zong (Pong) 156 Long per month, Zong (Pong) 156 Long per month, Zong (Pong) 156 Long per month, Zong (Pong) 156 Long per month 156 Long per m

MELL BOUTHWEWBOUTH PATES HETWORK ELEMENTS AND OTHER BETWEES

CALLERY COME (EELS)	200	7	9		3					
NRC-DS1 (resi) con 1 as		4		5	Y.	5	2	¥	ဋ	Z
	T XX	¥	Z. 22	17.23	W	2023	3			
WELCAST LOCAL LOOP - ACIT	- A 194	2	,				٤	2	¥	¥
MPC-DS1 interchine channel and DS1 (see) con Continue of	1	5	2	X	Ş	200	¥	\$421.47	ΨW	Ž
DAY CHARLES - LONG HOLL CONTROL OF THE CONTROL OF T		ž	\$3.17	3	ž	63.50	ĄN	52 53	1	****
	SOMAN	¥	25.40	2	2			2	5	2
HC-US1 Interoffice channel and DS1 Local Long Contribution . Macrael Sur Cont	77770	1				2	ž	Ş	¥	26.65
APC 051 Intercation channel and 051 (cont.)	Ŧ	5	٤	20 62	¥	823	¥	50.538	AM	¥N
AND SHARE I WANTED THE THE THE SALE OF THE	SOMAN	ž	ž	\$27.49	¥	\$153.37	A.A.	660.65		Ī
TACKS I INSTITUTE CHEMINE AND US I LOCAL LOCAL COMBINATION - MANUAL Sec Ord	SOMAN	Y.	Ž	98 99	3			3	5	ž
PICOSI interoffice channel and ISS Local (one Complessor, Married Co. Co.	ľ				٤	50	¥	ž	ž	ž
DO STIEND - LOUIS COLOR STORY - COLOR STORY	2	ž	ž	871.88	ş	86.06	¥	Ž	AM	1
	_								-	٤
DS3 Interpfice Cherried and D&3 Local Loop:		Ī								
Pecenting Charges										
۲										
the second instrumental bar mount	Z CES	≨	25.50	17 USL 3	Ą	46.50 At				
Sel Locgo pair male	41.640		ŀ			2	5	57/20	Ş	2
Managhas Changes Dardensted No. 8 - Han T.		٤		2	¥	200	ž	2	NA.	£ 15
Market Day of the Control of the Con	2	ž	\$1,121,50	\$717.60	AA	\$1.101	٩N	\$7.50 PB		
INTERIOR CHIEFTING - DAYS - Der mine per morten	11.500	¥		27 23	4				2	2
ION-Pacuring Charges - New EEL (Note 27) fore 25					5	3	ž	\$12.98	≨	88
	215	≨	U. 1715	7	1					
MC-DS3 interoffice - Adm		2			5	100	ž	Z.	ž	12.627
APC DS1 (acal loss - 1s		\$	2	Ž,	¥	. TO	×	53.55	≨	8:138
	× A	≨	8	\$761.B1	¥	\$::83	1			

100 001		•		•	3						
INHC-USA LOCAL LOOP - AOO!	7000			5		5	2	ž	ş	1	
NPC-DS3 interroting changes and DS3	Š	٤	513.74	\$545.54	¥	8673	**	1			
Control of the Contro	SOME	Y	6	5			5	Š	¥	221223	
NYK-LSS Interomoe channel and DS3 Local Loco Combination - Manual Sur On-	L		-	2	٤	\$3.50	≨	05.03	WA	S	
MRC-DS3 Interoffice channel and DS3 Local Torn Countries.	1	٤	23.40	ž	ž	ž	ž	4	1		
MIC DESIGNATION OF THE PARTY OF	- 1	≨	Ş	\$37.55	Ą	Aton Es	4		٤	S	
LACTORS MANUAL SVC ON	SOMAN	ž	¥	417 44	3		5	2	ž	ž	
LINE COST BRIDGE CHERNIS AND DIST LOSS LOND COmbination - Manual Svc Ord	SOLAN	YM.	1		5	200	٤	5690	¥	ž	
MACCOS IMPORTO CHARME and DSS Local Loop Comprision - Macual Sur One	Ŧ.		5	300	ź	88	Ş	¥	¥	1	
		٤	٤	88	¥	\$41.86	¥	ž	3	1	
STS-1 Interoffice Channel and STS-1 Local Loca-											
Recenting Charges											
11											
C.C. Town Per Communication for morals	NOT SI	¥X	20 277	6421 66	1						
and and and and	085	1	411.77		5	80.78	ž	5387.01	ž	\$400.21	
WINDOWS CHANNEL DESCRIBE - \$15-1 - Facility Termination per month	110750				٤	8	Ş	22.53	¥	500	
Principle Charmel - Dedicated - STS-1 - per man per month.	3	2	86.00	8/8/8	¥	\$1,101	¥	280085	VN.	20.00	
Not Recuring Charges - they REL ON- 2001	1	Ş	7.17	22.22	¥	214 04	¥N	96 30		3	
								200	٤	8	
Si - Concession - 18	UITES	×	20.73	17 1000							
WINCOLD INDICATES - ACC	2	3		3	ž	337	ž	\$624.06	٧X	29 1963	
MRC-515-1 Local Loco - 1g		5	201		Ş	\$404.36	¥	\$436.36	¥Z	N X	
MRC-STS-1 Local Loop - Addi		2	27	\$761.81	ž	\$611.30	ž	\$984 Q4	¥	5000	
MARC-618-1 searching channel and \$18-1 or at or at	3	٤	13.74	5545.54	≨	\$502.09	¥	C643 77	1		
MARC 818-1 Internation statement and area	2	\$	13.17	23.50	ž	5	3			3	
MPC.STS-1 Interview Agency and Sept.	SOMAN	ž	\$25.40	ž	¥Ž	· ·		2	2	2	
AND OTTO THE WAY OF THE SAME O	SOFER	≨	≨	£37 88	72	6.00	5	4	¥	8 8	
TOTAL CONTROL OF THE STATE OF T	BOMAN	ž	3	1		2	٤	20 0218	¥	ž	
FILE 9 9 1 Staroffice Chernel and STS-1 Local Locp Combination - Manual Sur	SOLAN	1		R	Ş	050018	¥	384.76	¥	ž	
NATO-513-1 Interoffice channel and 518-1 Local Local Compination - Lieunal Sec			2	2,072	ž	20.25	¥	ž	¥	4	
		5	2	278.2	¥	241.88	¥	ž	¥	¥	
D&3 Intercritos Chemiel and D&1 Local Lone	1										
Peterting Charges											
Distance more							Ī				
The state of the s	Š	≨	¥	12	13						

BOUNSEL	THEOUTH PATE
RETWORK (CELEMENTS
MO OTHER	REPAYORS
	2 2 6

F 2 2 2 2 2 2 2 2 2	2 2222222222	¥	18 19 19 19 19 19 19 19 19 19 19 19 19 19
2 2 2 2 2 2 2 2 2	\$\$\$\$\$\$\$\$\$\$ \$\$\$\$	\$ \$\$\$\$\$\$\$\$ \$ \$\$\$\$\$\$\$\$\$\$\$	22222222222222222222222222222222222222
MC TBD TBD TBD TBD NA 8720 36 81 2 96 81 84 61	201 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	N	3222222 222222222 ₂₂
2		£ 2555555 55555555555555555555555555555	**************************************
7888 X 288 X 288 X 101 101 114 101 115 20 20 20 20 20 20 20 20 20 20 20 20 20	#113.57 #404.34 #20.77 #20.77 #20.77 #12.29 #16.00 #15.00 #15.00 #15.00 #15.00 #15.00 #15.00 #15.00 #15.00 #15.00 #15.00 #15.00	# 1	817 88 817 88 817 88 817 88 817 88 817 818 817 818 817 817
2222222	33333333333	£	*****************
64 46 50 51 60 60 60 60 60 60 60 60 60 60 60 60 60	52009 52509 52509 5350 5350 5350 5350 5350 5350 5350 5	250.40 250.40 250.40 271.70 27	15 15 15 15 15 15 15 15 15 15 15 15 15 1
594.71 294.71 294.71 29.71 24.7 51.121.93 54.17 514.40	## 177 43 ## 177	25.00 20 20 20 20 20 20 20 20 20 20 20 20 2	2000 00 00 00 00 00 00 00 00 00 00 00 00
4 5 5 5 5 5 5 5 5	3333333333333	3444444	444444 44444
WELKY WELKY I LEXX X TELEVISION OF THE STATE	USE COLOR SOURCE	LEGION LOCIDIO	CEAL SOME CUTY CUTY CUTY CUTY CUTY CUTY CUTY CUTY
ENHANCED EXTENDED LINKS (EEL) OS 1 LODG per morth, Zowa 1 (Note 1) OS 1 LODG per morth, Zowa 2 (Note 1) OS 1 LODG per morth, Zowa 3 (Note 1) OS 1 LODG per morth, Zowa 3 (Note 1) OS 1 LODG per morth, Zowa 3 (Note 1) Interoffice Chernal - Dedicated - ISS 3 - Redht 7 (macrific) DS 3 (Zhernatics) (Striam per morth) DS 3 (Zhernatics) (Striam per morth) DS 3 (Zhernatics) (Striam per morth) DS 3 (Zhernatics) (Striam per morth) DS 3 (Zhernatics) (Striam per morth)	NRC, DSJ telenofice 1 (g) NRC, DSJ telenofice 1 (g) NRC, DSJ telenofice 1 (g) NRC, DSJ telenofice 1 (g) NRC, DSJ telenofice 1 (g) NRC, DSJ Conventitation System - Actf NRC, DSJ Conventitation System - Actf NRC, DSJ Conventitation System - Actf NRC, DSJ telenofice card - Actf NRC, DSJ telenofice of telenofice and DSJ Logal Logo Combination - Electronic Size Conventition of telenofice of telenofice and DSJ Logal Logo Combination - Manual Syst Conventition of telenofice and DSJ Logal Logo Combination - Manual Syst Conventition of telenofice and DSJ Logal Logo Combination - Manual Syst Conventition of telenofice and DSJ Logal Logo Combination - Manual Syst Conventi	17.5 I interorition Chemyal and D51 Local Loog: 1.000	A control Long per month, Extended A control Long per month, Extend 1 (beta 1) A control Long per month, Extend 1 (beta 1) A control Long per month, Extend 1 (beta 1) A control Long per month, Extend 1 (beta 1) A control Long per month, Extend 1 (beta 1) A control Long per month, Extend 1 (beta 1) A control Long per month, Extend 1 (beta 1) A control Long per month, Extend 1 (beta 1) A control Long per month, Extend 1 (beta 2) A control Long per month, Extend 1 (beta 2) A control Long per month, Extend 1 (beta 2) A control Long per month, Extend 1 (beta 2) A control Long per month, Extend 1 (beta 1) A control Long per month, Extend 1 (beta 1) A control Long per month, Extend 1 (beta 1) A control Long per month, Extend 1 (beta 1) A control Long per month, Extend 1 (beta 1) A control Extended Control Long 1 (beta 1) A control Extended Control Long 1 (beta 1) A control Extended Control Long 1 (beta 1) A control Extended Control Long 1 (beta 1) A control Extended Control Long 1 (beta 1) A control Extended Control Long 1 (beta 1) A control Extended Control Long 1 (beta 1) A control Extended Control Long 1 (beta 1) A control Extended Control Long 1 (beta 1) A control Extended Control Long 1 (beta 1) A control Extended Control Long 1 (beta 1) A control Extended Control Long 1 (beta 1) A control Extended Control Long 1 (beta 1) A control Extended Control Long 1 (beta 1) A control Extended Control Long 1 (beta 1) A control Long 1 (

RELIGIOTHMENTSOUTH MATT METWORK ELEMENTS AND OTHER REPAYORS

*	2012 2012 2012 2012 2012 2012 2012 2012	大学	845584 44444
8444 4444 44	*****	**************************************	84 # 12
22222222	180 180 180 180 180 180 180 180 180 180	180 117.4	80 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	*****	******	SET SEE SEE SEE SEE SEE SEE SEE SEE SEE
111111111111	2013年 2013	27.56 24.75 18.27 18.27 18.27 18.27 18.27 18.20	8 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2
*****	55555 55555555	\$\$\$\$\$\$ \$\$\$\$\$\$\$\$\$ \	885727 2873 2873 2873 2873 2873 2873 2873 2
322222222222	25.44 26.64 26.64 26.64 26.64 26.66 26.64 26.66 26.64 26.66 26.64 26.66 26.64	100 100 100 100 100 100 100 100 100 100	\$000 \$1540 \$1540 \$1540 \$1540 \$150 \$150 \$150 \$150 \$150 \$150 \$150 \$15
17 17 17 17 17 17 17 17 17 17 17 17 17 1	155.08 177.11 175.24 100.09 177.44 170.45 17	150 00 10 10 10 10 10 10 10 10 10 10 10 10	8 2 2 2 2 2 3 2 3 2 3 3 3 3 3 3 3 3 3 3
4 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3	222222 232222222 2322222 232222222	**************************************	5 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2
USOC UITYA UITYA UITYA UITYA UITYA UITYA UITYA UITYA UEALA SOMAN SOMAN	NA UNITED STATES OF STATES	MANOS MANOS	WEAL LEAL COMES CO
efon - Bec efon - Mer efon - Mer efon - Mer efon - Mer efon - Mer efon - Mer	Average to topo immercine Chemical and 4-wire 1980per Local Loop: Featuring Chemical Featuring Featuring Featuring Featuring Featuring Featuring Chemical Featuring	Recurring Charges Part P	2. Wing Analogy Votes Cares Large - Service Lares 2 2. Wing Analogy Votes Grads Large - Service Lares 2 2. Wing Analogy Votes Grads Large - Service Lares 3 2. Wing Analogy Votes Grads Large - Service Lares 4 2. Wing Analogy Votes Grads Large - Service Cares 4 2. Zares 3 2. Zares 3 2. Zares 4 2. Zares 4 2. Zares 4 2. Zares 4 2. Zares 4 2. Zares 5 2. Zares 6 2. Zares 6 2. Zares 6 2. Zares 6 2. Zares 7 2. Zares 7 2. Zares 7 2. Zares 7 2. Zares 7 2. Zares 8 2. Zares 8 2. Zares 9 2. Z

۳		
3	,	9
Ŧ	Σ	J
ξ	¥	۶
g	4	Z,
E	ũ	=
ý	¥	ě
\$	В	Ē
E	*	¢
ರ		9
3	*	3
3		
Z.		

3	ENHANCED EXTENDED (INKS (EELs)	SOC	7	7	Ϋ́	X	Γ¥	891	æ	36	Z.
	NAC-34-WIFE COMBINATION - Sentin As II' Comerson Oways - 1st	CONT	8 55	57 523	27 873 27 873	8 5	22.22	88.68	8 5	82.83	55.55
		300	8	167	\$1261	88	8	800	8	08.03	00 03
\pm	NPC, 844 WPE COMBINATION - Suich As Is' Conveyion Overgo - Discornaci - Addi	200	8	216.77	\$12.61	8	88	88	88	8	88
Ц	2-Wire Analog Voice Grade Loop - Service Level 2 - (reverse buildary)			Н	╁┼			Н			
\pm	2-Wire Analog Voice Grade Logo - Logoj Start - per mile per month. 2-Wire Analog Voice Grade Logo - Rev Bat - per month.	- CEARO	8 2	888	8 9 2	8 2 2	8 9 9	8 2	00.05	8 4	8 ×
	Zone)	UEARZ	37.98	Н	╁╅	\$17.27	\$17.65	Ħ	9	\$31.83	\$15.54
\pm	Zone 1	EAR	2 2	┿	+	X 22	X 8	+	3 2	28 25	29.53
	Zona 4	UEARZ	ž	H	1 1	ź	ž	Ħ	ž	ş	ž
\pm	NMC - Ordinarity Combined in GA (Note 5)	LEARS	Y.	\$	\$104.17	Ş	ž	×	¥	¥	X
Ħ	NPC: Addi	UEARZ	ş	ž	\$78.10	ž	ź	ž	ž	¥	ž
	NPC - Discornect Charge - 1st	EARS EARS	¥ ž	≨ ≴	¥	4 2	¥ 2	≨ ≵	≨ ≨	¥ ×	¥ ž
‡	NATC - Bectoric Syc Order, per LSR	SOMEC	≨ ≨	П	63.50	ž	¥	≨	≨	ž	ž
Ħ	NETC - Incremental Charge - Manuel Service Onter - 1st	SOMAN	≨:	П	816.94	≨:	¥:	¥	ž	ž	≨:
#	NATC - Incremental Charge - Manual Service Order - Add	SOMM	≨ ≨	Т	ž	ž ž	₹	٤ž	₹ ×	₹≨	źź
\exists	MRC-All Existing Combination "Switch As is" Convention Charge (Note 8)			П							
1	MACARA WINE COMMINATION - Suitch As Inf Omeration Origin - 1st Machine water Charles (1704) - Husen As Inf Omeration Owens - Add	200	8 = 2	2 2 2	7 8 8 8 8	5 9 5 23	2 22	3 E	3 2	22.23	2 = 2
Ħ	MIC 24-WPE COMPANTON - Tento As In Conveyton Owes - Decorned - 18	33045	8	116.77	112.61	8	8003	00.03	8	8	8
	•	33 18 18	8	\$16.77	\$12.61	8	8	88	8	8	8
\sharp	4-Whe Aratog Votes Grade Loop	11			П		-	П			
Ħ	4-Wire Analog Voice Grade Loop per mile per month	150	88	88	88	88	888	8 2	80.5	83	8 3
#	Zone 2	+	888	13.5	_		_	П	53655	2	\$19.55
Ħ	Zane 3	Н	29028	\$78.35	┪		-	7	87	8988	20 023
	Zone 4	+	≨	≨ I	¥		_	7	≨	ž	ž
\pm	AND COMMENT COMMENTS IN U.A. (NOTE 1)	UEALA	2	T	88 802	ž	¥	¥	ž	ž	ž
Ė	WPC: Add	VEAL	ž	П	\$170.57	≨	ž	≨	≨	≨	ž
Ħ	MHC - Discorrect Charge - 1st	UEAL	≨:	Т	≨ :	≨ :	ž	≨ :	≨ ŝ	≨ á	≨ 2
\pm	MRC - Decomed Charge - April -	23408	₹ ₹	T	88	\$ ≨	§ ≨	₹ ₹	ź	≨	Y.
İ	MRC - Incremental Charge - Manual Service Order - 1st	SOMAN	ž	П	≨	≨	ź	≨	≨	ž	ž
Ħ	NRC - Incremental Charge - Manual Service Order - Add	NAMOS S	≨:	≨:	27.83	ž	≨ 2	3 2	≨ \$	₹₹	ž
\pm	NRTC - Incremental Cherga - Menual Service Order - Decorned JERCAR Existing Combination "Serich As is" Conversion Cherge (Note 6)	LIVE SECTION	٤	1			5	¥			
Ħ	SPECIFICATE COMBRATION - Suits As In Consessor Overps - 1st	30 5	84.8	22.73	7 2	85.5	22	8	8 5	22.23	2 :
\pm	PRICES WITH CONSTRUCTOR - Theirs As In" Consisten Overs - And better the wate construction - Theirs As In" Consisten Overs - December 1st		800	\$16.77	\$12.61	900	88	800	8	8	8
$ \exists $	24-WPE COMBRATION - TAKEN As IN COMPUSED CHESS - DESERTED	30¥	89	816.77	\$12.61	800	8	80.03	00.00	8	8
\pm	2-Who BON Digital Grade Lego		Т					1 1	_		
H	2-Wire ISDN Loop per mile per morth	350	Т	8 4	7	8 2	8 4		_	8 9	8
#	Z-was ideal topics trade people money	XEN	П	X 22		7	\$21.15	1 1		21.40	23.54
\coprod	Zong	X	27.72	×17	282	-	28.96.3	2682	2	3 5	3 2
\pm	Zone 3 Zone 4	אכווה		ž	17	ž	ž	1 1	$\overline{}$	ž	ž
Ц	MRC - Ordinarity Combined in GA (Note 5)	20 111		,	7. 0.2		1	1	NA.	2	¥¥
1	MHC: Add	XZIIO	. ₹	ž	50.35	≨	Ş	¥	ş	ž	NA.
H	NATC - Discovnect Onerge - 1st	NIL2X	≨:	≨:	≨:	≨ :	ź	≨ s	≨ \$	¥	× ×
\pm	MFIC - Discornect Charge - Additional Charges - Add	SON	٤ž	٤ź	82	٤٤	ź	ž	¥	ž	ž
Ė	MRC - Incremental Charge - Manual Service Order - 1st	SOMAH	ž	\$:	18 8	≨:	¥:	≨:	≨:	ž	2 2
	MRC - Incremental Charge - Manual Service Order - Additional	NAMOS S	≨≨	≨ ≨	¥	≨ ≩	٤٤	§ §	ξŞ	ξ¥	ş
\exists	MRCAI Expens Combination Seven As Is Convention Charge (Note 6)			1	1	87.78	14.779	97779	W 773	× 739	554.13
1	ARCOR WITH COMBINATION - States As Inf. Concenter Charge - 14.	388	3 2 2 2 2	283	285	91 223	12.03	91.22	0.22	82	123
\coprod	MACO As of Comments	38	88	11677	112 61	88	88	88	3 3 8 8	88	88
Ī	Constitution Cardin - December	***									
	4-Wins 68 tapes Digital Grede Loop	CAS I	99	+	88	T	88	8 8	80.00	803	900
Ī	A-Web St tags Digital Graft Lang per mem.	3	ž	11	ş	П	ž	ž	29 67	V.	¥
H	Z079 I	8 2	200	8 2	3 22	≨ ≨	2 2	3 2 2 3	2 2	z s Z	2 2
\exists	Zone 2	874		1		1					

MELLBOUTHMEWBOUTH MATER METWORK ELEMENTS AND OTHER SERVICES

Н	9	ENHANCED EXTENDED LINKS (RELA)	г		1	8	×	5	878	9	S	X.
Н	+	Zone 3	99100	\$500.45	\$50.00	23	¥	\$36.48	158	<u>8</u>	\$52.76	565.73
1	+	MPC - Ordinarty Combined in QA (Note 6)			§	٤	٤	§	2	Ž	YZ.	Y.
H	Н	NRC - 1¢	25.00	ž	ž	52.0453	ž	ζZ.	٧×	ž	ž	ž
Н	Н	NATC - Addi	85 IO	ž	ž	\$241.20	¥	¥	¥	¥	ž	¥
1	+	NRC - Disconnect Dharge - 1st	25	ž	ž	¥	ž	ş	ž	ž	٧×	ž
1	+	NRC - Disconnect Charge - Addi	2	ž	ž	ž	¥	ž	ž	¥	ž	¥
±	+	MAC - EMPACHE SAG CARD DAY DOWN OF THE CARD OF THE CAR		2	2 3	2	1	¥ s	ž	Š	£ 5	Ž
t	╁	MRC - Incremental Champ - Manual Service Order - Arth	SOLLAN	2	ž	25	1	2	2	1	2	2
H	Н	NRC - Incremental Charge - Manual Service Order - Deconnect	SOMAN	≨	≨	≨	≨	¥	¥	2	≨	ž
Н	Н	MRC-All Extering Combinetion "Switch As Isr Conversion Charge (Note 6)										
1	4	MIC-24-WITE COMBINATION - "Selbch As In" Conversion Charge - 1st	30	8	123	258.43	25	\$4.23	\$54.09	254.00	854.26	\$54.13
+	+	NACES-WINE COMMINATION - SMICH AS IT COMMIND CHARGO - ART	3 2	2 2 2	10 11	8 2	2 XX	2222	2 22	27.70	0 2 2	225.17
H	Н	HPIC. SP-WME CONTRIATION - Senten As In" Conversion Charge - Discouract - Addition	NOCC	00 05	\$16.77	\$12.61	00 00	00.03	8	8	8	88
Н	+											
1	+	The second secon	(4.5)	Т	120	20.00	22.00	200	100	20.00	20.00	5
±	╀	A Mars Charle Charles are more from the comment of	19100	3 ≨	3 4	3 2	3 2	2	3 4	£32.67	3 5	2
E	╁╌	Zone 1	200		-	17821	ž	\$27.50	19 52	8	\$72	\$36.45
Е	Н	Zone (NOL64			53053	¥	\$47.25	\$33.94	780	\$45.01	\$45.07
1	4	Zone 3	200	П	╛	22.83	ž	\$36.46	548.51	180	\$52.78	\$65.75
1	+	Zong 4	300	Т	-	ž	ž	ž	2	ž	¥	ž
1	+	TOTAL ATTENDED TO THE CONTRACT OF THE CONTRACT	19.00	177	1	17.00		•	1	***	,,,	,
£	+	NRC - Arri	3	2	S Z	200	2	S Z	2 2	2	1	2
Ė	+	WRC - Decomed Dharps - 1st	19700	ž	ž	ž	₹	ž	ž	ž	ž	ž
E	H	NHC - Decomed Charge - Add1	300	٧×	ž	ž	≨	Ş	٧X	٧×	٧×	YZ.
Н	Н	NRC - Electronic Svc Order, per LSR	SOMEC	ž	ž	05.03	≨	٧	ž	ź	¥	٧×
1	4	MRC - Incremental Charge - Manual Service Order - 1st	SOMAN	ž	≨	200	ž	ž	ž	ž	Ş	ž
1	+	MRIC - Incremental Cherge - Manual Service Order - Addi	3	ž	≨:	20 65	≨	¥	≨:	ž	≨:	Ş
1	+	PRIC - Incremental Charge - Manual Service Order - Decorned	3	ž	ž	ž	ž	ž	Š	ž	¥	ď.
£	+	HER AN EXISTING CONDUMENT SWITCH AS IN CONVENCY CAMPING (1909 6)	VJ.	W 730	# 643	456	80,539	14.4	96.4.70	877	26.73	11.77
£	+	PACCE AN ARREST CONTRACTOR - CONTRACTOR CONTRACTOR CONTRACTOR - LINE	38	3 2	× 22	2 8	3 2 2	200	200	91 023	X ex	100
±	╀	PRC 26-WPR COMPRATION - Seatch As II' Consensor Oraco - Decorrect - 181	30¥5	88	\$16.77	\$1261	8 8	80 03	88	8 93	88	800
H	Н	-	SOM T	\$0.00	\$16.77	\$12.61	80.00	8000	80.08	80.00	\$0.00	00 03
Н	H											
\exists	\dashv	4-Wire DS1 Digital Loop					1	-				
1	+	+ Was DS1 Dignet Loop per mile per more?	2	B .	B :	B .	B	B	8	B	8 5	3
£	+	A WAY DO DOM (AND per ment)	3 2	24:33	98 530	2 62	200	22.00	8 53		200	5 2
Ŧ	╀	Zone 2	X SS	8	7.40	\$60.51	89.48	5.96.73	867.58	180	569.53	¥
Ė	╄	Zone 3	XX SO	\$152.29	5206.93	\$96.18	8162.34	\$197.57	95 96	8	8.2018	ž
Ė	+	Zomb 4	NS.XX	¥	¥	¥	¥	¥	\$127.47	Ž	¥	¥
Н	H	JOSC - Ordinarity Combined in QA (Non S)										
1	4	NRC - 1st	ğ	≨.	Į.	23.88	ž	Į.	≨:	¥:	<u> </u>	¥:
1	+	MMC Add	X 2	ž	2 3		2 3	¥ 2	2 3	2	S	2
±	+	March Discount Change - Arth	S S	ž	ž	ž	ž	ž	¥	2	ž	¥
±	╀	MFC - Electronic Sho Order, per LSR	SOME	2	ž	35.50	ž	į	¥	¥	¥	٧×
Н	Н	MRC - Incremental Charge - Manual Service Order - 1st	NAMOS	ž	¥	2018	¥	٧V	٧×	Ş	ž	ž
4	4	NRC - Incremental Charge - Manual Service Order - Add1	N N	ž	¥	27 68	¥:	ž	≨:	≨ :	¥:	¥:
1	┿	MATCH INCREMENTAL CHANGE - METALE SERVICE - DECORAGE		٤	٤	٤	ž	Ş	٤	٤	Т	5
1	╀	MACOSI CONSTANTON - Target As at Consensor Opera - Int	20045	28483	57.273	55.62	83	24.23	873	875	\$54.28	\$54.13
E	₽	MYCOST COMMUNICH - Selbo As In Comersion Chaps - April	CINCOC	11.23	822.75	66 92	\$32.16	12.23	\$32.16	01 223	Т	\$32.17
Н	Н	MPC-DB1-COMBENITION - Turbon As In" Conversion Overge - Discornect - 1st	200	00 03	11677	19 218	80 03	808	8	8	7	8
1	+	MRC DET COMBINATION - Tartest As In Comprise Owner - Decorrect - Add	8	8	11877	1921	8	8	BOR	8.0	7	3
±	+	Deliculate										
+	╁	One with the month.	2,500		81177	06 93	33	8538.38	564.38	53.53	\$15.53	53053
1	╀	facilities terranetion per month	UESPX	211966	\$404.58	1390 34	20.00	\$497.08	\$427.81	10 200	27.60	120021
Ė	Н	MRC - Ordhardy Combined in GA (Note 6)										
Н	Н	NPC - Facility Tembration - 1st	×	≨:	≨:	9	≨:	≨:	≨:	ž	ž	2
1	+	MPC - Facility Terrandigon - Add	X 90.94	2 2	2 2	200	2 3	Š	ž	5 5	Ş	ž
t	╁	MFC - Facility Termination - Decornect - Add	UEJPX	≨	ž	119 14	ž	¥	¥	Y.	≨	¥
Н	Н	NPC - Manual Src Order, per LSH	SOMAN	ž	ž	≨	ž	Ž:	ž	¥	\$	YY.
H	Н	NRC - Manuel Sire Order, per LSR disconnect	SOMAN	≨:	ž	¥ s	≨ :	≨ :	¥ s	ž	2 :	5 5
1	+	MAC: Electronic Syc Order, per LSR	200	2 3	\$ 3	2 3	5 3	2 2	2 2	2 3	2 3	2
土	+	MRC - Incremental Change Manual Svc Order - 1st	N N	€ ≨	€	23,123	ž	ź	¥	W.	ž	A.A.
±	+	MING - RESERVED CARREST AND COURT - 180	See	≱	¥	2,53	ž	. ₹	¥	¥N.	¥	ž
1	4	NAME: HOLD THE COURT OF THE COLUMN TO STATE O										

Attachment 2 Evital C Autes - Page 53

METWORK ELEMENTS METWORK ELEMENTS

25 25 25 25 25 25 25 25 25 25 25 25 25 2	2000 8000 874 26 8 8 8 8 8 9 0 0 0 0 0 0 0 0 0 0 0 0 0 0	\$10.00 \$15.53 \$10.53 \$497.00 \$427.01 \$597.01 \$431.32 \$400.21	*****	2222 2222 2222		2 T O O	\$41.27 \$24.69 \$11.70 \$469.66 \$611.36 \$701.71	22 23 22 23 22 23		54.23 \$4.00 \$4.00 \$4.33 \$54.13 522.4 \$23.16 \$23.10 \$22.25 50.00 \$0.00 \$0.00 \$0.00 \$0.00 80.00 \$0.00 \$0.00 \$0.00	10 85079 \$2038 11 82371 82,122 8		154.09 154.00 15 157.09 157.00 15 157.09 157.00 15 157.09 157.00	\$119.40 \$166.59 \$17000 \$47.57 \$183.50 \$2.260 \$1.73 \$1.677 \$173 \$183.50 \$773.73 \$667.00 \$500.66 \$773.40 \$170.54 NA NA NA NA NA NA NA
6.00 NA \$16.00 NA \$18.00 NA	9900	\$8.90 \$43.69 \$421.59 \$436.95	K3950 K3850 K300 K300 K300 K300 K300 K300 K300 K3	##		55,43 \$54.09 256.99 \$32.16 \$12.61 \$0.00	\$6.73 \$33.15 \$650.21 \$436.85 \$		2 × × × × × × × × × × × × × × × × × × ×	\$58.43 \$54.00 \$12.16 \$12.51 \$2.00 \$12.51 \$2.00 \$12.51 \$12.51 \$2.00 \$12.51 \$12.5	240.80 22.457 NA	118.14 NA 118.14 NA 118.14 NA 12.50 NA 13.55 NA 18.05 NA	AA 60 55 55 55 55 55 55 55 55 55 55 55 55 55	F27.25 \$166.58 \$119.45.59 \$2.750 \$2.7
N X X	50 50 50 50 50 50 50 50 50 50 50 50 50 5	\$10.08 \$11.77 \$419.06	\$ \$\$\$	2 2 2 3	2222 2222	25.03 21.13 20.03 22.73 22.73 23.73 24.73 27.73	\$7.00 \$8.80 \$1,123 \$646.60	\$ \$\$\$	22222 22222	# 22 00 00 00 00 00 00 00 00 00 00 00 00	3 81118 0 2,068	333333	14 14 14 14 14 14 14 14 14 14 14 14 14 1	5 VN VN VN S VN VN VN VN VN VN VN VN VN VN VN VN VN
OTTECT 181 SOMAN	000MN	ONE ILEMO	UCUSI UCUSI	SOMAN	SOME SOMIN SOMIN SOMIN SOMIN SOMIN SOMIN SOMIN SOMIN		GW2.11		SOME SOMAN SOMAN SOMAN SOMAN SOMAN SOMAN SOMAN SOMAN SOMAN SOMAN SOMAN SOMAN SOMAN SOMAN SOMAN SOMAN SOMAN	00000000000000000000000000000000000000	O-12.1	SOMAN SOMAN SOMAN SOMAN	2004) 2004) 2004)	ONTI
ENALANCED EXTENDED LINKS (RELs) MRC - Incremental Cost - Menual Svc. Order vs. Elect Discornact - Ist MRC - Incremental Cost - Menual Svc. Order vs. Elect Discornact - Additional Svc. Order vs. E	MCCSS COMMUNICATION STATE AND TO THE CONTROL CONTROL OF THE MCCSS COMMUNICATION - SHARE AS IF "Commission Charge - Is MCCSS COMMUNICATION - SHARE AS IF "Commission Charge - Add HOLD DOS COMMUNICATION - SHARE AS IF "Commission Charge - December - Is MCC DOS COMMUNICATION - SHARE AS IF "Commission Charge - December - Add HACK DOS COMMUNICATION - SHARE AS IF "Commission Charge - December - Add HACK DOS COMMUNICATION - SHARE AS IF "Commission Charge - December - Add HACK DOS COMMUNICATION - SHARE AS IT COMMUNICATION - DECEMBER - ADD HACK DOS COMMUNICATION - SHARE AS IT COMMUNICATION - DECEMBER - ADD HACK DOS COMMUNICATION - ADD HACK DOS COMMUNICATION - ADD HACK DOS COMMUNICATION - ADD HACK DOS COMMUNICATION - ADD HACK DOS COMMUNICATION - ADD HACK DOS COMMUNICATION - ADD HACK DOS COMMUNICATION - ADD HACK DOS COMMUNICATION - SHARE AS IT COMMISSION - ADD HACK DOS COMMUNICATION - A	519-1 Loost Loop De mile per month (activity learnington per month FPC - Ordinarity Combined to GA (Ness 8)	NRC - 815-1 - Facility Termination - 1st NRC - 515-1 - Facility Termination - Discountion - MRC - 515-1 - Facility Termination - Discountion - Indiana - NRC - 515-1 - Facility Termination - Discountion - Discounties - Addr	MPC - Marrial Sve Order, par LSR MPC - Marrial Sve Order, par LSR descrined MPC - Buctronic Sve Order, par LSR	HEV. SEGURANDE S	MC-819 (Oxeration - Seath As of Coveration Owige - Is MC-819 (Oxeration Owige - Is MC-819 (Oxeration Owige - Is MC-819 (Oxeration Owige - As MC-819 (Oxeration Owige - Oxeration Owige - Oxeration Owige - Oxeration Oxera	OC-3 Local Loop Der mit per month Led'hy kinnering per month Hed'hy kinnering per month Hed'hy kinnering comprined in GA (hose s)	HRC- OC3 - Facility Termination - 1st HRC- OC3 - Facility Termination - Acid HRC- OC3 - Facility Termination - Acid HRC- OC3 - Facility Termination - Decorated - Acid	MC - CC3 - Incremental Change - Jennal Sec Drote - 1st MC - CC3 - Incremental Change - Jennal Sec Drote - 1st MC - CC3 - Incremental Change - Jennal Sec Drote - Act MC - CC3 - Incremental Change - Jennal Sec Drote -	MO-OC-3 COMMUNION - Teach As if Constitute Chaps - 1st MC-OC-3 COMMUNION - Teach As if Constitute Chaps - Acri NPC-OC-3 COMMUNION - Teach As if Constitute Chaps - December - 1st MC-OC-3 COMMUNION - Teach As if Constitute Chaps - December - 1st	OC-12 Lecal Loop by mile per month teathy immediate per month reachy immediate per month reachy immediately Canadianed in OA (Note 5) NGC - Octobra Per Canadianed in OA (Note 5)	MMC - COL2 - Fazilly Termination - Addition	1900-18 Estimated Continues and constrained Continues and	COC-48 Lexel Loop per mile per month receipt termination per month COC-12 martinas on COC-48 Loop per month COC-12 martinas on COC-49 Loop per month COC-12 martinas on COC-49 Figure for the first on the little on
	 													

MET WORK ELEMENTS MET WORK ELEMENTS AND OTHER BERNOSS

# \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	\$252.17 \$200.00 \$000.00 \$000.00	80 X X X X X X X X	51.45 7.150 7.150 7.150 7.150 8.00 8.00 8.00 8.00 8.00 8.00 8.00 8.	╼┼┈╀┈╂┈╂╼╁╼┼╾╂╼┼╌╁╌		* 8	72 22 2 28 2 2 2 2 2 2 3 2 2
4222222222222	200 8 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	8 252222	25588 2788 8788	8 23232	25.00 25.00	10 00 00 00 00 00 00 00 00 00 00 00 00 0	27.23.23.25.25.25.25.25.25.25.25.25.25.25.25.25.
¥\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$	25.53 3 20.58 8 3.58 8	8 22222	2552 2552 8588 8588	8 3333333	8 5 8 8 5 8 8 8	NA NA NA NA NA NA NA NA NA NA NA NA NA N	23. 10 20. 23. 10 20. 20 20. 20 20. 20 20. 20 20. 20 20. 20 20 20 20 20 20 20 20 20 20 20 20 20 2
2	52 60 50 00 50 50 00 50 50 00 50 00 50 00 50 00 50 00 50 00 50 00	8 22222	\$2.56 \$2.00 \$0.00 \$0.00	8 333333	\$17.16 \$17.16 \$0.00	8	55 50 50 50 50 50 50 50 50 50 50 50 50 5
1222222222222	22.52 20.00	8 333333	22.22 20.00 20.00 21.22 22 22 22 22 22 22 22 22 22 22 22 22	8 1 2 2 2 2 2 2 2	52.2.2 50.00 80.00 00.00	\$ 222222	7 7 7 8 8 8 8
* = = = = = = = = = = = = = = = = = = =	25 88 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	8 3333333	25 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	8 2 222222	8 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	*8 222222 2 222222	27.58 2.58 8.58 8.58 8.58
68888888888888888888888888888888888888	526.43 512.61 512.61 513.91	200 200 200 200 200 200 200 200 200 200	25.00 12.00 12.00 13.00 10.00	200 St. 200 St	12.61 12.61 12.61	# 20 20 20 20 20 20 20 20 20 20 20 20 20	2 2 2 8 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2
435555555555555	522 522 756 77 76 71 86 77	8 222222	28.25 7.28.37 7.28.31 10.86.31	3 3 3 3 3 3 3 3 3	22 22 22 22 22 22 22 22 22 22 22 22 22	28 82 22 22 22 22 22 22 22 22 22 22 22 2	883 = 8 8 2 2 3
4 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	8 - 88 8 - 88 8 - 88 8 - 88	8 333333	20 00 19 15 20 00 19 15 20 00 19 15	333333	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	8 222222	2328 8:88 3
SOWNY SOWNY	UNCOC UNCOC UNCOC UNCOC UNCOC	ULDY2 EGMEN SOME SOME SOME SOME SOME SOME SOME SOME	2000NU 2000NU 2000NU	A REPORT OF THE PROPERTY OF TH	00000 NOON	ULDV4 ULDV4 SOUNN SOUNN SOUNN	20000 j
MYG. COLE INVESTIGNATION MYG. COLE INVESTIGNATION MYG. COLE INVESTIGNATION MYG. COLE INVESTIGNATION MYG. COLE INVESTIGNATION MYG. COLE INVESTIGNATION MYG. COLE INVESTIGNATION MYG. COLE INVESTIGNATION MYG. COLE INVESTIGNATION MYG. COLE INVESTIGNATION MYG. COLE INVESTIGNATION MYG. COLE INVESTIGNATION MYG. COLE INVESTIGNATION MYG. COLE INVESTIGNATION MYG. COLE INVESTIGNATION MYG. COLE INVESTIGNATION MYG. COLE IN ESTIMATED MYG. MYG. INVESTIGNATION MYG. COLE IN ESTIMATED MYG. MYG. INVESTIGNATION MYG. COLE IN ESTIMATED MYG. MYG. INVESTIGNATION MYG. COLE IN ESTIMATED MYG. MYG. INVESTIGNATION MYG. COLE IN ESTIMATED MYG. MYG. INVESTIGNATION MYG. COLE IN ESTIMATED MYG. MYG. INVESTIGNATION MYG. COLE IN ESTIMATED MYG. MYG. INVESTIGNATION MYG. COLE IN ESTIMATED MYG. MYG. INVESTIGNATION MYG. COLE IN INVESTIGNATION MYG. COLE IN INVESTIGNATION MYG. COLE IN INVESTIGNATION MYG. COLE IN INVESTIGNATION MYG. COLE IN INVESTIGNATION MYG. COLE IN INVESTIGNATION MYG. COLE IN INVESTIGNATION MYG. COLE IN INVESTIGNATION MYG. COLE IN INVESTIGNATION MYG. COLE IN INVESTIGNATION MYG. COLE IN INVESTIGNATION MYG. COLE IN INVESTIGNATION MYG. COLE IN INVESTIGNATION MYG. COLE IN INVESTIGNATION MYG. COLE INVESTIGNATION MYG. COLE IN INVESTIGNATION MYG. COLE IN INVESTIGNATION MYG. COLE IN INVESTIGNATION MYG. COLE IN INVESTIGNATION MYG. COLE IN INVESTIGNATION MYG. COLE IN INVESTIGNATION MYG. COLE IN INVESTIGNATION MYG. MYG. MYG. MYG. MYG. MYG. MYG. MYG.	Percoca constructor - Seen As ir Convento Organ - is in Percoca constructor Seen As ir Convento Organ - is in Percoca constructor Seen As ir Convento Organ - Anni Percoca constructor Seen As ir Convento Organ - Organization Anni Percoca Chemistri Anni Roca Chemistri Local Chemistri Deficialed - S. Whe VQ Moreth Perconting per north.	WRC Cutinarily Constituted in GA (News 6) WRC Sartes VOL Local Charrell - 18 WRC Sartes VOL Local Charrell - 18 WRC Sartes VOL Local Charrell - 18 WRC Sartes VOL Local Charrell - 18 WRC Sartes VOL Local Charrell - 18 WRC Sartes VOL - Incremental Charge - Marrell Ser Chitar - 18 WRC Sartes VOL - Incremental Charge - Marrell Ser Chi	MCS-84-WME COMBENTON - Sends As Y Convention Owige - 18 MPC S-4-WME COMBENTON - Sends As Y Convention Owige - 18 MPC S-54-WME COMBENTON - Sends As Y Convention Owige - Discovers - 18 MPC S-54-WME COMBENTON - Sends As Y Convention Owige - Discovers - 18 MPC S-54-WME COMBENTON - Sends As Y Convention Owige - Observed - Add Local Charmel - Dedicated - S-4MMe VG - Rev Bat. More Annual - Dedicated - S-4MMe VG - Rev Bat. More Annual - Dedicated - Sends - S	HIPC - Ordinarity Continued in 0.4 (Neas I) HIPC - Sariety Old Local Charmer - 15 HIPC - Sariety Old Local Charmer - 404 HIPC - Sariety Old Local Charmer - 404 HIPC - Sariety Old Local Charmer - 404 HIPC - Sariety Old Local Charmer - 404 HIPC - Sariety Old Local Charmer - 404 HIPC - Sariety Old Local Charmer - 404 HIPC - Sariety Old Local Charmer - 404 HIPC - Sariety Old Local Charmer - 404 HIPC - Sariety Old Local Charmer - 404 HIPC - Sariety Old Local Charmer - 404 HIPC - Sariety Old Local Charmer - 404 HIPC - Sariety Old Local Charmer - 404 HIPC - Sariety Old Local Charmer - 404 HIPC - Sariety Old Local Charmer - 404 HIPC - 804	8888	MATTER RECURS OF THE BETTONN MATTER RECURS OF THE BETTONN MICHARIN SECTION OF THE BETTONN MICHARIN SECTION OF THE BETTONN MICHARIN SECTION OF THE LEST MICHARIN SECTION OF THE LEST MICHARIN SECTION OF THE LEST MICHARIN SECTION OF THE LEST MICHARIN SECTION OF THE LEST MICHARIN SECTION OF THE MICHARIN OF THE MATTER SECTION OF THE MICHARIN SECTION OF THE MATTER SECTION OF THE	MoCas were consentron - Seats As by Commiss Owige - Ist MoCas were consentron - Seats As by Commiss Owige - Ist MoCas were consentron - Seats As by Commiss Owige - Discover: Ist MoCas were consentration - Seats As by Commiss Owige - Discover: And MoCas Were Commiss Owige - Discover: And Leon Chemis - Discover: And Commiss Owige - Discover: And Cas Mocas Owige - Discover: And Cas Mocas Owige - Discover: And Cas Mocas Owige - Discover: And Cas Mocas Owige - Discover: And Cas Mocas Owige - Discover: And Cas Ow

TAN MED	E	•
Ę	3	
1	Ž	į
9	Ē	1
Ē		

¥ 8	232222 2022 2022 2022 2022 2022 2022 20	25 25 25 25 25 25 25 25 25 25 25 25 25 2	12 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	100 100 100 100 100 100 100 100 100 100
2 8	₹₹₹₹₹ ₩₩₩ ₩₩₩₩	200 200 200 200 200 200 200 200 200 200	### ### ### ### ### ### ### ### #### ####	1 1 1 1 1 1 1 1 1 1
2 00 00	N N N N N N N N N N N N N N N N N N N	**************************************	A X X X X X X X X X X X X X X X X X X X	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2
80 08	N N N N N N N N N N N N N N N N N N N	**************************************	MA NA NA NA NA NA NA NA NA NA NA NA NA NA	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
¥0.00	**************************************	1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	# # # # # # # # # # # # # # # # # # #	25 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5
¥0.00	5 2 2 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3	25.00 25	254 00 00 00 00 00 00 00 00 00 00 00 00 00	10.12 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
8 0 00	100 100 100 100 100 100 100 100 100 100	6 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	2015 2015 2015 2015 2015 2015 2015 2015	##79 ##79 ##79 ##79 ##79 ##79 ##79 ##79
1 00 00 00	まままままま 2000 000 000 000 000 000 000 000 	1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	26.82 26.83 27.2	2002 2002 2002 2002 2003 2003 2003 2003
¥0 00	★ ★ ★ ★ ★ ★ ★ ★ ★ ★ ★ ★ ★ ★ ★ ★ ★ ★ ★	4 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	255 25 25 25 25 25 25 25 25 25 25 25 25	2012년 1212년 - 구독 독 독 독 독 독 국 국 등 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
USOC	LINGSO CHANNA SOME CHOPT CHOOSE CHANNA SOME CHOOSE CHANNA CHOOSE	CHOOSE SOME SOME SOME SOME SOME SOME SOME	CHOOSE CHARLES CONTROL	TATE TO SOURCE S
EMINACED EXTENDED LINGS (EELs) North Naturing parties per mile per mile per mile per mile per mile per mile per mile per mile per mile per mile per mile per mile per mile per mile per mile per mile per mile per mile mile mile mile mile mile mile mile	NRC: DSI LOGA Charmer National 18 NRC: DSI LOGA Charmer Add National 18 NRC: DSI LOGA Charmer Add National 18 NRC: DSI LOGA Charmer Add National 18 NRC: DSI LOGA Charmer National Charmer Add National Sto Charmer 18 NRC: DSI LOGA Charmer National Charmer Add National Sto Charmer 18 NRC: DSI LOGA Charmer National Charmer Add National Sto Charmer 18 NRC: DSI LOGA Charmer National Charmer Add National Sto Charmer Decornect Sto NRC Add National National Sto Charmer Decornect Sto NRC Add National National Sto Charmer Decornect Sto NRC-DSI COMBINATION 15 Seaton As if Convention Charmer 18 NRC: DSI COMBINATION 15 SEATON 18 NRC: DS	Local Cheminal - Dudicaled - DS3 DS3 Local Cheminal - Edith Vernicular - DS3 Local Cheminal - DS3 Local Cheminal - Edith Vernicular - DS3 Local Cheminal - Edith Vernicular - DS3 Local Cheminal Fedith Vernicular - DS3 Local Cheminal Fedith - Permination - Add MRC - DS3 Local Cheminal Fedith - Permination - Add MRC - DS3 Local Cheminal Fedith - Termination - Add MRC - DS3 Local Cheminal Fedith - DS3 Local Cheminal - Frommerical Cheminal - Manual Box Order - Dd3 NRC - DS3 Local Cheminal - Frommerical Cheminal Box Order - Dd3 NRC - DS3 Local Cheminal - Frommerical Cheminal Box Order - Dd3 NRC - DS3 Local Cheminal - Frommerical Cheminal Box Order - Dd3 NRC - DS3 Local Cheminal - Frommerical Cheminal Box Order - Decormect SO MRC- M Esterny Cortexination - Frommerical Cheminal Box Order - Decormect SO MRC- M Esterny Cortexination - Search As ar Convention Cheminal - Add MRC - DS3 COARBANT (NA - Search As ar Convention Cheminal - Decormect - 14 MRC - DS3 COARBANT (NA - Search As ar Convention Cheminal - Decormect - 14 MRC - DS3 COARBANT (NA - Search As ar Convention Cheminal - Decormect - 14 LA MRC - DS3 COARBANT (NA - Search As ar Convention Cheminal - Decormect - 14 LA MRC - DS3 COARBANT (NA - Search As ar Convention Cheminal - Decormect - 14 LA MRC - DS3 COARBANT (NA - Search As ar Convention Cheminal - Decormect - 14 LA MRC - DS3 COARBANT (NA - Search As ar Convention Cheminal - Decormect - 14 LA MRC - DS3 COARBANT (NA - Search As ar Convention Cheminal - 14 LA MRC - DS3 COARBANT (NA - Search As ar Convention Cheminal - 14 LA MRC - DS3 COARBANT (NA - Search - 14 ar Convention Cheminal - 14 LA MRC - DS3 COARBANT (NA - Search - 14 ar Convention Cheminal - 14 LA MRC - DS3 COARBANT (NA - Search - 14 ar Convention Cheminal - 14 LA MRC - DS3 COARBANT (NA - Search - 14 ar Convention Cheminal - 14 LA MRC - DS3 COARBANT (NA - Search - 14 ar Convention Cheminal - 14 LA MRC - DS3 COARBANT (NA - Search - 14 ar Convention Cheminal - 14 LA MRC - DS3 COARBANT (NA - Search - 14 ar Coarser (NA - 14 LA MRC - DS3 COARBANT (NA - Sear	STS-1 Local Charmal - Feelen Termineson on the Continues - Declared - \$152-1 Local Charmal - Feelen Termineson par month STS-1 Local Charmal - Feelen Termineson par month STS-1 Local Charmal - Feelen Termineson - 1 STS-1 Local Charmal - Feelen Termineson - 1 STS-1 Local Charmal - Feelen Termineson - 1 STS-1 Local Charmal - Feelen Termineson - 1 STS-1 Local Charmal - Feelen Termineson - 1 STS-1 Local Charmal - Feelen Termineson - 1 STS-1 Local Charmal - Feelen Termineson - 1 STS-1 Local Charmal - Feelen Termineson - 1 STS-1 Local Charmal - Feelen -	Liceal Chemes CCC 2 ptr Early Tembration Local Chemes CCC 2 ptr Failty Tembration HWG. CCC 3 ptr Failty Tembration 1 at HRG. CCC 3 ptr Failty Tembration 1 at HRG. CCC 3 ptr Failty Tembration 1 at HRG. CCC 3 ptr Failty Tembration 1 at HRG. CCC 3 ptr Failty Tembration 1 at HRG. CCC 3 ptr Failty Tembration 1 at HRG. CCC 3 ptr Failty Tembration 1 at HRG. CCC 3 ptr Failty Tembration 1 at HRG. CCC 3 ptr Failty Tembration 1 at HRG. CCC 3 ptr Failty Tembration 1 at HRG. CCC 3 ptr Failty Tembration 1 at HRG. CCC 3 ptr Failty Tembration 1 at HRG. CCC 3 ptr Failty Tembration 1 at HRG. CCC 3 ptr Failty Tembration 1 at HRG. CCC 3 ptr Failty Tembration 1 at HRG. CCC 3 ptr Failty Tembration 1 at HRG. CCC 3 ptr Failty Tembration 1 at HRG. CCC 3 ptr Failty Tembration 1 at HRG. CCC 3 ptr Failty Tembration 1 at HRG. CCC 3 ptr Failty Temp

PECUTH RAT	EL PROCENTS	-
PLLBOUTHER	METWORK	Section 1

本 	33	288 288 288 288 288 288 288 288 288	3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3
25 ₹ ₹ ₹ ₹ ₹ ₹ ₹ ₹ ₹ ₹ ₹ ₹ ₹ ₹ ₹ ₹ ₹ ₹ ₹	XQ 222222222222222222222222222222222222	T	25 2522 2 2008 2008 2008 2008 2008 2008
2	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	800 210 21 2 2 2 2 0 0 0 0 0 0 0 0 0 0 0	55 55555 8288 3888
2	┞╶╏═╏╶╏ ╌┩╌┦╌┩╌┦╌┩╌┩╌┩╌┦╌╏╏╏╏╏╏╏	888 88 88 88 88 88 88 88 88 88 88 88 88	14 14 14 14 14 14 14 14 14 14 14 14 14 1
2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	#####################################	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	22 22222 22222 22222
		8	22 22222
64 119.14 219.14 219.14 219.14 219.14 219.25		812 61 812 62 812 62 812 62 813 62 813 63 81	\$2 \$2 \$2 \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$
N N N N N N N N N N N N N N N N N N N	88 32 22 22 22 22 22 22 22 22 22 22 22 22	#####################################	25 2 2 2 2 2 2 3 3 3 3 3 3 3 3 3 3 3 3 3
4 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5			12 12 12 12 12 13 13 13 13 13 13 13 13 13 13 13 13 13
SOMMY SOMMY	TBA TBA TBA TBA TBA TBA TBA TBA TBA TBA	14200 14200	MALIN PALIN
PARACTOCIS ESTREMENT DIRANGEMENT AND THE SECONDARY OF THE	1 iii Addr. Electronic Decornaci-1 of Methods - Discornaci-1 of Methods - Discornaci-1 of Methods - Addr. Ad	HINDORFIGE CHARMANICAL STANDARY OF CONTRIBUTION CHARGE - DISCORNEY OF THE WORLD CHARGE - DISCORNEY OF THE WORLD CHARGE - DISCORNEY OF THE WORLD CHARGE - DISCORNEY OF THE WORLD CHARGE - DISCORNEY OF THE WORLD CHARGE - DISCORNEY OF THE WORLD CHARGE - DISCORNEY OF THE WORLD CHARGE - DISCORNEY OF THE WORLD CHARGE - DISCORNEY - DISCORNEY OF THE WORLD CHARGE - DISCORNEY	Intercribe Chemies - Dedicated 4-wire VG - per mis per morth Intercribe Chemies - Dedicated 4-wire VG - per mis per morth Intercribe Chemies - Dedicated 4-wire VG - per mis per morth Intercribe Chemies - Dedicated 4-wire VG - per mis per morth Intercribe Chemies - Dedicated 4-wire VG - per mis per morth Intercribe Chemies - Dedicated 4-wire VG - per mis per morth Intercribe Chemies - Dedicated 4-wire VG - per mis per morth Intercribe - Dedicated 4-wire VG - per mis per morth Intercribe - Dedicated 4-wire VG - per mis per morth Intercribe - Dedicated -

۲		
2	r	2
Ē	ğ	ÿ
₫	٩	5
Ē		ž
2	Ē	Ì
Ē	¥	δ
₫	ş	£
3	_	•

Z.	\$17.74	¥	\$ \$	≨ ≨	≨ ≨	154.13	25 -2 20 00 20 00	80 08	V . V	17.71	ž	≨≨	ź	£ £	¥.	22.5	88	3	2000	\$73 KB	Y.	ź	¥:	¥.	≨ ≨		2 2	88		19 07-03	ž	¥	£ 3	≨ ≨	4	2	88	8	25 55 27 57 55 27 57 57 57 57 57 57 57 57 57 57 57 57 57		111	
S	\$0.0373	٧×	≨ ≩	≨ ≨	≨ ≨	+	00 04 52 02	00 03	+	2003	ž	≱ ≨	ž	٤٤	YY.	22.2	88	3	+	8 8	2	5 ≨	≨ i	٤ź	≨ž	┰	H	88	++	60.40	ž	ž	£ <u>\$</u>	\$ \$	H	₩	C 80	++	E 1.2	-	5 5 5	
ž	\$17.40	¥N	≨ ≨	žž	žž	854.00	\$32.10	98 98	-	27.40	¥	≨ ≨	≨:	٤ź	\dashv	\vdash	88	11	+	\$71.28	¥	\$ 5	≨:	Ę ×	≨≨	20,730	0123	88	+-+	67.20	¥	≨ \$	£ \$	\$ \$	ž	8	60 00 10 00	8	\$12.98	_	\$ \$ \$	
SVS	\$0.0023 \$20.64	¥	≨≨	≨≨	≨ ≨	\$54.09	\$1.263	8	_	23023	ž	≨ ≨	\$ 3	٤ž	¥	91 62	88	N O	_	\$74.40	VX.	ž	≨ i	ž	≨ ≨	77.57	81 203	88	+-+	2000	¥	≨:	≨	≨ź	ž	824.09	9 00	800	20512	1 1	≨ ≨ ≨	
7	\$10.0384	¥	¥ ¥	≨≨	ŽŽ	254.23	\$32.24	90.03	10000	\$16.37	ž	≨ ≨	ži	\$ \$	٧×	27.73	88	3	50.780	07 (83	4	₹ ≱	≨:	٤ź	٤ž	100	1222	88		\$1,131,09	NA	≨:	5 5	žž	¥	277	2003	8	200		222	
KY	50 030 10 030	ž	žž	¥ ¥	ž ž	\$54.09	\$32.16	00 03	*0000	8883	ž	≨ ≨	¥	≨ ≩	¥	81 03	88	3	4600	888	V.	ž	≨ :	٤ ٤	≨≨	2	\$25 H	88		\$1,12.02	NA.	¥	£ ≨	≨≨	ž	864.08	2 S	88	32.5	П	\$ \$ \$	
ð	\$16.45	\$79.61	93.53	2 2 2 2 3 2 4 4 4 4 4 4 4 4 4 4 4 4 4 4	\$ \$	\$56.43	\$12.61	\$12.61	2000	\$16.46	\$79.61	50 SC SC SC SC SC SC SC SC SC SC SC SC SC	J 8	ž	ž	C7 955	\$12.61	10.51	W 46.27	178.47	53.69.57	11127	92	11.77	51.53	:	2 82	19.21		8717.80	4578.97	21.212	22.5	20 00	818 76	(7.15)	12 61	19:21	22		8 8 S	
	\$19.46	ž	žž	žž	źź	\$32.75	\$16.77	816.77	60,00	810 44	ž	ž ž	≨ \$	₹ ₹	ž	E as	16.7	,,,,,,,	6,449	199.79	¥	≨ ≨	≨ 3	≨	≨ ≨	1	52.23	4318		130	¥	≨ :	₹ ≨	≨≨	£	52.23	11677	11877	19.22	_	\$ \$ \$	1
¥	\$17.81	¥	≨ ≨	\$ \$	žž	\$54.03	\$0.00	80	en ma	19 213	¥	žž	≨ ≨	٤٤	ž	20 43	800	M/A	0000	23.00	M	5 ≨	≨ :	ź	≨≨	W.7.25	11 22	88 88		1200	¥¥	≨:	£ £	≨ ≨	ž	878	60 OS	00 00	8		≨ ≨≨	
DOSO	SOTIO SOTIO	UITOS	SOME	SOMAN	SOMAN	UNCCC	ON CO	33345	11.677	O:TD	ш	SOMEC	SOMAN	SOMAN	SOMAN	COOK	ONCO.	*****	41.677		IMTES	USTF1	SOLEC	SOMM	SOLAN	1			3	UTT	UITES	CTTES COLUMN	1 1		SOMAN	CINCOC	88 5 5 5	CMCCC	XX5.11		C C C C	
EMIANCED EXTENDED LINKS (RELS)	Interedition Chernal - Codicated - DSO - States Interedition Chernal - Dedicated - DSO - States - Par material per mortili Interedition Chernal - Dedicated - DSO - States - Recitivity Territoridors per mortili.	ARIC - Ordinarity Combined in GA (Note 5) ARIC - 4-ver 5640ps interofice Channel - Facility Termination - 1st	NRC - Express Src Order, per LSR	MRC - 4-wrs 56 ktps interchios Charnel - incrimental Charge - Manual Svc Ords NRC - 4-wrs 56 ktps interchios Charnel - incremental Charge - Manual Svc Ords	NATC: 4-498 50 IDAS RIBOTIOS CUBING - INCOMPANIE CARDO-MANUE SIG CITOS SON MATERIAL SIG CARD SON MATERIAL SIGNAL SON CARD SON CARDO SON	MCSHWE COMMATTON - Such As Is Constrain Dargs - IR	APIC-84 WITE COMBINATION - Tarch As Is' Consesson Dergs - Add MPC SIE-WITE COMBINATION - Switch As Is' Consesson Charge - Descript - 1st	MC, 24-WM2 COMMMITTON - Seaton As In Conversion Charge - Describes - Adril	Interoffice Cherries - Dedozing - DEG - 640pp	Integration Charmer - Dedicated - DSO - 84 ktops - Facility Termination per month IMMC - Ordinarity Combined in GA Michael Si	MFC - 4-wine 644dps Interoffice Channel - Facility Termination - 18.	NRC - 4-win 64 topp interprite Charma - Facety Terranego - Add 1 NRC - Bectraric Svc Order, per LSA	NAC - 4-wire 64 More Interoffice Chemis - Incremental Cherge - Manual Svc Onde	MFQ - 4-wire 64 logot interoffice Chemie - incremental Chenge - Manual Svc Outs	ANTC - 4-wire 64 ltgs interoffice Channel - Ingrenental Charge - Manuel Sirg Onte PRTC-AR Extering Combination *Switch As Is* Conversion Charge (Note 6)	ANC BE WINE COMMENTON - Sents As in Comments Owen - 1st here as were commentation - Sents As in Comment Owen - Anti-			Internative Change - Dedicated - DS 1	Intervition Chemis - Dedicated - DS1 - Facility Termination per month	MRC - Ordinarily Combined in GA (Mote 5)	arrel - Facility Terrainment - Add	er, per 158	MRC - DS1 interface Charma - incomental Charge-Manual Src Order - Add	MYC - DS1 interoffice Channel - Incremental Charge Manuel Svc Order - Decora	PRICAN Existing Combination "Switch As In" Conversion Charge (Note 6)	MPC-051 COMBINATION - Select As at Convention Charge - Acri	NEC DS1 COMBRATION - Switch As is Conversion Clarge - Discorned - Ast NHC, DS1 COMBRATION - Switch As is Conversion Clarge - Discorned - Ast	Interoffice Channel - Dedicated - DB3 - per rulls per month	Interchice Chemistry Described - USQ - Facility Termination per month	MRC - Definantly Combined in GA (Note 5) MRC - DS3 Interoffice Channel - Facility Termination - 1st	4	MRC - Estatoria Sve Crow per Libra MRC - DS3 Interprites Charrel - Incremental Charge - Manual Sve Order - 1st	MRC - DS3 Interoffice Chemiel - Incremental Charge - Manual Svc Order - Addition - MRC - DS3 Interoffice Chemiel - Incremental Charge - Manual Svc Order - Discontinuation	INTO - DCS Interplice Charmel - Incompanie Charge - Manuel Sec Order - Discon-	HRC-053 COMBRATION - Switch As Int Convenion Charge - 14	NRC-053 COMBNATION - "Switch As In" Convenion Charge - Add NRC-053 COMBINATION - "Switch As In" Convenion Charge - Disconnect - (III)	NETC. D&3 CONDINATION - Switch As it: Convention Chargs - Decorring - Ad-	Interoffice Channel - Dedoated - 515-1 Interoffice Channel - Dedoated - 515-1 - pe mile per month	WEIC - Ordinarity Combined in GA (Note 8)	NRC - \$1'\$-1 transfiles Channel - Facility Termination - 1st MRC - 515-1 transfiles Channel - Facility Termination - Add MRC - Facility Sec Onter Are 150	The second secon

П	Т	Т	П	Т	П	Т	П	Т	Т		П	Т	Т	П	Т	Т	П	T	Т	П	П	Т	П	Т	T	1	Т	П	7	Т	П		Т		П	Т	П	17	Т		Т	Т	11	\top	77	$\overline{}$	T	_	11	7	т	11	1
ž	≨.	€ 2	ž	554.13	\$32.17	8 8 8		41.1.4	25.23		≨	ž	5 ≥	₹	≨ s	٤ž	ž	1 739	223	88 88		-	\$1015	-	≨ 3	2	ž	ž	1	₹ ≨	≨	1	2 2	8 8 8		\$106.66	203118	2111	23	ź	2	ž	≨:	٤ž	¥	≨ ≨	ž	≨ ≨	≨:	YY	22.5	88	
೫	≨.	§ ≨	ž	82.53	23.23	3 3 8 8 8		26 36	28.3		≨:	≨ \$	€ ≨	ž	≨	ž	¥	× 73	22.23	8 8 8			\$11,132		≨ ;	2 3	ź	٧N	≨¦;	ž	¥	96 7 36	8 25	00.08			\$967.58		≨:	Š	ž	2 3	¥:	≨ ≨	ź	≨≨	ž	≨ ₹	≨:	V	8 X	88	
¥	≨:	Ž	¥	00 753	\$32.10	88 88		91.713	1,0,21		ž	≨	ž	¥	٤ź	ź	ž	60 Tab	\$32.10	3 3 3 8 8		530.38	\$7,676.00		≨ :	§ ž	¥	¥	≨ 5	≨ ≩	ž	664.00	2 22	00 03		20021	\$10,862		≨:	S ≨	≨:	٤٤	≨:	≨≨	ž	≨≨	ž	≨ ₹	ž:	V.	85.55	88	
SII	ź	ž	٧×	60 753	\$32.16	3 3 8 8		616 36	\$1,892.00		¥	≨ \$	ş	ž	¥ ×	٤٤	ž	6073	\$22.16	888		\$60.42	17,182.00		≨ 5		MA	NA.	≨ \$	≨ ≨	ž	664.70	2 2 2 2 3 3	00 03		\$102.43	\$11,480	******	¥	≨	¥.	§ ≨	¥	≨≨	ź	¥ ž	ź	ž	≨:	Ş	8 5 2 2 2 3 3 3 3	88	
4	≨:	5 ₹	ž	\$54.23	\$22.24	8 8 8 8		631 80	2,390		¥	ž	ž	ş	ž	ž	¥	273	12.23	000		\$74.44	\$11,517		ž	2 3	ž	¥	\$ \$	\$ ≥	ž	60.22	223	00.03		_	056713	_	\$	¥	≨:	≨ ≨	≨:	≨ ≨	ź	≨ ≨	ž	¥	ž	Ş	22.53	88	
Ş	¥:	ź	¥	854.08	\$32.16	3 3 8 8		60.00	330		≨:	≨ ≨	≨	ž	ž ž	§ §	ž	65.400	25.25	0000		284.88	\$12,344		ž	1	¥	ž	2 2	\$	ž	964.00	2	00 03		\$138.02	116,017		≨ :	₹	≨:	٤٤	≨:	≨ ≨	₹	≨ ≨	ź	ž	≨:	ž	88.65	88	
3	61.19	7183	1103	\$58.43	556.98	\$12.61		473	\$2 187.00		247.63	32.5	\$118	9	2,5	\$1800	81800	64.43	88	\$12.61		\$15.06	\$8,202.00		B	203	\$119.14	8	25.65	88	818.83	67 639	8	612 61 812 61		2.23	00 11 13		81,004.00	\$6.00	317.36	21914	112.31	\$ 9 6	2 /23	23.63	8001	88	8	BBILE	C7 655	19213	
1	≨:	ž	¥	\$32.75	332.78	\$16.77		55	53,043		≱:	ž	ž	ž	ž	ź	ž	ET 9.75	132.73	44.818		16.92	\$11,685		≨≨	2 3	≨	≨	≨ ≨	. ≥	≨	26.00	2,23	42 918			\$12.564		\$ 5	2	≨:	₹ ≨	≨:	ž ž	ź	≨ ≨	ž	≨ 5	ž	ž	22.23		
77	≨ :	ž	Υ¥	875	11.23	888		£7.78	22.475		≨:	2 3	ž	≨:	§ §	Ž	ž	W 733	11.23	00:05 80:00		\$19.26	\$9,763	***	ž	ž	ž	≨	ž	¥	≨	873	3	00 00 00 00 00 00		20 66	100111		ž	ž	≨	€ ≨	≨:	≨ ≨	€	≨ ≨	¥	≨ 5	ž	§	154 CB	88	
CROC	SOM	SOMM	SOMAN	UNCCC	30¥5	UNCCC		11 6372	TBA		¥ i	S A	TBA	SOMEC	3	SOMAN	SOMAN	333041	300	DOOMN DOOMN		XSJ.	TBA		4	V V	TBA	SOMEC	S	SOMAN	NAMOS	3	30	330H1		15XX	YEL YE		A S	TBA	40	<u> </u>	¥.	SOME?	SOMAN	NAMOS	MYNOS	SOMAN	SOLLAN	SONOS	COOM	88 3 3 3	
MANCED EXTENDED LINKS (EELA)	CHICASI VI I IMPRODUCE CIRCUM I INCREMENTA CANCOL SINCE OVO CITOR - 18	APC - \$15-1 Intercribe Channel - Incremental Change - Manual Svc Order - Disco	MRC - \$13-1 Interchica Charvet - Incremental Charge - Manual Svc Order - Disco	PERCAN Extending Combination "Switch As Is" Conversion Charge (Pote II) INFIC-\$15-1 COMBINATION - "Switch As Is" Conversion Charge - 1st	NRIC-STS-1 COMBINATION - SWICH As In Convention Charge - Addi	HHIC STS-LOWBINATION - Swith As it Conversion Charge - Decornect - I NHIC STS-LOWBINATION - Swith As it Conversion Charge - Disconnect - A			Interoffice Cremel - OC3 - per facility Terrelation	MRC - Ordinarity Combined in GA (Note 6)	MRC OCC Facility Termination - 1st	MAC - OCS - Facility Temporals - Ocs - 14	NRC - OC3 - Facility Termination - Disconnect - Addi	MRC - Electronic Sinc Order, per LSR	MEC. OCS. Incoments Change Manual Sec Order. Add.	MRC - OCC - Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-1st	MRC - QC3 -Incremental Coer - Manuel Svc. Order vs. Elect-Deportract-Addi	MMC-ARE EXIMINE CONCRETION - SWITCH AR IN: CORVERTION CHAINS (YOUR BY MICCOCA COMBRIATION - Switch As In' Corvenion Chains - 1st	MRC-OC-3 COMBRIATION - Switch As Ist Convention Cherge - April	NAC, OC-3 COMBINATION - Switch As III" Convention Charge - Discornact - La NFC, OC-3 COMBINATION - Switch As III" Convention Charge - Discornact - Ac	Marcellos Channel - OC12	marchice Channel - DC12 - per Mile	Interoffice Charmel - OC12 - per Pacifity Terratration	MRC - Ordinarily Combined in GA (Note 6)	MPC - CC12 - Facility 1 emination - 16	WC - OC12 - Facility Territorion - Decorporat - 1st	NFIQ - OC12 - Facility Termination - Discompct - Additi	NAC - Electronic 9vc Order, per LSR	MFC - OC12 - Incremental Chema - Manual Sec Order - 14 MFC - OC12 - Incremental Chema - Manual Suc Order - Addi	NPC - OC12 - Incremental Cost-Manual Svc. Order vs. Elect-Decornact-1st	ARC-0012 - Ingramental Cost-Manual Svc. Order vs. Elect-Discormact-Addi	MRC-All Existing Combination "Switch As is" Conversion Charge (Note 6) MCC-CC-13 Charlina Title "Suites As is Conversion Characters"	HRICOC-12 COMBRIATION - Tamon As In Convention Charge - Addi	1990-00-12 COMBINATION - Switch As is" Conversion Charge - Discorract - 1 1980-00-12 COMBINATION - Switch As is" Conversion Charge - Discorract - 1		Intercenting Charmai - OCAB - gar Mile	Interoffice Chernel - OCAB - per Facility Termination	MRC - Ordinarity Combined in GA (Note 5)	NRC-Ocki-Fadiby Termination - 14	NRC - OCUB - Immriage OC12 on OC48 - 1st	NPC - OCAB - Interface OC12 on OCAB - Addit	AFIC: OC48 - Facility Termination - Disconnect - Addi	WTC - OCAB Intentions OC12 on OC48 - Decormect - 1st	NHIC - OCAR - IMMITIGO OCIZ on OCAR - Disconnict - Addi NHIC - Flactionic Sec Order, par LSR	MRC - OCA6 - Facility Termination-Manual Svc Order vs Electronic-Decornact-1	NRC - OC48 - Facility Terminason-Manual Svc Order vs Electronic-Discornact-A NRC - OC48 - Interface - Manual Svc Order vs Electronic-Discornact-19	NRC - OCAR - Interface - Manual Svg Order vs Electronic-Discornect-Addit	NAC - OC-48 - Incremental Charge - Manual Syc Order 18	1870 - OC46 - Intertace OC12 on OC46 - Incremental Charge - Manual Svc Order	MPC - OCAS - Mentace OC12 on OCAS - Incremental Charge - Manual Svc Cross MPC-All Enteding Combination * Serfich As Is* Convension Charge (Note 6)	NRCOC-48 COMBINATION - Switch As In: Conversion Charge - 18	HAC OC 46 COMBINATION - SWIZT AS IS CONFIRM CHARGE LANGUAGE - I HAC OC 46 COMBINATION - SWIZT AS IS CONFIRM CHARGE - DISCOURCE - I HAC OC 46 COMBINATION - SWIZT AS IS CONFIRM CHARGE CHARGE - I	

EMANNED EXTENDED LINGS (PELS)	2085	7	F.	Ą	KY	3	S#1	S.	သွ	×
Chernelization:										
DSS Chemistred System on month								H		
DS3 Interface per month (DS1 COC)	NC ID	\$17.22	\$14.51	\$11.00	28.52	\$25.64 \$7.55	85.83 85.88	\$228	100001	\$222.96
NerC - DS) Chemetration - List	\$							Н		2
NRC - DS3 Chemetzeton - Add1	8	ž	ž	213:23	≨ ≨	≨ \$	≨:	≨:	≨:	ž
MRC-Cheme Advance (its	r Ci	≨	ş	\$1202	ź	٤ž	ž	≨ ≥	2 2	≨ ≥
NAC - Electronic Sec Order, par 159		≨ :	ž	98	≱.	₹	¥	ž	≨	ž
MRC - DS3 Chemekration - Incremental Charge - Manual Swc Order - 1st	SOMA	\$ ≨	ž	20.50	≨≨	≨ 3	≨:	≨:	≨.	ž
AUC. DC3 Chamelzation - Incremental Charge - Manual Sec Cross - Actri	NAMO3	≨	ž	\$38.43	ž	5 ≥	ž	2 3	ž	≨ş
MRC - 083 Chemelaston - Incremental Change - Mariai Sec Order - Decornect	W 5	≨ :	≨:	60023	ž	≨	≨	ž	V.	ž
NRC-All Exteting Combinedon "Buttch As is." Convention Charge (Note 5)			٤	90.00	ž	≨	ž	ž	٧	¥
HARCASASTS COMBINATION - SAMET AS IN Conventon Charge - 1st	SON)	848	57 CT 25	55.63	\$54.09	273	88.75	88488	864.28	2
MATCH DENSITY - COMMANATION - South As IN CONTRACT COMMAN	200	- E	23	88 82	\$32.16	\$32.24	\$32.16	622.10	273	2 2
MRC DSASTS-1 COMBINATION - SMICH As IF CONVESION CHAIRS - Discuss	18	88	1 2 2	\$12.61	88	8	000	8	00:03	800
De Characteristic					3	MA	3	B R	8	8
OSI Chemetral System nor month.	1									
OCU-OP drop imertace card per month (2.4-446m)	20.0	2000	7 7 7	8128.22	Б 00.23	1209.67	29 9+1\$	\$17.72	\$147.51	\$166.21
VG interface card per month.	101/0	20,000	4 4	8	X S	2 2	8	8	۲. 2	22 46
2-who ISON/BRITE card per month	Z Z	1773	22 22	12.71	3	2	2 2	2 2		2 2 2
MAC - CAMPANET COMMUNICATION (NOTE OF										2
MFC · OS1 Chamatization · Aeri		≨ :	ž	89 90 200 200 200 200 200 200 200 200 200 2	¥	Ş	¥	ž	ž	ž
MHC - Chemiel Activation VG - 14	67.0	ž	X	\$137.08	ž	Ž.	ş	Y.	Ş	ž
MRC - Chemnel Activation VG - Addr	DIVO	2	2	22.53	ž	≨ :	≨ :	ž	≨	ž
MRC - Chernel Activation OQL-OP- 1gt	10100	≨	ž	\$12.00	ž	2	≦ 3	≨ :	≨:	≨:
MRC - Chemel Activation OCU-DP - Addr	10100	¥	Ş	25	2	2	1	2 3	2	2
AND COMME ACTVAINS BATTE - 1st	S S	ž	¥	\$12.02	¥X	ž	\$	2	É	2 2
MAC. For heart for four and 60	<u>ა</u>	≨	¥	88	¥	ž	ž	ž	¥	1
1	_1	≨:	ž	05:03	٧V	¥	≨	٧Z	2	ž
MMC - DS1 Cherneltz aton - incremental Chernel-Manual Snc Christ - Arth		ž	2	27.73	≨:	ž	≨	٧×	٧V	٧×
MRC - DS1 Channelt aton - Incremental Charge - Manual Svc Order - Discorner	L	Y.	2 2	200		ž	ž	ž	ž	ž
MRC - DS1 Chemetration - Incremental Charge - Manual Svc Order - Discornect	1	Ş	₹	88	\$ \$	Ž	2 2	ž	¥	∑
MMC-All Existing Combinetion Switch As is "Conversion Charge (Note 6)	ı						5	٤	Š	Ş
WRCOSI COMBINATION - WASHING COMPANIES OF THE	88	8	22.72	\$58.43	8	134.23	87.09	\$64.00	82.28	54.13
NRC DS1 COMBRATION - Switch As In Convenien Chans - Decorrect - Le		E S	<u> </u>	8	2 2	12.23	\$1.20	61.50	52.23	132.17
JARC-DS1 COMBINATION - Switch As In' Convenion Charge - Decorned - Add		800	816.77	\$12.61	8	38	88	88	88	88
							2	3	3	3
To the										
OSI DSC Territorion with DS1 Rustrian		2	200	22.8	90	TBO	190	TBO	780	180
LOSS DSC Termination with DS1 Switching		3 2	277	2	2 5	2	2	90	TBO	g Q
MRC - Ordinarity Combined in QA:				871618		2	2	28	200	9
MFC - Outomer Configuration Establishment		2	16.53	16.23	180	081	180	9	180	Call Control
MACCOST DECEMBER OF THE PERSON		2	N C	9E C3	9	08	OE.	OBT	180	8
NRIC DS1 DSC Termination with DS0 Selection - Arth		3 6	2 2	625	2	0	2	OB.	180	TBO
NRC-D\$1 D\$C Termination with D\$0 Swarping - Discornact - 1st		2	100	2 8	2 2	2 2	2 6			9
MRC-DSI DSC Territorion with DSD Switching - Decorrect - Actril		2	24.86	20.16	8	2	2	2	200	2 2
MACCOST USC Temperation with MACCOST Selection - 1st		8	22/0	\$18.07	780	<u>1</u>	130	98	98	2
PERC. DS1 DSC Termination with NPC-DS1 Switching - Decorated - (at	T		× =	21.48	2	2	5	8	90	9
MRC- DS1 DSC Termination with MRC- DS1 Switching - Disconnect - Add7		08	£ 9	25.0	28	8	3 2	3 5	8 8	8
MACC OST DSC Termination with DS1 Switching - 18		æ	05.150	\$32.07	Ę	2	180	8	9	3 2
MRC OSJ DSC Termination with DS1 Seatthing - Discourses - 14	1		3 8	\$ 12	8	8	1BO	380	180	8
MRC- DSJ DSC Termination with DS I Switching - Disconnect - Addr		2 2	3 22	200	3 2	2 2	2 2	8 8	2 5	9
MRC-All Existing Combination "Switch As is" Conversion Charge (Note 6)		П						2	1	
MACCOS COMBINATION - Santa As In Convenion Change - 1st	8	8	22.22	55.623	88.83	275	84.8	82.00	27.53	5 2
MRC-DCS COMBINATION - Serton As in Convestion Change - Decorpact - 1st	300	+	24.7	8 2	= E	72.23	2 2 2	0.23	200	11.22
NATC- DC\$ COMBRACTION - Switch As Ist Convention Charge - Discornact - Act UK	33 ₹	++	316.77	\$12.61	88	88	88	88	38	88
Node (Gradinones)	T	T				1			+	
Node per month	LYCYL	\$15.77	\$16.35	\$13.96	\$18 11	\$1543	\$16.15	818.00	\$14.55	11,618
MMC - Ordensory Companied in GA:	27.00	12	1						Н	\prod
	T INCH	٤	ž	547 19 T	\$	¥	≨	¥	¥	¥

Š

METWORK ELEMENTS METWORK ELEMENTS AND OTHER SERVICES

4	ENHANCED EXTENDED LINKS (EELs)	OSO	7		₹9	A		37	5		
	NPC - Node - Adri	1500	VA.	1			5 3		2	ړ	E
	INRC - Node - Decornect - 1st	LON	2	1		¥ 3	¥ :	<u> </u>	ž	≨:	ž
	MR C - Node - Discornact - Addi	150	2 2	4	20.00	≨ :	٤ ا	≨ :	Ş	ž	≨
	MRC - Mode - Incremental Change - Manual Sawins Onter - 14		2	5	R	ž	ž	ş	¥	ž	٧¥
Г	NPC - Node - Incremental Change - Manual Sevice Orde - Add		\	5	2	ž:	≨	ş	ž	¥	ž
	MRC - Node - Incremental Otanos - Manuel Sarvice Order - Dismonent - 144	NAME OF THE PERSON	2	2	2	٤	ž	ž	ž	ž	ş
	NRC - Node - Incremental Charge - Manual Service Order - Discomment - Artific	NAMACS		5 3	2	2	2	Ş	ž	¥	ž
	MRC-All Establing Combination "Switch As is" Conversion Charge (Note 6)			5	٤	٤	Ę	٤	٤	ž	ž
	MRC-Node - Switch As In" Convention Charge - 1st	TINCOS:	m 73	X COL	170	90,190	1				
	MRC-Mode - Switch As Is' Convention Charge - Add	300	3 8	2 2	2 8	3	22.5	8	824.80	2	2
	MRC Node - Switch As In Convenion Chans - Decoment - List		8	2 2 2 2	2	2 2 2	2	572.16	23.10	823	123
	MPIC: Node - Switch As In" Convention Charge - Discorract - Additional Charges - Discorract - Additional Cha	COOK	88	14.8	212 61	38	38	88	88	8 8	8
						3	3	3	3	8	8
	Opidonal Features & Functione:		Ī				Ī		1		Ī
7	NRC - Clear Charmel Capability (B0ZS/ESF) Opton - Subsequent - per DS1 Charl	1000	3.24.65	21.64.52	\$184.62	\$184.91	\$184 65	\$184.60	6164 76	20,000	
	NRC - Cheer Chernel Capability (BIZZAESF) Option - Subsequent - per DS1 Charl		1972	263	82 78	ara	200	2 160	3	2 2	100.10
J	NRC - Clear Chernel Capability (B&ZS/ESF) Opton - Subsequent - per DS1 Char	2000 000 000 000 000 000 000 000 000 00	8	20.03	8	9	61 07		8 8	8 8	8 2
╛	MRC - Clear Chernal Capability (B82S/ESF) Option - Subsequent - per DS1 Cher		207	90	82.03	1 S	12.5	1	2	200	3 5
	NRC - Clear Charmel Capability (B82S/ESF) Cotton - Subsequent - Manual Service	SOME	282	27.50	1163	76 863	800	200	200		200
ゴ	MMC - Cheer Chermal Capacishy (B625/ESF) Opton - Subsequent - Manual Service	SOLLAN	23.83	1903	28 03	2	23.02	1813	1815	20.5	3 5
1											
┪	MRC - Oser Cherrist Capability (8625/SF) Opton - Subsequent - per DS1 Cherry		2 2 3	\$191.50	\$184.62	\$184.91	\$184.6K	S184 60	616476	70, 2010	
1	MRC - Clear Charmal Capability (B&ZS/ESF) Opton - Subsequent - per DS1 Char		1963	20.02	87.52	23.62	23.70	2 22	1	1	423.10
┪	MPC - Chear Cherriel Capachilly (B&ZS/SF) Option - Subsequent - per (251 Cherri	1808	8	2003	wa	8	1013		3	8 2	8 8
	MRC - Clear Channel Capability (B625/ESF) Option - Subsequent - per DS1 Charl		120	8	R	5	1	R	2 5	2 2	81
Н	MRC - Clear Chemist Capability (8825/55) Option - Subsequent - Manual Service	SOMAN	282	Fig			2	2/2		2	2
۲	MPC - Clear Charmet Canadally (BAZS/FSF) Ontion - Supergrave - Manual Com-		200		2 2 2		2	23.23	F. S.	я 8	2333
т			2	à	2	K i	2	2	22.22	80.03	8
⇈	0,000	Ì	Ī		Ī					1	
Н	White the subject to true-up.		Ī		Ī			1		1	
_	1 Geographically Desirenged UNE Zones and applicable rates have been	Ì								1	
-	established for certain strivides, as shown in this Agreement. Where										_
	Geographically Deeveraged UNE Zones and applicable rates are established.										
_	Statewide rates are obsolete. Further, Beliffouth is in the process of enhancing	_	_								
	is billing systems in order to accomodate this Geographically Desveraged UNE					_					
_	Zona Pale Shuctura. Until these entiencements are accomplished, settinged to		_	_		_		_	_		
_	Do mit ZUVI, the UNE Zone 1 rate will be billed for all services residing in Zones					_					_
_	11, 2, 3 of 4, 1.8., Rates for services residing in LINE Zones 2, 3 and UNE Zone 4,						_	_			-
	where apparatus, the rick be bried. Once balling emissionments are complete, as		_	,	_						-
_	SATISFACTOR CONT. CATALONIC IN THE AGREEMENT WILL BE DRIED. HERETED	-					-	_			
_	A THE PARTY OF THE			•					_		
-	Zone Designations by Cartest Office.	_	-		_		-,-			_	_
۰	Now EELs will only be available in the State of Georgia and in density. Zone I of the believes lift he have Read-one Engineer	Mary LAS	Ac in the Res	1	1					+	
Н	Florida - Marri, Orlando, Pt. Lauderdale		-				İ	İ	\dagger		
+	Louddans - New Orleans										Ī
+	M. CARONNE - Understoom, Chartotta								-		
Ť											
+	And the state of t			1	1						
+			y recurrence	2	1	1		+			
╀	- 2-see VG Loco + Votce Grade Interfers Card + DS1 Chemistration Seems + D	1		\dagger	1		1	1	1	1	1
┥	- DS1 Loop + DS1 Interface Card + DS3 Channel Latent Season + DS3 Internation Channel) Jenne		\dagger	ľ			1		1	
Н	DS3 Local Chemie + DS3 Interoffice Chemiet + DS3 Chemietzation System + D	St Interface	2		ĺ		T	t	Ì		Ī
Η	The Ordherfy Contitred in GA MRC agains to new combinations within the State of	Georgia			İ		<u> </u>				
H	6 The "Switch As Is" MPC is a conversion charge. One \$44 charge is applicable per circuit.	200				$\frac{1}{2}$		ļ		+	
_		-								\dagger	T
	Where the eath Commission has adopted rates for the rate elements contained	•					_				
-	Apply the same consistent with applicable FCC and Commission rules and orders.									•	_
!]

BELLBOUTHNEWBOUTH MATES HETWONE ELEMBITS AND OTHER BEWACES

Aracionara 2 Espesi C Falts - Page 84

DESCRIPTION	00 S n	7	Z	ΥS	¥	3	2	2	2	Z
OSS costs, per CLP, per month		٧¥		ž	¥	ž		\$305.00	ž	L.
HC - USS OLEC Daily Usage File: Recording, Per Message HC OSS OLEC Daily Usage File: Message Processing, Per Message		2000	8 8 8 9	\$0.0001275 \$0.0082546	0 0032357	\$0,000 ts	50 0001179	50000	\$0,0002862	900 03
RC - OSS OLEC DAIN Usage File: Message Distributor, Per Magnetic Tape		\$55.19	П	29.00	\$55.60	\$47,3000	254.62	19 451	24 75	1_1
Access Dally Usege File (ADUF)		50000	3		SO COCOCO	an manna	ac morning	100000	A DUCCUS	3
RC - ADUP, Message Processing, per message		100.03	П	20361003	100 03	-00 OS	ш	700 OS	100.00	700 OS
RC - ADUP, Data Transmision (COMMECT. DIRECT), per message		10,001	100 00	\$6 0000434	100.001	\$0.000000	100.03	100 OS	100.01	1000
Enhanced Optional Daily (bage File (ECOUF)		100								
Enfanced Optional Daily Usage File Message Processing per magnitic tape Figure Considering Control Con		ž ž	2 2	S S S S S S S S S S S S S S S S S S S	8	8 2 2	NA N	8 ≥	8 2 2	7 5
		٤	77	ž	Š	ž	Š	ď.	٤	ž
DAY EXT. THE TWO THEN THE DAY DESCRIPTION OF THE STATE OF	1	SO PONK	EN PROCESS	ES MALANA	12	EA MONTAGE	£0 000 131	th Amen	40 AUGUS 23 3	60 0006 203 En 0006 603
BXX Access Ten Digit Screening Svc. WilkX No. Delivery	Ц							2000	2	26 1000
for 0XX Numbers with Optional Complex Feetunes on curry	YAY X	≱ ≱	≨≱	\$ \$	9 9 9	≨≨	ž	3900003	≱≱	≨ 2
BIXI Access Yen Digit Screening Svs. W/POTS No. Delivery										
per quary field Optional Complex Features, per query	YN YN	≨ ≨	≨≨	≨≨	01000	≱ź	≨≨	\$0 00383 \$0 00431	≨ ₹	¥₹
AXX Access You Digit Screening Svc. W/800 No. Delivery										
for 800 Numbers, w/Ostonal Complex Feetures, per message	YN YN	\$ \$	≨ ≨	2 2	≨ ≨	≱ ₹	žž	≱≱	žž	≨ 2
BXX Access Ten Digit Screening Svc. WIPOTS No. Delivery										
twith Ontional Complex Fastures, one measure	YY	≱ ≵	\$	ž	≨ ≩	3 3	ž	¥2 2	≱ ‡	≨ 3
Reservation Charge par 833 number reserved					5	1	5	52	5	٤
7.0.12	XBBIX	11.13	2	íš X	\$10.08	88.29	97 97	87.08	88.93	00 003
INC. Incometé Chana Liana Savira Date : 14	SCHAN	100	≨ 3	9/3	61.13	2	8 2	20.00	200	9
WRC - Incremental Charge - Manual Service Order - Addi	SOUNN	. ≥	<u> </u>	. ₹	ž	. ≨	ž	¥	×	§ <u>\$</u>
Per 8XX e Established tulo POTS (sefeXX No.) Translations										
MC-14	ž	\$15.88	≨ ≨	\$12.81	8000	\$12.27	27.9	23.62	22.53	\$67.50
NRC-Decorned Charge 1st	¥	\$10.04	≨ ≩	2 2	Y Y	8 3	21.2	Z AZ	200	X
MRC - Disconnect Charge - Addi	¥	25 05	≨	ž	ž	5.73	38.03	ž	¥	ž
MAC - Incremental Charge - Menual Service Order - 1st	NYTOS	27.37	≨:	\$18	YN	\$18.14	22.523	81.13	ş	¥
MACC. Incremental Change - Manual Service Order - Changesta	S S	ž į	2 2	٤ş	2 2	¥ ::	¥ 9	≨ 5	ž	¥ s
Per 8XX 6 Established with POTS Translations								[
\$7-O#\$	Kertx	\$15.68	≨	\$12.81	\$30.50	\$12.27	19.04	23.62	\$22 K3	\$67.50
MAC: Additional Chams: 14	XBTX LEFTX	20.5	≨ 3	¥ 1	22.23	5 5	25 53	22.2	22.23	97.50
MRC - Decomed Charge - Adril	XTX	16 05	1	ž	ş	2,00	98	ž	. ≥	2
MRC - Incremental Charge - Manual Service Order - 1st	SOMAN	27.37	ž	81894	ş	\$18.14	\$25.62	21.38	MA	≨
MATC - Incremental Charge - Manual Service Order - Add	SOMM	≨ .	≨ :	≨ :	ž	≨ .	YY.	≨ :	ž	≨:
Carbonizad Ara of Sevice per EXX Number		617,13	٤	1	ž	2	8	٤	٤	٤
W.C. 16	ĭ 130 130 130 130 130 130 130 130 130 130	88.88	2	2 2	48	17.53	23 53	38	73 87	8
MRC. Strangerist Change - Manual Service Order - 1st	200	8 7	≨ 5	2 2	2	1 12	2	2 2	2 1	8
MRC - Insumental Charge - Manual Sevice Order - Addi	BOLLAN	≨	ž	₹	⋨	≨	≱	Υ×	₹	ž
Multiple inter LATA Carrier Routing per Carrier Requested per EXX 8	AC STATE	77.77	3	24.43	40.49	144 444	27.77	27.77	17	77.54
Water Aden	MARAIX	193	≨ ≨	88.03	72	38.03	# 53	#19	11.11	8 2
NRIC - Incremental Charge - Manual Service Order - Ist	NYNOS	≨ ŝ	≨ :	≨ :	≨:	≨ :	¥.	ž	¥	ž
Change Charte Set Round	L. Control	٤	٤	٤	Š	Ş	٤	2	Š	٤
X40.13	NBF AX	98.10	ž	II.U	\$11.24	10 (1	27.84	10 82	7.7	88
MAC Again	Ne Ax	20 97	≨:	87.00	81.18	2	3	80.5	3	9
MRC - Incremental Charge - Manual Service Order - Add	SOUVE	, ¥	٤٤	ž	٤ž	. ≨	X X	X Y	5 ¥2	٤ž
Cell Hending and Desirention Features	AUSTRA	77.77	47	1	14.77	# 7	11.00	11.11	77.79	¥ Y
KIRC: X68	NG DX	≨	≨	, ş	683	47	2 23	2 2	138	823
LIDS Common Transport per query	188			BC20000 03	900000	60 0000418	10 0000448	50000	\$0 0000442	80 000
L.DB Validation per query L.DB Originating Point Code Establishment or Change - 1810	8	3 3		\$50.00	8107.60		500142132	800	20 014 1003 26 1 62	SO Y
NRC : Incremental Charge : Bectrons Service Order				ž	ž	≨	₹	\$62.78	≨	ž

DESCRIPTION INC. Prominal Charge - Manual Service Order - 1st NRC - Prominant Charge - Manual Service Order - 1st	SOMAN	¥83	≟	A 20	≥ ≨	5 €	23.822	7 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	26.64 20.84	1N 891.00
	Name of the last	Š	П	Ş	NA.	¥	ž	≨	3	¥X
CCS7 Signaling Connection, per link (A link) per month		818.79	85.00	£17.65	618.31	410.48	19163	41.25.00	47.162	VX 45.8
i la		861/18	200075	\$131.98	1354.98	17834	\$16972	\$310.00	27.07	310.00
Ш	SOMAN	88.83	٤ž	2 8 E	ξž	\$16.14	\$134.08	≨≨	% ¥ ¥¥	≨≨
CCST Sharaking Connection, per link (8 link) (sinc increm as 0 link) per month	BOMAN	\$18.31 \$18.73	88.00 8.00	¥1,8	A.A. \$16.31	97 61 5	80.8152	A5 86 88	YN S	žŞ
NHC Necomed		S 1/15	00.00	8131.98	3354.98	7.77	\$169.72	1510.00	1911/01	\$10.00
MC - Incernetal Charge - Marual Service Order	SOUM	88.83	ž	75 PE	≨	\$1915	Π	≨	ž	žž
CCS7 Menutral Certainedica, per STP port per month.	POWA	\$148.72			9	\$161.99	\$18.05	\$132.88	\$156.33	NA 00.8803
CCST Standard Usege, per ISLP inneasege [44pt/cable when metalurament and billing capability extent.]				75000007	\$0.00003788G	0000000		\$0 0000	20 0000422	7
CCST Signaling Usage, per TCAP reseage [Activities with metallisment and billion countility action)		\$0,000 t	M00000	0/80000 08	\$0.000102042 \$0.0001062 \$0.0001115	\$0,000,082	\$0.0001115	80,00009	\$0 0001108	\$0,0000\$
CCST Uprathry Usage Surrogate, per link per LATA per mo (9)		\$378.12	84.00	19 0453	\$529.08	\$408.71	E5 90+1	8336.98	\$396.65	00.586.0
W.C		90 295	93 294	\$62.00	942.00	\$42.00	\$42.00	862.00	262.00	80.69
A CALL CANADA SERVICE			Т							
Operator Previded Call Hending per min - Using 85T LIDS Call Centries Access Territoria Clarce per call attents	×	\$1.21	8 4 3	96ZU996 06	\$1.603.6	16.03	8 13	Q	1213	¥
Operator Previded Call Handling per min - Using Foreign LIDS	ΎA	\$1.28	П	2018	\$1,6249	36 OK	\$1.24	\$1.24	\$1.25	5 ≨
Operator Provided Cell Henderg, per cell	¥¥	2 1	5 3	ž ž	5 5		ž ž	≨≨	80 ox	¥8
Fully Automated Cell Handling per cell - Using BST LIDS Fully Automated Cell Handling per cell - Halve Foreign LIDS	YA A	= 5	01.03	0.0778409	П	0. C	0.10728	100	\$01115808	5.
Professional recording of name (OCP atons)	П	00 000 03		00 000		\$7,000,00	\$7,000.00	87,000,00		8000
Promessonal recording of name (DA and OCP stone) ORAM or frost-and loading, ser 10PS series	T	2,000		87,000,00	П	\$7,000.00	0 000'43	17,000,00		\$7,000,00
AABS or heat-and loading, per IVS	USODS	223.00	223.00	223.00	\$225.00	\$225.00	8228.00	888	228.00	8 52 8
Rushs or e- automation locarity, per NAY energy Recording Charge per Branded Annoencement - Decornect - Infilial	USO02	2270.00		80 M	т	00 PZ	270.00	270.00		27000
	Υ¥	19.63	ž	≨	₹	¥	¥	ž	ž	ž
A Contraction of a River State Section and the Contraction			+							
Verficialon, per minute Verficialon, per minute Verficialon and Emergent Internation and Advisory	VA		IT	20.20	≨:	18 2			2	≱:
Vertication, per cell	į į	2	8 8	¥ (80	81.00	8 ¥	¥.	2 2 2	e ¥	¥8 8
Vermoston and Emergency Internet, per cell OHECTORY ASSISTANCE SERVICES	¥	≨	1	≨	\$11.11	¥	¥	99.03	Υ¥	8
Directory Assist Call Compution Access Tre (DACC), per call ettempti	≨ ā	0.72	0.0	0.10	01.08	90.70	01.03	2000	01.0	61.03
Number Services Interrupt per query	W.A	80.028	Т	10.0007497	80000	\$ 80 80 80 80 80 80 80 80 80 80 80 80 80 8	10.0166268	011008	\$0 0124036	¥ <u>~</u>
Number Services Intercept per Intercept Query Update Directory Assessora Access Service Cults, are coll	V _A	¥.65	П	¥¥ 5	П	¥ e va	¥ \$ \$	N S.C. MA	¥.	¥
Professional recording of name (OA atons)	П	L.	L	L	000000	딦	E1,000.00 E3,000.00 E3,000.00	00 000 C3	00 000 CS	2,000,00
DRAM or front and loading, per TOPS switch	П	++	-	∔ ∔	-1 1	00034	00.0553	25000	00 OS	0000
on leading, per KAP shall	XX	2000	88	270.00	20,00	220.00	270.00	22.000	822	23.58 23.08 23.08 23.08
Recording Charge per Branded Amoustoment - Discornect - Initial Recording Charge per Branded Amoustoment - Discornect - Subsequent	∀ ¥ ≱	5 5 2 2	≨≨	≨≨	¥	≨≨	¥	¥¥	¥	\$ \$
TO DESCRIPTION OF STREET WAS ASSESSED.										
Directory Transport - Local Charmel DS1, per month	žž	25 82	19 27 27 27	136.38 15.6.13	27.03	22.00	5 863 2 75 73	11	87.50	\$133.61
MRC - Add	¥	16.2 PM	17 923	\$312.80	15 97 25	67.06.73	14 36 28	60 2975	1 627	148
MRC - Decorated Outpe - Add	ŠÁ.		\prod	٤٤	2 2	223	88	<u>د</u> ا	≨ ≨	≨ ≨
ANC - Incomment Chep-Manual Sec Occe- NAC - 18	SCHOOL	g g		Z Z	≨≱	₹ 2	7. Se 12.	24.15	S X	ž ž
MRC - Incremental Charge-Menual Svc Ontar - MRC Disconnect Onestory Transport - Dedicated DS1 Level Interoffice per mile per and	SOLAW VA	20 692	¥0803	¥6.4523	NA 10 45	87.03	10 6596	NA \$0 5753	10.7598	¥67 878
	4 4 2	879 68 8196 15	2. 2. 2. 2. 2. 2. 2. 2. 3. 3. 3.	\$78.47	80 995 1 866 2 986 1 866 2 986	140.49	\$74.40	11.71.53	-	00 00
MRC - Add	¥2	\$148.18	3 2	S: 11.73	8231.18 173	\$106.69	\$147.31	816375	2,527	\$100.49
NRC: Decorned Chays - Add NRC: Promental Chans - Manual Service Crids - 1st	SOLAN	26.63	ž ž	3	źź	216 34	2 23	¥¥ PE	¥ E	33
NAC - Incremental Charge - Manual Service Order - Add1	SOMM	27.27	≨	¥	¥	1	282	40963	23.62	¥

STOTESTED AND BOOK

* * * * * * * * * * * * * * * * * * * *	≨ ≨ 2	\$ \$ \$	\$ \$ \$ \$	≱ ≱	\$ \$\$\$\$	5 55	33 33	33 3	 	3 3 3 3	3333
8C NA NA NA NA NA NA NA NA NA NA NA NA NA	10 0444 1127 23 127 23	9: YN	£ 12 £ 22 €	86 ZOG	10.03 80.000 04 80.000 04	¥ ¥ ¥ ¥ ¥ ¥ ¥	88 XX 88 XX	173 00 NA 173 00	18.028 NA 18.028 NA 18.028	6150 25 NA 50 0250662 50 005950	\$1.73 \$15.80 \$72.15 NA
MAC MAC MAC MAC MAC MAC MAC MAC MAC MAC	20 Oct 620	┡	7	53 00 ES	8172.08 NA NA 10.0023 10.0031 12.08	¥ 88 72	84 363 00 NA 87 378 NA	┞╶┠ ╌╂╼╂╼┼		8145 FK B NA 80 02 80	\$1 45 815 88 871 80 87
### ### ### ### ### ### ### ### ### ##	\$0.0447 \$128.17	1111	2 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	\$129.83 \$79.91	1131.54 143.77 10.0029 10.0075600	NA 169.31 80.35.98	88.379.00 NA NS.903 00.903	Ĭ	╺╏╸╏ ╶╏╌╏╌╏	\$108.90 \$40.44 \$0.025.6138 \$0.006.5161	51.75 51.60 51.28 51.28
12 0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	\$0.04.2 \$90.54	\$183.31 \$78.06	19 619	2 2 2 8 8 8	80 0029 10 10 10 10 10 10 10 10 10 10 10 10 10 1	NA \$153.25 \$78.08	28.315.00 NA NA E41.08	90 17	╺╂╶┧╼┨╌┨╌┩╸╏╶╽	25 25 25 25 25 25 25 25 25 25 25 25 25 2	\$1.70 \$15.89 \$34.61 \$21.97
10 0001 10 0001 10 0001 10 0001 10 0000 1	\$0.0193 \$120.78 NA	≨ ₹	3 3 3	23	3333	* * * *	33 33	≨ ≨ ≨	111111	3323	3333
20000188 200000188 200000188 20000000000	\$79.82 \$79.82 \$79.82 \$79.82	130.28 NA 124.53	¥ 88 ¥	£5.43 ¥X	\$35.44 NA \$0.0023 \$0.0795604	¥ %	86.348.00 KA KA KA	\$114.80 EX.	\$7008 NA \$70.08 NA	870.08 NA 80 000223 80.0003137	± 2 2 × 5 × 5 × 5 × 5 × 5 × 5 × 5 × 5 × 5
# 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	P00000 08	2 €	≨	≨ ≨	3 3 3 3	<u>8</u> ₹5	33 33	33 3 3	╁┼╂┼┼┼	2 2 2 2 2 2 2 2	3333
812.97 \$10003 \$10003 \$10003 \$10003 \$10003 \$150 \$150 \$175.46 \$1	\$128.56 \$128.56 KA	\$187.49 \$114.22	50 55 55 55 55 55 55 55 55 55 55 55 55 5	27 E	82000 82000 8317 8317 8317 8317 8317 8317 8317 8317	1 2 2 1 1 2 2 1 1 1 2 2 1 1 1 2 2 1 1 1 2 2 1 1 1 2 2 1 1 1 2 2 1	20 X 27 X 27 X 27 X 27 X 27 X 27 X 27 X	7 8 7 8 8 8 8 8 8	8117.98 817.90 8117.80	20 02 1 20 00 00 00 00 00 00 00 00 00 00 00 00	2831 215
BOUNN SOLWN NA	AN MAN	CWSE	CAMIP	CAMA	SAME SAME SAME SAME SAME SAME SAME SAME	25 25 25 25 25 25 25 25 25 25 25 25 25 2	BAPYX BAPTT BAPTT	EWING DIAME			BAPAS BAPAS BAPAS BAPAS
DESCRIPTION NRC: Incomeral Diago: Manual Service Order: Discorned: 181 NRC: Incomeral Diago: Manual Service Order: Discorned: 181 Selberge Commercial Orange: Manual Service Order: Discorned: Additional Commercial Orange: Manual Service Order: Discorned: Additional Commercial Orange or Discorned: Additional Service Orange or Selberge Orange or Selberge Orange or Selberge Orange or Selberge Orange or Selberge or Selbe	Directory Assistance Database Service charge per letting Obraciory Assistance Database Service, per month Obraciory Assistance Database Service, per month All per messes And - Ball-South AN SASS Access Service And - Ball-South AN SASS Access Service And - Ball-South AN SASS Access Service	Service Estatelerment Charge, ppr stein, print feelup MPC - Discorned Port Cornection - Dial/Shand Access MPC - Recorded	Pari Connection - ISDN Access NHC - Discovered User ID Cocks - per User ID Cocks	NHC - Discorned Society Card par Uses 10 Cods, tridg or replacement	WHC: Decornact Storage, per unit (1000) Storage, per unit (1000) Station per musik Station per musik AMI - Battlocath AMI footiet service	Service Examinent Charge, per ease, initial set-up. N-RC. M-RC. Moorwell Initialized Seesing, per castoner.	NATC: Concerned Infigure Access Charge, per Figure, per CN, Term Alempt ARC MRC Concerned Infigure Access Charge, per Figure per CN CM Annual Dates	NAC : Discorned NRC : Discorned NRC : Discorned NRC : Discorned	MRC - Discorned Charge, per tryger, per DN, IO-0git POOP MRC - Discorned MRC - Discorned MRC - Discorned MRC - Decorned MRC - Decorned MRC - Decorned MRC - Decorned MRC - Decorned	NTC: Decorned Over Charge, per query Type 1 Hode Charge, per AM Todes Subscripton, per rock, per query	Monthly Report - per AN Toulet Service Bellescription NRC - Discorned NRC - Discorned

Americana 2 Estate C Rates - Page 67

USOC AL FL QA	NA \$15.50 NA NA NA \$20.17 NA NA \$20.00	NA NA NA NA NA NA NA NA	UEAZZ 80.28 \$0.00 \$0.00 \$0.31 \$0.00 \$0.3898 \$0.00 \$0.0	UEACH 868 10.24 80.50 8082 80.22 80.798 80.19 10.10 10	11.03 11.03	**************************************	A set forth in applicative (PelSouth land or as responsibled by the parties upon request by either party.
DESCRIPTION Special Bardy - per AIN Toolkit Service Subscription HRC Cell Event Apport - per AIN Toolkit Service Subscription HRC HRC Cell Event Apport - per AIN Toolkit Service Subscription HRC Cell Event Special Bardy - per AIN Toolkit Service Subscription HRC HRC Cell Event Special Bardy - per AIN Toolkit Service Subscription HRC Cell Event Special Bardy - per AIN Toolkit Service Subscription	CALLE (Controller Community) for Called (Controller Co	For Line or Part Track, each Part Line or Part Track, each INSC Customized foulting per unique line clean couls, per request, per sentich INSC IN	PRC PAC IR PAC IR PAC IR PAC IR PAC PA	PC PC PC PC PC PC PC PC	### ### ### ### ### ### ### ### ### ##	DBI Crass Cennesse RC - Lott NRC - Marial Service Order - 1st NRC - Marial Service Order - Adri DBI Crass Cennesse RC - Lott RRC - Lott RRC - Lott RRC - Marial Service Order - 1st RRC - Marial Service Order - 1st RRC - Marial Service Order - 1st RRC - Marial Service Order - 1st	If no rate is blandfed in the context, the rate for the specific service or furnation will be reserved to the context of the c

BELLACUTHMEWROLTH FATES HETWORK ELEMENTS AND OTHER SERVICES

Attactories 2 Earlos C Rates - Page 68

DESCRIPTION 1 This charge is only applicable where signaling usage measurement or billing 2 This charge is only applicable where signaling usage measurement or billing 4 Prices for AM be determined upon development of mediation device. (Th) 5 Price for Line Cases Codes for Selective Routing shall be determined by the											
3 This charge is only applicable where signaling season measurement or billing completely does not be determined upon development of mediation device. (Th) 6 Phose for Line Class Codes for Selective Rouding shall be determined by the	DESCRIPTION	OSO	₹	ಪ	ð	¥	1	2	¥	ន	ž
4 Prices for AN to be obsermined upon development of mediation device. [This is Price for Une Cases Codes for Selective fourling shall be determined by the	3 This charge is only applicable where signaling usage measurement or billing countries one axis.										
6 Phts for Line Cleas Codes for Selective Rouding shall be determined by the	4 Prices for AN to be determined upon development of mediation device. (TN)										
	8 Price for Line Class Codes for Selective Routing shall be determined by the										

Virtue the same Commission has adopted rates for the rate sements contained hereby, it is the intent of the parties to reflect auch rates in this actibit and to study the same consistent with applicable FCC and Commission named good profess.

Attachment 3

Network Interconnection

TABLE OF CONTENTS

1.	Network Interconnection	3
2.	Interconnection Trunk Group Architectures	6
3.	Network Design And Management For Interconnection	12
4.	Local Dialing Parity	15
5.	Interconnection Compensation	15
6.	Frame Relay Service Interconnection	22
7.	Operational Support Systems (OSS)	24
Ra	tes	Exhibit A
Ba	sic Architecture	Exhibit B
On	ne Way Architecture	Exhibit C
Tw	o Way Architecture	Exhibit D
Su	pergroup Architecture	Fohihit F

The Parties shall provide interconnection with each other's networks for the transmission and routing of Local Traffic, Transit Traffic, Internet Service Provider-Bound ("ISP-Bound") Traffic, and exchange access (intraLATA toll and switched access) on the following terms:

1. Network Interconnection

All negotiated rates, terms and conditions set forth in this Attachment pertain only to the provision of network interconnection where NewSouth owns and provides its switch(es).

- 1.1 Network Interconnection for Call Transport and Termination may be provided by the Parties at any technically feasible point. Requests to BellSouth for interconnection at points other than as set forth in this Attachment may be made through the Bona Fide Request/New Business Request process set out in General Terms and Conditions.
- 1.1.1 An Interconnection Point (IP) is the physical telecommunications equipment interface that performs the interconnection function for BellSouth and NewSouth. Each Party is responsible for providing the network on its side of the IP. Furthermore, the IP must be located within the LATA in which Local Traffic is originated.
- 1.1.2 Pursuant to the provisions of this Attachment, the location of the initial IP in a given LATA shall be established by mutual agreement of the Parties. Subject to the requirements for installing additional IPs, as set forth below, existing IPs will be accepted as initial IPs and will not require re-grooming. When the Parties mutually agree to utilize two-way interconnection trunk groups for the exchange of Local Traffic, ISP-bound Traffic and IntraLATA Toll Traffic between each other, the Parties shall mutually agree to the location of IP(s).
- 1.1.2.1 If the Parties are unable to agree to a mutual initial IP, each Party, as originating Party, may establish a single IP in the LATA for the delivery of its originated Local Traffic, ISP-bound Traffic, and IntraLATA Toll Traffic to the other Party for call transport and termination by the terminating Party.
- 1.1.2.2 Additional IPs in a particular LATA may be established by mutual agreement of the Parties. Absent mutual agreement, in order to establish additional IPs in a LATA, the proposed additional IP must meet the following criteria: (1) the traffic between NewSouth and BellSouth at the proposed additional IP must exceed 8.9 million minutes of Local Traffic and ISP-bound Traffic per month for three consecutive months during the busy hour; and (2) any end office to be designated as an IP must be more than 20 miles from an existing IP. BellSouth will not designate an IP at an end office where physical or virtual collocation space or BellSouth fiber connectivity is not available.

Upon written notification from the Party requesting the establishment of an additional IP, the receiving Party has 20 business days to analyze, respond to, and negotiate in good faith the establishment of and location of such IP. Should the Parties disagree that the traffic volumes justify an additional IP or disagree as to the location of an additional IP, the Parties shall follow the Dispute Resolution process contained in the General Terms and Conditions of this Agreement to determine whether and how the additional IP should be established.

1.2 Interconnection via Dedicated Transport Facilities

- 1.2.1 The Parties shall institute a "bill and keep" compensation plan under which neither Party will charge the other Party recurring and nonrecurring charges for trunks (one-way or two-way), trunk ports and associated dedicated facilities for the exchange of Local Traffic (non-transit), ISP-bound Traffic, and IntraLATA Toll Traffic. Each Party has the obligation to install the appropriate trunks, trunk ports and associated facilities on its respective side of the IP and is responsible for bearing its own costs on its side of the IP. Both Parties, as appropriate, shall be compensated for the ordering of trunks, trunk ports and facilities used exclusively for transit traffic and for ancillary traffic types including, but not limited to, 911 and OS/DA. The Parties agree that charges for such trunks and facilities are as set forth in Exhibit A to this Attachment. Either Party may, at its option, choose to purchase such trunks and facilities from the other Party's tariff.
- 1.2.2 Pursuant to 1.2.1 above, as part of Local Interconnection Call Transport and Termination Service, the originating Party may obtain Local Channel facilities (i.e., entrance facilities) from the terminating Party from the originating Party's specified Interconnection Point to its Serving Wire Center. Such facilities may be purchased out of the terminating party's Commission approved access services tariff or as unbundled network elements at the rates set forth in Exhibit A to this Attachment. If tariffed access services are purchased, the portion of Local Channel facilities utilized for Local Traffic shall be determined based upon the application of the Percent Local Facility (PLF) Factor as defined in this Attachment. Additionally, the charges applied to the portion of the tariffed Local Channel used for Local Traffic as determined by the PLF are as set forth in Exhibit A to this Attachment. This factor shall be reported in addition to the switched dedicated transport jurisdictional factors specified in the BellSouth intrastate and interstate switched access tariffs.
- 1.2.3 Pursuant to 1.2.1 above, either Party may obtain Dedicated Interoffice Transport facilities from its designated Serving Wire Center to the other Party's first point of switching. Such facilities may be purchased out of the terminating party's access services tariff or as unbundled network elements at the rates set forth in Exhibit A to this Attachment. If tariffed access services are purchased, the portion of Dedicated Interoffice Transport facilities utilized for Local Traffic shall be determined based upon the application of the Percent Local Facility (PLF) Factor as defined in this Attachment. Additionally, the charges applied to the portion of the tariffed Dedicated

Interoffice Transport used for Local Traffic as determined by the PLF are as set forth in Exhibit A to this Attachment. This factor shall be reported in addition to the switched dedicated transport jurisdictional factors specified in the BellSouth intrastate and interstate switched access tariffs.

- 1.2.4 For the purposes of this Attachment, Local Channel (i.e., entrance facility) is defined as a flat-non-distance-sensitive rated switch transport facility between a Party's Interconnection Point and its Serving Wire Center.
- 1.2.5 For the purposes of this Attachment, Serving Wire Center is defined as the wire center owned by one Party from which the other Party would normally obtain dial tone for its Interconnection Point.
- 1.2.6 For the purposes of this Attachment, **Dedicated Interoffice Transport** is defined as a distance-sensitive rated switch transport facility between a Party's Serving Wire Center and the first point of switching on the other Party's common (shared) network.

1.3 Fiber Meet

- 1.3.1 Fiber Meet is an interconnection arrangement whereby the Parties physically interconnect their networks via an optical fiber interface (as opposed to an electrical interface) at which one Party's facilities, provisioning, and maintenance responsibility begins and the other Party's responsibility ends (i.e., Interconnection Point).
- 1.3.2 If NewSouth elects to interconnect with BellSouth pursuant to a Fiber Meet, NewSouth and BellSouth shall jointly engineer and operate a Synchronous Optical Network ("SONET") transmission system by which they shall interconnect their transmission and routing of Local Traffic via a Local Channel facility at either the DS0, DS1, or DS3 level. The Parties shall work jointly to determine the specific transmission system. However, NewSouth's SONET transmission must be compatible with BellSouth's equipment in the BellSouth Interconnection Wire Center, and the Data Communications Channel (DCC) must be turned off, unless otherwise mutually agreed to by the Parties.
- 1.3.3 BellSouth shall, wholly at its own expense, procure, install and maintain the agreed upon SONET equipment in the BellSouth Interconnection Wire Center ("BIWC").
- 1.3.4 NewSouth shall, wholly at its own expense, procure, install and maintain the agreed upon SONET equipment in the NewSouth Interconnection Wire Center ("NewSouth Wire Center").
- 1.3.5 BellSouth shall designate a Interconnection Point outside the BIWC as a Fiber Meet point, and shall make all necessary preparations to receive, and to allow and enable NewSouth to deliver, fiber optic facilities into the Interconnection Point with sufficient spare length to reach the fusion splice point at the Interconnection Point. BellSouth shall, wholly at its own expense, procure, install, and maintain the fusion splicing point in the Interconnection Point. A Common Language Location

Identification ("CLLI") code will be established for each Interconnection Point. The code established must be a building type code. All orders shall originate from the Interconnection Point (i.e., Interconnection Point to NewSouth, Interconnection Point to BellSouth).

- 1.3.6 NewSouth shall deliver and maintain such strands wholly at its own expense. Upon verbal request by NewSouth, BellSouth shall allow NewSouth access to the Fiber Meet entry point for maintenance purposes as promptly as possible.
- 1.3.7 The Parties shall jointly coordinate and undertake maintenance of the SONET transmission system. Each Party shall be responsible for maintaining the components of their own SONET transmission system.
- 1.3.8 Each Party will be responsible for (i) providing its own transport facilities to the Fiber Meet, and (ii) the cost to build-out its facilities to such Fiber Meet.
- 1.3.9 Neither Party shall charge the other for its portion of the Fiber Meet facility used exclusively for non-transit Local Traffic (i.e., the Local Channel). Charges for unbundled network element facilities shall be billed at the rates set forth in Attachment 3, Exhibit A. Charges for Switched and Special Access Services shall be billed in accordance with the applicable Access Service tariff (i.e. the BellSouth Interstate or Intrastate Access Services Tariff).

2. Interconnection Trunk Group Architectures

- 2.1 BellSouth and NewSouth shall establish interconnecting trunk groups and trunk group configurations between networks including the establishment of one-way or two-way trunks in accordance with the following provisions set forth in this Agreement. For trunking purposes, traffic will be routed based on the digits dialed by the originating end user and in accordance with the Local Exchange Routing Guide (LERG).
- 2.2 NewSouth shall establish an interconnection trunk group(s) to at least one BellSouth access tandem within the LATA for the delivery of NewSouth's originated local and intraLATA toll traffic and for the receipt and delivery of Transit Traffic. To the extent NewSouth desires to terminate local and intraLATA toll traffic to BellSouth and Transit Traffic to third parties subtending other BellSouth access tandems within the LATA, other than the one NewSouth has established interconnection trunk groups to, NewSouth shall establish trunk groups to such other BellSouth access tandems.
- 2.2.1 Notwithstanding the forgoing, NewSouth shall establish an interconnection trunk group(s) to all BellSouth access and local tandems in the LATA where NewSouth has homed (i.e., assigned) its NPA/NXXs. NewSouth shall home its NPA/NXXs on the BellSouth tandems that serve the Exchange Rate Center Areas to which the NPA/NXXs are assigned. The specified association between BellSouth tandems and Exchange Rate Centers is defined in the national Local Exchange Routing Guide (LERG). NewSouth shall enter its NPA/NXX access and/or local tandem homing arrangement into the LERG.

- 2.3 Switched Access traffic will be delivered to and by Interexchange Carriers (IXCs) based on NewSouth's NXX Access Tandem homing arrangement as specified by NewSouth in the Local Exchange Routing Guide (LERG).
- Any NewSouth interconnection request that deviates from the interconnection trunk group architectures as described in this Agreement that affects traffic delivered to NewSouth from a BellSouth switch that requires special BellSouth switch translations and other network modifications will require NewSouth to submit a Bona Fide Request/New Business Request via the Bona Fide Request/New Business Request Process set forth in Attachment 12 of this Agreement.
- Subject to 1.2.1, charges, both non-recurring and recurring, associated with interconnecting trunk groups between BellSouth and NewSouth are set forth in Exhibit A. To the extent a rate associated with the interconnecting trunk group is not set forth in Exhibit A, the interim rate shall be as set forth in the appropriate Party's Commission filed and effective tariff for Switched Access services. Once a cost based rate is established by BellSouth, the interim tariff rate shall be trued up and the cost based rate will be applied retroactively to the effective date of this agreement.
- Unless the Parties mutually agree otherwise, NewSouth shall be responsible for ordering and paying for any two way trunks carrying Transit Traffic. At such time as NewSouth is providing a transit function on 20% of all Transit Traffic, BellSouth and NewSouth will negotiate alternative compensation for two way trunks carrying Transit Traffic. At such time as NewSouth is providing the transit function on 30% of all Transit Traffic, the Parties shall execute an amendment implementing such alternative compensation for two way trunks carrying Transit Traffic.
- 2.7 All trunk groups will be provisioned as Signaling System 7 (SS7) capable where technically feasible. If SS7 is not technically feasible multi-frequency (MF) protocolsignaling shall be used.
- In cases where NewSouth is also an IXC, the IXC's Feature Group D (FG D) trunk group(s) must remain separate from the local interconnection trunk group(s).
- Unless in response to a blocking situation or for a project, when either Party orders interconnection trunk group augmentations, a Firm Order Confirmation (FOC) shall be returned to the ordering Party within four (4) business days from receipt of a valid error free ASR. A project is defined as a new trunk group or the request of 96 or more trunks on a single or multiple trunk group(s) in a given local calling area. Blocking situations and projects shall be managed through the BellSouth Interconnection Trunking Project Management group and NewSouth's equivalent trunking group.
- 2.10 Interconnection Trunk Groups for Exchange of Local, ISP-Bound, IntraLATA Toll and Transit Traffic
- 2.10.1 If the Parties' originated local, ISP-Bound and/or intraLATA toll traffic is exchanged utilizing the same two-way trunk group, the Parties shall mutually agree to use this

type of two-way interconnection trunk group with the quantity of trunks being mutually determined and the provisioning being jointly coordinated. Furthermore, the Interconnection Point(s) for two-way interconnection trunk groups transporting both Parties local, ISP-Bound and/or intraLATA toll shall be mutually agreed upon. NewSouth shall order such two-way trunks via the Access Service Request (ASR) process in place for Local Interconnection upon determination by the Parties, in a joint planning meeting, that such trunk groups shall be utilized. BellSouth will use the Trunk Group Service Request (TGSR) to request changes in trunking. Both Parties reserve the right to issue ASRs if so required, in the normal course of business. Furthermore, the Parties shall jointly review such trunk performance and forecasts on a periodic basis. The Parties use of two-way interconnection trunk groups for the transport of local, ISP-Bound and/or intraLATA toll traffic between the Parties does not preclude either Party from establishing additional one-way interconnection trunks for the delivery of its originated local, ISP-Bound and/or intraLATA toll traffic to the other Party.

2.10.2 BellSouth Access Tandem Interconnection Architectures

BellSouth Access Tandem Interconnection provides intratandem access to subtending end offices.

2.10.2.1 Basic Architecture

2.10.2.1.1 In this architecture, NewSouth's originating Local traffic, ISP-Bound traffic, and IntraLATA Toll and originating and terminating Transit Traffic is transported on a single two-way trunk group between NewSouth and BellSouth access tandem(s) within a LATA. This group carries intratandem Transit Traffic between NewSouth and Independent Companies, Interexchange Carriers, other CLECs and other network providers with which NewSouth desires interconnection and has the proper contractual arrangements. This group also carries NewSouth originated intertandem traffic transiting a single BellSouth access tandem destined to third party tandems such as an Independent Company tandem or other CLEC tandem. BellSouth originated Local, ISP-bound and IntraLATA Toll traffic is transported on a single one-way trunk group terminating to NewSouth. Other trunk groups for operator services, directory assistance, emergency services and intercept may be established if required. The LERG should be referenced for current routing and tandem serving arrangements. The Basic Architecture is illustrated in Exhibit B.

2.10.2.2 One-Way Trunk Group Architecture

2.10.2.2.1 In this architecture, the Parties interconnect using two one-way trunk groups. One one-way trunk group carries NewSouth-originated local, ISP-Bound, and intraLATA toll traffic destined for BellSouth end-users. The other one-way trunk group carries BellSouth-originated local, ISP-Bound, and intraLATA toll traffic destined for NewSouth end-users. A third two-way trunk group is established for NewSouth's originating and terminating Transit Traffic. This group carries intratandem Transit Traffic between NewSouth and Independent Companies, Interexchange Carriers, other CLECs and other network providers with which NewSouth desires

interconnection and has the proper contractual arrangements. This group also carries NewSouth originated intertandem traffic transiting a single BellSouth access tandem destined to third party tandems such as an Independent Company tandem or other CLEC tandem. Other trunk groups for operator services, directory assistance, emergency services and intercept may be established if required. The LERG should be referenced for current routing and tandem serving arrangements. The One-Way Trunk Group Architecture is illustrated in Exhibit C.

2.10.2.3 Two-Way Trunk Group Architecture

2.10.2.3.1 The Two-Way Trunk Group Architecture establishes one two-way trunk group to carry local, ISP-Bound, and intraLATA toll traffic between NewSouth and BellSouth. In addition, a two-way transit trunk group must be established for NewSouth's originating and terminating Transit Traffic. This group carries intratandem Transit Traffic between NewSouth and Independent Companies, Interexchange Carriers, other CLECs and other network providers with which NewSouth desires interconnection and has the proper contractual arrangements. This group also carries NewSouth originated intertandem traffic transiting a single BellSouth access tandem destined to third party tandems such as an Independent Company tandem or other CLEC tandem. Other trunk groups for operator services, directory assistance, emergency services and intercept may be established if required. The LERG should be referenced for current routing and tandem serving arrangements. The Two-Way Trunk Group Architecture is illustrated in Exhibit D.

2.10.2.4 Supergroup Architecture

2.10.2.4.1 In the Supergroup Architecture, the Parties' Local, ISP-Bound, and IntraLATA Toll and Transit Traffic are exchanged on a single two-way trunk group between NewSouth and BellSouth. This group carries intratandem Transit Traffic between the Parties and Independent Companies, Interexchange Carriers, other CLECs and other network providers with which NewSouth desires interconnection and has the proper contractual arrangements. This group also carries NewSouth originated intertandem traffic transiting a single BellSouth access tandem destined to third party tandems such as an Independent Company tandem or other CLEC tandem. Other trunk groups for operator services, directory assistance, emergency services and intercept may be established if required. The LERG should be referenced for current routing and tandem serving arrangements. In the event either Party desires to use Supergroup, the Parties will mutually agree on the compensation applicable for the portion of the trunks and facilities used for transit traffic. The Supergroup Architecture is illustrated in Exhibit E.

2.10.3 Local Tandem Interconnection

2.10.3.1 Local Interconnection trunk group(s) may be established at BellSouth local tandems for: (1) the delivery of NewSouth-originated Local Traffic transported and terminated by BellSouth to BellSouth end offices within the local calling area as defined in BellSouth's General Subscriber Services Tariff (GSST), section A3 served by those BellSouth local tandems, and (2) for local Transit Traffic transported by BellSouth for

- third party network providers who have also established an interconnection trunk group(s) at those BellSouth local tandems.
- 2.10.3.2 When a specified local calling area is served by more than one BellSouth local tandem. NewSouth must designate a "home" local tandem for each of its assigned NPA/NXXs and establish trunk connections to such local tandems. Where requested facilities are not available, NewSouth and BellSouth will route NewSouth's traffic, if possible, to a point where facilities are available. This alternative routing will be an interim solution until requested facilities are available. Additionally, NewSouth may choose to establish an interconnection trunk group(s) at the BellSouth local tandems where it has no codes homing but is not required to do so. NewSouth may deliver Local Traffic to a "home" BellSouth local tandem that is destined for other BellSouth or third party network provider end offices subtending other BellSouth local tandems in the same local calling area where NewSouth does not choose to establish an interconnection trunk group(s). It is NewSouth's responsibility to enter its own NPA/NXX local tandem homing arrangements into the Local Exchange Routing Guide (LERG) either directly or via a vendor in order for other third party network providers to determine appropriate traffic routing to NewSouth's codes. Likewise, NewSouth shall obtain its routing information from the LERG.
- 2.10.3.3. Notwithstanding establishing an interconnection trunk group(s) to BellSouth's local tandems, NewSouth must also establish an interconnection trunk group(s) to BellSouth access tandems within the LATA on which NewSouth has NPA/NXXs homed for the delivery of Interexchange Carrier Switched Access (SWA) and toll traffic, and traffic to Type 2A CMRS connections located at the access tandems. BellSouth shall not switch SWA traffic through more than one BellSouth access tandem. SWA, Type 2A CMRS or toll traffic routed to the local tandem in error will not be backhauled to the BellSouth access tandem for completion. (Type 2A CMRS interconnection is defined in BellSouth's A35 General Subscriber Services Tariff).
- 2.10.3.4 BellSouth's provisioning of local tandem interconnection assumes that NewSouth has executed the necessary local interconnection agreements with the other third party network providers subtending those local tandems in a manner consistent with the requirements of the Act.

2.10.4 Direct End Office-to-End Office Interconnection

- 2.10.4.1 Direct End Office-to-End Office one-way or two-way interconnection trunk groups allow for the delivery of a Party's originating local or intraLATA toll traffic to the terminating Party on a direct end office-to-end office basis.
- 2.10.4.2 The Parties shall utilize direct end office-to-end office trunk groups under the following conditions:
- 2.10.4.2.1 Tandem Exhaust If a tandem through which the Parties are interconnected is unable to, or is forecasted to be unable to support additional traffic loads for any period of time, the Parties will mutually agree on an alternative tandem trunking plan or end

- office trunking plan that will alleviate the tandem capacity shortage and ensure completion of traffic between NewSouth and BellSouth's subscribers.
- 2.10.4.2.2 Traffic Volume -To the extent either Party has the capability to measure the amount of traffic between a NewSouth switching center and a BellSouth end office, either Party shall install and retain direct end office trunking sufficient to handle actual or reasonably forecasted traffic volumes, whichever is greater, between a NewSouth switching center and a BellSouth end office where the traffic exceeds or is forecasted to exceed two DS1s of traffic per month. Either Party will install additional capacity between such points when overflow traffic between NewSouth's switching center and BellSouth's end office exceeds or is forecasted to exceed two DS1s of traffic per month. In the case of one way trunking, additional trunking shall only be required by the Party whose trunking has achieved the preceding usage threshold.
- 2.10.4.2.3 Mutual Agreement The Parties may install direct end office trunking upon mutual agreement in the absence of conditions (1) or (2) above and agreement will not unreasonably be withheld.

2.10.5 Transit Traffic Trunk Group

2.10.5.1 Transit Traffic trunks can either be two-way trunks or two one-way trunks ordered by NewSouth to deliver and receive local and intraLATA toll Transit Traffic from third parties, such as Independent Companies and other CLECs, via BellSouth access tandems (or BellSouth local tandems for Local Traffic), and Switched Access traffic to and from Interexchange Carriers via BellSouth access tandems pursuant to the Transit Traffic section of this Attachment. Establishing Transit Traffic trunks at BellSouth access and local tandems provides intratandem access to the third parties also interconnected at those tandems.

2.10.6 Toll Free Traffic

- 2.10.6.1 If NewSouth chooses BellSouth to handle Toll Free database queries from its switches, all NewSouth originating Toll Free traffic will be routed over the Transit Traffic Trunk Group.
- 2.10.6.2 All originating Toll Free Service (Toll Free) calls for which NewSouth requests that BellSouth perform the Service Switching Point ("SSP") function (i.e., perform the database query) shall be delivered using GR-394 format over the Transit Traffic Trunk Group. Carrier Code "0110" and Circuit Code (to be determined for each LATA) shall be used for all such calls.
- 2.10.6.3 NewSouth may handle its own Toll Free database queries from its switch. If so, NewSouth will determine the nature (local/intraLATA/interLATA) of the Toll Free call based on the response from the database. If the query determines that the call is a BellSouth local or intraLATA Toll Free number, NewSouth will route the post-query local or IntraLATA converted ten-digit local number to BellSouth over the local or intraLATA trunk group. If the query determines that the call is a third party (ICO or other CLEC) local or intraLATA Toll Free number, NewSouth will route the post-

query local or intraLATA converted ten-digit local number to BellSouth over the Transit Traffic Trunk Group. In such case, NewSouth is to provide a Toll Free billing record when appropriate. If the query reveals the call is an interLATA Toll Free number, NewSouth will route the post-query interLATA call (Toll Free number) directly from its switch for carriers interconnected with its network or over the Transit Traffic Trunk Group to the BellSouth Access Tandem for carriers not directly connected to its network. Calls will be routed to BellSouth over the local/intraLATA and Transit Traffic Trunk Groups within the LATA in which the calls originate.

2.10.6.4 All post-query Toll Free Service (Toll Free) calls for which NewSouth performs the SSP function, if delivered to BellSouth, shall be delivered using GR-394 format for calls destined to IXCs, and GR-317 format for calls destined to end offices that directly subtend the BellSouth access tandem.

3. Network Design And Management For Interconnection

- Network Management and Changes. Both Parties will work cooperatively with each other to install and maintain the most effective and reliable interconnected telecommunications networks, including but not limited to, the exchange of toll-free maintenance contact numbers and escalation procedures. Both Parties agree to provide public notice of changes in the information necessary for the transmission and routing of services using its local exchange facilities or networks, as well as of any other changes that would affect the interoperability of those facilities and networks.
- Interconnection Technical Standards. The interconnection of all networks will be based upon accepted industry/national guidelines for transmission standards and traffic blocking criteria. Interconnecting facilities shall conform, at a minimum, to the telecommunications industry standard of DS-1 pursuant to Bellcore Standard No. TR-NWT-00499. Signal transfer point, Signaling System 7 ("SS7") connectivity is required at each interconnection point. BellSouth will provide out-of-band signaling using Common Channel Signaling Access Capability where technically and economically feasible, in accordance with the technical specifications set forth in the BellSouth Guidelines to Technical Publication, TR-TSV-000905. Facilities of each Party shall provide the necessary on-hook, off-hook answer and disconnect supervision and shall hand off calling number ID (Calling Party Number) when technically feasible.
- Ouality of Interconnection. The local interconnection for the transmission and routing of telephone exchange service and exchange access that each Party provides to each other will be at least equal in quality to what it provides to itself and any subsidiary or affiliate, where technically feasible, or to any other Party to which each Party provides local interconnection.
- Network Management Controls. Both Parties will work cooperatively with each other to apply sound network management principles by invoking appropriate network management controls (e.g., call gapping) to alleviate or prevent network congestion.

- Common Channel Signaling. Both Parties will provide LEC-to-LEC Common Channel Signaling ("CCS") to each other, where available, in conjunction with all traffic in order to enable full interoperability of CLASS features and functions except for call return. All CCS signaling parameters will be provided, including automatic number identification ("ANI"), originating line information ("OLI") calling company category, charge number, etc. All privacy indicators will be honored, and each Party will cooperate with each other on the exchange of Transactional Capabilities Application Part ("TCAP") messages to facilitate full interoperability of CCS-based features between the respective networks. Neither Party shall alter the CCS parameters, or be a party to altering such parameters, or knowingly pass CCS parameters that have been altered in order to circumvent appropriate interconnection charges.
- 3.6 <u>Signaling Call Information</u>. BellSouth and NewSouth will send and receive 10 digits for Local Traffic. Additionally, BellSouth and NewSouth will exchange the proper call information, i.e. originated call company number and destination call company number, CIC, and OZZ, including all proper translations for routing between networks and any information necessary for billing.

3.7 Forecasting for Trunk Provisioning

- 3.7.1 Within six (6) months after execution of this agreement, NewSouth shall provide an initial interconnection trunk group forecast for each LATA that it shall provide service within BellSouth's region. Upon receipt of NewSouth's forecast, the Parties shall schedule and participate in a joint planning meeting to develop a joint interconnection trunk group forecast. Each forecast provided under this Section shall be deemed "Confidential Information" under the General Terms and Conditions Part A of this Agreement.
- 3.7.1.1 At a minimum, the forecast shall include the projected quantity of Transit Trunks, NewSouth-to-BellSouth one-way trunks ("NewSouth Trunks"), BellSouth-to-NewSouth one-way trunks ("Reciprocal Trunks") and/or two-way interconnection trunks, if the Parties have agreed to interconnect using two-way trunking to transport the Parties' local and intraLATA toll. The quantities shall be projected for a minimum of six months in advance and shall include the current year plus next two years total forecasted quantities. Considering NewSouth's provided forecast, the Parties shall mutually develop Reciprocal Trunk and/or two-way interconnection trunk forecast quantities for the time periods listed and to be included within the initial forecast.
- 3.7.1.2 Additionally, all forecasts shall include, at a minimum, trunk group type (local/intraLATA toll, Transit, Operator Services, 911, etc.), A location/Z location (CLLI codes for NewSouth location and BellSouth location where the trunks shall terminate), interface type (e.g., DS1), Direction of Signaling, Trunk Group Access Code, if known, (commonly referred to as the 2-6 code) and forecasted trunks in service each year (cumulative).

- Each Party shall exercise its best efforts to provide the quantity of interconnection trunks mutually forecasted. However, the provision of the forecasted quantity of interconnection trunks is subject to trunk terminations and facility capacity existing at the time the trunk order is submitted. Furthermore, the receipt and development of trunk forecasts does not imply any liability for failure to perform if capacity (trunk terminations or facilities) is not available for use at the forecasted time.
- 3.7.3 The submitting and development of interconnection trunk forecasts shall not replace the ordering process in place for local interconnection trunks.
- Once initial interconnection trunk forecasts have been developed, NewSouth shall continue to provide interconnection trunk forecasts on a semiannual basis or at otherwise mutually agreeable intervals. NewSouth shall use its best efforts to make the forecasts as accurate as possible based on reasonable engineering criteria. Interconnection trunk forecasts shall be updated and provided to BellSouth on an as needed basis, but no less frequently than semiannually and no more frequently than monthly. Upon receipt of NewSouth's forecast, including forecast updates, the Parties shall confer to mutually develop BellSouth Reciprocal Trunk and/or two-way interconnection trunk forecasted quantities for the listed time periods within such subsequent forecasts.

3.8 Trunk Utilization

- BellSouth and NewSouth shall monitor traffic on each interconnection trunk group 3.8.1 that is installed pursuant to the initial interconnection trunk requirements and subsequent forecasts six months after the initial installation of the trunks and any time after the end of a calendar quarter thereafter. Based on a review of the capacity utilization during such quarter for installed Reciprocal Trunk groups and/or two-way interconnection trunk groups, subject to the provisions of the section following, BellSouth may disconnect any Non-utilized Reciprocal Trunk(s) or Under-utilized Reciprocal Trunk(s), as defined in Section 3.8.1.1 below, and NewSouth shall refund to BellSouth associated trunk and facility charges paid by BellSouth retroactive to the date on which such trunks failed to maintain compliance with the threshold described in Section 3.8.1.1 below. In addition, BellSouth may request NewSouth to disconnect any Non-utilized or Under-utilized two-way interconnection trunk(s)if BellSouth has determined that the trunk group is not being utilized as described in Section 3.8.1.1 below, provided that the Parties have not otherwise agreed. NewSouth shall comply with such request, subject to Section 3.8.1.1 below.
- 3.8.1.1 The Parties agree that within 180 days of the installation of a trunk or trunks, the trunks will be utilized at 60 percent (60%) of the time consistent busy hour utilization level. The Parties agree that within 365 days of the installation of a trunk or trunks, the trunks will be utilized at eighty percent (80%) of the time consistent busy hour utilization level. Any trunk or trunks not meeting the minimum thresholds set forth in this Section are defined as "Under-utilized" trunks.

- 3.8.1.2 BellSouth's Local Interce mection Switching Center ("LISC") will notify NewSouth of any under-utilized reciprocal trunk groups and the number of trunks that BellSouth wishes to disconnect. BellSouth will provide supporting information either by email or facsimile to the designated NewSouth interface. NewSouth will provide concurrence with the disconnection in seven (7) business days of its receipt of such notification or will provide specific information supporting why the trunks should not be disconnected. Such supporting information should include expected Local Number Ported (LNP) and traffic volumes and the timeframes within which NewSouth expects to need such trunks. BellSouth's LISC project manager and Circuit Capacity Manager will discuss the information with NewSouth to determine if agreement can be reached on the number of trunks to be removed. If no agreement can be reached, BellSouth will issue disconnect orders to NewSouth. The due date of these orders will be four weeks from the date on which NewSouth received notice, in writing, of BellSouth's request to disconnect the underutilized trunk groups.
- 3.8.1.3 To the extent NewSouth requests BellSouth and BellSouth agrees to install additional Reciprocal and/or two-way interconnection trunks in any forecast period following the initial forecasting period that are not included in the forecast for that period (as such forecast may be revised from time to time), such trunks may be provisioned by BellSouth subject to the conditions set forth in the preceding sections.
- To the extent that any Final interconnection trunk group is utilized at, or, based on trend (incorporating linear regression analysis using Erlang B theory with weekly tracked historical traffic data per trunk group engineered at a P.01 grade of service) will reach within six weeks, a time-consistent busy hour utilization level of eighty percent (80%) or greater, the Parties shall negotiate in good faith for the installation of augmented facilities.

4. Local Dialing Parity

- BellSouth and NewSouth shall provide local and toll dialing parity as described in the Act and required by FCC rules, regulations and policies. Dialing parity shall be provided for all originating telecommunications services that require dialing to route a call. BellSouth and NewSouth shall permit similarly situated telephone exchange service end users to dial the same number of digits to make a local telephone call notwithstanding the identity of the end user's or the called party's telecommunications service provider. In addition, NewSouth end users shall experience at least the same service quality level as BellSouth end users in terms of post-dial delay, call completion rate and transmission quality.
- 5. Interconnection Compensation
- 5.1 Compensation for Call Transportation and Termination for Local Traffic

- Page 16 For reciprocal compensation between the Parties pursuant to this Attachment, Local 5.1.1 Traffic is defined as any telephone call that is originated by an end user of one Party and terminated to an end user of the other Party within a given LATA on that other Party's network, except for those calls that are originated or terminated through switched access arrangements.
- 5.1.1.1 Additionally, Local Traffic includes any cross boundary, voice-to-voice intrastate, interLATA or interstate, interLATA calls between specific wire centers established as a local call by the ruling regulatory body.
- For purposes of this Attachment, ISP-bound Traffic is defined as any telephone call to 5.1.1.2 an ESP/ISP that is dialed using a local dialing pattern (7 or 10 digits) by the calling party to an ESP/ISP server physically located within a given LATA ("ISP-bound traffic").
- 5.1.2 For purposes of this Agreement and for traffic between the Parties originating from and directed to the exchanges subject to this Agreement, the Parties agree to a billand-keep arrangement for usage on Local Traffic and ISP-bound traffic. Such billand-keep arrangement includes any per minute of use rate elements associated with the transport and termination of Local Traffic and ISP-bound Traffic (including, but not limited to end office switching, tandem switching, and common transport).
- For the purposes of this Attachment, Common (Shared) Transport is defined as the 5.1.2.1 transport of the originating Party's traffic by the terminating Party over the terminating Party's common (shared) facilities between the terminating Party's tandem switch and end office switch and/or between the terminating Party's tandem switches.
- 5.1.2.2 For the purposes of this Attachment, Tandem Switching is defined as the function that establishes a communications path between two switching offices through a third switching office (the Tandem switch).
- 5.1.2.3 For the purposes of this Attachment, End Office Switching is defined as the function that establishes a communications path between the trunk side and line side of the End Office switch.
- 5.1.3 Neither Party shall represent Switched Access Traffic as Local Traffic for purposes of payment of reciprocal compensation.
- 5.1.4 Pursuant to the definition of Local Traffic in this Attachment, and for the purpose of delivery of one Party's originating traffic to the other, Local Traffic and ISP-Bound Traffic delivered to a terminating Party's end users physically located within the LATA in which the call originated and within which the Party's end user's NPA/NXX is assigned shall be subject to bill-and-keep. If either Party assigns NPA/NXXs to specific BellSouth rate centers within the LATA and assigns numbers from those NPA/NXXs to its end users physically located outside of that LATA, the originating Party's traffic originating from within the LATA where the NPA/NXXs are assigned and delivered to the terminating Party's customer physically located Version 2000: 6/15/00 05/18/01

outside of such LATA, shall not be deemed Local Traffic, and such traffic will not be subject to bill-and-keep. Further, the Parties agree to identify such interLATA traffic to each other and to compensate each other for originating and transporting such interLATA traffic to each other at the originating Party's Commission approved tariffed switched access rates.

- Percent Local Use. Each Party shall report to the other a Percent Local Usage 5.1.5. ("PLU"). The application of the PLU will determine the amount of Local Traffic and ISP-Bound Traffic, per minute of use, subject to bill and keep under this agreement. For purposes of developing the PLU, each Party shall consider every local call, every call to an enhanced/information service provider (including Internet service providers), and every long distance call, excluding Transit Traffic. Each Party shall update its PLU on the first of January, April, July and October of the year and shall send it to the other Party to be received no later than 30 calendar days after the first of each such month based on local usage for the past three months ending the last day of December, March, June and September, respectively. Requirements associated with PLU calculation and reporting shall be as set forth in BellSouth's Percent Local Use Reporting Guidebook, as it is amended from time to time. Notwithstanding the foregoing, where the terminating Party has message recording technology that identifies the jurisdiction of traffic terminated as defined in this Agreement, such information, in lieu of the PLU factor, shall at the terminating Party's option be utilized to determine the appropriate local usage compensation to be paid.
- Percent Local Facility. Each Party shall report to the other a Percent Local Facility ("PLF"). The application of the PLF will determine the portion of switched dedicated transport ordered via BellSouth's E6 tariff to be billed per the local jurisdiction rates. The PLF shall be applied to multiplexing, local channel and interoffice channel switched dedicated transport ordered from BellSouth's E6 tariff and utilized in the provision of local interconnection trunks. Each Party shall update its PLF on the first of January, April, July and October of the year and shall send it to the other Party to be received no later than 30 calendar days after the first of each such month to be effective the first bill period the following month, respectively. Requirements associated with PLU and PLF calculation and reporting shall be as set forth in BellSouth's Percent Local Use/Percent Local Facility Reporting Guidebook, as it is amended from time to time.
- Percent Interstate Usage. Each Party shall report to the other the projected Percent Interstate Usage ("PIU"). All jurisdictional report requirements, rules and regulations for Interexchange Carriers specified in BellSouth's Intrastate Access Services Tariff will apply to NewSouth. After interstate and intrastate traffic percentages have been determined by use of PIU procedures, the PLU and PLF factors will be used, to the extent applicable, for application and billing of local interconnection. Each Party shall update its PIUs on the first of January, April, July and October of the year and shall send it to the other Party to be received no later than 30 calendar days after the first of each such month, for all services showing the percentages of use (PIUs, PLU, and PLF) for the past three months ending the last day of December, March, June and September. Notwithstanding the foregoing, where the terminating Party has message

recording technology that identifies the jurisdiction of traffic terminated as defined in this Agreement, such information, in lieu of the PIU and PLU factors, shall at the terminating Party's option be utilized to determine the appropriate local usage compensation to be paid.

Audits. On thirty (30) days written notice, each Party must provide the other the ability and opportunity to conduct an annual audit to ensure the proper billing of traffic. BellSouth and NewSouth shall retain records of call detail for a minimum of nine months from which a PLF, PLU and/or PIU can be ascertained. The audit shall be accomplished during normal business hours at an office designated by the Party being audited. Audit requests shall not be submitted more frequently than one (1) time per calendar year. Audits shall be performed by a mutually acceptable independent auditory paid for by the Party requesting the audit. The PLF, PLU and/or PIU shall be adjusted based upon the audit results and shall apply to the usage for the quarter the audit was completed, to the usage for the quarter prior to the completion of the audit, and to the usage for the two quarters following the completion of the audit. If, as a result of an audit, either Party is found to have overstated the PLF, PLU and/or PIU by twenty percentage points (20%) or more, that Party shall reimburse the auditing Party for the cost of the audit.

5.5 Rate True-up

This section applies only to Tennessee and other rates that are interim or expressly subject to true-up under this attachment.

- 5.5.1 The interim prices for Unbundled Network Elements and Other Services and Local Interconnection shall be subject to true-up according to the following procedures:
- 5.5.2 The interim prices shall be trued-up, either up or down, based on final prices determined either by further agreement between the Parties, or by a final order (including any appeals) of the Commission which final order meets the criteria of (3) below. The Parties shall implement the true-up by comparing the actual volumes and demand for each item, together with interim prices for each item, with the final prices determined for each item. Each Party shall keep its own records upon which the true-up can be based, and any final payment from one Party to the other shall be in an amount agreed upon by the Parties based on such records. In the event of any disagreement as between the records or the Parties regarding the amount of such true-up, the Parties agree that the body having jurisdiction over the matter shall be called upon to resolve such differences, or the Parties may mutually agree to submit the matter to the Dispute Resolution process in accordance with the provisions of Section 12 of the General Terms and Conditions and Attachment 1 of the Agreement.
- 5.5.3 The Parties may continue to negotiate toward final prices, but in the event that no such agreement is reached within nine (9) months, either Party may petition the Commission to resolve such disputes and to determine final prices for each item. Alternatively, upon mutual agreement, the Parties may submit the matter to the Dispute Resolution Process set forth in the General Terms and Conditions of the

Agreement, so long as they file the resulting Agreement with the Commission as a "negotiated agreement" under Section 252(e) of the Act.

- A final order of this Commission that forms the basis of a true-up shall be the final order as to prices based on appropriate cost studies, or potentially may be a final order in any other Commission proceeding which meets the following criteria:
 - (a) BellSouth and CLEC is entitled to be a full Party to the proceeding;
 - (b) It shall apply the provisions of the federal Telecommunications Act of 1996, including but not limited to Section 252(d)(1) (which contains pricing standards) and all then-effective implementing rules and regulations; and,
 - (c) It shall include as an issue the geographic deaveraging of unbundled element prices, which deaveraged prices, if any are required by said final order, shall form the basis of any true-up.

5.6 Compensation for 8XX Traffic

- 5.6.1 Compensation for 8XX Traffic. Each Party shall compensate the other pursuant to the appropriate switched access charges, including the database query charge as set forth in the Party's Commission or FCC filed and approved intrastate or interstate switched access tariffs.
- 5.6.2 Records for 8XX Billing. Each Party will provide to the other the appropriate records necessary for billing intraLATA 8XX customers. The records provided will be in a standard EMI format.
- 5.6.3 8XX Access Screening. BellSouth's provision of 8XX Toll Free Database (TFD) to NewSouth requires interconnection from NewSouth to BellSouth 8XX SCP. Such interconnections shall be established pursuant to BellSouth's Common Channel Signaling Interconnection Guidelines and Bellcore's CCS Network Interface Specification document, TR-TSV-000905. NewSouth shall establish CCS7 interconnection at the BellSouth Local Signal Transfer Points serving the BellSouth 8XX SCPs that NewSouth desires to query. The terms and conditions for 8XX TFD are set out in BellSouth's Intrastate Access Services Tariff as amended.

5.7 Mutual Provision of Switched Access Service

Switched Access Traffic. Switched Access Traffic is defined as telephone calls requiring local transmission or switching services for the purpose of the origination or termination of Telephone Toll Service. Switched Access Traffic includes the following types of traffic: Feature Group A, Feature Group B, Feature Group C, Feature Group D, toll free access (e.g., 800/877/888), 900 access, and their successors or similar Switched Exchange Access Services. The Parties have been unable to agree as to whether "Voice-Over-Internet Protocol" transmissions ("VOIP") which cross LATA boundaries constitute Switched Access Traffic. Notwithstanding the foregoing, and without waiving any rights with respect to either Party's position as to

the jurisdictional nature of VOIP, the Parties agree to abide by any effective and applicable FCC rules and orders regarding the nature of such traffic and the compensation payable by the Parties for such traffic, if any.

- When one Party's end office switch, subtending the other Party's Access Tandem 5.8.2 switch for receipt or delivery of switched access traffic, provides an access service connection to or from an interexchange carrier ("IXC") by either a direct trunk group to the IXC utilizing the other Party's facilities, or via the other Party's tandem switch, each Party will provide its own access services to the IXC and bill on a multi-bill, multi-tariff meet-point basis. Each Party will bill its own access services rates to the IXC with the exception of the interconnection charge. The interconnection charge will be billed by the Party providing the end office function. Each party will use the Multiple Exchange Carrier Access Billing (MECAB) guidelines to establish meet point billing for all applicable traffic. Thirty (30)-day billing periods will be employed for these arrangements. For tandem routed traffic, the tandem company agrees to provide to the Initial Billing Company as defined in MECAB, at no charge, all the switched access detail usage data, recorded at the access tandem, within no more than sixty (60) days after the recording date. The Initial Billing Company will provide the switched access summary usage data, for all originating and terminating traffic, to all Subsequent Billing Companies as defined in MECAB within 10 days of rendering the initial bill to the IXC. Each Party will notify the other when it is not feasible to meet these requirements so that the customers may be notified for any necessary revenue accrual associated with the significantly delayed recording or billing. As business requirements change, data reporting requirements may be modified as necessary.
- In the event that either Party fails to provide switched access detailed usage data to the other Party within 90 days after the recording date and the receiving Party is unable to bill and/or collect access revenues due to the sending Party's failure to provide such data within said time period, then the Party failing to send the data as specified herein shall be liable to the other Party in an amount equal to the unbillable or uncollectible revenues. Each company will provide complete documentation to the other to substantiate any claim of unbillable access revenues and a negotiated settlement will be agreed upon between the Parties.
- Each Party will retain for a minimum period of sixty (60) days, access message detail sufficient to recreate any data which is lost or damaged by their company or any third party involved in processing or transporting data.
- 5.8.5 Each Party agrees to recreate the lost or damaged data within forty-eight (48) hours of notification by the other or by an authorized third party handling the data.
- Each Party also agrees to process the recreated data within forty-eight (48) hours of receipt at its data processing center.
- 5.8.7 All claims should be filed with the other Party within 120 days of the receipt of the date of the unbillable usage.

Version 2Q00: 6/15/00

- 5.8.8 The Initial Billing Company shall keep records of its billing activities relating to jointly-provided Intrastate and Interstate access services in sufficient detail to permit the Subsequent Billing Party to, by formal or informal review or audit, to verify the accuracy and reasonableness of the jointly-provided access billing data provided by the Initial billing Party. Each Party agrees to cooperate in such formal or informal reviews or audits and further agrees to jointly review the findings of such reviews or audits in order to resolve any differences concerning the findings thereof.
- NewSouth agrees not to deliver switched access traffic to BellSouth for termination except over NewSouth ordered switched access trunks and facilities.

5.9 Transit Traffic Service

- 5.9.1 Each Party shall provide tandem switching and transport services for the other Party's Transit Traffic. Transit Traffic is traffic originating on one Party's network that is switched and/or transported by the other Party and delivered to a third party's network, or traffic originating on a third Party's network that is switched and/or transported by one Party and delivered to the other Party's network. Rates for local Transit Traffic shall be the applicable Call Transport and Termination charges as set forth in Exhibit A to this Attachment. Rates for intraLATA toll and Switched Access Transit Traffic shall be the applicable charges as set forth in the applicable Party's Commission approved Interstate or Intrastate Switched Access tariffs. Switched Access Transit Traffic presumes that one Party's switch is subtending the other Party's Access Tandem for switched access traffic to and from one Party's end users utilizing the other Party's facilities, either by direct trunks with the IXC, or via the BellSouth Access Tandem. Billing associated with all Transit Traffic shall be pursuant to Multiple Exchange Carrier Access Billing (MECAB) guidelines. Pursuant to these guidelines, the Initial Billing Company shall provide summary usage data, for all originating and terminating Transit Traffic, to all Subsequent Billing Companies. Traffic between NewSouth and Wireless Type 1 third parties shall not be treated as Transit Traffic from a routing or billing perspective. Traffic between NewSouth and Wireless Type 2A or third parties utilizing UNE-P shall not be treated as Transit Traffic from a routing or billing perspective until BellSouth and the Wireless carrier or third party utilizing UNE-P have the capability to properly meet-point-bill in accordance with MECAB guidelines.
- 5.9.2. The delivery of traffic which transits the BellSouth network and is transported to another carrier's network is excluded from any BellSouth billing guarantees and will be delivered at the rates stipulated in this Agreement to a terminating carrier. BellSouth agrees to deliver this traffic to the terminating carrier; provided, however, that NewSouth is solely responsible for negotiating and executing any appropriate contractual agreements with the terminating carrier for the receipt of this traffic through the BellSouth network. BellSouth will not be liable for any compensation to the terminating carrier or to NewSouth. Both Parties' agree to compensate the other for any charges or costs for the delivery of Local Transit Traffic to a connecting carrier on behalf of the other Party.

Additionally, the Parties agree that any billing to a third party or other telecommunications carrier under this section shall be pursuant to MECAB procedures.

6. Frame Relay Service Interconnection

- In addition to the Local Interconnection services set forth above, BellSouth will offer a network to network Interconnection arrangement between BellSouth's and NewSouth's frame relay switches as set forth below. The following provisions will apply only to Frame Relay Service and Exchange Access Frame Relay Service in those states in which NewSouth is certified and providing Frame Relay Service as a Local Exchange Carrier and where traffic is being exchanged between NewSouth and BellSouth Frame Relay Switches in the same LATA.
- The Parties agree to establish two-way Frame Relay facilities between their respective Frame Relay Switches to the mutually agreed upon Frame Relay Service point(s) of interconnection ("POI(s)") within the LATA. All POIs shall be within the same Frame Relay Network Serving Areas as defined in Section A40 of BellSouth's General Subscriber Service Tariff except as set forth in this Attachment.
- Upon the request of either Party, such interconnection will be established where BellSouth and NewSouth have Frame Relay Switches in the same LATA. Where there are multiple Frame Relay switches in one central office, an interconnection with any one of the switches will be considered an interconnection with all of the switches at that central office for purposes of routing packet traffic.
- 6.4 The Parties agree to provision local and IntraLATA Frame Relay Service and Exchange Access Frame Relay Service (both intrastate and interstate) over Frame Relay interconnection facilities between the respective Frame Relay switches and the POIs.
- 6.5 The Parties agree to assess each other reciprocal charges for the facilities that each provides to the other according to the Percent Local Circuit Use Factor (PLCU), determined as follows:
- 6.5.1 If the data packets originate and terminate in locations in the same LATA, and consistent with the local definitions of the Agreement, the traffic is considered local. Frame Relay framed packet data is transported within Virtual Circuits (VC). For the purposes of this Agreement, if all the data packets transported within a VC remain within the LATA, then consistent with the local definitions in this Agreement, the traffic on that VC is local ("Local VC").
- 6.5.2 If the originating and terminating locations of the two way packet data traffic are not in the same LATA, the traffic on that VC is interLATA ("InterLATA VC").
- 6.5.3 The PLCU is determined by dividing the total number of Local VCs, by the total number of VCs on each Frame Relay facility. To facilitate implementation,

NewSouth may determine its PLCU in aggregate, by dividing the total number of Local VCs in a given LATA by the total number VCs in that LATA. The Parties agree to renegotiate the method for determining PLCU, at BellSouth's request, and within 90 days, if BellSouth notifies NewSouth that it has found that this method does not adequately represent the PLCU.

- 6.5.4 If there are no VCs on a facility when it is billed, the PLCU will be zero.
- 6.5.5 BellSouth will provide the circuit between the Parties' respective Frame Relay Switches. The Parties will be compensated as follows: BellSouth will invoice, and NewSouth will pay, the total non-recurring and recurring charges for the circuit based upon the rates set forth in BellSouth's Interstate Access Tariff, FCC No. 1. NewSouth will then invoice, and BellSouth will pay, an amount calculated by multiplying the BellSouth billed charges for the circuit by one-half of NewSouth's PLCU.
- The Parties agree to compensate each other for Frame Relay network-to-network interface (NNI) ports based upon the NNI rates set forth in BellSouth's Interstate Access Tariff, FCC No. 1. Compensation for each pair of NNI ports will be calculated as follows: BellSouth will invoice, and NewSouth will pay, the total non-recurring and recurring charges for the NNI port. NewSouth will then invoice, and BellSouth will pay, an amount calculated by multiplying the BellSouth billed non-recurring and recurring charges for the NNI port by NewSouth's PLCU.
- Each Party agrees that there will be no charges to the other Party for its own subscriber's Permanent Virtual Circuit (PVC) rate elements for the local PVC segment from its Frame Relay switch to its own subscriber's premises. PVC rate elements include the Data Link Connection Identifier (DLCI) and Committed Information Rate (CIR).
- For the PVC segment between the NewSouth and BellSouth Frame Relay switches, compensation for the PVC charges is based upon the rates in BellSouth's Interstate Access Tariff, FCC No. 1.
- 6.9 Compensation for PVC rate elements will be calculated as follows:
- If NewSouth orders a VC connection between a BellSouth subscriber's PVC segment and a PVC segment from the BellSouth Frame Relay switch to the NewSouth Frame Relay switch, BellSouth will invoice, and NewSouth will pay, the total non-recurring and recurring PVC charges for the PVC segment between the BellSouth and NewSouth Frame Relay switches. If the VC is a Local VC, NewSouth will then invoice and BellSouth will pay, the total nonrecurring and recurring PVC charges billed for that segment. If the VC is not local, no compensation will be paid to NewSouth for the PVC segment.
- 6.9.2 If BellSouth orders a Local VC connection between a NewSouth subscriber's PVC segment and a PVC segment from the NewSouth Frame Relay switch to the

BellSouth Frame Relay switch, BellSouth will invoice, and NewSouth will pay, the total non-recurring and recurring PVC and CIR charges for the PVC segment between the BellSouth and NewSouth Frame Relay switches. If the VC is a Local VC, NewSouth will then invoice and BellSouth will pay the total non-recurring and recurring PVC and CIR charges billed for that segment. If the VC is not local, no compensation will be paid to NewSouth for the PVC segment.

- 6.9.3 The Parties agree to compensate each other for requests to change a PVC segment or PVC service order record, according to the Feature Change charge as set forth in the BellSouth access tariff BellSouth Tariff FCC No, 1.
- 6.9.4 If NewSouth requests a change, BellSouth will invoice and NewSouth will pay a Feature Change charge for each affected PVC segment.
- 6.9.4.1 If BellSouth requests a change to a Local VC, NewSouth will invoice and BellSouth will pay a Feature Change charge for each affected PVC segment.
- 6.9.5 The Parties agree to limit the sum of the CIR for the VCs on a DS1 NNI port to not more than three times the port speed, or not more than six times the port speed on a DS3 NNI port.
- 6.9.6 Except as expressly provided herein, this Agreement does not address or alter in any way either Party's provision of Exchange Access Frame Relay Service or interLATA Frame Relay Service. All charges by each Party to the other for carriage of Exchange Access Frame Relay Service or interLATA Frame Relay Service are included in the BellSouth access tariff BellSouth Tariff FCC No. 1.
- NewSouth will identify and report quarterly to BellSouth the PLCU of the Frame Relay facilities it uses, per section 6.5.3 above.
- Either Party may request a review or audit of the various service components, consistent with the provisions of section E2 of the BellSouth State Access Services tariffs or Section 2 of the BellSouth FCC No.1 Tariff.
- If during the term of this Agreement, BellSouth obtains authority to provide interLATA Frame Relay in any State, the Parties agree to renegotiate this arrangement for the exchange of Frame Relay Service Traffic within one hundred eighty (180) days of the date BellSouth receives interLATA authority. In the event the Parties fail to renegotiate this Section 6 within the one hundred eighty-day period, they will submit this matter to the appropriate State commission(s) for resolution.

7. Operational Support Systems (OSS)

The terms, conditions and rates for OSS are as set forth in General Terms of this Agreement.

BELLSOUTHWEWSOUTH RATES LOCAL INTERCONNECTION

PEOCHASIA						RATES BY STATE	ATE			
LOCAL INTERCONNECTION (CALL TRANSPORT AND TERMINATIONS		¥	런	Ą	\$	1	NA.	2		
End Office Switching, per mou	HAF							١	200	Z
Direct Local Interconnection, per mou (same as End Office Substitution in Eli	¥	\$0.0018	¥	\$0.0016333	\$0.002562	\$0.00210	£0 000 3771	60.0046	2000000	
Tandem Switching, per mou		ž	\$0.005	₹	¥	ž	Y.	414	CEZEI MO CA	6100.03
Tandem Local Interconnection, ner may /inchartee and office	¥	\$0.00063	\$0.00029	\$0.0006757	\$0.001096	80000	\$0,000,024	200	VV.	¥
Multiple Tearlest Switching new form (1997)	٧×	ž	\$0.00325	¥	٧N	AM		00000	*U.UU0043	\$0.000676
I cost informaction	٧×	ž	\$0.00125	NA NA	₹N		¥ :	ž	٧V	٧×
Total market in the control (appear to the control of the control	N/A	ž	\$0.00125	1		5	¥Z	¥	¥Z	AN
Lancett imermediary Charge, per mou"	ž	\$0.0015	NAME OF THE PERSON OF THE PERS		¥.	ž	¥	¥	¥	¥
(Trins charge is applicable only to transit traffic and is applied in addition to			5	5	90.00.00	¥	¥	٧×	¥X	¥
Taylor Autoca										
Thomas crawdee - For trumps not exploct to bill and keep										
Inferim chames both non-required and analysis										
main or was believed by the second control of the second control o								-		
At such time so Ballowsh America of the Country of				-					-	
from movement the Deutse about 100 miles and 100 miles for such interconnecting										
The sort rates and shell the control of the such final cost										
Installed with this Attachment.										
Installiation I funk Side Service - per DS0										
NHC - 18t	TOPL	00000	27 2000							
NHC - Add	100	2000	250.43	\$333.28	\$334.09	\$334.94	\$334.11	£333 54	£235 14	6000
	#	6.00	\$57.38	\$56.84	\$57.12	86.95\$	\$56.98	\$58 AR	\$57 to	27.53
INTEROFFICE TRANSPORT - For transport not subject to bill and leave									0	10.75
Common (Shared) Transport										
Common (Shared) Transment nee mile nee man										
Common (Shared) Tracement Equilibre To Live	¥	\$0.00001	\$0.000012	\$0.00000	\$0.0000049	\$0 0000000	\$0,00000	10000		
Intervalled Charles I tampout recauses I emmination per mou	N/A	\$0.00045	\$0,000	-	\$0,000,00	1	10000000	\$0.0001	\$0.0000121	\$0.00004
Inter-Add Charles I rensport - Dedicated - VG				-	90.000		\$0.0004281	\$0.00034	\$0.0004672	\$0.00036
O WILL STATEMENT - DEDICATE - Z-WIFE VG										
Z-ware VG - Der mile per month	11.5XF	Sh Cristo	2000	20000						
Z-Wire VG - Facility Termination per month	11 5XF	\$19.40	2000	20.000	ž	20.0384	٧×	\$0.0282	\$0.0373	\$0.0173
NHC - 2-wire VG - Facility Termination -1st	11 5XE	6107 44	X0.0X	/0//	ž	\$19.10	٧×	\$18.01	\$21.42	£18.33
INRC - 2-whe VG - Facility Termination - Add1	11 SXF	648.03	90.10	2/8/61	ž	\$76.20	¥	\$137.48	\$136.44	\$55.30
INRC 2-wire VG -Facility Termination - Disconnect Charge -1st	11 KYE	227.55	3 3	80.05	ž	\$34.54	٧V	\$52.58	\$51.37	\$17.27
NRC - 2-whe VG - Facility Termination - Disconnect Charge - Addr	11 KYE	200	5 2	≨:	¥	\$28.03	٧×	≨	ž	20708
NRC - Manual Svc Order, per LSR	SOWAN	8 1	\$12.78	≨.	¥	\$5.37	٧×	¥	×	6361
NRC - Manual Svc Order, per LSR disconnect	NAMOS	5 3	8 2	≨ :	¥	ž	٧	ž	ž	8100
NRC - Electronic Svc Order, per LSR	SOME	5 5	2	ž	ž	ž	×	¥	ž	A N
NRC - Electronic Svc Order, per LSR disconnect	SOME	80.32	e s	22.52	ž	\$3.50	٧V	\$3.50	03 63	63.50
NRC - 2-wire VG - Incremental Charge - Manual Svc Order - 1st	NAMA	50.00	7.47	ž	ž	ž	VY	٧×	¥	NA.
NRC - 2-wire VG - Incremental Charge-Manual Svc Order - Addr	SOUTH	157.75	ž	\$18.94	V	\$18.14	¥	\$38.07	530 63	2
NRC - 2-wire VG - Incremental Charge—Manual Svc Order-Disconnect - 1-d	NAMO	/6/24	ž	\$18.94	¥	\$18.14	٧×	\$38.07	539 63	V V
NRC - 2-wire VG - Incremental Charge-Manual Syc Onde-Decorated, Address	SON SON	\$12.9/	ž	ž	¥	\$8.06	ž	ž	₹N	5
	SCHAN	\$12.97	ž	¥	٧×	90.83	≨	¥X		\
Interoffice Transport - Dedicated - 2 Wire VG - Kentucky & Mississippi		+	1						- L	4
2-Wire VG - per mile per month	1 KNC		+							T
2-Wire VG - Facility Termination per month	100	5	≨	ž	\$0.0301	¥	\$0.0323	ž	4N	1
NRC - 2-wire VG - Fecility Termination -1st	100	ž	ž	٧V	\$27.66	≨	\$21.33	ž	42	2
NRC - 2-wire VG - Facility Termination - Add1	75.	≨	ž	Y.	\$142.31	≨	\$106.72	ž	Y.	¥ 2
INRC - 2-with VG -Facility Termination - Disconnect Charge -1st	TANCE OF THE PARTY	≨ :	¥	٧×	\$58.21	٧×	\$48.83	ž	NA.	5 5
Version acad ossession	1L5NF	_	٧×	V.	¥	¥Z	\$38.05	¥	¥ 2	٤:
									- S	۲ ۲

Attachment 3
Exhibit A
Rates - Page 2

BELLSOUTHNEWSOUTH RATES LOCAL INTERCONNECTION

Description		+				RATES BY STATE	rate			
NAC-2-wire VG - Earling Termination Pa	USOC	٧L	F	8	ž	-	92	-		
NPC - Married Sec Order Sec 180	1L5NF	¥	ž	ž	Ž	5 2	200	2	သင	N.
NAC. Went Co. Co.	SOMAN	¥	ž	ž	\$10.00		27.76	ž	ž	Ν
NBC Flexible St. O. O	SOMAN	¥Z	ž	¥N	NAN AN	5	ž	¥	¥	٧¥
NOT ENAMED OF CASE OF THE PARTY	SOMEC	ž	ž	YZ	5 65	5	ž	¥	ž	Ϋ́
NBC 3 min VO 1	SOMEC	ž	ž	¥	3 2	5	25.50	ž	ž	¥
AND A WAY INCIDENTIAL Charge Manual Svc Order - 1st	SOMAN	ž	Ž	T	٤	Ž:	ž	ž	٧×	¥
INTIC - 2-Wife VG - Incremental Charge-Manual Svc Order - Add'i	SOMAN	¥2	2	٤	YZ:	ž	\$25.52	¥	٧×	¥Ν
INFC - 2-wire VG - Incremental Charge-Manual Svc Order-Disconnect1st	SOMAN	1	5	٤:	ž	ž	\$25.52	٧	٧X	Ž
NHC - 2-wire VG - Incremental Charge-Manual Svc Order-Disconnect-Add1	SOMAN	Ž		ž	ž	ž	\$11.34	Ϋ́	¥X	¥
		5	٤	ž	¥ Z	٧×	\$11.34	Ϋ́	ž	ž
Imeroffice Transport - Dedicated - 050 - 5664 KBPS										
DSO - per mile per month	11 8 YOK	60 0330	1000							
DS0 - Facility Termination per month		20.00	8000	\$0.0222	٧	\$0.0384	ž	\$0.0282	\$0,007.2	60 4 3 20
NRC - DS0 - Facility Termination - 1st	TL3XK	\$17.81	\$19.31	\$16.45	¥	\$18.37	ž	\$17.40	\$20.24	\$0.1/S
NPC - DS0 - Facility Termination - Adm	XX	\$107.11	SB1.1	\$79.61	¥Z	\$76.20	ž	\$137.49	136.13	\$17.74
NRC - DS0 -Facility Termination - Discount Charge 122	TSXK	54827	\$54.83	\$36.08	٧X	\$34.54	ΨV	6K2 KB	20.00	S20.00
MRC - DSO - Facility Termination - Discourage - 18	LSX	\$37.16	\$31.01	≨	ž	\$28.03	¥N.	A14	15.15	\$17.37
MRC - Married Syc Order part Sp	1.5×K	\$5.88	\$12.78	ž	ž	\$5.37	2	2 3	Š.	\$27.96
Mac. Magnet Sup Octor 150 4	SOMAN	٧×	\$21.58	ž	¥2	N.		٤	42	\$3.51
MBC - Flections Section 1	SOMAN	٧×	£3.84	ž	¥2	42	£ :	ž	ž	\$19.99
NAC ELECTRIC SVC COOK, Der LINE	SOMEC	\$3.50	\$2.77	23 50	42	5 65	Ž	¥N.	¥	¥
AND DOS TOO DOS TO THE CONTROL OF THE CONTROL	SOMEC	٧×	\$0.42	₹N		8:2	٤	\$3.50	\$3.50	\$3.50
MINO NOS Incremental Charge - Manual Svc Order - 1st	SOMAN	\$27.37	ž	\$18 04	42	5	ž	¥	ž	¥
Inno-Loui Indemental Charge-Manual Svc Order - Addi	SOMAN	\$27.37	ž	2180	4	10.14	¥2	\$38.07	\$39.63	٧×
INNC - USU -Indemental ChargeManual Svc Order-Disconnect1st	SOMAN	\$12.97	2	1	٤	\$18.14	ž	\$38.07	\$39.63	¥
INFC - DS0 -incremental Charge Manual Svc Order-Discornect Add1	SOMAN	\$12.07	2		٤	33.08	¥	ž	Ϋ́	٧V
			5	٤	٤	88.08	≨	Ϋ́	٧×	¥
Interoffice Transport , Darkestal, Den sess vonc										
DSO - Der mile per month										
DS0 - Facility Termination ner month	¥KCT.	ž	ž	ž	\$0.0301	ž	\$0.0323	ž	ΨN	
NPC - DSO - Facility Termination - 1st	TCDNK	≨	≨	٧V	\$26.95	ž	250 64	¥	42	¥2.
NRC - DSO - Facility Termination - Add1	TLSNK.	ž	≨	Ź	\$142.31	ž	\$106.72	ž	42	5
NPC - DS0 -Facility Termination - Discoursed Chams - 144	ANC.	ž	ž	ž	\$56.21	ž	\$48.83	ž	4 2	5
NRC - DS0 - Facility Termination - Disconnect Chams - Arter	LONK	ž	ž	ž	¥	¥	\$38.05	ž	¥X	
NRC - Manual Svc Order, per LSR	TEST STATES	ž	≨ :	ž	¥	٧×	\$7.23	٧×	ž	¥
NRC - Manual Svd Order, per LSR disconnect	NAMOS	X	ž	ž	\$19.99	¥	٧×	ž	¥	¥
INRC - Electronic Svc Order, per LSR	CONCO	۲.	٤	ž	ž	ž	¥	¥	¥	¥2
NRC - Electronic Svc Order, per LSR disconnect	SOME	5	Š	ž	23.50	ž	\$3.50	٧¥	¥	V.
INRC - DS0 -Incremental Charge-Manual Svc Order - 1st	STATE OF THE PARTY	٤	Ž	¥	ž	ž	٧V	¥	¥	¥X
NRC -DS0 - Incremental Charge Manual Svc Order - Add1	NAME OF THE PERSON	Ž	ž	ž	\$37.21	¥	\$25.52	٧×	٧×	¥
NRC - DS0 -Incremental Charge—Manual Svc Order-Disconnect1et	SOUTH	≨ :	ž	ž	\$37.21	٧¥	\$25.52	ž	Ϋ́	¥Z
NRC - DS0 -Incremental ChargeManual Svc Order-Disconnect Arti	SOM SO	٤	ž	ž	ž	٧×	\$11.31	ž	ž	¥
	SCHOOL	ž	ž	¥	₹	٧×	\$11.34	₹	ž	¥Z
Interoffice Transport - Dedicated - DS1										
DS1 - per mile per month	11.53	60 6000	60000	2001						
DS1 -Facility Termination per month	11.50	670 60	600 70	420.43	₹.	50.7831	ž	\$0.5753	\$0.7598	\$0.3525
NRC - DS1-Facility Termination - 1st	11.5%	£100 18	87.89	3/84/	ž	\$93.40	¥	\$71.29	\$94.98	\$75.83
Version 3000:09/29/00		1 21 221	10.01	10.7416	٧	\$140.49	¥	\$217.17	\$216.27	\$145.98

Attachment 3 Exhibit A Rates - Page 3

BELLSOUTHNEWSOUTH RATES LOCAL INTERCONNECTION

						2000				
DESCRIPTION	Con	;				HAIES BY STATE	ATE			
NRC - DS1 - Facility Termination - Add1	2000	4	2	δ	Ž	2	E	Ž,	Ç	2
NRC - DS1 - Facility Termination - Disconnect Chame - 1st	TC T	\$148.18	£ 18	\$111.75	ž	\$106.69	ž	\$183.75	4169 70	2000
NRC - DS1 - Facility Termination - Discounsed Change Addition	TXCT	\$25.44	\$30.30	¥	¥Z	\$20.00	¥X.	2	100.70	5103.83
NPC - Manual Svc Order, per I SP	1.5%	\$20.42	\$26.76	¥	ž	\$18.34	T N		٤	\$19.55
NRC - Margal Syc Order per 1 CB Alexandra	SOMAN	¥	\$21.56	ž	ž	AN.	1	٤:	Ž	\$14.99
INRC - Flectionic Sec Order per 100	SOMAN	¥	\$3.84	₹	ž	4N	2 2	Ž.	¥.	\$19.99
NBC - Flectonic On Other Control	SOMEC	\$3.50	\$2.77	\$3.50	₹Z	63 63	٤	¥	Ą	¥
MDC Dot here. 12.	SOMEC	ž	\$0.42	ΨN	1	C'54	Ž	\$3.50	\$3.50	\$3.50
Mich Do. Historical Charge—Manual SVc Order - 1st	SOMAN	\$27.37	¥X	£18 64		V.	٤	¥	Ϋ́	٧
MIND TO 1 Incremental Charge - Manual Svc Order - Add'	SOMAN	\$27.37	ž	\$18 94	2 2	218.14	¥	\$38.07	\$39.63	٧N
And Do. 1. Indemental Charge - Manual Svc Order-Disconnect - 1st	SOMAN	\$12.97	2	NA.		\$10.14	ž	\$38.07	\$39.63	٧V
NHC - USI - Incremental Charge-Manual Svc Order-Disconnect—Addri	SOMAN	\$12.97	ž	ž	2 2	90.92	ž	¥.	٧×	¥Χ
Interoffice Transport - Dedicated - DS1 - Kentucky & Missississ						810	٤	¥	₹	¥
DS1 - per mile per month										
DS1 -Facility Termination per month	1 SNL	ž	¥	¥	\$0.4500	ž	\$0.6598	4 2		
NRC - DS1-Facility Termination - 1st	1.5ML	ž	ž	¥	\$55.05	٧×	\$74.40	2	٤	\$
NRC - DS1 - Facility Termination - Addi	אכן:	¥	≨	ž	\$298.18	¥	\$196.28	ž	2	4
NRC - DS1 - Facility Termination - Disconnect Charge - 1st	TING!	Ž	Ş	¥	\$231.23	ΝA	\$147.31	ž	¥2	2 2
NRC - DS1 - Facility Termination - Disconnect Charge - Addr		5	ž	ž	ž	¥	\$26.56	ž	¥Z	N N
NRC - Manual Svc Order, per LSR	SOMAN	2	ž	¥	ž	¥	\$21.61	ž	¥	₹ N
NRC - Menual Svc Order, per LSR disconnect	SOMAN	5 5	<u> </u>	ž	\$19.99	ž	Ş	٧×	ž	AN
NRC - Electronic Svc Order, per LSR	SOME	414	ž	≨ :	¥	¥	٧V	ž	¥	Y.
NRC - Electronic Svc Order, per LSR deconnect	SOME	4	4	ž	\$3.50	ž	\$3.50	¥X	¥	Y.
NRC - DS1 - Incremental Charge-Manual Svc Order - 1st	SOMAN	S Y	٤	٤	ž	٧	¥	٧	ž	¥
NRC-DS1 - Incremental Charge-Manual Svc Order - Add1	SOMAN	Ž	5 2	£ .	ž:	¥	\$25.52	٧	¥	ž
NHC - DS1 - Incremental Charge-Manual Svc Order-Disconnect-1st	SOMAN	¥	\$ 2	2 2	¥ •	≨	\$25.52	٧×	NA	ΑN
NHC - DS1 - Incremental Charge - Manuel Svc Order-Disconnect - Add1	SOMAN	2	ž	5 2	Ž.	ž	\$11.31	¥	٧×	¥
				5	Š	ž	\$11.34	ž	ž	¥
Interornical Transport - Dedicated - DS3										
CSS T are per mortin	1L5XM	2	21.12	800	414					
USS -Ficility emination per month	1L5XM	\$898.15	\$1.121.83	\$788 OO	5 2	50.4	¥	\$12.98	\$8.13	\$5.89
MICCOSS - Facility (emination - 1st	1L5XM	\$511.77	\$557.69	\$511.10	5 2	101,10	ž	\$720.38	\$967.70	\$760.20
NBC OCS FINE TO THE DESCRIPTION OF THE DESCRIPTION	1L5XM	\$330.92	\$325.61	\$330.77	¥Z	\$304 OU	¥ 4	3.	\$808.72	\$625.91
NBC - DC3 - Facility Lettingston - Decorated Charge - 1st	1L5XM	\$121.72	\$111.58	\$122.31	ž	\$10018		00/8/00	\$423.45	\$311.38
INRC - Manual Sec Order, part 158	1L5XM	\$118.54	\$108.34	\$119.14	ž	\$99.46	¥X	2	ž	\$103.36
NRC - Marual Svc Order, per I SR discounsed	SOMAN	Ş	\$21.58	₹	¥	¥	¥	ž	5 2	8100.39
NRC - Electronic Svc Order, per LSR	SONAN	ž	5 3.84	ž	¥	¥X	٧×	¥	4 2	4 13.33
NRC - Electronic Svc Order, per LSR disconnect	SOME	3	\$2.77	\$3.50	٧×	\$3.50	ž	\$3.50	\$3.50	63.50
NRC - DS3 - Incremental Charge - Marrual Syc Order - 1st	SOME	¥ 6	20.42	ž	٧×	¥	NA	٧×	2	AN
NPC - DS3 - Incremental Charge - Manual Svc Order - Add7	COLLAN	28.48	≨ :	\$37.55	¥Z	\$50.25	¥Ν	\$91.28	\$54.26	Ž
NRC - DS3 - Incremental Charge Manual Svc Order-Disconnect 1st	SOMAN	\$10.03	٤	66.72	ž	\$50.25	٧×	\$91.28	\$54.26	¥X
NRC - DS3 - Incremental Charge - Manual Svc Order-Disconnect - Add1	SOMAN	\$19.03	§ 2	\$10.03	ž	\$20.94	¥Z	¥	¥	¥X
			i	33.0	ξ.	\$50.9E	¥	YN.	¥	¥
Intercrice Transport - Dedicated - DS3 - Kentucky & Mississippi										
Door - Per miss per month	1LSNM	ž	ž	¥	\$12.62	¥Z	415.00	AIN		
COST COMPY TETHINGHOUT DOT MOTHER	1L5NM	ž	ž	ž	\$1.204	ž	87.44.20	5	Y.	¥.
Version 3000-09/29/00							B	<u> </u>	~ Z	- V

Version 3000:09/29/00

BELLSOUTHMEWSOUTH RATES LOCAL INTERCONNECTION

Attachment 3 Exhibit A Rates - Page 4

\$19.02 \$24.16 \$19.99 ZZ 2 2 8 8 ¥ × N A A A ž ž ž ž YZZZ \$24.83 \$55.52 \$5.51 \$19.99 NA \$3.50 \$277.35 \$233.26 \$33.18 ž ž ž ž ž ž \$18.05 \$562.46 \$91.57 NA NA S3.50 83.50 843.64 813.55 \$37.20 \$534.81 \$462.81 NA \$553.80 \$89.69 S Z Z Z Z \$35.68 \$534.48 \$462.69 NA ZZZZZZ \$42.17 \$125.56 \$118.79 NA NA \$3.50 NA \$3.50 \$64.97 \$27.08 \$17.83 \$84.36 \$77.69 \$8.95 NA NA \$3.50 \$3.50 \$11.34 \$686.7 \$19.03 \$495.25 \$86.56 \$78.58 \$9.84 NA NA \$3.50 ž \$16.05 \$38.91 \$494.83 \$435.28 \$46.85 \$11.34 ž RATES BY STATE \$14.94 \$59.75 \$59.75 \$53.68 \$6.60 NA NA \$3.50 NA \$3.50 NA \$3.50 \$11.40 \$352.75 \$61.33 \$300.30 \$24.15 \$21.31 \$346.23 \$516.89 NA NA \$3.50 NA 893.12 893.12 NA NA \$585.15 \$98.53 \$11.89 NA NA NA NA NA \$23.38 \$585.15 \$98.53 \$22.28 \$538.95 \$464.94 NA X X X X X X X X \$13.91 \$382.95 8 2 2 2 2 2 2 2 2 2 2 2 2 2 \$14.99 \$54.05 \$38.36 \$356.15 NA NA \$386.34 \$66.36 #57.91 #5.82 #2.75 #5.75 #5.75 NA NA \$387.21 \$6.72 \$6.73 \$6.73 \$1.58 \$2.75 대원된원원원원원원원원원 \$242.45 \$226.44 \$41.13 ≨≨ \$14.61 \$494.65 \$77.81 \$77.63 \$15.77 \$502.43 \$86.68 \$78.71 \$8.53 NA NA NA NA S3.50 \$3.50 \$3.50 \$18.73 \$3.50 NA \$27.37 \$18.37 \$35.52 \$503.57 \$442.84 \$46.28 **422222222** ⊀ા≨ ≨ž 1LENIM 1L TEFV2
TEFV2
TEFV2
TEFV2
TEFV2
SOMAN SOMAN SOMEC SOMAN SOMAN TEFV4
TEFV4
TEFV4
TEFV4
TEFV4
SOMAN
SOMAN
SOMAN TEPHG TEPHG TEPHG TEFHG SOMAN NRC - 2-wire VG - Facility Termination - 1st
NRC - 2-wire VG - Facility Termination - Add1
NRC - 2-wire VG - Facility Termination - Disconnect - 1st
NRC - 2-wire VG - Facility Termination - Disconnect - Add1
NRC - Aminal Svc Order, per LSR disconnect
NRC - Manual Svc Order, per LSR disconnect
NRC - Electronic Svc Order, per LSR disconnect
NRC - Electronic Svc Order, per LSR disconnect
NRC - Electronic Svc Order, per LSR disconnect
NRC - 2-wire VG - Incremental Charge - Manual Svc Order - Add1
NRC - 2-wire VG - Incremental Charge - Manual Svc Order - Add1
NRC - 2-wire VG - Incremental Charge - Manual Svc Order - Add1
NRC - 2-wire VG - Incremental Cost - Manual Svc Order - Add1 NRC - DS3 - Facility Termination - 1st
NRC - DS3 - Facility Termination - Add1
NRC - DS3 - Facility Termination - Discornect Charge - 1st
NRC - DS3 - Facility Termination - Discornect Charge - Add1
NRC - Manual Svc Order, per LSR
NRC - Manual Svc Order, per LSR
NRC - Electronic Svc Order, per LSR
NRC - Electronic Svc Order, per LSR
NRC - Electronic Svc Order, per LSR
NRC - Electronic Svc Order, per LSR
NRC - DS3 - Incremental Charge - Manual Svc Order - 1st
NRC - DS3 - Incremental Charge - Manual Svc Order - Add1
NRC - DS3 - Incremental Charge - Manual Svc Order-Discornect - 1st
NRC - DS3 - Incremental Charge - Manual Svc Order-Discornect - 1st
NRC - DS3 - Incremental Charge - Manual Svc Order-Discornect - 1st NRC - 4-Wire VG - Discorrect Chg - 1st
NRC - 4-Wire VG - Discorrect Chg - Add1
NRC - Manual Svc Order, per LSR
NRC - Bectroric Svc Order, per LSR
NRC - Electroric Svc Order, per LSR
NRC - Electroric Svc Order, per LSR
NRC - Electroric Svc Order, per LSR
NRC - Electroric Svc Order, per LSR
NRC - 4-Wire VG - Incremental Charge--Manual Svc Order - 1st
NRC - 4-Wire VG - Incremental Charge--Manual Svc Order - Add1 Local Channel - Dedicated - 2-Wire VG Monthly Recurring per month Local Channel - Dedicated - 4-Wire VG DS1 Monthly Recurring per month NRC - DS1 - 1st NRC - DS1 - Add1 NRC - DS1 - Disconnect Crg - 1st NRC - DS1 - Disconnect Crg - 1st Monthly Recurring per month NRC - 4-Wire VG - 1st NRC - 4-Wire VG - AddT ocal Channel - Dedicated - DS1 Local Channel - Declicated DESCRIPTION

Version 3000:09/29/00

BELLSOUTHANEWSOUTH RATES LOCAL INTERCONNECTION

Affachment 3 Exhibit A Rates - Page 5

\$726.16 \$411.64 \$103.36 \$100.59 NA \$3.50 \$50.31 \$12.61 \$12.61 \$19.99 NA NA NA \$185.94 \$21.71 \$10.46 \$14.21 \$1.46 YZZZZ * * * * * * * * \$735.42 \$498.58 \$234.30 \$11.99 \$12.05 \$8.68 NA NA \$25.59 \$8.92 NA NA 22222 \$562.25 \$527.88 \$56.25 NA NA \$498.87 \$77.90 \$77.90 \$43.32 \$15.78 \$11.78 NA NA NA NA NA NA NA NA S3.50 \$1.82 \$1.82 \$1.82 \$1.82 \$1.82 \$1.82 22222 \$533.33 \$526.67 \$493.71 \$40.87 NA NA \$3.50 NA \$31.49 \$31.49 \$25.35 \$247.40 \$79.94 \$65.20 \$5.58 \$11.36 ž RATES BY STATE \$594.71 \$396.54 \$113.75 \$110.80 NA NA NA NA NA NA NA NA S50.25 \$20.94 \$20.94 \$182.64 \$50.46 \$7.55 \$12.29 \$18.00 NA NA S12.73 \$12.43 NA Z Z Z Z Z Z \$1,091.00 \$661.23 NA NA NA NA NA S19.99 \$19.99 NA \$3.50 NA S87.71 \$3.50 \$93.12 NA NA \$236.32 \$425.41 \$303.33 NA NA NA NA NA NA NA ž \$12.31 \$119.14 NA NA NA NA NA S37.55 \$37.65 \$37.65 \$37.65 \$318.03 \$550.01 \$639.50 A X3.50 XX XX \$12.02 \$11.02 \$11.02 \$12.02 \$12.02 \$13.50 \$13.50 \$13.50 \$10.82 \$10.82 **\$**\$\$\$\$\$ \$570.08 \$903.37 \$528.05 \$154.90 \$21.56 \$21.56 \$21.56 \$21.56 \$3.84 NA \$21.56 \$2.75 \$0.42 \$250.97 \$188.00 \$188.00 \$14.40 \$13.16 \$21.56 \$13.16 **22222** \$541.78 \$640.54 \$426.28 \$121.72 \$118.54 NA NA NA NA NA S3.50 NA S3.50 S18.48 \$19.03 \$19.03 \$188.51 \$11.67 **22222** SOMAN SOMEC SOMEC SOMAN SOMAN TEFHU
TEFHU
TEFHU
TEFHU
SOMAN
SOMAN
SOMAN
SOMAN
SOMAN
SOMAN SATICS SATICS SATICS SATICS SATICS SATICS SATICS SAMAN SOMAN SOMAN SOMAN CSOC SATINS SATINS SATINS SATINS SATINS NRC - Manual Svc Order, per LSR
NRC - Manual Svc Order, per LSR decorrect
NRC - Electronic Svc Order, per LSR decorrect
NRC - Electronic Svc Order, per LSR disconnect
Charnel System - Incremental Cost - Manual Svc. Order vs. Electronic - 1st
Charnel System - Incremental Cost - Manual Svc. Order vs. Electronic - 1st NRC - Electronic Svc Order, per LSR
NRC - Electronic Svc Order, per LSR disconnect
NRC - DS3 - Incremental Charge - Manual Svc Order - 1st
NRC - DS3 - Incremental Charge - Manual Svc Order - 1st
NRC - DS3 - Incremental Charge - Manual Svc Order - AddT
NRC - DS3 - Incremental Charge - Manual Svc Order - Disconnect - 1st
NRC - DS3 - Incremental Charge - Manual Svc Order - Disconnect - AddT NRC - Manual Svc Order, per LSR discornect
NRC - Electronic Svc Order, per LSR
NRC - Electronic Svc Order, per LSR
NRC - DSI - Incremental Charge - Manual Svc Order - 1st
NRC - DSI - Incremental Charge - Manual Svc Order - 1st
NRC - DSI - Incremental Charge - Manual Svc Order - 1st DSS Channelization (DSS to DS1) - Kentucky & Mississippi per Channelized System (28 DS1) per month NRC - 1st NRC - Ad1 DSS - Facility Termination per month
NRC - DSS - Facility Termination - 1st
NRC - DSS - Facility Termination - Add1
NRC - DSS - Facility Termination - Disconnect - 1st
NRC - DSS - Facility Termination - Disconnect - Add1
NRC - Manual Svc Order, per LSR
NRC - Manual Svc Order, per LSR CHANNELIZATION
DS3 Channelization (DS3 to DS1)
per Channelizad System (28 DS1) per month
NRC - 1st NRC - Manual Svc Order, per LSR NRC - Add1
NRC -1st - Disconnect
NRC -Add1 - Disconnect
Det interlece per month (COCI) Local Channel - Dedicated - D83 NRC -1st - Disconnect
NRC -Add1 - Disconnect
per Interface per month (COC) NRC - Add'I NRC - 1st DESCRIPTION

Attachmeni 3 Exhibit A Rates - Page 6

						RATES BY STATE	ATE			
NO. CON	nsoc	*	•	2	;			-		
NO.	SATCO	YZ.	12	5	2	3	MS	Š	သင	ž
I CON	SATCO	72	5	٤	\$19.99	ž	¥	ž	ΨN	A V
ADD M	SOMAN	Ž	\	ž	ž	ž	¥	¥	¥	42
NHC - Marriel Svc Order, per LSR disconnect	SOMAN	2	¥ :	ž	83.50	¥	\$3.50	ž	AN N	5
MHC - Electronic Svc Order, per LSR	SOMEC	5	٤	ž	¥	¥	ž	¥	· N	2
NHC - Electronic Svc Order, per LSR disconnect	-	£ :	₹	¥	٧	¥	\$28.95	72		4
Chairmel System - Incremental Cost - Manual Svc. Order vs. Electronic - 1st		¥ :	≨	ž	٧×	Ž	51.08	12		Y.
Chennel System - Indemental Cost - Manual Svc. Order vs. Electronic - Artri	TOTAL STATE OF THE PARTY OF THE	٤	ž	¥Ζ	\$41.47	ž	\$16.97	2	¥ :	Ž
	1	٤	≨	≨	\$11.99	₹	NA	1	2	Ž.
DS1 Channelization (DS1 to DS0)								2	¥2	ž
per Channelized System (24 DS0) per month										
NRC - 1st	SAIC	\$136.82	\$153.60	\$126.22	ž	\$209.87	NAME OF THE PERSON OF THE PERS	1		
NRC - Add1	SATCI	\$197.98	\$182.14	\$198.22	ž	\$193.63	2	277718	\$147.51	\$165.21
NRC - 1sr - Disconnect	SATCI	\$123.12	\$125.18	\$123.59	ž	\$118.37	2 2	20/18	\$220.89	\$197.21
NRC -Add! - Disconnect	SAICI	\$30.18	\$19.52	\$31.03	ž	226 44	42	2010	\$137.15	\$119.99
- Interface (COCI)	SATCI	\$18.88	\$18.14	\$19.75	ž	\$16.83		2	¥.	\$25.66
per OCU-DP(data) card per morth (2 4-84khe)								×1.14	ž	\$15.81
NRC - 1st	SATSA	21 .66	\$2.20	\$1.86	¥	63.13	Į.			
NRC - Add1	SATSA	\$12.05	\$13.18	\$12.02	ž	\$12.20		S X	25.34	\$2.46
per BRITE card per month	SATSA	88 GS	\$9.43	\$8.66	¥N.	S B		\$15.78	\$12.05	\$12.61
NRC - 1st	SAISA	2341	\$3.83	\$3.71	¥	2 18	42	27.1.50	88.83	\$9.03
NRC - Add1	SAISA	\$12.05	\$13.18	\$12.02	٧N	\$12.20		2	12.21	\$3.33
per VG card per month (DS0)	SATSA	\$8.69	\$9.43	\$8.66	¥	8	١	\$10.76	\$12.05	\$12.61
NAC - 1st	SATSA	8 9.08	\$1.45	\$1.17	¥	21.63		87118	39 33	\$9.03
NRC - Add1	SAISA	\$12.05	\$13.16	\$12.02	¥	\$12.20	Y N	8.10	51.47	\$1.25
NRC - Manual Svc Order, per LSB	VS IVS	89.68	\$9.43	\$8.66	ž	88	NA.	23.70	\$12.05	\$12.61
NRC - Manual Svc Order, per LSR disconnect	SOMAN	≨.	\$21.58	¥	¥X	¥	VZ.	474	8	\$9.03
NRC - Electronic Svc Order, per LSR	NAME OF THE PERSON	ž	23.84	¥	AN	ž	V.	Ž	¥ .	\$19.99
NRC - Electronic Svc Order, per LSR disconnect		8	52.77	\$3.50	٧V	\$3.50	ž	\$3.50	2	AN C
Channel System - Incremental Cost - Manuel Syc. Order vs. Electronic 1st	$\frac{1}{1}$	ž	\$0.43	¥	٧V	ž	¥	42	06:22	93.50
Channel System - Incremental Cost - Manual Svc. Order vs. Electronic - Add	1	10.01	≨	\$14.75	¥	\$19.74	ž	\$28.13	425 EQ	YN S
Incremental Cost-Manual Svc. Order vs. Elect -Discormect - 1st	-	87.76	≨ :	\$6.55	ž	\$8.77	¥	\$13.33	88	00.024
Incremental Cost-Manuel Svc. Order vs. Elect -Disconnect - Add1	SOMAN	\$1 0450	≨ à	\$10.70	¥	\$12.43	٧N	\$18.26	¥	\$14.21
1 Chancelland			5	37.02	ž	ž	¥	\$1.48	¥z	\$1.48
De Commission (DS) to DS() - Remucky & Mississippi										
And the second system (24 USU) per month	SASTN1	≨	V.	₹N	10000					
NBC - AAC	SASTN1	٧V	ž	Ž	230.53	2 3	\$146.87	¥	¥	Ϋ́
NPC : 1st - Demonstra	SASTN1	ž	ž	Ž	\$184.20	\$ 2	25.11.52	≨:	¥	ΝA
NDC AACH COMME	SASTN1	₹	ž	٧×	N N	5	\$ 104.00	×.	¥	¥
- Independent Contraction	SASTN1	ž	ž	¥	¥2	ž	200.00	٧V	٧V	¥
Oer OCI DOVINION COMPANY OF THE PROPERTY OF TH					2	Ę	88.11.88	٧V	٧V	¥
NPC . 1st	SATSA	¥	ž	ž	20	AM	60.00			
NAC. Add	SATSA	≨	ž	≨	\$15.88	×2	41 A B B	ž	¥	ž
Der BRITE card nar month	SATSA	¥	¥	٧×	\$11.36	42	30.11	\$	Ž.	ž
- Ca2	SATSA	¥	ž	ž	2	1	2000	¥.	٧×	₹
2007 - Add	SATSA	¥	ž	ž	\$15.86	2	200	Ý.	∀ Z	¥
The Vis card not month (Dec)	SATSA	¥	ž	ž	\$113	¥2	611.05	YZ.	٧×	¥
NBC 1st	SATSA	٧×	¥X	ž	\$1.40	¥2	31.15	Ž	٧×	٧
EDA . CRX	SATSA	ž	٧Z	ž	\$15.86	¥2	21.43	Ž:	٧×	Ϋ́
Version 3000:092900	SATSA	ž	٧×	¥	\$11.36	Ž	\$1135	¥2	¥.	ž
							23.1.1	5	×z	ž

BELLSOUTHMEWSOUTH RATES LOCAL INTERCONNECTION

DESCRIPTION						RATES BY STATE	ATE			
NRC - Manual Svc Order, per LSR	USOC	¥	2	ď	¥	5	2	4	8	
NRC - Marual Svc Order per I SB decourse	NAMO.	ž	¥	¥	\$ 19.90	N N		2	٦	Z
NBC - Electronic Syc Order and Sp	SOMAN	ž	ž	ž	٧N	4	5	ž	¥	V.
UCI INTERPRETATION	SOMEC	ž	¥N	· ·		5	ž	¥	ž	Ϋ́Ν
Character Care Order, per LSR disconnect	SOMEC	2	2	٤	2	YN.	\$3.50	ž	¥	AN
Committee Oystem - McComental Cost - Manual Svc. Order vs. Electronic - 1st	SOMAN	12	5	٤	ž	ž	Ϋ́	ž	¥	V N
Chamel System - Incremental Cost - Manual Svc. Order vs. Electronic - Addri	NAMO	١	٤:	¥Z	\$41.47	¥	\$19.74	ž	AM	
Incremental Cost-Manuel Svc. Order vs. Elect -Discornect - 1st	S S S S S S S S S S S S S S S S S S S	٤	ž	¥	\$11.99	ž	58.77	¥N.	44	\$
Incremental Cost-Manual Svc. Order vs. Elect - Discoursed - Andre	NAME OF	≨ :	ž	¥	¥	ž	\$12.43	42		¥N.
	NAMA NAMA	≨	¥	ž	ž	V.	NA.		¥Z.	Y.
Local Interconnection Mid-span Mask							5	¥	¥	¥
Local Charges - Dedicated - Des										
DS: Month Boundary										
NAC. DOL. 14	TEFHG	\$14.84	\$17.27	415 05	100					
NBC CON A44	TEFHG	\$546.69	\$553 10	CEAR BO	87.73	\$19.93	\$16.48	\$16.58	\$15.87	\$17.90
NBC Det Description	TEFHG	\$471.61	2772	270 87	07.75	27.0	\$548.77	\$546.22	\$548.72	\$547.48
No. 10 - Uscomed Cho - 1st	TEFHG	CEL	TAN	100	10.27	\$473.34	\$271.68	\$471.20	\$473.36	\$472.20
INTO 151 - Uscomect Chg - Add1	TEFHG		200	ngi	90	180	TBD	TBO	TBD	TBO
NHC - DS1 - Incremental Cherge - Manual Svc Order - 1st	POMAN	200		GB.	180	TB0	180	180	TBO	Tal
NRC - DS1 - Incremental ChargeManual Svc Order - Add1	NAME OF THE PERSON	203.32	80/03	\$88.87	\$89.09	\$89.34	589.03	288.92	690 24	001
NRC - DS1 - Incremental Charge - Manual Syc Order-Discoursed	111100	70.55	90.0705	88.8666	89.0679	89.3388	89,0255	Cap or	2000	4 500
	COMPAN	3	TBO	TBO	TBD	780	CBT	TBO	403.54	400.14
Local Chennel - Dedicated - D&s								CO	2	180
DS3 - Facility Termination ner month										
NPC - DS3 - Eacility Tambodia 444	TEFE	\$214.60	\$245.65	\$231.61	\$275.80	CO07 11	10000			
NOC. DOS FILE T	TEFHU	\$1,066.46	\$1.079.15	\$1.064.72	\$1 067 27	41 070 00	X18.84	\$232.80	\$222.56	\$285.50
Mary - Color -	TEFHU	\$670.88	678 85 OB	00 000	10.100	81,070.38	\$1,056.63	\$1,085.55	\$1,070.42	\$1,068,00
NPC - USI - Disconnect Chg - 1st	TEFFU	E	TEC	100	20/1.45	\$6/3.34	\$670.98	\$670.31	\$673.37	\$671.85
NHC - DS1 - Disconnect Chg - Add1	TEFAL	TBC	26	200	2	180	TBD	TBD	TB0	CBT
NRC - DS3 -Incremental Charge - Manual Svc Order - 1st	SOMAN	07.702		OB.	IBO	TBD	TBO	180	180	TBD
NRC - DS3 - Incremental Charge Manual Svc Order - Add7	NAMO	0,100	10.00	5.	\$94.57	\$94.84	\$94.50	\$94.41	29.492	604 63
NRC - DS1 - Incremental Charge - Manual Svc Order-Disconnect	SOMAN	100	20.02	20.75	\$94.57	\$94.84	\$94.50	\$94.41	20.402	2010
		3	ng i	TBD	TBD	TB0	TBD	TBD	780	TBD
NOTES:			1							
		1								
If no rate is blandflad in the contrary the sate to the the sate		†								
et fort in applicable BellSouth tariff or as negotiated by the Parties upon request by either Party.										
								- Control		-
Where the state Commission has adopted rates for the rate elements contained to the										
is the intent of the parties to reflect such rates in this earthar and to south the source.		1						†	†	
consistent with applicable FCC and Commission rules and orders.										
										_

Attachment 4

Physical Collocation

BELLSOUTH PHYSICAL COLLOCATION

1. Scope of Attachment

1.1 Scope of Attachment. The rates, terms, and conditions contained within this Attachment shall only apply when NewSouth is occupying the Collocation Space as a sole occupant or as a Host within a Premises location pursuant to Section 4.

All the negotiated rates, terms and conditions set forth in this Attachment pertain to collocation and the provisioning of Collocation Space.

- Right to occupy. BellSouth shall offer to NewSouth collocation on rates, terms, and 1.2 conditions that are just, reasonable, non-discriminatory and consistent with the rules of the Federal Communications Commission ("FCC"). Subject to Section 4 of this Attachment, BellSouth hereby grants to NewSouth a right to occupy that certain area designated by BellSouth within a BellSouth Premises, of a size which is specified by NewSouth and agreed to by BellSouth (hereinafter "Collocation Space"). BellSouth Premises include BellSouth Central Offices and Serving Wire Centers, as well as all buildings or similar structures owned or leased by BellSouth that house BellSouth Network Facilities and all structures that house facilities on public rights-of-way, including but not limited to, vaults containing loop concentrators and other similar structures. To the extent this Attachment does not include all the necessary rates, terms and conditions for BellSouth Premises other than BellSouth Central Offices, the Parties will negotiate said rates, terms, and conditions at the request for collocation at BellSouth Premises other than a Central Office. Notwithstanding the foregoing, BellSouth shall consider in its designation for cageless collocation any unused space within the BellSouth Premises. The size specified by NewSouth may contemplate a request for space sufficient to accommodate NewSouth's growth within a two year period.
- 1.2.1 Space Reclamation. In the event of space exhaust within a Central Office Premises, BellSouth may include in its documentation for the Petition for Waiver filing any unused space in the Central Office Premises. NewSouth will be responsible for any justification of unused space within its space, if such justification is required by the appropriate state commission.
- Use of Space. NewSouth shall use the Collocation Space for the purposes of installing, maintaining and operating NewSouth's equipment (to include testing and monitoring equipment) necessary to interconnect with BellSouth services and facilities, including access to unbundled network elements, for the provision of telecommunications services. Pursuant to Section 5 following, NewSouth may at its option, place NewSouth-owned fiber entrance facilities to the Collocation Space. In addition to, and not in lieu of, interconnection to BellSouth services and facilities, NewSouth may connect to other interconnectors within the designated BellSouth Premises by purchasing BellSouth's Special Access Services from BellSouth's FCC

or State Access Services Tariffs. The Collocation Space may be used for no other purposes except as specifically described herein or authorized in writing by BellSouth.

- 1.4 <u>Rates and charges</u>. NewSouth agrees to pay the rates and charges identified in Exhibit A attached hereto.
- 1.5 <u>Service Coordination</u>. The Parties shall coordinate, where necessary, to ensure that the Collocation Space is provisioned in accordance with the specifications submitted by NewSouth in its Application, as affirmed by the Bona Fide Firm Order or as jointly amended thereafter. BellSouth will continue to provide the necessary infrastructure to support NewSouth's request(s) during NewSouth's occupancy of the Collocation Space.

2. Space Notification

- Availability of Space. Upon submission of an application pursuant to Section 6, BellSouth will permit NewSouth to physically collocate, pursuant to the terms of this Attachment, at any BellSouth Premises, unless BellSouth has determined that there is no space available due to space limitations or that physical collocation is not practical for technical reasons.
- 2.1.1 Unless otherwise specified, BellSouth will respond to an application within ten (10) calendar days as to whether space is available or not available within a BellSouth Premises. This interval excludes National Holidays. If the amount of space requested is not available, BellSouth will notify NewSouth of the amount of space that is available.
- 2.1.2 BellSouth will respond to a Florida Application within fifteen (15) calendar days as to whether space is available or not available within a BellSouth Premises. If the amount of space requested is not available, BellSouth will notify NewSouth of the amount of space that is available.
- 2.1.3 BellSouth will respond to a Louisiana Application within ten (10) calendar days for space availability for one (1) to ten (10) Applications; fifteen (15) calendar days for eleven (11) to twenty (20) Applications; and for more than twenty (20) Applications, it is increased by five (5) calendar days for every five additional Applications received within five (5) business days. If the amount of space requested is not available, BellSouth will notify NewSouth of the amount of space that is available.
- 2.1.4 BellSouth will respond to a Mississippi Application within ten (10) business days as to whether space is available or not available within a BellSouth Premises. If the amount of space requested is not available, BellSouth will notify NewSouth of the amount of space that is available.

- 2.2 Reporting. Upon request from NewSouth, BellSouth will provide a written report ("Space Availability Report") specifying the amount of Collocation Space available at the Premises requested, the number of collocators present at the Premises, any modifications in the use of the space since the last report on the Premises requested and the measures BellSouth is taking to make additional space available for collocation arrangements.
- 2.2.1 The request from NewSouth for a Space Availability Report must be written and must include the Premises and Common Language Location Identification ("CLLI") code of the Premises. Such information regarding Premises and CLLI code is located in the National Exchange Carriers Association (NECA) Tariff FCC No. 4.
- 2.2.2 BellSouth will respond to a request for a Space Availability Report for a particular Premises within ten (10) calendar days (in Mississippi, 10 business days) of receipt of such request. BellSouth will make best efforts to respond in ten (10) calendar (in Mississippi, 10 business days) days to such a request when the request includes from two (2) to five (5) Premises within the same state. The response time for requests of more than five (5) Premises shall be negotiated between the Parties. If BellSouth cannot meet the ten calendar day (in Mississippi, 10 business days) response time, BellSouth shall notify in writing NewSouth and inform NewSouth of the time frame under which it can respond.
- Denial of Application. After notifying NewSouth that BellSouth has no available space in the requested Premises ("Denial of Application"), BellSouth will allow NewSouth, upon request, to tour the entire Premises within ten (10) calendar days (in Mississippi, 10 business days) of such Denial of Application. In order to schedule said tour within ten (10) calendar days (in Mississippi, 10 business days), the request for a tour of the Premises must be received by BellSouth within five (5) calendar days of the Denial of Application.
- 2.4 <u>Filing of Petition for Waiver</u>. Upon Denial of Application BellSouth will timely file a petition with the Commission pursuant to 47 U.S.C. § 251(c)(6).
- Waiting List. On a first-come, first-served basis governed by the date of receipt of an application or Letter of Intent, BellSouth will maintain a waiting list of requesting carriers who have either received a Denial of Application or, where it is publicly known that the Premises is out of space, have submitted a Letter of Intent to collocate. BellSouth will notify the telecommunications carriers on the waiting list when space becomes available according to how much space becomes available and the position of telecommunications carrier on said waiting list. NewSouth must submit an updated, complete, and correct application to BellSouth within 30 business days or notify BellSouth in writing that NewSouth wants to maintain its place on the waiting list either without accepting such space or accepting an amount of space less than its original request. If NewSouth does not submit such an application or notify BellSouth in writing as described above, BellSouth will offer such space to the next

CLEC on the waiting list and remove NewSouth from the waiting list. Upon request, BellSouth will advise NewSouth as to its position on the list.

- In Florida, on a first-come, first-served basis governed by the date of receipt of an Application or Letter of Intent, BellSouth will maintain a waiting list of requesting carriers who have either received a Denial of Application or, where it is publicly known that the Premises is out of space, have submitted a Letter of Intent to collocate. Sixty (60) calendar days prior to space becoming available, if known, BellSouth will notify the Florida PSC and the telecommunications carriers on the waiting list by mail when space becomes available according to the position of telecommunications carrier on said waiting list. If not known sixty (60) calendar days in advance, BellSouth shall notify the Florida PSC and the telecommunications carriers on the waiting list within two days of the determination that space is available.
- Public Notification. BellSouth will maintain on its Interconnection Services website a notification document that will indicate all Central Offices that are without available space. BellSouth shall update such document within ten (10) calendar days (in Mississippi, 10 business days) of the Denial of Application date. BellSouth will also post a document on its Interconnection Services website that contains a general notice where space has become available in a Central Office previously on the space exhaust list. BellSouth shall allocate said available space pursuant to the waiting list referenced in Section 2.5.
- 2.7 <u>State Agency Procedures</u>. Notwithstanding the foregoing, should any state or federal regulatory agency impose procedures or intervals different than procedures or intervals set forth in this section, whether now in effect or that become effective after execution of this Agreement, those procedures or intervals shall supersede the requirements set forth herein for all applications submitted for the first time after the effective date thereof.

3. Collocation Options

Cageless. In accordance and compliance with local building code, BellSouth shall allow NewSouth to collocate NewSouth's equipment and facilities without requiring the construction of a cage or similar structure and without requiring the creation of a separate entrance to the Collocation Space. BellSouth shall allow NewSouth to have direct access to its equipment and facilities 24 hours/day, 7days/week, but may require NewSouth to use a central entrance to the BellSouth Premises. BellSouth shall make cageless collocation available in single bay increments pursuant to Section 7. Except where NewSouth's equipment requires special technical considerations (e.g., special cable racking, isolated ground plane), BellSouth shall assign cageless Collocation Space in conventional equipment rack lineups where feasible. For equipment requiring special technical considerations, NewSouth must provide the equipment layout, including spatial dimensions for such equipment pursuant to generic requirements contained in BellCore (Telcordia) GR-63-Core and shall be

responsible for constructing all special technical requirements associated with such equipment pursuant to Section 6.5 following.

- Cages and Adjacent Arrangement Enclosures. At NewSouth's option and upon request, BellSouth shall construct enclosures in compliance with NewSouth's collocation request and in accordance and compliance with local building code. At NewSouth's request, BellSouth shall permit NewSouth to subcontract the construction of physical collocation arrangements with a contractor certified by BellSouth ("BellSouth Certified Contractor"), provided however, that BellSouth shall not unreasonably withhold approval of contractors. When NewSouth elects to have BellSouth construct an enclosure, space enclosure charges will apply as set forth in Exhibit A to this Attachment. The space enclosure charges set forth in Exhibit A to this Attachment will not apply in cases where NewSouth elects to use a BellSouth Certified Contractor to construct an enclosure.
- When NewSouth subcontracts the construction, NewSouth must arrange with a 3.2.1 BellSouth Certified Contractor to construct a collocation arrangement enclosure in accordance with BellSouth's reasonable and nondiscriminatory guidelines and specifications and at NewSouth's sole expense. BellSouth will provide guidelines and specifications upon request. Where local building codes require enclosure specifications more stringent than BellSouth's standard enclosure specification, NewSouth and NewSouth's BellSouth Certified Contractor must comply with local building code requirements. NewSouth's BellSouth Certified Contractor shall be responsible for filing and receiving any and all necessary permits and/or licenses for such construction. BellSouth shall cooperate with NewSouth and provide, at NewSouth's expense, the documentation, including architectural drawings, necessary for NewSouth to obtain the zoning, permits and/or other licenses. BellSouth shall pass on to NewSouth its reasonable costs of providing the documentation. The BellSouth Certified Contractor shall bill NewSouth directly for all work performed for NewSouth pursuant to this Attachment and BellSouth shall have no liability for nor responsibility to pay such charges imposed by the BellSouth Certified Contractor. NewSouth must provide the local BellSouth building contact with two Access Keys used to enter the locked enclosure. Except in case of emergency, BellSouth will not access NewSouth's locked enclosure prior to notifying NewSouth.
- 3.2.2 BellSouth has the right to review NewSouth's plans and specifications prior to allowing construction to start. BellSouth shall complete its review within 15 calendar days. BellSouth has the right to inspect the enclosure after construction to make sure it is designed and constructed according to BellSouth's reasonable and nondiscriminatory guidelines and specifications and to require NewSouth to remove or correct at NewSouth's cost any structure that does not meet these standards.
- 3.2 <u>Shared (Subleased) Caged Collocation</u>. NewSouth may allow other telecommunications carriers to share NewSouth's caged collocation arrangement pursuant to terms and conditions agreed to by NewSouth ("Host") and other telecommunications carriers ("Guests") and pursuant to this section in accordance and

compliance with local building code, except where the BellSouth Premises is located within a leased space and BellSouth is prohibited by said lease from offering such an option. NewSouth shall notify BellSouth in writing upon execution of any agreement between the Host and its Guest within ten (10) calendar days of its execution. Further, such notice shall include the name of the Guest(s) and the term of the agreement, and shall contain a certification by NewSouth that said agreement imposes upon the Guest(s) the same terms and conditions for Collocation Space as set forth in this Attachment between BellSouth and NewSouth.

- NewSouth shall be the sole interface and responsible Party to BellSouth for the purpose of submitting applications for initial and additional equipment placements of Guest; for assessment of rates and charges contained within this Attachment; and for the purposes of ensuring that the safety and security requirements of this Attachment are fully complied with by the Guest, its employees and agents. In the event the Host and Guest jointly submit an initial Application, only one Application Fee will be assessed. A separate initial Guest application shall require the assessment of a Subsequent Application Fee, as set forth in Exhibit A, if this application is not the initial application made for the arrangement. Notwithstanding the foregoing, Guest may arrange directly with BellSouth for the provision of the interconnecting facilities between BellSouth and Guest and for the provision of the services and access to unbundled network elements and BellSouth will bill the Guest directly for such services.
- 3.2.2 NewSouth shall indemnify and hold harmless BellSouth from any and all claims, actions, causes of action, of whatever kind or nature arising out of the presence of NewSouth's Guests in the Collocation Space except to the extent caused by BellSouth's sole negligence, gross negligence, or willful misconduct.
- In making shared caged arrangements available, whether or not NewSouth serves as Host, BellSouth may not increase the cost of site preparation or nonrecurring charges above the cost of provisioning such a shared arrangement of similar dimensions and material to a single collocating party.
- 3.2.4 BellSouth will not place unreasonable restrictions on NewSouth's use of a cage, and as such will allow NewSouth to contract with other CLECs to share the cage in a sublease-type arrangement. If two (2) or more CLECs who have interconnection agreements with BellSouth utilize a shared collocation cage, BellSouth will permit each CLEC to order UNEs to and provision service from that shared collocation space, regardless of which CLEC was the original Collocator.
- Adjacent Collocation. BellSouth will provide adjacent collocation arrangements ("Adjacent Arrangement") where space within the Premises is legitimately exhausted, subject to technical feasibility, where the Adjacent Arrangement does not interfere with access to existing or planned structures or facilities on the Premises property and where permitted by zoning and other applicable state and local regulations. The Adjacent Arrangement shall be constructed or procured by NewSouth and in

conformance with BellSouth's reasonable and nondiscriminatory design and construction specifications. Further, NewSouth shall construct, procure, maintain and operate said Adjacent Arrangement(s) pursuant to all of the terms and conditions set forth in this Attachment. Reasonable and nondiscriminatory rates shall be negotiated at the time of the request for the Adjacent Arrangement.

- Should NewSouth elect such option, NewSouth must arrange with a BellSouth 3.4.1 Certified Contractor to construct an Adjacent Arrangement structure in accordance with BellSouth's reasonable and nondiscriminatory guidelines and specifications. BellSouth will provide guidelines and specifications upon request. Where local building codes require enclosure specifications more stringent than BellSouth's standard specification, NewSouth and NewSouth's BellSouth Certified Contractor must comply with local building code requirements. NewSouth's BellSouth Certified Contractor shall be responsible for filing and receiving any and all necessary zoning, permits and/or licenses for such construction. NewSouth's BellSouth Certified Contractor shall bill NewSouth directly for all work performed for NewSouth pursuant to this Attachment and BellSouth shall have no liability for nor responsibility to pay such charges imposed by the BellSouth Certified Contractor. NewSouth must provide the local BellSouth building contact with two cards, keys or other access device used to enter the locked enclosure. Except in cases of emergency, BellSouth shall not access NewSouth's locked enclosure prior to notifying NewSouth. BellSouth will notify NewSouth by telephone of such access within a reasonable time.
- 3.4.2 BellSouth maintains the right to review NewSouth's plans and specifications prior to construction of an Adjacent Arrangement(s). BellSouth shall complete such review within 15 business days. BellSouth may inspect the Adjacent Arrangement(s) following construction and prior to the Commencement Date, as defined in Section 4.1 following, to ensure the design and construction comply with BellSouth's reasonable and nondiscriminatory guidelines and specifications. BellSouth may require NewSouth, at NewSouth's sole cost, to correct any deviations from BellSouth's reasonable and nondiscriminatory guidelines and specifications found during such inspection(s), up to and including removal of the Adjacent Arrangement, within five (5) business days of BellSouth's inspection for deviations that may cause harm to personnel or property and within a reasonable period of time for other deviations, unless the Parties mutually agree to an alternative time frame.
- NewSouth shall provide a concrete pad, the structure housing the arrangement, heating/ventilation/air conditioning ("HVAC"), lighting, and all facilities that connect the structure (i.e. racking, conduits, etc.) to the BellSouth point of interconnection. At NewSouth's option, and where the local authority having jurisdiction permits, BellSouth shall provide an AC power source and access to physical collocation services and facilities subject to the same nondiscriminatory requirements as applicable to any other physical collocation arrangement. NewSouth's BellSouth Certified Contractor shall be responsible for filing and receiving any and all necessary zoning, permits and/or licenses for such arrangement.

3.4.4 BellSouth shall allow Shared (Subleased) Caged Collocation within an Adjacent Arrangement pursuant to the terms and conditions set forth in Section 3.3 preceding.

4. Occupancy

- 4.1 <u>Commencement Date</u>. The "Commencement Date" shall be the day NewSouth's equipment becomes operational as described in Article 4.2, following.
- Occupancy. BellSouth will notify NewSouth in writing that the Collocation Space is ready for occupancy. NewSouth must notify BellSouth in writing that collocation equipment installation is complete and is operational with BellSouth's network. BellSouth may, for good cause and on a reasonable and nondiscriminatory basis, not accept orders for interconnected service until receipt of such notice. For purposes of this paragraph, NewSouth's telecommunications equipment will be deemed operational when cross-connected to BellSouth's network for the purpose of service provision.
- Termination. Except where otherwise agreed to by the Parties, NewSouth may 4.3 terminate occupancy in a particular Collocation Space upon thirty (30) calendar days prior written notice to BellSouth. Upon termination of such occupancy, NewSouth at its expense shall remove its equipment and other property from the Collocation Space. NewSouth shall have thirty (30) calendar days from the termination date to complete such removal, including the removal of all equipment and facilities of NewSouth's Guests; provided, however, that NewSouth shall continue payment of monthly fees to BellSouth until such date as NewSouth has fully vacated the Collocation Space. Should NewSouth or NewSouth's Guest fail to vacate the Collocation Space within thirty (30) calendar days from the termination date, BellSouth shall have the right to remove the equipment and other property of NewSouth or NewSouth's Guest at NewSouth's expense and with no liability for damage or injury to NewSouth or NewSouth's Guest's property unless caused by the gross negligence or intentional misconduct of BellSouth. Upon termination of occupancy, NewSouth shall surrender such Collocation Space to BellSouth in the same condition as when first occupied by the NewSouth except for ordinary wear and tear unless otherwise agreed to by the Parties. NewSouth shall be responsible for the cost of removing any enclosure, together with all support structures (e.g., racking, conduits), of an Adjacent Collocation arrangement at the termination of occupancy and restoring the grounds to their original condition.

5. Use of Collocation Space

5.1 Equipment Type. In accordance with applicable FCC and State Commission rules and orders, BellSouth permits the collocation of any type of equipment necessary for interconnection to BellSouth's network or for access to unbundled network elements in the provision of telecommunications services. Such equipment used or useful for

interconnection and access to unbundled network elements includes, but is not limited to transmission equipment including, but not limited to, optical terminating equipment and multiplexers, and digital subscriber line access multiplexers, routers, asynchronous transfer mode multiplexers, and remote switching modules. Nothing in this section requires BellSouth to permit collocation of equipment used solely to provide enhanced services.

- 5.1.1 Such equipment must at a minimum meet the following BellCore (Telcordia) Network Equipment Building Systems (NEBS) General Equipment Requirements: Criteria Level 1 requirements as outlined in the BellCore (Telcordia) Special Report SR-3580, Issue 1; equipment design spatial requirements per GR-63-CORE, Section 2; thermal heat dissipation per GR-063-CORE, Section 4, Criteria 77-79; acoustic noise per GR-063-CORE, Section 4, Criterion 128, and National Electric Code standards. BellSouth may not impose safety requirements on NewSouth's equipment that are more stringent than the safety requirements it imposes on its own equipment. BellSouth may not object to the collocation of NewSouth's equipment on the ground that the equipment fails to comply with NEBS performance standards. If BellSouth denies collocation of a competitor's equipment, citing safety standards, BellSouth must provide to NewSouth within five (5) business days of the denial a list of all equipment that BellSouth locates with the premises in question, together with an affidavit attesting that all of that equipment meets or exceeds the safety standard that BellSouth contends NewSouth's equipment fails to meet. In the event that BellSouth believes that the collocated equipment will not be or is not being used for interconnection or access to unbundled network elements or determines that NewSouth's equipment does not meet NEBS Level 1 safety requirements, NewSouth will be given ten (10) calendar days to comply with the requirements or remove the equipment from the collocation space. If the parties do not resolve the dispute, BellSouth or NewSouth may file a complaint at the Commission seeking a formal resolution of the dispute.
- 5.1.2 NewSouth shall not use the Collocation Space for marketing purposes nor shall it place any identifying signs or markings in the area surrounding the Collocation Space or on the grounds of the Premises.
- NewSouth shall place a plaque or other identification affixed to NewSouth's equipment necessary to identify NewSouth's equipment, including a list of emergency contacts with telephone numbers.
- Upon request, NewSouth will certify in writing to BellSouth that the equipment is necessary for interconnection or access to unbundled network elements. In the event that the Parties have a dispute about the type of equipment to be collocated, BellSouth may, within thirty (30) calendar days from the receipt of the written certification, file a complaint with the Commission seeking a formal determination that the equipment cannot be collocated in a BellSouth Premises. While the dispute is pending, BellSouth will not prevent or unreasonably delay installation of the disputed equipment in the Collocation space; however, NewSouth will not activate the

equipment during the pendency of the dispute. NewSouth will be responsible for all costs incurred as a result of the installation should removal or modification of the equipment be required by the Commission's ruling.

- Entrance Facilities. NewSouth may elect to place NewSouth-owned or NewSouth-5.2 leased fiber entrance facilities into the Collocation Space. BellSouth will provide an interconnection point or points, physically accessible by both BellSouth and NewSouth, at which the fiber optic cable carrying NewSouth's circuits can enter BellSouth's premises, provided that BellSouth shall designate interconnection points as close as possible to its premises. NewSouth will provide and place fiber cable at the point of entrance of sufficient length to be pulled through conduit and into the splice location. NewSouth will provide and install a sufficient length of fire retardant riser cable, to which the entrance cable will be spliced, which will extend from the splice location to NewSouth's equipment in the Collocation Space. In the event NewSouth utilizes a non-metallic, riser-type entrance facility, a splice will not be required. NewSouth must contact BellSouth for instructions prior to placing the entrance facility cable in the manhole. NewSouth is responsible for maintenance of the entrance facilities. At NewSouth's option, BellSouth will accommodate where technically feasible a microwave entrance facility pursuant to separately negotiated terms and conditions. BellSouth will permit interconnection of copper or coaxial cable if such interconnection is first approved by the Commission.
- Dual Entrance. BellSouth will provide at least two interconnection points at each Premises where there are at least two such interconnection points available and where capacity exists. Upon receipt of a request for physical collocation under this Attachment, BellSouth shall provide NewSouth with information regarding BellSouth's capacity to accommodate dual entrance facilities. If conduit in the serving manhole(s) is available and is not reserved for another purpose for utilization within 12 months of the receipt of an application for collocation, BellSouth will make the requested conduit space available for installing a second entrance facility to NewSouth's arrangement. The location of the serving manhole(s) will be determined at the reasonable and nondiscriminatory discretion of BellSouth. Where dual entrance is not available due to lack of capacity, BellSouth will so state in the Application Response. BellSouth shall not deny a collocation application solely for the reason that dual entrance facilities are not available.
- Shared Use. NewSouth may utilize spare capacity on an existing interconnector entrance facility for the purpose of providing an entrance facility to another NewSouth collocation arrangement within the same BellSouth Premises. NewSouth must arrange with BellSouth for BellSouth to splice the utilized entrance facility capacity to NewSouth-provided riser cable.
- Splicing in the Entrance Manhole. Although not generally permitted, should NewSouth request a splice to occur in the entrance manhole(s), BellSouth, at its reasonable and nondiscriminatory discretion, may grant such a request. When the request for a splice is granted to NewSouth by BellSouth, NewSouth shall ensure its

employees or agents entering and/or performing work in the entrance manhole(s) are trained and comply with BellSouth procedures and OSHA requirements regarding access to manholes and that BellSouth personnel are notified and present for all entrances and work performed in the entrance manhole(s). Manhole covers shall be properly closed and secured at the conclusion of entry and/or work. Advance notification to BellSouth shall occur at a minimum of 48 hours prior to desired entry for normal work activities and at a minimum of 2 hours prior to desired entry in an out of service condition.

- 5.4 Demarcation Point. BellSouth, in a reasonable and nondiscriminatory manner, will designate the point(s) of demarcation between NewSouth's equipment and/or network and BellSouth's network. Each Party will be responsible for maintenance and operation of all equipment/facilities on its side of the demarcation point. For 2-wire and 4-wire connections to BellSouth's network, the demarcation point shall be a common block on the BellSouth designated conventional distributing frame. NewSouth shall be responsible for providing, and a supplier certified by BellSouth ("NewSouth's BellSouth Certified Supplier") shall be responsible for installing and properly labeling/stenciling, the common block, and necessary cabling pursuant to Section 6.4. For DS1 and DS3 connections, the demarcation point shall be a BellSouth provided DSX panel. For fiber connections, the demarcation point shall be a BellSouth provided LGX panel. NewSouth or its agent must perform all required maintenance to equipment/facilities on its side of the demarcation point, pursuant to Section 5.5, following, and may self-provision cross-connects that may be required within the Collocation Space to activate service requests. At NewSouth's option and expense, a Point of Termination ("POT") bay or frame may be placed in the Collocation Space, but will not serve as the demarcation point. NewSouth must make arrangements with a BellSouth Certified Supplier for such placement.
- In the event NewSouth's collocation space is at such a distance from the demarcation point that an intermediary transmission device is needed to prevent signal degradation and ensure compliance with industry standards, BellSouth, at its own expense, shall install such transmission device. In such cases, BellSouth will notify NewSouth and all relevant vendors that the distance poses a risk of degradation and that larger gauge cable should be used.
- NewSouth's Equipment and Facilities. NewSouth, or if required by this Attachment, NewSouth's BellSouth Certified Supplier, is solely responsible for the design, engineering, installation, testing, provisioning, performance, monitoring, maintenance and repair of the equipment and facilities used by NewSouth. Such equipment and facilities may include but are not limited to cable(s); equipment; and point of termination connections.
- BellSouth's Access to Collocation Space. From time to time BellSouth may require access to the Collocation Space. BellSouth retains the right to access such space for the purpose of making BellSouth equipment and building modifications (e.g., running, altering or removing racking, ducts, electrical wiring, HVAC, and cables).

BellSouth will give three (3) business days' notice to NewSouth when access to the Collocation Space is required. NewSouth may elect to be present whenever BellSouth performs work in the Collocation Space. The Parties agree that NewSouth will not bear any of the expense associated with this work.

- Access. Pursuant to Section 11, NewSouth shall have access to the Collocation Space twenty-four (24) hours a day, seven (7) days a week. NewSouth agrees to provide the name and social security number or date of birth or driver's license number of each employee, contractor, or agents of NewSouth provided with access keys or cards ("Access Keys") prior to the issuance of said Access Keys. BellSouth shall issue Access Keys within thirty (30) calendar days of such request for such Access Keys. Access Keys shall not be duplicated under any circumstances. NewSouth agrees to be responsible for all Access Keys and for the return of all said Access Keys in the possession of NewSouth employees, contractors, Guests, or agents after termination of the employment relationship, contractual obligation with NewSouth or upon the termination of this Attachment or the termination of occupancy of an individual collocation arrangement.
- 5.7.1 BellSouth will not repeatedly delay NewSouth's entry into a Premises or access to its collocated equipment. BellSouth will provide NewSouth with reasonable access to restroom facilities and parking.
- Lost or Stolen Access Keys. NewSouth shall notify BellSouth in writing immediately in the case of lost or stolen Access Keys. Should it become necessary for BellSouth to re-key buildings or deactivate a card as a result of a lost Access Key(s) or for failure to return an Access Key(s), NewSouth shall pay for all reasonable costs associated with the re-keying or deactivating the card. Likewise, if NewSouth must re-key the lock on its collocation cage as a result of BellSouth losing the NewSouth provided key, BellSouth shall pay the reasonable costs for re-keying the cage and replacing keys.
- 5.8 Interference or Impairment. Notwithstanding any other provisions of this Attachment, equipment and facilities placed in the Collocation Space shall not interfere with or impair service provided by BellSouth or by any other interconnector located in the Premises; shall not endanger or damage the facilities of BellSouth or of any other interconnector, the Collocation Space, or the Premises; shall not compromise the privacy of any communications carried in, from, or through the Premises; and shall not create an unreasonable risk of injury or death to any individual or to the public. If BellSouth, on a reasonable and nondiscriminatory basis, determines that any equipment or facilities of NewSouth violates the provisions of this paragraph, BellSouth shall give written notice to NewSouth, which notice shall direct NewSouth to cure the violation within forty-eight (48) hours of NewSouth's actual receipt of written notice or, at a minimum, to commence curative measures within 24 hours and to exercise reasonable diligence to complete such measures as soon as possible thereafter. After receipt of the notice, the Parties agree to consult immediately and, if necessary, to inspect the arrangement. If NewSouth fails to take curative action

within 48 hours or if the violation is of a character which poses an immediate and substantial threat of damage to property, injury or death to any person, or interference/impairment of the services provided by BellSouth or any other interconnector, then and only in that event, BellSouth may take such action as it deems appropriate to correct the violation, including without limitation the interruption of electrical power to NewSouth's equipment. BellSouth will endeavor, but is not required, to provide notice to NewSouth prior to taking such action and shall have no liability to NewSouth for any damages arising from such action, except to the extent that such action by BellSouth constitutes willful misconduct.

- Personalty and its Removal. Subject to requirements of this Attachment, NewSouth may place or install in or on the Collocation Space such facilities and equipment, including storage for spare equipment, as it deems desirable for the conduct of business, provided that such equipment is telecommunications equipment, does not violate floor loading requirements, nor imposes or could impose or contains or could contain environmental conditions or hazards. Personal property, facilities and equipment placed by NewSouth in the Collocation Space shall not become a part of the Collocation Space, even if nailed, screwed or otherwise fastened to the Collocation Space, but shall retain their status as personalty and may be removed by NewSouth at any time. Any damage caused to the Collocation Space by NewSouth's employees, agents or representatives during the removal of such property shall be promptly repaired by NewSouth at its expense.
- Alterations. In no case shall NewSouth or any person acting on behalf of NewSouth make any rearrangement, modification, improvement, addition, repair, or other alteration which could affect in any way space, power, HVAC, and/or safety considerations to the Collocation Space or the BellSouth Premises without the written consent of BellSouth, which consent shall not be unreasonably withheld. The cost of any such specialized alterations shall be paid by NewSouth.
- Janitorial Service. NewSouth shall be responsible for the general upkeep and cleaning of the Caged Collocation Space and, if using a contractor, shall arrange directly with a BellSouth Certified Contractor for janitorial services. BellSouth shall provide a list of such contractors on a site-specific basis upon request.

6. Ordering and Preparation of Collocation Space

Should any state regulatory or federal agency impose procedures or intervals different than procedures or intervals set forth in this section, whether now in effect or that become effective after execution of this Agreement, those procedures or intervals shall supersede the requirements set forth herein for all applications submitted for the first time after the effective date thereof.

- 6.2 <u>Application for Space</u>. NewSouth shall submit an application document when NewSouth or NewSouth's Guest(s), as defined in Section 3.3, desires to request or modify the use of the Collocation Space.
- Initial Application. For NewSouth or NewSouth's Guest(s) initial equipment placement, NewSouth shall submit to BellSouth a Physical Expanded Interconnection Application Document ("Application"), together with payment of the Application Fee as stated in Exhibit A. The Application is Bona Fide when it is complete and accurate, meaning that all required fields on the application are completed with the appropriate type of information. The Bona Fide Application shall contain a detailed description and schematic drawing of the equipment to be placed in NewSouth's Collocation Space(s) and an estimate of the amount of square footage required.
- Subsequent Application Fee. In the event NewSouth or NewSouth's Guest(s) desire 6.2.2 to modify the use of the Collocation Space in a manner not reflected in the original application, NewSouth shall complete an Application document detailing all information regarding the modification to the Collocation Space. BellSouth shall determine what modifications, if any, to the Premises are required to accommodate the change requested by NewSouth in the Application. Such necessary modifications to the Premises may include but are not limited to, floor loading changes, changes necessary to meet HVAC requirements, changes to power plant requirements, and equipment additions. Where the subsequent Application does not require assessment for provisioning or construction work by BellSouth (e.g., NewSouth proposes to install a splitter), no Subsequent Application Fee will be required. The fee for an Application where the modification requested has limited effect (i.e., does not require assessment related to capital expenditure by BellSouth; e.g., NewSouth proposes to enhance power by adding a fuse) shall be the Subsequent Application Fee as set forth in Exhibit A. If the modification requires capital expenditure assessment (e.g., NewSouth proposes to add cable racking or increase floor space), the full Application Fee for the appropriate state shall apply.
- Application Response. In Alabama, Kentucky, North Carolina, and Tennessee, in addition to the notice of space availability pursuant to Section 2.1, BellSouth will respond within ten (10) calendar days of receipt of an Application stating whether the Application is Bona Fide, and if it is not Bona Fide, the items necessary to cause the Application to become Bona Fide. When space has been determined to be available, BellSouth will provide a written response ("Application Response") within twenty-three (23) business days, which will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 7.
- In South Carolina, in addition to the notice of space availability pursuant to Section 2.1, BellSouth will respond within ten (10) calendar days as to whether the Application is Bona Fide, and if it is not Bona Fide, the items necessary to cause the Application to become Bona Fide. When space has been determined to be available, BellSouth will provide a written response ("Application Response") within thirty (30)

calendar days of receipt of a Bona Fide Application. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 7. When multiple applications are submitted in a state within a fifteen (15) calendar day window, BellSouth will respond to the Bona Fide Applications as soon as possible, but no later than the following: within thirty (30) calendar days for Bona Fide Applications 1-5; within thirty-six (36) calendar days for Bona Fide Applications 6-10; within forty-two (42) calendar days for Bona Fide Applications 11-15. Response intervals for multiple Bona Fide Applications submitted within the same timeframe for the same state in excess of 15 must be negotiated. All negotiations shall consider the total volume from all requests from telecommunications companies for collocation.

- In Florida, within fifteen (15) calendar days of receipt of a Bona Fide Application, BellSouth will respond as to whether space is available or not available within a particular Premises. Additionally, when space has been determined to be available or when a lesser amount of space than that requested is available, then with respect to the space available, BellSouth will provide a written response ("Application Response") including sufficient information to enable NewSouth to place a Firm Order. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 7. When NewSouth submits ten (10) or more Applications within ten (10) calendar days, the initial fifteen (15) day response period will increase by ten (10) days for every additional ten (10) Applications or fraction thereof.
- In Georgia, in addition to the notice of space availability pursuant to Section 2.1, BellSouth will respond as to whether the Application is Bona Fide, and if it is not Bona Fide, the items necessary to cause the Application to become Bona Fide. When space has been determined to be available for caged or cageless arrangements, BellSouth will provide a written response ("Application Response") within thirty (30) calendar days of receipt of a Bona Fide Application. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 7.
- In Louisiana, in addition to the notice of space availability pursuant to Section 2.1, BellSouth will respond as to whether the Application is Bona Fide, and if it is not Bona Fide, the items necessary to cause the Application to become Bona Fide. BellSouth will respond as to whether space is available or not available within a particular Premises in accordance with Section 2. When space has been determined to be available, BellSouth will provide a written response ("Application Response") within thirty (30) calendar days of receipt of a Bona Fide Application. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 7. BellSouth will respond with a full Application Response within 30 calendar days for one to ten Applications; 35 calendar days for eleven to

twenty Applications; and for requests of more than twenty Application it is increased by five calendar days for every five Applications received within five business days.

- In Mississippi, in addition to the notice of space availability pursuant to Section 2., 6.3.5 BellSouth will respond as to whether the Application is Bona Fide, and if it is not Bona Fide, the items necessary to cause the Application to become Bona Fide. When space has been determined to be available, BellSouth will provide a written response ("Application Response") within thirty (30) business days of receipt of a Bona Fide Application. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 7. When multiple applications are submitted in a state within a fifteen (15) business day window, BellSouth will respond to the Bona Fide Applications as soon as possible, but no later than the following: within thirty (30) business days for Bona Fide Applications 1-5; within thirty-six (36) business days for Bona Fide Applications 6-10; within forty-two (42) business days for Bona Fide Applications 11-15. Response intervals for multiple Bona Fide Applications submitted within the same timeframe for the same state in excess of 15 must be negotiated. All negotiations shall consider the total volume from all requests from telecommunications companies for collocation.
- Application Modifications. If a modification or revision is made to any information in the Bona Fide Application for Physical Collocation or the Bona Fide Application for Adjacent Collocation, with the exception of modifications to Customer Information, Contact Information or Billing Contact Information, either at the request of NewSouth or necessitated by technical considerations, the application will be considered a new Application for the purposes of the response and provisioning intervals. If, at any time, BellSouth needs to reevaluate NewSouth's Bona Fide Application as a result of changes requested by NewSouth to NewSouth's original application, then BellSouth will charge NewSouth a Subsequent Application Fee. Major changes such as requesting additional space or adding additional equipment may require NewSouth to resubmit the Application with an Application Fee.
- Bona Fide Firm Order. In Alabama, Kentucky, North Carolina, and Tennessee, NewSouth shall indicate its intent to proceed with equipment installation in a BellSouth Premises by submitting a Bona Fide Firm Order to BellSouth. A Bona Fide Firm Order requires NewSouth to complete the Application/Inquiry process described in Section 6.2, preceding, and submit the Physical Expanded Interconnection Firm Order document (BSTEI-1P-F) indicating acceptance of the Application Response provided by BellSouth ("Bona Fide Firm Order"). The Bona Fide Firm Order must be received by BellSouth no later than five (5) business days after NewSouth's receipt of BellSouth's Application Response to NewSouth's Bona Fide Application. If the Bona Fide Firm Order is not received within this five (5) business day period, the construction and provisioning interval shall be extended day for day for each day after the 5th business day. If the Bona Fide Firm Order is not received within (30) calendar days of the Application Response, the Application will expire.

- In South Carolina, NewSouth shall indicate its intent to proceed with equipment installation in a BellSouth Premises by submitting a Physical Expanded Interconnection Firm Order document ("Firm Order") to BellSouth. A Firm Order shall be considered Bona Fide when NewSouth has completed the Application/Inquiry process described in Section 6.2, preceding and has submitted the Firm Order document indicating acceptance of the Application Response provided by BellSouth. The Bona Fide Firm Order must be received by BellSouth no later than thirty (30) calendar days after BellSouth's Application Response to NewSouth's Bona Fide Application or the Application will expire.
- In Mississippi, NewSouth shall indicate its intent to proceed with equipment installation in a BellSouth Premises by submitting a Physical Expanded Interconnection Firm Order document ("Firm Order") to BellSouth. A Firm Order shall be considered Bona Fide when NewSouth has completed the Application/Inquiry process described in Section 6.2, preceding and has submitted the Firm Order document indicating acceptance of the Application Response provided by BellSouth. The Bona Fide Firm Order must be received by BellSouth no later than thirty (30) business days after BellSouth's Application Response to NewSouth's Bona Fide Application or the Application will expire.
- 6.5.3 BellSouth will establish a firm order date based upon the date BellSouth is in receipt of a Bona Fide Firm Order. BellSouth will acknowledge the receipt of NewSouth's Bona Fide Firm Order within seven (7) calendar days of receipt indicating that the Bona Fide Firm Order has been received. A BellSouth response to a Bona Fide Firm Order will include a Firm Order Confirmation containing the firm order date.
- 6.5.4 BellSouth will permit one accompanied site visit to NewSouth's designated collocation arrangement location after receipt of the Bona Fide Firm Order without charge to NewSouth.
- NewSouth must submit to BellSouth the completed Access Control Request Form (RF-2906-C) for all employees or agents requiring access to the BellSouth Premises a minimum of 30 calendar days prior to the date NewSouth desires access to the Collocation Space. NewSouth may submit such a request at any time subsequent to BellSouth's receipt of the Bona Fide Firm Order. In the event NewSouth desires access to the Collocation Space after submitting such a request but prior to access being approved, BellSouth shall permit NewSouth to access the Collocation Space, accompanied by a security escort at NewSouth's expense. NewSouth must request escorted access at least three (3) business days prior to the date such access is desired.
- 6.6 Construction and Provisioning
- 6.6.1 In Alabama (Caged Only) Kentucky, Tennessee and North Carolina, BellSouth will complete construction for collocation arrangements within seventy-six (76) business days from receipt of an Application or as agreed to by the Parties. Under extraordinary conditions, the Company will complete construction for collocation

arrangements within ninety-one (91) business days. Examples of extraordinary conditions include, but are not limited to, extended license or permitting intervals; major Company equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. In the event NewSouth submits a forecast as described in the following section three (3) months or more prior to the application date, the above intervals shall apply. In the event NewSouth submits such a forecast between two (2) months and three (3) months prior to the application date, the above intervals may be extended by one (1) additional month. In the event NewSouth submits such a forecast less than two (2) months prior to the application date, the above intervals may be extended by sixty (60) calendar days. BellSouth will attempt to meet standard intervals for unforecasted requests and any interval adjustments will be discussed with NewSouth at the time the application 's received. Raw space, which is space lacking the necessary infrastructure to provide collocation space (including but not limited to HVAC, Power, etc.), conversion time frames fall outside the normal intervals and are negotiated on an individual case basis. Additionally, installations to existing collocation arrangements for line sharing or line splitting, which include adding cable, adding cable and splitter, and adding a splitter, will be forty five (45) business days from receipt of an Application.

- 6.6.1.1 To be considered a timely and accurate forecast, NewSouth must submit to the Company the CLEC Forecast Form, as set forth in Exhibit C attached hereto, containing the following information: Central Office/Serving Wire Center CLLI, number of Caged square feet and/or Cageless bays, number of DS0, DS1, DS3 frame terminations, number of fused amps and planned application date.
- In Alabama, BellSouth will complete construction for cageless collocation arrangements under ordinary conditions as soon as possible and within a maximum of sixty (60) calendar days from receipt of a Bona Fide Firm Order and ninety (90) calendar days for extraordinary conditions or as agreed to by the Parties. Ordinary conditions are defined as space available with only minor changes to support systems required, such as but not limited to, HVAC, cabling and the power plant(s). Extraordinary conditions are defined to include but are not limited to major BellSouth equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. The Parties may mutually agree to renegotiate an alternative provisioning interval or BellSouth may seek a waiver from this interval from the Commission.
- In Florida, BellSouth will complete construction for collocation arrangements as soon as possible and within a maximum of 90 calendar days from receipt of a Bona Fide Firm Order or as agreed to by the Parties. For changes to collocation space after initial space completion, BellSouth will complete construction for collocation arrangements as soon as possible and within a maximum of 45 calendar days from

receipt of a Bona Fide Firm Order or as agreed to by the Parties. If BellSouth does not believe that construction will be completed within the relevant time frame and BellSouth and NewSouth cannot agree upon a completion date, within 45 calendar days of receipt of the Bona Fide Firm Order for an initial request, and within 30 calendar days for Augmentations, BellSouth may seek an extension from the Florida PSC.

- In Georgia, BellSouth will complete construction for caged collocation arrangements 6.6.4 under ordinary conditions as soon as possible and within a maximum of 90 calendar days from receipt of a Bona Fide Firm Order or as agreed to by the Parties. BellSouth will use best efforts to complete construction for cageless collocation arrangements under ordinary conditions as soon as possible and within a maximum of 60 calendar days from receipt of a Bona Fide Firm Order and 90 calendar days for extraordinary conditions or as agreed to by the Parties. Ordinary conditions are defined as space available with only minor changes to support systems required, such as but not limited to, HVAC, cabling and the power plant(s). Extraordinary conditions are defined to include but are not limited to major BellSouth equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. BellSouth may elect to renegotiate an alternative provisioning interval with NewSouth or seek a waiver from this interval from the Commission.
- In Louisiana, BellSouth will complete construction for collocation arrangements 6.6.5 under ordinary conditions as soon as possible and within a maximum of 90 calendar days from receipt of a Bona Fide Firm Order for an initial request, and within 60 calendar days for an Augmentation, or as agreed to by the Parties. Ordinary conditions are defined as space available with only minor changes to support systems required, such as but not limited to, HVAC, cabling and the power plant(s). BellSouth will complete construction of all other Collocation Space ("extraordinary conditions") within 120 calendar days of the receipt of a Bona Fide Firm Order. Examples of extraordinary conditions include but are not limited to, extended license or permitting intervals; major BellSouth equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. BellSouth may elect to renegotiate an alternative provisioning interval with NewSouth or seek a waiver from this interval from the Commission.
- In Mississippi, excluding the time interval required to secure the appropriate government licenses and permits, BellSouth will complete construction for collocation arrangements under ordinary conditions as soon as possible and within a maximum of 120 calendar days from receipt of a Bona Fide Firm Order or as agreed to by the Parties. Ordinary conditions are defined as space available with only minor changes to support systems required, such as but not limited to, HVAC, cabling and the power plant(s). Excluding the time interval required to secure the appropriate

government licenses and permits, Be lSouth will complete construction of all other Collocation Space ("extraordinary conditions") within 180 calendar days of the receipt of a Bona Fide Firm Order. Examples of extraordinary conditions include but are not limited to, extended license or permitting intervals; major BellSouth equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. BellSouth may elect to renegotiate an alternative provisioning interval with NewSouth or seek a waiver from this interval from the Commission.

- In South Carolina, BellSouth will complete the construction and provisioning activities for cageless and caged collocation arrangements as soon as possible, but no later than 90 calendar days from receipt of a bona fide firm order. BellSouth may elect to renegotiate an alternative provisioning interval with NewSouth or seek a waiver from this interval from the Commission.
- Joint Planning Meeting. Unless otherwise agreed to by the Parties, a joint planning meeting or other method of joint planning between BellSouth and NewSouth will commence within a maximum of fifteen (15) calendar days from BellSouth's receipt of a Bona Fide Firm. At such meeting, the Parties will agree to the preliminary design of the Collocation Space and the equipment configuration requirements as reflected in the Bona Fide Application and affirmed in the Bona Fide Firm Order. BellSouth will complete all design work following the joint planning meeting. BellSouth will provide the floor plans to NewSouth within two weeks of a request for cageless collocation. For caged collocation, BellSouth will provide floor plans in the Application Response.
- Unless otherwise agreed, the Collocation Space completion due date and Access Customer Termination Location (ACTL) codes will be provided to NewSouth during the joint planning meeting. BellSouth shall deliver Connecting Facilities Assignments (CFA) prior to Space Acceptance unless a POT Bay either serves as the demarcation point or has been installed by NewSouth in its collocation space. In the event a POT Bay does exist, the parties shall work cooperatively in the provisioning of CFAs in a timely manner.
- 6.8 Permits. Each Party or its agents will diligently pursue filing for the permits required for the scope of work to be performed by that Party or its agents within ten (10) calendar days of the completion of finalized construction designs and specifications.
- Acceptance Walk Through. NewSouth and BellSouth will complete an acceptance walk through of each Collocation Space requested from BellSouth by NewSouth within fifteen (15) calendar days of BellSouth's notifying NewSouth that the collocation space is ready for occupancy. BellSouth will correct any deviations to NewSouth's original or jointly amended requirements within seven (7) calendar days after the walk through, unless the Parties jointly agree upon a different time frame.

- Use of BellSouth Certified Supplier. NewSouth shall select a supplier which has 6.10 been approved as a BellSouth Certified Supplier to perform all engineering and installation work required in the Collocation Space as set forth in TR 73503, which is consistent with industry standards. NewSouth may utilize its own employees to perform such work provided that NewSouth has been certified by BellSouth to perform such work. NewSouth must select BellSouth Certified Suppliers for transmission equipment, switching equipment and power equipment. BellSouth shall provide NewSouth with a list of BellSouth Certified Suppliers upon request. BellSouth shall not unreasonably withhold approval of any contractor proposed by NewSouth that meets the standard BellSouth criteria. The BellSouth Certified Supplier(s) shall be responsible for installing NewSouth's equipment and components, extending power cabling to the BellSouth power distribution frame, performing operational tests after installation is complete, and notifying BellSouth's equipment engineers and NewSouth upon successful completion of installation. The BellSouth Certified Supplier shall bill NewSouth directly for all work performed for NewSouth pursuant to this Attachment and BellSouth shall have no liability for nor responsibility to pay such charges imposed by the BellSouth Certified Supplier. BellSouth shall consider certifying NewSouth or any supplier proposed by NewSouth.
- 6.11 Alarm and Monitoring. BellSouth shall place environmental alarms in the Premises for the protection of BellSouth equipment and facilities. NewSouth shall be responsible for placement, monitoring and removal of environmental and equipment alarms used to service NewSouth's Collocation Space. Upon request, BellSouth will provide NewSouth with applicable tariffed service(s) to facilitate remote monitoring of collocated equipment by NewSouth. Both Parties shall use best efforts to notify the other of any verified environmental hazard known to that Party. The Parties agree to utilize and adhere to the Environmental Hazard Guidelines identified as Exhibit B attached hereto.
- Basic Telephone Service. Upon request of NewSouth, BellSouth will provide basic telephone service to the Collocation Space under the rates, terms and conditions of the current tariff offering for the service requested.
- Virtual Collocation Transition. BellSouth offers Virtual Collocation pursuant to the rates, terms and conditions set forth in its F.C.C. Tariff No. 1. For the interconnection to BellSouth's network and access to BellSouth unbundled network elements, NewSouth may purchase 2-wire and 4-wire cross-connects as set forth in Exhibit A, and NewSouth may place within its Virtual Collocation arrangements the telecommunications equipment set forth in Section 5.1. In the event physical Collocation Space was previously denied at a location due to technical reasons or space limitations, and that physical Collocation Space has subsequently become available, NewSouth may transition its virtual collocation arrangements to physical collocation arrangements and pay the appropriate non-recurring fees for physical collocation and for the rearrangement or reconfiguration of services terminated in the virtual collocation arrangement, as outlined in the appropriate BellSouth tariffs. In the event that BellSouth knows when additional space for physical collocation may

become available at the location requested by NewSouth, such information will be provided to NewSouth in BellSouth's written denial of physical collocation. To the extent that (i) physical Collocation Space becomes available to NewSouth within 180 calendar days of BellSouth's written denial of NewSouth's request for physical collocation, and (ii) NewSouth was not informed in the written denial that physical Collocation Space would become available within such 180 calendar days, then NewSouth may transition its virtual collocation arrangement to a physical collocation arrangement and will receive a credit for any nonrecurring charges previously paid for such virtual collocation. NewSouth must arrange with a BellSouth Certified Supplier for the relocation of equipment from its virtual Collocation Space to its physical Collocation Space and will bear the cost of such relocation.

- NewSouth may request the conversion of any existing virtual collocation 6.13.1 arrangements to physical collocation arrangements. BellSouth will authorize the conversion of virtual collocation arrangements to physical collocation arrangements without requiring the relocation of the virtual arrangement where there are no extenuating circumstances or technical reasons that would cause the arrangement to become a safety hazard within the Premises or otherwise being in conformance with the terms and conditions of this Attachment and where (1) there is no change to the arrangement; and (2) the conversion of the virtual arrangement would not cause the arrangement to be located in the area of the Premises reserved for BellSouth's forecast of future growth; and (3) due to the location of the virtual collocation arrangement, the conversion of said arrangement to a physical arrangement would not impact BellSouth's ability to secure its own facilities. Notwithstanding the foregoing, if the BellSouth Premises is at or nearing space exhaust, BellSouth may authorize the conversion of the virtual arrangement to a physical arrangement even though BellSouth could no longer secure its own facilities.
- 6.14 <u>Cancellation</u>. If, at anytime prior to Space Acceptance, NewSouth cancels its order for the Collocation Space(s), BellSouth will bill NewSouth the applicable non-recurring rate for any and all work processes for which work has begun.
- 6.15 <u>Licenses.</u> NewSouth, at its own expense, will be solely responsible for obtaining from governmental authorities, and any other appropriate agency, entity, or person, all rights, privileges, and licenses necessary or required to operate as a provider of telecommunications services to the public or to occupy the Collocation Space.

7. Rates and Charges

Recurring Charges. The recurring charges for space preparation begin on the date that NewSouth executes the written document accepting the Collocation Space pursuant to Section 6.9, or on the date NewSouth first occupies the Collocation Space, whichever is sooner. If NewSouth fails to schedule and complete a walkthrough pursuant to Section 6.9 within fifteen (15) days after BellSouth releases the space for occupancy,

then BellSouth shall begin billing NewSouth for recurring charges as of the sixteenth (16th) day after BellSouth releases the Collocation Space.

- Order Processing and monthly recurring charges for Central Office Modifications, assessed per arrangement, per square foot, and Common Systems Modifications, assessed per arrangement, per square foot for cageless and per cage for caged collocation. NewSouth shall remit payment of the nonrecurring Firm Order Processing Fee coincident with submission of a Bona Fide Firm Order. The charges recover the costs associated with preparing the Collocation Space, which includes survey, engineering of the Collocation Space, design and modification costs for network, building and support systems.. In the event NewSouth opts for cageless space, the space preparation fees will be assessed based on the total floor space dedicated to NewSouth as prescribed in Section 7.6.
- Space Preparation Fee in Georgia. In Georgia, the Space Preparation Fee is a one-time fee, assessed per arrangement, per location. It recovers a portion of costs associated with preparing the Collocation Space, which includes survey, engineering of the Collocation Space, design and modification costs for network, power, building and support systems. This is a set fee of \$100 per square foot as established by the Georgia Public Service Commission Order in Docket No. 7061-U.. In the event NewSouth opts for non-enclosed space, the space preparation fee will be assessed based on the total floor space dedicated to NewSouth as prescribed in Section 7.6.
- Space Preparation Fee in North Carolina. In North Carolina, space preparation fees consist of monthly recurring charges for Central Office Modifications, assessed per arrangement, per square foot; Common Systems Modifications, assessed per arrangement, per square foot for cageless and per cage for caged collocation; and Power, assessed per the nominal –48V DC ampere requirements specified by NewSouth on the Bona Fide Application.. The charges recover the costs associated with preparing the Collocation Space, which includes survey, engineering of the Collocation Space, design and modification costs for network, building and support systems.. In the event NewSouth opts for cageless space, the space preparation fees will be assessed based on the total floor space dedicated to NewSouth as described in Section 7.6.
- 7.5 <u>Cable Installation</u>. Cable Installation Fee(s) are assessed per entrance fiber placed.
- Floor Space. The floor space charge includes reasonable charges for lighting, HVAC, and other allocated expenses associated with maintenance of the Premises but does not include amperage necessary to power NewSouth's equipment. When the Collocation Space is enclosed, NewSouth shall pay floor space charges based upon the number of square feet so enclosed. When the Collocation Space is not enclosed, NewSouth shall pay floor space charges based upon the following floor space calculation: [(depth of the equipment lineup in which the rack is placed) + (0.5 x maintenance aisle depth) + (0.5 x wiring aisle depth)] X (width of rack and spacers).

For purposes of this calculation, the depth of the equipment lineup shall consider the footprint of equipment racks plus any equipment overhang. BellSouth will assign unenclosed Collocation Space in conventional equipment rack lineups where feasible. In the event NewSouth's collocated equipment requires special cable racking, isolated grounding or other treatment which prevents placement within conventional equipment rack lineups, NewSouth shall be required to request an amount of floor space sufficient to accommodate the total equipment arrangement. Floor space charges are due beginning with the date on which BellSouth releases the Collocation Space for occupancy or on the date NewSouth first occupies the Collocation Space, whichever is sooner.

- 7.7 Power. BellSouth shall make available –48 Volt (-48V) DC power for NewSouth's Collocation Space at a BellSouth Power Board or BellSouth Battery Distribution Fuse Bay ("EDFB") at NewSouth's option within the Premises.
- Recurring charges for -48V DC power will be assessed per ampere per month based 7.7.1 upon the BellSouth Certified Supplier engineered and installed power feed fused ampere capacity. Rates include redundant feeder fuse positions (A&B) and common cable rack to NewSouth's equipment or space enclosure. NewSouth is responsible for contracting with a BellSouth Certified Supplier for power distribution feeder cable runs from a BellSouth BDFB or power board to NewSouth's equipment. When obtaining power from a BellSouth BDFB or miscellaneous fuse positions on a BellSouth power board, power cables must be engineered, furnished and installed by NewSouth using a BellSouth Certified power Supplier. Determination of the BellSouth BDFB or BellSouth power board as the power source will be made at BellSouth's sole, but reasonable, discretion. When obtaining power from a BDFB, fuses and power cables (A&B) must be engineered (sized), and installed by NewSouth's BellSouth Certified Supplier. When obtaining power from a BellSouth power board, power cables (A&B) must be engineered (sized), and installed by NewSouth's BellSouth Certified power Supplier. NewSouth's BellSouth Certified Supplier must also provide a copy of the engineering power specification prior to the Commencement Date. BellSouth will provide the power feeder cable support structure between the BellSouth BDFB or power board and NewSouth's arrangement area. NewSouth shall contract a BellSouth Certified Supplier who will be responsible for the following: power cable support structure within NewSouth's arrangement; power cable feeds; terminations of cable. Any terminations at a BellSouth power board must be performed by a BellSouth Certified power Supplier. NewSouth shall comply with all applicable National Electric Code (NEC), BellSouth TR-73503, BellCore (Telcordia) and ANSI Standards regarding power cabling.
- 7.7.2 If BellSouth has not previously invested in power plant capacity for collocation at a specific site, NewSouth has the option to add its own dedicated power plant; provided, however, that such work shall be performed by a BellSouth Certified Supplier who shall comply with BellSouth's reasonable and nondiscriminatory guidelines and specifications. Where the addition of NewSouth's dedicated power plant results in construction of a new power plant room, upon termination of this

Agreement, NewSouth shall have the right to remove its equipment from the power plant room, but shall otherwise leave the room intact.

- 7.7.3 If NewSouth elects to install its own DC Power Plant, BellSouth shall provide AC power to feed NewSouth's DC Power Plant. Charges for AC power will be assessed per breaker ampere per month. Rates include the provision of commercial and standby AC power. When obtaining power from a BellSouth service panel, protection devices and power cables must be engineered (sized), and installed by NewSouth's BellSouth Certified Supplier except that BellSouth shall engineer and install protection devices and power cables for Adjacent Collocation. NewSouth's BellSouth Certified Supplier must also provide a copy of the engineering power specification prior to the Commencement Date. Charges for AC power shall be assessed pursuant to the rates specified in Exhibit A. AC power voltage and phase ratings shall be determined on a per location basis. At NewSouth's option, NewSouth may arrange for AC power in an Adjacent Collocation arrangement from a retail provider of electrical power.
- 7.8 Cable Record charges. These charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.
- 7.9 Security Escort. A security escort will be required whenever NewSouth or its approved agent desires access to the entrance manhole or must have access to the Premises after the one accompanied site visit allowed pursuant to Section 6.4.2 prior to completing BellSouth's Security Training requirements and/or prior to Space Acceptance. Rates for a security escort are assessed in one-half (1/2) hour increments according to the schedule appended hereto as Exhibit A.
- Rate "True-Up". It is the Parties' intention that the rates contained in Exhibit A 7.10 hereto are the Commission approved rates where such rates exist. To the extent that no such Commission approved rates exist, the Parties agree that the prices reflected as interim herein shall be "trued-up" (up or down) based on final prices either determined by further agreement or by final order in a proceeding involving BellSouth before the regulatory authority for the state in which the services are being performed or any other body having jurisdiction over this Agreement (hereinafter "Commission"). Under the "true-up" process, the interim price for each service shall be multiplied by the volume of that service purchased to arrive at the total interim amount paid for that service ("Total Interim Price"). The final price for that service shall be multiplied by the volume purchased to arrive at the total final amount due ("Total Final Price"). The Total Interim Price shall be compared with the Total Final Price. If the Total Final Price is more than the Total Interim Price, NewSouth shall pay the difference to BellSouth. If the Total Final Price is less than the Total Interim Price, BellSouth shall pay the difference to NewSouth. Each Party shall keep its own records upon which a "true-up" can be based and any final payment from one Party to the other shall be in an amount agreed upon by the Parties based on such records. In the event of any disagreement as between the records or the Parties regarding the

amount of such "true-up," the Parties agree that the Commission shall be called upon to resolve such differences.

Other. If no rate is identified in the contract, the rate for the specific service or function will be negotiated by the Parties upon request by either Party. Payment of all other charges under this Attachment shall be due thirty (30) calendar days after receipt of the bill (payment due date). NewSouth will pay a late payment charge of the lessor of the legal rate or one and one-half percent (1-1/2%) assessed monthly on any balance which remains unpaid after the payment due date.

8. Insurance

- NewSouth shall, at its sole cost and expense, procure, maintain, and keep in force insurance as specified in this Section 8 and underwritten by insurance companies licensed to do business in the states applicable under this Attachment and having a Best's Insurance Rating of B++X (B++ ten).
- 8.2 NewSouth shall maintain the following specific coverage:
- 8.2.1 Commercial General Liability coverage in the amount of ten million dollars (\$10,000,000.00) or a combination of Commercial General Liability and Excess/Umbrella coverage totaling not less than ten million dollars (\$10,000,000.00). BellSouth shall be named as an Additional Insured on the Commercial General Liability policy as specified herein.
- 8.2.2 Statutory Workers Compensation coverage and Employers Liability coverage in the amount of one hundred thousand dollars (\$100,000.00) each accident, one hundred thousand dollars (\$100,000.00) each employee by disease, and five hundred thousand dollars (\$500,000.00) policy limit by disease.
- 8.2.3 All Risk property coverage on a full replacement cost basis insuring all of NewSouth's real and personal property situated on or within BellSouth's Central Office location(s).
- NewSouth may elect to purchase business interruption and contingent business interruption insurance, having been advised that BellSouth assumes no liability for loss of profit or revenues should an interruption of service occur.
- The limits set forth in Section 8.2 above may be increased by BellSouth from time to time during the term of this Attachment upon thirty (30) days notice to NewSouth to at least such minimum limits as shall then be customary with respect to comparable occupancy of BellSouth structures.
- All policies purchased by NewSouth shall be deemed to be primary and not contributing to or in excess of any similar coverage purchased by BellSouth. All insurance must be in effect on or before the date equipment is delivered to BellSouth's

Premises and shall remain in effect for the term of this Attachment or until all NewSouth's property has been removed from BellSouth's Premises, whichever period is longer. If NewSouth fails to maintain required coverage, BellSouth may pay the premiums thereon and seek reimbursement of same from NewSouth.

NewSouth shall submit certificates of insurance reflecting the coverage required pursuant to this Section a minimum of ten (10) business days prior to the commencement of any work in the Collocation Space. Failure to meet this interval may result in construction and equipment installation delays. NewSouth shall forward a certificate of insurance and notice of cancellation/non-renewal to BellSouth at the following address:

BellSouth Telecommunications, Inc. Attn.: Risk Management Coordinator 675 W. Peachtree Street Rm. 17H53 Atlanta, Georgia 30375

- NewSouth must conform to recommendations made by BellSouth's fire insurance company to the extent BellSouth has agreed to, or shall hereafter agree to, such recommendations.
- Self-Insurance. If NewSouth's net worth exceeds five hundred million dollars (\$500,000,000), NewSouth may elect to request self-insurance status in lieu of obtaining any of the insurance required in Sections 8.2.1 and 8.2.3. NewSouth shall provide audited financial statements to BellSouth thirty (30) days prior to the commencement of any work in the Collocation Space. BellSouth shall then review such audited financial statements and respond in writing to NewSouth in the event that self-insurance status is not granted to NewSouth. If BellSouth approves NewSouth for self-insurance, NewSouth shall annually furnish to BellSouth, and keep current, evidence of such net worth that is attested to by one of NewSouth's corporate officers. The ability to self-insure shall continue so long as the NewSouth meets all of the requirements of this Section. If the NewSouth subsequently no longer satisfies this Section, NewSouth is required to purchase insurance as indicated by Sections 8.2.1 and 8.2.3.
- The net worth requirements set forth in Section 8.7 may be increased by BellSouth from time to time during the term of this Attachment upon thirty (30) days' notice to NewSouth to at least such minimum limits as shall then be customary with respect to comparable occupancy of BellSouth structures.
- Failure to comply with the provisions of this Section will be deemed a material breach of this Attachment.

9. Mechanics Liens

(BellSouth or NewSouth), or any improvement thereon by reason of or arising out of any labor or materials furnished or alleged to have been furnished or to be furnished to or for the other Party or by reason of any changes, or additions to said property made at the request or under the direction of the other Party, the other Party directing or requesting those changes shall, within thirty (30) calendar days after receipt of written notice from the Party against whose property said lien has been filed, either pay such lien or cause the same to be bonded off the affected property in the manner provided by law. The Party causing said lien to be placed against the property of the other shall also defend, at its sole cost and expense, on behalf of the other, any action, suit or proceeding which may be brought for the enforcement of such liens and shall pay any damage and discharge any judgment entered thereon.

10. Inspections

BellSouth shall conduct an inspection of NewSouth's equipment and facilities in the Collocation Space(s) prior to the activation of facilities between NewSouth's equipment and equipment of BellSouth. BellSouth may conduct an inspection if NewSouth adds equipment and may otherwise conduct routine inspections at reasonable intervals mutually agreed upon by the Parties. BellSouth shall provide NewSouth with a minimum of forty-eight (48) hours or two (2) business days, whichever is greater, advance notice of all such inspections. All costs of such inspection shall be borne by BellSouth.

11. Security and Safety Requirements

The security and safety requirements set forth in this section are as stringent as the 11.1 security requirements BellSouth maintains at its own premises either for their own employees or for authorized contractors. Only BellSouth employees, BellSouth Certified Contractors and authorized employees, authorized Guests, pursuant to Section 3.3, preceding, or authorized agents of NewSouth will be permitted in the BellSouth Premises. NewSouth shall provide its employees and agents with picture identification which must be worn and visible at all times while in the Collocation Space or other areas in or around the Premises. The photo Identification card shall bear, at a minimum, the employee's name and photo, and the NewSouth name. BellSouth reserves the right to remove from its premises any employee of NewSouth not possessing identification issued by NewSouth or who have violated any of BellSouth's policies as outlined in the CLEC Security Training documents. NewSouth shall hold BellSouth harmless for any damages resulting from such removal of its personnel from BellSouth premises. NewSouth shall be solely responsible for ensuring that any Guest of NewSouth is in compliance with all subsections of this Section 11.

- 11.1.1 NewSouth will be required, at its own expense, to conduct a statewide investigation of criminal history records for each NewSouth employee being considered for work on the BellSouth Premises, for the states/counties where the NewSouth employee has worked and lived for the past five years. Where state law does not permit statewide collection or reporting, an investigation of the applicable counties is acceptable.
- NewSouth will be required to administer to their personnel assigned to the BellSouth Premises security training either provided by BellSouth, or meeting criteria defined by BellSouth.
- 11.1.3 NewSouth shall not assign to the BellSouth Premises any personnel with records of felony criminal convictions. NewSouth shall not assign to the BellSouth Premises any personnel with records of misdemeanor convictions, except for misdemeanor traffic violations, without advising BellSouth of the nature and gravity of the offense(s). BellSouth reserves the right to refuse building access to any NewSouth personnel who have been identified to have misdemeanor criminal convictions. Notwithstanding the foregoing, in the even that NewSouth chooses not to advise BellSouth of the nature and gravity of any misdemeanor conviction, NewSouth may, in the alternative, certify to BellSouth that it shall not assign to the BellSouth Premises any personnel with records of misdemeanor convictions (other than misdemeanor traffic violations).
- 11.1.4 Neither Party shall not knowingly assign to the BellSouth Premises any individual who was a former employee of the other Party and whose employment with the other Party was terminated for a criminal offense whether or not the other Party sought prosecution of the individual for the criminal offense.
- Neither Party shall not knowingly assign to the BellSouth Premises any individual who was a former contractor of the other Party and whose access to a BellSouth Premises was revoked due to commission of a criminal offense whether or not the other Party sought prosecution of the individual for the criminal offense.
- 11.1.6 For each NewSouth employee requiring access to a BellSouth Premises pursuant to this Attachment, NewSouth shall furnish BellSouth, prior to an employee gaining such access, a certification that the aforementioned background check and security training were completed. The certification will contain a statement that no felony convictions were found and certifying that the security training was completed by the employee. If the employee's criminal history includes misdemeanor convictions, NewSouth will disclose the nature of the convictions to BellSouth at that time. In the alternative, NewSouth may certify to BellSouth that it shall not assign to the BellSouth Premises any personnel with records of misdemeanor convictions other than misdemeanor traffic violations.
- 11.1.7 At BellSouth's request, NewSouth shall promptly remove from the BellSouth's Premises any employee of NewSouth BellSouth does not wish to grant access to its premises 1) pursuant to any investigation conducted by BellSouth or 2) prior to the

initiation of an investigation in the event that an employee of NewSouth is found interfering with the property or personnel of BellSouth or another CLEC, provided that an investigation shall promptly be commenced by BellSouth.

- 11.2 Notification to BellSouth. BST reserves the right to interview NewSouth's employees, agents, or contractors in the event of wrongdoing in or around BellSouth's property or involving BellSouth's or another CLEC's property or personnel, provided that BellSouth shall provide reasonable notice to NewSouth's Security contact of such interview. NewSouth and its contractors shall reasonably cooperate with BellSouth's investigation into allegations of wrongdoing or criminal conduct committed by, witnessed by, or involving NewSouth's employees, agents, or contractors. Additionally, BellSouth reserves the right to bill NewSouth for all reasonable costs associated with reasonable and nondiscriminatory investigations involving its employees, agents, or contractors if it is established and mutually agreed in good faith that NewSouth's employees, agents, or contractors are responsible for the alleged act. BellSouth shall bill NewSouth for BellSouth property which is stolen or damaged where an investigation determines the culpability of NewSouth's employees, agents, or contractors and where NewSouth agrees, in good faith, with the results of such investigation. NewSouth shall notify BellSouth in writing immediately in the event that the CLEC discovers one of its employees already working on the BellSouth premises is a possible security risk. Upon request of the other Party, the Party who is the employer shall discipline consistent with its employment practices, up to and including removal from the BellSouth Premises, any employee found to have violated the security and safety requirements of this section. NewSouth shall hold BellSouth harmless for any damages resulting from such removal of its personnel from BellSouth premises.
- Use of Supplies. Unauthorized use of telecommunications equipment or supplies either Party, whether or not used routinely to provide telephone service (e.g. plug-in cards,) will be strictly prohibited and handled appropriately. Costs associated with such unauthorized use may be charged to the offending Party, as may be all associated investigative costs.
- Use of Official Lines. Except for non-toll calls necessary in the performance of their work, neither Party shall use the telephones of the other Party on the BellSouth Premises. Charges for unauthorized telephone calls may be charged to the offending Party, as may be all associated investigative costs.
- Accountability. Full compliance with the Security requirements of this section shall in no way limit the accountability of either Party to the other for the improper actions of its employees.

12. Destruction of Collocat.on Space

In the event a Collocation Space is wholly or partially damaged by fire, windstorm, 12.1 tornado, flood or by similar causes to such an extent as to be rendered wholly unsuitable for NewSouth's permitted use hereunder, then either Party may elect within ten (10) business days after such damage, to terminate this Attachment, and if either Party shall so elect, by giving the other written notice of termination, both Parties shall stand released of and from further liability under the terms hereof. If the Collocation Space shall suffer only minor damage and shall not be rendered wholly unsuitable for NewSouth's permitted use, or is damaged and the option to terminate is not exercised by either Party, BellSouth covenants and agrees to proceed promptly without expense to NewSouth, except for improvements not the property of BellSouth, to repair the damage. BellSouth shall have a reasonable time within which to rebuild or make any repairs, and such rebuilding and repairing shall be subject to delays caused by storms, shortages of labor and materials, government regulations, strikes, walkouts, and causes beyond the control of BellSouth, which causes shall not be construed as limiting factors, but as exemplary only. NewSouth may, at its own expense, accelerate the rebuild of its collocated space and equipment provided however that a BellSouth Certified Contractor is used and the necessary space preparation has been completed. Rebuild of equipment must be performed by a BellSouth Certified Vendor. If NewSouth's acceleration of the project increases the cost of the project, then those additional charges will be incurred by NewSouth. Where allowed and where practical, NewSouth may erect a temporary facility while BellSouth rebuilds or makes repairs. In all cases where the Collocation Space shall be rebuilt or repaired, NewSouth shall be entitled to an equitable abatement of rent and other charges, depending upon the unsuitability of the Collocation Space for NewSouth's permitted use, until such Collocation Space is fully repaired and restored and NewSouth's equipment installed therein (but in no event later than thirty (30) calendar days after the Collocation Space is fully repaired and restored). Where NewSouth has placed an Adjacent Arrangement pursuant to section 3.4, NewSouth shall have the sole responsibility to repair or replace said Adjacent Arrangement provided herein. Pursuant to this section, BellSouth will restore the associated services to the Adjacent Arrangement.

13. Eminent Domain

If the whole of a Collocation Space or Adjacent Arrangement shall be taken by any public authority under the power of eminent domain, then this Attachment shall terminate with respect to such Collocation Space or Adjacent Arrangement as of the day possession shall be taken by such public authority and rent and other charges for the Collocation Space or Adjacent Arrangement shall be paid up to that day with proportionate refund by BellSouth of such rent and charges as may have been paid in advance for a period subsequent to the date of the taking. If any part of the Collocation Space or Adjacent Arrangement shall be taken under eminent domain, BellSouth and NewSouth shall each have the right to terminate this Attachment with

respect to such Collocation Space or Adjacent Arrangement and declare the same null and void, by written notice of such intention to the other Party within ten (10) business days after such taking.

14. Nonexclusivity

NewSouth understands that this Attachment is not exclusive and that BellSouth may enter into similar agreements with other Parties. Assignment of space pursuant to all such agreements shall be determined by space availability and made on a first come, first served basis.

15. Notice of Non-Emergency and Emergency Work

- BellSouth shall provide NewSouth with written notice five (5) business days prior to those instances where BellSouth or its subcontractors may be performing non-emergency work that has a substantial likelihood of directly affecting the Collocation Space occupied by NewSouth, or that is directly related to circuits that support NewSouth equipment.
- BellSouth will inform NewSouth by telephone of emergency related activity that BellSouth or its subcontractors may be performing that has a substantial likelihood of directly affecting the Collocation Space occupied by NewSouth, or is directly related to circuits that support NewSouth equipment. Notification of any emergency related activity shall be made as soon as practicable after BellSouth learns that such emergency activity is necessary so that NewSouth can take any action required to monitor or protect its service.

EXHIBIT A: BELLSOUTH/CLEC-1 RATES – ALABAMA PHYSICAL COLLOCATION

Rates marked with an asterisk (*) are interim and are subject to true-up

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1BA	Application Fee	Per request	NA	\$3,760.00
PE1CA	Subsequent Application Fee	Per request	NA.	\$3,134.00
				Minimum
	Space Preparation Fees			
PE1SJ	(Note4)			
PE1SK	Firm Order Processing* Central Office Modifications*			\$1,211.00
PE1SL		Per sq. ft.	\$2.24	
1 1101	Common Systems	Per sq. ft.	\$3.01	
PE1SM	Modifications – Cageless* Common Systems	D		
. 210141	Modifications - Caged*	Per cage	\$102.16	
	Wodincations - Caged			
	Space Enclosure (100 sq. ft.			
054044	minimum)			
PE1BW	Welded Wire-mesh	Per first 100 sq. ft.	\$178.65	NA
PE1CW	Welded Wire-mesh	Per add'l 50 sq. ft.	\$17.52	NA
PE1PJ	Floor Space			
FEILO	Floor Space	Per sq. ft.	\$3.68	NA
PE1BD	Cable Installation	Per cable	NA	£4.754.00
		T. G. Gabie	IVA	\$1,751.00
PE1PM	Cable Support Structure	Per entrance cable	\$19.67	NA
·	Power			
	-48V DC Power*	Per amp	60.00	***
PE1FB	120V AC Power single phase*	Per breaker amp	\$9.00 \$5.63	NA
PE1FD	240V AC Power single phase*	Per breaker amp	\$11.26	•
PE1FE	120V AC Power three phase*	Per breaker amp	\$16.89	•
PE1FG	277 AC Power three phase*	Per breaker amp	\$38.99	•
	Power (Note 3)			
PE1PL	-48V DC Power	Per amp	\$7.14	
			Ψ1.14	
	Cross Connects			
PE1P2	2-wire	Per cross connect		First/Add'l
PE1P4	4-wire		\$.031	\$33.68/\$31.79
PE1P1	DS-1		\$.062	\$33.63/\$31.67
PE1P3	DS-3		\$1.28	\$52.93/\$39.87
PE1F2	2-fiber		\$16.27	\$51.99/\$38.59
,	- ··- • ·	1 1	\$3.23	\$52.00/\$38.60

	10	ARAMA (agatinus d)		
USOC	Rate Element Description	ABAMA (continued)		
	Trace Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1F4	4-fiber			
	. 11001		\$5.73	\$64.54/\$51.14
PE1AX	Security Access System Security System*	Per central office	\$52.27	
PE1A1	New Access Card Activation*	Per card	\$.059	\$5 5.57
PE1AA	Administrative change, existing card*	Per card	\$1000	\$15.58
PE1AR	Replace lost or stolen card*	Per card		₽4E EG
PE1AK	Initial Key	Per key	NA	\$45.56 \$26.19
PE1AL	Replace lost or stolen key	Per key	NA NA	\$26.19 \$26.19
PE1SR	Space Availability Report*	Per premises		
		requested		\$2,150.00
	POT Bay Arrangements	Description		
	Prior to 6/1/99	Per cross connect		
PE1PE	2-Wire Cross-Connect		\$0.08	A LA
PE1PF	4-Wire Cross-Connect		\$0.17	NA NA
PE1PG	DS1 Cross-Connect		\$0.69	NA NA
PE1PH	DS3 Cross-Connect		\$4.74	NA NA
PE1B2	2-Fiber Cross-Connect		\$32.02	NA
PE1B4	4-Fiber Cross-Connect		\$40.48	NA
	Cable Records ¹			Note 2
			·	Initial/Subsequent
PE1CR	Cable Records	Per request	NA	\$1708/\$1166
PE1CD	VG/DS0 Cable	Per cable record	NA I	\$923.51/\$923.51
PE1CO	VG/DS0 Cable	Per each 100 pair	NA	\$18.02/\$18.02
PE1C1 PE1C3	DS1	Per T1TIE	NA	\$8.44/\$8.44
PE1CB	DS3	Per T3TIE	NA	\$29.53/\$29.53
, 5,00	Fiber Cable	Per cable record	NA	\$278.95/\$278.95

	ALABAMA (continued)				
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)	
	Security Escort	Per half hr/add'l			
PE1BT	Basic Time	riali fir	NA	\$33.85/\$21.45	
PE1OT	Overtime		NA	\$44.09/\$27.71	
PE1PT Note(s):	Premium Time		NA NA	\$54.33/\$33.96	

N/A refers to rate elements which do not have a negotiated rate.

Note1: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.

Note 3: These Power rates will only apply for existing collocation arrangements provisioned prior to the execution of this agreement and Augments that make use of existing power.

Note 4: Recurring charges for Space Preparation will not apply to existing collocation arrangements for which NewSouth paid non-recurring Space Preparation charges.

EXHIBIT A: BELLSOUTH/CLEC-1 RATES - FLORIDA PHYSICAL COLLOCATION

USOC		Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1BA	Application Fee	Per request		\$3,791.00
DETOA	Out of the second of the secon			Ψ5,791.00
PE1CA	Subsequent Application Fee	Per request	NA	\$3,160.00
	Space Propagation Force			70,100,0
	Space Preparation Fees (Note 5)			
PE1SJ	Firm Order Processing			
PE1SK	Central Office Modifications	Por or 4		\$1,211.0
PE1SL	Common Systems	Per sq. ft.	\$2.58	
	Modifications - Cageless	Per sq. ft.	\$2.96	
PE1SM	Common Systems	Dor con		
	Modifications - Caged	Per cage	\$100.66	
	Oayeu	-		· · · · · · · · · · · · · · · · · · ·
	Space Enclosure (100 sq. ft.	 		
	minimum)			
PE1BW		Per first 100 sq. ft.	¢005.00	
PE1CW	Wire Cage	Per add'l 50 sq. ft.	\$205.93 \$20.20	N/
		1 01 444 100 34. 16.	\$20.20	NA
PE1PJ	Floor Space	Per sq. ft.	\$6.57	
			Ψ0.07	NA
PE1BD	Cable Installation	Per cable		\$1,826.00
PE1PM	Oatt O			Ψ1,020.00
PEIPM	Cable Support Structure		\$21.66	NA
	Power			
PE1FB	-48V DC Power	Per amp	\$8.86	NA
PE1FD	120V AC Power single phase	Per breaker amp	\$5.62	-
PE1FE	240V AC Power single phase	Per breaker amp	\$11.26	-
PE1FG	120V AC Power three phase	Per breaker amp	\$16.88	-
	277 AC Power three phase	Per breaker amp	\$38.98	-
PE1PL	Power (Note 4)			
	-48V DC Power	Don o		-
	407 DO I ONE!	Per amp	\$7.14	
	Cross Connects	 		
ļ	2-wire	Per cross sonner		First/Add'l
1	4-wire	Per cross connect Per cross connect	\$.074	\$34.53/\$32.51
	DS1	Per cross connect	\$.148	\$34.54/\$32.53
	DS3	Per cross connect	\$1.29	\$54.15/\$40.94
	2-fiber	Per cross connect	\$17.48	\$53.28/\$39.65
	4-fiber	Per cross connect	\$2.96	\$53.28/\$39.66
		1 - Si Cioss Collinect	\$5.66	\$66.08/\$52.47

	FL FL	ORIDA (continued)		
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1AX	Security Access System Security System	Per premises	\$89.48	
PE1A1 PE1AA	New Access Card Activation Administrative change, existing card	Per card Per card	\$.06	\$56.03 \$15.71
PE1AR	Replace lost or stolen card	Per card		
PE1AK PE1AL	Initial Key Replace lost or stolen key	Per key Per key	NA NA	\$45.93 \$26.41 \$26.41
PE1SR	Space Availability Report	Per premises requested		\$2,168.00
	POT Bay (Note 1)		NA	NA
	Cable Records ²			Note 3
PE1CR	Cohla Da			initial/subsequent
PE1CD PE1CO	Cable Records VG/DS0 Cable VG/DS0 Cable	Per request Per cable record Per each 100 pair	NA NA	\$1709/\$1166 \$923.86/\$923.86
PE1C1 PE1C3 PE1CB	DS1 DS3 Fiber Cable	Per T1TIE Per T3TIE Per cable record	NA NA NA	\$18.03/\$18.03 \$8.44/\$8.44 \$29.54/\$29.54
· · · · · · · · · · · · · · · · · · ·		i el capie record	NA NA	\$279.05/\$279.05
DE400	Security Escort	Per ¼ hour		
PE1BQ PE1OQ PE1PQ	Basic Time Overtime Premium Time		NA NA	\$10.89 \$13.64
Note(s):	remum nine		NA NA	\$16.40

N/A refers to rate elements which do not have a negotiated rate.

Note 1; POT Bays: BellSouth's Florida specific rates were established in the Florida Public Service Commission Docket No. 960833. The Commission did not set permanent rates for POT Bays, given the assumption by the Parties to the Proceeding that they will always provide their own POT Bays. It will be necessary for CLEC-1 to provide its own POT Bays per BellSouth specifications and provide the necessary information from which BellSouth can inventory.

Note 2: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

Note 3: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.

Note 4: These Power rates will only apply for existing collocation arrangements provisioned prior to the execution of this agreement and Augments that make use of existing power.

Note 5: Recurring charges for Space Preparation will not apply to existing collocation arrangements for which NewSouth paid non-recurring Space Preparation charges.

EXHIBIT A: BELLSOUTH/CLEC-1 RATES - GEORGIA PHYSICAL COLLOCATION

Rates marked with an asterisk (*) are interim and subject to true-up

USUC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1BA	Application Fee	Per request	NA	\$3,755.00
55101				Ψ0,700.00
PE1CA	Subsequent Application Fee	Per request	NA	\$3,130.00
				Minimum
PE1BB	Canada Describina			
FEIBB	Space Preparation Fee	Per sq. ft.	NA	\$100.00
PE1BW	Space Enclosure (100 sq. ft. minimum)			
PE1CW	Welded Wire-mesh	Per first 100 sq. ft.	\$187.36	NA
FEICAA	Welded Wire-mesh	Per add'l 50 sq. ft.	\$18.38	NA
PE1PJ	Floor Space Zone A	Per sq. ft.		
PE1PK	Zone B	Per sq. ft.	\$4.47	NA
		r er sq. it.	\$4.47	NA
PE1BD	Cable Installation	Per cable	NA	\$1,693.00
				Ψ1,093.00
PE1PM	Cable Support Structure	Per entrance cable	\$19.26	NA
	Dawes			
PE1PL	Power -48V DC Power			
PE1FB	120V AC Power single phase*	Per amp	\$5.00	NA
PE1FD	240V AC Power single phase*	Per breaker amp	\$5.52	-
PE1FE	120V AC Power three phase*	Per breaker amp	\$11.05	-
PE1FG	277 AC Power three phase*	Per breaker amp	\$16.58	•
	217 / O T OWEI tillee phase	Per breaker amp	\$38.27	•
	Cross Connects	Per cross connect		Fig. 1/A 2 14
PE1P2	2-wire	. Si SiOSS COMMECT	\$0.031	First/Add'l
PE1P4	4-wire		\$0.061	\$33.76/\$31.86
PE1P1	DS-1		\$1.13	\$33.77/\$31.80
PE1P3	DS-3		\$14.43	\$53.05/\$39.99 \$53.4/\$39.71
PE1F2	2-fiber		\$2.86	\$52.14/\$38.71 \$52.14/\$38.72
PE1F4	4-fiber	1	\$5.08	\$64.74/\$51.31

USOC	Rate Element Description	ORGIA (continued)		T
	riate Liement Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1AX	Society Access Continue			
	Security Access System Security System*	Per premises	\$40.00	
PE1A1	New Access Card Activation*	Per card	\$.058	\$55.5°
PE1AA	Administrative change, existing card*	Per card	4.555	\$15.56
PE1AR	Replace lost or stolen card*	Per card		045.50
PE1AK	Initial Key	Per key	NA	\$45.50
PE1AL	Replace lost or stolen key	Per key	NA NA	\$26.16 \$26.16
DETOD				Ψ20.10
PE1SR	Space Availability Report*	Per premises		\$2,148.00
		requested		
	POT Bay Arrangements	Per cross-connect		
	Prior to 6/1/99	rei cross-connect		
PE1PE	2-Wire Cross-Connect		\$0.40	A.
PE1PF	4-Wire Cross-Connect		\$0.40 \$1.20	NA NA
PE1PG	DS1 Cross-Connect		\$1.20 \$1.20	NA NA
PE1PH	DS3 Cross-Connect		\$8.00	NA
PE1B2	2 Fiber Cross-Connect		\$38.79	NA
PE1B4	4 Fiber Cross-Connect		\$52.31	NA NA
			402.01	INA
	Cable Records ¹			Note 2
				Initial/subsequent
PE1CR	Cable Records	Per request	NA	4.1994 10. 1.1.1.1
PE1CD	VG/DS0 Cable	Per cable record	· 1	\$1706/\$1164
PE1CO	VG/DS0 Cable	Per each 100 pair	NA	\$922.38/\$922.38
PE1C1	DS1	Per T1TIE	NA NA	\$18.00/\$18.00
PE1C3	DS3	Per T3TIE	NA I	\$8.43/\$8.43
PE1CB	Fiber Cable	Per cable record	NA NA	\$29.49/\$29.49
				\$278.61/\$278.61
	Socurity Econd			
	Security Escort	Per half hr./Add'l half hr.		
PE1BT	Basic Time		NA	\$33.81/\$21.42
E1OT	Overtime		NA	\$44.03/\$27.67
PE1PT	Premium Time		NA	\$54.26/\$33.92

N/A refers to rate elements which do not have a negotiated rate.

Note1: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.

EXHIBIT A: BELLSOUTH/CLEC-1 RATES - KENTUCKY PHYSICAL COLLOCATION

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1BA	Application Fee	Per request	NA	\$3,761.00
			14/1	Ψ3,701.00
PE1CA	Subsequent Application Fee	Per request	NA	\$3,135.00
				ψο, 133.00 Minimum
				withinati
	Space Preparation Fees			
PE1SJ	(Note 4)			
PE1SK	Firm Order Processing*			\$1,202.00
	Central Office Modifications*	Per sq. ft.	\$2.38	+ -,
PE1SL	Common Systems	Per sq. ft.	\$3.30	
DETCH	Modifications - Cageless*		·	
PE1SM	Common Systems	Per cage	\$112.11	
	Modifications - Caged*			
	Space Englacers (400 - 4)			
	Space Enclosure (100 sq. ft. minimum)			
PE1BW				
PE1CW	1	Per first 100 sq. ft.	\$189.85	NA
LICTY	Weided Wire-mesh	Per add'l 50 sq. ft.	\$18.62	NA
PE1PJ	Floor Space	Por on 4		
	· iou. Opass	Per sq. ft.	\$8.20	NANA
PE1BD	Cable Installation	Per cable	NA	04.755.00
		1 Ci Cable	INA	\$1,755.00
PE1PM	Cable Support Structure	Per entrance cable	\$20.14	NA
			Ψ20.14	INA
	Power			
55.55	-48V DC Power*	Per amp	\$8.77	NA
PE1FB	120V AC Power single phase*	Per breaker amp	\$5.58	
PE1FD	240V AC Power single phase*	Per breaker amp	\$11.16	_
PE1FE	120V AC Power three phase*	Per breaker amp	\$16.74	_
PE1FG	277 AC Power three phase*	Per breaker amp	\$38.65	•
DE1D!	401/ 00 0			
PE1PL	-48V DC Power (Note 3)	Per amp	\$7.68	NA
	Cross Constant			
PE1P2	Cross Connects	Per cross connect		First/Add'l
PE1P4	2-wire		\$0.037	\$33.67/\$31.78
PE1P1	4-wire		\$0.075	\$33.66/\$31.70
PE1P3	DS-1		\$1.51	\$52.97/\$39.90
PE1F2	DS-3		\$19.15	\$52.04/\$38.62
PE1F4	2-fiber		\$3.80	\$52.04/\$38.63
<u> </u>	4-fiber		\$6.75	\$64.59/\$51.18

	KEN	ITUCICY (continued)		
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1AX	Security Access System	Per premises	\$78.11	
	Security System*	r or premises	\$78.11	
PE1A1	New Access Card Activation	Per card	\$.059	\$55.59
PE1AA	Administrative change, existing card	Per card	4.000	\$15.59
PE1AR	Replace lost or stolen card	Per card		\$45.58
PE1AK	Initial Key	Per key	NA	\$26.20
PE1AL	Replace lost or stolen key	Per key	NA NA	\$26.20 \$26.20
PE1SR	Space Availability Report	Dor promises		
. = 1011	Opace Availability Report	Per premises		\$2,151
		requested		
	POT Bay Arrangements Prior to 6/1/99	Per cross-connect		
PE1PE	2-Wire Cross-Connect		00.00	
PE1PF	4-Wire Cross-Connect		\$0.06	NA
PE1PG	DS1 Cross-Connect		\$0.15 \$0.58	NA
PE1PH	DS3 Cross-Connect		\$0.58 \$4.51	NA
PE1B2	2 Fiber Cross-Connect		\$38.79	NA
PE1B4	4 Fiber Cross-Connect		\$52.31	NA NA
-	Security Escort	Per half hr./Add'l		
	•	half hr.		
PE1BT	Basic Time		NA	\$33.86/\$21.46
PE10T	Overtime		NA	\$44.10/\$27.72
PE1PT	Premium Time		NA	\$54.35/\$33.97
	Cable Records			
	Odbie Necolds			Note 2
				Initial/subsequent
PE1CR	Cable Records	Per request	NA	\$1709/1166
PE1CD	VG/DS0 Cable	Per cable record	NA	\$923.83/\$923.83
PE1CO PE1C1	VG/DS0 Cable	Per each 100 pair	NA	\$18.03/\$18.03
PE1C1	DS1	Per T1TIE	NA	\$8.44/\$8.44
PE1CB	DS3 Fiber Cable	Per T3TIE	NA	\$29.54/\$29.54
LICE	Fiber Cable	Per cable record	NA	\$279.05/\$279.05
Vote(s):				

N/A refers to rate elements which do not have a negotiated rate.

Note1: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.

Note 3: These Power rates will only apply for existing collocation arrangements provisioned prior to the execution of this agreement and Augments that make use of existing power.

Note 4: Recurring charges for Space Preparation will not apply to existing collocation arrangements for which NewSouth paid non-recurring Space Preparation charges...

EXHIBIT A: BELLSOUTH/CLEC-1 RATES - LOUISIANA PHYSICAL COLLOCATION

Rates marked with an asterisk (*) are interim and are subject to true-up.

USOC	narked with an asterisk (*) are in Rate Element Description	Unit	Recurring	Non-Recurring
			Rate (RC)	Rate (NRC)
PE1BA	Application Fee	Per request	NA NA	
			IAV	\$3756.00
PE1CA	Subsequent Application Fee	Per request	NA	60101.00
			130	\$3131.00
				Minimum
	Space Preparation Fees		· · · · · · · · · · · · · · · · · · ·	
	(Note 4)			:
PE1SJ	Firm Order Processing*			\$1,200.00
PE1SK	Central Office Modifications*	Per sq. ft.	\$2.60	\$1,200.00
PE1SL	Common Systems	Per sq. ft.	\$3.15	
	Modifications - Cageless*		ΨΟ. 13	
PE1SM	Common Systems	Per cage	\$105.87	
	Modifications - Caged*		Ψ105.07	
	Space Enclosure (100 sq. ft.			
	minimum)	•		
PE1BW	Welded Wire-mesh	Per first 100 sq. ft.	\$207.06	NA
PE1CW	Welded Wire-mesh	Per add'l 50 sq. ft.	\$20.31	NA NA
			Ψ2.0.07	NA
PE1PJ	Floor Space	Per sq. ft.	\$5.94	NA
				, NA
PE1BD	Cable Installation	Per cable	NA	\$1,753.00
55454				\$1,700.00
PE1PM	Cable Support Structure	Per entrance cable	\$21.16	NA
55454	Power			
PE1PL	-48V DC Power*	Per amp	\$9.20	NA
PE1FB	120V AC Power single phase*	Per breaker amp	\$5.66	_
PE1FD	240V AC Power single phase*	Per breaker amp	\$11.34	-
PE1FE	120V AC Power three phase*	Per breaker amp	\$17.00	_
PE1FG	277 AC Power three phase*	Per breaker amp	\$39.26	_
DE4D1	401/505			
PE1PL	-48V DC Power (Note 3)	Per amp	\$7.15	NA
DE4DO	Cross Connects	Per cross connect		First/Add'l
PE1P2	2-wire		\$0.036	\$33.61/\$31.76
PE1P4	4-wire		\$0.073	\$33.53/\$31.58
PE1P1	DS-1		\$1.20	\$52.80/\$39.76
PE1P3	DS-3	<u> </u>	\$15.26	\$51.86/\$38.49

	LO	JISIANA (continued)		
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
DE4E0	Cross Connects (continued)	Per cross connect		First/Add'l
PE1F2	2-fiber 4-fiber		\$3.03	\$51.86/\$38.49
I LIF4	4-11ber		\$5.38	\$64.36/\$50.99
PE1AX	Society Assessed			
	Security Access System Security System*	Per premises	\$60.60	
PE1A1	New Access Card Activation*	Per card	\$.060	\$55.51
PE1AA	Administrative change, existing card*	Per card	,	\$15.57
PE1AR	Replace lost or stolen card	Per card		\$45.51
PE1AK PE1AL	Initial Key	Per key	NA	\$26.16
PEIAL	Replace lost or stolen key	Per key	NA	\$26.16
PE1SR	Space Availability Report*	Por promises		
	Tepon	Per premises requested		\$2,148
PE1PE	POT Bay Arrangements Prior to 6/1/99	Per cross-connect		
PE1PE	2-Wire Cross-Connect		\$0.0776	NA
PE1PG	4-Wire Cross-Connect DS1 Cross-Connect		\$0.1552	NA
PE1PH	DS3 Cross-Connect		\$0.6406	NA Ì
PE1B2	2 Fiber Cross-Connect		\$4.75	NA
PE1B4	4 Fiber Cross-Connect		\$47.44	NA
			\$63.97	NA
	Cable Records ¹			Note 2
				Initial/subsequent
PE1CR	Cable Records	Per request	NA	\$1706/\$1165
PE1CD	VG/DS0 Cable	Per cable record	NA	\$922.51/\$922.51
PE1CO PE1C1	VG/DS0 Cable	Per each 100 pair	NA	\$18.00/\$18.00
PE1C3	DS1 DS3	Per T1TIE	NA	\$8.43/\$8.43
PE1CB	Fiber Cable	Per T3TIE	NA (\$29.49/\$29.49
. 2.00	- Inci Capie	Per cable record	NA	\$278.65/\$278.65

LOUISIANA (continued)				
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
	Security Escort	Per half hr./Add'l	(100)	Hate (MAC)
PE1BT	Basic Time	half hr.		
PE1OT			NA	\$33.97/\$21.53
PE1PT	Overtime		NA I	\$44.25/\$27.81
Note/s)	Premium Time		NA	\$54.53/\$34.09

N/A refers to rate elements which do not have a negotiated rate.

Note1: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.

Note 3: These Power rates will only apply for existing collocation arrangements provisioned prior to the execution of this agreement and Augments that make use of existing power.

Note 4: Recurring charges for Space Preparation will not apply to existing collocation arrangements for which NewSouth paid non-recurring Space Preparation charges...

EXHIBIT A: BELLSOUTH/CLEC-1 RATES - MISSISSIPPI PHYSICAL COLLOCATION

Rates marked with an asterisk (*) are interim and are subject to true-up.

Nates I	narked with an asterisk (*) ar			
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring
PE1BA	Application Fee	Per request	NA NA	Rate (NRC)
		Torrequest	IVA	\$3,755.00
PE1CA	Subsequent Application Fee	Per request	NA	\$3,130.00
	, , , , , , , , , , , , , , , , , , , ,		147	Minimum
				iviiiiiiuni
	Space Preparation Fees			
	(Note 4)			
PE1SJ	Firm Order Processing*			\$1,200.00
PE1SK	Central Office Modifications*	Per sq. ft.	\$2.61	Ψ1, 2 00.00
PE1SL	Common Systems	Per sq. ft.	\$2.88	
	Modifications - Cageless*	·	,	
PE1SM	Common Systems	Per cage	\$97.85	
	Modifications - Caged*			
	Space Enclosure(100 sq. ft.	`		
PE1BW	minimum)			
PE1CW	Welded Wire-mesh	Per first 100 sq. ft.	\$208.30	NA
PETCVV	Welded Wire-mesh	Per add'l 50 sq. ft.	\$20.43	NA NA
PE1PJ	Floor Space	Dance (
1 2 11 0	1 loor Space	Per sq. ft.	\$6.53	
PE1BD	Cable Installation	Per cable	NA NA	£4.074.00
		1 el cable	NA NA	\$1,871.00
PE1PM	Cable Support Structure	Per entrance cable	\$19.90	NA
			Ψ10.50	INA
	Power			
PE1PL	-48V DC Power*	Per amp	\$8.96	NA
PE1FB	120V AC Power single phase*	Per breaker amp	\$5.61	-
PE1FD	240V AC Power single phase*	Per breaker amp	\$11.23	
PE1FE	120V AC Power three phase*	Per breaker amp	\$16.84	-
PE1FG	277 AC Power three phase*	Per breaker amp	\$38.89	
PE1PL	404.00.0	_		
PEIPL	-48V DC Power (Note 3)	Per amp	\$6.93	NA NA
	Cross Connects			
PE1P2	2-wire	Per cross connect		First/Add'l
PE1P4	4-wire		\$.038	\$33.65/\$31.77
	7-4411.0	11	\$.076	\$33.46/\$31.52

	MISSISSIPPI (continued)				
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)	
DE4.D4	Cross Connects (continued)	Per cross connect		First/Add'l	
PE1P1	DS-1		\$1.30		
PE1P3	DS-3		\$16.55	\$51.78/\$38.43	
PE1F2 PE1F4	2-fiber		\$3.28	\$51.78/\$38.43	
PE 1F4	4-fiber		\$5.83	\$64.27/\$50.91	
			 		
PE1AX	Security Access System Security System*	Per premises	\$85.54		
PE1A1	New Access Card Activation*	Per card	\$.061	\$55.50	
PE1AA	Administrative change, existing	Per card	\$.001	\$55.50 \$45.50	
	card*			\$15.56	
PE1AR	Replace lost or stolen card	Per card		\$45.50	
PE1AK	Initial Key	Per key	NA	\$26.16	
PE1AL	Replace lost or stolen key	Per key	NA NA	\$26.16	
PE1SR	Space Availability Report*	B			
1011	Space Availability Report	Per premises .		\$2,147.00	
		requested			
	POT Bay Arrangements	Per cross-connect	<u> </u>		
	Prior to 6/1/99	1 of oross-connect			
PE1PE	2-Wire Cross-Connect		\$0.1195	NA	
PE1PF	4-Wire Cross-Connect		\$0.2389	NA NA	
PE1PG	DS1 Cross-Connect		\$0.9862	NA NA	
PE1PH	DS3 Cross-Connect		\$5.81	NA NA	
PE1B2	2 Fiber Cross-Connect		\$38.79	NA NA	
PE1B4	4 Fiber Cross-Connect		\$52.31	, NA	
	Cable Records				
	Cable Records			Note 2	
_				Initial/subsequent	
PE1CR	Cable Records	Per request	NA	\$1706/1164	
PE1CD	VG/DS0 Cable	Per cable record	NA	\$922.28/\$922.28	
PE1CO	VG/DS0 Cable	Per each 100 pair	NA	\$18.00/\$18.00	
PE1C1	DS1	Per T1TIE	NA	\$8.42/\$8.42	
PE1C3	DS3	Per T3TIE	NA	\$29.49/\$29.49	
PE1CB	Fiber Cable	Per cable record	NA	\$278.58/\$278.58	
				i	

	MISSISSIPPI (continued)				
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)	
	Security Escort	Per half hr./Add'l half hr.		riate (iiiio)	
PE1BT	Basic Time		NA	\$33.80/\$21.42	
PE1OT	Overtime		NA	\$44.03/\$27.67	
PE1PT	Premium Time		NA	\$54.26/\$33.92	

N/A refers to rate elements which do not have a negotiated rate.

Note1: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.

Note 3: These Power rates will only apply for existing collocation arrangements provisioned prior to the execution of this agreement and Augments that make use of existing power.

Note 4: Recurring charges for Space Preparation will not apply to existing collocation arrangements for which NewSouth paid non-recurring Space Preparation charges.

EXHIBIT A: BELLSOUTH/CLEC-1 RATES - NORTH CAROLINA PHYSICAL COLLOCATION

Rates marked with an asterisk (*) are interim and are subject to true-up.

	marked with an asterisk (*) are interim and are subject to true-up.					
USOC	Rate Element Description	Unit	Recurring	Non-Recurring		
DEADA	Application Foot		Rate (RC)	Rate (NRC)		
PE1BA	Application Fee*	Per request	NA NA	\$3,850.00		
PE1CA	Subsequent Application Fee	Dos sociosal				
FEICA	Subsequent Application Fee	Per request	NA	\$3,119.00		
				Minimum		
	Space Preparation Fees					
	(Note 4)					
	Central Office Modification*	Per sq. ft.	\$1.57			
	Common Systems Modification	Per sq. ft.	\$1.57 \$3.26			
	- Cageless*	1 01 34. 16.	φ3.20			
	Common Systems Modification	Per cage	\$110.79			
	- Caged*	3.5	4 1.0.70			
	Power*	Per nominal -48v	\$5.76			
		DC Amp				
	Space Enclosure (100 sq. ft.					
DEADW	minimum)					
PE1BW	Welded Wire-mesh*	Per first 100 sq. ft.	\$102.76	NA		
PE1CW	Welded Wire-mesh*	Per add'l 50 sq. ft.	\$10.44	NA NA		
PE1PJ	Floor Space*	Dor on 4	00.45			
	1 loci opace	Per sq. ft.	\$3.45	NA NA		
PE1BD	Cable Installation*	Per cable	NA	\$2,305.00		
		1 or oabio		\$2,303.00		
PE1PM	Cable Support Structure*	Per entrance cable	\$21.33	NA		
			<u> </u>			
	Power		-			
	-48V DC Power*	Per amp	\$6.65	NA		
PE1FB	120V AC Power single phase*	Per breaker amp	\$5.50	-		
PE1FD	240V AC Power single phase*	Per breaker amp	\$11.01	-		
PE1FE	120V AC Power three phase*	Per breaker amp	\$16.51	-		
PE1FG	277 AC Power three phase*	Per breaker amp	\$38.12	•		
PE1PL	49V DC Power (Note 0)	5				
	-48V DC Power (Note 3)	Per amp	\$5.00	NA		
	Cross Connects (Note 1)	Per cross connect		First/Add'l		
PE1P2	2-wire*	. or oross connect	\$0.32	\$41.78/\$39.23		
PE1P4	4-wire*		\$0.64	\$41.91/\$39.25		
PE1P1	DS-1*		\$2.34	\$71.02/\$51.08		
PE1P3	DS-3*		\$42.84	\$69.84/\$49.43		
PE1F2	2-fiber		\$2.94	\$51.97/\$38.59		
PE1F4	4-fiber		\$5.62	\$64.53/\$51.15		

	NORTH	CAROLINA (continue	ed)	
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1AX	Socurity Access System	Por promises	\$41.03	
	Security Access System Security System*	Per premises		
PE1A1	New Access Card Activation*	Per card	\$.062	\$55.30
PE1AA	Administrative change, existing card*	Per card		\$15.51
PE1AR	Replace lost or stolen card	Per card	_	\$45.34
PE1AK	Initial Key	Per key	NA	\$26.18
PE1AL	Replace lost or stolen key	Per key	NA NA	\$26.18
PE1SR	Space Availability Report*	Por promises		£0.140.00
FEISH	Space Availability Report	Per premises requested		\$2,140.00
	POT Bay Arrangements Prior to 6/1/99	Per cross-connect		
PE1PE	2-Wire Cross-Connect		\$0.10	NA
PE1PF	4-Wire Cross-Connect		\$0.19	NA NA
PE1PG	DS1 Cross-Connect		\$0.79	NA NA
PE1PH	DS3 Cross-Connect		\$4.85	NA NA
PE1B2	2 Fiber Cross-Connect		\$45.30	NA.
PE1B4	4 Fiber Cross-Connect		\$61.09	NA NA
	Security Escort	Per half hr./Add'l half hr.		- 1
PE1BT	Basic Time		NA	\$42.92/\$25.56
PE1OT	Overtime		NA	\$54.51/\$32.44
PE1PT	Premium Time		NA	\$66.10/\$39.32
	Cable Records ¹			Note 2
				Initial/subsequent
PE1CR	Cable Records	Per request	NA	\$1707/\$1165
PE1CD	VG/DS0 Cable	Per cable record	NA	\$923.08/\$923.08
PE1CO	VG/DS0 Cable	Per each 100 pair	NA	\$18.02/\$18.02
PE1C1	DS1	Per T1TIE	NA	\$8.43/\$8.43
PE1C3	DS3	Per T3TIE	NA	\$29.51/\$29.51
PE1CB	Fiber Cable	Per cable record	NA	\$278.82/\$278.82
		1		

EXHIBIT A: BELLSOUTH/CLEC-1 RATES - NORTH CAROLINA PHYSICAL COLLOCATION (continued)

Note(s):

N/A refers to rate elements which do not have a negotiated rate.

Note1: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.

Note 3: These Power rates will only apply for existing collocation arrangements provisioned prior to the execution of this agreement and Augments that make use of existing power.

Note 4: Recurring charges for Space Preparation will not apply to existing collocation arrangements for which NewSouth paid non-recurring Space Preparation charges...

EXHIBIT A: BELLSOUTH/CLEC-1 RATES – SOUTH CAROLINA PHYSICAL COLLOCATION

Rates marked with an asterisk (*) are interim and are subject to true-up.

USOC	Rate Element Description	Unit	Recurring	Non-Recurring
			Rate (RC)	Rate (NRC)
PE1BA	Application Fee	Per request	NA NA	\$3768.00
PE1CA	Subsequent Application Fee	Per request	NA	\$3,141.00
				Minimum
	Space Preparation Fees			
	(Note 4)			
PE1SJ	Firm Order Processing*			\$1,204.00
PE1SK	Central Office Modifications*	Per sq. ft.	\$2.75	\$1,204.00
PE1SL	Common Systems	Per sq. ft.	\$3.24	
	Modifications - Cageless*	1 01 04.10.	Ψ0.24	
PE1SM	Common Systems	Per cage	\$110.17	
	Modifications - Caged*	· · · · · · · · · ·	4	
	Space Enclosure (100 sq. ft.			
,	minimum)			
PE1BW	Welded Wire-mesh	Per first 100 sq. ft.	\$219.19	NA
PE1CW	Welded Wire-mesh	Per add'l 50 sq. ft.	\$21.50	NA NA
55.01		<u> </u>		· · · · · · · · · · · · · · · · · · ·
PE1PJ	Floor Space	Per sq. ft.	\$3.95	NA_
PE1BD	Cable Installation	Per cable	NA	\$1,621.00
1 2100	Cable installation	rei cable	INA	\$1,021.00
PE1PM	Cable Support Structure	Per entrance cable	\$21.33	NA
				•
	Power			
	-48V DC Power*	Per amp	\$9.19	NA
PE1FB	120V AC Power single phase*	Per breaker amp	\$5.67	-
PE1FD	240V AC Power single phase*	Per breaker amp	\$11.36	-
PE1FE	120V AC Power three phase*	Per breaker amp	\$17.03	•
PE1FG	277 AC Power three phase*	Per breaker amp	\$39.33	•
PE1PL	-48V DC Power (Note 3)	Por omo	\$7.00	NA
) ha () ha	TOT DO FOWER (NOTE 3)	Per amp	\$7.09	IVA
	Cross Connects	Per cross connect		First/Add'l
PE1P2	2-wire	1 ST STORES CONTINECT	\$.034	\$33.75/\$31.86
PE1P4	4-wire		\$.068	\$33.71/\$31.75
PE1P1	DS-1		\$1.12	\$53.05/\$39.96
PE1P3	DS-3		\$14.21	\$52.11/\$38.68
PE1F2	2-fiber		\$2.82	\$52.11/\$38.69
PE1F4	4-fiber		\$5.01	\$64.69/\$51.26

	SOUTH CAR()LINA (continued)					
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)		
35.43						
PE1AX	Security Access System Security System*	Per premises	\$74.12			
PE1A1	New Access Card Activation*	Per card	\$.060	\$55.70		
PE1AA	Administrative change, existing card*	Per card		\$15.62		
PE1AR	Replace lost or stolen card	Per card		\$45.66		
PE1AK	Initial Key	Per key	NA	\$26.25		
PE1AL	Replace lost or stolen key	Per key	NA.	\$26.25		
PE1SR	Space Availability Report*	Per premises requested		\$2,155.00		
	POT Bay Arrangements Prior to 6/1/99	Per cross-connect				
PE1PE	2-Wire Cross-Connect		\$0.1091	NA		
PE1PF	4-Wire Cross-Connect		\$0.2181	NA		
PE1PG	DS1 Cross-Connect	•	\$0.9004	NA.		
PE1PH	DS3 Cross-Connect		\$5.64	NA		
PE1B2	2 Fiber Cross-Connect		\$37.36	NA		
PE1B4	4 Fiber Cross-Connect		\$50.38	NA		
	Security Escort	Per half hr./Add'l half hr.		.,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		
PE1BT	Basic Time		NA	\$33.92/\$21.50		
PE10T	Overtime	-	NA	\$44.19/\$27.77		
PE1PT	Premium Time		NA NA	\$54.45/\$34.04		
	Cable Records ¹			Note 2		
				Initial/subsequent		
PE1CR	Cable Records	Per request	NA NA	\$1712/\$1168		
PE1CD	VG/DS0 Cable	Per cable record	NA	\$925.57/\$925.57		
PE1CO	VG/DS0 Cable	Per each 100 pair	NA	\$18.06/\$18.06		
PE1C1	DS1	Per T1TIE	NA	\$8.45/\$8.45		
PE1C3	DS3	Per T3TIE	NA	\$29.59/\$29.59		
PE1CB	Fiber Cable	Per cable record	NA	\$279.57/\$279.57		

EXHIBIT A: BELLSOUTH/CLEC-1 RATES – SOUTH CAROLINA PHYSICAL COLLOCATION (continued)

Note(s):

N/A refers to rate elements which do not have a negotiated rate.

Note1: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.

Note 3: These Power rates will only apply for existing collocation arrangements provisioned prior to the execution of this agreement and Augments that make use of existing power.

Note 4: Recurring charges for Space Preparation will not apply to existing collocation arrangements for which NewSouth paid non-recurring Space Preparation charges.

EXHIBIT A: BELLSOUTH/CLEC-1 RATES – TENNESSEE PHYSICAL COLLOCATION

* Rates a	* Rates are interim and are subject to true-up.						
usoc	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)			
PE1BA	Application Fee	Per request	NA	\$3,767.00			
PE1CA	Subsequent Application Fee	Per request	NA	\$3,140.00			
				Minimum			
		·					
	Space Preparation Fees						
	(Note 4)						
PE1SJ	Firm Order Processing*			\$1,204.00			
PE1SK	Central Office Modifications*	Per sq. ft.	\$2.74				
PE1SL	Common Systems	Per sq. ft.	\$2.95				
	Modifications - Cageless*						
PE1SM	Common Systems	Per cage	\$100.14				
	Modifications - Caged*						
	Space Enclosure (100 sq. ft.						
	minimum)						
PE1BW	Welded Wire-mesh	Per first 100 sq. ft.	\$218.53	NA			
PE1CW	Welded Wire-mesh	Per add'l 50 sq. ft.	\$21.44	NA NA			
PE1PJ	Floor Space	Per sq. ft.	\$6.75	NA			
		<u> </u>		A. ====			
PE1BD	Cable Installation	Per cable	NA NA	\$1,757.00			
DEADM	Ophia Command Observations	Dor ontropo onblo	640.00	NIA.			
PE1PM	Cable Support Structure	Per entrance cable	\$19.80	NA			
	Power						
	-48V DC Power*	Per amp	\$8.87	NA			
PE1FB	120V AC Power single phase*	Per breaker amp	\$5.60	-			
PE1FD	240V AC Power single phase*	Per breaker amp	\$11.22				
PE1FE	120V AC Power three phase*	Per breaker amp	\$16.82				
PE1FG	277 AC Power three phase*	Per breaker amp	\$38.84	_			
	Errito i over ance priase	T or broaker amp	400.01				
PE1PL	-48V DC Power (Note 3)	Per amp	\$5.00	NA			
<u> </u>							
	Cross Connects	Per cross connect		First/Add'l			
PE1P2	2-wire		\$0.033	\$33.82/\$31.92			
PE1P4	4-wire		\$0.066	\$33.94/\$31.95			
PE1P1	DS-1		\$1.51	\$53.27/\$40.16			
PE1P3	DS-3		\$19.26	\$52.37/\$38.89			
PE1F2	2-fiber		\$3.82	\$52.37/\$38.89			
PE1F4	4-fiber		\$6.79	\$65.03/\$51.55			

	TENN	IESSEE (continued)		
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1AX	Security Access System Security System	Per premises	\$55.99	
PE1A1	New Access Card Activation	Per card	\$.059	\$55.67
PE1AA	Administrative change, existing card	Per card		\$15.61
PE1AR	Replace lost or stolen card	Per card		\$45.64
PE1AK	Initial Kev	Per key	NA	\$26.24
PE1AL	Replace lost or stolen key	Per key	NA	\$26.24
PE1SR	Space Availability Report*	Per premises requested		\$2,154.00
	POT Bay Arrangements Prior to 6/1/99	Per cross-connect		
PE1PE	2-Wire Cross-Connect		\$0.40	NA
PE1PF	4-Wire Cross-Connect	.	\$1.20	NA
PE1PG	DS1 Cross-Connect		\$1.20	NA
PE1PH	DS3 Cross-Connect		\$8.00	NA
PE1B2	2 Fiber Cross-Connect		\$38.79	NA
PE184	4 Fiber Cross-Connect		\$52.31	NA
	Security Escort	Per half hr./Add'l half hr.		
PE1BT	Basic Time		NA	\$33.91/\$21.49
PE1OT	Overtime		NA	\$44.17/\$27.76
PE1PT	Premium Time		NA	\$54.42/\$34.02
	Cable Records			Note 2
				Initial/subsequent
PE1CR	Cable Records	Per request	NA	\$1711/\$1168
PE1CD	VG/DS0 Cable	Per cable record	NA	\$925.06/\$925.06
PE1CO	VG/DS0 Cable	Per each 100 pair	NA	\$18.05/\$18.05
PE1C1	DS1	Per T1TIE	NA	\$8.45/\$8.45
PE1C3	DS3	Per T3TIE	NA	\$29.57/\$29.57
PE1CB	Fiber Cable	Per cable record	NA	\$279.42/\$279.42
Note(s):				

N/A refers to rate elements which do not have a negotiated rate.

Note1: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.

Note 3: These Power rates will only apply for existing collocation arrangements provisioned prior to the execution of this agreement and Augments that make use of existing power.

Note 4: Recurring charges for Space Preparation will not apply to existing collocation arrangements for which NewSouth paid non-recurring Space Preparation charges.

EXHIBIT B
Page 1 of 4

ENVIRONMENTAL AND SAFETY PRINCIPLES

The following principles provide basic guidance on environmental and safety issues when applying for and establishing Physical Collocation arrangements.

1. GENERAL PRINCIPLES

- Compliance with Applicable Law. BellSouth and NewSouth agree to comply with applicable federal, state, and local environmental and safety laws and regulations including U.S. Environmental Protection Agency (USEPA) regulations issued under the Clean Air Act (CAA), Clean Water Act (CWA), Resource Conservation and Recovery Act (RCRA), Comprehensive Environmental Response, Compensation and Liability Act (CERCLA), Superfund Amendments and Reauthorization Act (SARA), the Toxic Substances Control Act (TSCA), and OSHA regulations issued under the Occupational Safety and Health Act of 1970, as amended and NFPA and National Electrical Codes (NEC) and the NESC ("Applicable Laws"). Each Party shall notify the other if compliance inspections are conducted by regulatory agencies and/or citations are issued that relate to any aspect of this Attachment.
- Notice. BellSouth and NewSouth shall provide notice to the other, including Material Safety Data Sheets (MSDSs), of known and recognized physical hazards or Hazardous Chemicals existing on site or brought on site. Each Party is required to provide specific notice for known potential Imminent Danger conditions. NewSouth should contact 1-800-743-6737 for BellSouth MSDS sheets.
- 1.3 Practices/Procedures. BellSouth may make available additional environmental control procedures for NewSouth to follow when working at a BellSouth Premises (See Section 2, below). These practices/procedures will represent the regular work practices required to be followed by the employees and contractors of BellSouth for environmental protection. NewSouth will require its contractors, agents and others accessing the BellSouth Premises to comply with these practices. Section 2 lists the Environmental categories where BST practices should be followed by CLEC when operating in the BellSouth Premises.
- 1.4 Environmental and Safety Inspections. BellSouth reserves the right to inspect the NewSouth space with proper notification. BellSouth reserves the right to stop any NewSouth work operation that imposes Imminent Danger to the environment, employees or other persons in the area or Facility.
- 1.5 <u>Hazardous Materials Brought On Site</u>. Any hazardous materials brought into, used, stored or abandoned at the BellSouth Premises by NewSouth are owned by NewSouth. NewSouth will indemnify BellSouth for claims, lawsuits or damages to

persons or property caused by these materials. Without prior written BellSouth approval, no substantial new safety or environmental hazards can be created by NewSouth or different hazardous materials used by NewSouth at BellSouth Facility. NewSouth must demonstrate adequate emergency response capabilities for its materials used or remaining at the BellSouth Facility.

- 1.6 Spills and Releases. When contamination is discovered at a BellSouth Premises, the Party discovering the condition must notify BellSouth. All Spills or Releases of regulated materials will immediately be reported by NewSouth to BellSouth.
- Coordinated Environmental Plans and Permits. BellSouth and NewSouth will coordinate plans, permits or information required to be submitted to government agencies, such as emergency response plans, spill prevention control and countermeasures (SPCC) plans and community reporting. If fees are associated with filing, BellSouth and NewSouth will develop a cost sharing procedure. If BellSouth's permit or EPA identification number must be used, NewSouth must comply with all of BellSouth's permit conditions and environmental processes, including environmental "best management practices (BMP)" (see Section 2, below) and/or selection of BST disposition vendors and disposal sites.
- Environmental and Safety Indemnification. BellSouth and NewSouth shall indemnify, defend and hold harmless the other Party from and against any claims (including, without limitation, third-party claims for personal injury or death or real or personal property damage), judgments, damages, (including direct and indirect damages, and punitive damages), penalties, fines, forfeitures, costs, liabilities, interest and losses arising in connection with the violation or alleged violation of any Applicable Law or contractual obligation or the presence or alleged presence of contamination arising out of the acts or omissions of the indemnifying Party, its agents, contractors, or employees concerning its operations at the Facility.

2. CATEGORIES FOR CONSIDERATION OF ENVIRONMENTAL ISSUES

When performing functions that fall under the following Environmental categories on BellSouth's Premises, NewSouth agrees to comply with the applicable sections of the current issue of BellSouth's Environmental and Safety Methods and Procedures (M&Ps), incorporated herein by this reference. NewSouth further agrees to cooperate with BellSouth to ensure that NewSouth's employees, agents, and/or subcontractors are knowledgeable of and satisfy those provisions of BellSouth's Environmental M&Ps which apply to the specific Environmental function being performed by NewSouth, its employees, agents and/or subcontractors.

The most current version of reference documentation must be requested from BellSouth.

ENVIRONMENTAL CATEGORIES	ENVIRONMENTAL ISSUES	ADDRESSED BY THE FOLLOWING DOCUMENTATION
Disposal of hazardous material or other regulated material (e.g., batteries, fluorescent	Compliance with all applicable local, state, & federal laws and regulations	 Std T&C 450 Fact Sheet Series 17000
tubes, solvents & cleaning materials)	Pollution liability insurance	• Std T&C 660-3
materials)	EVET approval of contractor	Approved Environmental Vendor List (Contact E/S Management)
Emergency response	Hazmat/waste release/spill firesafety emergency	 Fact Sheet Series 1700 Building Emergency Operations Plan (EOP) (specific to and located on Premises)
Contract labor/outsourcing for services with environmental implications to be performed on BellSouth Premises	Compliance with all applicable local, state, & federal laws and regulations	• Std T&C 450
(e.g., disposition of hazardous material/waste; maintenance of storage tanks)	Performance of services in accordance with BST's environmental M&Ps	 Std T&C 450-B (Contact E/S for copy of appropriate E/S M&Ps.)
	Insurance	• Std T&C 660
Transportation of hazardous material	Compliance with all applicable local, state, & federal laws and regulations	Std T&C 450 Fact Sheet Series 17000
	Pollution liability insurance	• Std T&C 660-3
	EVET approval of contractor	Approved Environmental Vendor List (Contact E/S Management)
Maintenance/operations work	Compliance with all application	• Std T&C 450
which may produce a waste	local, state, & federal laws and regulations	
Other maintenance work	_	
	Protection of BST employees and equipment	 29CFR 1910.147 (OSHA Standard) 29CFR 1910 Subpart O
		(OSHA Standard)
Janitorial services	All waste removal and disposal	P&SM Manager -

	must conform to all applicable federal, state and local regulations	Procurement
	All Hazardous Material and Waste	Fact Sheet Series 17000
	Asbestos notification and protection of employees and equipment	 GU-BTEN-001BT, Chapter 3 BSP 010-170-001BS (Hazcom)
Manhole cleaning	Compliance with all applicable local, state, & federal laws and regulations	 Std T&C 450 Fact Sheet 14050 BSP 620-145-011PR Issue A, August 1996
	Pollution liability insurance	• Std T&C 660-3
	EVET approval of contractor	Approved Environmental Vendor List (Contact E/S Management)
Removing or disturbing building materials that may contain asbestos	Asbestos work practices	GU-BTEN-001BT, Chapter 3

3. **DEFINITIONS**

Generator. Under RCRA, the person whose act produces a Hazardous Waste, as defined in 40 CFR 261, or whose act first causes a Hazardous Waste to become subject to regulation. The Generator is legally responsible for the proper management and disposal of Hazardous Wastes in accordance with regulations.

<u>Hazardous Chemical</u>. As defined in the U.S. Occupational Safety and Health (OSHA) hazard communication standard (29 CFR 1910.1200), any chemical which is a health hazard or physical hazard.

Hazardous Waste. As defined in section 1004 of RCRA.

<u>Imminent Danger</u>. Any conditions or practices at a facility which are such that a danger exists which could reasonably be expected to cause immediate death or serious harm to people or immediate significant damage to the environment or natural resources.

Spill or Release. As defined in Section 101 of CERCLA.

4. ACRONYMS

E/S – Environmental/Safety

EVET - Environmental Vendor Evaluation Team

<u>DEC/LDEC</u> - Department Environmental Coordinator/Local Department Environmental Coordinator

<u>GU-BTEN-001BT</u> - BellSouth Environmental Methods and Procedures

NESC - National Electrical Safety Codes

P&SM - Property & Services Management

Std. T&C - Standard Terms & Conditions

Attachment 5

Access to Numbers and Number Portability

Version1Q00: 3/6/00 11/7/00

TABLE OF CONTENTS

1.	Non-Discriminatory Access To Telephone Numbers	3
2.	Number Portability Permanent Solution	3
3.	Service Provider Number Portability	4
4.	SPNP Implementation	4
5.	Transition To Permanent Number Portability	7
6.	True-Up	7
Ra	ates	Exhibit A

ACCESS TO NUMBERS AND NUMBER PORTABILITY

1. Non-Discriminatory Access to Telephone Numbers

- 1.1 During the term of this Agreement, NewSouth shall contact the North American Numbering Plan Administrator, Neustar, for the assignment of numbering resources. In order to be assigned a Central Office Code, NewSouth will be required to complete the Central Office Code (NXX) Assignment Request and Confirmation Form (Code Request Form) in accordance with Industry Numbering Committee's Central Office Code (NXX) Assignment Guidelines (INC 95-0407-008).
- 1.2 For the purposes of the resale of BellSouth's telecommunications services by NewSouth, BellSouth will provide NewSouth with on line access to telephone numbers for reservation on a first come first served basis. Such reservations of telephone numbers, on a pre-ordering basis shall be for a period of forty-five (45) days. NewSouth acknowledges that there may be instances where there is a shortage of telephone numbers in a particular CLLIC and in such instances BellSouth may request that NewSouth cancel its reservations of numbers. NewSouth shall comply with such request.
- 1.3. Further, upon NewSouth request and for the purposes of the resale of BellSouth's telecommunications services by NewSouth, BellSouth will reserve up to 100 telephone numbers per Common Language Location Identifier Code (CLLIC), for NewSouth's sole use. Such telephone number reservations shall be transmitted to NewSouth via electronic file transfer. Such reservations shall be valid for forty-five (45) days from the reservation date. NewSouth acknowledges that there may be instances where there is a shortage of telephone numbers in a particular CLLIC and in such instances BellSouth shall use its best efforts to reserve for a forty-five (45) day period a sufficient quantity for NewSouth's reasonable need in that particular CLLIC.

2. Number Portability Permanent Solution

- 2.1 The FCC, the Commissions, and industry forums have developed and BellSouth is implementing a permanent approach to providing service provider number portability. Both Parties will implement a permanent approach as developed and approved by the Commission, the FCC and industry forums. Consistent with the requirements to move to Permanent Number Portability (PNP) as set forth in Section 5 of this Attachment, Interim Service Provider Number Portability (SPNP) may be available only until such permanent solution is implemented in an end office.
- 2.2 <u>End User Line Charge</u>. Recovery of charges associated with implementing PNP through a monthly charge assessed to end users has been authorized by the FCC. This

end user line charge will be as filed in FCC No. 1 and will be billed to NewSouth where NewSouth is a subscriber to local switching or where NewSouth is a reseller of BellSouth telecommunications services. This charge will not be discounted.

3. Service Provider Number Portability

- 3.1 <u>Definition</u>. Until the industry-wide permanent solution is implemented in an end office, BellSouth shall provide Service Provider Number Portability ("SPNP"). SPNP is an interim service arrangement whereby an end user who switches subscription of his local exchange service from BellSouth to a CLEC, or vice versa, is permitted to retain the use of his existing assigned telephone number, provided that the end user remains at the same location for his local exchange service or changes locations and service providers but stays within the same serving wire center of his existing number.
- Methods of Providing Number Portability. SPNP is available through either remote call forwarding or direct inward dialing trunks, at the election of NewSouth. Remote call forwarding (SPNP-RCF) is an existing switch-based BellSouth service that redirects calls within the telephone network. Direct inward dialing trunks (SPNP-DID) allow calls to be routed over a dedicated facility to the NewSouth switch that serves the subscriber.
- 3.3 Signaling Requirements. SS7 Signaling is required for the provision of SPNP services. SPNP-DID is available from BellSouth on a per DS0, DS1, or DS3 basis. Where SPNP-DID is technically feasible and is provided on a DS1 or a DS3 basis, the applicable channelization rates are those specified in Section E6 in BellSouth's Intrastate Access Tariffs, incorporated herein by this reference. SPNP is available only for basic local exchange service.

3.4 Rates

Rates for SPNP are set out in Exhibit A to this Attachment. If no rate is identified in the Attachment, the rate for the specific service or function will be as set forth in applicable BellSouth tariff or as negotiated by the Parties upon request by either Party.

4. SPNP Implementation

4.1 SPNP is available only where a CLEC or BellSouth is currently providing, or will begin providing concurrent with provision of SPNP, basic local exchange service to the affected end user. SPNP for a particular telephone number is available only from the central office originally providing local exchange service to the end user. SPNP for a particular assigned telephone number will be disconnected when any end user, Commission, BellSouth, or CLEC initiated activity (e.g., a change in exchange boundaries) would normally result in a telephone number change had the end user retained his initial local exchange service.

Version1Q00: 3/6/00 11/7/00

- 4.2 SPNP-RCF, as contemplated by this Agreement, is a telecommunications service whereby a call dialed to an SPNP-RCF equipped telephone number is automatically forwarded to an assigned seven- or ten- digit telephone number within the local calling area as defined in BellSouth's General Subscriber Services Tariff. The forwarded-to number shall be specified by the CLEC or BellSouth, as appropriate. The forwarding Party will provide identification of the originating telephone number, via SS7 signaling, to the receiving Party. Identification of the originating telephone number to the SPNP-RCF end user cannot be guaranteed, however. SPNP-RCF provides a single call path for the forwarding of no more than one simultaneous call to the receiving Party's specified forwarded-to number.
- SPNP-DID service, as contemplated by this Agreement, provides trunk side access to 4.3 end office switches for direct inward dialing to the other Party's premises equipment from the telecommunications network to lines associated with the other Party's switching equipment and must be provided on all trunks in a group arranged for inward service. A SPNP-DID trunk termination charge, provided with SS7 Signaling only, applies for each trunk voice grade equivalent. In addition, direct facilities are required from the end office where a ported number resides to the end office serving the ported end user customer. The rates for a switched local channel and switched dedicated transport apply as set forth in Exhibit A of this Attachment. Transport mileage will be calculated as the airline distance between the end office where the number is ported and the Point of Interface ("POI") using the V&H coordinate method. SPNP-DID must be established with a minimum configuration of two channels and one unassigned telephone number per switch, per arrangement for control purposes. Transport facilities arranged for SPNP-DID may not be mixed with any other type of trunk group, with no outgoing calls placed over said facilities. SPNP-DID will be provided only where such facilities are available and where the switching equipment of the ordering Party is properly equipped. Where SPNP-DID service is required from more than one wire center or from separate trunk groups within the same wire center, such service provided from each wire center or each trunk group within the same wire center shall be considered a separate service. Only customer-dialed sent-paid calls will be completed to the first number of a SPNP-DID number group; however, there are no restrictions on calls completed to other numbers of a SPNP-DID number group. Sent-paid calls refer to those calls placed by an end user who physically deposits currency in a public telephone. Interface group arrangements provided for terminating the switched transport at the Party's terminal location are as set forth in of BellSouth's Intrastate Access Services Tariff, § E6.1.3.A as amended from time to time.
- 4.3.1 SPNP-DID Service requires ordering consecutive telephone numbers in blocks of twenty. To order non-consecutive telephone numbers or telephone numbers in less than blocks of twenty, the NBR process must be used. SS7 Signaling is required for the provision of either of these services.

- The calling Party shall be responsible for payment of the applicable charges for sent-4.4 paid calls to the SPNP number. For collect, third-party, or other operator-assisted non-sent paid calls to the ported telephone number, BellSouth or the CLEC shall be responsible for the payment of charges under the same terms and conditions for which the end user would have been liable for those charges. Either Party may request that the other block collect and third party non-sent paid calls to the SPNP-assigned telephone number. If a Party does not request blocking, the other Party will provide itemized local usage detail for the billing of non-sent paid calls on the monthly bill of usage charges provided at the individual end user account level. The detail will include itemization of all billable usage. Each Party shall have the option of receiving this usage data on a daily basis via a data file transfer arrangement. This arrangement will utilize the existing industry uniform standard, known as EMI standards, for exchange of billing data. Files of usage data will be created daily for the optional service. Usage originated and recorded in the sending BellSouth RAO will be provided in unrated or rated format, depending on processing system. CLEC usage originated elsewhere and delivered via CMDS to the sending BellSouth RAO shall be provided in rated format.
- Each Party shall be responsible for obtaining authorization from the end user for the 4.5 handling of the disconnection of the end user's service, the provision of new local service and the provision of SPNP services. Each Party shall be responsible for coordinating the provision of service with the other to assure that its switch is capable of accepting SPNP ported traffic. Each Party shall be responsible for providing equipment and facilities that are compatible with the other's service parameters, interfaces, equipment and facilities and shall be required to provide sufficient terminating facilities and services at the terminating end of an SPNP call to adequately handle all traffic to that location and shall be solely responsible to ensure that its facilities, equipment and services do not interfere with or impair any facility, equipment, or service of the other Party or any of its end users. In the event that either Party determines in its reasonable judgment that the other Party will likely impair or is impairing, or interfering with any equipment, facility or service or any of its end users, that Party may either refuse to provide SPNP service or may terminate SPNP service to the other Party after providing appropriate notice.
- Each Party shall be responsible for providing an appropriate intercept announcement service for any telephone numbers subscribed to SPNP services for which it is not presently providing local exchange service or terminating to an end user. Where either Party chooses to disconnect or terminate any SPNP service, that Party shall be responsible for designating the preferred standard type of announcement to be provided.
- Each Party shall be the other Party's single point of contact for all repair calls on behalf of each Party's end user. Each Party reserves the right to contact the other Party's customers if deemed necessary for maintenance purposes.

- Neither Party shall be responsible for adverse effects on any service, facility or equipment from the use of SPNP services. End-to-end transmission characteristics may vary depending on the distance and routing necessary to complete calls over SPNP facilities and the fact that another carrier is involved in the provisioning of service. Therefore, end-to-end transmission characteristics cannot be specified by either Party for such calls. Neither Party shall be responsible to the other if any necessary change in protection criteria or in any of the facilities, operation, or procedures of either renders any facilities provided by the other Party obsolete or renders necessary modification of the other Party's equipment.
- For terminating IXC traffic ported to either Party which requires use of either Party's tandem switching, the tandem provider will bill the IXC tandem switching, the interconnection charge, and a portion of the transport, and the other Party will bill the IXC local switching, the carrier common line and a portion of the transport. If the tandem provider is unable to provide the necessary access records to permit the other Party to bill the IXC directly for terminating access to ported numbers, then the tandem provider will bill the IXC full terminating switched access charges at the tandem provider's rate and will compensate the other Party at the tandem Party's tariff rates via a process to be mutually agreed to by the Parties to estimate the amount of ported switched access revenues due the other Party. If an intraLATA toll call is delivered, the delivering Party will pay terminating access rates to the other Party. This subsection does not apply in cases where SPNP-DID is utilized for number portability.

5. Transition to Permanent Number Portability

- Once a PNP is implemented in an end office both Parties must withdraw their SPNP offerings. The transition from existing SPNP arrangements to PNP shall occur within one hundred twenty (120) days from the date PNP is implemented in the end office. Neither Party shall charge the other Party for conversion from SPNP to PNP. The Parties shall comply with any SPNP/PNP transition processes established by the FCC and State commissions and appropriate industry number portability work groups.
- Notwithstanding the foregoing, the Parties acknowledge that the FCC has determined once LNP has been deployed pursuant to the FCC's orders, rules and regulations, that all local exchange carriers (LECs) have the duty to provide LNP. Therefore, either Party, at any time, may seek appropriate legal or regulatory relief concerning the transition from INP to LNP or other related issues.

6. True-up

This section applies only to Tennessee and other rates that are interim or expressly subject to true-up under this attachment.

6.1 The interim prices for Network Elements and Other Services and Local Interconnection shall be subject to true-up according to the following procedures:

The interim prices shall be trued-up, either up or down, based on final prices determined either by further agreement between the Parties, or by a final order (including any appeals) of the Commission which final order meets the criteria of (3) below. The Parties shall implement the true-up by comparing the actual volumes and demand for each item, together with interim prices for each item, with the final prices determined for each item. Each Party shall keep its own records upon which the true-up can be based, and any final payment from one Party to the other shall be in an amount agreed upon by the Parties based on such records. In the event of any disagreement as between the records or the Parties regarding the amount of such true-up, the Parties agree that the body having jurisdiction over the matter shall be called upon to resolve such differences, or the Parties may mutually agree to submit the matter to the Dispute Resolution process in accordance with the provisions in the General Terms and Conditions and Attachment 1 of this Agreement.

- The Parties may continue to negotiate toward final prices, but in the event that no such agreement is reached within nine (9) months, either Party may petition the Commission to resolve such disputes and to determine final prices for each item. Alternatively, upon mutual agreement, the Parties may submit the matter to the Dispute Resolution Process set forth in the General Terms and Conditions and Attachment 1 of the Agreement incorporated herein by reference, so long as they file the resulting Agreement with the Commission as a "negotiated Agreement" under Section 252(e) of the Act.
- A final order of this Commission that forms the basis of a true-up shall be the final order as to prices based on appropriate cost studies, or potentially may be a final order in any other Commission proceeding which meets the following criteria:
 - (a) BellSouth and CLEC is entitled to be a full Party to the proceeding;
 - (b) It shall apply the provisions of the federal Telecommunications Act of 1996, including but not limited to Section 252(d)(1) (which contains pricing standards) and all then-effective implementing rules and regulations; and,
 - (c) It shall include as an issue the geographic deaveraging of network element prices, which deaveraged prices, if any are required by said final order, shall form the basis of any true-up.

Attachment 5 Exhibit A Rates - Page 1

BELLSOUTHMEWSOUTH RATES SERVICE PROVIDER NUMBER PORTABILITY

The control of the	DESCRIPTION BITTERIN BERTIEF PROVIDER PRIMISER FORTABLITY HEFT IN ENTRE							31416				
NA		80	200		5	ď	\$	5	97 2	2	ş	2
NA	ICF, per number ported (Business Line), 10 pertra				を かいまり	Mary Inc.	354 P W 41 63	Section of the second	STATIONALL S	大学をかっている大大二	14.65 July 10.00 July	
NA NA 81.15 NA NA 81.15 NA NA 82.27 82.24 81.66 82.27 NA 80.29 80.244 80.20 80.20 NA 80.20 80.244 80.20 80.20 NA 80.20	ICF, per number ported (Residence Line). 6 cettre	N. C.	2 6	≨ :	ž	¥	¥	ž	₹	\$2.25	2	42
NA \$229 \$234 \$166 \$217 \$0.066 NA \$0.05 \$0.0641 \$0.01 \$0.006 NA \$0.05 \$0.0641 \$0.01 \$0.006 NA \$0.05 \$0.0644 \$0.01 \$0.006 NA \$0.05 \$0.0644 \$0.01 \$0.006 \$0.0644 \$0.01 \$0.006 \$0.0644 \$0.01 \$0.006 \$0.0644 \$0.00 \$0.006 \$0.0644 \$0.00 \$0.006	CF, per number ported (Bushness Line)	N	2 3	≨ .	≨	¥	ž	ž	ž	\$1.15	2	2
NA \$0.05 \$0.0544 \$0.777 \$0.7046 NA NA \$2.29 \$2.0544 \$0.50 NA NA \$2.29 \$2.0544 \$0.50 NA \$2.07 \$0.0544 \$0.50 NA \$2.07 \$2.076 \$0.0544 \$0.50 NA \$2.07 \$2.076 \$2.070 \$	INRC - Electronic		2	2	¥	25.03	Ź		22.34	2 2	25 13	9
NA \$205 \$20544 \$5050 NA NA \$204 \$23.4 \$1.66 \$2.17 NA \$204 \$20.4 \$50.50 NA NA \$202 \$2.84 \$2.73 \$1.37 NA \$202 \$2.84 \$2.73 \$1.37 NA \$202 \$2.84 \$2.73 \$1.37 NA \$202 \$2.84 NA NA NA \$201 \$2.84 \$2.73 \$1.37 NA \$2.01 \$2.84 NA NA NA \$2.01 \$2.84 NA NA NA \$2.01 \$2.85 \$2.85 NA NA \$2.01 \$2.85 \$2.85 NA NA \$2.01 \$2.84 NA NA NA \$2.01 \$2.94 NA NA NA \$2.01 \$2.94 NA NA NA \$2.01 \$2.94 NA NA NA \$2.01 \$2.94 NA NA NA \$2.01 \$2.94 NA NA NA \$2.01 \$2.94 NA NA NA \$2.01 \$2.94 NA NA NA \$2.01 \$2.94 NA NA NA \$2.01 \$2.94 NA NA NA \$2.01 \$2.94 NA NA NA \$2.01 \$2.94 NA NA NA \$2.01 \$2.94 NA NA NA \$2.01 \$2.94 NA NA NA \$2.01 \$2.94 NA NA NA \$2.01 \$2.94 NA NA NA \$2.01 \$2.94 NA NA NA \$2.01 \$2.94 NA NA NA \$2.01 \$2.94 NA NA	NPC - Disconnect Chame	N.	2	\$0.65	ž	\$0.51	¥		\$0.6441	20.71	\$0.704A	
NA \$2.29 \$2.34 \$1.06 \$2.17 \$2.00	CF, Der Rumber Bortert (Residence 1 has)		ď	20.03	ž	¥	ž		\$0.0644	52.62	N N	5
NA \$0.49 \$0.644 \$0.77 \$0.7046 NA NA \$0.054 \$0.50 NA NA St. 0.055			Ē	\$2.13	¥	\$2.03	ž	l	22.34	81.8	20.53	٤
NA \$0.05 \$0.0644 \$0.50 NA NA \$0.383 \$0.382 \$0.3854 NA NA \$2.02 \$2.33 \$1.37 NA \$2.02 \$2.84 \$2.73 \$1.37 NA \$2.02 \$2.84 NA NA NA \$2.01 \$2.85 \$2.85 NA NA \$2.01 \$2.85 \$2.85 NA NA \$2.01 \$2.85 \$2.85 NA \$2.02 \$2.85 \$2.85 NA \$2.02 \$2.85 \$2.85 NA \$2.02 \$2.85 \$2.25 NA \$2.02 \$2.25 NA \$2.02 \$2.25 NA \$2.02 \$2.25 NA \$2.02 \$2.25 NA \$2.02 \$2.25 NA \$2.02 \$2.25 NA \$2.02 \$2.25 NA \$2.02 \$2.25 NA \$2.02 \$2.25 NA \$2.02 \$2.25 NA \$2.02 \$2.25 NA \$2.02 \$2.25 NA \$2.02 \$2.25 NA \$2.02 \$2.25 NA \$2.02 \$2.25 NA \$2.02 \$2.25 NA \$2.02 \$2.25 NA \$2.02 \$2.00 NA \$2.02 \$2.00 NA \$2.02 \$2.00 NA \$2.02 \$2.00 NA \$2.00 \$2.00 NA \$2.00 NA \$2.00 \$2.00 NA \$2.00 NA \$2.00 NA \$2.00 NA \$2.00 NA \$2.00 NA \$2.00 NA \$2.00 NA \$2.00 \$2.00 NA \$2.00 NA \$2.00 NA \$2.00 NA \$2.00 NA \$2.00 \$2.00 NA \$2.00 NA \$2.00 \$2.00 NA \$2.00 \$	MDC - Discovery Channel		£	\$0.65	NA	\$0.51	ž		\$0.6441	2, 55	60 7048	C7
NA \$0.30 \$0.303 \$0.32 \$0.3844 NA NA S.0.20 \$2.84 \$2.73 \$1.37 NA S.0.303 \$1.37 NA S.0.303 \$1.37 S.0.304 S.0.3	C. add cancely for structure as bounded	Ž	æ	\$0.07	٧¥	≨	ž		779005	S	AN 42	2
NA \$2.00 \$2.84 \$2.73 \$1.37 NA S2.01 \$2.84 NA NA NA NA NA NA NA N	The state of the s	2	*	20.33	¥	\$0.2836	ž		\$0.38.38	25	2000	٤
NA \$2.02 \$2.84 \$2.73 \$1.37	2F, per service order, per location	(++) Bus	* INPBO									5
NA \$2.02 \$2.84 \$2.73 \$1.37 NA \$2.02 \$2.84 \$2.73 \$1.37 NA \$2.01 \$2.84 NA NA NA NA \$2.01 \$2.84 NA NA NA \$2.01 \$2.85 \$2.85 \$2.85 NA \$2.85 \$2.85 NA \$2.85 \$2.85 NA \$2.85 \$2.85 NA \$2.85 \$2.85 NA NA \$2.85 NA NA \$2.85 NA	MPC-1st	71										
NA \$2.00 \$2.84 \$2.73 \$1.37 NA NA NA NA NA NA NA N	MRC - Add1			*	≨ :	\$2.10	¥	\$2.00	\$2.84	\$2.73	\$1.37	\$25.00
NA \$2.01 \$2.64 NA NA NA NA NA NA NA N	NRC - Discormed - 1st		# :	1	ž	\$2.10	¥	\$2.02	\$2.84	\$2.73	\$1.37	\$25.00
NA \$2.01 \$2.84 NA NA NA NA NA S18.14 \$25.52 \$46.80 NA S18.14 \$25.52 \$46.80 NA S18.14 \$25.52 \$46.80 NA S18.14 \$26.52 \$46.80 NA S18.17 NA \$20.90 \$11.17 NA NA NA NA NA NA NA N	NRC - Disconnect - Add1			1	ž	¥	ž	\$2.01	\$2.84	ž	₹	Ž
NA \$18.14 \$25.52 \$45.80 NA NA \$11.41 \$16.06 NA \$44.70 NA \$44.7	MRC - Incremental Charge - Manual Service Order - 1st	100	- 1	*	Ì	¥	-	\$2.01	\$2.84	¥	ž	¥Z
NA \$18.14 \$25.52 \$45.80 NA \$11.41 \$16.06 NA \$44.70 NA \$44.70 NA \$44.70 NA \$44.70 NA \$44.70 NA \$44.70 NA \$44.70 NA \$40.80 \$11.77 NA \$44.70 NA \$40.80 \$11.77 NA \$44.70 NA \$2.25 \$2.25 \$2.25 \$2.25 NA \$44.70 NA \$2.02 \$2.25 \$2.25 NA NA NA \$2.02 \$2.25 \$2.25 NA NA \$2.02 \$2.25 \$2.25 NA NA \$2.02 \$2.25 \$2.25 NA NA \$2.01 \$2.25 \$2.25 \$2.25 NA NA \$2.01 \$2.25 \$2.25 \$2.25 NA NA \$2.01 \$2.25 \$2.25 \$2.25 NA NA \$2.01 \$2.25 \$2.25 NA NA \$2.01 \$2.25 \$2.25 NA NA \$2.01 \$2.25 \$2.25 NA NA \$2.01 \$2.25 \$2.25 NA NA \$2.01 \$2.25 \$2.25 NA NA \$2.01 \$2.25 \$2.25 NA NA \$2.01 \$2.25 \$2.25 NA NA \$2.02 \$2.25 \$2.25 NA NA \$2.02 \$2.25 \$2.25 NA NA \$2.02 \$2.25 \$2.25 NA NA \$2.02 \$2.25 \$2.25 NA NA \$2.02 \$2.25 \$2.25 NA NA \$2.25 \$2.25 \$2.25 NA NA \$2.25 \$2.25 \$2.25 NA NA \$2.25 \$2.25 NA NA \$2.25 \$2.25 NA NA \$2.25 \$2.25 NA NA \$2.25 \$2.25 NA NA \$2.25 \$2.25 NA NA \$2.25 \$2.25 NA NA \$2.25 \$2.25 NA NA \$2.25 \$2.25 NA NA \$2.25 \$2.25 NA NA \$2.25 \$2.25 NA NA \$2.25 \$2.25 NA NA \$2.25 \$2.25 NA NA \$2.25 \$2.25 NA NA \$2.25 \$2.25 NA NA \$2.25 \$2.25 NA NA \$2.25 \$2.25 NA NA \$2.25 \$2.25 NA NA NA \$2.25 \$2.25 NA NA NA \$2.25 NA NA \$2.25 NA NA \$2.25 NA NA \$2.25 NA NA \$2.25 NA NA NA NA \$2.25	NRC - Incremental Charge - Manual Service Order - Addit	56	- 1	\$27.37		ž		\$18.14	\$25.52	\$45.80	ž	2
NA \$11.41 \$16.06 NA \$44.70 NA \$0.90 \$11.71 \$22.26 \$2.25 NA \$0.90 \$11.77 \$2.25 NA \$0.90 \$11.77 \$2.25 NA \$0.90 \$11.77 \$2.25 NA \$0.90 \$11.77 \$2.25 NA \$0.90 \$11.77 \$2.25 NA \$2.00 \$11.77 \$2.25 NA \$2.00 \$11.77 \$2.25 NA \$2.00 \$11.77 \$2.25 NA \$2.00 \$2.84 \$2.73 \$1.37 NA \$2.01 \$2.84 \$2.73 \$1.37 NA \$2.01 \$2.84 \$2.73 \$1.37 NA \$2.01 \$2.84 \$2.84 NA \$2.01 \$2.85 \$2.84 NA \$2.01 \$2.85 \$2.84 NA \$11.41 \$18.00 \$NA \$NA \$NA \$NA \$NA \$NA \$NA \$NA \$1.45 NA \$12.46 \$11.50 \$11.43 \$13.16 NA \$12.46 \$11.50 \$11.43 \$13.16 NA \$12.46 \$13.76 \$217.66 \$218.03 NA \$12.46 \$13.76 \$11.43 \$13.16 NA \$12.46 \$13.76 \$11.43 \$13.16 NA \$12.46 \$13.76 \$11.43 \$13.16 NA \$12.46 \$13.76 \$11.43 \$13.16 NA \$12.46 \$13.76 \$11.43 \$13.16 NA \$12.46 \$13.76 \$11.43 \$13.16 NA \$12.46 \$13.76 \$11.43 \$13.16 NA \$18.75 \$24.77 \$NA \$NA \$NA \$NA \$NA \$NA \$NA \$NA \$NA \$NA	NRC - Incremental Charge - Manual Service Order - Disconnect - 144	000		\$27.37	1	≨		\$18.14	\$25.52	\$45.80	ž	Ž
NA \$10.00 NA NA \$13.16 NA \$44.70 NA \$2.25 NA \$2.25 NA \$2.25 NA NA \$2.00 \$1.17 NA NA NA \$2.00 \$1.17 NA NA NA NA NA \$2.00 \$1.17 NA NA NA NA NA \$2.00 \$1.17 NA \$44.70 NA \$2.01 \$2.00 NA \$44.70 NA \$1.01 \$2.00 NA \$1.01 \$2.00 NA \$1.01 \$2.00 NA \$1.01 \$2.00 NA \$1.01 \$2.00 NA \$1.01 \$2.00 NA \$1.01 \$2.00 NA NA \$1.00 \$1.01 \$2.00 NA NA \$1.00 \$1.00 NA NA \$1.00 NA NA \$1.00 NA \$1.00 NA NA \$1.00 NA \$1.00 NA NA \$1.00 NA \$1.00 NA NA \$1.00 NA NA \$1.00 NA \$1.00 NA NA \$1.00 NA NA \$1.00 NA \$1.00 NA NA \$1.00 N	MRC - Incremental Charge - Marrial Sarvice Order - Discourses - Add	OS .	Z X	517.77	1	¥		11.112	\$16.06	ž	244 70	1
N	CHARLES AND THE CHARLES AND AND AND AND AND AND AND AND AND AND		WAN WAN	\$17.77	ş.	≨	ı		L	ž	\$44.70	Ž
NA \$0.89 \$1.17 \$1.25 \$2.25 NA NA NA \$0.80 \$1.17 \$1.17 \$1.25 \$2.25 NA \$1.17 \$2.25 \$2.25 NA \$2.25 \$2.25 NA \$2.25 \$2.25 NA \$2.25 \$2.25 NA \$2.25 \$2.25 NA \$2.25 \$2.25 NA \$2.25 \$2.25 NA \$2.25 \$2.25 NA \$2.25 \$2.25 NA \$2.25 \$2.25 NA \$2.27 \$2.25 NA \$2.27 \$2.24 NA \$2.27 \$2.24 NA \$2.27 \$2.24 NA \$2.27 \$2.24 NA \$2.27 \$2.24 NA \$2.27 \$2.24 NA \$2.27 \$2.24 NA \$2.27 \$2.24 NA NA \$2.25) per number ported, Residence - NRC			7 72 67		Ser dell's				West Triples	1. A. T. A. T.	
NA \$0.90 \$1.17 \$2.25 NA NA NA S0.90 \$1.17 \$2.25 NA NA \$0.90 \$1.17 NA NA NA S2.02 \$2.94 \$2.73 \$1.37 NA \$2.01 \$2.94 \$2.73 \$1.37 NA \$2.01 \$2.94 \$2.73 \$1.37 NA \$2.01 \$2.94 NA \$2.95 \$2.94 NA \$2.95 \$2.94 NA \$2.95 \$2.94 NA \$2.95 \$2.94 NA \$2.95 \$2.94 NA NA \$1.44 \$25.52 \$45.90 NA NA NA \$1.44 \$1.50 NA NA NA NA NA NA NA NA NA NA \$2.95 \$2.9) per number ported, Residence - MRC - Disconnect	TATE OF THE PARTY	4	21.10	٤	603	ž	\$0.69	\$1.17	\$2.25	\$2.25	ž
NA \$0.09 \$1.17 \$2.25 \$2.25 NA \$2.00 \$2.04 \$1.17 NA NA NA NA NA \$2.00 \$2.04 \$2.73 \$1.37 NA \$2.00 \$2.04 \$2.04 \$2.73 \$1.37 NA \$2.01 \$2.04 \$2.04 NA \$44.70 NA \$10.41 \$2.02 \$45.00 NA NA \$11.41 \$10.44 \$1.41 \$1.41) per number ported, Bushness - NRC	GNL			٤	ž	≨.	8	\$1.17	ž	NA	ž
NA \$2.00 \$1.17 NA NA NA NA NA S. 1.07 NA \$2.00 \$2.04 \$2.73 \$1.37 NA \$2.00 \$2.04 \$2.73 \$1.37 NA \$2.00 \$2.04 \$2.73 \$1.37 NA \$2.01 \$2.04 NA \$44.70 NA \$1.04 \$2.05 \$45.00 NA NA \$11.41 \$16.00 NA NA \$11.41 \$16.00 NA NA \$11.41 \$16.00 NA NA \$11.41 \$16.00 NA NA \$11.40 \$13.70 \$13.10 NA \$120.00 \$13.70 \$13.10 NA \$120.00 \$13.70 \$13.10 NA \$120.00 \$13.70 \$13.10 NA \$10.75 \$24.77 NA NA \$10.75 \$24.77 NA NA NA \$10.75 \$24.77 NA NA NA \$10.75 \$24.77 NA NA NA \$10.75 \$24.77 NA NA NA NA \$10.75 \$24.77 NA NA NA NA \$10.75 \$24.77 NA NA	Per number ported, Buelness - NRC - Disconnect	dNL	4		5	3	ž	200	\$1.17	\$2.25	\$2.25	ž
NA \$2.02 \$2.84 \$2.73 \$1.37 NA \$2.02 \$2.84 \$2.73 \$1.37 NA \$2.01 \$2.84 \$2.73 \$1.37 NA \$18.14 \$25.52 \$45.80 NA NA \$18.14 \$18.00 NA NA NA \$11.41 \$18.00 NA NA \$12.46 \$13.76 \$11.43 \$13.16 NA \$12.46 \$13.76 \$11.43 \$13.16 NA \$12.46 \$13.76 \$11.43 \$13.16 NA \$12.46 \$13.76 \$11.43 \$13.16 NA \$12.46 \$13.76 \$11.43 \$13.16 NA \$12.46 \$13.76 \$11.43 \$13.16 NA \$12.46 \$13.76 \$11.43 \$13.16 NA \$12.46 \$13.76 \$11.43 \$13.16 NA \$12.46 \$13.76 \$11.43 \$13.16 NA \$18.75 \$24.71 NA NA) per service order, per location				Š	ž	ž	808	\$1.17	≨	ž	ž
NA \$2.02 \$2.84 \$2.73 \$1.37 NA \$2.02 \$2.84 \$2.73 \$1.37 NA \$2.02 \$2.84 NA \$44.70 NA \$1.37 NA \$1.37 NA \$1.37 NA \$1.37 NA \$1.37 NA \$1.34 \$2.32 \$45.80 NA NA \$1.34 \$1.3	ABC - 18	dy'T		77.5	1	0,00						
NA \$2.02 \$2.84 NA \$4.70 NA \$2.01 \$2.84 NA \$44.70 NA \$2.01 \$2.84 NA \$44.70 NA \$4.80 NA \$44.70 NA \$41.41 \$18.04 NA \$41.41 \$18.04 NA \$11.41 \$18.00 NA NA \$11.41 \$18.00 NA NA \$11.41 \$18.00 NA NA \$11.42 \$13.76 \$13.76 \$13.76 NA \$13.76 \$13.76 NA \$13.76 NA \$13.76 \$13.76 NA \$13.76 \$13.76 NA \$13.76 NA \$13.76 \$13.76 NA \$13.76 NA \$13.76 \$13.76 NA NA \$13.75 \$13.76 \$13.76 NA NA \$18.75 \$13.76 \$13.76 NA NA \$18.75 \$13.76 \$13.76 NA NA \$18.75 \$13.76 \$13.76 NA NA \$18.75 \$13.76 NA NA \$18.75 \$13.76 NA NA \$18.75 \$13.76 NA NA \$18.75 \$13.76 NA NA \$18.75 \$13.76 NA NA \$18.75 \$13.76 NA NA \$18.75 \$13.76 NA NA \$18.75 \$13.76 NA NA \$18.75 \$13.76 NA NA \$18.75 \$13.76 NA NA \$18.75 \$13.76 NA NA \$18.75 \$13.76 NA NA \$18.75 \$13.76 NA NA \$18.75 \$13.76 NA NA \$18.75 \$13.76 NA NA \$18.75 \$13.76 NA NA NA NA NA NA NA NA NA NA NA NA NA	MRC - Add1	QN.L			1	20.20	ž	23.02	22.84	\$2.73	\$1.37	≨
NA \$2.01 \$2.84 NA \$44.70 NA \$16.14 \$25.52 \$45.80 NA NA \$16.14 \$25.52 \$45.80 NA NA \$11.41 \$16.00 NA NA NA \$12.46 \$13.76 \$11.43 \$13.16 NA \$12.46 \$13.76 \$13.76 NA \$12.46 \$13.76 \$13.76 NA \$12.46 \$13.76 NA \$12.46 \$13.76 NA \$12.46 \$13.76 NA \$12.46 \$13.76 NA \$12.46 \$13.76 NA \$12.46 \$13.76 NA \$12.46 \$13.76 NA \$12.46 \$13.76 NA \$13.76 NA \$13.76 NA \$18.75	NPC - Disconnect - 1st	dAT.	1		5	24.10	≨:	23.03	25.84	\$2.73	\$1.37	ž
NA \$201 \$284 NA \$44.70 NA \$18.14 \$25.52 \$45.80 NA NA \$18.14 \$18.00 NA NA NA \$18.44 \$18.00 NA NA NA \$12.46 \$13.76 \$11.41 \$13.16 NA \$17.85 \$18.00 NA NA NA NA S7.85 \$49.86 NA NA NA NA NA NA NA NA NA NA NA NA NA	MRC - Disconnect - Addi	DAT	l		5	٤:	≨ :	\$2.01	25.84	ž	\$44.70	₹
NA \$18.14 \$25.52 \$45.80 NA NA \$18.14 \$25.52 \$45.80 NA NA ST.24 \$18.06 NA NA ST.25 \$49.86 NA NA \$12.46 \$171.69 \$211.43 \$13.16 NA \$12.46 \$171.69 \$211.43 \$13.16 NA \$12.46 \$13.78 \$171.69 \$218.03 NA \$10.75 \$24.71 NA NA \$10.75 \$24.71 NA NA NA NA NA NA NA NA NA NA NA NA NA	NRC - Incremental Charge - Manual Service Order - 1st	NOS	-	107.03	5 5	5	ž	22.0	25.52	≨	\$44.70	ž
NA \$18.44 \$15.52 NA NA NA NA STATE NA NA STATE STATE NA NA STATE STATE STATE STATE NA NA STATE S	NRC - Incremental Charge - Manual Service Order - Add1	NOS	-	27.17	5 2	4,014	٤	\$10.14	\$25.52	245.80	Ν	≱
NA \$11.41 \$18.00 NA NA NA NA NA SI A SI A SI A SI A SI A	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	MOS	-	17.77	1 2	\$ 2	\	\$10.14	\$25.52	25.80	3	¥
NA \$12.46 \$13.16 NA \$12.46 \$13.16 NA \$12.80 \$171.68 \$217.88 \$218.03 NA \$17.85 \$13.16 NA \$37.85 \$13.76 NA \$37.85 \$24.71 NA \$18.75 \$24.71 NA \$18.75 \$24.71 NA \$18.75 \$24.71	INHC - Incremental Charge - Manual Service Order - Disconnect - Add1	MOS	-	11.77	Ž		5		878	ž	≨	¥
NA \$12.00 \$13.76 \$13.16 NA \$27.05 \$13.16 NA \$27.05 \$13.06 NA \$27.05 \$13.06 NA \$27.05 \$24.71 NA NA), per trunk termination, initial	2		7811	1 N	200	٤	17.11	\$16.06	≨	≨	ž
NA \$17.68 \$217.68 \$218.03), per trunk termination, initial - NRC	2	-	17.17	5 5	\$10.73	٤:	\$12.46	\$13.78	\$11.43	\$13.16	¥
NA \$1.05 \$13.16 NA \$7.05 NA \$7	2), per trunk termination, Initial - Discornect	2		50.63	5 2	193.4	٤:	\$129.00	\$177.68	\$217.68	\$218.03	ž
NA \$12.46 \$13.78 \$11.43 \$13.16 NA \$10.75 \$24.71 NA NA	3, per thunk termination, Subsequent	2			5 2	42	٤	537.05	249.86	≨	¥	≨
NA \$16.75 \$24.71 NA NA NA), per Inusk termination, Subsequent - NRC			36.13	5 5	\$10.73	≨ :	\$12.46	\$13.78	\$11.43	\$13.16	2
NA 1 2/3/2 2/3/2 NA NA NA NA NA NA NA N), per trunk termination, Subsequent - Discornact	2	ŀ	25.00	\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \	2 2	ž	\$37.85	\$20.69	\$73.56	\$73.63	¥
Table is identified in the contract, the rate for the apeculic service or function will be as set forth in applicable BaltSouth built or as negotiated by the Parties upon neguest by attent Pany. 1 Until the FCC issues its order implementing a cost recovery mechanism for permanent number portability, the Company will track its costs of providing historia With sufficient detail to verify the costs. This will inclinate the Florida PSCs consideration of the recovery of these costs in Docket PROFITED TO THE FORTAL TO THE PROFITED TO THE PR	TES:					5	S.	\$16.75	17.7	ž	ž	≨
1 Until the FCC tassus its order implementing a cost recovery mechanism for permanent number portability, the Company will track its costs of providing his firm SPNP with sufficient detail to verify the costs. This will incitate the Florida PSCs consideration of the recovery of these costs in Decisal PSCs or additional provideration of the recovery of these costs in Decisal PSCs consideration of the recovery of these costs in Decisal PSCs consideration of the recovery of these costs in Decisal PSCs consideration of the recovery of these costs in Decisal PSCs consideration of the recovery of these costs in Decisal PSCs consideration of the recovery of these costs in Decisal PSCs consideration of the recovery of these costs in Decisal PSCs consideration of the recovery of these costs in Decisal PSCs consideration of the recovery of these costs in Decisal PSCs consideration of the recovery of these costs in Decisal PSCs consideration of the recovery of these costs in Decisal PSCs consideration of the recovery of these costs in Decisal PSCs consideration of the recovery of these costs in Decisal PSCs consideration of the recovery of these costs in Decisal PSCs consideration of the recovery of these costs in Decisal PSCs consideration of the recovery of these costs in Decisal PSCs consideration of the recovery of these costs in Decisal PSCs consideration of the recovery of these costs in Decisal PSCs consideration of the recovery of th	o rate is identified in the contract, the rate for the apecific service or function w	II be as set forth in amicrable RetSouth	access as see	o de la companya de l		1						
portability, the Company will track its costs of providing international recovery of these costs in Deciset in SPAP with sufficient detail in verify the costs. This will inclinate the Florida PSCs consideration of the necessary of these costs in Deciset in Space 273-TP. (FL.)	1 [httl ft-a FCC lees see its cortex implications as seen as FCC lees.	4			diameter and	and and an inch	×.					
the costs. This will british the Florida PSCs consideration of the recovery of these costs in Docket 8,900-71-17, FL. P. P. P. P. P. P. P. P. P. P. P. P. P.	portability, the Company will track its costs of providing interim SPNP with	or permanent number sufficient detail to verify										
A PARCY ALL ALL ALL ALL ALL ALL ALL ALL ALL AL	the costs. This will tacificate the Florida PSCs consideration of the recova- sentature (FI)	ry of these costs in Docket										
	Station of Class and the season of the seaso											

Attachment 6 Ordering and Provisioning

TABLE OF CONTENTS

1.	Quality of Ordering And Provisioning	3
2.	Access To Operational Support Systems	4
3.	Miscellaneous Ordering And Provisioning Guidelines	. 6

ORDERING AND PROVISIONING

1. Quality of Ordering and Provisioning

BellSouth shall provide ordering and provisioning services to NewSouth that are equal to the ordering and provisioning services BellSouth provides to itself or any other CLEC, where technically feasible. Reasonable and nondiscriminatory guidelines for ordering and provisioning are set forth in the various pre-ordering, ordering and provisioning guides, as appropriate, and as they are amended from time to time during this Agreement. The guides may be referenced at the following site:

http://www.interconnection.bellsouth.com/guides/guides_p.html.

BellSouth shall provide advance notification of such guides via carrier notification letters posted to BellSouth's web site.

- 1.2 BellSouth shall provide all ordering and provisioning services to NewSouth during the same business hours of operation that BellSouth provisions service to its affiliates or end users. Ordering and provisioning support required by NewSouth outside of these hours will be considered outside of normal business hours and will be subject to overtime billing.
- 1.2.1 For purposes of this Agreement, BellSouth's regular working hours are defined as follows:

Monday – Friday – 8:00 a.m. – 5:00 p.m. (Excluding Holidays)

(Resale/UNE non-coordinated, coordinated orders and order coordination-time specific)

Saturday - 8:00 a.m. -5:00 p.m. (Excluding Holidays)

(Resale/UNE non-coordinated orders)

It is understood and agreed that BellSouth technicians involved in provisioning service to NewSouth may work shifts outside of BellSouth's regular working hours as defined in Section 1.2 above (e.g., the employee's shift ends at 7:00 p.m. during daylight savings time). To the extent that NewSouth requests that work necessarily required in the provisioning of service to be performed outside BellSouth's regular working hours and that work is performed by a BellSouth technician during his or her scheduled shift such that BellSouth does not incur any additional costs in performing the work on behalf of NewSouth, BellSouth will not assess NewSouth additional charges beyond the rates and charges specified in this Agreement.

2. Access to Operations Support Systems

- 2.1 BellSouth shall provide NewSouth access to operations support systems ("OSS") functions for pre-ordering, ordering and provisioning, maintenance and repair and billing. Access to OSS is available through a variety of means, including electronic interfaces. BellSouth also provides manual options. The OSS functions available to CLECs through electronic interfaces are:
- Pre-Ordering. BellSouth provides electronic access to the following pre-ordering 2.2 functions: service address validation, telephone number selection, service and feature availability, due date information, and upon Commission approval of confidentiality protections, to customer record information. Access is provided through the Local Exchange Navigation System (LENS) interface, or the Telecommunications Access Gateway (TAG) interface. Customer Record Information includes but is not limited to, customer specific information in CRIS and RSAG. In addition, NewSouth shall provide to BellSouth access to customer record information including electronic access where available. Otherwise, NewSouth shall provide paper copies of customer record information within a reasonable period of time upon request by BellSouth. NewSouth agrees not to view, copy, or otherwise obtain access to the customer record information of any customer without that customer's permission and further agrees that it will obtain access to customer record information only in strict compliance with applicable laws, rules, or regulations of the State in which the service is provided.
- 2.2.1 <u>Interfaces.</u> BellSouth shall make available the following interfaces to NewSouth for access to pre-order functions: LENS; and TAG. Each such interface shall be available on a non-discriminatory basis in connection with pre-ordering for Resale services and UNEs that are available electronically.
- 2.2.2 The Parties <u>acknowledge</u> that ordering requirements necessitate the use of current, real time pre-order information to accurately build service orders. Each pre-order interface shall be available except for downtime attributable to maintenance and upload, twenty-four (24) hours a day, seven (7) days a week.
- 2.2.3 NewSouth shall be <u>permitted</u> to reserve a number, including, without limitation, a vanity number, for up to thirty (30) days for End Users.
- 2.2.4 All CSR data exchanged must be in English text, and not only USOC or FID format, provided that such <u>information</u> is maintained in textual format by BellSouth. All other data shall be in a mutually agreed upon nomenclature.
- Upon request, BellSouth shall <u>provide</u> NewSouth with pre-order information in batch transmission to the extent available or provided to any other Telecommunications Carrier on the same terms and conditions and at the same rates.
- 2.2.6 Pre-ordering functions shall be provided at parity as measured by the Performance Measurement metrics included in Attachment 9 hereto.

- 2.3 Service Ordering and Provisioning. BellSouth provides electronic options for the exchange of ordering and provisioning information. Access is provided through an Electronic Data Interchange (EDI) interface, or the TAG ordering interface for non-complex and certain complex resale requests and certain network elements. NewSouth may integrate the EDI interface with the TAG pre-ordering interface or the TAG ordering interface. BellSouth provides integrated pre-ordering, ordering and provisioning capability through the LENS interface for non-complex and certain complex resale service requests and Unbundled Network Elements.
- 2.3.1 For generation of Resale service orders, ordering flows shall be available via such electronic interfaces for each of the following ordering functions: Conversion ("as is" or "with changes"); Change (features, listings, long distance); New Connect; Disconnect; From and To (change of premises with same service).
- 2.3.2 BellSouth shall provide to NewSouth electronic and manual interfaces for transmitting orders and receiving Firm Order Confirmation ("FOC"), completion notices, Due-Date Jeopardies, Design Layout Records, and, as available, other provisioning data and information. BellSouth shall provide NewSouth with a FOC for each Resale and UNE order. The FOC includes: purchase order number, telephone number, Local Service Request number, due date, and Service Order number.
- 2.3.3 BellSouth shall provision Resale Services and UNEs as prescribed in NewSouth service order requests. Access to status on electronically-submitted Resale services and UNEs shall be provided via the electronic interfaces. Access to status on manually-submitted service order requests shall be provided on BellSouth's Internet website.
- 2.3.4 BellSouth shall provide notice of a lack of facilities availability at parity to that BellSouth provides to itself, its Affiliates, or any other Telecommunications Carrier.
- NewSouth to report and monitor service troubles and obtain repair services.

 BellSouth shall offer NewSouth service trouble reporting in a non-discriminatory manner that provides NewSouth the equivalent ability to report and monitor service troubles that BellSouth provides to itself. BellSouth also provides NewSouth an estimated time to repair, an appointment time or a commitment time, as appropriate, on trouble reports. BellSouth provides several options for electronic trouble reporting. For exchange services, BellSouth offers NewSouth non-discriminatory access to the Trouble Analysis Facilitation Interface (TAFI). In addition, BellSouth offers an industry standard, machine-to-machine Electronic Communications Trouble Administration (ECTA) Gateway interface. For designed services, BellSouth provides non-discriminatory trouble reporting through the ECTA Gateway. BellSouth also offers ECTA functionality through the human-to-machine EC-CPM/TA

interface. If NewSouth requests BellSouth to repair a trouble after normal working hours, NewSouth will be billed the appropriate overtime charges associated with this request pursuant to BellSouth's tariffs, in the event BellSouth is required to pay overtime charges to the technicians repairing or isolating the service.

- 2.5 <u>Change Management</u>. BellSouth provides a collaborative process for change management of the electronic interfaces through the Electronic Interface Change Control Process ("EICCP). Reasonable and nondiscriminatory guidelines for this process are set forth in the EICCP document, and as it is amended from time to time during this agreement.
- Migration of NewSouth to New Software Releases for National Standard Machine-to-Machine Electronic Interfaces. Pursuant to the change management process, BellSouth will issue new software releases for new industry standards for its industry standard, machine-to-machine electronic interfaces. When a new release of new industry standards is implemented, BellSouth will continue to support both the new release (N) and the prior release (N-1). When BellSouth makes the next release (N+1), BellSouth will eliminate support for the (N-1) release and support the two newest releases (N and N+1). Thus, BellSouth will always support the two most current releases. BellSouth will issue documents to NewSouth with sufficient notice to allow NewSouth to make the necessary changes to their systems and operations to migrate to the newest release in a timely fashion.
- 2.7 Rates. To the extent approved by the Commission or otherwise agreed to by the Parties, all costs incurred by BellSouth to develop and implement operational interfaces to the OSS shall be recovered from the carriers that use the services. Charge for use of OSS shall be as set forth in the General Terms and Conditions of of this Agreement.
- 2.8 The electronic OSS Charges rather than the manual ordering charges shall apply to Local Service Request submitted by NewSouth when BellSouth's existing electronic interfaces normally utilized by NewSouth are unavailable for reasons other than scheduled maintenance or other scheduled activities for which advance notification is required and provided by BellSouth.

3. Miscellaneous Ordering and Provisioning Guidelines

- Pending Orders. To ensure the most efficient use of facilities and resources, orders placed in the hold or pending status by NewSouth will be held for a maximum of thirty (30) days from the date the order is placed on hold. After such time, if NewSouth wishes to reinstate an order, NewSouth may be required to submit a new service order.
- 3.2 <u>Single Point of Contact</u>. NewSouth will be the single point of contact with BellSouth for ordering activity for network elements and other services used by NewSouth to

provide services to its end users, except that BellSouth may accept an order directly from another CLEC, or BellSouth, acting with authorization of the affected end user. NewSouth and BellSouth shall each execute a blanket letter of authorization with respect to customer orders. The Parties shall each be entitled to adopt their own internal processes for verification of customer authorization for orders, provided, however, that such processes shall comply with applicable state and federal law including, until superseded, the FCC guidelines and orders applicable to Presubscribed Interexchange Carrier (PIC) changes including Un-PIC. Pursuant to such an order, BellSouth may disconnect any network element associated with the service to be disconnected and being used by NewSouth to provide service to that end user and reuse such network elements or facilities to enable such other LEC to provide service to the end user. BellSouth will notify NewSouth that such an order has been processed, but will not be required to notify NewSouth in advance of such processing.

- Use of Facilities. When a customer of NewSouth elects to discontinue service and transfer service to another local exchange carrier, including BellSouth, BellSouth shall have the right to reuse the facilities provided to NewSouth by BellSouth for retail or resale service, loop and/or port for that customer. In addition, BellSouth may disconnect and reuse facilities (i) where BellSouth is providing switching; (ii) when the facility is in a denied state (i.e., service is no longer being provided over the facility but the facility has not been disconnected); and (iii) BellSouth has received an order to establish new service or transfer of service from a customer or a customer's CLEC at the same address served by the denied facility.
- 3.3.1 Upon receipt of a service order, BellSouth will do the following:
- 3.3.1.1 Process disconnect and reconnect orders to provision the service which shall be due dated using current interval guidelines.
- Reuse the serving facility for the retail, resale service, or network element at the same location.
- 3.3.1.3 Notify NewSouth after the disconnect order has been completed.
- 3.4 <u>Contact Numbers</u>. The Parties agree to provide one another with toll-free contact numbers for the purpose of ordering, provisioning and maintenance of services.

BellSouth shall provide single points of contact ("SPOC") for the provisioning of Resale Services (LCSC) and UNEs (UNE Center) ordered by NewSouth. Preordering and ordering electronic interfaces shall be available, subject to downtime for scheduled maintenance and other scheduled activities for which advance notification is required and provided by BellSouth, seven (7) days a week, 24 hours a day. BellSouth shall provide access to assistance for technical issues such as connectivity and passwords related to LENS, TAG and TAFI, and to the "EDI Central Group" for

- technical problems with EDI. Assistance will be available by telephone during normal business hours and through other contacts on nights, weekends and holidays.
- 3.5 <u>Subscription Functions</u>. In cases where BellSouth performs subscription functions for an inter-exchange carrier (i.e. PIC and LPIC changes via Customer Account Record Exchange (CARE)), BellSouth will provide the affected inter-exchange carriers with the Operating Company Number (OCN) of the local provider for the purpose of obtaining end user billing account and other end user information required under subscription requirements.
- 3.6 <u>Cancellation Charges</u>. If NewSouth cancels an order for network elements or other services, any costs incurred by BellSouth in conjunction with the provisioning of that order will be recovered in accordance with FCC No. 1 Tariff, Section 5.4.
- 3.7 <u>Disaster Recovery Plan.</u> BellSouth's Disaster Recover Plan is as set forth in Attachment 11 of this Agreement.
- 3.8 Ordering and Provisioning Information. BellSouth shall provide the following to NewSouth upon request:
- Design Layout Records ("DLRs") for designed unbundled Network Elements where applicable;
- 3.8.2 Advance information on the details and requirements for planning and implementation of NPA splits; and
- 3.8.3 Access to the Regional Street Address Guide ("RSAG") information via LENS, TAG or RoboTAGTM pre-ordering.
- Each Party shall establish mutually acceptable methods and procedures for handling all misdirected calls from the other Party's End Users. Each Party, on a reciprocal basis, shall refer all misdirected calls that it receives from the other Party's End Users to a designated number of the other Party as set forth in this section. NewSouth and BellSouth each shall be responsible for providing the other party with its current toll free number. The foregoing shall apply only when the Party receiving such call knows or has reason to know that the call is misdirected from an End User of the other Party hereto.
- 3.10 BellSouth shall provide order format specifications to NewSouth for all available services, features, and functions and for ancillary data required by BellSouth to provision those services.
- 3.11 BellSouth shall provide NewSouth with standard expected provisioning intervals for all unbundled Network Elements.
- 3.12 BellSouth shall not reconfigure any NewSouth service arrangements of any NewSouth End User for Resale services, UNEs or Combinations, unless so directed by NewSouth. Any NewSouth End User that contacts BellSouth regarding a change

to its NewSouth service (excluding changes in its local service provider) shall be advised to contact NewSouth. Any BellSouth End User that contacts NewSouth regarding a change in BellSouth service (excluding changes in its local service provider) shall be advised to contact BellSouth.

- 3.13 The Parties shall provide a generic intercept referral message that includes any new telephone number of an End User for the same period of time that BellSouth currently provides such a message for its own End Users. The intercept message shall be similar in format to the intercept referral message currently provided by BellSouth for its own End Users.
- 3.14 BellSouth shall perform all pre-testing necessary to ensure the services ordered meet the specifications outlined in the technical service description provided by BellSouth for the service being ordered.
- Any written "leave behind" materials that BellSouth technicians provide to NewSouth End Users shall be non-branded materials that do not identify the work being performed as being by BellSouth. These materials shall include, without limitation, non-branded forms for the Customer and non-branded "not at home" cards.
- 3.16 If a NewSouth End User requests a change of service at the time of installation, BellSouth technicians shall direct them to contact NewSouth directly and provide a toll-free number supplied by NewSouth. When a BellSouth employee visits the premise of an NewSouth End User, the BellSouth employee shall inform the Customer that he or she is there acting on behalf of NewSouth.
- 3.17 BellSouth shall provide telephone and/or facsimile notification to NewSouth of any NewSouth end user service requests and charges therefore not authorized on the NewSouth service request, and obtain NewSouth's approval prior to commencing work.
- Each Party shall train and direct its employees who have contact with End Users of the other Party in the process of provisioning, maintenance or repair not to disparage the other Party or its services in any way to the other Party's End Users.
- When NewSouth places an order, NewSouth shall specify a requested Due Date, and BellSouth shall specify a Due Date based on the applicable intervals. In the event NewSouth's requested date is less than the standard interval, NewSouth shall contact BellSouth by telephone or use the expedite request field on the order as directed by BellSouth and the Parties shall negotiate an expedited Due Date as set forth in this Agreement. Expedite charges shall be as set-forth in Attachment 2 of this Agreement. BellSouth shall not complete the order prior to the Due Date unless authorized by NewSouth. If BellSouth misses the Due Date, BellSouth shall promptly notify NewSouth of the revised installation Due Date. If NewSouth requests that an order be expedited, BellSouth shall notify NewSouth of the status of the order (i) by the end of the same Business Day when such expedite requests are made prior to noon; or (ii) by noon the following Business Day otherwise.

- 3.20 NewSouth and BellSouth shall agree to escalation procedures and contacts for resolving questions and disputes related to ordering and provisioning procedures or to the processing of individual orders, subject ultimately to the dispute resolution provisions of this Agreement. The Parties shall use best efforts to notify each other of any modifications to these contacts within ten (10) days of any such modifications.
- 3.21 BellSouth shall transmit to NewSouth a FOC or, in the alternative, notification of the lack of available facilities within time periods specified hereafter, after BellSouth's receipt of a complete and correct order from NewSouth, provided, however, that an order for complex services requiring a service inquiry shall be deemed received for these purposes only after completion of the service inquiry. The FOC shall contain a commitment date, which shall be established on a nondiscriminatory basis with respect to installation dates for comparable orders at such time. If NewSouth uses LENS, EDI, or any other electronic interface for the submission of the order, the FOC or notification shall be posted by BellSouth in such interface within twenty-four (24) hours of receipt of the order for all fully mechanized requests. If NewSouth does not use these interfaces, or these interfaces are not available for the service or UNE being ordered, BellSouth shall transmit the FOC or notification by telecopier to a toll-free number provided by NewSouth within forty-eight (48) hours of BellSouth's receipt of the order. When NewSouth submits a complete and correct LSR for SPNP and an associated unbundled Loop simultaneously, BellSouth shall likewise issue a FOC for both the Loop and the SPNP simultaneously.
- 3.22 For Local Service Requests submitted via an electronic interface, BellSouth shall notify NewSouth via the same electronic interface, of Rejections/Errors contained in any of the data element(s) field(s) contained on any NewSouth Local Service Request. For Local Service Requests submitted manually, BellSouth shall notify NewSouth by facsimile of such Rejections and Errors. BellSouth will notify NewSouth of Rejections or Errors in 95% of mechanized orders within one (1) hour from BellSouth's receipt of the order. BellSouth will notify NewSouth of Rejections or Errors in 85% of non-mechanized and partially mechanized orders within forty-eight (48) hours from BellSouth's receipt of the order.

Attachment 7

Billing and Billing Accuracy Certification

TABLE OF CONTENTS

1.	Payment and Billing Arrangements	3
2.	Billing Accuracy Certification	6
3.	Billing Disputes	7
4.	RAO Hosting	8
5.	Optional Daily Usage File	11
6.	Access Daily Usage File	14
7.	Enhanced Optional Daily Usage File	17
Rя	ites Ex	hibit A

BILLING AND BILLING ACCURACY CERTIFICATION

1. Payment and Billing Arrangements

All negotiated rates, terms and conditions set forth in this Attachment pertain to billing and billing accuracy certifications.

- Billing. BellSouth agrees to provide billing through the Carrier Access Billing System (CABS) and through the Customer Records Information System (CRIS) depending on the particular service(s) that NewSouth requests. BellSouth will bill and record in accordance with this Agreement those charges NewSouth incurs as a result of NewSouth purchasing from BellSouth Network Elements and Other Services as set forth in this Agreement. BellSouth will format all bills in CBOS Standard or CLUB/EDI format, depending on the type of service ordered. For those services where standards have not yet been developed, BellSouth's billing format will change as necessary when standards are finalized by the industry forum.
- 1.1.1 For any service(s) BellSouth orders from NewSouth, NewSouth shall bill BellSouth in CABS format or in accordance with industry standards.
- 1.1.2 If either Party requests multiple billing media or additional copies of bills, the Billing Party will provide these at a reasonable cost.
- Master Account. After receiving certification as a local exchange company from the appropriate regulatory agency, NewSouth will provide the appropriate BellSouth account manager the necessary documentation to enable BellSouth to establish a master account for Local Interconnection, Network Elements and Other Services, and/or resold services. Such documentation shall include the Application for Master Account, proof of authority to provide telecommunications services, an Operating Company Number ("OCN") assigned by the National Exchange Carriers Association ("NECA"), Carrier Identification Code (CIC), Group Access Code (GAC), Access Customer Name and Address (ACNA) and a tax exemption certificate, if applicable.
- Payment Responsibility. Payment of all charges will be the responsibility of the billed Party. The billed Party shall make payment to the billing Party for all services billed. The billing Party is not responsible for payments not received by the billed Party from the billed Party's customer. The billing Party will not become involved in billing disputes that may arise between the billed Party and the billed Party's customer. Payments made to the billing Party as payment on account will be credited to an accounts receivable master account and not to an end user's account.
- Payment Due. The payment will be due on or before the next bill date (i.e., same date in the following month as the bill date) and is payable in immediately available funds. Payment is considered to have been made when received by the billing Party.

Version 1Q00:3/6/00 05/18/01

If the payment due date falls on a Sunday or on a Holiday which is observed on a Monday, the payment due date shall be the first non-Holiday day following such Sunday or Holiday. If the payment due date falls on a Saturday or on a Holiday which is observed on Tuesday, Wednesday, Thursday, or Friday, the payment due date shall be the last non-Holiday day preceding such Saturday or Holiday. If payment is not received by the payment due date, a late payment penalty, as set forth in Section 1.7, below, shall apply.

- 1.5 <u>Tax Exemption</u>. Upon proof of tax exempt certification from NewSouth, the total amount billed to NewSouth will not include those taxes or fees for which the CLEC is exempt. NewSouth will be solely responsible for the computation, tracking, reporting and payment of all taxes and like fees associated with the services provided to the end user of NewSouth.
- Late Payment. If any portion of the payment is received by the billing Party after the payment due date as set forth preceding, or if any portion of the payment is received by the billing Party in funds that are not immediately available to the billing Party, then a late payment penalty shall be due to the billing Party. The late payment penalty shall be the portion of the payment not received by the payment due date times a late factor and will be applied on a per bill basis. When BellSouth is the billing Party, the late factor shall be as set forth in Section A2 of the General Subscriber Services Tariff, Section B2 of the Private Line Service Tariff or Section E2 of the Intrastate Access Tariff, as appropriate. When NewSouth is the billing Party the late factor shall be one and one-half percent (1.5%) per month. The billed Party will be charged a fee for all returned checks as set forth in Section A2 of the General Subscriber Services Tariff or pursuant to the applicable state law.
- 1.7 <u>Discontinuing Service to NewSouth</u>. The procedures for discontinuing service to NewSouth are as follows:
- 1.7.1 BellSouth reserves the right to suspend or terminate service for nonpayment of services or in the event of prohibited, unlawful or improper use of BellSouth facilities or service or any other violation or noncompliance by NewSouth of the rules and regulations contained in BellSouth's tariffs.
- 1.7.2 If payment of <u>Undisputed Amounts</u> due, as described in Section 3 of this Attachment, is not received by the bill date in the month after the original bill date, BellSouth may provide written notice to NewSouth that additional applications for service will be refused and that any pending orders for service will not be completed if payment is not received by the fifteenth day following the date of the notice. In addition, BellSouth may, at the same time, give thirty (30) days notice to NewSouth at the billing address to discontinue the provision of existing services to NewSouth at any time thereafter.

- 1.7.3 In the case of such discontinuance, all billed charges, as well as applicable termination charges, shall become due.
- 1.7.4 If BellSouth does not discontinue the provision of the services involved on the date specified in the thirty days notice and NewSouth's noncompliance continues, nothing contained herein shall preclude BellSouth's right to discontinue the provision of the services to NewSouth without further notice.
- If payment is not received or satisfactory arrangements made for payment by the date given in the written notification, NewSouth's services will be discontinued. Upon discontinuance of service on NewSouth's account, service to the NewSouth's end users will be denied. BellSouth will reestablish service at the request of the end user or NewSouth for BellSouth to reestablish service upon payment of the appropriate connection fee and subject to BellSouth's normal application procedures. NewSouth is solely responsible for notifying the end user of the proposed service disconnection. If within fifteen (15) days after an end user's service has been denied and no arrangements to reestablish service have been made consistent with this subsection, the end user's service will be disconnected.
- 1.8 Deposit Policy. When purchasing services from BellSouth, NewSouth will be required to complete the BellSouth Credit Profile and provide information regarding credit worthiness. Based on the results of the credit analysis, BellSouth reserves the right to secure the account with a suitable form of security deposit. Such security deposit shall take the form of cash, an Irrevocable Letter of Credit (BellSouth form), Surety Bond (BellSouth form) or, in BellSouth's reasonable and nondiscriminatory discretion, some other form of security. Any such security deposit shall in no way release NewSouth from its obligation to make complete and timely payments of its bill. Such security shall be required prior to the inauguration of service. If, in the reasonable opinion of BellSouth, circumstances so warrant and/or gross monthly billing has increased beyond the level initially used to determine the level of security, BellSouth reserves the right to request additional security. Interest on a security deposit, if provided in cash, shall accrue and be paid in accordance with the terms in the appropriate BellSouth tariff. Security deposits collected under this Section shall not exceed two months' estimated billing. In the event NewSouth fails to remit to BellSouth any deposit requested pursuant to this Section, service to NewSouth may be terminated in accordance with the terms of Section 1.7 of this Attachment, and any security deposits will be applied to NewSouth's account(s).
- Notices. Notwithstanding anything to the contrary in this Agreement, all bills and notices regarding billing matters, including notices relating to security deposits, to rejection of additional orders from NewSouth and to disconnection of services for nonpayment of charges, shall be forwarded to the individual and/or address provided by NewSouth in establishment of its billing accounts with BellSouth, or to the individual and/or address subsequently provided by NewSouth as the contact for

billing information. All monthly bills and the notices described in this Section shall be forwarded to the same individual and/or address; provided, however, upon written notice from NewSouth to BellSouth's billing organization, a final notice of disconnection of services purchased by NewSouth under this Agreement shall be sent via certified mail to the individual(s) listed in the Notices provision of the General Terms and Conditions of this Agreement at least 30 days before BellSouth takes any action to terminate such services.

Rates. Rates for Optional Daily Usage File (ODUF), Enhanced Optional Daily Usage File (EODUF), Access Daily Usage File (ADUF), and Centralized Message Distribution Service (CMDS) are set out in Exhibit A to this Attachment. If no rate is identified in this Attachment, the rate for the specific service or function will be as set forth in applicable BellSouth tariff or as negotiated by the Parties upon request by either Party.

2. Billing Accuracy Certification

- Upon request, BellSouth and NewSouth will agree upon a billing quality assurance program for all billing elements covered in this Agreement that will eliminate the need for post-billing reconciliation. Appropriate terms for access to any BellSouth documents, systems, records, and procedures for the recording and billing of charges will be part of that program.
- As part of the billing quality assurance program, BellSouth and NewSouth will develop standards, measurements, and performance requirements for a local billing measurements process. On a regular basis BellSouth will provide NewSouth with mutually agreed upon performance measurement data that substantiates the accuracy, reliability, and integrity of the billing process for local billing. In return, NewSouth will pay all bills received from BellSouth in full by the payment due date.
- 2.3 Local billing discrepancies will be addressed in an orderly manner via a mutually agreed upon billing exemption process.
- 2.3.1 Each Party agrees to notify the other Party upon identifying a billing discrepancy. The Parties shall endeavor to resolve any billing discrepancy within sixty (60) calendar days of the notification date. A mutually agreed upon escalation process will be established for resolving local billing discrepancies as part of the billing quality assurance program.
- 2.3.2 Closure of a specific billing period will occur by joint agreement of the Parties whereby the Parties agree that such billing period is closed to any further analysis and financial transactions except those resulting from regulatory mandates. Closure will take place within a mutually agreed upon time interval from the bill date. The month being closed represents those charges that were billed or should have been billed by the designated bill date.

Version IQ00:3/6/00 05/18/01

3. Billing Disputes

- Where the Parties have not agreed upon a billing quality assurance program, or where such process does not encompass all billing disputes, billing disputes shall be handled pursuant to the terms of this section.
- 3.1.1 Each Party agrees to notify the other Party in writing upon the discovery of a billing dispute. In the event of a billing dispute, the Parties will endeavor to resolve the dispute within sixty (60) calendar days of the notification date. If the Parties are unable within the 60 day period to reach resolution, then the aggrieved Party may pursue dispute resolution in accordance with the General Terms and Conditions of this Agreement.
- 3.1.2 For purposes of this Section 3, a billing dispute means a dispute of a specific amount of money actually billed by either Party. The dispute must be clearly explained by the disputing Party and supported by written documentation, which clearly shows the basis for disputing the charges. By way of example and not by limitation, a billing dispute will not include the refusal to pay all or part of a bill or bills when no written documentation is provided to support the dispute, nor shall a billing dispute include the refusal to pay other amounts owed by the billed Party until the dispute is resolved. Claims by the billed Party for damages of any kind will not be considered a billing dispute for purposes of this Section 3. Once the billing dispute is resolved, the disputing Party will make immediate payment on any of the disputed amount owed to the billing Party or the billing Party shall have the right to pursue normal treatment procedures. Any credits due to the disputing Party, pursuant to the billing dispute, will be applied to the disputing Party's account by the billing Party immediately upon resolution of the dispute.
- 3.2 If a Party disputes a charge and does not pay such charge by the payment due date, or if a payment or any portion of a payment is received by either Party after the payment due date, or if a payment or any portion of a payment is received in funds which are not immediately available to the other Party, then a late payment penalty shall be assessed. For bills rendered by either Party for payment, the late payment charge for both Parties shall be calculated based on the portion of the payment not received by the payment due date times the late factor as set forth in the following BellSouth tariffs: for services purchased from the General Subscribers Services Tariff for purposes of resale and for ports and non-designed loops, Section A2 of the General Subscriber Services Tariff; for services purchased from the Private Line Tariff for purposes of resale, Section B2 of the Private Line Service Tariff; and for network elements and other services and local interconnection charges, Section E2 of the Access Service Tariff. In no event, however, shall interest be assessed by either Party on any previously assessed late payment charges. The Parties shall assess interest on previously assessed late payment charges only in a state where it has the authority

pursuant to its tariffs. No interest or late payment penalties shall be assessed in the event that the billed Party prevails in a billing dispute.

4. RAO Hosting

- 4.1 RAO Hosting, Calling Card and Third Number Settlement System (CATS) and Non-Intercompany Settlement System (NICS) services provided to NewSouth by BellSouth will be in accordance with the methods and practices regularly adopted and applied by BellSouth to its own operations during the term of this Agreement, including such reasonable and nondiscriminatory revisions as may be made from time to time by BellSouth. BellSouth will use best efforts to provide NewSouth with 30 days' advanced notice of such revisions.
- 4.2 NewSouth shall furnish all relevant information required by BellSouth for the provision of RAO Hosting, CATS and NICS.
- Compensation amounts, if applicable, will be billed by BellSouth to NewSouth on a monthly basis in arrears. Amounts due from one Party to the other (excluding adjustments) are payable within thirty (30) days of receipt of the billing statement.
- NewSouth must have its own unique hosted RAO code. Requests for establishment of RAO status where BellSouth is the selected Centralized Message Distribution System (CMDS) interfacing host, require written notification from NewSouth to the BellSouth RAO Hosting coordinator at least eight (8) weeks prior to the proposed effective date. The proposed effective date will be mutually agreed upon between the Parties with consideration given to time necessary for the completion of required Telcordia (formerly BellCore) functions. BellSouth will request the assignment of an RAO code from its connecting contractor, currently Telcordia (formerly BellCore), on behalf of NewSouth and will coordinate all associated conversion activities.
- 4.5 BellSouth will receive messages from NewSouth that are to be processed by BellSouth, another LEC or CLEC in the BellSouth region or a LEC outside the BellSouth region.
- 4.6 BellSouth will perform invoice sequence checking, standard EMI format editing, and balancing of message data with the EMI trailer record counts on all data received from NewSouth.
- 4.7 All data received from NewSouth that is to be processed or billed by another LEC or CLEC within the BellSouth region will be distributed to that LEC or CLEC in accordance with the Agreement(s) which may be in effect between BellSouth and the involved LEC or CLEC.

- All data received from NewSouth that is to be placed on the CMDS network for distribution outside the BellSouth region will be handled in accordance with the agreement(s) which may be in effect between BellSouth and its connecting contractor (currently Telcordia (formerly BellCore)).
- 4.9 BellSouth will receive messages from the CMDS network that are destined to be processed by NewSouth and will forward them to NewSouth on a daily basis.
- 4.10 Transmission of message data between BellSouth and NewSouth will be via CONNECT:Direct.
- 4.11 All messages and related data exchanged between BellSouth and NewSouth will be formatted in accordance with accepted industry standards for EMI formatted records and packed between appropriate EMI header and trailer records, also in accordance with accepted industry standards.
- 4.12 NewSouth will ensure that the recorded message detail necessary to recreate files provided to BellSouth will be maintained for back-up purposes for a period of three (3) calendar months beyond the related message dates.
- Should it become necessary for NewSouth to send data to BellSouth more than sixty (60) days past the message date(s), NewSouth will notify BellSouth in advance of the transmission of the data. If there will be impacts outside the BellSouth region, BellSouth will work with its connecting contractor and NewSouth to notify all affected Parties.
- In the event that data to be exchanged between the two Parties should become lost or destroyed, both Parties will work together to determine the source of the problem. Once the cause of the problem has been jointly determined and the responsible Party (BellSouth or NewSouth) identified and agreed to, the company responsible for creating the data (BellSouth or NewSouth) will make every effort to have the affected data restored and retransmitted. If the data cannot be retrieved, the responsible Party will be liable to the other Party for any resulting lost revenue. Lost revenue may be a combination of revenues that could not be billed to the end users and associated access revenues. Both Parties will work together to estimate the revenue amount based upon historical data through a method mutually agreed upon. The resulting estimated revenue loss will be paid by the responsible Party to the other Party within three (3) calendar months of the date of problem resolution, or as mutually agreed upon by the Parties.
- Should an error be detected by the EMI format edits performed by BellSouth on data received from NewSouth, the entire pack containing the affected data will not be processed by BellSouth. BellSouth will notify NewSouth of the error condition. NewSouth will correct the error(s) and will resend the entire pack to BellSouth for processing. In the event that an out-of-sequence condition occurs on subsequent

- packs, NewSouth will resend these packs to BellSouth after the pack containing the error has been successfully reprocessed by BellSouth.
- 4.16 In association with message distribution service, BellSouth will provide NewSouth with associated intercompany settlements reports (CATS and NICS) as appropriate.
- 4.17 In no case shall either Party be liable to the other for any direct or consequential damages incurred as a result of the obligations set out in this Agreement.

4.18 RAO Compensation

- 4.18.1 Rates for message distribution service provided by BellSouth for NewSouth are as set forth in Exhibit A to this Attachment.
- 4.18.2 Rates for data transmission associated with message distribution service are as set forth in Exhibit A to this Attachment.
- Data circuits (private line or dial-up) will be required between BellSouth and NewSouth for the purpose of data transmission. Where a dedicated line is required, NewSouth will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. NewSouth will also be responsible for charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to NewSouth. Additionally, all message toll charges associated with the use of the dial circuit by NewSouth will be the responsibility of NewSouth. Associated equipment on the BellSouth end, including a modem, will be negotiated on a case by case basis between the Parties.
- 4.18.4 All equipment, including modems and software that is required on the NewSouth end for the purpose of data transmission will be the responsibility of NewSouth.

4.19 <u>Intercompany Settlements Messages</u>

- 4.19.1 This Section addresses the settlement of revenues associated with traffic originated from or billed by NewSouth as a facilities based provider of local exchange telecommunications services outside the BellSouth region. Only traffic that originates in one Bell operating territory and bills in another Bell operating territory is included. Traffic that originates and bills within the same Bell operating territory will be settled on a local basis between NewSouth and the involved company (ies), unless that company is participating in NICS.
- 4.19.2 Both traffic that originates outside the BellSouth region by NewSouth and is billed within the BellSouth region, and traffic that originates within the BellSouth region

Version IQ00:3/6/00

and is billed outside the BellSouth region by NewSouth, is covered by this Agreement (CATS). Also covered is traffic that either is originated by or billed by NewSouth, involves a company other than NewSouth, qualifies for inclusion in the CATS settlement, and is not originated or billed within the BellSouth region (NICS).

- 4.19.3 Once NewSouth is operating within the BellSouth territory, revenues associated with calls originated and billed within the BellSouth region will be settled via Telcordia (formerly BellCore)'s, its successor or assign, NICS system.
- 4.19.4 BellSouth will receive the monthly NICS reports from Telcordia (formerly BellCore), its successor or assign, on behalf of NewSouth. BellSouth will distribute copies of these reports to NewSouth on a monthly basis.
- 4.19.5 BellSouth will receive the monthly Calling Card and Third Number Settlement System (CATS) reports from Telcordia (formerly BellCore), its successor or assign, on behalf of NewSouth. BellSouth will distribute copies of these reports to NewSouth on a monthly basis.
- 4.19.6 BellSouth will collect the revenue earned by NewSouth from the Bell operating company in whose territory the messages are billed (CATS), less a per message billing and collection fee of five cents (\$0.05), on behalf of NewSouth. BellSouth will remit the revenue billed by NewSouth to the Bell operating company in whose territory the messages originated, less a per message billing and collection fee of five cents (\$0.05), on behalf on NewSouth. These two amounts will be netted together by BellSouth and the resulting charge or credit issued to NewSouth via a monthly Carrier Access Billing System (CABS) miscellaneous bill.
- 4.19.7 BellSouth will collect the revenue earned by NewSouth within the BellSouth territory from another CLEC also within the BellSouth territory (NICS) where the messages are billed, less a per message billing and collection fee of five cents (\$0.05), on behalf of NewSouth. BellSouth will remit the revenue billed by NewSouth within the BellSouth region to the CLEC also within the BellSouth region, where the messages originated, less a per message billing and collection fee of five cents (\$0.05). These two amounts will be netted together by BellSouth and the resulting charge or credit issued to NewSouth via a monthly Carrier Access Billing System (CABS) miscellaneous bill.

BellSouth and NewSouth agree that monthly netted amounts of less than fifty dollars (\$50.00) will not be settled.

5. Optional Daily Usage File

- Upon written request from NewSouth, BellSouth will provide the Optional Daily 5.1 Usage File (ODUF) service to NewSouth pursuant to the terms and conditions set forth in this section.
- 5.2 The NewSouth shall furnish all relevant information required by BellSouth for the provision of the Optional Daily Usage File.
- The Optional Daily Usage Feed will contain billable messages that were carried over 5.3 the BellSouth Network and processed in the BellSouth Billing System, but billed to a NewSouth customer.

Charges for delivery of the Optional Daily Usage File will appear on the NewSouth's monthly bills. The charges are as set forth in Exhibit A to this Attachment.

- The Optional Daily Usage Feed will contain both rated and unrated messages. All 5.4 messages will be in the standard Alliance for Telecommunications Industry Solutions (ATIS) EMI record format.
- 5.5 Messages that error in the billing system of the NewSouth will be the responsibility of the NewSouth. If, however, the NewSouth should encounter significant volumes of errored messages that prevent processing by the NewSouth within its systems, BellSouth will work with the NewSouth to determine the source of the errors and the appropriate resolution.
- 5.6 The following specifications shall apply to the Optional Daily Usage Feed.
- 5.6.1 Usage To Be Transmitted
- 5.6.1.1 The following messages recorded by BellSouth will be transmitted to the NewSouth:
 - Message recording for per use/per activation type services (examples: Three Way Calling, Verify, Interrupt, Call Return, ETC.)
 - Measured billable Local
 - Directory Assistance messages
 - IntraLATA Toll
 - WATS & 800 Service
 - N11
 - Information Service Provider Messages
 - Operator Services Messages
 - Operator Services Message Attempted Calls (Network Element only)
 - Credit/Cancel Records
 - Usage for Voice Mail Message Service

Version 1Q00:3/6/00

- 5.6.1.2 Rated Incollects (originated in BellSouth and from other companies) can also be on Optional Daily Usage File. Rated Incollects will be intermingled with BellSouth recorded rated and unrated usage. Rated Incollects will not be packed separately.
- 5.6.1.3 BellSouth will perform duplicate record checks on records processed to Optional Daily Usage File. Any duplicate messages detected will be deleted and not sent to NewSouth.
- 5.6.1.4 In the event that NewSouth detects a duplicate on Optional Daily Usage File they receive from BellSouth, NewSouth will drop the duplicate message (NewSouth will not return the duplicate to BellSouth).

5.6.2 <u>Physical File Characteristics</u>

- The Optional Daily Usage File will be distributed to NewSouth via an agreed medium with CONNECT:Direct being the preferred transport method. The Daily Usage Feed will be a variable block format (2476) with an LRECL of 2472. The data on the Daily Usage Feed will be in a non-compacted EMI format (175 byte format plus modules). It will be created on a daily basis (Monday through Friday except holidays). Details such as dataset name and delivery schedule will be addressed during negotiations of the distribution medium. There will be a maximum of one dataset per workday per OCN.
- Data circuits (private line or dial-up) may be required between BellSouth and NewSouth for the purpose of data transmission. Where a dedicated line is required, NewSouth will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. NewSouth will also be responsible for any charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to NewSouth. Additionally, all message toll charges associated with the use of the dial circuit by NewSouth will be the responsibility of NewSouth. Associated equipment on the BellSouth end, including a modem, will be negotiated on a case by case basis between the Parties. All equipment, including modems and software that is required on NewSouth end for the purpose of data transmission will be the responsibility of NewSouth.

5.6.3 <u>Packing Specifications</u>

A pack will contain a minimum of one message record or a maximum of 99,999 message records plus a pack header record and a pack trailer record. One transmission can contain a maximum of 99 packs and a minimum of one pack.

5.6.3.2 The OCN, From RAO, and Invoice Number will control the invoice sequencing. The From RAO will be used to identify to NewSouth which BellSouth RAO that is sending the message. BellSouth and NewSouth will use the invoice sequencing to control data exchange. BellSouth will be notified of sequence failures identified by NewSouth and resend the data as appropriate.

The data will be packed using ATIS EMI records.

5.6.4 Pack Rejection

NewSouth will notify BellSouth within one business day of rejected packs (via the mutually agreed medium). Packs could be rejected because of pack sequencing discrepancies or a critical edit failure on the Pack Header or Pack Trailer records (i.e. out-of-balance condition on grand totals, invalid data populated). Standard ATIS EMI Error Codes will be used. NewSouth will not be required to return the actual rejected data to BellSouth. Rejected packs will be corrected and retransmitted to NewSouth by BellSouth.

5.6.5 Control Data

NewSouth will send one confirmation record per pack that is received from BellSouth. This confirmation record will indicate NewSouth received the pack and the acceptance or rejection of the pack. Pack Status Code(s) will be populated using standard ATIS EMI error codes for packs that were rejected by NewSouth for reasons stated in the above section.

5.6.6 Testing

Upon request from NewSouth, BellSouth shall send test files to NewSouth for the Optional Daily Usage File. The Parties agree to review and discuss the file's content and/or format. For testing of usage results, BellSouth shall request that NewSouth set up a production (LIVE) file. The live test may consist of NewSouth's employees making test calls for the types of services NewSouth requests on the Optional Daily Usage File. These test calls are logged by NewSouth, and the logs are provided to BellSouth. These logs will be used to verify the files. Testing will be completed within 30 calendar days from the date on which the initial test file was sent.

6. Access Daily Usage File

6.1. Upon written request from NewSouth, BellSouth will provide the Access Daily Usage File (ADUF) service to NewSouth pursuant to the terms and conditions set forth in this section.

Version 1Q00:3/6/00 05/18/01

- 6.2 The NewSouth shall furnish all relevant information required by BellSouth for the provision of the Access Daily Usage File.
- 6.3 The Access Daily Usage Feed will contain access messages associated with a port that NewSouth has purchased from BellSouth
- Charges for delivery of the Access Daily Usage File will appear on the NewSouth's' monthly bills. The charges are as set forth in Exhibit A to this Attachment. All messages will be in the standard Alliance for Telecommunications Industry Solutions (ATIS) EMI record format.
- Messages that error in the billing system of the NewSouth will be the responsibility of the NewSouth. If, however, the NewSouth should encounter significant volumes of errored messages that prevent processing by the NewSouth within its systems, BellSouth will work with the NewSouth to determine the source of the errors and the appropriate resolution.
- 6.6 <u>Usage To Be Transmitted</u>
- 6.6.1 The following messages recorded by BellSouth will be transmitted to NewSouth:

Originating and terminating interstate and intrastate access records associated with a port.

Terminating access records for undetermined jurisdiction access records associated with a port.

When NewSouth purchases Network Element ports from BellSouth and calls are made using these ports, BellSouth will handle the calls as follows:

Originating from Network Element and carried by Interexchange Carrier:

BellSouth will bill network element to CLEC and send access record to the CLEC via ADUF

Originating from network element and carried by BellSouth (NewSouth is BellSouth's toll customer):

BellSouth will bill resale toll rates to NewSouth and send toll record for the end user toll billing purposes via ODUF (Optional Daily Usage File). Access record will be sent to NewSouth via ADUF.

Terminating on network element and carried by Interexchange Carrier:

05/18/01

BellSouth will bill network element to NewSouth and send access record to NewSouth.

Terminating on network element and carried by BellSouth:

BellSouth will bill network element to NewSouth and send access record to NewSouth.

- BellSouth will perform duplicate record checks on records processed to the Access Daily Usage File. Any duplicate messages detected will be dropped and not sent to NewSouth.
- In the event that NewSouth detects a duplicate on the Access Daily Usage File they receive from BellSouth, NewSouth will drop the duplicate message (NewSouth will not return the duplicate to BellSouth.)
- 6.6.5 <u>Physical File Characteristics</u>
- 6.6.5.1 The Access Daily Usage File will be distributed to NewSouth via an agreed medium with CONNECT:Direct being the preferred transport method. The Daily Usage Feed will be a fixed block format (2476) with an LRECL of 2472. The data on the Daily Usage Feed will be in a non-compacted EMI format (210 byte format plus modules). It will be created on a daily basis (Monday through Friday except holidays). Details such as dataset name and delivery schedule will be addressed during negotiations of the distribution medium. There will be a maximum of one dataset per workday per OCN.
- Data circuits (private line or dial-up) may be required between BellSouth and NewSouth for the purpose of data transmission. Where a dedicated line is required, NewSouth will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. NewSouth will also be responsible for any charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to NewSouth. Additionally, all message toll charges associated with the use of the dial circuit by NewSouth will be the responsibility of NewSouth. Associated equipment on the BellSouth end, including a modem, will be negotiated on a case by case basis between the Parties. All equipment, including modems and software that is required on NewSouth end for the purpose of data transmission will be the responsibility of NewSouth.

05/18/01

6.6.6 Packing Specifications

- 6.6.6.1 A pack will contain a minimum of one message record or a maximum of 99,999 message records plus a pack header record and a pack trailer record. One transmission can contain a maximum of 99 packs and a minimum of one pack.
- The OCN, From RAO, and Invoice Number will control the invoice sequencing. The From RAO will be used to identify to NewSouth which BellSouth RAO that is sending the message. BellSouth and NewSouth will use the invoice sequencing to control data exchange. BellSouth will be notified of sequence failures identified by NewSouth and resend the data as appropriate.

The data will be packed using ATIS EMI records.

6.6.7 Pack Rejection

NewSouth will notify BellSouth within one business day of rejected packs (via the mutually agreed medium). Packs could be rejected because of pack sequencing discrepancies or a critical edit failure on the Pack Header or Pack Trailer records (i.e. out-of-balance condition on grand totals, invalid data populated). Standard ATIS EMI Error Codes will be used. NewSouth will not be required to return the actual rejected data to BellSouth. Rejected packs will be corrected and retransmitted to NewSouth by BellSouth.

6.6.8 <u>Control Data</u>

NewSouth will send one confirmation record per pack that is received from BellSouth. This confirmation record will indicate NewSouth received the pack and the acceptance or rejection of the pack. Pack Status Code(s) will be populated using standard ATIS EMI error codes for packs that were rejected by NewSouth for reasons stated in the above section.

6.6.9 Testing

Upon request from NewSouth, BellSouth shall send test files to NewSouth for the Access Daily Usage File. Testing shall consist of actual calls made from live accounts. A call log shall be supplied along with test request information. The Parties agree to review and discuss the file's content and/or format.

Enhanced Optional Daily Usage File

7.1 Upon written request from NewSouth, BellSouth will provide the Enhanced Optional Daily Usage File (EODUF) service to NewSouth pursuant to the terms and conditions

Version 1Q00:3/6/00

set forth in this section. EODUF will only be sent to existing ODUF subscribers who request the EODUF option.

- 7.2 The NewSouth shall furnish all relevant information required by BellSouth for the provision of the Enhanced Optional Daily Usage File.
- 7.3 The Enhanced Optional Daily Usage File (EODUF) will provide usage data for local calls originating from resold Flat Rate Business and Residential Lines.

Charges for delivery of the Enhanced Optional Daily Usage File will appear on the NewSouth' monthly bills. The charges are as set forth in Exhibit A to this Attachment.

- 7.4 All messages will be in the standard Alliance for Telecommunications Industry Solutions (ATIS) EMI record format.
- Messages that error in the billing system of the NewSouth will be the responsibility of the NewSouth. If, however, the NewSouth should encounter significant volumes of errored messages that prevent processing by the NewSouth within its systems, BellSouth will work with the NewSouth to determine the source of the errors and the appropriate resolution.
- 7.6 The following specifications shall apply to the Optional Daily Usage Feed.
- 7.6.1 <u>Usage To Be Transmitted</u>
- 7.6.1.1 The following messages recorded by BellSouth will be transmitted to the NewSouth:

Customer usage data for flat rated local call originating from CLEC end user lines (1FB or 1FR). The EODUF record for flat rate messages will include:

Date of Call
From Number
To Number
Connect Time
Conversation Time
Method of Recording
From RAO
Rate Class
Message Type
Billing Indicators
Bill to Number

7.6.1.2 BellSouth will perform duplicate record checks on EODUF records

Version 1Q00:3/6/00

- processed to Optional Daily Usage File. Any duplicate messages detected will be deleted and not sent to NewSouth.
- 7.6.1.3 In the event that NewSouth detects a duplicate on Enhanced Optional Daily Usage File they receive from BellSouth, NewSouth will drop the duplicate message (NewSouth will not return the duplicate to BellSouth).

7.6.2 Physical File Characteristics

- 7.6.2.1 The Enhanced Optional Daily Usage Feed will be distributed to NewSouth over their existing Optional Daily Usage File (ODUF) feed. The EODUF messages will be intermingled among NewSouth's Optional Daily Usage File (ODUF) messages. The EODUF will be a variable block format (2476) with an LRECL of 2472. The data on the EODUF will be in a non-compacted EMI format (175 byte format plus modules). It will be created on a daily basis (Monday through Friday except holidays).
- NewSouth for the purpose of data transmission. Where a dedicated line is required, NewSouth will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. NewSouth will also be responsible for any charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to NewSouth. Additionally, all message toll charges associated with the use of the dial circuit by NewSouth will be the responsibility of NewSouth. Associated equipment on the BellSouth end, including a modem, will be negotiated on a case by case basis between the Parties. All equipment, including modems and software that is required on NewSouth end for the purpose of data transmission will be the responsibility of NewSouth.

7.6.3 <u>Packing Specifications</u>

- 7.6.3.1 A pack will contain a minimum of one message record or a maximum of 99,999 message records plus a pack header record and a pack trailer record. One transmission can contain a maximum of 99 packs and a minimum of one pack.
- 7.6.3.2 The Operating Company Number (OCN), From Revenue Accounting Office (RAO), and Invoice Number will control the invoice sequencing. The From RAO will be used to identify to NewSouth which BellSouth RAO that is sending the message. BellSouth and NewSouth will use the invoice sequencing to control data exchange. BellSouth will be notified of sequence failures identified by NewSouth and resend the data as appropriate.

Version 1Q00:3/6/00 05/18/01

The data will be packed using ATIS EMI records.

BELLSOUTH/NEWSOUTH RATES ODUF/EDQUF/ADUF/CMDS

<

Attachment 8

Rights-of-Way, Conduits and Pole Attachments

Rights-of-Way, Conduits and Pole Attachments

BellSouth will provide nondiscriminatory access to any pole, duct, conduit, or right-of-way owned or controlled by BellSouth pursuant to 47 U.S.C. § 224, as amended by the Act, pursuant to terms and conditions of a license agreement subsequently negotiated with BellSouth's Competitive Structure Provisioning Center.

Attachment 9

Performance Measurements

Performance Measurements

Upon a particular Commission's issuance of an Order pertaining to Performance Measurements and/or Remedies in a proceeding expressly applicable to all CLECs generally, BellSouth shall implement in that state such Performance Measurements as of the date specified by the Commission.

for NewSouth

BellSouth Standard Interconnection Agreement

Agreement Effective Date:	Date:		Agreement Expiration Date:
Account Manager:			Account Manager Tel No:
Attachment Name/Number	Section Number	Version Date	. Planned Activities
Terms/Conditions PartA	1	2/29/00	
	2	2/29/00	
	3	2/29/00	
	4	2/29/00	
	5	2/29/00	
	9	2/29/00	
	7	2/29/00	
	∞	2/29/00	
	6	2/29/00	
	10	2/29/00	
	11	2/29/00	
	12	2/29/00	
	13	2/29/00	
	14	2/29/00	
	15	2/29/00	
	16	2/29/00	
	17	2/29/00	
	18	2/29/00	
	61	2/29/00	
	20	2/29/00	
	21	2/29/00	
	22	2/29/00	
	23	2/29/00	
	24	2/29/00	
	25	2/29/00	
	26	2/29/00	
Terms/Conditions Part B		2/29/00	
1-Resale	1	2/29/00	

1-Resale Version 1Q00:3/6/00

Attachment 10-Residence

for NewSouth

BellSouth Standard Interconnection Agreement

Name/Number	Date 2/29/00 2/29/00 2/29/00 2/29/00 2/29/00 2/29/00 2/29/00 2/29/00 2/29/00 2/29/00 2/29/00 2/29/00	
2 3 4 4 5 6 6 10 10 11 11 11 12 13 Exhibit B Exhibit D Exhibit D Exhibit D Exhibit D	2/29/00 2/29/00 2/29/00 2/29/00 2/29/00 2/29/00 2/29/00 2/29/00 2/29/00	
3	2/29/00 2/29/00 2/29/00 2/29/00 2/29/00 2/29/00 2/29/00 2/29/00	
4 5 6 7 7 8 8 9 9 9 10 11 11 12 12 13 13 Exhibit A Exhibit B Exhibit C Exhibit D Exhibit D Exhibit D Exhibit D Exhibit D	2/29/00 2/29/00 2/29/00 2/29/00 2/29/00 2/29/00 2/29/00	
5 6 7 7 8 8 9 10 11 11 12 12 Exhibit B Exhibit C Exhibit D	2/29/00 2/29/00 2/29/00 2/29/00 2/29/00 2/29/00 2/29/00	
6 7 8 8 9 10 10 11 11 12 13 Exhibit B Exhibit D Exhibit D	2/29/00 2/29/00 2/29/00 2/29/00 2/29/00 2/29/00	
7 8 9 9 10 10 11 11 12 13 Exhibit B Exhibit D Exhibit D	2/29/00 2/29/00 2/29/00 2/29/00 2/29/00	
8 9 10 10 11 12 13 Exhibit B Exhibit C Exhibit D Exhibit D	2/29/00 2/29/00 2/29/00 2/29/00 2/29/00	
9 10 11 11 12 13 Exhibit A Exhibit B Exhibit B Exhibit D Exhibit D	2/29/00 2/29/00 2/29/00 2/29/00	
10 11 12 13 13 Exhibit A Exhibit B Exhibit C Exhibit D	2/29/00	
11 12 13 13 Exhibit A Exhibit B Exhibit D Exhibit D	2/29/00	
12 13 Exhibit A Exhibit B Exhibit B Exhibit C Exhibit D	2/29/00	
Exhibit A Exhibit B Exhibit B Exhibit C Exhibit D	00/06/6	
Exhibit A Exhibit B Exhibit C Exhibit D Exhibit D	2017	
Exhibit B Exhibit C Exhibit D	2/29/00	
Exhibit D Exhibit D	2/29/00	
Exhibit D	2/29/00	
1:3:3:	2/29/00	
EXMONE	2/29/00	
Exhibit F	2/29/00	•
Exhibit G	2/29/00	
	2/29/00	
2-Network Elements & 1 Other Services	2/29/00	
2	2/29/00	
3	2/29/00	
4	2/29/00	
5	2/29/00	
9	2/29/00	
7	2/29/00	
∞	2/29/00	
6	2/29/00	
10	2/29/00	

Attachment 10-Residence

for NewSouth

NewSouth BellSouth Standard Interconnection Agreement

Attachment	Section	Version	Planned Activities
Name/Number	Number	Date	
	11	2/29/00	
	12	2/29/00	
	13	2/29/00	
	14	2/29/00	
	15	2/29/00	
	16	2/29/00	
	17	2/29/00	
	Exhibit A	2/29/00	
	Exhibit B	2/29/00	
	Exhibit C	2/29/00	
3-Local Interconnection		2/29/00	
	2	2/29/00	
	3	2/29/00	
	4	2/29/00	
	\$	2/29/00	
	9	2/29/00	
	7	2/29/00	
	8	2/29/00	
	Exhibit A	2/29/00	
	Exhibit B	2/29/00	
	Exhibit C	2/29/00	
	Exhibit D	2/29/00	
	Exhibit E	2/29/00	
4-Physical Collocation	_	2/29/00	
	2	2/29/00	
	3	2/29/00	
	4	2/29/00	
	5	2/29/00	
	9	2/29/00	
	7	2/29/00	
	8	2/29/00	

Attachment 10-Residence

Version 1Q00:3/6/00

for NewSouth

NellSouth Standard Interconnection Agreement

Attachment	Continu	1/2	D
Member	Section	noisia	Figured Activities
Name/Number	Number	Date	
	6	2/29/00	
	10	2/29/00	
	11	2/29/00	
	12	2/29/00	
	13	2/29/00	
	14	2/29/00	
	Exhibit A	2/29/00	
	Exhibit B	2/29/00	
5-Access to Numbers &		2/29/00	
Number Portability	1		
	2	2/29/00	
	3	2/29/00	
	4	2/29/00	
	5	2/29/00	
	9	2/29/00	
	L	2/29/00	
	8	2/29/00	
	Exhibit A	2/29/00	
6-Ordering/Provisioning	1	2/29/00	
	2	2/29/00	
	3	2/29/00	
7-Billing & Billing Accuracy Certification		2/29/00	
	2	2/29/00	
	3	2/29/00	
	4	2/29/00	
	5	2/29/00	
	9	2/29/00	
	7	2/29/00	
	Exhibit A	2/29/00	
8-ROW/Conduits/PoleAtt	1	2/29/00	

Attachment 10-Residence

Version 1Q00:3/6/00

Attachment 10-Residence

AGREEMENT IMPLEMENTATION TEMPLATE (Residence)

for NewSouth

BellSouth Standard Interconnection Agreement

Planned Activities																	
Version Date	2/29/00	2/29/00	2/29/00	2/29/00	2/29/00	2/29/00	2/29/00	2/29/00	2/29/00	2/29/00	2/29/00	2/29/00	2/29/00	2/29/00	2/29/00	2/29/00	
Section Number	Pre-Ordering	Ordering	Provisioning	Maint/Repair	Billing	Opr Svcs/DA	E911	Trunk Grp Perf	Collocation	Appendix A	Appendix B	Appendix C					
Attachment Name/Number	9-Perf Measurement												10-Executive Summary		11-Disaster Recovery		

AGREEMENT IMPLEMENTATION TEMPLATE (Business)

for NewSouth

BellSouth Standard Interconnection Agreement

Agreement Expiration Date:	Account Manager Tel No:	Planned Activities																												
		Version	Date																											
Date:		Section No.		1	2	3	4	 9	7	8	6	10	11	12	13	14	15	16	17	18	19	20	21	22	23	24	25	26 .		1
Agreement Effective Date:	Account Manager:	Attachment	Name	Terms/Conditions PartA																									Terms/Conditions Part B	1-Resale

Attachment 10-Business

AGREEMENT IMPLEMENTATION TEMPLATE (Business)

'n

for NewSouth BellSouth Standard Interconnection Agreement

Attachment	Section No.	Version	Planned Activities
Name		Date	
	2		
	3		
	4		
	5		
	9		
	4		
	&		
	6		
	10		
	11		
	12		
	13		
	Exhibit A		
	Exhibit B		
	Exhibit C		
	Exhibit D		
	Exhibit E		
	Exhibit F		
	Exhibit G		
	Exhibit H		
2-Network Elements & Other Services			
	2		
	3		
	4		
	\$		
	9		
	7		
	80		
	6		
	10		

Attachment 10-Business

AGREEMENT IMPLEMENTATION TEMPLATE (Business)

for NewSouth BellSouth Standard Interconnection Agreement

Attachment	Section No.	Version	Planned Activities
Name		Date	
	11		
	12		
	13		
	14		
	15		
	16		
	17		
	Exhibit A		
	Exhibit B		
	Exhibit C		
3-Local Interconnection	1		
	2		
	3		
	4		
	5		
	9		
	7		
	8		
	Exhibit A		
4-Physical Collocation	1		
	2		
	3		
	4		
	5		
	9		
	7		
	80		
	6		
	10		
	11		
	12		

AGREEMENT IMPLEMENTATION TEMPLATE (Business) for

for NewSouth BellSouth Standard Interconnection Agreement

Attachment	Section No.	Version	Planned Activities
Name		Date	
	13		
	14		
	Exhibit A		
	Exhibit B		
5-Access to Numbers & Number Portability	-		
	2		
	3		
	4		
	5		
	9		
	7		
	8		
	Exhibit A		
6-Ordering/Provisioning			
	2		
	3		
7-Billing & Billing Accuracy Certification	1		
	2		
	3		
	4		
	5		
	9		
	7		
	Exhibit A		
8-ROW/Conduits/PoleAtt			
9-Perf Measurement	Pre-Ordering		
	Ordering		
	Provisioning		
	Maint/Repair		

AGREEMENT IMPLEMENTATION TEMPLATE (Business)

for NewSouth

BellSouth Standard Interconnection Agreement

	_	-	- -		$\overline{}$				
Planned Activities						•			
Version	Date								
Section No.		Billing	Opr Svcs/DA	E911	Trunk Grp Perf	Collocation	Appendix A	Appendix B	Appendix C
Attachment	Name								

Attachment 11 BellSouth Disaster Recovery Plan

2000 BELLSOUTH

DISASTER RECOVERY PLANNING

For

CLECS

CONTENTS PAGE 4 1.0 Purpose 4 2.0 Single Point of Contact 4 3.0 Identifying the Problem 5 Site Control 3.1 6 3.2 **Environmental Concerns** 4.0 The Emergency Control Center (ECC) 6 7 5.0 Recovery Procedures 7 5.1 CLEC Outage 7 5.2 BellSouth Outage 8 5.2.1 Loss of Central Office 8 5.2.2 Loss of a Central Office with Serving Wire Center Functions 8 5.2.3 Loss of a Central Office with Tandem Functions 9 5.2.4 Loss of a Facility Hub 5.3 Combined Outage (CLEC and BellSouth Equipment 9 9 6.0 T1 Identification Procedures 10 7.0 Acronyms

1.0 PURPOSE

In the unlikely event of a disaster occurring that affects BellSouth's long-term ability to deliver traffic to a Competitive Local Exchange Carrier (CLEC), general procedures have been developed to hasten the recovery process. Since each location is different and could be affected by an assortment of potential problems, a detailed recovery plan is impractical. However, in the process of reviewing recovery activities for specific locations, some basic procedures emerge that appear to be common in most cases.

These general procedures should apply to any disaster that affects the delivery of traffic for an extended time period. Each CLEC will be given the same consideration during an outage and service will be restored as quickly as possible.

This document will cover the basic recovery procedures that would apply to every CLEC.

2.0 SINGLE POINT OF CONTACT

When a problem is experienced, regardless of the severity, the BellSouth Network Management Center (NMC) will observe traffic anomalies and begin monitoring the situation. Controls will be appropriately applied to insure the sanity of BellSouth's network; and, in the event that a switch or facility node is lost, the NMC will attempt to circumvent the failure using available reroutes.

BellSouth's NMC will remain in control of the restoration efforts until the problem has been identified as being a long-term outage. At that time, the NMC will contact BellSouth's Emergency Control Center (ECC) and relinquish control of the recovery efforts. Even though the ECC may take charge of the situation, the NMC will continue to monitor the circumstances and restore traffic as soon as damaged network elements are revitalized.

The telephone number for the BellSouth Network Management Center in Atlanta, as published in Telcordia's National Network Management Directory, is 404-321-2516.

3.0 IDENTIFYING THE PROBLEM

During the early stages of problem detection, the NMC will be able to tell which CLECs are affected by the catastrophe. Further analysis and/or first hand observation will determine if the disaster has affected CLEC equipment only; BellSouth equipment only or a combination. The initial restoration activity will be largely determined by the equipment that is affected.

Once the nature of the disaster is determined and after verifying the cause of the problem, the NMC will initiate reroutes and/or transfers that are jointly agreed upon by the affected CLECs' Network Management Center and the BellSouth NMC. The type and percentage of controls used will depend upon available network capacity. Controls necessary to stabilize the situation will be invoked and the NMC will attempt to re-establish as much traffic as possible.

For long term outages, recovery efforts will be coordinated by the Emergency Control Center (ECC). Traffic controls will continue to be applied by the NMC until facilities are re-established. As equipment is made available for service, the ECC will instruct the NMC to begin removing the controls and allow traffic to resume.

3.1 SITE CONTROL

In the total loss of building use scenario, what likely exists will be a smoking pile of rubble. This rubble will contain many components that could be dangerous. It could also contain any personnel on the premises at the time of the disaster. For these reasons, the local fire marshal with the assistance of the police will control the site until the building is no longer a threat to surrounding properties and the companies have secured the site from the general public.

During this time, the majority owner of the building should be arranging for a demolition contractor to mobilize to the site with the primary objective of reaching the cable entrance facility for a damage assessment. The results of this assessment would then dictate immediate plans for restoration, both short term and permanent.

In a less catastrophic event, i.e., the building is still standing and the cable entrance facility is usable, the situation is more complex. The site will initially be controlled by local authorities until the threat to adjacent property has diminished. Once the site is returned to the control of the companies, the following events should occur.

An initial assessment of the main building infrastructure systems (mechanical, electrical, fire & life safety, elevators, and others) will establish building needs. Once these needs are determined, the majority owner should lead the building restoration efforts. There may be situations where the site will not be totally restored within the confines of the building. The companies must individually determine their needs and jointly assess the cost of permanent restoration to determine the overall plan of action.

Multiple restoration trailers from each company will result in the need for designated space and installation order. This layout and control is required to maximize the amount of restoration equipment that can be placed at the site, and the priority of placements.

Care must be taken in this planning to insure other restoration efforts have logistical access to the building. Major components of telephone and building equipment will need to be removed and replaced. A priority for this equipment must also be jointly established to facilitate overall site restoration. (Example: If the AC switchgear has sustained damage, this would be of the highest priority in order to regain power, lighting, and HVAC throughout the building.)

If the site will not accommodate the required restoration equipment, the companies would then need to quickly arrange with local authorities for street closures, rights of way or other possible options available.

3.2 ENVIRONMENTAL CONCERNS

In the worse case scenario, many environmental concerns must be addressed. Along with the police and fire marshal, the state environmental protection department will be on site to monitor the situation.

Items to be concerned with in a large central office building could include:

- 1. Emergency engine fuel supply. Damage to the standby equipment and the fuel handling equipment could have created "spill" conditions that have to be handled within state and federal regulations.
- 2. Asbestos containing materials that may be spread throughout the wreckage. Asbestos could be in many components of building, electrical, mechanical, outside plant distribution, and telephone systems.
- 3. Lead and acid. These materials could be present in potentially large quantities depending upon the extent of damage to the power room.
- 4. Mercury and other regulated compounds resident in telephone equipment.
- 5. Other compounds produced by the fire or heat.

Once a total loss event occurs at a large site, local authorities will control immediate clean up (water placed on the wreckage by the fire department) and site access.

At some point, the companies will become involved with local authorities in the overall planning associated with site clean up and restoration. Depending on the clean up approach taken, delays in the restoration of several hours to several days may occur.

In a less severe disaster, items listed above are more defined and can be addressed individually depending on the damage.

In each case, the majority owner should coordinate building and environmental restoration as well as maintain proper planning and site control.

4.0 THE EMERGENCY CONTROL CENTER (ECC)

The ECC is located in the Colonnade Building in Birmingham, Alabama. During an emergency, the ECC staff will convene a group of pre-selected experts to inventory the damage and initiate corrective actions. These experts have regional access to BellSouth's personnel and equipment and will assume control of the restoration activity anywhere in the nine-state area.

In the past, the ECC has been involve with restoration activities resulting from hurricanes, ice storms and floods. They have demonstrated their capabilities during these calamities as well as

during outages caused by human error or equipment failures. This group has an excellent record of restoring service as quickly as possible.

During a major disaster, the ECC may move emergency equipment to the affected location, direct recovery efforts of local personnel and coordinate service restoration activities with the CLECs. The ECC will attempt to restore service as quickly as possible using whatever means is available; leaving permanent solutions, such as the replacement of damaged buildings or equipment, for local personnel to administer.

Part of the ECC's responsibility, after temporary equipment is in place, is to support the NMC efforts to return service to the CLECs. Once service has been restored, the ECC will return control of the network to normal operational organizations. Any long-term changes required after service is restored will be made in an orderly fashion and will be conducted as normal activity.

5.0 RECOVERY PROCEDURES

The nature and severity of any disaster will influence the recovery procedures. One crucial factor in determining how BellSouth will proceed with restoration is whether or not BellSouth's equipment is incapacitated. Regardless of who's equipment is out of service, BellSouth will move as quickly as possible to aid with service recovery; however, the approach that will be taken may differ depending upon the location of the problem.

5.1 CLEC OUTAGE

For a problem limited to one CLEC (or a building with multiple CLECs), BellSouth has several options available for restoring service quickly. For those CLECs that have agreements with other CLECs, BellSouth can immediately start directing traffic to a provisional CLEC for completion. This alternative is dependent upon BellSouth having concurrence from the affected CLECs.

Whether or not the affected CLECs have requested a traffic transfer to another CLEC will not impact BellSouth's resolve to re-establish traffic to the original destination as quickly as possible.

5.2 BELLSOUTH OUTAGE

Because BellSouth's equipment has varying degrees of impact on the service provided to the CLECs, restoring service from damaged BellSouth equipment is different. The outage will probably impact a number of Carriers simultaneously. However, the ECC will be able to initiate immediate actions to correct the problem.

A disaster involving any of BellSouth's equipment locations could impact the CLECs, some more than others. A disaster at a Central Office (CO) would only impact the delivery of traffic to and from that one location, but the incident could affect many Carriers. If the Central Office is a Serving Wire Center (SWC), then traffic from the entire area to those Carriers served from that switch would also be impacted. If the switch functions as an Access Tandem, or there is a tandem in the building, traffic from every CO to every CLEC could be interrupted. A disaster that destroys a facility hub could disrupt various traffic flows, even though the switching equipment may be unaffected.

The NMC would be the first group to observe a problem involving BellSouth's equipment. Shortly after a disaster, the NMC will begin applying controls and finding re-routes for the

completion of as much traffic as possible. These resources may involve delivering traffic to alternate Carriers upon receiving approval from the CLECs involved. In some cases, changes in translations will be required. If the outage is caused by the destruction of equipment, then the ECC will assume control of the restoration.

5.2.1 Loss of a Central Office

When BellSouth loses a Central Office, the ECC will

- a) Place specialists and emergency equipment on notice;
- b) Inventory the damage to determine what equipment and/or functions are lost;
- c) Move containerized emergency equipment and facility equipment to the stricken area, if necessary;
- d) Begin reconnecting service for Hospitals, Police and other emergency agencies; and
- e) Begin restoring service to CLECs and other customers.

5.2.2 Loss of a Central Office with Serving Wire Center Functions

The loss of a Central Office that also serves as a Serving Wire Center (SWC) will be restored as described in section 5.2.1.

5.2.3 Loss of a Central Office with Tandem Functions

When BellSouth loses a Central Office building that serves as an Access Tandem and as a SWC, the ECC will

- a) Place specialists and emergency equipment on notice;
- b) Inventory the damage to determine what equipment and/or functions are lost;
- c) Move containerized emergency equipment and facility equipment to the stricken area, if necessary;
- d) Begin reconnecting service for Hospitals, Police and other emergency agencies;
- e) Re-direct as much traffic as possible to the alternate access tandem (if available) for delivery to those CLECs utilizing a different location as a SWC;
- f) Begin aggregating traffic to a location near the damaged building. From this location, begin re-establishing trunk groups to the CLECs for the delivery of traffic normally found on the direct trunk groups. (This aggregation point may be the alternate access tandem location or another CO on a primary facility route.)
- g) Begin restoring service to CLECs and other customers.

5.2.4 Loss of a Facility Hub

In the event that BellSouth loses a facility hub, the recovery process is much the same as above. Once the NMC has observed the problem and administered the appropriate controls, the ECC will assume authority for the repairs. The recovery effort will include

- a) Placing specialists and emergency equipment on notice;
- b) Inventorying the damage to determine what equipment and/or functions are lost;
- c) Moving containerized emergency equipment to the stricken area, if necessary;
- d) Reconnecting service for Hospitals, Police and other emergency agencies; and
- e) Restoring service to CLECs and other customers. If necessary, BellSouth will aggregate the traffic at another location and build temporary facilities. This alternative would be viable for a location that is destroyed and building repairs are required.

5.3 COMBINED OUTAGE (CLEC AND BELLSOUTH EQUIPMENT)

In some instances, a disaster may impact BellSouth's equipment as well as the CLECs'. This situation will be handled in much the same way as described in section 5.2.3. Since BellSouth and the CLECs will be utilizing temporary equipment, close coordination will be required.

6.0 T1 IDENTIFICATION PROCEDURES

During the restoration of service after a disaster, BellSouth may be forced to aggregate traffic for delivery to a CLEC. During this process, T1 traffic may be consolidated onto DS3s and may become unidentifiable to the Carrier. Because resources will be limited, BellSouth may be forced to "package" this traffic entirely differently then normally received by the CLECs. Therefore, a method for identifying the T1 traffic on the DS3s and providing the information to the Carriers is required.

7.0 ACRONYMS

CO - Central Office (BellSouth)

DS3 - Facility that carries 28 T1s (672 circuits)

ECC - Emergency Control Center (BellSouth)

CLEC - Competitive Local Exchange Carrier

NMC - Network Management Center

SWC - Serving Wire Center (BellSouth switch)

T1 - Facility that carries 24 circuits

Hurricane Information

During a hurricane, BellSouth will make every effort to keep CLECs updated on the status of our network. Information centers will be set up throughout BellSouth Telecommunications. These centers are not intended to be used for escalations, but rather to keep the CLEC informed of network related issues, area damages and dispatch conditions, etc.

Hurricane-related information can also be found on line at http://www.interconnection.bellsouth.com/network/disaster/dis_resp.htm. Information concerning Mechanized Disaster Reports can also be found at this website by clicking on CURRENT MDR REPORTS or by going directly to http://www.interconnection.bellsouth.com/network/disaster/mdrs.htm.

BST Disaster Management Plan

BellSouth maintenance centers have geographical and redundant communication capabilities. In the event of a disaster removing any maintenance center from service another geographical center would assume maintenance responsibilities. The contact numbers will not change and the transfer will be transparent to the CLEC.

Attachment 12

Bona Fide Request and New Business Requests Process

BONA FIDE REQUEST AND NEW BUSINESS REQUESTS PROCESS

- 1.0 The Parties agree that NewSouth is entitled to order any Network Element, Interconnection option, service option or Resale Service required to be made available by the Communications Act of 1934, as modified by the Telecommunications Act of 1996 (the "Act"), FCC requirements or State Commission requirements. NewSouth also shall be permitted to request the development of new or revised facilities or service options, which are not required by the Act. Procedures applicable to requesting the addition of such facilities or service options are specified in this Attachment 12.
- Bona Fide Requests ("BFR") are to be used when NewSouth makes a request of BellSouth to provide a new or modified network element, interconnection option, or other service option pursuant to the Act that was not previously included in the Agreement. New Business Requests ("NBRs") are to be used when NewSouth makes a request of BellSouth to provide a new or custom capability or function to meet NewSouth's business needs that was not previously included in the Agreement. The BFR/NBR process is intended to facilitate the two-way exchange of information between NewSouth and BellSouth, necessary for accurate processing of requests in a consistent and timely fashion.
- 3.0 A BFR/NBR shall be submitted in writing by NewSouth and shall specifically identify the required service date, technical requirements, space requirements and/or such specifications that clearly define the request such that BellSouth has sufficient information to analyze and prepare a response. Such a request also shall include a NewSouth's designation of the request as being (i) pursuant to the Telecommunications Act of 1996 (i.e. a "BFR") or (ii) pursuant to the needs of the business (i.e. a "NBR"). The request shall be sent to NewSouth's Account Executive.
- 4.0 NewSouth may cancel a BFR or NBR at any time. If NewSouth cancels the request more than three (3) business days after submitting it, NewSouth shall pay BellSouth's reasonable and demonstrable costs of processing and/or implementing the BFR or NBR up to the date of cancellation. If NewSouth does not cancel a BFR or NBR, NewSouth shall pay BellSouth's reasonable and demonstrable costs of processing and implementing the request.
- Within fifteen (15) business days of its receipt of a BFR or NBR from NewSouth, BellSouth shall respond to NewSouth by providing a preliminary analysis of such Interconnection, Network Element, or other facility or service option that is the subject of the BFR or NBR. The preliminary analysis shall confirm that BellSouth will either offer access to the Interconnection, Network Element, or other facility or service

option, or provide an explanation of why it is not technically feasible and/or why the request does not qualify as an Interconnection, Network Element, or is not otherwise required to be provided under the Act.

- 6.0 If BellSouth determines that the Interconnection, Network Element, or other facility or service option that is the subject of the BFR is technically feasible, BellSouth shall propose a firm price and a detailed implementation plan within forty (40) business days after receipt of the BFR. BellSouth may, but shall not be required, to provide a firm time and cost proposal for a NBR.
- 7.0 Within thirty (30) business days after its receipt of (i) a refusal of BellSouth to provide a BFR or NBR price quote, or (ii) the BFR or NBR price quote and implementation plan from BellSouth, NewSouth must either confirm or cancel its order for such facility or service option. If it believes such quote is not consistent with the requirements of the Act, NewSouth may at that time seek FCC or state Commission arbitration of its request, as appropriate. Any such arbitration applicable to Network Elements and/or Interconnection shall be conducted in accordance with standards prescribed in Section 252 of the Act.
- Unless NewSouth agrees otherwise, all prices shall be consistent with the pricing principles of the Act, FCC and/or the State Commission.
- 9.0 If either Party to a BFR or NBR believes that the other Party is not requesting, negotiating, or processing the Bona Fide Request in good faith, or disputes a determination, or price or cost quote, such Party may seek FCC or state Commission resolution of the dispute, as appropriate.
- 10.0 Upon agreement to the terms of a BFR or NBR, an amendment to the Agreement may be required.